

## Thameslink, Southern and Great Northern Franchise

Train Service Requirements

December 2015

#### **TSGN Franchise – Train Service Requirements**

#### 1. Interpretations and Definitions

- 1.1 Unless otherwise stated each Train Service Requirement shall be interpreted in accordance with the Franchise Agreement between the Secretary of State and the Franchisee dated
- 1.2 Unless otherwise stated words and expressions defined in the Franchise Agreement shall have the same meaning in each Train Service Requirement.
- 1.3 In the Train Service Requirement unless the context otherwise requires, the following words and expressions shall have the following meanings:

## "Average Journey Time Specification"

means the specification of the average journey time for Eligible Passenger Services between designated stations as set out in the Non-Core Specification Tables (in columns headed "Journey times", under the sub-heading "Average") and in the Core Specification Tables (in columns headed "Average journey times"), with the average journey time being specified in minutes;

#### "Capacity Requirement"

means a requirement on the Franchisee to ensure that its Timetable and Train Plan include at least a specified minimum number of:

- (a) vehicles comprised within
  Passenger Services of Specified
  Descriptions arriving at Specified
  Stations during each Morning Peak
  and Morning High Peak, as more
  particularly described in the
  relevant table labelled "Morning
  Peak" of each Capacity Table; and
- (b) vehicles comprised within Passenger Services of Specified Descriptions departing from Specified Stations during each Evening Peak and Evening High Peak, as more particularly described in the relevant table labelled "Evening Peak" of each Capacity Table;

#### "Capacity Tables"

means each of the "Morning Peak" and "Evening Peak" tables labelled CR1, CR2, CR3, CR4, CR5 and CR6;

## "Core Specification Tables"

means the tables labelled "Thameslink Core: Minimum Service Specification and

First and Last Train Specification" (Table 1: Northbound and Table 2: Southbound);

#### "Direct Service"

means a Passenger Service from A to B where passengers may board the Passenger Service at A and subsequently alight at B, without needing to change from one train to another;

#### "Eligible Non-Stop Passenger Service"

means a Passenger Service that is required to:

- (a) operate from London Victoria to Gatwick Airport (or vice versa, as they case may be) without calling at any intermediate stations (but which for the avoidance of doubt may also serve stations south of Gatwick Airport); and
- (b) complete the journey from London Victoria to Gatwick Airport (or vice versa, as the case may be) within the specified Maximum Journey Time Requirement.

#### "Evening High Peak"

means the period between 1730 and 1829 (inclusive) during a Weekday;

## "First and Last Train Specification"

means the specification of the latest permissible time at which the First Train of the Day may be operated, and the earliest permissible time at which the Last Train of the Day may be operated, between designated stations as specified in the Core Specification Tables and the Non-Core Specification Tables;

#### "First Train"

means the first Passenger Service in such Day that counts towards the obligation of the Franchisee to operate a specified number of trains per day;

#### "Last Train"

means the last Passenger Service in such Day that counts towards the obligation of the Franchisee to operate a specified number of trains per day;

## "Maximum Journey Time Specification"

means the specification of the maximum permissible journey time for Eligible Passenger Services between designated stations as set out in the Non-Core Specification Tables (in columns headed "Journey times", under the sub-heading "Maximum") and the Core Specification Tables (in columns headed "Maximum

journey times"), with the maximum journey time being specified in minutes;

## "Minimum Service Specification"

#### means:

- (a) in relation to any Non-Core Specification Table, the specification of:
  - (i) the minimum number of Direct Services that are required to be operated between designated stations during specified intervals as set out in columns headed "Total"; and
  - (ii) the minimum number of Direct Services that are required to be operated during each hour between designated stations during specified intervals as set out in columns headed "TPH",

in each case specified in the relevant tables referred to in each of TSR4, TSR5, TSR6 and TSR7 (each such Direct Service as referred to in paragraph (a) to be known as an "Eligible Passenger Service") and provided that it is a requirement of the Minimum Service Specification that:

- (x) each Eligible Passenger Service shall comply with any Maximum Journey Time Specification that applies in respect of such Eligible Passenger Service; and
- (y) the average journey time of all the Eligible Passenger Services which comply with the Maximum Journey Time Specification referred to in (a)(x) above shall be no greater than any Average Journey Time Specification that applies in respect of such services;
- (b) in relation to any Core Specification Table, the specification of:
  - (i) the minimum number of Direct Services that are required to be operated between designated stations and/or the frequency

with which such services are to be operated;

(ii) the days of the week, and times of the day, during which such services are required to be operated,

in each case specified in the relevant Core Specification Table (each such Direct Service as referred to in paragraph (b) to be known as an "Eligible Core Passenger Service") and provided that it is a requirement of the Minimum Service Specification that:

(x) each Eligible Core Passenger Service shall comply with any Maximum Journey Time Specification that applies in respect of such Eligible Core Passenger Service and shall call (as a minimum) St **Pancras** at International, Farringdon, Thameslink, Blackfriars and the designated intermediate stations identified in the column headed "Principal stations served" in the relevant row of the Specification Tables (provided that a Passenger Service which is scheduled to pass but not call at City Thameslink at any time before 0515 on a Monday, or during any period commencing at 2315 on a Weekday and ending at 0515 on the following day, or at any time after 2115 on a Saturday, or at any time on a Sunday, shall be regarded as being compliant); and

(y) the average journey time of all the Eligible Core Passenger Services which comply with the Maximum Journey Time Specification referred to in (a)(x) above shall be no greater than any Average Journey Time Specification that applies in respect of such services;

"Morning High Peak"

means the period between 0800 and 0859 (inclusive) during a Weekday;

#### "Non-Core Specification Tables"

means the tables referred to as "ECML 1", "ECML 2", "MML 1", "MML 2", "South East London and Kent 1", "South East London and Kent 2", "BML Inner 1", "BML Inner 2", "BML Outer 1", "BML Outer 2", "West Coastway", "East Coastway" and "Milton Keynes to Croydon" in paragraph 1.4 below;

## "Relevant London Station"

means, in relation to a Passenger Service that operates through the Core Thameslink Route (as such term is defined in Schedule 6.3 (TLP/TRSP Related Provisions) of the Franchise Agreement):

- (a) St Pancras International, in the case of a southbound service;
- (b) London Bridge, in the case of a northbound service that is scheduled to call there;
- (c) London Blackfriars, in the case of a northbound service that is not scheduled to call at London Bridge.

## "Service Level Commitments"

means the documents labelled "SLC1", "SLC2" and "SLC3", which set out certain requirements referred to in paragraphs 3.1, 4.1 and 5.1 below;

## "Specified Description"

means the description of the Passenger Service, specified in Column (2) of the Capacity Tables headed "Specified description";

#### "Specified Station"

means the stations specified in Column (2) of the Capacity Tables headed "Specified station"; and

#### "TSR"

means the Train Service Requirement.

1.4 In this Train Service Requirement, references to tables comprising Non-Core Specification Tables are to be construed as follows:

#### "ECML1"

means the tables labelled "Table 1: Northbound" and "Table 2: Southbound" in the worksheet headed "ECML1: East Coast Mainline Minimum Service Specification and First and Last Train Specification";

#### "ECML2"

means the tables labelled "Table 1: Northbound" and "Table 2: Southbound" in the worksheet labelled "ECML2: East Coast Mainline Minimum Service Specification and First and Last Train Specification";

"MML1"

means the tables labelled "Table 1: Northbound" and "Table 2: Southbound" in the worksheet labelled "MML1: Midland Mainline Minimum Service Specification and First and Last Train Specification";

"MML2"

means the tables labelled "Table 1: Northbound" and "Table 2: Southbound" in the worksheet labelled "MML2: Midland Mainline Minimum Service Specification and First and Last Train Specification";

"South East London and Kent 1"

means the tables labelled "Table 1: Southbound" and "Table 2: Northbound" in the worksheet labelled "South East London and Kent 1: Minimum Service Specification and First and Last Train Specification";

"South East London and Kent 2"

means the tables labelled "Table 1: Southbound" and "Table 2: Northbound" in the worksheet labelled "South East London and Kent 2: Minimum Service Specification and First and Last Train Specification";

"BML Inner 1"

means the tables labelled "Table 1: Southbound" and "Table 2: Northbound" in the worksheet labelled "BML Inner 1: Brighton Main Line Minimum Service Specification (Inner Services only) and First and Last Train Specification";

"BML Inner 2"

means the tables labelled "Table 1: Southbound" and "Table 2: Northbound" in the worksheet labelled "BML Inner 2: Brighton Main Line Minimum Service Specification (Inner Services only) and First and Last Train Specification";

"BML Outer 1"

means the tables labelled "Table 1: Southbound", "Table 2: Northbound", "Table 3: Redhill Outbound" and "Table 4: Redhill Inbound" in the worksheet labelled "BML Outer 1: Brighton Main Line Minimum Service Specification (Outer Services only) and First and Last Train Specification";

"BML Outer 2"

means the tables labelled "Table 1: Southbound", "Table 2: Northbound", "Table 3: Redhill Outbound" and "Table 4: Redhill Inbound" in the worksheet labelled "BML Outer 2: Brighton Main Line Minimum

Service Specification (Outer Services only) and First and Last Train Specification";

#### "West Coastway"

means the tables labelled "Table 1: Westbound", "Table 2: Westbound", "Table 3: Eastbound" and "Table 4: Eastbound" in the worksheet labelled "West Coastway: Minimum Service Specification and First and Last Train Specification";

#### "East Coastway"

means the tables labelled "Table 1: Eastbound" and "Table 2: Westbound" in the worksheet labelled "East Coastway: Minimum Service Specification and First and Last Train Specification"; and

## "Milton Keynes to Croydon"

means the tables labelled "Table 1: Northbound", "Table 2: Northbound", "Table 3: Southbound" and "Table 4: Southbound" in the worksheet labelled "Milton Keynes-Croydon: Minimum Service Specification and First and Last Train Specification".

#### 2. Construction and Interpretation

#### 2.1 Franchise Agreement Obligation

- (a) Without limitation to any provision of the Franchise Agreement the Franchisee is required:
  - (i) pursuant to paragraph 1 of Schedule 1.1 (Service Development) of the Franchise Agreement to seek Train Slots from Network Rail in accordance with the applicable TSR with the intention that the working timetable issued by Network Rail at the end of its timetable development process is consistent with such TSR; and
  - (ii) to comply with the provisions of paragraph 7 of Schedule 1.1 (Service Development) of the Franchise Agreement including by providing the capacity necessary to enable the Capacity Requirement to be complied with.

#### 2.2 Days and Times of Day

- (a) Except to the extent the context otherwise requires,
  - references to a day mean the period commencing at 0200 on one day and ending at 0159 on the following day and references to particular days of the week shall be construed accordingly;

- (ii) references to periods of time and periods of days include the times and days such periods start and finish; and
- (iii) all references to time are to the twenty-four hour clock.

#### 2.3 Interval or Frequency of Services

- (a) Where a Minimum Service Specification specifies a number of Passenger Services per hour, indicated by "TPH" in the relevant column heading, an hour will be interpreted to be a clock face hour beginning at "xx00" and ending at "xx59" and frequency shall be measured at the location indicated in the relevant table.
- (b) For the purposes of complying with the Minimum Service Specification set out in any of the Non-Core Specification Tables, a Passenger Service that is scheduled to arrive at or depart from a particular station:
  - (i) at 55, 56, 57, 58 or 59 minutes past the hour may instead be treated as though it were scheduled to arrive at (or depart from) that station in the immediately following hour (i.e. a service scheduled to arrive at 0857 may be treated as though it were scheduled to arrive in the 0900-0959 hour); and
  - (ii) on the hour, or at 01, 02, 03 or 04 minutes past the hour, may instead be treated as though it were scheduled to arrive at (or depart from) that station in the immediately preceding hour (i.e. a service scheduled to arrive at 0902 may be treated as though it were scheduled to arrive in the 0800-0859 hour),

provided that any single Passenger Service is counted only once.

#### 2.4 References to "London"

- (a) In tables labelled "ECML 1", "ECML 2", "MML 1" and "MML 2", references to:
  - (i) Direct Services to or from "London" are to be construed as meaning Direct Services between the specified station and either London Kings Cross, Moorgate or London St Pancras International;
  - (ii) arrival times in "London" are to be construed as meaning the arrival time at London Kings Cross, Moorgate or London St Pancras International; and
  - (iii) departure times from "London" are to be construed as meaning the departure time from London Kings Cross, Moorgate or London St Pancras International.
- (b) In tables labelled "South East London and Kent 1", "South East London and Kent 2", "BML Inner 1", "BML Inner 2", "BML Outer 1" and "BML Outer 2", references to:

- (i) direct services to or from "London" are to be construed as meaning Direct Services between each specified station and any one or more of London Victoria, London Bridge or London Blackfriars;
- (ii) arrival times in "London" are to be construed as meaning the arrival time at London Victoria, London Bridge or London Blackfriars (or the earliest arrival time at one of these stations, in the case of a train which calls at both London Blackfriars and London Bridge); and
- (iii) departure times from "London" are to be construed as meaning the departure time from London Victoria, London Bridge or London Blackfriars (or the latest departure time from one of these stations, in the case of a train which calls at both London Blackfriars and London Bridge).

## 2.5 Interpretation of requirements in tables headed "Thameslink Core"

For the purposes of the tables labelled "Thameslink Core":

- (a) Passenger Services between any particular "Origin" and "Destination" station operate at "half-hourly intervals" if the interval between the scheduled arrival time of each pair of consecutive services at the Relevant London Station is no less than 25 minutes and no more than 35 minutes;
- (b) The references to days of the week or times of the day in the columns headed "Operation" are included only for convenience and are not intended to supplement, negate or otherwise amend the requirements as specified in the columns headed "Service arrives at destination station".

#### 2.6 Requirements in tables labelled "BML Outer 1" and "BML Outer 2"

In the tables labelled "BML Outer 1" and "BML Outer 2", the requirements specified in Tables 3 and 4 are in addition to those specified in Tables 1 and 2. Accordingly a Passenger Service that is counted towards meeting a requirement specified in Table 1 or Table 2 may not also be counted towards meeting a requirement specified in Table 3 or Table 4.

#### 2.7 Requirements in tables labelled "West Coastway"

In the tables labelled "West Coastway", the requirements specified in Tables 2 and 4 are in addition to those specified in Tables 1 and 3. Accordingly a Passenger Service that is counted towards meeting a requirement specified in Table 1 or 3 may not also be counted towards meeting a requirement specified in Table 2 or 4.

## 2.8 References to "inner services", "outer services" and "Cambridge Express"

(a) For the purposes of the tables labelled "ECML 1" and "ECML 2":

- (i) a "Cambridge Express Service" is a Passenger Service which is scheduled to call at London Kings Cross, Cambridge and no more than two other stations between London Kings Cross and Cambridge (and which may, but is not required to, be extended to serve stations north of Cambridge);
- (ii) an "ECML Outer Service" is a Passenger Service, not being a Cambridge Express Service, which is scheduled to call at or pass through Welwyn North station and which, between Welwyn North and Finsbury Park, is not scheduled to call at any station other than Welwyn North, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park;
- (iii) an "ECML Inner Service" is a Passenger Service which is neither a Cambridge Express Service nor an ECML Outer Service;
- (iv) both ECML Inner Services and ECML Outer Services may be counted towards meeting the requirements specified in the tables in relation to Finsbury Park, Potters Bar, Hatfield and Welwyn Garden City;
- only ECML Inner Services may be counted towards meeting the requirements specified in the rows labelled "ECML inner services only except Finsbury Park, Potters Bar, Hatfield and Welwyn Garden City";
- (vi) only ECML Outer Services may be counted towards meeting the requirements specified in the rows labelled "ECML outer services only";
- (vii) only Cambridge Express Services may be counted towards meeting the requirements specified in the tables in the rows labelled "CEx only";
- (viii) only Cambridge Express Services and ECML Outer Services may be counted towards meeting the requirements specified for Waterbeach, Ely, Littleport, Downham Market, Watlington and Kings Lynn.
- (b) For the purposes of the tables labelled "South East London & Kent 1" and "South East London & Kent 2":
  - (i) a "**Kent Outer Service**" is a Passenger Service which is not scheduled to call at any stations between Bromley South and Elephant and Castle (other than those two stations themselves);
  - (ii) a "**Kent Inner Service**" is a Passenger Service which is not a Kent Outer Service;
  - (iii) only Kent Outer Services may be counted towards meeting the requirements specified in the tables in the rows labelled "Kent Outer Services only";

- (iv) only Kent Inner Services may be counted towards meeting the requirements specified in the tables in the rows labelled "Kent Inner Services only";
- (v) Passenger Services operating on routes covered by the sheets labelled "BML Inner 1" and "BML Inner 2" may not be counted towards the requirements specified for Elephant and Castle in the tables labelled "South East London & Kent 1" and "South East London & Kent 2".
- (c) For the purposes of the tables labelled "BML Inner 1", "BML Inner 2", "BML Outer 1" and "BML Outer 2":
  - (i) a "**BML Outer Service**" is a Passenger Service which either:
    - (A) calls at or passes through Coulsdon South station, or by-passes Coulsdon South station via the Quarry Lines, and which is not scheduled to call at any station between Coulsdon South and London Victoria, London Blackfriars or (as the case may be) London Bridge other than Coulsdon South, Purley, South Croydon, East Croydon, Norwood Junction, Clapham Junction, London Victoria, New Cross Gate, London Bridge, Tulse Hill, Herne Hill, Elephant and Castle or London Blackfriars; or
    - (B) calls at or passes through Sanderstead station and which is not scheduled to call at any station between Sanderstead and London Victoria or (as the case may be) London Bridge other than Sanderstead, Purley, South Croydon, East Croydon, Norwood Junction, Clapham Junction, London Victoria or London Bridge.
  - (ii) a "BML Inner Service" is a Passenger Service which is not a BML Outer Service;
  - (iii) both BML Inner Services and BML Outer Services may be counted towards meeting the requirements specified in the tables labelled "BML Inner 1" and "BML Inner 2" for Purley, South Croydon, East Croydon, Norwood Junction, New Cross Gate and Clapham Junction:
  - (iv) only BML Inner Services may be counted towards meeting the requirements specified in those tables for stations other than Purley, South Croydon, East Croydon, Norwood Junction, New Cross Gate and Clapham Junction;
  - (v) only BML Outer Services may be counted towards meeting the requirements specified in the tables labelled "BML Outer 1" and "BML Outer 2";

- (vi) Passenger Services operating on routes covered by the sheets labelled "South East London & Kent 1" and "South East London & Kent 2" may not be counted towards meeting the requirements specified for Elephant and Castle in the sheets labelled "BML Inner 1" and "BML Inner 2"; and
- (vii) Passenger Services operating between East Croydon and Elephant and Castle may not be counted towards meeting the requirements specified for Elephant and Castle in the sheet labelled "BML Inner 1".

#### 2.9 References to "passing through" particular stations

For the purposes of paragraph 2.8 above, a Passenger Service is treated as "passing through" a particular station ("Station A") only if (in addition to the Passenger Service being scheduled to pass through that station) it is possible for passengers to travel on that Passenger Service between a London Station and a station that is further away from London than Station A, where "London Station" means either London Kings Cross, London St Pancras International, London Victoria or London Bridge.

#### 2.10 References to "Gatwick Airport (Express Services)"

In the tables labelled "BML Outer 1" and "BML Outer 2", the requirement in relation to rows labelled "Gatwick Airport (Express Services)" is to provide at least the specified number of Eligible Non-Stop Passenger Services between London Victoria and Gatwick Airport. Such services may additionally be counted towards meeting the requirements specified in the rows labelled "Gatwick Airport (including Gatwick Express)" in those tables.

#### 2.11 Bank Holidays

Except to the extent that the Secretary of State may agree otherwise:

- the Franchisee shall not be required to deliver the TSR on Christmas Day or Boxing Day, but on Boxing Day shall be required to operate services in accordance with paragraph 2.11(f) below;
- (b) in relation to New Year's Day the Franchisee shall be required to deliver the same service specified to be operated on Sundays in the Timetable then in force, supplemented by the following Passenger Services to be operated in the early hours of the morning of New Year's Day:
  - (i) twelve services departing from London Victoria, at 20 minute intervals between approximately 0105 and 0505, to East Croydon calling at all intermediate stations via Norbury;
  - eight services departing from London Blackfriars, at 30 minute intervals between approximately 0030 and 0430 to Sutton calling at all intermediate stations via Hackbridge;

- (iii) eight services departing from Farringdon, at 30 minute intervals between approximately 0030 and 0430, to St Albans calling at all intermediate stations;
- (iv) four services departing from London Kings Cross, at 30 minute intervals between approximately 0030 and 0230 (inclusive), to Hertford North calling at all intermediate stations;
- (v) four services departing from London Kings Cross, at 30 minute intervals between approximately 0030 and 0230, to Welwyn Garden City calling at all intermediate stations;
- (vi) three services departing from London Kings Cross, at hourly intervals between approximately 0001 and 0300, calling at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin and all stations to Peterborough;
- (vii) three services departing from London Kings Cross, at hourly intervals between approximately 0001 and 0300, calling at Finsbury Park, Stevenage, Hitchin, Letchworth Garden City, Baldock, Ashwell and Morden, Royston and Cambridge;
- (c) in relation to all other Bank Holidays the Franchisee shall be required to deliver the same service specified to be operated on Saturdays in the Timetable then in force;
- (d) in relation to the weekdays between Christmas Day and New Year's Day that are not Bank Holidays the Franchisee shall be required to deliver the same service specified to be operated on Saturdays in the Timetable then in force, and shall:
  - (i) use all reasonable endeavours to supplement that service with such additional Passenger Services during the Peak (in particular between London Bridge and the Sussex Coast and between Bedford and London Blackfriars) and such additional early-morning services (in particular in relation to BML Inner Services (as such term is defined in paragraph 2.8(c) above)) as are necessary to fulfil the Franchisee's obligations under clause 5.2 (General Obligations) of the Franchise Agreement; and
  - (ii) operate services in longer formations (compared to the Train Plan that would the Franchisee would normally operate on Saturdays) to the extent necessary to comply with the Franchisee's obligations under paragraph 7 (Planning to meet Target Passenger Demand) of Schedule 1.1 (Service Development ) of the Franchise Agreement.
- (e) on 24<sup>th</sup> December the Franchisee shall not be required to operate any Passenger Services after 2200 hours, and may make such reasonable reductions in the Passenger Services after 1300 hours

to reflect reduced demand and the need to ensure that the cessation of train services occurs in an operationally efficient manner as the Secretary of State may agree, such agreement not to be unreasonably withheld; and

- (f) in relation to Boxing Day, from approximately 0830 (departures from Origin) until approximately 2100 (arrivals at Destination) except where otherwise stated, the Franchisee is required to provide:
  - (i) services between London Victoria and Gatwick Airport at half-hourly intervals, to operate from approximately 0500 (departures from Origin) until approximately 2300 (arrivals at Destination);
  - (ii) services between London Victoria and Three Bridges at half-hourly intervals, calling intermediately at Clapham Junction, East Croydon and Gatwick Airport. Alternate services shall be extended to call at Haywards Heath and Brighton to provide an hourly service between those stations and London Victoria. The requirements for these services may be combined with those between London Victoria and Gatwick Airport; and
  - (iii) stopping services at half-hourly intervals between:
    - (A) London Victoria and Sutton via Selhurst; and
    - (B) London Victoria and East Croydon via Crystal Palace.

#### 2.12 Bank holidays: exceptions

The requirements specified in paragraph 2.11(b)(i), paragraph 2.11(d) (in so far as that paragraph relates to Southern Services) and paragraph 2.11(f) shall not apply prior to the Combined Effective Date.

#### 2.13 No prioritisation of any aspect of any of the TSRs

In the case of each TSR, the obligations of the Franchisee are in relation to the entire TSR and the Franchisee shall not be permitted to prioritise the delivery of any element of the TSR to the detriment of any other. In any case where the effect of the Minimum Service Specification is to require the operation of more than one Passenger Service before the time specified as the latest permissible time at which the First Train of the Day may be operated, or after the time specified as the earliest permissible time at which the Last Train of the Day may be operated, the Minimum Service Specification shall continue to apply to its full extent.

#### 2.14 Passenger Services not required by the TSRs

(a) Subject to compliance with all relevant provisions of the Franchise Agreement the Franchisee is not restricted by any of the TSR from providing any specification of Passenger Services (including departures from any stations, passenger carrying capacity and departures prior to the latest specified time for the First Train or

- after the earliest specified time for the Last Train) additional to that required for the purposes of complying with such TSR.
- (b) For the avoidance of doubt, any specification of such additional Passenger Services may be disregarded for the purposes of determining compliance with any Average Journey Time Specification or Maximum Journey Time Specification contained within this TSR.

#### 3. Train Service Requirement - TSR1

- 3.1 TSR1 consists of the following elements:
  - (a) a requirement to operate Passenger Services in accordance with the requirements specified in the Service Level Commitment labelled "SLC1"; and
  - (b) the Capacity Requirement in the tables labelled "CR1".

#### 4. Train Service Requirement - TSR2

- 4.1 TSR2 consists of the following elements:
  - (a) a requirement to operate Passenger Services in accordance with the requirements specified in the Service Level Commitment labelled "SLC2"; and
  - (b) the Capacity Requirement in the tables labelled "CR2".

#### 5. Train Service Requirement – TSR3

- 5.1 TSR3 consists of the following elements:
  - (a) a requirement to operate Passenger Services in accordance with the requirements specified in the Service Level Commitments labelled "SLC2" and "SLC3"; and
  - (b) the Capacity Requirement specified in the tables labelled "CR3".

#### 6. Train Service Requirement - TSR4

- 6.1 TSR4 consists of the following elements:
  - (a) the Capacity Requirements specified in the table labelled "CR4"; and
  - (b) the Minimum Service Specification (including the Average Journey Time Specification and Maximum Journey Time Specification contained therein) and the First and Last Train Specification described in the following tables (as defined in paragraph 1.4 above):
    - (i) ECML 1;
    - (ii) MML 1;

- (iii) South East London and Kent 1;
- (iv) BML Inner 1;
- (v) BML Outer 1;
- (vi) West Coastway;
- (vii) East Coastway; and
- (viii) Milton Keynes to Croydon.

#### 7. Train Service Requirement - TSR5

- 7.1 TSR5 consists of the following elements:
  - (a) the Capacity Requirement specified in the tables labelled "CR5"; and
  - (b) the Minimum Service Specification (including the Average Journey Time Specification and Maximum Journey Time Specification contained therein) and the First and Last Train Specification described in the following tables (as defined in paragraph 1.4 above):
    - (i) ECML 1;
    - (ii) MML 1;
    - (iii) South East London and Kent 1;
    - (iv) BML Inner 1;
    - (v) BML Outer 1;
    - (vi) West Coastway;
    - (vii) East Coastway; and
    - (viii) Milton Keynes to Croydon.

#### 8. Train Service Requirement - TSR6

- 8.1 TSR6 consists of the following elements:
  - (a) the Capacity Requirement specified in the tables labelled CR6; and
  - (b) the Minimum Service Specification (including the Average Journey Time Specification and Maximum Journey Time Specification contained therein) and the First and Last Train Specification described in the following tables (as defined in paragraph 1.4 above):
    - (i) ECML 2;
    - (ii) MML 2;

- (iii) South East London and Kent 2;
  (iv) BML Inner 2;
  (v) BML Outer 2;
  (vi) West Coastway;
  (vii) East Coastway;
  (viii) Milton Keynes to Croydon; and
- (ix) the Core Specification Tables.
- 8.2 For the avoidance of doubt, a Passenger Service that contributes to meeting a requirement specified in the Core Specification Tables may also be counted as contributing to meeting relevant requirements specified in the other tables listed in paragraph 8.1(b)(i) to (viii) above.
- 9. Train Service Requirement TSR7
- 9.1 TSR7 consists of the following elements:
  - (a) the Capacity Requirement specified in the tables labelled "CR6"; and
  - (b) the Minimum Service Specification (including the Average Journey Time Specification and Maximum Journey Time Specification contained therein) and the First and Last Train Specification described in the following tables (as defined in paragraph 1.4 above):
    - (i) ECML 2;
    - (ii) MML 2;
    - (iii) South East London and Kent 2;
    - (iv) BML Inner 2;
    - (v) BML Outer 2;
    - (vi) West Coastway;
    - (vii) East Coastway;
    - (viii) Milton Keynes to Croydon; and
    - (ix) the Core Specification Tables.
- 9.2 For the avoidance of doubt, a Passenger Service that contributes to meeting a requirement specified in the Core Specification Tables may also be counted as contributing to meeting relevant requirements specified in the other tables listed in paragraph 9.1(b)(i) **Error! Reference source not found.**to (viii) above.

## SLC1

## TSGN FRANCHISE (FORMER THAMESLINK/GN SERVICES)

# SERVICE LEVEL COMMITMENT SLC1

Version 2

#### Part 1

#### **Service Level Commitment – General Provisions**

#### 1. CONSTRUCTION

The following provisions of this Part 1 to this SLC1 shall apply in respect of the Service Level Commitment set out in Part 2 to this SLC1.

#### 2. DAYS AND TIMES OF DAY

- 2.1 Except to the extent the context otherwise requires, references to a day mean the period commencing at 02:00 on one day and ending at 01:59 on the following day and references to Weekdays and particular days of the week shall be construed accordingly.
- 2.2 References to periods of times and periods of days include the times and days such periods start and finish.
- 2.3 All references to time are to the twenty-four hour clock.

#### 3. SERVICES

- 3.1 Except where expressly indicated to the contrary, references to services, all services or any part or any proportion of services are to be construed as references to the Passenger Services (or the relevant part or proportion thereof) required to be included by the Franchisee in its Timetable pursuant to Schedule 1.1 of the Franchise Agreement and do not include such additional railway passenger services as the Franchisee may be permitted to provide from time to time under this Agreement.
- 3.2 Except where expressly indicated to the contrary, where an interval or frequency is specified for a service, such specification shall apply at the departure point for the relevant service.
- 3.3 Except where expressly indicated to the contrary, all services are to run in both directions and the requirements of the Service Level Commitment (including any interval between services, frequency of service or stopping pattern) are to apply in each direction.
- 3.4 Except where expressly indicated to the contrary, references to "Summer" shall mean that period in any year from the last Saturday in May until the third Saturday in September and "Winter" is defined as any period other than "Summer."

#### 4. STATIONS

4.1 Except where expressly indicated to the contrary and subject always to compliance with the other provisions of the Service Level Commitment (including any maximum Journey Times) nothing in the Service Level Commitment shall prevent services which are required to be included by, or on behalf of, the Franchisee in the Timetable pursuant to Schedule 1.1 calling at any stations which are not specified in the Service Level Commitment or any relevant part of it.

4.2 The Franchisee may designate any station, at which a service calls, as a Request Stop Station, where such service need only call when a passenger or intending passenger wishes to join such service or leave such service at such station. Any such designation by the Franchisee shall be clearly advertised in advance to intending passengers (including in the Timetable).

#### 5. CONNECTIONS

- 5.1 Except where expressly indicated to the contrary, a service which is required to be included in the Timetable shall be provided so as to enable travel between the stations specified without passengers being required to change train. Where a service may be provided by a Connection or where a Connection is required to be provided between two services, such service or combination of services may be provided without change of train being required.
- 5.2 Except where expressly indicated to the contrary, where Connections are required to be provided by the Franchisee, the Franchisee shall ensure that the interval or waiting period between the two relevant services is of a sufficient duration to allow passengers a reasonable period of time to transfer between such services.
- 5.3 Except where expressly indicated to the contrary, where services are subject to maximum Journey Times, such Journey Times shall not apply where the services are required to be, or may be, provided by Connections.

#### 6. PEAK, OFF-PEAK AND CONTRA-PEAK

- 6.1 Peak means, in relation to any service a Weekday Service (except Christmas and Bank Holidays) which arrives at London (London Bridge, London Blackfriars, City Thameslink, Farringdon, London Kings Cross or London St. Pancras International), between 07:00 and 09:59 (the "Morning" Peak) and departs from London (London Bridge, London Blackfriars, City Thameslink, Farringdon, London Kings Cross or London St. Pancras International), between 16:00 and 18:59 (the "Evening" Peak)
- 6.2 Contra-Peak means in relation to any service, those services originating from the point of destination of Peak services during the Peak periods defined in paragraph 6.1.
- 6.3 Off-Peak means in relation to any service any period which is not Peak.

#### 7. NOT USED

#### 8. SERVICE INTERVALS

- 8.1 Except where expressly indicated to the contrary, where the Franchisee is required to include within the Timetable, services for a specified period (which period shall commence and end with an Early Service or Late Service), with a specified interval between each such service during such period, then the following shall apply:
- (a) the minimum number of services to be included in the Timetable in such period shall be determined in accordance with the following formula:

$$S = \frac{TM}{IM}$$

#### where:

- S is the minimum number of services to be included in the Timetable;
- TM is the total minutes in the specified period; and
- IM is the number of minutes in the specified interval, rounded down to the nearest whole number;
- (b) the interval between any two services during such period may be extended by:
  - (i) five minutes; or,
  - (ii) Thameslink Route only: services may be extended by six minutes, when required to travel through the St. Pancras International London Blackfriars section
  - (iii) if greater, an amount of minutes equivalent to one-sixth of the specified interval between each service rounded down to the nearest whole number,

subject to the other provisions of this paragraph 8.1 and such extension not exceeding ten minutes;

(c) the interval between any two services may be reduced, subject to the other provisions of this paragraph 8.1, the interval between each service required in accordance with paragraph 8.1(a) for such period, by such amount as the Franchisee may determine;

#### 8. SERVICE INTERVALS Contd.

(d) the interval between services during such specified period shall be such that, for any period which occurs during such specified period and has a duration equivalent to:

$$((4 \times SI) + E)$$
 minutes

where:

SI is equal to the specified interval between such services; and

E is equal to the extension permitted to such interval under paragraph 8.1(b), at least four services shall be included in the Timetable in such period; and

- (e) the intervals between the start of such period of time and the first service in such period (save where such period of time starts with an Early Service) and between the last service in such period and the end of such period (save where such period of time ends with a Late Service) shall each be no more than the interval between each service specified for such period and (save where such period of time starts and/or ends with an Early Service/Late Service) one of such intervals shall be no more than half such interval between each service.
- 8.2 Except where expressly indicated to the contrary, where one period ends and another period begins immediately thereafter and services are specified at different intervals for each such period, the maximum interval between the last service in the first period and the first service in the second period shall be the longer of the maximum intervals allowed for each such period.

## 9. Key Output 0 Services – General Description of North and South of London Blackfriars

9. Services on First Capital Connect SLC 1 Routes F and G may be extended South of London Blackfriars towards Sevenoaks, as part of the Key Output Zero specified outputs.

For the avoidance of doubt:

- (a) services (from Kentish Town and all stations North of that location) extended South of London Blackfriars shall be classified as London and South Eastern Railways ("Southeastern") services from London Blackfriars and at all stations South of that location, **except that** services originating from or terminating at Elephant & Castle to or from stations North of Elephant & Castle shall be classified as Thameslink Southern & Great Northern services.
- (b) services (from Sevenoaks and other points of origin South of Blackfriars, normally operated by London and South Eastern Railways) extended North of London Blackfriars shall be classified as Thameslink Southern & Great Northern services from London Blackfriars and all stations North of that location, **except that** services originating from or terminating at City Thameslink to or from stations South of City Thameslink shall be classified as London and South Eastern Railways ("Southeastern") services.

#### **INDEX TO ROUTES**

#### **ROUTE**

#### (<u>GN</u>)

- A. London (Moorgate/ Kings Cross) to Hertford North and Letchworth Garden City.
- B. London (Moorgate/ Kings Cross) to Welwyn Garden City.
- C. London (Moorgate/ Kings Cross) and Stevenage to Huntingdon and Peterborough.
- D. London (Kings Cross) and Stevenage to Cambridge.
- E. London (Kings Cross) and Kings Lynn.

#### (Thameslink)

- F. London Blackfriars to Bedford.
- G. London Blackfriars to St Albans and Luton.
- H. London (St. Pancras International/London Bridge) to Gatwick Airport and Brighton.
- I. London (St. Pancras International) to Wimbledon and Sutton.
- J. London (St. Pancras International) to Hackbridge and Sutton.

## ROUTE A: LONDON (MOORGATE / KINGS CROSS) TO HERTFORD NORTH AND LETCHWORTH GARDEN CITY.

#### 1. Route Definition

1.1 Services shall be provided between:

London (Moorgate), Old Street, Essex Road, Highbury and Islington and Drayton Park, or

London Kings Cross and

Finsbury Park and Hertford North, calling at Harringay, Hornsey, Alexandra Palace, Bowes Park, Palmers Green, Winchmore Hill, Grange Park, Enfield Chase, Gordon Hill, Crews Hill, Cuffley, and Bayford.

1.2 Services in paragraph 1.1 shall extend to or originate from Letchworth Garden City, calling at Watton-at-Stone, Stevenage and Hitchin, as described below.

## 2. <u>Service Pattern Monday to Friday: London (Moorgate / Kings Cross) to Hertford</u> North.

2.1 Between the Early Service and 09:05, a half-hourly interval service shall be provided, calling at the stations described in paragraph 1.1. After 23:15 an hourly interval may apply.

Between 07:00 and 08:45 an additional four services shall be provided from London (Moorgate) to Hertford North which, subject to the following provisions, shall call at the stations described in paragraph 1.1. All four services may omit to call at Bowes Park, Grange Park, Crews Hill and Bayford. The first two services may additionally omit to call at Harringay and Hornsey.

Between 07:45 and 08:45, three further services shall be provided from London (Moorgate) to Gordon Hill, at 20-minute intervals. These services shall call at all stations mentioned in paragraph 1.1, except for Crews Hill, Cuffley and Bayford, and two of these services may omit to call at Harringay and Hornsey.

2.2 Between 09:05 and 16:29 a 20-minute interval service shall be provided calling at the stations described in paragraph 1.1. One interval may extend up to 30 minutes.

Between 16:15 and 16:30, one additional service shall be provided from London (Moorgate) to Gordon Hill, calling at the intermediate stations described in paragraph 1.1. This service may omit to call at Bowes Park and Grange Park.

- 2.3 The Early Service and the following two services shall originate from London (Kings Cross). Two of these services shall extend to Stevenage, calling at Watton-at-Stone.
- 2.4 Between 06:40 and 22:00 all services shall originate from Moorgate and call at all intermediate stations between Moorgate and Finsbury Park.

- 2.5 Five of the services specified in 2.1 between 06:45 and 09:15 shall extend to Stevenage calling at intermediate stations, two of which shall be extended further to Letchworth Garden City calling at intermediate stations.
- 2.6 At hourly intervals between 09:45 and 15:15, services specified in paragraph 2.2 shall extend to Letchworth Garden City, calling at intermediate stations.

One further service between 09:05 and 09:45 shall so extend.

- 2.7 Between 15:00 and 16:29, three services specified in paragraph 2.2 shall extend to Stevenage calling at Watton-at-Stone. The last of these three services may omit to call at Harringay, Hornsey, Crews Hill and Bayford
- 2.8 Services from London between 16:30 and 19:44 shall call as shown below:

Para 2.8 part	Service timed from Moorgate	No. of Services	Maximum Interval
(a)	Moorgate, Old Street, Essex Road, Highbury & Islington, Drayton Park, Finsbury Park, Palmers Green, Winchmore Hill, Enfield Chase, Gordon Hill.	23	15 minutes, with one exception of 20 minutes
(b)	Harringay, Hornsey	8	30 minutes
(c)	Alexandra Palace	18	20 minutes, with one exception of 25 minutes
(d)	Bowes Park, Grange Park	15	15 minutes, with one exception of 20 minutes
(e)	Crews Hill, Bayford	7	30 minutes, with one exception of 35 minutes
(f)	Cuffley, Hertford North	17	15 minutes, with one exception of 20 minutes
(g)	Watton-at-Stone, Stevenage	6	30 minutes with two exceptions of 35 minutes
(h)	Hitchin, Letchworth Garden City	2	Between 18:45 and 19:30, two services shall be extended to Letchworth Garden City, calling at all intermediate stations

2.9 Between 19:45 and 22:00, services shall be provided from London (Moorgate) to Hertford North, calling at the stations described in paragraph 1.1, at 20-minute intervals. The first of these services shall extend to Stevenage, calling additionally at Watton-at-Stone. Thereafter, every third service shall extend to Letchworth Garden City, calling at the stations described in paragraph 1.2.

Between 20:00 and 22:00 another two services shall be extended from Hertford North to Letchworth Garden City, calling at the stations described in paragraph 1.2.

An additional service shall be provided between Moorgate and Finsbury Park between 20:00 and 20:30, calling at the stations described in paragraph 1.1.

- 2.10 Between 22:00 and 23:30 a half-hourly service shall be provided from London (Kings Cross) to Hertford North, calling at the stations described in paragraph 1.1, with two services at an hourly interval being extended to Letchworth Garden City, calling at the stations described in paragraph 1.2.
- 2.11 Between 23:30 and 01:10, two further services (including the Late Service) shall be provided at an hourly interval from London (Kings Cross) to Hertford North, calling at the stations described in paragraph 1.1. The Late Service need not call at Harringay, Hornsey, Bowes Park, Grange Park, Crews Hill, Bayford and Watton-on-Stone. These services shall be extended to Letchworth Garden City calling at Hitchin.

## 3. <u>Service Pattern Monday to Friday: Hertford North to London (Moorgate / Kings Cross).</u>

- 3.1 Between 04:45 and 05:00, the Early Service shall originate from Letchworth Garden City and operate to London Kings Cross, calling at the stations described in paragraph 1.1. In addition, between the Early Service and 06:45, two further services from Hertford North to London (Kings Cross / Moorgate) shall originate from Letchworth Garden City.
- 3.2 Between the Early Service and 06:45, and between 16:40 and 21:15, services shall be provided from Hertford North to London Moorgate at half-hourly intervals, calling at the stations described in paragraph 1.1. Two of these services departing from Hertford North between 17:00 and 18:00 shall originate from Stevenage, calling at Watton-at-Stone.

Between 06:00 and 06:15, one additional service shall be provided from Stevenage, calling at stations specified in paragraph 1.1. It need not call at Bayford, Crews Hill, Grange Park, Bowes Park, Harringay and Hornsey.

Between 16:40 and 19:40, six additional services shall be provided from Hertford North, calling at Cuffley, Gordon Hill, Winchmore Hill, Palmers Green, Finsbury Park and intermediate stations to London (Moorgate). The first such service shall original from Letchworth Garden City calling at the stations described in paragraph 1.2, and shall call additionally at Bayford, Crews Hill and Alexandra Palace. The second such service shall call additionally at Alexandra Palace. The other four shall originate from Stevenage, calling at Watton-at-Stone.

Between 17:15 and 19:30, six additional services shall be provided from Gordon Hill to London (Moorgate) at maximum intervals of 30 minutes, calling at stations specified in paragraph 1.1. The first need not call at Grange Park, Bowes Park, Harringay and Hornsey.

Between 19:45 and 20:15, one additional service shall be provided from Hertford North to Moorgate, calling at Cuffley, Gordon Hill, Enfield Chase, Winchmore Hill, Palmers Green, Finsbury Park and then intermediate stations to Moorgate.

3.3 Services departing from Hertford North towards London between 06:45 and 09:00 shall call at maximum intervals at the stations as shown below:

Para 3.7 part	Service timed at Hertford North	No. of Services	Maximum Interval
(a)	Letchworth Garden City, Hitchin	1	Not applicable
(b)	Stevenage, Watton-at-Stone	5	30 minutes, with one exception of 35 minutes permitted
(c)	Hertford North, Cuffley	12	20 minutes,

(d)	Bayford, Crews Hill	5	30 minutes, with one
			exception of 40 minutes
			permitted

- 3.4 Between 07:00 and 07:40, two additional services at a maximum interval of 30 minutes and between 08:10 and 09:30 five additional services at a maximum interval of 25 minutes shall be provided from Gordon Hill to London (Moorgate), calling at all intermediate stations. Two of these services may omit to call at Harringay and Hornsey.
- 3.5 Services departing from Gordon Hill towards London between 07:00 and 09:30, including those from Hertford North, shall call at maximum intervals at stations shown below:

Para 3.5 part	Services timed at Gordon Hill (07:00 - 09:30)	No. of Services	Maximum Interval
(a)	Gordon Hill, Enfield Chase, Finsbury Park, Drayton Park, Highbury & Islington, Essex Road, Old Street, London (Moorgate)	18	15 minutes
(b)	Grange Park	13	15 minutes, with one exception of 20 minutes permitted
(c)	Winchmore Hill, Palmers Green	14	15 minutes, with one exception of 21 minutes permitted
(d)	Bowes Park, Alexandra Palace	16	15 minutes, with one exception of 20 minutes permitted
(e)	Hornsey, Harringay	12	15 minutes, with two exceptions of 20 minutes permitted

- 3.6 Between 09:00 and 16:40, a 20-minute interval service shall be provided from Hertford North to London (Moorgate), calling at the stations described in paragraphs 1.1 and 1.2. One interval of 30 minutes shall be permitted before 09:30.
  - Between 16:15 and 16:30, one additional service shall be provided from Hertford North to London (Moorgate), calling at the stations described in paragraphs 1.1 This service may omit Bayford, Crews Hill, Grange Park, Bowes Park, Harringay and Hornsey.
- 3.7 Between 09:45 and 16:30 seven services shall originate from Letchworth Garden City at hourly intervals and call at all intermediate stations. Between 0900 and 1020, two of these services shall originate from Stevenage calling at Watton-at-Stone, at an hourly interval. Between 1620 and 1640 one service

- shall originate at Letchworth Garden City calling at the stations described in paragraph 1.2.
- 3.8 Between 21:15 and including the Late Service, three services at hourly intervals shall originate from Letchworth Garden City and call at all intermediate stations.
  - Two additional services shall be provided between Hertford North and London (Kings Cross) calling at the stations described in paragraph 1.1 to allow for a half-hourly interval between 21:15 and the Late Service.
- 3.9 All services between 21:30 until and including the Late Service shall terminate at London (Kings Cross).

#### 4. <u>Service Pattern: Saturdays</u>

- 4.1 Between and including the Early and Late Services, 40 services shall be provided from London (Kings Cross) to Hertford North, at half-hourly intervals, calling at all stations specified in paragraph 1.1. The Late service and one service departing between 23:15 and 23:30 shall be extended from Hertford North to Letchworth Garden City, calling at all intermediate stations.
- 4.2 Between and including the Early and Late Services, 38 services shall be provided from Hertford North to London (Kings Cross), at half-hourly intervals, calling at all stations specified in paragraph 1.1. One of these services need not call at Harringay.
- 4.3 Between 06:15 and 22:30 services shall be extended at hourly intervals from Hertford North to Stevenage, calling at Watton-at-Stone.
- 4.4 Between 07:15 and 23:45, services shall originate at hourly intervals from Stevenage, calling additionally at Watton-at-Stone. The Early, the second and the fourth services from Hertford North shall originate from Letchworth Garden City and call at Hitchin, Stevenage and Watton-at-Stone.

#### 5. <u>Service Pattern: Sundays</u>

- 5.1 Between and including the Early and Late Services, 36 services shall be provided at half-hourly intervals from London (Kings Cross) to Hertford North, calling at the stations in paragraph 1.1. One service an hour shall be extended to Stevenage, calling at Watton-at-Stone.
- 5.2 Between and including the Early and Late Services, 35 services shall be provided at half-hourly intervals from Hertford North to London (Kings Cross), calling at the stations in paragraph 1.1. One service per hour shall have originated from Stevenage, calling at Watton-at-Stone.
- 5.3 Two services shall be provided from Hertford North to London (Kings Cross), at half hourly intervals, the second one half an hour before the Early Service, calling at the stations specified in paragraph 1.1
- The Late service shall be provided from London (Kings Cross) to Letchworth Garden City, calling at all intermediate stations.

#### 6. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Hertford North	06:20	06:15	06:45
Hertford North to London (Moorgate / Kings Cross)	06:00	06:00	08:30

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Hertford North	01:00	00:45	23:15
Hertford North to London (Moorgate / Kings Cross)	23:30	23:15	23:30

#### 7. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Finsbury Park to Hertford North	37 minutes, with 12 exceptions of 38 minutes permitted.	38 minutes	38 minutes
Hertford North to Finsbury Park	36 minutes.	35 minutes	35 minutes

#### ROUTE B: LONDON (MOORGATE / KINGS CROSS) TO WELWYN GARDEN CITY.

#### 1. Route Definition

- 1.1 Services shall be provided between London (Moorgate) and Welwyn Garden City, calling at Old Street, Essex Road, Highbury & Islington, Drayton Park, Finsbury Park, Harringay, Hornsey, Alexandra Palace, New Southgate, Oakleigh Park, New Barnet, Hadley Wood, Potters Bar, Brookmans Park, Welham Green and Hatfield.
- 1.2 Services in paragraph 1.1 may extend to or originate from London (Kings Cross) as described below.

#### 2. <u>Service Pattern Monday to Friday: London Departures.</u>

2.1 Between the Early Service and 09:14 a half-hourly service shall be provided, calling at the stations described in paragraph 1.1. Between 06:00 and 06:20, one of these services shall depart from London Kings Cross, calling at the stations described in paragraph 1.1.

Between 08:15 and 08:30, one additional service shall be provided from London (Moorgate) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

Between 08:30 and 09:00 one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at Finsbury Park, Potters Bar, Hatfield and Welwyn Garden City.

- 2.2 Between 09:15 and 16:59, a 20 minute interval service shall be provided, calling at the stations described in paragraph 1.1. Two intervals of up to 30 minutes between services departing between 16:00 and 16:59 to Welwyn Garden City shall be permitted.
- 2.3 Between 17:00 and 19:29, 9 services shall be provided with a maximum interval of 20 minutes.
- 2.4 Between 16:45 and 19:50, six additional services shall be provided from London (Kings Cross) to Welwyn Garden City, calling at Finsbury Park, Potters Bar and Hatfield. These services may call additionally at Alexandra Palace, New Southgate, Oakleigh Park, New Barnet and Hadley Wood.

Between 15:30 and 15:45, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City calling at Finsbury Park, New Southgate, Oakleigh Park, New Barnet, Potters Bar and Hatfield.

- 2.5 Between 19:30 and 22:00, a half-hourly service shall be provided from London (Moorgate) to Welwyn Garden City, calling at the stations described in paragraph 1.1.
- 2.6 Between 22:00 and 23:59, a half-hourly service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations described in paragraph 1.1.

Between 00;00 and 01:15, two additional services shall be provided from London (Kings Cross) to Welwyn Garden City, at hourly interval, calling at the stations defined in paragraph 1.1,

#### 3. Service Pattern Monday to Friday: Welwyn Garden City Departures.

3.1 Between and including the Early Service and 08:59, ten services shall be provided to Moorgate at a maximum interval of 45 minutes, calling at the stations described in paragraph 1.1. Two departures between 07:20 and 08:20 need not call at Alexandra Palace, and one of these need not call at Hornsey and Harringay, provided that the maximum interval between services at these stations does not exceed 35 minutes.

Between 08:15 and 08:30, one additional service shall be provided to Moorgate, calling at the stations specified in paragraph 1.1.

3.2 Between 08:00 and 10:00, three additional services shall be provided at a maximum interval of one hour from Welwyn Garden City to London (Kings Cross), calling at Hatfield, Potters Bar and Finsbury Park. Two of these services shall also call at New Barnet, Oakleigh Park, New Southgate and Alexandra Palace. The other additional service shall also call at New Barnet.

Two additional services shall be provided to London (Kings Cross) at hourly intervals, calling at the stations described in paragraph 1.1. The second service shall depart 40 minutes after the Early Service.

Between 04:00 and 04:15, one additional service shall be provided to London (Kings Cross), calling at the stations described in paragraph 1.1.

- 3.3 Between 09:00 and 16:14, services shall be provided at 20-minute intervals, calling at the stations described in paragraph 1.1.
- 3.4 Between 16:15 and 21:14, a half-hourly service shall be provided calling at the stations described in paragraph 1.1.

Between 16:30 and 18:29, three additional services shall be provided to London (Kings Cross) at a maximum interval of 70 minutes, calling only at Hatfield, Potters Bar and Finsbury Park.

3.5 Between 21:15 and 23:00 a half-hourly service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations described in paragraph 1.1.

#### 4. Service Pattern: Saturdays

4.1 Between and including the Early and Late Services, 36 services shall be provided from London (Kings Cross) to Welwyn Garden City at half-hourly service intervals, calling at the stations specified in paragraph 1.1.

Between 00:30 and 00:45, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

Between 01:00 and 01:15, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

4.2 Between and including the Early and Late Services, 35 services shall be provided from Welwyn Garden City to London (Kings Cross) at half-hourly intervals, calling at the stations specified in paragraph 1.1.

Between 04:00 and 04:15, one additional service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations specified in paragraph 1.1.

Between 05:15 and 05:45, one additional service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations specified in paragraph 1.1.

#### 5. <u>Service Pattern: Sundays</u>

- 5.1 Between and including the Early and Late Services, 35 services shall be provided from London (Kings Cross) to Welwyn Garden City at half-hourly intervals calling at all stations specified in paragraph 1.1.
- 5.2 Between and including the Early and Late Services, 34services shall be provided from Welwyn Garden City to London (Kings Cross) at half-hourly intervals, calling at all stations specified in paragraph 1.1.

#### 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Saturday Su		Sunday
London (Moorgate / Kings Cross) to Welwyn Garden City	07:00	07:00	07:30
Welwyn Garden City to London (Moorgate / Kings Cross)	05:00	07:00	07:30

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Welwyn Garden City	23:30	23:30	23:30
Welwyn Garden City to London (Moorgate / Kings Cross)	22:45	22:45	22:45

#### 7. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Finsbury Park to Welwyn Garden City	36 minutes, with five exceptions of 37 minutes permitted	35 minutes, with three exceptions of 36 minutes and two exceptions of 41 minutes permitted	35 minutes, with seven exceptions of 38 minutes and one exception of 39 minutes permitted
Welwyn Garden City to Finsbury Park	35 minutes	35 minutes	35 minutes

## ROUTE C: LONDON (KINGS CROSS) AND STEVENAGE TO HUNTINGDON AND PETERBOROUGH.

#### 1. Route Definition

- 1.1 Services shall be provided between London (Kings Cross) and Peterborough (the Stopping Service), calling at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 1.2 Services shall be provided between London (Kings Cross) and Peterborough (the Semi-Fast Service), calling at Finsbury Park, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 1.3 The services in paragraphs 1.1 and 1.2 shall be planned in such a way to provide a regular interval service to and from London and stations between Hitchin and Peterborough.

# 2. <u>Service Pattern Monday To Friday (The Stopping Service): London (Kings Cross) Departures.</u>

- One additional service shall be provided between 05:15 and 05:30 calling at the stations described in paragraph 1.1.
- 2.2 Between and including the Early Service and 15:45, an hourly service shall be provided calling at the stations described in paragraph 1.1.
- 2.3 Two additional services shall be provided between 00:30 and 01:45 at hourly intervals calling at the stations described in paragraph 1.1. One of these services need not call at Arlesey and Sandy.

# 3. <u>Service Pattern Monday to Friday (the Stopping Service): Peterborough Departures.</u>

- 3.1 One additional service shall be provided between 03:15 and 03:30, which shall call at Huntingdon, St Neots, Biggleswade, Hitchin, Stevenage and Finsbury Park. This service shall not be subject to a maximum journey time.
- 3.2 Two additional services shall be provided between 04:00 and 07:00 at a maximum interval of 2 hours, calling at the stations described in paragraph 1.1. The second of these services may omit to call at Hatfield and Potters Bar.
- 3.3 Between and including the Early and Late Services, 13 services shall be provided from Peterborough to London (Kings Cross) at hourly intervals calling at the stations described in paragraph 1.1. There may be one extended interval of one hour 15 minutes between services departing from Peterborough after 20:00.

#### 4. Service pattern Monday to Friday (The Semi-Fast Service): London Departures

- 4.1 Between the Early Service and 16:29, an hourly service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2. The last of these services shall call additionally at Knebworth, but may terminate at Huntingdon.
- 4.2 Evening Peak semi-fast services from London (Kings Cross) to Peterborough shall be provided, calling at the stations described in paragraph 1.2, as shown in the following table:

Para 4.2 part	Services departing from Kings Cross to:	No. of Services	Service Requirement
(a)	Finsbury Park, Stevenage. Hitchin, Arlesey, Biggleswade, Sandy, St. Neots, Huntingdon, Peterborough	6	Half-hourly between 16:30 and 19:29, with one interval of up to 40 minutes permitted.  Welwyn North One of these services departing between 16:30 and 17;00 shall call.  Knebworth One service departing between 17:00 and 17:30, and another service departing between 18:30 and 19:00, shall call.
(b)	St. Neots, Huntingdon, Peterborough	6	Half-hourly between 16:30 and 19:29.  Hitchin and Biggleswade The first and fifth of these services shall call at Biggleswade. The other four of these services shall call at Hitchin.

- 4.3 Between 19:45 and 19:59 an additional service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2. This train will not be subject to a maximum journey time.
- 4.4 Between 20:00 and the Late Service, an hourly service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2.
- 4.5 Between 20:00 and 21:15, two additional services shall be provided at hourly intervals from London (Kings Cross) to Peterborough, calling at St Neots and Huntingdon. The later of these services shall call additionally at Biggleswade.

Between 22:00 and 23:15, two additional services shall be provided from London (King Cross) to Peterborough to arrive at hourly intervals, calling at Biggleswade, St Neots and Huntingdon only.

On a Saturday morning between 00:00 and 00:15 one additional service shall be provided from London (Kings Cross) to Peterborough, calling at Stevenage, Hitchin, Biggleswade, St Neots and Huntingdon.

# 5. <u>Service Pattern Monday to Friday (the Semi-Fast Service): Peterborough Departures.</u>

- 5.1 One additional service shall be provided between 05:00 and 05:15 calling at the stations described in paragraph 1.2.
- 5.2 Morning Peak semi-fast services from Peterborough to London (Kings Cross) shall be provided at the stations described in paragraph 1.2, as shown in the following table:

Para 5.2 part	Services departing from Peterborough to:	No. of Services	Service Requirement
(a)	Huntingdon, St. Neots, Sandy, Biggleswade, Arlesey, Hitchin, Stevenage, Finsbury Park, Kings Cross	6	Between 05:30 and 09:00, a half- hourly service shall be provided calling at the stations described in paragraph 1.2.  The first of these services need not call at Sandy, Biggleswade or Arlesey.  One interval of 45 minutes and an additional interval of 55 minutes shall be permitted for services departing from Peterborough.
(b)	Huntingdon, St. Neots, Kings Cross	5	Between 06:15 and 08:00, five additional services shall be provided at half-hourly intervals from Peterborough to London (Kings Cross), one interval may be extended to 40 minutes.  Three of these services shall depart Peterborough between 07:10 and 07:50.  Biggleswade The first and the fourth of these services shall call.  Stevenage

	Four of these services shall call.

- 5.3 One additional service shall be provided from Peterborough to London (Kings Cross) between 09:00 and 09:29, calling at Huntingdon, St Neots, Biggleswade and Stevenage.
  - One additional service shall be provided from Peterborough to London (Kings Cross) between 09:15 and 09:30, calling at the stations specified in paragraph 1.2. This service need not to call at Finsbury Park.
- 5.4 Between 09:30 and the Late Service, an hourly service shall be provided from Peterborough to London (Kings Cross), calling at the stations described in paragraph 1.2. Services departing Peterborough between 10:45 and 14:00 need not call at Arlesey. There may be one interval of up to 1 hour 15 minutes between 16:45 and 18:15.
- 5.5 One service shall depart Huntingdon between 17:45 and 18:15, calling at St. Neots, Hitchin, Stevenage and Finsbury Park.

#### 6. Service Pattern: Saturdays

- 6.1 Between and including the Early and Late Services, 19services (the Stopping Service) shall be provided at hourly intervals from London (Kings Cross) to Peterborough calling at all stations specified in paragraph 1.1. There may be one interval of 1 hour 15 minutes.
- 6.2 Between and including the Early and Late Services, 18 services (the Stopping Service) shall be provided at hourly intervals from Peterborough to London (Kings Cross) calling at all stations specified in paragraph 1.1
- 6.3 Between and including the Early and Late Services, 18services (the Semi-Fast Service) shall be provided at hourly intervals from London (Kings Cross) to Peterborough calling at the stations specified in paragraph 1.2.
  - Between 16:30 and 18:45, three additional services shall be provided from London (Kings Cross) to Peterborough at hourly intervals, calling at Biggleswade, St Neots and Huntingdon.
- 6.4 Between and including the Early and Late Services, 18 services (the Semi-Fast Service) shall be provided at hourly intervals from Peterborough to London (Kings Cross) calling at the stations specified in paragraph 1.2.
- 6.5 Three additional services shall depart from Peterborough to London (Kings Cross) between 08:00 and 10:15, at hourly intervals, calling at Huntingdon, St Neots and Biggleswade.
- 6.6 Between 03:00 and 04:15, two additional services may be provided from Peterborough to London (Kings Cross) calling at stations specified in paragraph 1.1, except Knebworth, Welwyn North and Harringay. One service may omit to call at Sandy, Arlesey, Welwyn Garden City, Hatfield, Potters Bar.

- 6.7 Between 05:00 and 05:30, one additional service shall be provided from Peterborough to London (Kings Cross) calling at stations specified in paragraph 1.1.
- 6.8 Between 09:15 and 09:45 a service to London (Kings Cross) defined in paragraph 6.2 may originate from Huntingdon, calling at stations specified in paragraph 1.1.
- 6.9 Not used
- 6.10 The Late service specified in 6.3 shall call at Finsbury Park, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 6.11 Until the end of the Thameslink/GN franchise the late stopping service from London Kings Cross may terminate at Hitchin, in which case an additional service shall be provided from London Kings Cross to Peterborough calling at the stations described in paragraph 1.2.

The additional service shall provide a connection not exceeding five minutes at Stevenage with the Late Stopping service.

#### 7. Service Pattern: Sundays

7.1 Between and including the Early and Late Services, 16 Semi-Fast services shall be provided at hourly intervals from London (Kings Cross) to Peterborough, calling at the stations specified in paragraph 1.2.

Between 16:15 and 18:45 three additional services shall be provided from London (Kings Cross) to Peterborough, at least thirty minutes apart, calling at Biggleswade, St Neots and Huntingdon. One of these services shall also call at Sandy.

One additional service shall be provided between 16:15 and 18:45 terminating at Royston Herts as part of Route D, calling at Finsbury Park, Potters Bar, Hatfield Herts, Welwyn Garden City, Welwyn North, Knebworth, Stevenage and Hitchin. This service may be extended to Cambridge.

7.2 Between and including the Early and Late Services, 17 Semi-Fast services shall be provided at hourly intervals from Peterborough to London (Kings Cross), calling at the stations specified in paragraph 1.2.

Between 09:00 and 11:30, three additional services shall be provided from Peterborough to London (Kings Cross) at hourly intervals, calling at Huntingdon, St Neots and Biggleswade.

Between 23:00 and 23:15, an additional service shall be provided between Peterborough and London (Kings Cross), calling at Huntingdon, St Neots, Sandy, Biggleswade, Arlesley, Hitchin, Stevenage and Welwyn Garden City.

7.3 Two additional services shall be provided from London (Kings Cross) after the Late Service, departing between 00:30 and 01:45 at a maximum interval of one hour, calling at all stations specified in paragraph 1.2, except that the second of these service need not call at Arlesey and Sandy.

### 8. <u>Early and Late Services</u>

### Early Service arrives at destination no later than:

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Peterborough (Stopping Service)	08:15	07:00	N/A
Peterborough to London (Kings Cross) (Stopping Service)	12:00	07:00	N/A
London (Kings Cross) to Stevenage and Peterborough (Semi-Fast Service)	07:45	07:45	09:45
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	06:45	07:15	07:30

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday		Sunday	
London (Kings Cross) to Peterborough (Stopping Service)	01:30	23:45	N/A	
Peterborough to London (Kings Cross) (Stopping Service)	22:15	22:15	N/A	
London (Kings Cross) to Stevenage and Peterborough (Semi-Fast Service)	23:15	23:15	23:15	
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	21:45	22:46	21:45	

## 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
-------	-----------------	----------	--------

London (Kings Cross) to Peterborough (Stopping Service)	1 hour 33 minutes, with one exceptions of 1 hour 37 minutes and two further exceptions of 1 hour 45 minutes permitted	1 hour 29 minutes, with two exceptions of 1 hour 34 minutes and one exception of 1 hour 40 minutes permitted	N/A
Peterborough to London (Kings Cross) (Stopping Service)	1 hour 34 minutes, with three exceptions of up to 1 hour 38 minutes and one further exception of 1 hour 44 minutes permitted	1 hour 32 minutes, with five exceptions of 1 hour 35 minutes and one exception of 1 hour 43 minutes permitted	N/A

## 9. <u>Maximum Journey Times Contd.</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Stevenage and Peterborough (Semi- Fast Service)	1 hour 19 minutes, with three exceptions of 1 hour 22 minutes and one exception of 1 hour and 26 minutes permitted	1 hour 17 minutes, with one exception of 1 hour 21 minutes and two exceptions of 1 hour 32 minutes permitted	1 hour 15 minutes, with three exceptions of 1 hour and 17 minutes, two exceptions of one hour 28 minutes, one exception of 1 hour 33 minutes and two exceptions of 1 hour 50 minutes permitted
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	1 hour 20 minutes, with two exceptions of 1 hour 25 minutes permitted	1 hour 22 minutes, with two exceptions of 1 hour 31 minutes permitted	1 hour 16 minutes, with five exceptions of 1 hour 19 minutes and two exceptions of 1 hour 24 minutes permitted

#### ROUTE D: LONDON (KINGS CROSS) AND STEVENAGE TO CAMBRIDGE.

#### 1. Route Definition

- 1.1 Stopping Services shall be provided between London (Kings Cross) and Cambridge, calling at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Letchworth, Baldock, Ashwell & Morden, Royston, Meldreth, Shepreth, and Foxton.
- 1.2 Semi-Fast Services shall be provided between London (Kings Cross) and Cambridge, calling at Finsbury Park, Stevenage, Hitchin, Letchworth Garden City, Baldock and Royston.
- 1.3 **Limited Stops** shall be provided for the service specified in paragraph 1.2, at Ashwell & Morden, Meldreth, Shepreth and Foxton.
- 1.4 Services may be provided in conjunction with those specified in Route E.

# 2. <u>Service Pattern Monday to Friday (Stopping Service): London (Kings Cross)</u> <u>Departures.</u>

- 2.1 Between and including the Early Service and 16:29, 11 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at the stations described in paragraph 1.1.
- 2.2 Between 16:30 and 19:29 Evening Peak services shall be provided as described in the table below:

Para 2.2 part	Services departing from Kings Cross to:	No. of Services	Service Requirement
2.2.	Stations as described in paragraph 1.1.	7	Services shall depart London (Kings Cross) between 16:30 and 19:29 at half-hourly intervals. One interval of 20 minutes shall be permitted between 16:30 and 17:00.  The service departing between 17:00 and 17:30 need not call at Baldock or Ashwell & Morden.  Four of these services need not call at Potters Bar. The service departing between 17:30 and 18:00 need not call at Meldreth, Shepreth and Foxton.

2.2A Between 18:45 and 19:00 an additional service shall be provided from Royston to Cambridge, calling at all intermediate stations. This service may

- be formed from a portion which detaches at Royston from a London (Kings Cross) to Kings Lynn service that is required to be provided under Route E.
- 2.3 Between 19:30 up to and including the Late Service, four additional services shall be provided at hourly intervals from London (Kings Cross) to Cambridge, calling at the stations in paragraph 1.1.
- 2.4 Between 19:30 and 23:45 five additional services shall be provided at hourly intervals from London (Kings Cross) and Letchworth Garden City, calling at the stations in paragraph 1.1. The first of these services need not call at Potters Bar. The services which depart between 20:30 and 23:30 must be extended to Royston, calling at Royston only.

#### 3. Service Pattern Monday to Friday (Stopping Service): Cambridge Departures.

- 3.1 Between 05:30 and 05:45, an additional service shall be provided from Cambridge to London (Kings Cross), calling at the stations described in paragraph 1.1.
- 3.2 Between 05:00 and 08:45 Morning Peak services shall be provided as described in the table below:

Para 3.2 part	Services departing from Cambridge to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1.	2	One additional service from Cambridge to London (Kings Cross) shall depart in each of the time periods 06:20 - 06:40 and 08:20 - 08:40.  The earlier of these two services need not call at Potters Bar.
(b)	Foxton, Shepreth, Meldreth, Royston, Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, and Finsbury Park.	2	Additional services from Cambridge to London (Kings Cross) shall be provided between 06:45 and 07:30.
(c)	Royston, Ashwell & Morden, Baldock, Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park.	2	Additional services from Royston to London (Kings Cross) shall depart between 05:00 and 06:00.
(d)	Letchworth Garden City, Hitchin, Stevenage, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park	2	Additional services from Letchworth Garden City to London (Kings Cross) shall

			depart between 06:45 and 07:30.
(e)	Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park	1	An additional service from Letchworth Garden City to London (Kings Cross) shall depart between 08:30 and 08:45.

- 3.3 Between 09:30 and 10:30 an additional service shall be provided from Letchworth Garden City to London (Kings Cross) calling at the stations described in paragraph 1.1.
- 3.4 Between and including the Early Service and Late Services, 15 services shall be provided from Cambridge to London (Kings Cross) at hourly intervals calling at the stations described in paragraph 1.1. The interval between the penultimate service and Late Service may be extended up to 1 hour 30 minutes. The Late Service shall originate from Kings Lynn as part of a service specified in Route E.

# 4. <u>Service Pattern Monday to Friday (Semi-Fast Service): London (Kings Cross)</u> Departures.

- 4.1 Between and including the Early Service and 15:59, 11 services shall be provided at hourly intervals from London Kings Cross to Cambridge calling at the stations specified in paragraph 1.2. Those services departing London (Kings Cross) before 09:00 must additionally call at Ashwell & Morden, Meldreth, Shepreth, and Foxton. The Early Service shall be extended to Kings Lynn as part of a service specified in Route E.
- 4.2 Between 16:15 and 18:59, five additional services shall be provided from London (Kings Cross) to Royston at half-hourly intervals, calling at Stevenage, Letchworth Garden City, Baldock and Ashwell & Morden.. One interval between services departing London (Kings Cross) before 17:30 may be extended up to 1 hour 15 minutes.

One of these services departing London (Kings Cross) between 18:00 and 18:30 shall call additionally at Welwyn North and Hitchin, but need not call at Stevenage.

One of these services departing London (Kings Cross) between 18:30 and 19:00 shall call additionally at Hitchin.

Between 19:00 and 19:30 another additional service shall be provided between London (Kings Cross) and Royston, calling at Welwyn North, Letchworth Garden City, Baldock, Ashwell & Morden, and Royston.

The additional services specified in this paragraph shall be subject to a Maximum Journey Time from London (Kings Cross) to Royston not exceeding 46 minutes, with one exception of 50 minutes permitted.

4.3 Between 20:00 and the Late Service three services shall be provided at hourly intervals from London (Kings Cross) to Cambridge calling at the stations

described in paragraph 1.2. These services must call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.

4.4 Between 23:50 and 00:20 one additional service shall be provided between London (Kings Cross) and Cambridge, calling at the stations described in paragraph 1.2. This service must call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.

On Saturday mornings between 00:30 and 00:45, an additional service shall be provided between London (Kings Cross) and Cambridge, calling at Stevenage, Hitchin, Letchworth, Baldock and Royston.

#### 5. Service Pattern Monday to Friday (Semi-Fast Service): Cambridge Departures

5.1 Between 06:15 and 09:00, six additional services shall be provided from Royston to London (Kings Cross) calling at Ashwell & Morden, Baldock, Letchworth Garden City, Hitchin, Stevenage and Finsbury Park. Two services departing Royston between 06:30 and 07:00 and between 07:30 and 08:00 respectively must call additionally at Knebworth. The first of the six services must start back from Cambridge, and need not call at Foxton, Shepreth, Meldreth, Ashwell & Morden and Baldock.

Two services departing from Royston between 07:30 and 08:00 may omit to call at Finsbury Park. One of these services may omit to call at Stevenage. One service shall call additionally at Welwyn North.

The additional services specified in this paragraph shall be subject to a Maximum Journey Time from Royston to London (Kings Cross) not exceeding 51 minutes, with two exceptions of up to 54 minutes permitted.

5.2 Between and including the Early and Late Services, 15 services shall be provided at hourly intervals from Cambridge to London (Kings Cross), calling at the stations described in paragraph 1.2.

Two of these services departing Cambridge between 10:15 and 10:45 and between 16:15 and 16:45 respectively must start back from Ely, calling at Waterbeach.

The Early Service need not call at Stevenage or Finsbury Park.

The Early Service and the four services departing from Cambridge between 16:00 and 20:00must additionally call at Foxton, Shepreth, Meldreth and Ashwell & Morden.

One interval may be extended to 1 hours and 40 minutes.

Six services originating from Cambridge after 17:00 shall originate from Kings Lynn as part of services specified in Route E.

#### 6. Service Pattern Saturdays: Stopping Service.

6.1 Between and including the Early and Late Services, 18 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.1.

6.2 Between and including the Early and Late Services, 17services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.1.

Between 23:00 and 23:15, an additional service shall be provided from Cambridge to London (Kings Cross), calling at the stations specified in paragraph 1.1.

An additional service shall be provided between Royston and London Kings Cross departing between 05:00 and 05:30, calling at stations specified in paragraph 1.1.

#### 7. Service Pattern Saturdays: Semi-Fast Service.

7.1 Between and including the Early and the Late Services, 16 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.2.

Between 05:30 and 05:45, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.2. This service shall call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.

- 7.2 Between and including the Early and the Late Services, 16 services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.2.
- 7.3 Two additional services shall be provided from London Kings Cross to Cambridge departing between 23:00 and 23:59, calling at the stations specified in paragraph 1.1. One of these services need only call at Stevenage, Letchworth and Royston.

#### 8. Service Pattern Sundays: Stopping Service

8.1 Between and including the Early and the Late Services, 17 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.1.

An additional service shall be provided between 16:00 and 19:00 from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.1.

8.2 Between and including the Early and the Late Services, 16services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.1.

Between 08:45 and 11:00, three additional services shall be provided from Letchworth to London (Kings Cross) at hourly intervals, calling at the stations specified in paragraph 1.1.

#### 9. Service Pattern Sundays: Semi-Fast Service

9.1 Between the Early and the Late Services, 15 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.2.

Between 00;00 and 00:15, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at Stevenage and Hitchin.

Between 06:30 and 06:45, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.2.

9.2 Between the Early and the Late Services, 17 services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.2.

#### 10. <u>Limited Stops (Semi-Fast Service)</u>

#### Ashwell & Morden, Meldreth, Shepreth and Foxton.

- 10.1 On Mondays to Fridays, services departing from London (Kings Cross) between 05:30 and 08:00 (including the Early Service) shall call.
- 10.2 On Saturdays the Early Service and the second service from London (Kings Cross) shall call.

#### 11. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
London (Kings Cross) to Cambridge (Stopping Service)	07:30	07:30	09:30	
Cambridge to London (Kings Cross) ( Stopping Service)	10:30	07:30	08:30	
London (Kings Cross) to Cambridge (Semi-fast Service)	07:00	08:15	10:00	
Cambridge to London (Kings Cross) ( Semi-fast Service)	09:10	07:45	07:45	

#### 11. Early and Late Services Contd.

	Late Service departs no earlier than:				
Route	Monday - Friday Saturday Sunday				
London (Kings Cross) to Cambridge (Stopping Service)	23:00 23:00 00:00				

Cambridge to London (Kings Cross) ( Stopping Service)	23:15	18:30	21:45
London (Kings Cross) to Cambridge (Semi-fast Service)	22:45	21:45	22:45
Cambridge to London (Kings Cross) ( Semi-fast Service)	22:15	22:30	22:15

## 12. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Cambridge (Stopping Service)	oridge (Stopping exceptions of 1		1 hour 21 minutes, with two exceptions of 1 hour 25 minutes, two exceptions of 1 hour 29 minutes and one exception of 1 hour 32 minutes permitted
Cambridge to London (Kings Cross) (Stopping Service)	1 hour 26 minutes, with one exception of 1 hour 30 minutes permitted	1 hour 25 minutes, with one exception of 1 hour 28 minutes and one exception of 1 hour 35 minutes permitted	1 hour 24 minutes, with one exception of 1 hour 28 minutes permitted
London (Kings Cross) to Cambridge (Semi-Fast Service)	1 hour 7 minutes, with three exceptions of 1 hour 9 minutes and two exceptions of 1 hour 13 minutes permitted	1 hour 3 minutes, with two exceptions of 1 hour and 7 minutes permitted	1 hour 2 minutes, with one exception of 1 hour 9 minutes and two exceptions of 1 hour 11 minutes permitted

## 12. <u>Maximum Journey Times Contd.</u>

Route	Monday - Friday	Saturday	Sunday
-------	-----------------	----------	--------

Cambridge to London (Kings Cross) ( Semi-fast Service)	1 hour 4 minutes, with four exceptions of 1 hour 9 minutes and one exception of 1 hour 11 minutes permitted	1 hour 4 minutes, with one exception of 1 hour 6 minutes and one exception of 1 hour 19 minutes permitted	1 hour 2 minutes, with two exceptions of 1 hour 6 minutes, two exceptions of 1 hour 13 minutes and one exception of 1 hour 17 minutes permitted
--	--	---	---

#### ROUTE E: LONDON (KINGS CROSS) TO CAMBRIDGE AND KINGS LYNN.

#### 1. Route Definition

- 1.1 Services shall be provided between London (Kings Cross) and Kings Lynn, calling at Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.
- 1.2 Services may be provided by those specified in Route D.

#### 2. Service Pattern: Monday to Friday

- 2.1 Between and including the Early and Late Services, 14 services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1. From 16:40 onwards, all services shall call additionally at Letchworth Garden City and Royston, except that one service need not call at either Letchworth Garden City or Royston and one service need only call at Royston.
- 2.2 Between 20:00 and 23:30, four additional services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn, calling at the stations specified in Paragraph 1.1. These services shall call additionally at Letchworth Garden City and Royston.

Between 20:30 and 23:00 an hourly service shall operate from London (Kings Cross) to Ely, calling at Letchworth, Royston, Cambridge and Waterbeach. The first of these services shall extend to Kings Lynn, calling at Littleport, Downham Market and Watlington.

2.3 Between 08:00 and 19:30,12 additional services shall be provided at hourly intervals from London (Kings Cross) to Cambridge. These services, when combined with those specified in paragraph 2.1, shall provide a half-hourly service between London and Cambridge. The first of these services shall extend to Ely, but need not call at Waterbeach.

From 17:00 (departure time from London Kings Cross) one of these services shall call additionally at Letchworth Garden City and two of these services shall call additionally at Royston.

Between 17:00 and 19:30 (departure time from London Kings Cross), three services shall be extended to Ely, but need not call at Waterbeach. The

second of these three services shall be further extended to Kings Lynn, calling at Downham Market.

- 2.4 Between 07:00 and 07:30 one additional service shall be provided from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1.
- 2.5 Between 06:15 and 07:00, two additional services shall be provided from Cambridge to Kings Lynn at an interval of 40 minutes, calling at the stations specified in paragraph 1.1. The second of these services shall originate from Kings Cross as part of a service specified in Route D.
- 2.6 Between 15:15 and 16:30 two services shall be provided at an hourly interval from Cambridge to Ely, calling at Waterbeach. The later of these services shall be extended to Downham Market, but need not call at Littleport.
- 2.7 Between and including the Early and Late Services, 11services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross), calling at stations specified in paragraph 1.1.

Services departing Kings Lynn up to 09:00 shall call additionally at Royston and Letchworth Garden City. One of these services need not call at Letchworth Garden City.

Between 07:00 and 07:15 an additional service shall be provided from Kings Lynn to Cambridge, calling at all intermediate stations. This service shall be extended to London (Kings Cross) calling at the stations specified in paragraph 1.1 and additionally at Royston and Letchworth Garden City. It may fulfil this requirement by attaching at Cambridge to the first of the two services required to be provided by paragraph [2.8] below.

2.8 Between 07:15 and 08:45 two additional services shall be provided at an hourly interval from Kings Lynn to London (Kings Cross) calling at the stations in paragraph 1.1.

The first of these services shall call additionally at Royston and Letchworth Garden City and need not call at Watlington, Littleport or Waterbeach.

Between 17:00 and 17:30 one additional service shall be provided from Downham Market to London (Kings Cross). This service need not call at Littleport and Waterbeach.

- 2.9 Between 10:00 and 18:30, nine additional services shall be provided at hourly intervals from Cambridge to London Kings Cross. These services when combined with those specified in paragraph 2.7 shall provide a half-hourly service from Cambridge to London.

  Between 05:00 and 05:15, one additional service shall be provided from Cambridge to London Kings Cross. This service must call additionally at Royston and Letchworth Garden City.
- 2.10 Between 18:30 and 22:00, a half-hourly service shall be provided from Cambridge to London (Kings Cross). Two services departing from Cambridge between 18;30 and 19:00 and between 20:30 and 21:00 respectively shall start back from Ely, but need not call at Waterbeach.

- 2.11 Between 04:45 and 05:00, one additional service shall be provided from Kings Lynn to London (Kings Cross) calling at Watlington, Downham Market, Littleport, Ely, Waterbeach, Cambridge, Royston and Letchworth Garden City.
- 2.12 Between 06:00 and 06:15 one service shall be provided from Kings Lynn to London (Kings Cross), calling at Downham Market, Ely, Cambridge and Royston.
- 2.13 Between 16:30 and 22:35, seven services shall be provided at hourly intervals from Kings Lynn to Cambridge, calling at the stations specified in Paragraph 1.1. These services shall be extended to London (Kings Cross) as part of services specified in Route D.

#### 3. Service Pattern Saturdays

- 3.1 Between and including the Early and Late Services, 16 services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1. The Late service may terminate at Cambridge.
- 3.2 Between 08:00 and 19:30, 12 additional services shall be provided from London (Kings Cross) and Cambridge. These services when combined with those specified in paragraph 3.1 shall provide a half hourly service between Cambridge and London.
- 3.3 Between and including the Early and Late Services, 17 services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross) calling at stations specified in paragraph 1.1. One interval may be reduced to 40 minutes.
  - Between 09:15 and 09:30, an additional service shall be provided from Kings Lynn to London (Kings Cross) calling at Downham Market, Ely and Cambridge.
- 3.4 Between 07:00 and 20:30,13 additional services shall be provided between Cambridge and London (Kings Cross). These services when combined with those specified in paragraph 3.3 shall provide a half hourly service between Cambridge and London.
- 3.5 Between 22:00 and 23:15, two additional services shall be provided from London (Kings Cross) to Kings Lynn, at hourly intervals, calling at Letchworth, Royston, Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.
- 3.6 Between 06:30 and 06:45, one additional service shall be provided from Cambridge to Kings Lynn calling at Waterbeach, Ely, Littleport, Downham Market and Watlington.

#### 4. Service Pattern Sundays

4.1 Between and including the Early Service and 13:59, five services shall be provided at two-hourly intervals from London (Kings Cross) and Kings Lynn calling at stations specified in paragraph 1.1.

Between 07:45 and 08:00, an additional service shall be provided from London (Kings Cross) to Kings Lynn, calling at Finsbury Park, Stevenage, Hitchin, Lechworth, Baldock, Royston, Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.

- 4.2 From 14:00 up to and including the Late Service, ten services shall be provided at hourly intervals from London (Kings Cross) and Kings Lynn calling at stations specified in paragraph 1.1.
- 4.3 Between and including the Early Service and 11:59, four services shall be provided at two-hourly intervals from Kings Lynn to London (Kings Cross) calling at all stations specified in paragraph 1.1.
- 4.4 From 12:00 up to and including the Late Service, eleven services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross) calling at all stations specified in paragraph 1.1.

Between 17:45 and 18:00, one additional service shall be provided from Kings Lynn to London (Kings Cross), at a half hourly interval with the services specified in the paragraph above, calling at the stations specified in paragraph 1.1.

4.5 One additional service shall be provided from Cambridge to Kings Lynn to arrive at Kings Lynn no more than two hours before the Early Service.

### 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:				
Route	Monday - Friday Saturday Sunday				
London (Kings Cross) to Kings Lynn	08:30	08:30	11:00		
Kings Lynn to London (Kings Cross)	08:00	07:45	10:15		

	Late Service departs no earlier than:				
Route	Monday - Friday Saturday Sunday				
London (Kings Cross) to Kings Lynn	19:30	21:30	23:00		
Kings Lynn to London (Kings Cross)	15:45	21:30	22:15		

### 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Kings Lynn	1 hour 39 minutes, with five exceptions of 1 hour 41 minutes, four exceptions of 1 hour 44 minutes and one exception of 2 hours permitted	1 hour 36 minutes, with four exceptions of 1 hour 41 minutes, two exception of 1 hour 48 minutes and 1 exception of 1 hour 57 minutes permitted	1 hour 35 minutes, with one exception of 1 hour 44 minutes and one exception of 2 hours permitted
Kings Lynn to London (Kings Cross)	•		1 hour 40 minutes, with one exception of 1 hour 41 minutes, one exception of 1 hour 51 minutes and one exception of 2 hours 4 minutes permitted

#### ROUTE F: BEDFORD TO LONDON BLACKFRIARS

#### 1. Route Definition

- 1.1 Services shall be provided between Bedford and London Blackfriars calling at Flitwick, Harlington, Leagrave, Luton, Luton Airport Parkway, Harpenden, St. Albans, St. Pancras International, Farringdon and City Thameslink.
- 1.2 **Limited Stops** shall be made at Radlett, Elstree, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town as described.
- 1.3 Services may be combined with those specified in Routes H, I and J to provide cross-London services with one train fulfilling both requirements.

#### 1.4 <u>City Thameslink</u>

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.5 **Farringdon**

Services need not call at Farringdon between 00:45 and 05:00.

#### 2. Service Pattern: Mondays to Fridays

2.1 Between 02:00 and 05:20 six services shall be provided from Bedford to London Blackfriars at a maximum interval of 1 hour.

Three of these services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

2.2 Between 05:30 and the Early Service morning peak services shall be provided from Bedford to London Blackfriars as described in the table below:

Para 2.2 part	Services departing from Bedford to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1.	(11)	These services shall be provided at 20 minute intervals. One interval of 25 minutes may be permitted. These services shall depart between 05:30 and the Early Service.  A maximum interval of 20 minutes shall apply. One interval of 24 minutes shall be permitted.

Para 2.2	Services departing from Bedford to:	No. of Services	Service Requirement
part (a) Contd.			One service departing Bedford between 07:00 and 07:30 shall call additionally at Radlett, Elstree & Borehamwood and West Hampstead Thameslink.  Four of these services shall call additionally at West Hampstead Thameslink.  One service may start from Flitwick and shall call additionally at Radlett, Elstree & Borehamwood and West Hampstead
(b)	Flitwick, Luton, Harpenden, St. Albans, St. Pancras International, Farringdon, City Thameslink and London Blackfriars.	9	Thameslink. These additional services shall depart Bedford between 05:45 and 08:00. These services shall be provided at 20 minute intervals. Three services departing between 07:00 and 08:00 shall call at Leagrave. Two of these services may be extended to Elephant & Castle. Two services need not call at Flitwick

2.3 Between and including the Early and Late Services 41 services shall be provided at 15 minute intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1.

Two services per hour at half hourly intervals may call additionally at West Hampstead Thameslink.

2.4 Between 09:00 and 10:30 three additional services shall be provided from Bedford to London Blackfriars at a maximum interval of 50 minutes, calling at Flitwick, Luton, Luton Airport Parkway, Harpenden, St. Albans, St. Pancras International, Farringdon and City Thameslink.

These services may be extended to Elephant & Castle.

2.5 Between 15:45 and 17:45, four additional services shall be provided from Bedford to London Blackfriars, calling at the stations in paragraph 1.1.

All of these services need not call at Harlington and Leagrave. Two services need not call at Harpenden and one service need not call at Flitwick.

2.6 Between 18:00 and the Late Service, two additional services shall be provided from Bedford to London Blackfriars at an interval of 30 minutes.

These services shall call additionally at West Hampstead Thameslink.

These services need not call at Flitwick, Harlington or Leagrave, but may call additionally at Radlett and Elstree & Borehamwood.

2.7 Between the Late Service and 22:29 eight services shall be provided from Bedford to London Blackfriars at a maximum 30 minute interval calling at the stations in paragraph 1.1.

Seven of these services shall call additionally at West Hampstead Thameslink.

2.8 Between 22:30 and 01:59, five services shall be provided from Bedford to London Blackfriars at a maximum 60 minute interval calling at the stations in paragraph 1.1.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

2.9 Between 02:00 and 05:59 five services shall be provided from London Blackfriars to Bedford calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour and 20 minutes.

These services shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

2.10 Between 06:00 and the Early Service four services shall be provided from London Blackfriars to Bedford at a maximum interval of 30 minutes calling at the stations in paragraph 1.1.

Three of these services shall call additionally at West Hampstead Thameslink.

2.11 Between and including the Early and Late Services 38 services shall be provided at 15 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1. Between 07:30 and 09:00 there may be one interval of 25 minutes.

Two services per hour at half houly intervals may call additionally at West Hampstead Thameslink.

2.12 Between 08:15 and 09:30, four additional services shall be provided at 15 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. One interval of 35 minutes may be permitted.

These services need not call at Harpenden and Harlington. Three services need not call at Leagrave. One service need not call at St. Albans.

2.13 Between 16:00 and 16:30 one additional service shall be provided from London Blackfriars to Bedford calling at City Thameslink, Farringdon, St. Pancras International, St. Albans, Harpenden, Luton and Flitwick.

This service may originate from Elephant & Castle.

2.14 Between the Late Service and 19:09 evening peak services shall be provided from London Blackfriars to Bedford as described in the table below:

Para 2.14 part	Services departing from London Blackfriars to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1	12	Services shall be provided at 20 minute intervals.  Two services departing London Blackfriars between 17:30 and 18:30 shall call additionally at West Hampstead Thameslink, Mill Hill Broadway, Elstree & Borehamwood and Radlett.  One of these services may originate from Elephant & Castle.  Two further services shall call additionally at West Hampstead Thameslink
(b)	City Thameslink, Farringdon, St. Pancras International, St. Albans, Harpenden, Luton, Leagrave, Flitwick and Bedford	8	Services shall be provided at 30 minute intervals.  One service need not call at Harpenden  One service shall call additionally at Leagrave.  One service need not call at Flitwick.

2.15 Between 19:10 and 21:59, twelve services shall be provided at 15 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1.

Two services an hour shall call additionally at West Hampstead Thameslink.

2.16 Between 22:00 and 01:30 seven services shall be provided at 30 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

Services departing London Blackfriars after 00:01 shall call additionally at Kentish Town, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

#### 3. Service Pattern: Saturdays

3.1 Between 02:00 and the Early Service nine services shall be provided at half hourly intervals, calling at the stations is paragraph 1.1. An interval of 60 minutes may be permitted between the first and second services and one exception of 40 minutes may be permitted.

Seven services shall call additionally at West Hampstead Thameslink and four shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood and Kentish Town.

- 3.2 Between and including the Early and Late Services 50 services shall be provided at 15 minute intervals from Bedford to London Blackfriars, calling at the stations in paragraph 1.1. After 18:00 intervals of 20 minutes may be permitted. Two services per hour at 30 minute intervals shall call additionally at West Hampstead Thameslink.
- 3.3 Between the Late Service and 23:45 eleven services shall be provided from Bedford to London Blackfriars at half-hourly intervals, calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink. Three services departing after 22:00 shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood and Kentish Town.

3.4 Between 02:00 and the Early Service nine services shall be provided from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour 15 minutes.

Services departing before 06:00 shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett. Four services departing between 06:00 and 07:30 shall call additionally at West Hampstead Thameslink.

3.5 Between and including the Early and Late Services 58 services shall be provided at 15 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. Two services per hour at 30 minute intervals shall call additionally at West Hampstead Thameslink.

3.6 Between the Late Service and 01:15 four services shall be provided at half hourly intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1

These services shall call additionally at West Hampstead Thameslink.

A further four services shall be provided at half hourly intervals from London Blackfriars to Bedford calling at the stations specified in Paragraph 1.1 and additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

#### 4. Service Pattern: Sundays

4.1 Between 05:30 and the Early Service four services shall be provided at half hourly intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1. One service may terminate at London St Pancras.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

4.2 Between the Early and Late Services 25 services shall be provided at half hourly intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

4.3 Between the Early Service and 17:45, 20 additional services shall be provided at thrity minute intervals from Bedford to London Blackfriars, calling at the stations in paragraph 1.1.

These services may call additionally at West Hampstead Thameslink.

4.4 Between the Late Service and 02:00 ten services shall be provided from Bedford to London Blackfriars, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 60 minutes.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

4.5 Between the Early and Late Services 36 services shall be provided at half hourly intervals between London Blackfriars and Bedford, calling at the stations in paragraph 1.1. One interval of 1 hour shall be permitted before 07:00. The early service and the second service may originate at St Pancras International.

These services shall call additionally at West Hampstead Thameslink. All services before 10:00 and four services between 22:00 and 00:15 shall call additionally at Kentish Town, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

4.6 Between 10:30 and 19:45, 19 additional services shall be provided at 30 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

4.7 Between the Late Service and 01:15 two additional services shall be provided at a 30 minute interval from London Blackfriars and Bedford, calling at the stations in paragraph 1.1.

This service shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

#### 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:				
Route	Monday - Saturday Sunday				
Bedford to London Blackfriars	09:45	07:45	09:15		
London Blackfriars to Bedford	08:40	09:00	07:30		

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Bedford to London Blackfriars	18:30	18:30	19:45
London Blackfriars to Bedford	16:30	21:45	23:45

#### 6. <u>Maximum Journey Times</u>

Route	Route Monday - Friday		Sunday	
Bedford to London Blackfriars	1 hour and 11 minutes, with 9 exceptions of 1 hour and 16 minutes	1 hour and 13 minutes, with 10 exceptions of 1 hour and 25 minutes permitted.	1 hour and 15 minutes, with 9 exceptions of 1 hour and 25 minutes permitted	
London Blackfriars to Bedford	1 hour and 11 minutes, with 8 exceptions of 1 hour and 14 minutes	1 hour and 15 minutes, with 11 exceptions of 1 hour and 28 minutes permitted	1 hour and 15 minutes, with 10 exceptions of 1 hour and 28 minutes permitted	

#### ROUTE G: LUTON AND ST. ALBANS TO LONDON BLACKFRIARS

#### 1. Route Definition

- 1.1 Services shall be provided between Luton and London Blackfriars, calling at Luton Airport Parkway, Harpenden, St. Albans, Radlett, Elstree, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink, Kentish Town, St. Pancras International, Farringdon and City Thameslink.
- 1.2 Limited Stops shall be provided at Leagrave, Harlington, Flitwick and Bedford as described.
- 1.3 Services may be combined with those specified in Routes H, I and J to provide cross-London services with one train fulfilling both requirements.

#### 1.4 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.5 **Farringdon**

Services need not call at Farringdon between 00:45 and 05:00.

#### 2. Service Pattern: Mondays To Fridays

2.1 Between 05:15 and 06:00 one service shall be provided from St. Albans to London Blackfriars calling at the stations described in paragraph 1.1.

This service shall start from Luton calling at Luton Airport Parkway and Harpenden.

2.2 Between 06:00 and the Early Service morning peak services shall be provided from Luton and St. Albans calling at stations at the maximum intervals as shown below:

Para 2.2 part	Services timed at St. Albans	No. of Services	Service Requirement
(a)	Luton, Luton Airport Parkway, Harpenden	8	Services shall be provided at 30 minute intervals. One interval of 38 minutes may be permitted.
(b)	St. Albans, Radlett, Elstree & Borehamwood, West Hampstead Thameslink, London St. Pancras International, Farringdon, City Thameslink, London Blackfriars	14	Services shall be provided at 20 minute intervals.

Para 2.2 part	Services timed at St. Albans	No. of Services	Service Requirement
(c)	Mill Hill Broadway, Hendon, Cricklewood, Kentish Town	10	Services shall be provided at 20 minute intervals.

2.3 Between and including the Early and Late Services 44 services shall be provided at 15-minute intervals from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

There may be one interval of 35 minutes between 18:00 and 19:00.

After 19:00, services shall be provided at half hourly intervals.

Two of the services specified in paragraph 2.3 in each hour shall originate from Luton calling at Luton Airport Parkway and Harpenden.

Two intervals of up to 70 minutes may be permitted.

2.5 Between 09:00 and 09:15, an additional service shall be provided from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

This service shall originate at Luton, calling at Luton Airport Parkway and Harpenden and need not call at Mill Hill Broadway, Hendon, Cricklewood or Kentish Town.

2.6 Between 17:30 and 19:45, five additional services shall be provided at an interval of 30 minutes from St. Albans to London Blackfriars, calling at Radlett, Elstree & Borehamwood, West Hampstead Thameslink, St. Pancras International, Farringdon and City Thameslink.

One interval of 45 minutes may be permitted.

Three services shall originate at Luton calling at Luton Airport Parkway and Harpenden.

2.7 Between the Late Service and 23:30 three services shall be provided from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

These services shall originate from Luton calling at Luton Airport Parkway and Harpenden.

2.8 Between 05:50 and 06:54 three services shall be provided from London Blackfriars to St. Albans calling at the stations described in paragraph 1.1.

A maximum interval of 30 minutes shall apply.

One service shall extend to Bedford calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.

2.9 Between 06:55 and the Early Service, morning Contra-peak services shall be provided from London Blackfriars calling at stations at the maximum intervals as shown below:

Para. 2.9 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(a)	City Thameslink, Farringdon, St. Pancras International, West Hampstead Thameslink, Elstree & Borehamwood, Radlett and St. Albans.	15	Services shall be provided at 20 minutes intervals with one exception of 25 minutes.
(b)	Kentish Town, Cricklewood, Hendon, Mill Hill Broadway	9	Services shall be provided at 20 minutes intervals with one exception of 30 minutes.
(c)	Harpenden, Luton Airport Parkway and Luton	8	Services shall be provided at 20 minutes intervals with one interval of 30 minutes and one interval of 50 minutes.

- 2.10 Between and including the Early and Late Services 28 services shall be provided at 15-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.
- 2.11 Two of the services specified in paragraph 2.10 in each hour shall be extended to Luton calling at Harpenden and Luton Airport Parkway. One service between 11:45 and 12:15 shall be extended to Bedford.
- 2.12 Between the Late Service and 18:54, evening peak services shall be provided from London Blackfriars calling at stations at the intervals as shown below:

Para. 2.12 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(a)	Elephant & Castle	2	Two services may originate from Elephant & Castle.
(b)	City Thameslink, Farringdon, St. Pancras International, West Hampstead Thameslink, Mill Hill Broadway, Elstree & Borehamwood, Radlett and St. Albans	10	Services shall be provided at 15 minute intervals
(c)	Kentish Town, Cricklewood and Hendon	9	Services shall be provided at 15 minutes intervals with one interval of 20 minutes permitted.

Para. 2.12 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(d)	Harpenden, Luton Airport Parkway, Luton	4	Services shall be provided at 30 minute intervals.

2.13 Between 18:55 and 19:54, four services shall be provided at 15-minute intervals from London Blackfriars to St. Albans, calling at the stations in paragraph 1.1.

Two of these services shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.

2.14 Between 19:55 and 21:54, eight services shall be provided at 15-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.

Four of these services shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.

Between 21:55 and 23:55, four services shall be provided at 30-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.

These services shall be extended to Bedford calling at Leagrave, Harlington, Flitwick and Bedford.

#### Additional services from Kentish Town towards London

2.15 Between 09:50 and 16:30, fourteen services shall be provided at 30-minute intervals from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

Between 05:45 and 06:00, one additional service shall be provided from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

Between 16:00 and 16:15, one additional service shall be provided from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

- 2.16 Between 17:00 and 17:30, one service shall be provided from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.
- 2.17 Between 20:00 and 21:00, two services shall be provided at a 30-minute interval from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

#### Additional services from London to Kentish Town

- 2.18 Between 08:40 and 10:00, four services shall be provided at a maximum interval of 40 minutes from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.
  - Between 05:30 and 05:45, one additional service shall be provided from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.
- 2.19 Between 10:00 and 16:15, thirteen services shall be provided at 30-minute intervals from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.
- 2.20 Between 19:30 and 21:10, four services shall be provided at a maximum interval of 30 minutes from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.

#### 3. Service Pattern Saturdays

- 3.1 Between 05:30 and the Early Service three additional services at half-hourly intervals shall be provided from Luton to London Blackfriars, calling at the stations in paragraph 1.1.
- 3.2 Between and including the Early and Late Services 53 services shall be provided at 15-minute intervals from St. Albans to London Blackfriars, calling at the stations in paragraph 1.1. One interval of 20 minutes shall be permitted.
- Two of the services specified in paragraph 3.2 each hour departing before 08:15 shall have originated from Luton, calling at Luton Airport Parkway and Harpenden.
- 3.4 Between the Late Service and 23:00, six additional services shall be provided at half hourly intervals from Luton to London Blackfriars, calling at the stations in paragraph 1.1.
- 3.5 Between 06:00 and the Early Service four additional services shall be provided at half hourly intervals from London Blackfriars to Luton, calling at the stations in paragraph 1.1. One service may be extended to Bedford calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.
- 3.6 Between and including the Early and Late Services 56 services shall be provided at 15-minute intervals from London Blackfriars to St. Albans, calling at the stations in paragraph 1.1.
- Two of the services specified in paragraph 3.6 each hour shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.
- 3.8 Between the Late Service and 23:45, five additional services shall be provided at half hourly intervals from London Blackfriars to St Albans, calling at the stations in paragraph 1.1.
  - Four of these services shall be extended to Bedford, calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.

One interval of 35 minutes shall be permitted.

#### 4. Service Pattern Sundays

- 4.1 Between and including the Early and Late Services 25 services shall be provided at half hourly intervals between Luton and London Blackfriars, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from London Blackfriars and Luton, calling at the stations in paragraph 1.1.

.

### 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
St. Albans to London Blackfriars	09:50	08:00	09:30
London Blackfriars to St. Albans	10:45	08:35	11:15

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday	
St. Albans to London Blackfriars	21:30	20:00	20:30	
London Blackfriars to St. Albans	16:15	21:30	21:30	

## 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
St. Albans to London Blackfriars	50 minutes	47 minutes	42 minutes
London Blackfriars to St. Albans	46 minutes	44 minutes	43 minutes

# ROUTE H: BRIGHTON AND GATWICK AIRPORT TO LONDON ST. PANCRAS INTERNATIONAL

#### 1. Route Definition

1.1 Services (stopping services) shall be provided between Brighton and London St. Pancras International, calling at Preston Park, Hassocks, Burgess Hill, Wivelsfield, Haywards Heath, Balcombe, Three Bridges, Gatwick Airport, East Croydon, London Bridge, London Blackfriars, City Thameslink and Farringdon.

On Sundays these services need not call at Balcombe, Preston Park and Wivelsfield.

- 1.2 Services (fast services) shall be provided between Brighton and London St. Pancras International, calling at Haywards Heath, Three Bridges, Gatwick Airport, East Croydon, London Bridge, London Blackfriars, City Thameslink and Farringdon.
- 1.3 Limited Stops shall be made at Tulse Hill, Herne Hill, and Elephant & Castle as described.
- 1.4 Services may be combined with those specified in Routes F and G to provide cross-London services with one train fulfilling both requirements.

#### 1.5 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 08:45 and after 21:15 on Saturdays or all day on Sundays.

#### 1.6 **Farringdon**

Services after between 00:45 and 05:00 need not call at Farringdon.

#### 2. Service Pattern Monday To Friday (Stopping Services)

2.1 Between 02:00 and 04:59 four services shall be provided from Three Bridges to St. Pancras International, calling at Gatwick Airport, East Croydon and London Blackfriars.

One of these services shall also call at London Bridge.

Between 05:15 and 05:45, one additional service shall be provided from Gatwick Airport to St Pancras International calling at the stations specified in Paragraph 1.1.

2.2 Between 05:00 and the Early Service services shall be provided from Brighton to London St. Pancras International as described in the table below:

Para 2.2 part	Services departing from Brighton to:	No. of Services	Service Requirement
(a)	Preston Park, Hassocks, Haywards Heath, Gatwick Airport, East Croydon, Blackfriars, City Thameslink,	11	Services shall be provided at 25 minute intervals with one exception of 30 minutes and one exception of 40 minutes permitted.

Farringdon and London St. Pancras International	

Para 2.2 part	Services departing from Brighton to:	No. of Services	Service Requirement
(b)	Burgess Hill, Three Bridges	10	Services shall be provided at 25 minute intervals with three exceptions of 40 minutes permitted.
(c)	Wivelsfield	8	Services shall be provided at a maximum interval of 65 minutes.
(d)	Balcombe	6	Services shall be provided at a maximum interval of 90 minutes.
(e)	London Bridge	6	Six services shall call.
(f)	One of these services shall cal	l at Tulse Hill	
(g)	Three of these services shall c	all at Herne Hill	

2.3 Between and including the Early and Late Services fifteen services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1.

One service shall call additionally at Tulse Hill and Elephant & Castle.

Balcombe need only be served hourly.

One service need not call at Wivelsfield and two services need not call at London Bridge.

2.4 Between the Late Service and 18:00, three services shall be provided from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 60 minutes.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Tulse Hill.

Two of these services need not call at London Bridge, one need not call at Wivelsfield and one need not call at Balcombe.

2.5 Between 18:00 and 23.59, twelve services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1.

Balcombe need only be served hourly.

Two services need not call at Preston Park and Wivelsfield.

2.6 Between 02:15 and 04:14 three services shall be provided from St. Pancras International to Three Bridges, calling at London Blackfriars, East Croydon and Gatwick Airport.

One of these services may be extended to Brighton, calling at Haywards Heath.

These services may call additionally at Selhurst.

2.7 Between 04:15 and the Early Service, ten services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1. One interval of 40 minutes may be permitted.

One service departing London St. Pancras International between 06:10 and 06:30 shall call at Balcombe.

One service shall call additionally at Tulse Hill. Two of these services may call additionally at Selhurst.

Five of these services need not call at Balcombe, two services need not call at Wivelsfield and four need not call at London Bridge.

Two of these services need not call at Wivelsfield and Preston Park

2.8 Between and including the Early and Late Services 16 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1. One interval of 40 minutes may be permitted.

Balcombe need only be served hourly.

2.9 Between the Late Service and 18:15, five services shall be provided from St. Pancras International to Brighton calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 30 minutes.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Redhill.

One service may terminate at Three Bridges, calling additionally at Elephant & Castle and Redhill.

One service may omit to call at Balcombe, one service may omit to call at Preston Park and three services may omit to call at Wivelsfield.

2.10 Between 18:45 and 23:14 nine services shall be provided at 30-minute intervals from St. Pancras International to Brighton calling at the stations in described paragraph 1.1.

Balcombe need only be served hourly.

2.11 Between 23:15 and 02:14 five services shall be provided at a 30 minute interval from St. Pancras International to Three Bridges, calling at Farringdon, London Blackfriars, London Bridge, East Croydon and Gatwick Airport. The

interval between the penultimate and last of these services may be extended to 60 minutes.

#### 3. <u>Service Pattern Monday to Friday (fast services)</u>

3.1 Between and including the Early and Late Services thirteen services shall be provided at 30 minute intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.2.

The Early Service shall call additionally at Hassocks and Burgess Hill.

3.2 Between the Late Service and 17:30 four services shall be provided from Brighton to St. Pancras International, calling at the stations described in paragraph 1.2.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Tulse Hill.

One service may call additionally at Burgess Hill.

Two services need not call at London Bridge.

3.3 Between 17:45 and 20:45 six services shall be provided at 30 minute intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.2.

One of these services may originate from Three Bridges and two services shall call additionally at Burgess Hill.

3.4 Between 06:15 and the Early Service three services shall be provided from St. Pancras International to Brighton, calling at the stations described in paragraph 1.2.

These services shall call additionally at Three Bridges and two services shall call additionally at Burgess Hill. One service shall call additionally at Balcombe and Hassocks.

One service need not call at London Bridge.

- 3.5 Between and including the Early and Late Services fifteen services shall be provided at 30 minute intervals from St. Pancras International to Brighton calling at the stations described in paragraph 1.2.
- 3.6 Between the Late Service and 19:45 four services shall be provided from St. Pancras International to Brighton, calling at the stations described in paragraph 1.2. After 18:00 these services shall be provided at 30 minutes intervals with one interval of 40 minutes permitted.

All of these services shall call additionally at Burgess Hill, Three of these services shall call additionally at Three Bridges and Hassocks; one of these services shall call additionally at Balcombe, and one service shall call additionally at Preston Park.

#### 4. <u>Service Pattern Saturday (stopping service)</u>

- 4.1 Between 02:10 and 05:14, five services shall be provided from Three Bridges to St. Pancras International, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour.
- 4.2 Between 05:15 and the Early Service, two services shall be provided at hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.
- 4.3 Between the Early and Late Services 32 services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.
  - Balcombe need only be served hourly.
- 4.4 Between the Late Service and 00:10, two services shall be provided from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.
  - The maximum interval between these services shall be 1 hour and 10 minutes.
- 4.5 Between 02:00 and 04:14, four services shall be provided at hourly intervals from St. Pancras International to Three Bridges, calling at the stations in paragraph 1.1.
  - These services may call additionally at Selhurst.
- 4.6 Between 04:15 and the Early Service four services shall be provided from St. Pancras International to Brighton calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour.
  - The first two services may call additionally at Selhurst.
- 4.7 Between the Early and Late Services 27 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations in paragraph 1.1.
  - Balcombe need only be served hourly.
- 4.8 Between the Late Service and 23:09, seven services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations in paragraph 1.1.
  - Balcombe need only be served hourly.
- 4.9 Between 23:10 and 00:40, three services shall be provided at 30-minute intervals from St. Pancras International to Three Bridges, calling at the stations in paragraph 1.1.
  - Two services shall be extended to Brighton, calling at the stations in paragraph 1.1.
  - One service need not call at Balcombe, Wivelsfield and Preston Park.

#### 5. Service Pattern Saturday (fast services)

- 5.1 One additional service departing between 06:00 and the Early Service shall be provided between Brighton and the stations in paragraph 1.2.
- 5.2 Between and including Early and Late Services 28 services shall be provided at maximum intervals of 30 minutes between Brighton and St. Pancras International, calling at the stations in paragraph 1.2.
- 5.3 Between and including Early and Late Services 28 services shall be provided at half hourly intervals between St. Pancras International and Brighton, calling at the stations in paragraph 1.2.

The Early service shall call additionally at Three Bridges.

#### 6. <u>Service Pattern Sunday (stopping services)</u>

6.1 Between and including the Early and Late Services 36 services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.

Between 04:30 and the Early Service, two additional services shall be provided from Three Bridges to London Blackfriars, calling at the stations described in paragraph 1.1. One service shall be extended to Brighton, calling at the stations described in paragraph 1.1.

Between 08:45 and 22:00, 23 additional services shall be provided from Three Bridges to St. Pancras International, calling at the stations described in paragraph 1.1.

Services need not call at London Bridge.

6.2 Between and including the Early and Late Services 30 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1.

Between 06:30 and the Early Service, one additional service shall be provided between London Blackfriars and Brighton, calling at the stations described in paragraph 1.1.

Between 08:45 and 18:30, 20 additional services shall be provided at half hourly intervals from St. Pancras International to Three Bridges, calling at the stations described in paragraph 1.1.

Services need not call at London Bridge.

6.3 Between the Late Service and 01:10 six additional services shall be provided from St. Pancras International and Brighton, calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 60 minutes.

Three of these services need not call at Haywards Heath, Burgess Hill, Hassocks and Brighton. Services need not call at London Bridge.

No service required.

## 8. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	10:45	08:15	07:45
St. Pancras International to Brighton (stopping)	10:30	08:00	09:00
Brighton to St. Pancras International (fast)	10:30	08:30	Not applicable
St. Pancras International to Brighton (fast)	10:40	08:00	Not applicable

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	16:00	21:50	23:30
St. Pancras International to Brighton (stopping)	16:15	19:15	21:15
Brighton to St. Pancras International (fast)	14:50	20:15	Not applicable
St. Pancras International to Brighton (fast)	16:00	19:30	Not applicable

## 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	1 hour and 27 minutes, with three exceptions of 1 hour and 29 minutes permitted.	1 hour and 29 minutes	1 hour and 19 minutes, with one exception of 1hour and 24 minutes permitted.
St. Pancras International to Brighton (stopping)	1 hour and twenty three minutes, with three exceptions of 1 hour and 32 minutes permitted.	1 hour and 29 minutes	1 hour and 30 minutes
Brighton to St. Pancras International (fast)	1 hour and 14 minutes, with one exception of 1 hour and 18 minutes permitted.	1 hour and 17 minutes	Not applicable

## 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
St. Pancras International to Brighton (fast)	1 hour and 16 minutes, with two exceptions of 1 hour and 21 minutes permitted.	1 hour and 18 minutes, with two exceptions of 1 hour and 23 minutes permitted.	Not applicable

#### ROUTE I: SUTTON AND WIMBLEDON TO LONDON ST. PANCRAS INTERNATIONAL

#### 1. Route Definition

- 1.1 Services shall be provided between Sutton and London St. Pancras International calling at West Sutton, Sutton Common, St Helier, Morden South, South Merton, Wimbledon Chase, Wimbledon, Haydons Road, Tooting, Streatham, Tulse Hill, Herne Hill, Loughborough Junction, Elephant and Castle, London Blackfriars, City Thameslink and Farringdon.
- 1.2 Services may be combined with those specified in Routes F, G and J to provide cross-London services with one train fulfilling both requirements. Services when combined with those in Route J shall when combined provide a 15 minute service between St. Pancras International and Streatham.

#### 1.3 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.4 Farringdon

Services after between 00:45 and 05:00 need not call at Farringdon.

#### 2. Service Pattern: Monday To Friday

- 2.1 Between 05:30 and the Early Service two additional services shall be provided from Streatham to St. Pancras International calling at the stations in paragraph 1.1 except that these services may originate at Selhurst and one need not call at Loughborough Junction. The maximum interval between these services shall be 30 minutes.
- 2.2 Between and including the Early and Late Services 30 services shall be provided from Sutton to St. Pancras International at 30 minute intervals calling at the stations in paragraph 1.1. The interval between the first and second service may be 45 minutes.
- 2.3 Between the Late Service and 23:15 four additional services shall be provided from Sutton to St. Pancras International, calling at the stations in paragraph 1.1.
- 2.4 Between 05:00 and the Early Service five additional services shall be provided from St. Pancras International to Sutton calling at the stations in paragraph 1.1. The maximum interval between these services shall be one hour.
- 2.5 Between and including the Early and Late Services 31 services shall be provided from St. Pancras International to Sutton at 30 minute intervals calling at the stations in paragraph 1.1. One interval of 40 minutes and one interval of 45 minutes may be permitted.

The Early and Late Services need not call at Loughborough Junction.

#### 3. Service Pattern Saturday

- 3.1 Between 05:30 and 07:00 three additional services shall be provided between Selhurst and St. Pancras International, calling at the stations in paragraph 1.1. The first of these need not call at Loughborough Junction.
- 3.2 Between 06:45 and the Early Service one additional service shall be provided from Wimbledon to St. Pancras International, calling at the stations in paragraph 1.1.
- 3.3 Between and including the Early and Late Services 30 services shall be provided at half hourly intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1. One interval of 45 minutes shall be permitted between the penultimate and Late Services.
- 3.4 Between the Late Service and 23:30 two additional services shall be provided from Wimbledon to St. Pancras International, calling at the stations in paragraph 1.1. The second service need not call at Loughborough Junction.
- 3.5 Between 06:00 and the Early Service one additional service shall be provided from London Blackfriars to Sutton, calling at the stations in paragraph 1.1, except Loughborough Junction.
- 3.5 Between and including the Early and Late Services 35 services shall be provided at half hourly intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1.

#### 4. <u>Service Pattern Sunday</u>

- 4.1 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals between Sutton and St. Pancras International, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 25 services shall be provided at half hourly intervals between St. Pancras International and Sutton, calling at the stations in paragraph 1.1.

#### 5. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Wimbledon)	07:30	08:15	10:45
St. Pancras International to Sutton (via Wimbledon)	09:45	07:45	10:15

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday

Sutton to St. Pancras International (via Wimbledon)	20:15	21:45	21:30
St. Pancras International to Sutton (via Wimbledon)	23:30	23:45	21:00

## 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Wimbledon)	1 hour and three minutes with five exceptions of 1 hour and seven minutes permitted.	1 hour and 7 minutes, with one exception of 1 hour 12 minutes permitted.	59 minutes
St. Pancras International to Sutton (via Wimbledon)	1 hour and three minutes, with seven exceptions of 1 hour and 9 minutes permitted.	1 hour and 4 minutes, with one exception of 1hour 7 minutes and one exception of 1hour 12 minutes permitted.	59 minutes

# ROUTE J: SUTTON AND MITCHAM JUNCTION TO LONDON ST. PANCRAS INTERNATIONAL

#### 1. Route Definition

- 1.1 Services shall be provided between Sutton and London St. Pancras International, calling at Carshalton, Hackbridge, Mitcham Junction, Streatham, Tulse Hill, Herne Hill, Loughborough Junction, Elephant & Castle, London Blackfriars, City Thameslink and Farringdon.
- 1.2 Services may be combined with those specified in Routes F, G and I to provide cross-London services with one train fulfilling both requirements. Services when combined with those in Route I shall when combined provide a 15 minute service between St. Pancras International and Streatham.

#### 1.3 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays.

#### 1.4 **Farringdon**

Services after between 00:45 and 05:00 need not call at Farringdon.

#### 2. <u>Service Pattern: Monday To Friday</u>

2.1 Between and including the Early and Late Services twenty services shall be provided at 30 minute intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1. One service departing Sutton between 16:15 and 16:45 need not call at Loughborough Junction.

Between 06:30 and the Early Service five additional services shall be provided from Sutton to St. Pancras International at 30 minute intervals, calling at the stations in paragraph 1.1. The interval between the first and the second service may be extended to one hour.

Between the Late Service and 21:30, four additional services shall be provided from Sutton to St. Pancras International at 30 minute intervals, calling at the stations in paragraph 1.1.

- 2.2 Between 05:50 and the Early Service, one additional service shall be provided from St. Pancras International to Sutton calling at the stations in Paragraph 1.1.
- 2.3 Between and including the Early and Late Services 29 services shall be provided at 30 minute intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1. One interval of 45 minutes and two intervals of 40 minutes may be permitted.

Between the Late Service and 21:00, one additional service shall be provided from St. Pancras International to Sutton calling at the stations in Paragraph 1.1.

#### 3. Service Pattern: Saturday

3.1 Between and including the Early and Late Services 29 services shall be provided at half hourly intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1.

3.2

3.3 Between and including the Early and Late Services 31 services shall be provided at half hourly intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1.

#### 4. Service Pattern Sunday

- 4.1 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from Sutton to London Blackfriars, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from London Blackfriars to Sutton, calling at the stations in paragraph 1.1.

#### 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Mitcham Junction)	10:30	08:00	11:00*
St. Pancras International to Sutton (via Mitcham Junction)	07:30	06:45	10:00

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Mitcham Junction)	19:00	21:00	21:45
St. Pancras International to Sutton (via Mitcham Junction)	20:15	20:45	20:00*

<sup>\*</sup>May terminate/start at London Blackfriars

#### 6. <u>Maximum Journey Times</u>

Sutton to St. Pancras International (via Mitcham Junction)	48 minutes, with one exception of 50 minutes permitted.	51 minutes	Not applicable
St. Pancras International to Sutton (via Mitcham Junction)	48 minutes, with seven exceptions of 52 minutes permitted.	50 minutes	Not applicable

## SLC<sub>2</sub>

# TSGN FRANCHISE (FORMER THAMESLINK/GN SERVICES)

# SERVICE LEVEL COMMITMENT SLC2

Version 2

#### Part 1

#### **Service Level Commitment – General Provisions**

#### 1. CONSTRUCTION

The following provisions of this Part 1 to this SLC2 shall apply in respect of the Service Level Commitment set out in Part 2 to this SLC2.

#### 2. DAYS AND TIMES OF DAY

- 2.1 Except to the extent the context otherwise requires, references to a day mean the period commencing at 02:00 on one day and ending at 01:59 on the following day and references to Weekdays and particular days of the week shall be construed accordingly.
- 2.2 References to periods of times and periods of days include the times and days such periods start and finish.
- 2.3 All references to time are to the twenty-four hour clock.

#### 3. SERVICES

- 3.1 Except where expressly indicated to the contrary, references to services, all services or any part or any proportion of services are to be construed as references to the Passenger Services (or the relevant part or proportion thereof) required to be included by the Franchisee in its Timetable pursuant to Schedule 1.1 of the Franchise Agreement and do not include such additional railway passenger services as the Franchisee may be permitted to provide from time to time under this Agreement.
- 3.2 Except where expressly indicated to the contrary, where an interval or frequency is specified for a service, such specification shall apply at the departure point for the relevant service.
- 3.3 Except where expressly indicated to the contrary, all services are to run in both directions and the requirements of the Service Level Commitment (including any interval between services, frequency of service or stopping pattern) are to apply in each direction.
- 3.4 Except where expressly indicated to the contrary, references to "Summer" shall mean that period in any year from the last Saturday in May until the third Saturday in September and "Winter" is defined as any period other than "Summer."

#### 4. STATIONS

4.1 Except where expressly indicated to the contrary and subject always to compliance with the other provisions of the Service Level Commitment (including any maximum Journey Times) nothing in the Service Level Commitment shall prevent services which are required to be included by, or on behalf of, the Franchisee in the Timetable pursuant to Schedule 1.1 calling at any stations which are not specified in the Service Level Commitment or any relevant part of it.

4.2 The Franchisee may designate any station, at which a service calls, as a Request Stop Station, where such service need only call when a passenger or intending passenger wishes to join such service or leave such service at such station. Any such designation by the Franchisee shall be clearly advertised in advance to intending passengers (including in the Timetable).

#### 5. CONNECTIONS

- 5.1 Except where expressly indicated to the contrary, a service which is required to be included in the Timetable shall be provided so as to enable travel between the stations specified without passengers being required to change train. Where a service may be provided by a Connection or where a Connection is required to be provided between two services, such service or combination of services may be provided without change of train being required.
- 5.2 Except where expressly indicated to the contrary, where Connections are required to be provided by the Franchisee, the Franchisee shall ensure that the interval or waiting period between the two relevant services is of a sufficient duration to allow passengers a reasonable period of time to transfer between such services.
- 5.3 Except where expressly indicated to the contrary, where services are subject to maximum Journey Times, such Journey Times shall not apply where the services are required to be, or may be, provided by Connections.

#### 6. PEAK, OFF-PEAK AND CONTRA-PEAK

- 6.1 Peak means, in relation to any service a Weekday Service (except Christmas and Bank Holidays) which arrives at London (London Bridge, London Blackfriars, City Thameslink, Farringdon, London Kings Cross or London St. Pancras International), between 07:00 and 09:59 (the "Morning" Peak) and departs from London (London Bridge, London Blackfriars, City Thameslink, Farringdon, London Kings Cross or London St. Pancras International), between 16:00 and 18:59 (the "Evening" Peak).
- 6.2 Contra-Peak means in relation to any service, those services originating from the point of destination of Peak services during the Peak periods defined in paragraph 6.1.
- 6.3 Off-Peak means in relation to any service any period which is not Peak.

#### 7. NOT USED

#### 8. SERVICE INTERVALS

- 8.1 Except where expressly indicated to the contrary, where the Franchisee is required to include within the Timetable, services for a specified period (which period shall commence and end with an Early Service or Late Service), with a specified interval between each such service during such period, then the following shall apply:
- (a) the minimum number of services to be included in the Timetable in such period shall be determined in accordance with the following formula:

$$S = \frac{TM}{IM}$$

#### where:

- S is the minimum number of services to be included in the Timetable;
- TM is the total minutes in the specified period; and
- IM is the number of minutes in the specified interval, rounded down to the nearest whole number;
- (b) the interval between any two services during such period may be extended by:
  - (i) five minutes; or,
  - (ii) Thameslink Route only: services may be extended by six minutes, when required to travel through the St. Pancras International London Blackfriars section
  - (iii) if greater, an amount of minutes equivalent to one-sixth of the specified interval between each service rounded down to the nearest whole number,

subject to the other provisions of this paragraph 8.1 and such extension not exceeding ten minutes;

(c) the interval between any two services may be reduced, subject to the other provisions of this paragraph 8.1, the interval between each service required in accordance with paragraph 8.1(a) for such period, by such amount as the Franchisee may determine;

#### 8. SERVICE INTERVALS Contd.

(d) the interval between services during such specified period shall be such that, for any period which occurs during such specified period and has a duration equivalent to:

$$((4 \times SI) + E)$$
 minutes

where:

SI is equal to the specified interval between such services; and

E is equal to the extension permitted to such interval under paragraph 8.1(b), at least four services shall be included in the Timetable in such period; and

- (e) the intervals between the start of such period of time and the first service in such period (save where such period of time starts with an Early Service) and between the last service in such period and the end of such period (save where such period of time ends with a Late Service) shall each be no more than the interval between each service specified for such period and (save where such period of time starts and/or ends with an Early Service/Late Service) one of such intervals shall be no more than half such interval between each service.
- 8.2 Except where expressly indicated to the contrary, where one period ends and another period begins immediately thereafter and services are specified at different intervals for each such period, the maximum interval between the last service in the first period and the first service in the second period shall be the longer of the maximum intervals allowed for each such period.

#### **INDEX TO ROUTES**

#### **ROUTE**

#### (GN)

- A. London (Moorgate/ Kings Cross) to Hertford North and Letchworth Garden City.
- B. London (Moorgate/ Kings Cross) to Welwyn Garden City.
- C. London (Moorgate/ Kings Cross) and Stevenage to Huntingdon and Peterborough.
- D. London (Kings Cross) and Stevenage to Cambridge.
- E. London (Kings Cross) and Kings Lynn.

#### (Thameslink)

- F. London Blackfriars to Bedford.
- G. London Blackfriars to St Albans and Luton.
- H. London (St. Pancras International/London Bridge) to Gatwick Airport and Brighton.
- I. London (St. Pancras International) to Wimbledon and Sutton.
- J. London (St. Pancras International) to Hackbridge and Sutton.

#### (Former LSER Thameslink)

- K. London Blackfriars to Beckenham Junction
- L. London Blackfriars to Sevenoaks via Catford and Otford

# ROUTE A: LONDON (MOORGATE / KINGS CROSS) TO HERTFORD NORTH AND LETCHWORTH GARDEN CITY.

#### 1. Route Definition

1.1 Services shall be provided between:

London (Moorgate), Old Street, Essex Road, Highbury and Islington and Drayton Park, or

London Kings Cross and

Finsbury Park and Hertford North, calling at Harringay, Hornsey, Alexandra Palace, Bowes Park, Palmers Green, Winchmore Hill, Grange Park, Enfield Chase, Gordon Hill, Crews Hill, Cuffley, and Bayford.

1.2 Services in paragraph 1.1 shall extend to or originate from Letchworth Garden City, calling at Watton-at-Stone, Stevenage and Hitchin, as described below.

## 2. <u>Service Pattern Monday to Friday: London (Moorgate / Kings Cross) to Hertford</u> North.

2.1 Between the Early Service and 09:05, a half-hourly interval service shall be provided, calling at the stations described in paragraph 1.1. After 23:15 an hourly interval may apply.

Between 07:00 and 08:45 an additional four services shall be provided from London (Moorgate) to Hertford North which, subject to the following provisions, shall call at the stations described in paragraph 1.1. All four services may omit to call at Bowes Park, Grange Park, Crews Hill and Bayford. The first two services may additionally omit to call at Harringay and Hornsey.

Between 07:45 and 08:45, three further services shall be provided from London (Moorgate) to Gordon Hill, at 20-minute intervals. These services shall call at all stations mentioned in paragraph 1.1, except for Crews Hill, Cuffley and Bayford, and two of these services may omit to call at Harringay and Hornsey.

2.2 Between 09:05 and 16:29 a 20-minute interval service shall be provided calling at the stations described in paragraph 1.1. One interval may extend up to 30 minutes.

Between 16:15 and 16:30, one additional service shall be provided from London (Moorgate) to Gordon Hill, calling at the intermediate stations described in paragraph 1.1. This service may omit to call at Bowes Park and Grange Park.

- 2.3 The Early Service and the following two services shall originate from London (Kings Cross). Two of these services shall extend to Stevenage, calling at Watton-at-Stone.
- 2.4 Between 06:40 and 22:00 all services shall originate from Moorgate and call at all intermediate stations between Moorgate and Finsbury Park.

- 2.5 Five of the services specified in 2.1 between 06:45 and 09:15 shall extend to Stevenage calling at intermediate stations, two of which shall be extended further to Letchworth Garden City calling at intermediate stations.
- 2.6 At hourly intervals between 09:45 and 15:15, services specified in paragraph 2.2 shall extend to Letchworth Garden City, calling at intermediate stations.

One further service between 09:05 and 09:45 shall so extend.

- 2.7 Between 15:00 and 16:29, three services specified in paragraph 2.2 shall extend to Stevenage calling at Watton-at-Stone. The last of these three services may omit to call at Harringay, Hornsey, Crews Hill and Bayford
- 2.8 Services from London between 16:30 and 19:44 shall call as shown below:

Para 2.8 part	Service timed from Moorgate	No. of Services	Maximum Interval
(a)	Moorgate, Old Street, Essex Road, Highbury & Islington, Drayton Park, Finsbury Park, Palmers Green, Winchmore Hill, Enfield Chase, Gordon Hill.	23	15 minutes, with one exception of 20 minutes
(b)	Harringay, Hornsey	8	30 minutes
(c)	Alexandra Palace	18	20 minutes, with one exception of 25 minutes
(d)	Bowes Park, Grange Park	15	15 minutes, with one exception of 20 minutes
(e)	Crews Hill, Bayford	7	30 minutes, with one exception of 35 minutes
(f)	Cuffley, Hertford North	17	15 minutes, with one exception of 20 minutes
(g)	Watton-at-Stone, Stevenage	6	30 minutes with two exceptions of 35 minutes
(h)	Hitchin, Letchworth Garden City	2	Between 18:45 and 19:30, two services shall be extended to Letchworth Garden City, calling at all intermediate stations

2.9 Between 19:45 and 22:00, services shall be provided from London (Moorgate) to Hertford North, calling at the stations described in paragraph 1.1, at 20-minute intervals. The first of these services shall extend to Stevenage, calling additionally at Watton-at-Stone. Thereafter, every third service shall extend to Letchworth Garden City, calling at the stations described in paragraph 1.2.

Between 20:00 and 22:00 another two services shall be extended from Hertford North to Letchworth Garden City, calling at the stations described in paragraph 1.2.

An additional service shall be provided between Moorgate and Finsbury Park between 20:00 and 20:30, calling at the stations described in paragraph 1.1.

- 2.10 Between 22:00 and 23:30 a half-hourly service shall be provided from London (Kings Cross) to Hertford North, calling at the stations described in paragraph 1.1, with two services at an hourly interval being extended to Letchworth Garden City, calling at the stations described in paragraph 1.2.
- 2.11 Between 23:30 and 01:10, two further services (including the Late Service) shall be provided at an hourly interval from London (Kings Cross) to Hertford North, calling at the stations described in paragraph 1.1. The Late Service need not call at Harringay, Hornsey, Bowes Park, Grange Park, Crews Hill, Bayford and Watton-on-Stone. These services shall be extended to Letchworth Garden City calling at Hitchin.

# 3. <u>Service Pattern Monday to Friday: Hertford North to London (Moorgate / Kings Cross).</u>

- 3.1 Between 04:45 and 05:00, the Early Service shall originate from Letchworth Garden City and operate to London Kings Cross, calling at the stations described in paragraph 1.1. In addition, between the Early Service and 06:45, two further services from Hertford North to London (Kings Cross / Moorgate) shall originate from Letchworth Garden City.
- 3.2 Between the Early Service and 06:45, and between 16:40 and 21:15, services shall be provided from Hertford North to London Moorgate at half-hourly intervals, calling at the stations described in paragraph 1.1. Two of these services departing from Hertford North between 17:00 and 18:00 shall originate from Stevenage, calling at Watton-at-Stone.

Between 06:00 and 06:15, one additional service shall be provided from Stevenage, calling at stations specified in paragraph 1.1. It need not call at Bayford, Crews Hill, Grange Park, Bowes Park, Harringay and Hornsey.

Between 16:40 and 19:40, six additional services shall be provided from Hertford North, calling at Cuffley, Gordon Hill, Winchmore Hill, Palmers Green, Finsbury Park and intermediate stations to London (Moorgate). The first such service shall original from Letchworth Garden City calling at the stations described in paragraph 1.2, and shall call additionally at Bayford, Crews Hill and Alexandra Palace. The second such service shall call additionally at Alexandra Palace. The other four shall originate from Stevenage, calling at Watton-at-Stone.

Between 17:15 and 19:30, six additional services shall be provided from Gordon Hill to London (Moorgate) at maximum intervals of 30 minutes, calling at stations specified in paragraph 1.1. The first need not call at Grange Park, Bowes Park, Harringay and Hornsey.

Between 19:45 and 20:15, one additional service shall be provided from Hertford North to Moorgate, calling at Cuffley, Gordon Hill, Enfield Chase, Winchmore Hill, Palmers Green, Finsbury Park and then intermediate stations to Moorgate.

3.3 Services departing from Hertford North towards London between 06:45 and 09:00 shall call at maximum intervals at the stations as shown below:

Para 3.7 part	Service timed at Hertford North	No. of Services	Maximum Interval
(a)	Letchworth Garden City, Hitchin	1	Not applicable
(b)	Stevenage, Watton-at-Stone	5	30 minutes, with one exception of 35 minutes permitted
(c)	Hertford North, Cuffley	12	20 minutes,

(d)	Bayford, Crews Hill	5	30 minutes, with one
			exception of 40 minutes
			permitted

- 3.4 Between 07:00 and 07:40, two additional services at a maximum interval of 30 minutes and between 08:10 and 09:30 five additional services at a maximum interval of 25 minutes shall be provided from Gordon Hill to London (Moorgate), calling at all intermediate stations. Two of these services may omit to call at Harringay and Hornsey.
- 3.5 Services departing from Gordon Hill towards London between 07:00 and 09:30, including those from Hertford North, shall call at maximum intervals at stations shown below:

Para 3.5 part	Services timed at Gordon Hill (07:00 - 09:30)	No. of Services	Maximum Interval
(a)	Gordon Hill, Enfield Chase, Finsbury Park, Drayton Park, Highbury & Islington, Essex Road, Old Street, London (Moorgate)	18	15 minutes
(b)	Grange Park	13	15 minutes, with one exception of 20 minutes permitted
(c)	Winchmore Hill, Palmers Green	14	15 minutes, with one exception of 21 minutes permitted
(d)	Bowes Park, Alexandra Palace	16	15 minutes, with one exception of 20 minutes permitted
(e)	Hornsey, Harringay	12	15 minutes, with two exceptions of 20 minutes permitted

- 3.6 Between 09:00 and 16:40, a 20-minute interval service shall be provided from Hertford North to London (Moorgate), calling at the stations described in paragraphs 1.1 and 1.2. One interval of 30 minutes shall be permitted before 09:30.
  - Between 16:15 and 16:30, one additional service shall be provided from Hertford North to London (Moorgate), calling at the stations described in paragraphs 1.1 This service may omit Bayford, Crews Hill, Grange Park, Bowes Park, Harringay and Hornsey.
- 3.7 Between 09:45 and 16:30 seven services shall originate from Letchworth Garden City at hourly intervals and call at all intermediate stations. Between 0900 and 1020, two of these services shall originate from Stevenage calling at Watton-at-Stone, at an hourly interval. Between 1620 and 1640 one service

- shall originate at Letchworth Garden City calling at the stations described in paragraph 1.2.
- 3.8 Between 21:15 and including the Late Service, three services at hourly intervals shall originate from Letchworth Garden City and call at all intermediate stations.

Two additional services shall be provided between Hertford North and London (Kings Cross) calling at the stations described in paragraph 1.1 to allow for a half-hourly interval between 21:15 and the Late Service.

3.9 All services between 21:30 until and including the Late Service shall terminate at London (Kings Cross).

#### 4. Service Pattern: Saturdays

- 4.1 Between and including the Early and Late Services, 40 services shall be provided from London (Kings Cross) to Hertford North, at half-hourly intervals, calling at all stations specified in paragraph 1.1. The Late service and one service departing between 23:15 and 23:30 shall be extended from Hertford North to Letchworth Garden City, calling at all intermediate stations.
- 4.2 Between and including the Early and Late Services, 38 services shall be provided from Hertford North to London (Kings Cross), at half-hourly intervals, calling at all stations specified in paragraph 1.1. One of these services need not call at Harringay.
- 4.3 Between 06:15 and 22:30 services shall be extended at hourly intervals from Hertford North to Stevenage, calling at Watton-at-Stone.
- 4.4 Between 07:15 and 23:45, services shall originate at hourly intervals from Stevenage, calling additionally at Watton-at-Stone. The Early, the second and the fourth services from Hertford North shall originate from Letchworth Garden City and call at Hitchin, Stevenage and Watton-at-Stone.

#### 5. <u>Service Pattern: Sundays</u>

- 5.1 Between and including the Early and Late Services, 36 services shall be provided at half-hourly intervals from London (Kings Cross) to Hertford North, calling at the stations in paragraph 1.1. One service an hour shall be extended to Stevenage, calling at Watton-at-Stone.
- 5.2 Between and including the Early and Late Services, 35 services shall be provided at half-hourly intervals from Hertford North to London (Kings Cross), calling at the stations in paragraph 1.1. One service per hour shall have originated from Stevenage, calling at Watton-at-Stone.
- 5.3 Two services shall be provided from Hertford North to London (Kings Cross), at half hourly intervals, the second one half an hour before the Early Service, calling at the stations specified in paragraph 1.1.
- The Late service shall be provided from London (Kings Cross) to Letchworth Garden City, calling at all intermediate stations.

#### 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London (Moorgate / Kings Cross) to Hertford North	06:20	06:15	06:45	
Hertford North to London (Moorgate / Kings Cross)	06:00	06:00	08:30	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Hertford North	01:00	00:45	23:15
Hertford North to London (Moorgate / Kings Cross)	23:30	23:15	23:30

## 7. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Finsbury Park to Hertford North	37 minutes, with 12 exceptions of 38 minutes permitted.	38 minutes	38 minutes
Hertford North to Finsbury Park	36 minutes.	35 minutes	35 minutes

#### ROUTE B: LONDON (MOORGATE / KINGS CROSS) TO WELWYN GARDEN CITY.

#### 1. Route Definition

- 1.1 Services shall be provided between London (Moorgate) and Welwyn Garden City, calling at Old Street, Essex Road, Highbury & Islington, Drayton Park, Finsbury Park, Harringay, Hornsey, Alexandra Palace, New Southgate, Oakleigh Park, New Barnet, Hadley Wood, Potters Bar, Brookmans Park, Welham Green and Hatfield.
- 1.2 Services in paragraph 1.1 may extend to or originate from London (Kings Cross) as described below.

#### 2. <u>Service Pattern Monday to Friday: London Departures.</u>

2.1 Between the Early Service and 09:14 a half-hourly service shall be provided, calling at the stations described in paragraph 1.1. Between 06:00 and 06:20, one of these services shall depart from London Kings Cross, calling at the stations described in paragraph 1.1.

Between 08:15 and 08:30, one additional service shall be provided from London (Moorgate) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

Between 08:30 and 09:00 one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at Finsbury Park, Potters Bar, Hatfield and Welwyn Garden City.

- 2.2 Between 09:15 and 16:59, a 20 minute interval service shall be provided, calling at the stations described in paragraph 1.1. Two intervals of up to 30 minutes between services departing between 16:00 and 16:59 to Welwyn Garden City shall be permitted.
- 2.3 Between 17:00 and 19:29, 9 services shall be provided with a maximum interval of 20 minutes.
- 2.4 Between 16:45 and 19:50, six additional services shall be provided from London (Kings Cross) to Welwyn Garden City, calling at Finsbury Park, Potters Bar and Hatfield. These services may call additionally at Alexandra Palace, New Southgate, Oakleigh Park, New Barnet and Hadley Wood.

Between 15:30 and 15:45, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City calling at Finsbury Park, New Southgate, Oakleigh Park, New Barnet, Potters Bar and Hatfield.

- 2.5 Between 19:30 and 22:00, a half-hourly service shall be provided from London (Moorgate) to Welwyn Garden City, calling at the stations described in paragraph 1.1.
- 2.6 Between 22:00 and 23:59, a half-hourly service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations described in paragraph 1.1.

Between 00;00 and 01:15, two additional services shall be provided from London (Kings Cross) to Welwyn Garden City, at hourly interval, calling at the stations defined in paragraph 1.1.

#### 3. Service Pattern Monday to Friday: Welwyn Garden City Departures.

3.1 Between and including the Early Service and 08:59, ten services shall be provided to Moorgate at a maximum interval of 45 minutes, calling at the stations described in paragraph 1.1. Two departures between 07:20 and 08:20 need not call at Alexandra Palace, and one of these need not call at Hornsey and Harringay, provided that the maximum interval between services at these stations does not exceed 35 minutes.

Between 08:15 and 08:30, one additional service shall be provided to Moorgate, calling at the stations specified in paragraph 1.1.

3.2 Between 08:00 and 10:00, three additional services shall be provided at a maximum interval of one hour from Welwyn Garden City to London (Kings Cross), calling at Hatfield, Potters Bar and Finsbury Park. Two of these services shall also call at New Barnet, Oakleigh Park, New Southgate and Alexandra Palace. The other additional service shall also call at New Barnet.

Two additional services shall be provided to London (Kings Cross) at hourly intervals, calling at the stations described in paragraph 1.1. The second service shall depart 40 minutes after the Early Service.

Between 04:00 and 04:15, one additional service shall be provided to London (Kings Cross), calling at the stations described in paragraph 1.1.

- 3.3 Between 09:00 and 16:14, services shall be provided at 20-minute intervals, calling at the stations described in paragraph 1.1.
- 3.4 Between 16:15 and 21:14, a half-hourly service shall be provided calling at the stations described in paragraph 1.1.

Between 16:30 and 18:29, three additional services shall be provided to London (Kings Cross) at a maximum interval of 70 minutes, calling only at Hatfield, Potters Bar and Finsbury Park.

3.5 Between 21:15 and 23:00 a half-hourly service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations described in paragraph 1.1.

#### 4. Service Pattern: Saturdays

4.1 Between and including the Early and Late Services, 36 services shall be provided from London (Kings Cross) to Welwyn Garden City at half-hourly service intervals, calling at the stations specified in paragraph 1.1.

Between 00:30 and 00:45, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

Between 01:00 and 01:15, one additional service shall be provided from London (Kings Cross) to Welwyn Garden City, calling at the stations specified in paragraph 1.1.

4.2 Between and including the Early and Late Services, 35 services shall be provided from Welwyn Garden City to London (Kings Cross) at half-hourly intervals, calling at the stations specified in paragraph 1.1.

Between 04:00 and 04:15, one additional service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations specified in paragraph 1.1.

Between 05:15 and 05:45, one additional service shall be provided from Welwyn Garden City to London (Kings Cross), calling at the stations specified in paragraph 1.1.

#### 5. Service Pattern: Sundays

- 5.1 Between and including the Early and Late Services, 35 services shall be provided from London (Kings Cross) to Welwyn Garden City at half-hourly intervals calling at all stations specified in paragraph 1.1.
- 5.2 Between and including the Early and Late Services, 34services shall be provided from Welwyn Garden City to London (Kings Cross) at half-hourly intervals, calling at all stations specified in paragraph 1.1.

#### 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Welwyn Garden City	07:00	07:00	07:30
Welwyn Garden City to London (Moorgate / Kings Cross)	05:00	07:00	07:30

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London (Moorgate / Kings Cross) to Welwyn Garden City	23:30	23:30	23:30
Welwyn Garden City to London (Moorgate / Kings Cross)	22:45	22:45	22:45

#### 7. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Finsbury Park to Welwyn Garden City	36 minutes, with five exceptions of 37 minutes permitted	35 minutes, with three exceptions of 36 minutes and two exceptions of 41 minutes permitted	35 minutes, with seven exceptions of 38 minutes and one exception of 39 minutes permitted
Welwyn Garden City to Finsbury Park	35 minutes	35 minutes	35 minutes

## ROUTE C: LONDON (KINGS CROSS) AND STEVENAGE TO HUNTINGDON AND PETERBOROUGH.

#### 1. Route Definition

- 1.1 Services shall be provided between London (Kings Cross) and Peterborough (the Stopping Service), calling at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 1.2 Services shall be provided between London (Kings Cross) and Peterborough (the Semi-Fast Service), calling at Finsbury Park, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 1.3 The services in paragraphs 1.1 and 1.2 shall be planned in such a way to provide a regular interval service to and from London and stations between Hitchin and Peterborough.

## 2. <u>Service Pattern Monday To Friday (The Stopping Service): London (Kings Cross) Departures.</u>

- 2.1 One additional service shall be provided between 05:15 and 05:30 calling at the stations described in paragraph 1.1.
- 2.2 Between and including the Early Service and 15:45, an hourly service shall be provided calling at the stations described in paragraph 1.1.
- 2.3 Two additional services shall be provided between 00:30 and 01:45 at hourly intervals calling at the stations described in paragraph 1.1. One of these services need not call at Arlesey and Sandy.

# 3. <u>Service Pattern Monday to Friday (the Stopping Service): Peterborough Departures.</u>

- 3.1 One additional service shall be provided between 03:15 and 03:30, which shall call at Huntingdon, St Neots, Biggleswade, Hitchin, Stevenage and Finsbury Park. This service shall not be subject to a maximum journey time.
- 3.2 Two additional services shall be provided between 04:00 and 07:00 at a maximum interval of 2 hours, calling at the stations described in paragraph 1.1. The second of these services may omit to call at Hatfield and Potters Bar.

3.3 Between and including the Early and Late Services, 13 services shall be provided from Peterborough to London (Kings Cross) at hourly intervals calling at the stations described in paragraph 1.1. There may be one extended interval of one hour 15 minutes between services departing from Peterborough after 20:00.

#### 4. Service pattern Monday to Friday (The Semi-Fast Service): London Departures

- 4.1 Between the Early Service and 16:29, an hourly service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2. The last of these services shall call additionally at Knebworth, but may terminate at Huntingdon.
- 4.2 Evening Peak semi-fast services from London (Kings Cross) to Peterborough shall be provided, calling at the stations described in paragraph 1.2, as shown in the following table:

Para 4.2 part	Services departing from Kings Cross to:	No. of Services	Service Requirement
(a)	Finsbury Park, Stevenage. Hitchin, Arlesey, Biggleswade, Sandy, St. Neots, Huntingdon, Peterborough	6	Half-hourly between 16:30 and 19:29, with one interval of up to 40 minutes permitted.  Welwyn North One of these services departing between 16:30 and 17;00 shall call.  Knebworth One service departing between 17:00 and 17:30, and another service departing between 18:30 and 19:00, shall call.
(b)	St. Neots, Huntingdon, Peterborough	6	Half-hourly between 16:30 and 19:29.  Hitchin and Biggleswade The first and fifth of these services shall call at Biggleswade. The other four of these services shall call at Hitchin.

- 4.3 Between 19:45 and 19:59 an additional service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2. This train will not be subject to a maximum journey time.
- 4.4 Between 20:00 and the Late Service, an hourly service shall be provided from London (Kings Cross) to Peterborough, calling at the stations described in paragraph 1.2.
- 4.5 Between 20:00 and 21:15, two additional services shall be provided at hourly intervals from London (Kings Cross) to Peterborough, calling at St Neots and Huntingdon. The later of these services shall call additionally at Biggleswade.

Between 22:00 and 23:15, two additional services shall be provided from London (King Cross) to Peterborough to arrive at hourly intervals, calling at Biggleswade, St Neots and Huntingdon only.

On a Saturday morning between 00:00 and 00:15 one additional service shall be provided from London (Kings Cross) to Peterborough, calling at Stevenage, Hitchin, Biggleswade, St Neots and Huntingdon.

# 5. <u>Service Pattern Monday to Friday (the Semi-Fast Service): Peterborough Departures.</u>

- 5.1 One additional service shall be provided between 05:00 and 05:15 calling at the stations described in paragraph 1.2.
- 5.2 Morning Peak semi-fast services from Peterborough to London (Kings Cross) shall be provided at the stations described in paragraph 1.2, as shown in the following table:

Para 5.2 part	Services departing from Peterborough to:	No. of Services	Service Requirement
(a)	Huntingdon, St. Neots, Sandy, Biggleswade, Arlesey, Hitchin, Stevenage, Finsbury Park, Kings Cross	6	Between 05:30 and 09:00, a half- hourly service shall be provided calling at the stations described in paragraph 1.2.  The first of these services need not call at Sandy, Biggleswade or Arlesey.  One interval of 45 minutes and an additional interval of 55 minutes shall be permitted for services departing from Peterborough.
(b)	Huntingdon, St. Neots, Kings Cross	5	Between 06:15 and 08:00, five additional services shall be provided at half-hourly intervals from Peterborough to London (Kings Cross), one interval may be extended to 40 minutes.  Three of these services shall depart Peterborough between 07:10 and 07:50.  Biggleswade The first and the fourth of these services shall call.  Stevenage

call.
Call.

5.3 One additional service shall be provided from Peterborough to London (Kings Cross) between 09:00 and 09:29, calling at Huntingdon, St Neots, Biggleswade and Stevenage.

One additional service shall be provided from Peterborough to London (Kings Cross) between 09:15 and 09:30, calling at the stations specified in paragraph 1.2. This service need not to call at Finsbury Park.

5.4 Between 09:30 and the Late Service, an hourly service shall be provided from Peterborough to London (Kings Cross), calling at the stations described in paragraph 1.2. Services departing Peterborough between 10:45 and 14:00 need not call at Arlesey. There may be one interval of up to 1 hour 15 minutes between 16:45 and 18:15.

One additional service shall be provided from Peterborough to London (Kings Cross) between 22:30 and 23:30 calling at the stations specified in paragraph 1.2.

5.5 One service shall depart Huntingdon between 17:45 and 18:15, calling at St. Neots, Hitchin, Stevenage and Finsbury Park.

#### 6. Service Pattern: Saturdays

- 6.1 Between and including the Early and Late Services, 19services (the Stopping Service) shall be provided at hourly intervals from London (Kings Cross) to Peterborough calling at all stations specified in paragraph 1.1. There may be one interval of 1 hour 15 minutes.
- 6.2 Between and including the Early and Late Services, 18 services (the Stopping Service) shall be provided at hourly intervals from Peterborough to London (Kings Cross) calling at all stations specified in paragraph 1.1
- 6.3 Between and including the Early and Late Services, 18services (the Semi-Fast Service) shall be provided at hourly intervals from London (Kings Cross) to Peterborough calling at the stations specified in paragraph 1.2.

Between 16:30 and 18:45, three additional services shall be provided from London (Kings Cross) to Peterborough at hourly intervals, calling at Biggleswade, St Neots and Huntingdon.

- 6.4 Between and including the Early and Late Services, 18 services (the Semi-Fast Service) shall be provided at hourly intervals from Peterborough to London (Kings Cross) calling at the stations specified in paragraph 1.2.
- 6.5 Three additional services shall depart from Peterborough to London (Kings Cross) between 08:00 and 10:15, at hourly intervals, calling at Huntingdon, St Neots and Biggleswade.
- 6.6 Between 03:00 and 04:15, two additional services may be provided from Peterborough to London (Kings Cross) calling at stations specified in

- paragraph 1.1, except Knebworth, Welwyn North and Harringay. One service may omit to call at Sandy, Arseley, Welwyn Garden City, Hatfield, Potters Bar.
- 6.7 Between 05:00 and 05:30, one additional service shall be provided from Peterborough to London (Kings Cross) calling at stations specified in paragraph 1.1.
- 6.8 Between 09:15 and 09:45 a service to London (Kings Cross) defined in paragraph 6.2 may originate from Huntingdon, calling at stations specified in paragraph 1.1.
- 6.9 Not used
- 6.10 The Late service specified in 6.3 shall call at Finsbury Park, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.
- 6.11 Until the end of the Thameslink/GN franchise the late stopping service from London Kings Cross may terminate at Hitchin, in which case an additional service shall be provided from London Kings Cross to Peterborough calling at the stations described in paragraph 1.2.

The additional service shall provide a connection not exceeding five minutes at Stevenage with the Late Stopping service.

#### 7. Service Pattern: Sundays

7.1 Between and including the Early and Late Services, 16 Semi-Fast services shall be provided at hourly intervals from London (Kings Cross) to Peterborough, calling at the stations specified in paragraph 1.2.

Between 16:15 and 17:45 two additional services shall be provided from London (Kings Cross) to Peterborough, at hourly intervals, calling at Finsbury Park, Potters Bar, Hatfield Herts, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Arlesey, Biggleswade, Sandy, St Neots and Huntingdon.

Between 16:15 and 18:45 three additional services shall be provided from London (Kings Cross) to Peterborough, at least thirty minutes apart, calling at Biggleswade, St Neots and Huntingdon. One of these services shall also call at Sandy.

One additional service shall be provided after the first service described in the paragraph above, and before the second service described in the paragraph above, terminating at Cambridge as part of Route D, calling at Finsbury Park, Potters Bar, Hatfield Herts, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Letchworth, Royston and Cambridge.

Another additional service shall be provided between 18:30 and 19:00 from London (Kings Cross) to Peterborough, calling at Biggleswade, Sandy, St Neots, Huntingdon and Peterborough.

7.2 Between and including the Early and Late Services, 17 Semi-Fast services shall be provided at hourly intervals from Peterborough to London (Kings Cross), calling at the stations specified in paragraph 1.2.

Between 09:00 and 11:30, three additional services shall be provided from Peterborough to London (Kings Cross) at hourly intervals, calling at Huntingdon, St Neots and Biggleswade.

Between 23:00 and 23:15, an additional service shall be provided between Peterborough and London (Kings Cross), calling at Huntingdon, St Neots, Sandy, Biggleswade, Arlesley, Hitchin, Stevenage and Welwyn Garden City.

7.3 Two additional services shall be provided from London (Kings Cross) after the Late Service, departing between 00:30 and 01:45 at a maximum interval of one hour, calling at all stations specified in paragraph 1.2, except that the second of these service need not call at Arlesey and Sandy.

### 8. Early and Late Services

#### Early Service arrives at destination no later than:

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Peterborough (Stopping Service)	08:15	07:00	N/A
Peterborough to London (Kings Cross) (Stopping Service)	12:00	07:00	N/A
London (Kings Cross) to Stevenage and Peterborough (Semi-Fast Service)	07:45	07:45	09:45
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	06:45	07:15	07:30

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday	
London (Kings Cross) to Peterborough (Stopping Service)	01:30	23:45	N/A	
Peterborough to London (Kings Cross) (Stopping Service)	22:15	22:15	N/A	

London (Kings Cross) to Stevenage and Peterborough (Semi-Fast Service)	23:15	23:15	23:15
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	21:45	22:46	21:45

# 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Peterborough (Stopping Service)	1 hour 33 minutes, with one exceptions of 1 hour 37 minutes and two further exceptions of 1 hour 45 minutes permitted	1 hour 29 minutes, with two exceptions of 1 hour 34 minutes and one exception of 1 hour 40 minutes permitted	N/A
Peterborough to London (Kings Cross) (Stopping Service)	1 hour 34 minutes, with three exceptions of up to 1 hour 38 minutes and one further exception of 1 hour 44 minutes permitted	1 hour 32 minutes, with five exceptions of 1 hour 35 minutes and one exception of 1 hour 43 minutes permitted	N/A

# 9. <u>Maximum Journey Times Contd.</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Stevenage and Peterborough (Semi- Fast Service)	1 hour 19 minutes, with three exceptions of 1 hour 22 minutes and one exception of 1 hour and 26 minutes permitted	1 hour 17 minutes, with one exception of 1 hour 21 minutes and two exceptions of 1 hour 32 minutes permitted	1 hour 15 minutes, with three exceptions of 1 hour and 17 minutes, two exceptions of one hour 28 minutes, one exception of 1 hour 33 minutes and two exceptions of 1 hour 50 minutes permitted
Peterborough to Stevenage and London (Kings Cross) (Semi-Fast Service)	1 hour 20 minutes, with two exceptions of 1 hour 25 minutes permitted	1 hour 22 minutes, with two exceptions of 1 hour 31 minutes permitted	1 hour 16 minutes, with five exceptions of 1 hour 19 minutes and two exceptions of 1 hour 24 minutes permitted

#### ROUTE D: LONDON (KINGS CROSS) AND STEVENAGE TO CAMBRIDGE.

# 1. Route Definition

- 1.1 Stopping Services shall be provided between London (Kings Cross) and Cambridge, calling at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Letchworth, Baldock, Ashwell & Morden, Royston, Meldreth, Shepreth, and Foxton.
- 1.2 Semi-Fast Services shall be provided between London (Kings Cross) and Cambridge, calling at Finsbury Park, Stevenage, Hitchin, Letchworth Garden City, Baldock and Royston.
- 1.3 **Limited Stops** shall be provided for the service specified in paragraph 1.2, at Ashwell & Morden, Meldreth, Shepreth and Foxton.
- 1.4 Services may be provided in conjunction with those specified in Route E.

# 2. <u>Service Pattern Monday to Friday (Stopping Service): London (Kings Cross)</u> <u>Departures.</u>

- 2.1 Between and including the Early Service and 16:29, 11 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at the stations described in paragraph 1.1.
- 2.2 Between 16:30 and 19:29 Evening Peak services shall be provided as described in the table below:

Para 2.2	Services departing from Kings Cross to:	No. of Services	Service Requirement
part 2.2.	Stations as described in paragraph 1.1.	7	Services shall depart London (Kings Cross) between 16:30 and 19:29 at half-hourly intervals. One interval of 20 minutes shall be permitted between 16:30 and 17:00.  The service departing between 17:00 and 17:30 need not call at Baldock or Ashwell & Morden.  Four of these services need not call at Potters Bar. The service departing between 17:30 and 18:00 need not call at Meldreth, Shepreth and Foxton.

2.2A Between 18:45 and 19:00 an additional service shall be provided from Royston to Cambridge, calling at all intermediate stations. This service may

- be formed from a portion which detaches at Royston from a London (Kings Cross) to Kings Lynn service that is required to be provided under Route E.
- 2.3 Between 19:30 up to and including the Late Service, four additional services shall be provided at hourly intervals from London (Kings Cross) to Cambridge, calling at the stations in paragraph 1.1.
- 2.4 Between 19:30 and 23:45 five additional services shall be provided at hourly intervals from London (Kings Cross) and Letchworth Garden City, calling at the stations in paragraph 1.1. The first of these services need not call at Potters Bar. The services which depart between 20:30 and 23:30 must be extended to Royston, calling at Royston only.

#### 3. Service Pattern Monday to Friday (Stopping Service): Cambridge Departures.

- 3.1 Between 05:30 and 05:45, an additional service shall be provided from Cambridge to London (Kings Cross), calling at the stations described in paragraph 1.1.
- 3.2 Between 05:00 and 08:45 Morning Peak services shall be provided as described in the table below:

Para 3.2 part	Services departing from Cambridge to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1.	2	One additional service from Cambridge to London (Kings Cross) shall depart in each of the time periods 06:20 - 06:40 and 08:20 - 08:40.  The earlier of these two services need not call at Potters Bar.
(b)	Foxton, Shepreth, Meldreth, Royston, Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, and Finsbury Park.	2	Additional services from Cambridge to London (Kings Cross) shall be provided between 06:45 and 07:30.
(c)	Royston, Ashwell & Morden, Baldock, Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park.	2	Additional services from Royston to London (Kings Cross) shall depart between 05:00 and 06:00.
(d)	Letchworth Garden City, Hitchin, Stevenage, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park	2	Additional services from Letchworth Garden City to London (Kings Cross) shall

			depart between 06:45 and 07:30.
(e)	Letchworth Garden City, Hitchin, Stevenage, Knebworth, Welwyn North, Welwyn Garden City, Hatfield, Potters Bar and Finsbury Park	1	An additional service from Letchworth Garden City to London (Kings Cross) shall depart between 08:30 and 08:45.

- 3.3 Between 09:30 and 10:30 an additional service shall be provided from Letchworth Garden City to London (Kings Cross) calling at the stations described in paragraph 1.1.
- 3.4 Between and including the Early Service and Late Services, 15 services shall be provided from Cambridge to London (Kings Cross) at hourly intervals calling at the stations described in paragraph 1.1. The interval between the penultimate service and Late Service may be extended up to 1 hour 30 minutes. The Late Service shall originate from Kings Lynn as part of a service specified in Route E.
  - 3.5 Between 17:00 and 17:30 an additional service shall be provided from Royston to London (Kings Cross) calling at Letchworth Garden City, Finsbury Park and London (Kings Cross).

# 4. <u>Service Pattern Monday to Friday (Semi-Fast Service): London (Kings Cross)</u> <u>Departures.</u>

- 4.1 Between and including the Early Service and 15:59, 11 services shall be provided at hourly intervals from London Kings Cross to Cambridge calling at the stations specified in paragraph 1.2. Those services departing London (Kings Cross) before 09:00 must additionally call at Ashwell & Morden, Meldreth, Shepreth, and Foxton. The Early Service shall be extended to Kings Lynn as part of a service specified in Route E.
- 4.2 Between 16:15 and 18:59, five additional services shall be provided from London (Kings Cross) to Royston at half-hourly intervals, calling at Stevenage, Letchworth Garden City, Baldock and Ashwell & Morden.. One interval between services departing London (Kings Cross) before 17:30 may be extended up to 1 hour 15 minutes.

One of these services departing London (Kings Cross) between 18:00 and 18:30 shall call additionally at Welwyn North and Hitchin, but need not call at Stevenage.

One of these services departing London (Kings Cross) between 18:30 and 19:00 shall call additionally at Hitchin.

Between 19:00 and 19:30 another additional service shall be provided between London (Kings Cross) and Royston, calling at Welwyn North, Letchworth Garden City, Baldock, Ashwell & Morden, and Royston.

The additional services specified in this paragraph shall be subject to a Maximum Journey Time from London (Kings Cross) to Royston not exceeding 46 minutes, with one exception of 50 minutes permitted.

A further additional service will be provided between London (Kings Cross) and Royston between 22:00 and 23:00, calling at the stations described in paragraph 1.1. The maximum journey time is 60 minutes.

- 4.3 Between 20:00 and the Late Service three services shall be provided at hourly intervals from London (Kings Cross) to Cambridge calling at the stations described in paragraph 1.2. These services must call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.
- 4.4 Between 23:50 and 00:20 one additional service shall be provided between London (Kings Cross) and Cambridge, calling at the stations described in paragraph 1.2. This service must call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.

On Saturday mornings between 00:30 and 00:45, an additional service shall be provided between London (Kings Cross) and Cambridge, calling at Stevenage, Hitchin, Letchworth, Baldock and Royston.

# 5. Service Pattern Monday to Friday (Semi-Fast Service): Cambridge Departures

5.1 Between 06:15 and 09:00, six additional services shall be provided from Royston to London (Kings Cross) calling at Ashwell & Morden, Baldock, Letchworth Garden City, Hitchin, Stevenage and Finsbury Park. Two services departing Royston between 06:30 and 07:00 and between 07:30 and 08:00 respectively must call additionally at Knebworth. The first of the six services must start back from Cambridge, and need not call at Foxton, Shepreth, Meldreth, Ashwell & Morden and Baldock.

Two services departing from Royston between 07:30 and 08:00 may omit to call at Finsbury Park. One of these services may omit to call at Stevenage. One service shall call additionally at Welwyn North.

The additional services specified in this paragraph shall be subject to a Maximum Journey Time from Royston to London (Kings Cross) not exceeding 51 minutes, with two exceptions of up to 54 minutes permitted.

5.2 Between and including the Early and Late Services, 15 services shall be provided at hourly intervals from Cambridge to London (Kings Cross), calling at the stations described in paragraph 1.2.

Two of these services departing Cambridge between 10:15 and 10:45 and between 16:15 and 16:45 respectively must start back from Ely, calling at Waterbeach.

The Early Service need not call at Stevenage or Finsbury Park.

The Early Service and the four services departing from Cambridge between 16:00 and 20:00must additionally call at Foxton, Shepreth, Meldreth and Ashwell & Morden.

One interval may be extended to 1 hours and 40 minutes.

Six services originating from Cambridge after 17:00 shall originate from Kings Lynn as part of services specified in Route E.

#### 6. Service Pattern Saturdays: Stopping Service.

- 6.1 Between and including the Early and Late Services, 18 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.1.
- 6.2 Between and including the Early and Late Services, 17services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.1.

Between 23:00 and 23:15, an additional service shall be provided from Cambridge to London (Kings Cross), calling at the stations specified in paragraph 1.1. This service will start back from Kings Lynn and calling in addition at Kings Lynn, Watlington, Downham Market, Littleport, Ely, Waterbeach and Cambridge.

An additional service shall be provided between Royston and London Kings Cross departing between 05:00 and 05:30, calling at stations specified in paragraph 1.1.

#### 7. Service Pattern Saturdays: Semi-Fast Service.

7.1 Between and including the Early and the Late Services, 16 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.2.

Between 05:30 and 05:45, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.2. This service shall call additionally at Ashwell & Morden, Meldreth, Shepreth and Foxton.

- 7.2 Between and including the Early and the Late Services, 16 services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.2.
- 7.3 Two additional services shall be provided from London Kings Cross to Cambridge departing between 23:00 and 23:59, calling at the stations specified in paragraph 1.1. One of these services need only call at Stevenage, Letchworth and Royston.

#### 8. Service Pattern Sundays: Stopping Service

8.1 Between and including the Early and the Late Services, 17 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.1.

An additional service shall be provided between 16:00 and 19:00 from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.1.

8.2 Between and including the Early and the Late Services, 16 services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.1.

Between 08:45 and 11:00, three additional services shall be provided from Letchworth to London (Kings Cross) at hourly intervals, calling at the stations specified in paragraph 1.1.

### 9. Service Pattern Sundays: Semi-Fast Service

9.1 Between the Early and the Late Services, 15 services shall be provided at hourly intervals from London Kings Cross to Cambridge, calling at stations specified in paragraph 1.2.

Between 00:00 and 00:15, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at Stevenage and Hitchin.

Between 06:30 and 06:45, an additional service shall be provided from London (Kings Cross) to Cambridge, calling at the stations specified in paragraph 1.2.

9.2 Between the Early and the Late Services, 17 services shall be provided at hourly intervals from Cambridge to London Kings Cross, calling at stations specified in paragraph 1.2.

# 10. Limited Stops (Semi-Fast Service)

Ashwell & Morden, Meldreth, Shepreth and Foxton.

- 10.1 On Mondays to Fridays, services departing from London (Kings Cross) between 05:30 and 08:00 (including the Early Service) shall call.
- 10.2 On Saturdays the Early Service and the second service from London (Kings Cross) shall call.

### 11. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
London (Kings Cross) to Cambridge (Stopping Service)	07:30	07:30	09:30	
Cambridge to London (Kings Cross) ( Stopping Service)	10:30	07:30	08:30	
London (Kings Cross) to Cambridge (Semi-fast Service)	07:00	08:15	10:00	

Cambridge to London (Kings Cross) ( Semi-fast	09:10	07:45	07:45
Service)			

# 11. Early and Late Services Contd.

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday	
London (Kings Cross) to Cambridge (Stopping Service)	23:00	23:00	00:00	
Cambridge to London (Kings Cross) ( Stopping Service)	23:15	18:30	21:45	
London (Kings Cross) to Cambridge (Semi-fast Service)	22:45	21:45	22:45	
Cambridge to London (Kings Cross) ( Semi-fast Service)	22:15	22:30	22:15	

# 12. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Cambridge (Stopping Service)	1 hour 26 minutes, with two exceptions of 1 hour 30 minutes and two exceptions of 1 hour 32 minutes and three exceptions of 1 hour 36 minutes permitted	1 hour 22 minutes, with two exceptions of 1 hour and 37 minutes permitted	1 hour 21 minutes, with two exceptions of 1 hour 25 minutes, two exceptions of 1 hour 29 minutes and one exception of 1 hour 32 minutes permitted
Cambridge to London (Kings Cross) (Stopping Service)	1 hour 26 minutes, with one exception of 1 hour 30 minutes permitted	1 hour 25 minutes, with one exception of 1 hour 28 minutes and one exception of 1 hour 35 minutes permitted	1 hour 24 minutes, with one exception of 1 hour 28 minutes permitted

London (Kings Cross) to Cambridge (Semi-Fast Service)	1 hour 7 minutes, with three exceptions of 1 hour 9 minutes and two exceptions of 1 hour 13 minutes permitted	1 hour 3 minutes, with two exceptions of 1 hour and 7 minutes permitted	1 hour 2 minutes, with one exception of 1 hour 9 minutes and two exceptions of 1 hour 11 minutes permitted
---	--	--	--

## 12. <u>Maximum Journey Times Contd.</u>

Route	Monday - Friday	Saturday	Sunday
Cambridge to London (Kings Cross) ( Semi-fast Service)	1 hour 4 minutes, with four exceptions of 1 hour 9 minutes and one exception of 1 hour 11 minutes permitted	1 hour 4 minutes, with one exception of 1 hour 6 minutes and one exception of 1 hour 19 minutes permitted	1 hour 2 minutes, with two exceptions of 1 hour 6 minutes, two exceptions of 1 hour 13 minutes and one exception of 1 hour 17 minutes permitted

# ROUTE E: LONDON (KINGS CROSS) TO CAMBRIDGE AND KINGS LYNN.

# 1. Route Definition

- 1.1 Services shall be provided between London (Kings Cross) and Kings Lynn, calling at Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.
- 1.2 Services may be provided by those specified in Route D.

## 2. <u>Service Pattern: Monday to Friday</u>

- 2.1 Between and including the Early and Late Services, 14 services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1. From 16:40 onwards, all services shall call additionally at Letchworth Garden City and Royston, except that one service need not call at either Letchworth Garden City or Royston and one service need only call at Royston.
- 2.2 Between 20:00 and 23:30, four additional services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn, calling at the stations specified in Paragraph 1.1. These services shall call additionally at Letchworth Garden City and Royston.

Between 20:30 and 23:00 an hourly service shall operate from London (Kings Cross) to Ely, calling at Letchworth, Royston, Cambridge and Waterbeach. The first of these services shall extend to Kings Lynn, calling at Littleport, Downham Market and Watlington.

2.3 Between 08:00 and 19:30,12 additional services shall be provided at hourly intervals from London (Kings Cross) to Cambridge. These services, when combined with those specified in paragraph 2.1, shall provide a half-hourly service between London and Cambridge. The first of these services shall extend to Ely, but need not call at Waterbeach.

From 17:00 (departure time from London Kings Cross) one of these services shall call additionally at Letchworth Garden City and two of these services shall call additionally at Royston.

Between 17:00 and 19:30 (departure time from London Kings Cross), three services shall be extended to Ely, but need not call at Waterbeach. The second of these three services shall be further extended to Kings Lynn, calling at Downham Market.

Between 23:00 and 23:59 one additional service shall be provided between London (Kings Cross) and Cambridge.

- 2.4 Between 07:00 and 07:30 one additional service shall be provided from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1.
- 2.5 Between 06:15 and 07:00, two additional services shall be provided from Cambridge to Kings Lynn at an interval of 40 minutes, calling at the stations specified in paragraph 1.1. The second of these services shall originate from Kings Cross as part of a service specified in Route D.
- 2.6 Between 15:15 and 16:30 two services shall be provided at an hourly interval from Cambridge to Ely, calling at Waterbeach. The later of these services shall be extended to Downham Market, but need not call at Littleport.
- 2.7 Between and including the Early and Late Services, 11services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross), calling at stations specified in paragraph 1.1.

Services departing Kings Lynn up to 09:00 shall call additionally at Royston and Letchworth Garden City. One of these services need not call at Letchworth Garden City.

Between 07:00 and 07:15 an additional service shall be provided from Kings Lynn to Cambridge, calling at all intermediate stations. This service shall be extended to London (Kings Cross) calling at the stations specified in paragraph 1.1 and additionally at Royston and Letchworth Garden City. It may fulfil this requirement by attaching at Cambridge to the first of the two services required to be provided by paragraph [2.8] below.

2.8 Between 07:15 and 08:45 two additional services shall be provided at an hourly interval from Kings Lynn to London (Kings Cross) calling at the stations in paragraph 1.1.

The first of these services shall call additionally at Royston and Letchworth Garden City and need not call at Watlington, Littleport or Waterbeach.

Between 17:00 and 17:30 one additional service shall be provided from Downham Market to London (Kings Cross). This service need not call at Littleport and Waterbeach.

- 2.9 Between 10:00 and 18:30, nine additional services shall be provided at hourly intervals from Cambridge to London Kings Cross. These services when combined with those specified in paragraph 2.7 shall provide a half-hourly service from Cambridge to London.

  Between 05:00 and 05:15, one additional service shall be provided from Cambridge to London Kings Cross. This service must call additionally at Royston and Letchworth Garden City.
- 2.10 Between 18:30 and 22:00, a half-hourly service shall be provided from Cambridge to London (Kings Cross). Two services departing from Cambridge between 18;30 and 19:00 and between 20:30 and 21:00 respectively shall start back from Ely, but need not call at Waterbeach.
  - Between 22:00 and 23:00 one further additional service shall be provided between Cambridge and London (Kings Cross).
- 2.11 Between 04:45 and 05:00, one additional service shall be provided from Kings Lynn to London (Kings Cross) calling at Watlington, Downham Market, Littleport, Ely, Waterbeach, Cambridge, Royston and Letchworth Garden City.
- 2.12 Between 06:00 and 06:15 one service shall be provided from Kings Lynn to London (Kings Cross), calling at Downham Market, Ely, Cambridge and Royston.
- 2.13 Between 16:30 and 22:35, seven services shall be provided at hourly intervals from Kings Lynn to Cambridge, calling at the stations specified in Paragraph 1.1. These services shall be extended to London (Kings Cross) as part of services specified in Route D.

# 3. Service Pattern Saturdays

3.1 Between and including the Early and Late Services, 16 services shall be provided at hourly intervals from London (Kings Cross) to Kings Lynn calling at stations specified in paragraph 1.1. The Late service may terminate at Cambridge.

An additional service from London (Kings Cross) to Kings Lynn shall be provided between 21:00 and 21:30, calling at the stations specified in paragraph 1.1

3.2 Between 08:00 and 21:00, 12 additional services shall be provided from London (Kings Cross) to Cambridge. These services when combined with those specified in paragraph 3.1 shall provide a half hourly service between London and Cambridge between 08:00 and 19:30.

Between 18:30 and 19:30 one of these services shall be extended to Ely running non-stop between Cambridge and Ely.

Between 20:00 and 21:00 one of these services shall be extended to Ely calling at Letchworth, Royston, Cambridge, Waterbeach and Ely.

An additional service from London (Kings Cross) to Cambridge shall be provided between 22:30 and 23:00, calling at Letchworth Garden City, Royston and Cambridge.

3.3 Between and including the Early and Late Services, 17 services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross) calling at stations specified in paragraph 1.1. One interval may be reduced to 40 minutes.

Between 09:15 and 09:30, an additional service shall be provided from Kings Lynn to London (Kings Cross) calling at Downham Market, Ely and Cambridge.

3.4 Between 07:00 and 20:30,13 additional services shall be provided between Cambridge and London (Kings Cross). These services when combined with those specified in paragraph 3.3 shall provide a half hourly service between Cambridge and London.

Between 08:30 and 09:30 one of these services will start from Ely and run fast to Cambridge.

Between 21:00 and 23:30, three additional services will operate between Cambridge and London (Kings Cross), running non-stop to London Kings Cross.

- 3.5 Between 22:00 and 23:15, two additional services shall be provided from London (Kings Cross) to Kings Lynn, at hourly intervals, calling at Letchworth, Royston, Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.
- 3.6 Between 06:30 and 06:45, one additional service shall be provided from Cambridge to Kings Lynn calling at Waterbeach, Ely, Littleport, Downham Market and Watlington.

# 4. Service Pattern Sundays

4.1 Between and including the Early Service and 13:59, thive services shall be provided at two-hourly intervals from London (Kings Cross) and Kings Lynn calling at stations specified in paragraph 1.1.

Between 07:45 and 08:00, an additional service shall be provided from London (Kings Cross) to Kings Lynn, calling at Finsbury Park, Stevenage, Hitchin, Lechworth, Baldock, Royston, Cambridge, Waterbeach, Ely, Littleport, Downham Market and Watlington.

4.2 From 14:00 up to and including the Late Service, ten services shall be provided at hourly intervals from London (Kings Cross) and Kings Lynn calling at stations specified in paragraph 1.1.

- 4.3 Between and including the Early Service and 11:59, four services shall be provided at two-hourly intervals from Kings Lynn to London (Kings Cross) calling at all stations specified in paragraph 1.1.
- 4.4 From 12:00 up to and including the Late Service, eleven services shall be provided at hourly intervals from Kings Lynn to London (Kings Cross) calling at all stations specified in paragraph 1.1.
  - Between 17:45 and 18:00, one additional service shall be provided from Kings Lynn to London (Kings Cross), at a half hourly interval with the services specified in the paragraph above, calling at the stations specified in paragraph 1.1.
- 4.5 One additional service shall be provided from Cambridge to Kings Lynn to arrive at Kings Lynn no more than two hours before the Early Service.

# 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
London (Kings Cross) to Kings Lynn	08:30	08:30	11:00	
Kings Lynn to London (Kings Cross)	08:00	07:45	10:15	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London (Kings Cross) to Kings Lynn	19:30	21:30	23:00
Kings Lynn to London (Kings Cross)	15:45	21:30	22:15

# 6. <u>Maximum Journey Times</u>

Route Monday - Friday		Saturday	Sunday
London (Kings Cross) to Kings Lynn	1 hour 39 minutes, with five exceptions of 1 hour 41 minutes, four exceptions of 1 hour 44 minutes and one exception of 2 hours permitted	1 hour 36 minutes, with four exceptions of 1 hour 41 minutes, two exception of 1 hour 48 minutes and 1 exception of 1 hour 57 minutes permitted	1 hour 35 minutes, with one exception of 1 hour 44 minutes and one exception of 2 hours permitted
Kings Lynn to London (Kings Cross)	1 hour 49 minutes, with two exceptions of 1 hour 52 minutes, three exceptions of 1 hour 59 minutes and one exception of 2 hours 12 minutes permitted	1 hour 39 minutes, with three exceptions of 2 hours and one exception of 2 hours 13 minutes permitted	1 hour 40 minutes, with one exception of 1 hour 41 minutes, one exception of 1 hour 51 minutes and one exception of 2 hours 4 minutes permitted

### ROUTE F: BEDFORD TO LONDON BLACKFRIARS

# 1. Route Definition

- 1.1 Services shall be provided between Bedford and London Blackfriars calling at Flitwick, Harlington, Leagrave, Luton, Luton Airport Parkway, Harpenden, St. Albans, St. Pancras International, Farringdon and City Thameslink.
- 1.2 **Limited Stops** shall be made at Radlett, Elstree, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town as described.
- 1.3 Services may be combined with those specified in Routes H, I and J to provide cross-London services with one train fulfilling both requirements.

# 1.4 <u>City Thameslink</u>

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.5 **Farringdon**

Services need not call at Farringdon between 00:45 and 05:00.

# 2. Service Pattern: Mondays to Fridays

2.1 Between 02:00 and 05:20 six services shall be provided from Bedford to London Blackfriars at a maximum interval of 1 hour.

Three of these services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

2.2 Between 05:30 and the Early Service morning peak services shall be provided from Bedford to London Blackfriars as described in the table below:

Para 2.2 part	Services departing from Bedford to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1.	(11)	These services shall be provided at 20 minute intervals. One interval of 25 minutes may be permitted. These services shall depart between 05:30 and the Early Service.  A maximum interval of 20 minutes shall apply. One interval of 24 minutes shall be permitted.

Para 2.2 part	Services departing from Bedford to:	No. of Services	Service Requirement
(a) Contd.			One service departing Bedford between 07:00 and 07:30 shall call additionally at Radlett, Elstree & Borehamwood and West Hampstead Thameslink.  Four of these services shall call additionally at West Hampstead Thameslink.  One service may start from Flitwick and shall call
			additionally at Radlett, Elstree & Borehamwood and West Hampstead Thameslink.
(b)	Flitwick, Luton, Harpenden, St. Albans, St. Pancras International, Farringdon, City Thameslink and London	9	These additional services shall depart Bedford between 05:45 and 08:00.
	Blackfriars.		These services shall be provided at 20 minute intervals.
			Three services departing between 07:00 and 08:00 shall call at Leagrave.
			Two of these services may be extended to Elephant & Castle.
			Two services need not call at Flitwick

2.3 Between and including the Early and Late Services 41 services shall be provided at 15 minute intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1.

Two services per hour at half hourly intervals may call additionally at West Hampstead Thameslink.

2.4 Between 09:00 and 10:30 three additional services shall be provided from Bedford to London Blackfriars at a maximum interval of 50 minutes, calling at Flitwick, Luton, Luton Airport Parkway, Harpenden, St. Albans, St. Pancras International, Farringdon and City Thameslink.

These services may be extended to Elephant & Castle.

2.5 Between 15:45 and 17:45, four additional services shall be provided from Bedford to London Blackfriars, calling at the stations in paragraph 1.1.

All of these services need not call at Harlington and Leagrave. Two services need not call at Harpenden and one service need not call at Flitwick.

2.6 Between 18:00 and the Late Service, two additional services shall be provided from Bedford to London Blackfriars at an interval of 30 minutes.

These services shall call additionally at West Hampstead Thameslink.

These services need not call at Flitwick, Harlington or Leagrave, but may call additionally at Radlett and Elstree & Borehamwood.

2.7 Between the Late Service and 22:29 eight services shall be provided from Bedford to London Blackfriars at a maximum 30 minute interval calling at the stations in paragraph 1.1.

Seven of these services shall call additionally at West Hampstead Thameslink.

2.8 Between 22:30 and 01:59, five services shall be provided from Bedford to London Blackfriars at a maximum 60 minute interval calling at the stations in paragraph 1.1.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

2.9 Between 02:00 and 05:59 five services shall be provided from London Blackfriars to Bedford calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour and 20 minutes.

These services shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

2.10 Between 06:00 and the Early Service four services shall be provided from London Blackfriars to Bedford at a maximum interval of 30 minutes calling at the stations in paragraph 1.1.

Three of these services shall call additionally at West Hampstead Thameslink.

2.11 Between and including the Early and Late Services 38 services shall be provided at 15 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1. Between 07:30 and 09:00 there may be one interval of 25 minutes.

Two services per hour at half houly intervals may call additionally at West Hampstead Thameslink.

2.12 Between 08:15 and 09:30, four additional services shall be provided at 15 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. One interval of 35 minutes may be permitted.

These services need not call at Harpenden and Harlington. Three services need not call at Leagrave. One service need not call at St. Albans.

2.13 Between 16:00 and 16:30 one additional service shall be provided from London Blackfriars to Bedford calling at City Thameslink, Farringdon, St. Pancras International, St. Albans, Harpenden, Luton and Flitwick.

This service may originate from Elephant & Castle.

2.14 Between the Late Service and 19:09 evening peak services shall be provided from London Blackfriars to Bedford as described in the table below:

Para 2.14 part	Services departing from London Blackfriars to:	No. of Services	Service Requirement
(a)	Stations as described in paragraph 1.1	12	Services shall be provided at 20 minute intervals.  Two services departing London Blackfriars between 17:30 and 18:30 shall call additionally at West Hampstead Thameslink, Mill Hill Broadway, Elstree & Borehamwood and Radlett.  One of these services may originate from Elephant & Castle.  Two further services shall call additionally at West Hampstead Thameslink
(b)	City Thameslink, Farringdon, St. Pancras International, St. Albans, Harpenden, Luton, Leagrave, Flitwick and Bedford	8	Services shall be provided at 30 minute intervals.  One service need not call at Harpenden  One service shall call additionally at Leagrave.  One service need not call at Flitwick.

2.15 Between 19:10 and 21:59, twelve services shall be provided at 15 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1.

Two services an hour shall call additionally at West Hampstead Thameslink.

2.16 Between 22:00 and 01:30 seven services shall be provided at 30 minute intervals from London Blackfriars to Bedford calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

Services departing London Blackfriars after 00:01 shall call additionally at Kentish Town, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

# 3. Service Pattern: Saturdays

3.1 Between 02:00 and the Early Service nine services shall be provided at half hourly intervals, calling at the stations is paragraph 1.1. An interval of 60 minutes may be permitted between the first and second services and one exception of 40 minutes may be permitted.

Seven services shall call additionally at West Hampstead Thameslink and four shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood and Kentish Town.

- 3.2 Between and including the Early and Late Services 50 services shall be provided at 15 minute intervals from Bedford to London Blackfriars, calling at the stations in paragraph 1.1. After 18:00 intervals of 20 minutes may be permitted. Two services per hour at 30 minute intervals shall call additionally at West Hampstead Thameslink.
- 3.3 Between the Late Service and 23:45 eleven services shall be provided from Bedford to London Blackfriars at half-hourly intervals, calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink. Three services departing after 22:00 shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood and Kentish Town.

3.4 Between 02:00 and the Early Service nine services shall be provided from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour 15 minutes.

Services departing before 06:00 shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett. Four services departing between 06:00 and 07:30 shall call additionally at West Hampstead Thameslink.

3.5 Between and including the Early and Late Services 58 services shall be provided at 15 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1. Two services per hour at 30 minute intervals shall call additionally at West Hampstead Thameslink.

3.6 Between the Late Service and 01:15 four services shall be provided at half hourly intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1

These services shall call additionally at West Hampstead Thameslink.

A further four services shall be provided at half hourly intervals from London Blackfriars to Bedford calling at the stations specified in Paragraph 1.1 and additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

#### 4. Service Pattern: Sundays

4.1 Between 05:30 and the Early Service four services shall be provided at half hourly intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1. One service may terminate at London St Pancras.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

4.2 Between the Early and Late Services 25 services shall be provided at half hourly intervals from Bedford to London Blackfriars calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

4.3 Between the Early Service and 17:45, 20 additional services shall be provided at thrity minute intervals from Bedford to London Blackfriars, calling at the stations in paragraph 1.1.

These services may call additionally at West Hampstead Thameslink.

4.4 Between the Late Service and 02:00 ten services shall be provided from Bedford to London Blackfriars, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 60 minutes.

These services shall call additionally at Radlett, Elstree & Borehamwood, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink and Kentish Town.

4.5 Between the Early and Late Services 36 services shall be provided at half hourly intervals between London Blackfriars and Bedford, calling at the stations in paragraph 1.1. One interval of 1 hour shall be permitted before 07:00. The early service and the second service may originate at St Pancras International.

These services shall call additionally at West Hampstead Thameslink. All services before 10:00 and four services between 22:00 and 00:15 shall call additionally at Kentish Town, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

4.6 Between 10:30 and 19:45, 19 additional services shall be provided at 30 minute intervals from London Blackfriars to Bedford, calling at the stations in paragraph 1.1.

These services shall call additionally at West Hampstead Thameslink.

4.7 Between the Late Service and 01:15 two additional services shall be provided at a 30 minute interval from London Blackfriars and Bedford, calling at the stations in paragraph 1.1. This service shall call additionally at Kentish Town, West Hampstead Thameslink, Cricklewood, Hendon, Mill Hill Broadway, Elstree & Borehamwood and Radlett.

# 5. <u>Early and Late Services</u>

	Early Service arrives at destination no later than:			
Route	Monday - Saturday Sund			
Bedford to London Blackfriars	09:45	07:45	09:15	
London Blackfriars to Bedford	08:40	09:00	07:30	

	Late Service departs no earlier than:		
Route	Monday - Friday	Sunday	
Bedford to London Blackfriars	18:30	18:30	19:45
London Blackfriars to Bedford	16:30	21:45	23:45

# 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday Sunday	
Bedford to London Blackfriars	1 hour and 11 minutes, with 9 exceptions of 1 hour and 16 minutes	1 hour and 13 minutes, with 10 exceptions of 1 hour and 25 minutes permitted.	1 hour and 15 minutes, with 9 exceptions of 1 hour and 25 minutes permitted
London Blackfriars to Bedford	AVCANTIONS Of 1		1 hour and 15 minutes, with 10 exceptions of 1 hour and 28 minutes permitted

#### ROUTE G: LUTON AND ST. ALBANS TO LONDON BLACKFRIARS

#### 1. Route Definition

- 1.1 Services shall be provided between Luton and London Blackfriars, calling at Luton Airport Parkway, Harpenden, St. Albans, Radlett, Elstree, Mill Hill Broadway, Hendon, Cricklewood, West Hampstead Thameslink, Kentish Town, St. Pancras International, Farringdon and City Thameslink.
- 1.2 Limited Stops shall be provided at Leagrave, Harlington, Flitwick and Bedford as described.
- 1.3 Services may be combined with those specified in Routes H, I and J to provide cross-London services with one train fulfilling both requirements.

#### 1.4 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.5 **Farringdon**

Services need not call at Farringdon between 00:45 and 05:00.

## 2. Service Pattern: Mondays To Fridays

2.1 Between 05:15 and 06:00 one service shall be provided from St. Albans to London Blackfriars calling at the stations described in paragraph 1.1.

This service shall start from Luton calling at Luton Airport Parkway and Harpenden.

2.2 Between 06:00 and the Early Service morning peak services shall be provided from Luton and St. Albans calling at stations at the maximum intervals as shown below:

Para 2.2 part	Services timed at St. Albans	No. of Services	Service Requirement
(a)	Luton, Luton Airport Parkway, Harpenden	8	Services shall be provided at 30 minute intervals. One interval of 38 minutes may be permitted.
(b)	St. Albans, Radlett, Elstree & Borehamwood, West Hampstead Thameslink, London St. Pancras International, Farringdon, City Thameslink, London Blackfriars	14	Services shall be provided at 20 minute intervals.

Para 2.2 part	Services timed at St. Albans	No. of Services	Service Requirement
(c)	Mill Hill Broadway, Hendon, Cricklewood, Kentish Town	10	Services shall be provided at 20 minute intervals.

2.3 Between and including the Early and Late Services 44 services shall be provided at 15-minute intervals from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

There may be one interval of 35 minutes between 18:00 and 19:00.

After 19:00, services shall be provided at half hourly intervals.

Two of the services specified in paragraph 2.3 in each hour shall originate from Luton calling at Luton Airport Parkway and Harpenden.

Two intervals of up to 70 minutes may be permitted.

2.5 Between 09:00 and 09:15, an additional service shall be provided from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

This service shall originate at Luton, calling at Luton Airport Parkway and Harpenden and need not call at Mill Hill Broadway, Hendon, Cricklewood or Kentish Town.

2.6 Between 17:30 and 19:45, five additional services shall be provided at an interval of 30 minutes from St. Albans to London Blackfriars, calling at Radlett, Elstree & Borehamwood, West Hampstead Thameslink, St. Pancras International, Farringdon and City Thameslink.

One interval of 45 minutes may be permitted.

Three services shall originate at Luton calling at Luton Airport Parkway and Harpenden.

2.7 Between the Late Service and 23:30 three services shall be provided from St. Albans to London Blackfriars calling at the stations in paragraph 1.1.

These services shall originate from Luton calling at Luton Airport Parkway and Harpenden.

2.8 Between 05:50 and 06:54 three services shall be provided from London Blackfriars to St. Albans calling at the stations described in paragraph 1.1.

A maximum interval of 30 minutes shall apply.

One service shall extend to Bedford calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.

2.9 Between 06:55 and the Early Service, morning Contra-peak services shall be provided from London Blackfriars calling at stations at the maximum intervals as shown below:

Para. 2.9 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(a)	City Thameslink, Farringdon, St. Pancras International, West Hampstead Thameslink, Elstree & Borehamwood, Radlett and St. Albans.	15	Services shall be provided at 20 minutes intervals with one exception of 25 minutes.
(b)	Kentish Town, Cricklewood, Hendon, Mill Hill Broadway	9	Services shall be provided at 20 minutes intervals with one exception of 30 minutes.
(c)	Harpenden, Luton Airport Parkway and Luton	8	Services shall be provided at 20 minutes intervals with one interval of 30 minutes and one interval of 50 minutes.

- 2.10 Between and including the Early and Late Services 28 services shall be provided at 15-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.
- 2.11 Two of the services specified in paragraph 2.10 in each hour shall be extended to Luton calling at Harpenden and Luton Airport Parkway. One service between 11:45 and 12:15 shall be extended to Bedford.
- 2.12 Between the Late Service and 18:54, evening peak services shall be provided from London Blackfriars calling at stations at the intervals as shown below:

Para. 2.12 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(a)	Elephant & Castle	2	Two services may originate from Elephant & Castle.
(b)	City Thameslink, Farringdon, St. Pancras International, West Hampstead Thameslink, Mill Hill Broadway, Elstree & Borehamwood, Radlett and St. Albans	10	Services shall be provided at 15 minute intervals
(c)	Kentish Town, Cricklewood and Hendon	9	Services shall be provided at 15 minutes intervals with one interval of 20 minutes permitted.

Para. 2.12 part	Services timed from London Blackfriars	No. of Services	Service Requirement
(d)	Harpenden, Luton Airport Parkway, Luton	4	Services shall be provided at 30 minute intervals.

2.13 Between 18:55 and 19:54, four services shall be provided at 15-minute intervals from London Blackfriars to St. Albans, calling at the stations in paragraph 1.1.

Two of these services shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.

2.14 Between 19:55 and 21:54, eight services shall be provided at 15-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.

Four of these services shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.

Between 21:55 and 23:55, four services shall be provided at 30-minute intervals from London Blackfriars to St. Albans calling at the stations in paragraph 1.1.

These services shall be extended to Bedford calling at Leagrave, Harlington, Flitwick and Bedford.

# Additional services from West Hampstead Thameslink/Kentish Town towards London

2.15 Between 09:50 and 16:30, fourteen services shall be provided at 30-minute intervals from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

Twelve of these services will start back from West Hampstead Thameslink.

Between 05:45 and 06:00, one additional service shall be provided from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

Between 16:00 and 16:15, one additional service shall be provided from West Hampstead Thameslink to London Blackfriars, calling at Kentish Town, St. Pancras International, Farringdon and City Thameslink.

2.16 Between 17:00 and 17:30, one service shall be provided from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink. This train will start back from West Hampstead Thameslink.

2.17 Between 20:00 and 21:00, two services shall be provided at a 30-minute interval from Kentish Town to London Blackfriars, calling at St. Pancras International, Farringdon and City Thameslink.

One of these services will start back from West Hampstead Thameslink.

# Additional services from London to Kentish Town/West Hampstead Thameslink

2.18 Between 08:40 and 10:00, four services shall be provided at a maximum interval of 40 minutes from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.

Between 05:30 and 05:45, one additional service shall be provided from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.

2.19 Between 10:00 and 16:15, thirteen services shall be provided at 30-minute intervals from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.

Ten of these services shall be extended to West Hampstead Thameslink.

2.20 Between 19:30 and 21:10, four services shall be provided at a maximum interval of 30 minutes from London Blackfriars to Kentish Town, calling at City Thameslink, Farringdon and St. Pancras International.

Three of these services shall be extended to West Hampstead Thameslink.

### 3. Service Pattern Saturdays

- 3.1 Between 05:30 and the Early Service three additional services at half-hourly intervals shall be provided from Luton to London Blackfriars, calling at the stations in paragraph 1.1.
- 3.2 Between and including the Early and Late Services 53 services shall be provided at 15-minute intervals from St. Albans to London Blackfriars, calling at the stations in paragraph 1.1. One interval of 20 minutes shall be permitted.
- Two of the services specified in paragraph 3.2 each hour departing before 08:15 shall have originated from Luton, calling at Luton Airport Parkway and Harpenden.
- 3.4 Between the Late Service and 23:00, six additional services shall be provided at half hourly intervals from Luton to London Blackfriars, calling at the stations in paragraph 1.1.
- 3.5 Between 06:00 and the Early Service four additional services shall be provided at half hourly intervals from London Blackfriars to Luton, calling at the stations in paragraph 1.1. One service may be extended to Bedford calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.

- 3.6 Between and including the Early and Late Services 56 services shall be provided at 15-minute intervals from London Blackfriars to St. Albans, calling at the stations in paragraph 1.1.
- Two of the services specified in paragraph 3.6 each hour shall be extended to Luton, calling at Harpenden and Luton Airport Parkway.
- 3.8 Between the Late Service and 23:45, five additional services shall be provided at half hourly intervals from London Blackfriars to St Albans, calling at the stations in paragraph 1.1.

Four of these services shall be extended to Bedford, calling at Harpenden, Luton Airport Parkway, Luton, Leagrave, Harlington and Flitwick.

One interval of 35 minutes shall be permitted.

# 4. <u>Service Pattern Sundays</u>

- 4.1 Between and including the Early and Late Services 25 services shall be provided at half hourly intervals between Luton and London Blackfriars, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from London Blackfriars and Luton, calling at the stations in paragraph 1.1.

## .

### 5. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
St. Albans to London Blackfriars	09:50	08:00	09:30
London Blackfriars to St. Albans	10:45	08:35	11:15

	Late Service departs no earlier than:		:
Route	Monday - Friday	Saturday	Sunday
St. Albans to London Blackfriars	21:30	20:00	20:30
London Blackfriars to St. Albans	16:15	21:30	21:30

# 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
St. Albans to London Blackfriars	50 minutes	47 minutes	42 minutes

London Blackfriars to St. Albans	46 minutes	44 minutes	43 minutes
-------------------------------------	------------	------------	------------

# ROUTE H: BRIGHTON AND GATWICK AIRPORT TO LONDON ST. PANCRAS INTERNATIONAL

### 1. Route Definition

1.1 Services (stopping services) shall be provided between Brighton and London St. Pancras International, calling at Preston Park, Hassocks, Burgess Hill, Wivelsfield, Haywards Heath, Balcombe, Three Bridges, Gatwick Airport, East Croydon, London Blackfriars, City Thameslink and Farringdon.

On Sundays these services need not call at Balcombe, Preston Park and Wivelsfield.

- 1.2 Services (fast services) shall be provided between Brighton and London St. Pancras International, calling at Haywards Heath, Three Bridges, Gatwick Airport, East Croydon, London Blackfriars, City Thameslink and Farringdon.
- 1.2A Services (London Bridge services) shall be provided between Brighton and London Bridge, calling at Burgess Hill, Haywards Heath, Three Bridges, Gatwick Airport and East Croydon until the Combined Effective Date.
- 1.3 Limited Stops shall be made at Tulse Hill, Herne Hill, and Elephant & Castle as described.
- 1.4 Services may be combined with those specified in Routes F and G to provide cross-London services with one train fulfilling both requirements.

#### 1.5 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 08:45 and after 21:15 on Saturdays or all day on Sundays.

# 1.6 **Farringdon**

Services between 00:45 and 05:00 need not call at Farringdon.

### 2. Service Pattern Monday To Friday (Stopping Services)

2.1 Between 02:00 and 04:59 four services shall be provided from Three Bridges to St. Pancras International, calling at Gatwick Airport, East Croydon and London Blackfriars.

Between 0515 and 0545, one additional service shall be provided from Gatwick Airport to St Pancras International calling at the stations specified in Paragraph 1.1.

2.2 Between 05:00 and the Early Service services shall be provided from Brighton to London St. Pancras International as described in the table below:

Para 2.2 part	Services departing from Brighton to:	No. of Services	Service Requirement
(a)	Preston Park, Hassocks, Haywards Heath, Gatwick Airport, East Croydon, Blackfriars, City Thameslink,	10	Services shall be provided at 25 minute intervals with one exception of 30 minutes and

Farringdon and London St.	one exception of 40 minutes
Pancras International	permitted.

Para 2.2 part	Services departing from Brighton to:	No. of Services	Service Requirement
(b)	Burgess Hill, Three Bridges	9	Services shall be provided at 25 minute intervals with three exceptions of 40 minutes permitted.
(c)	Wivelsfield	7	Services shall be provided at a maximum interval of 65 minutes.
(d)	Balcombe	6	Services shall be provided at a maximum interval of 90 minutes.
(e)	Not used		
(f)	One of these services shall call at Tulse Hill		
(g)	Three of these services shall c	all at Herne Hill	

2.3 Between and including the Early and Late Services fifteen services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1.

One service shall call additionally at Tulse Hill and Elephant & Castle.

Balcombe need only be served hourly.

One service need not call at Wivelsfield.

2.4 Between the Late Service and 18:00, three services shall be provided from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 60 minutes. The last two of these services shall start at Elephant & Castle.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Tulse Hill.

One of these services need not call at Wivelsfield and one need not call at Balcombe.

2.5 Between 18:00 and 23.59, twelve services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.1.

Balcombe need only be served hourly.

Two services need not call at Preston Park and Wivelsfield.

2.6 Between 02:15 and 04:14 three services shall be provided from St. Pancras International to Three Bridges, calling at London Blackfriars, East Croydon and Gatwick Airport.

One of these services may be extended to Brighton, calling at Haywards Heath.

These services may call additionally at Selhurst.

2.7 Between 04:15 and the Early Service, ten services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1. One interval of 40 minutes may be permitted.

One service departing London St. Pancras International between 06:10 and 06:30 shall call at Balcombe.

One service shall call additionally at Tulse Hill. Two of these services may call additionally at Selhurst.

Five of these services need not call at Balcombe and two services need not call at Wivelsfield.

Two of these services need not call at Wivelsfield and Preston Park

2.8 Between and including the Early and Late Services 16 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1. One interval of 40 minutes may be permitted.

Balcombe need only be served hourly.

2.9 Between the Late Service and 18:15, five services shall be provided from St. Pancras International to Brighton calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 30 minutes.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Redhill.

One service may terminate at Three Bridges, calling additionally at Elephant & Castle and Redhill.

One service may omit to call at Balcombe, one service may omit to call at Preston Park and two services may omit to call at Wivelsfield.

2.10 Between 18:45 and 23:14 nine services shall be provided at 30-minute intervals from St. Pancras International to Brighton calling at the stations in described paragraph 1.1.

Balcombe need only be served hourly.

2.11 Between 23:15 and 02:14 five services shall be provided at a 30 minute interval from St. Pancras International to Three Bridges, calling at Farringdon,

London Blackfriars, East Croydon and Gatwick Airport. The interval between the penultimate and last of these services may be extended to 60 minutes.

## 3. Service Pattern Monday to Friday (fast services)

- 3.1 Between and including the Early and Late Services thirteen services shall be provided at 30 minute intervals from Elephant & Castle to St. Pancras International, calling at London Blackfriars, City Thameslink and Farringdon.
- 3.2 Between the Late Service and 17:30 four services shall be provided from Brighton to St. Pancras International, calling at the stations described in paragraph 1.2. Two of these services shall start at East Croydon and one of them shall start at Blackfriars.

One service shall call additionally at Elephant & Castle and one service shall call additionally at Tulse Hill.

One service may call additionally at Burgess Hill.

3.3 Between 17:45 and 20:45 four services shall be provided at 30 minute intervals from Brighton to St. Pancras International, calling at the stations described in paragraph 1.2.

One of these services may originate from Three Bridges, one service shall originate from Elephant & Castle and two services shall call additionally at Burgess Hill.

3.4 Between 06:15 and the Early Service three services shall be provided from St. Pancras International to Brighton, calling at the stations described in paragraph 1.2.

These services shall call additionally at Three Bridges and two services shall call additionally at Burgess Hill. One service shall call additionally at Balcombe and Hassocks. One service may terminate at Elephant & Castle.

- 3.5 Between and including the Early and Late Services fifteen services shall be provided at 30 minute intervals from St. Pancras International to Elephant & Castle calling at Farringdon, City Thameslink and London Blackfriars.
- 3.6 Between the Late Service and 19:45 four services shall be provided from St. Pancras International to Brighton, calling at the stations described in paragraph 1.2. After 18:00 these services shall be provided at 30 minutes intervals with one interval of 40 minutes permitted. One of these services may terminate at Three Bridges. Two of these services may terminate at Elephant & Castle.

All of these services shall call additionally at Burgess Hill, Three of these services shall call additionally at Three Bridges and Hassocks; one of these services shall call additionally at Balcombe, and one service shall call additionally at Preston Park.

#### 3A. Service Pattern Monday to Friday (London Bridge services)

- 3A.1 The services in this paragraph 3A shall be provided under this SLC2 until the Combined Effective Date.
- 3A.2 Between 08:15 and 15:45, a half-hourly service shall be provided from Brighton to London Bridge, calling at the stations in paragraph 1.2A.
- 3A.3 One additional service between 05:45 and 06:00, and three further additional services at half-hourly intervals between 17:15 and 18:45, shall be provided from Brighton to London Bridge, also calling at the stations in paragraph 1.2A.
- 3A.4 Between 09:30 and 17:00, a half-hourly service shall be provided from London Bridge to Brighton, calling at the stations in paragraph 1.2A.
- 3A.5 One additional service between 07:00 and 07:15, and three further additional services at half-hourly intervals between 18:30 and 20:00, shall be provided from London Bridge to Brighton, also calling at the stations in paragraph 1.2A.

## 3B. <u>Service Pattern Monday to Friday (Daytime Terminating/Originating Three Bridges services)</u>

Between 09:30 and 18:00, sixteen services shall be provided between Three Bridges and London Blackfriars, calling at Gatwick Airport, East Croydon and London Blackfriars. The maximum interval between these services shall be one hour. Three further additional services with the same station calls shall be provided between 19:00 and 21:00.

Between 08:30 and 16:30, sixteen services shall be provided between London Blackfriars and Three Bridges, calling at East Croydon and Gatwick Airport and Three Bridges.

Three further services shall be provided between 17:45 and 19:30. One of these may terminate at Gatwick Airport.

#### 4. Service Pattern Saturday (stopping service)

- 4.1 Between 02:10 and 05:14, five services shall be provided from Three Bridges to St. Pancras International, calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour.
- 4.2 Between 05:15 and the Early Service, two services shall be provided at hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.
- 4.3 Between the Early and Late Services 32 services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.
  - Balcombe need only be served hourly.
- 4.4 Between the Late Service and 00:10, two services shall be provided from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.

The maximum interval between these services shall be 1 hour and 10 minutes.

4.5 Between 02:00 and 04:14, four services shall be provided at hourly intervals from St. Pancras International to Three Bridges, calling at the stations in paragraph 1.1.

These services may call additionally at Selhurst.

4.6 Between 04:15 and the Early Service four services shall be provided from St. Pancras International to Brighton calling at the stations in paragraph 1.1. The maximum interval between these services shall be 1 hour.

The first two services may call additionally at Selhurst.

4.7 Between the Early and Late Services 27 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations in paragraph 1.1.

Balcombe need only be served hourly.

4.8 Between the Late Service and 23:09, seven services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations in paragraph 1.1.

Balcombe need only be served hourly.

4.9 Between 23:10 and 00:40, three services shall be provided at 30-minute intervals from St. Pancras International to Three Bridges, calling at the stations in paragraph 1.1.

Two services shall be extended to Brighton, calling at the stations in paragraph 1.1

One service need not call at Balcombe, Wivelsfield and Preston Park.

#### 5. <u>Service Pattern Saturday (fast services)</u>

- 5.1 One additional service departing between 06:00 and the Early Service shall be provided between Brighton and the stations in paragraph 1.2.
- 5.2 Between and including Early and Late Services 28 services shall be provided at maximum intervals of 30 minutes between Elephant & Castle and St. Pancras International, calling at London Blackfriars, City Thameslink, Farrington and St. Pancras International.
- 5.3 Between and including Early and Late Services 28 services shall be provided at half hourly intervals between St. Pancras International and Elephant & Castle, calling at Farringdon, City Thameslink and London Blackfriars.

#### 5A. Service Pattern Saturday (London Bridge services)

- 5A.1 The services in this paragraph 5A shall be provided under this SLC2 until the Combined Effective Date.
- 5A.2 Between 07:30 and 18:00, a half-hourly service shall be provided from Brighton to London Bridge, calling at the stations in paragraph 1.2A.
- 5A.3 Between 07:30 and 19:00, a half-hourly service shall be provided from London Bridge to Brighton, calling at the stations in paragraph 1.2A.

## <u>5B Service Pattern Saturday (Daytime Originating/Terminating Three Bridges services)</u>

Between 08:30 and 22:15, twenty-six services shall be provided between Three Bridges and London Blackfriars calling at Gatwick Airport and East Croydon. The maximum interval between these services shall be one hour.

The first three of these services may start from Gatwick Airport.

Between 07:00 and 20:00, twenty-six services shall be provided between London Blackfriars and Three Bridges, calling at East Croydon and Gatwick Airport.

#### 6. Service Pattern Sunday (stopping services)

6.1 Between and including the Early and Late Services 36 services shall be provided at half hourly intervals from Brighton to St. Pancras International, calling at the stations in paragraph 1.1.

Between 04:30 and the Early Service, two additional services shall be provided from Three Bridges to London Blackfriars, calling at the stations described in paragraph 1.1. One service shall be extended to Brighton, calling at the stations described in paragraph 1.1.

Between 08:45 and 22:00, 23 additional services shall be provided from Elephant and Castle to St. Pancras International, calling at London Blackfriars, Farringdon and St. Pancras International.

Nineteen of these services shall start back from Three Bridges calling additionally at Gatwick Airport and East Croydon.

A further three services shall start back from Gatwick Airport calling at East Croydon.

6.2 Between and including the Early and Late Services 30 services shall be provided at half hourly intervals from St. Pancras International to Brighton, calling at the stations described in paragraph 1.1.

Between 06:30 and the Early Service, one additional service shall be provided between London Blackfriars and Brighton, calling at the stations described in paragraph 1.1.

Between 08:45 and 18:30, twenty additional services shall be provided at half hourly intervals from St. Pancras International to Three Bridges, calling at

Farringdon, London Blackfriars, East Croydon, Gatwick Airport and Three Bridges.

6.3 Between the Late Service and 01:10 six additional services shall be provided from St. Pancras International and Brighton, calling at the stations described in paragraph 1.1. The maximum interval between these services shall be 60 minutes.

Three of these services need not call at Haywards Heath, Burgess Hill, Hassocks and Brighton.

#### 7. Service Pattern Sundays (fast services and London Bridge services)

No service required.

#### 8. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	10:45	08:15	07:45
St. Pancras International to Brighton (stopping)	10:30	08:00	09:00
Brighton to St. Pancras International (fast)	10:30	08:30	Not applicable
St. Pancras International to Brighton (fast)	10:40	08:00	Not applicable

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	16:00	21:50	23:30
St. Pancras International to Brighton (stopping)	16:15	19:15	21:15
Brighton to St. Pancras International (fast)	14:50	20:15	Not applicable
St. Pancras International to Brighton (fast)	16:00	19:30	Not applicable

#### 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Brighton to St. Pancras International (stopping)	1 hour and 30 minutes, with three exceptions of 1 hour and 29 minutes permitted.	1 hour and 29 minutes	1 hour and 19 minutes, with one exception of 1hour and 24 minutes permitted.
St. Pancras International to Brighton (stopping)	1 hour and 30 minutes, with three exceptions of 1 hour and 32 minutes permitted.	1 hour and 29 minutes	1 hour and 30 minutes
Brighton to St. Pancras International (fast)	1 hour and 14 minutes, with one exception of 1 hour and 18 minutes permitted.	1 hour and 17 minutes	Not applicable

#### 9. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
St. Pancras International to Brighton (fast)	1 hour and 16 minutes, with two exceptions of 1 hour and 21 minutes permitted.	1 hour and 18 minutes, with two exceptions of 1 hour and 23 minutes permitted.	Not applicable

#### ROUTE I: SUTTON AND WIMBLEDON TO LONDON ST. PANCRAS INTERNATIONAL

#### 1. Route Definition

- 1.1 Services shall be provided between Sutton and London St. Pancras International calling at West Sutton, Sutton Common, St Helier, Morden South, South Merton, Wimbledon Chase, Wimbledon, Haydons Road, Tooting, Streatham, Tulse Hill, Herne Hill, Loughborough Junction, Elephant and Castle, London Blackfriars, City Thameslink and Farringdon.
- 1.2 Services may be combined with those specified in Routes F, G and J to provide cross-London services with one train fulfilling both requirements. Services when combined with those in Route J shall when combined provide a 15 minute service between St. Pancras International and Streatham.

#### 1.3 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays or all day on Sundays.

#### 1.4 Farringdon

Services after between 00:45 and 05:00 need not call at Farringdon.

#### 2. Service Pattern: Monday To Friday

- 2.1 Between 05:30 and the Early Service two additional services shall be provided from Streatham to St. Pancras International calling at the stations in paragraph 1.1 except that these services may originate at Selhurst and one need not call at Loughborough Junction. The maximum interval between these services shall be 30 minutes.
- 2.2 Between and including the Early and Late Services 30 services shall be provided from Sutton to St. Pancras International at 30 minute intervals calling at the stations in paragraph 1.1. The interval between the first and second service may be 45 minutes.
- 2.3 Between the Late Service and 23:15 four additional services shall be provided from Sutton to St. Pancras International, calling at the stations in paragraph 1.1.
- 2.4 Between 05:00 and the Early Service five additional services shall be provided from St. Pancras International to Sutton calling at the stations in paragraph 1.1. The maximum interval between these services shall be one hour.
- 2.5 Between and including the Early and Late Services 31 services shall be provided from St. Pancras International to Sutton at 30 minute intervals calling at the stations in paragraph 1.1. One interval of 40 minutes and one interval of 45 minutes may be permitted.

The Early and Late Services need not call at Loughborough Junction.

#### 3. Service Pattern Saturday

- 3.1 Between 05:30 and 07:00 three additional services shall be provided between Selhurst and St. Pancras International, calling at the stations in paragraph 1.1. The first of these need not call at Loughborough Junction.
- 3.2 Between 06:45 and the Early Service one additional service shall be provided from Wimbledon to St. Pancras International, calling at the stations in paragraph 1.1.
- 3.3 Between and including the Early and Late Services 30 services shall be provided at half hourly intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1. One interval of 45 minutes shall be permitted between the penultimate and Late Services.
- 3.4 Between the Late Service and 23:30 two additional services shall be provided from Wimbledon to St. Pancras International, calling at the stations in paragraph 1.1. The second service need not call at Loughborough Junction.
- 3.5 Between 06:00 and the Early Service one additional service shall be provided from London Blackfriars to Sutton, calling at the stations in paragraph 1.1, except Loughborough Junction.
- 3.5 Between and including the Early and Late Services 35 services shall be provided at half hourly intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1.

#### 4. <u>Service Pattern Sunday</u>

- 4.1 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals between Sutton and St. Pancras International, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 25 services shall be provided at half hourly intervals between St. Pancras International and Sutton, calling at the stations in paragraph 1.1.

#### 5. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Wimbledon)	07:30	08:15	10:45
St. Pancras International to Sutton (via Wimbledon)	09:45	07:45	10:15

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Wimbledon)	20:15	21:45	21:30
St. Pancras International to Sutton (via Wimbledon)	23:30	23:45	21:00

## 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Wimbledon)	1 hour and three minutes with five exceptions of 1 hour and seven minutes permitted.	1 hour and 7 minutes, with one exception of 1 hour 12 minutes permitted.	59 minutes
St. Pancras International to Sutton (via Wimbledon)	1 hour and three minutes, with seven exceptions of 1 hour and 9 minutes permitted.	1 hour and 4 minutes, with one exception of 1hour 7 minutes and one exception of 1hour 12 minutes permitted.	59 minutes

## ROUTE J: SUTTON AND MITCHAM JUNCTION TO LONDON ST. PANCRAS INTERNATIONAL

#### 1. Route Definition

- 1.1 Services shall be provided between Sutton and London St. Pancras International, calling at Carshalton, Hackbridge, Mitcham Junction, Streatham, Tulse Hill, Herne Hill, Loughborough Junction, Elephant & Castle, London Blackfriars, City Thameslink and Farringdon.
- 1.2 Services may be combined with those specified in Routes F, G and I to provide cross-London services with one train fulfilling both requirements. Services when combined with those in Route I shall when combined provide a 15 minute service between St. Pancras International and Streatham.

#### 1.3 **City Thameslink**

Services need not call at City Thameslink before 06:00 and after 23:10 on Mondays to Fridays and before 09:00 and after 21:15 on Saturdays.

#### 1.4 **Farringdon**

Services after between 00:45 and 05:00 need not call at Farringdon.

#### 2. <u>Service Pattern: Monday To Friday</u>

2.1 Between and including the Early and Late Services twenty services shall be provided at 30 minute intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1. One service departing Sutton between 16:15 and 16:45 need not call at Loughborough Junction.

Between 06:30 and the Early Service five additional services shall be provided from Sutton to St. Pancras International at 30 minute intervals, calling at the stations in paragraph 1.1. The interval between the first and the second service may be extended to one hour.

Between the Late Service and 21:30, four additional services shall be provided from Sutton to St. Pancras International at 30 minute intervals, calling at the stations in paragraph 1.1.

- 2.2 Between 05:50 and the Early Service, one additional service shall be provided from St. Pancras International to Sutton calling at the stations in Paragraph 1.1.
- 2.3 Between and including the Early and Late Services 29 services shall be provided at 30 minute intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1. One interval of 45 minutes and two intervals of 40 minutes may be permitted.

Between the Late Service and 21:00, one additional service shall be provided from St. Pancras International to Sutton calling at the stations in Paragraph 1.1.

#### 3. Service Pattern: Saturday

- 3.1 Between and including the Early and Late Services 29 services shall be provided at half hourly intervals from Sutton to St. Pancras International, calling at the stations in paragraph 1.1.
- 3.2 Between and including the Early and Late Services 31 services shall be provided at half hourly intervals from St. Pancras International to Sutton, calling at the stations in paragraph 1.1.

#### 4. Service Pattern Sunday

- 4.1 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from Sutton to London Blackfriars, calling at the stations in paragraph 1.1.
- 4.2 Between and including the Early and Late Services 24 services shall be provided at half hourly intervals from London Blackfriars to Sutton, calling at the stations in paragraph 1.1.

#### 5. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Mitcham Junction)	10:30	08:00	11:00*
St. Pancras International to Sutton (via Mitcham Junction)	07:30	06:45	10:00

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Mitcham Junction)	19:00	21:00	21:45
St. Pancras International to Sutton (via Mitcham Junction)	20:15	20:45	20:00*

<sup>\*</sup>May terminate/start at London Blackfriars

## 6. <u>Maximum Journey Times</u>

Route	Monday - Friday	Saturday	Sunday
Sutton to St. Pancras International (via Mitcham Junction)	48 minutes, with one exception of 50 minutes permitted.	51 minutes	Not applicable
St. Pancras International to Sutton (via Mitcham Junction)	48 minutes, with seven exceptions of 52 minutes permitted.	50 minutes	Not applicable

#### Route K: London Blackfriars to Beckenham Junction

#### 1. ROUTE DEFINITION

1.1 Services shall be provided between London Blackfriars and Orpington calling at Elephant & Castle, Herne Hill, West Dulwich, Sydenham Hill, Penge East, Kent House, Beckenham Junction, Shortlands, Bromley South, Bickley and Petts Wood.

#### 2. Service Pattern Mondays to Fridays

#### **Towards Orpington**

2.1 Between 08:00 and 08:30, one service shall be provided between London Blackfriars and Beckenham Jn, with no requirement to call at intermediate stations.

2.2

During the Evening Peak between 16:00 - 18:30, two services shall be provided at an interval of at least one hour from London Blackfriars to Beckenham Junction, calling at the stations specified in Paragraph 1.1. . One service shall call at Loughborough Junction.

#### **Towards London**

- 2.3 NOT USED
- 2.4 Between 04:45-05:15, one service shall be provided from Orpington to London Blackfriars, calling at the stations specified in Paragraph 1.1.
- 2.5 NOT USED
- 2.6 NOT USED
- 2.7 During the Morning Peak four services shall be provided from Orpington to London Blackfriars, calling at the stations specified in Paragraph 1.1. Two services may originate from Beckenham Junction. These services shall call additionally at Loughborough Junction. Two of these services shall arrive at London Blackfriars between 08:00 and 09:00.
- 2.8 Between 18:15 and 18:45, one service shall be provided between Beckenham Junction and London Blackfriars, calling at the stations described in paragraph 1.1. plus Loughborough Junction.

#### 3 SERVICE PATTERN SATURDAYS

3.1 There are no trains specified on the route on Saturdays

#### 4 SERVICE PATTERN SUNDAYS

4.1 There are no trains specified on the route on Sundays

#### Route L: London Blackfriars to Sevenoaks via Catford and Otford

#### 1. ROUTE DEFINITION

1.1. Services shall be provided between London Blackfriars and Sevenoaks calling at Elephant & Castle, Denmark Hill, Peckham Rye, Nunhead, Crofton Park, Catford, Bellingham, Beckenham Hill, Ravensbourne, Shortlands, Bromley South, Bickley, St. Mary Cray, Swanley, Eynsford, Shoreham, Otford and Bat & Ball.

#### 2. Service Pattern Mondays to Fridays

#### **Towards Sevenoaks**

2.1 Between and including the Early and Late Services, services shall be provided from London Blackfriars to Sevenoaks at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The interval between the Early Service and second service may be extended to 45 minutes, provided an additional service is provided from Bromley South to Sevenoaks between 0600 and 0630, calling at the stations specified in Paragraph 1.1.

Between 07:30 and 09:00, five additional services shall be provided from London Blackfriars with a maximum interval of 30 minutes, calling at the stations specified in Paragraph 1.1.

- (a) One service need not call at Peckham Rye, Nunhead, Crofton Park, Bellingham, Beckenham Hill, Ravensbourne and Shortlands.
- (b) One service may terminate at Orpington, calling at Petts Wood.
- (c) One service may terminate at Beckenham Junction, calling at Elephant & Castle only.
- (d) Two services may terminate at Bromley South. One service need not call at Crofton Park, Beckenham Hill, Ravensbourne and Shortlands. One service may call at Bromley South only.
- 2.2 During the Evening Peak, the following additional service shall be provided from **London Blackfriars**, calling at the stations specified in Paragraph 1.1:
  - (a) between 1615-1645, one service to **Orpington.** After Bickley, this service shall call at Petts Wood.
- 2.3 During the Evening Peak, one service to Sevenoaks need not call at Bickley and St Mary Cray.

#### **Towards London**

- 2.4 Between the Early Service and 0600 services arriving at London Blackfriars need not call at Sevenoaks, Bat & Ball, Otford, Shoreham Eynsford, St Mary Cray and Swanley but alternatively may start from Orpington, calling at Petts Wood.
- 2.5 Between and including the Early and Late Services, services shall be provided from Sevenoaks to London Blackfriars at half-hourly intervals, calling at the stations specified in Paragraph 1.1, except:

- (a) during the Morning Peak, services need not stop at certain stations, as follows:
  - (i) Ravensbourne, Beckenham Hill 2 services
  - (ii) Crofton Park 2 services
  - (iii) Nunhead 2 services
  - (iv) Peckham Rye 1 service
- (b) One interval between services in the Morning Peak may be extended to 40 minutes.

The interval between the Early Service and 06:30 may be extended to one hour.

- 2.6 During the Morning Peak, the following additional services shall be provided to London Blackfriars, calling at the stations specified in Paragraph 1.1:
  - (a) between 0730-0900, two services from Orpington at a minimum interval of 45 minutes, calling additionally at Petts Wood. One of these services need not call at Ravensbourne and Beckenham Hill.
  - (b) between 0800-0900, two services from Bromley South. One of these services need not call at Crofton Park.

Between 16:15 and 17:45, two additional services shall be provided, calling at the stations specified in Paragraph 1.1. One service need not call at Shoreham, Eynsford, Bickley, Shortlands, Ravensbourne, Beckenham Hill and Crofton Park. One service may start from Orpington, calling at Petts Wood.

Between 19:30 and 20:00, one additional service shall be provided from Sevenoaks to Bromley South, calling at the stations specified in Paragraph 1.1.

#### 3 SERVICE PATTERN SATURDAYS

#### **Towards Sevenoaks**

3.1 Between and including the Early and Late Services, services shall be provided from London Blackfriars to Sevenoaks at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### **Towards London**

3.2 Between and including the Early and Late Services, services shall be provided from Sevenoaks to London Blackfriars at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### 4 SERVICE PATTERN SUNDAYS

#### **Towards Sevenoaks**

4.1 Between and including the Early and Late Services, services shall be provided from London Blackfriars to Sevenoaks at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### **Towards London**

4.2 Between and including the Early and Late Services, services shall be provided from Sevenoaks to London Blackfriars at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

4.3 The Early Service and the second service shall operate from Orpington to London Blackfriars, calling at Petts Wood, Bickley then calling at stations specified in Paragraph 1.1.

#### 5 EARLY AND LATE SERVICES

	Early Service arrives at destination no later than:		
Route	Monday - Saturday Sunday		
London Blackfriars to Sevenoaks	0630	0720	0850
Sevenoaks to London Blackfriars	0520*	0645	0735*

<sup>\*</sup>May originate at Orpington

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Blackfriars to Sevenoaks	2330	2330	2330
Sevenoaks to London Blackfriars	2230	2230	2230

#### 5 MAXIMUM JOURNEY TIMES

Route	Monday - Friday	Saturday	Sunday
London Blackfriars to Sevenoaks	1 hour and 5 minutes* 1 service may take up to 1 hour 7 minutes	1 hour and 5 minutes	1 hour and 5 minutes
Sevenoaks to London Blackfriars	1 hour and 5 minutes*	1 hour and 5 minutes	1 hour and 5 minutes

<sup>\*</sup> Services in the Morning and Evening Peaks may take up to 1 hour and 15 minutes.

## SLC3

## TSGN FRANCHISE (FORMER SOUTH CENTRAL SERVICES)

# SERVICE LEVEL COMMITMENT SLC3

Version 1

22nd May 2014

#### **Part 1 - Service Level Commitment – General Provisions**

#### 1. CONSTRUCTION

The following provisions of this Part 1 to this SLC3 shall apply in respect of the Service Level Commitment set out in Part 2 to this SLC3.

#### 2. DAYS AND TIMES OF DAY

- 2.1 Except to the extent the context otherwise requires, references to a day mean the period commencing at 0200 on one day and ending at 0159 on the following day and references to Weekdays and particular days of the week shall be construed accordingly.
- 2.2 References to periods of times and periods of days include the times and days such periods start and finish.
- 2.3 All references to time are to the twenty-four hour clock.

#### 3. SERVICES

- 3.1 Except where expressly indicated to the contrary, references to services, all services or any part or any proportion of services are to be construed as references to the Passenger Services (or the relevant part or proportion thereof) required to be included by the Franchisee in its Timetable pursuant to Schedule 1.1 of the Franchise Agreement and do not include such Additional Passenger Services as the Franchisee may be permitted to provide from time to time under this Franchise Agreement.
- 3.2 Except where expressly indicated to the contrary, where an interval or frequency is specified for a service, such specification shall apply at the departure point for the relevant service.
- 3.3 Except where expressly indicated to the contrary, all services are to run in both directions and the requirements of the Service Level Commitment (including any interval between services, frequency of service or stopping pattern) are to apply in each direction.
- 3.4 Except where expressly indicated to the contrary, references to "Summer" shall mean that period in any year from the last Saturday in May until the third Saturday in September and "Winter" is defined as any period other than "Summer."

#### 4. STATIONS

- 4.1 Except where expressly indicated to the contrary and subject always to compliance with the other provisions of the Service Level Commitment (including any maximum Journey Times) nothing in the Service Level Commitment shall prevent services which are required to be included by, or on behalf of, the Franchisee in the Timetable pursuant to Schedule 1.1 of the Franchise Agreement calling at any stations which are not specified in the Service Level Commitment or any relevant part of it.
- 4.2 The 'core' stations as specified in the Route Definition, when applied to a service defined in the Service Pattern which does not run over the full length of the Route, shall apply only to stations which are on the shorter journey being specified.
- 4.3 Where services on Routes A to G are required or permitted to depart from or arrive at London Bridge, such services shall not be required to call at London Victoria and Clapham Junction, but shall otherwise be required to call at the same stations as the London Victoria services. Maximum Journey Times shall apply to such services.

#### 5. CONNECTIONS

- 5.1 Except where expressly indicated to the contrary, a service which is required to be included in the Timetable shall be provided so as to enable travel between the stations specified without passengers being required to change train. Where a service may be provided by a Connection or where a Connection is required to be provided between two services, the two relevant services must be provided without a further change of train being required.
- 5.2 Except where expressly indicated to the contrary, where Connections are required to be provided by the Franchisee, the Franchisee shall ensure that the interval or waiting period between the two relevant services is of a sufficient duration to allow passengers a reasonable period of time to transfer between such services. For certain locations guidance on these times can be found in the Network Rail Rules of the Plan or in the station index of the National Rail timetable.
- 5.3 Except where expressly indicated to the contrary, where services are subject to maximum Journey Times, such Journey Times shall not apply where the services are required to be, or may be, provided by Connections.

#### 6. PEAK AND OFF-PEAK

- 6.1 Peak means, in relation to any service a Weekday Service (except Christmas and Bank Holidays) which arrives at London Victoria, London Bridge or London Blackfriars, between 0700 and 0959 (the "Morning" Peak) and departs from London Victoria, London Bridge or London Blackfriars, between 1600 and 1859 (the "Evening" Peak).
- 6.2. Off-Peak means in relation to any service any period which is not Peak.

#### 7. NOT USED

#### 8. SERVICE INTERVALS

- 8.1 Except where expressly indicated to the contrary, where the Franchisee is required to include within the Timetable, services for a specified period (which period shall commence and end with an Early Service or Late Service), with a specified interval between each such service during such period, then the following shall apply:
  - (a) the minimum number of services to be included in the Timetable in such period shall be determined in accordance with the following formula:

$$S = \frac{TM}{IM}$$

where:

S is the minimum number of services to be included in the Timetable:

TM is the total minutes in the specified period; and

IM is the number of minutes in the specified interval, rounded down to the nearest whole number;

- (b) the interval between any two services during such period may be extended by:
  - (i) five minutes; or,
  - (ii) if greater, an amount of minutes equivalent to one-sixth of the specified interval between each service rounded down to the nearest whole number, subject to the other provisions of this Paragraph 8.1 and such extension not exceeding ten minutes;
- (c) the interval between any two services may be reduced, subject to the other provisions of this Paragraph 8.1, the interval between each service required in accordance with Paragraph 8.1(a) for such period, by such amount as the Franchisee may determine;
- (d) the interval between services during such specified period shall be such that, for any period which occurs during such specified period and has a duration equivalent to:

$$((4 \times SI) + E)$$
 minutes

where:

- SI is equal to the specified interval between such services; and
- E is equal to the extension permitted to such interval under Paragraph 8.1(b), at least four services shall be included in the Timetable in such period; and

- (e) the intervals between the start of such period of time and the first service in such period (save where such period of time starts with an Early Service) and between the last service in such period and the end of such period (save where such period of time ends with a Late Service) shall each be no more than the interval between each service specified for such period and (save where such period of time starts and/or ends with an Early Service/Late Service) one of such intervals shall be no more than half such interval between each service.
- 8.2 Except where expressly indicated to the contrary, where one period ends and another period begins immediately thereafter and services are specified at different intervals for each such period, the maximum interval between the last service in the first period and the first service in the second period shall be the longer of the maximum intervals allowed for each such period.

#### 9. SERVICE PORTION COMBINATIONS

9.1 Except where expressly indicated to the contrary, where the Franchisee is required to include within the Timetable, services may be formed by combining no more than two service requirements for a proportion of the journey, except to the extent the Authority otherwise agrees.

#### 10. Part 2 Service Level Commitment

#### Section 1 – Gatwick Airport and Brighton Services

**Route A:** London Victoria – Gatwick Airport (Express Services)

**Route B1: London Victoria - Brighton (Fast Services)** 

**Route B2:** London Victoria - Brighton (Semi-fast Services)

**Route B3:** London Victoria - Brighton (Stopping Services)

**Route B4: London Bridge - Brighton** 

#### **Section 2 – Sussex Coast Services**

Route C1: London Victoria – Littlehampton via Hove

Route C2: London Victoria – Littlehampton via Hove (Stopping Services)

Route D1: London Victoria – Southampton Central via Horsham

Route D2: London Victoria – Portsmouth Harbour via Horsham

Route D3: London Victoria – Bognor Regis via Horsham

**Route E1: London Victoria – Hastings via Lewes** 

Route E2: London Victoria – Eastbourne via Lewes

#### Section 3 – Surrey & Sussex Borders

**Route F1:** London Bridge – Horsham via Redhill (Semi-Fast Services)

Route F2: London Bridge – Reigate via Redhill

Route F3: London Bridge – Tonbridge via Redhill

Route G1: London Victoria – East Grinstead

**Route G2:** London Bridge – Uckfield

#### **Section 4 – East Coastway Services**

**Route H1: Brighton - Lewes** 

Route H2: Brighton - Seaford

**Route H3: Brighton - Hastings** 

Route H4: Brighton - Ashford

#### Section 5 – West Coastway Services

- **Route J1:** Brighton Hove
- **Route J2:** Brighton West Worthing
- **Route J3:** Brighton Portsmouth Harbour
- **Route J4:** Brighton Southampton Central
- **Route J5:** Littlehampton Bognor Regis
- **Route J6:** Littlehampton Portsmouth & Southsea

#### **Section 6 – South London Lines**

- Route K: London Victoria Horsham via Hackbridge, Epsom and Dorking (semi-fast)
- Route L: London Victoria Epsom via Hackbridge
- Route M: London Victoria Epsom via Selhurst, West Croydon and Sutton
- Route N: London Victoria Epsom Downs via Selhurst, West Croydon and Sutton
- Route O: London Victoria Sutton via Selhurst and West Croydon
- Route P: London Victoria Caterham via Selhurst and East Croydon
- Route Q: London Victoria –Sutton via Crystal Palace and West Croydon
- Route R: London Victoria London Bridge via Crystal Palace
- **Route S: NOT USED**
- Route T: London Bridge West Croydon via Peckham Rye, Tulse Hill and Selhurst
- Route U: London Bridge Beckenham Junction via Tulse Hill and Crystal Palace
- Route V1 London Bridge Norwood Junction via Forest Hill (Stopping Services)
- Route V2 London Bridge Epsom via Forest Hill and West Croydon
- Route W: London Bridge Sutton and Wimbledon via Tulse Hill and Hackbridge
- Route X: London Bridge Caterham via East Croydon
- Route Y: London Bridge Tattenham Corner via East Croydon (semi-fast)

#### **Section 7 –West London Line**

Route Z: East Croydon – Clapham Junction - Watford Junction - Milton Keynes Central via Selhurst, Kensington Olympia and Harrow & Wealdstone

#### **Section 1 – Gatwick Airport and Brighton Services**

#### **Route A** LONDON VICTORIA – GATWICK AIRPORT (Express Services)

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Gatwick Airport, calling at no intermediate stations.
- 1.2 There are no Limited Stops on this route.

#### 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

- 2.1 <sup>i</sup>Between and including the Early and Late Services, four services per hour shall be provided at quarter-hourly intervals between London Victoria and Gatwick Airport.
- 2.2 Services departing before 0545 and after 2315 are permitted to extend the Maximum Journey Time by 7 minutes.

#### Additional Services and Variations

#### Towards Gatwick Airport

- 2.3 Two additional services shall be provided from London Victoria to Gatwick Airport at an interval of sixty minutes, the second service departing thirty minutes earlier than the Early Service.
- 2.4 One additional service shall be provided from London Victoria to Gatwick Airport departing thirty minutes later than the Late Service. This service is permitted to extend the Maximum Journey Time by 20 minutes.

#### Towards London Victoria

- 2.5 Two additional services shall be provided from Gatwick Airport to London Victoria at an interval of forty-five minutes, the second service departing thirty minutes earlier than the Early Service. One of these services is permitted to extend the Maximum Journey Time by 7 minutes.
- 2.6 One additional morning Peak service shall be provided from Gatwick Airport to London Victoria.
- 2.7 One additional service shall be provided from Gatwick Airport to London Victoria departing forty-five minutes later than the Late Service. This service is permitted to extend the Maximum Journey Time by 20 minutes.

#### 3. Service Pattern – Saturdays

#### Core Service Specification

- 3.1 Between and including the Early and Late Services, four services per hour shall be provided at quarter-hourly intervals between London Victoria and Gatwick Airport.
- 3.2 Services departing before 0545 and after 2315 are permitted to extend the Maximum Journey Time by 6 minutes, with the exception of one service which is permitted to extend the Maximum Journey Time by 7 minutes

#### **Additional Services and Variations**

#### Towards Gatwick Airport

- 3.3 Two additional services shall be provided from London Victoria to Gatwick Airport at an interval of sixty minutes, the second service departing thirty minutes earlier than the Early Service.
- 3.4 One additional service shall be provided from London Victoria to Gatwick Airport departing thirty minutes later than the Late Service. This service is permitted to extend the Maximum Journey Time by 20 minutes.

#### Towards London Victoria

- 3.5 Two additional services shall be provided from Gatwick Airport to London Victoria at an interval of forty-five minutes, the second service departing thirty minutes earlier than the Early Service.
- 3.6 One additional service shall be provided from Gatwick Airport to London Victoria departing forty-five minutes later than the Late Service. This service is permitted to extend the Maximum Journey Time by 20 minutes.

#### 4. Service Pattern – Sundays

#### Core Service Specification

4.1 Between and including the Early and Late Services, four services per hour shall be provided at quarter-hourly intervals between London Victoria and Gatwick Airport.

#### Additional Services and Variations

#### Towards Gatwick Airport

- 4.2 Two additional services shall be provided from London Victoria to Gatwick Airport at an interval of sixty minutes, the second service departing thirty minutes earlier than the Early Service.
- 4.3 One additional service shall be provided from London Victoria to Gatwick Airport departing thirty minutes later than the Late Service. This service is

permitted to extend the Maximum Journey Time by 20 minutes.

#### Towards London Victoria

- 4.4 Two additional services shall be provided from Gatwick Airport to London Victoria at an interval of forty-five minutes, the second service departing thirty minutes earlier than the Early Service.
- 4.5 One additional service shall be provided from Gatwick Airport to London Victoria departing forty-five minutes later than the Late Service. This service is permitted to extend the Maximum Journey Time by 20 minutes.

#### 5. Limited Stops

There are no Limited Stops on this route

#### 6. Early and Late Services

	Early Service arrives at destination no later than:				
Route	Monday - Friday Saturday Sunday				
London Victoria to Gatwick Airport	0545	0545	0545		
Gatwick Airport to London Victoria	0630	0630	0630		

	Late Service departs no earlier than:				
Route	Monday - Friday Saturday Sunday				
London Victoria to Gatwick Airport	0000	0000	0000		
Gatwick Airport to London Victoria	0045	0045	0045		

## 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Gatwick Airport	30 minutes with five exceptions of 31 minutes and one exception of 33 minutes permitted	30 minutes with one exception of 32 minutes permitted	35 minutes with eighteen exceptions of 38 minutes permitted
Gatwick Airport to London Victoria	30 minutes with seven exceptions of 32 minutes permitted	30 minutes	37 minutes with two exceptions of 38 minutes and a further exception of 40 minutes permitted

#### **Route B1 LONDON VICTORIA - BRIGHTON (Fast Services)**

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Brighton, calling at Clapham Junction and East Croydon.
- 1.2 There is no requirement for Limited Stops on this route.

#### 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Brighton, calling at some or all of the stations specified in Paragraphs 1.1 and 1.2.

#### Additional Services and Variations

#### Towards Brighton

- 2.2 One interval as specified in Paragraph 2.1 may be extended to forty minutes.
- 2.3 Three additional evening peak services shall be provided departing at half hourly intervals between 1730 and 1900 from London Victoria to Brighton. These services may be provided by combining with a service from London Victoria to Gatwick Airport specified in Route A Paragraph 2.1, omitting to call at Clapham Junction and East Croydon. The first of these services is permitted to extend the Maximum Journey Time by 10 minutes and the second of these services is permitted to extend the Maximum Journey Time by 7 minutes.
- 2.4 Three Evening Peak services departing no earlier than 1715, as specified in Paragraph 2.1, may be provided by combining with a service from London Victoria to Gatwick Airport specified in Route A Paragraph 2.1, omitting to call at Clapham Junction and East Croydon. One interval may be extended to 40 minutes.

One additional service shall be provided from London Victoria to Brighton departing between 0600 and 0615, calling at Clapham Junction, East Croydon, Gatwick Airport and Three Bridges. This service is permitted to extend the Maximum Journey Time by 7 minutes.

One additional service shall be provided from London Victoria to Brighton departing between 0730 and 0745, calling at Clapham Junction, East Croydon, Gatwick Airport, Three Bridges, Haywards Heath and Burgess Hill. This service is permitted to extend the Maximum Journey Time by 4 minutes.

Two additional services shall be provided from London Victoria to Three Bridges departing between 1615 and 1645 at a half hourly interval, calling at Gatwick Airport. These two services as provided as part of the requirements

#### for Route A.

#### Towards London Victoria

- 2.5 Seven additional morning Peak services departing between 0615 and 0830 shall be provided by combining with services from Gatwick Airport to London Victoria specified in Route A Paragraph 2.1, omitting to call at Clapham Junction and East Croydon. These services are permitted to extend the Maximum Journey Time by up to 15 minutes
- 2.6 One additional Morning Peak service shall be provided from Brighton to London Victoria departing between 0645 and 0655. This service is permitted to extend the Maximum Journey Time by 16 minutes.

One additional service shall be provided from Brighton to Victoria departing between 1915 and 1930, calling at Gatwick Airport, East Croydon and Clapham Junction. This service is permitted to extend the Maximum Journey Time by 7 minutes.

#### 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Brighton, calling at some or all of the stations specified in Paragraphs 1.1 and 1.2.

#### Additional Services and Variations

#### Towards Brighton

3.2 There are no variations to the core service from London Victoria to Brighton specified in Paragraph 3.1.

#### Towards London Victoria

3.3 There are no variations to the core service from London Victoria to Brighton specified in Paragraph 3.1.

#### 4. Service Pattern – Sundays

#### Core Service Specification

4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Brighton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Brighton

4.2 There are no variations to the core service from London Victoria to Brighton

specified in Paragraph 4.1.

#### Towards London Victoria

4.3 There are no variations to the core service from Brighton to London Victoria specified in Paragraph 3.1.

#### 5. Limited Stops

There is no requirement for Limited Stops on this route.

#### 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Brighton	0915	0800	1000	
Brighton to London Victoria	0945	0745	1000	

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Brighton	2300	2300	2030	
Brighton to London Victoria	2215	2145	2000	

#### 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Brighton	52 minutes, with seven exceptions of up to 55 minutes, two exceptions of up to 58 minutes, four exceptions of up to 1 hour 3 minutes and two further exceptions of up to 1 hour 8 minutes permitted	51 minutes with five exceptions of 55 minutes permitted	52 minutes
Brighton to London Victoria	51 minutes with nine exceptions of up to 52 minutes and four exceptions of up to 55 minutes permitted	51 minutes	minutes with one exception of 55 minutes permitted

#### **Route B2 London Victoria - Brighton (Semi-Fast Services)**

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Brighton, calling at Clapham Junction, East Croydon, Gatwick Airport, Burgess Hill and Hassocks.
- 1.2 There are no Limited Stops on this route.

#### 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Brighton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Brighton

2.2 Between the Early and Late Service as specified in Paragraph 2.1, services from London Victoria may omit to call at Clapham Junction.

#### Towards London Victoria

2.3 There are no additions or variations to the services from Brighton to London Victoria as specified in Paragraph 2.1.

#### 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Brighton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Brighton

3.2 Between and including the Early and Late Service as specified in Paragraph 3.1, services from London Victoria may omit to call at Clapham Junction.

#### Towards London Victoria

3.3 There are no additions or variations to the services from Brighton to London Victoria as specified in Paragraph 3.1.

One additional service shall be provided from Brighton to London Victoria, departing half an hour later than the Late Service, calling at Hassocks,

Gatwick Airport and East Croydon.

#### 4. Service Pattern – Sundays

#### Core Service Specification

#### 4.1

Between and including the Early and Late services, one service per hour shall be provided from London Victoria to Brighton, calling at Clapham Junction, East Croydon and Gatwick Airport.

Between and including the Early and Late services, one service per hour shall be provided from Brighton to London Victoria, calling at Gatwick Airport, East Croydon and Clapham Junction.

#### 5. Limited Stops

There are no Limited Stops on this route

#### 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Brighton	1000	0900	1015	
Brighton to London Victoria	1030	0830	0945	

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Brighton	1545	1945	2100	
Brighton to London Victoria	1515	1815	2030	

#### 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Brighton	1 hour 1 minute	1 hour 5 minutes	N/A
Brighton to London Victoria	59 minutes	1 hour 3 minutes	N/A

#### **Route B3** London Victoria - Brighton (Stopping Services)

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Brighton, calling at Clapham Junction, East Croydon, Purley, Horley, Gatwick Airport, Three Bridges and Haywards Heath.
- 1.2 Limited Stops shall be made at Coulsdon South, Merstham, Redhill, Balcombe, Wivelsfield, Burgess Hill, Hassocks and Preston Park.

#### 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

2.1 There is no core service specification for Mondays to Fridays.

#### Additional Services and Variations

#### Towards Brighton

- 2.2 Between 0001 and 0030, one service shall be provided, calling at the stations specified in Paragraph 1.1, and at Coulsdon South, Merstham and Redhill. This service may terminate at Gatwick Airport.
- 2.3 Between 0030 and 0130, one service shall be provided, calling at the stations specified in Paragraph 1.1. This service is permitted to extend the Maximum Journey Time by 10 minutes.
- 2.4 Between 0130 and 0330, two services shall be provided at hourly intervals, calling at the stations specified in Paragraph 1.1. These services may terminate at Three Bridges.
- 2.5 Between 0330 and 0430, one service shall be provided from London Victoria, calling at the stations specified in Paragraph 1.1. This service is permitted to extend the Maximum Journey Time by 8 minutes.
- 2.6 Between 0445 and 0515, one service shall be provided from London Victoria, calling at the stations specified in Paragraph 1.1, and at all the stations specified in Paragraph 1.2. This service is permitted to extend the Maximum Journey Time by 13 minutes.
- 2.7 Between 0620 and 0635, one service shall be provided from London Victoria, calling at the stations specified in Paragraph 1.1, and at all the stations specified in Paragraph 1.2. *This service may omit to call at Balcombe*. This service is permitted to extend the Maximum Journey Time by 7 minutes.

One additional service shall be provided from London Victoria to Brighton departing between 0815 and 0830, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Three Bridges, Haywards Heath, Wivelsfields and Preston Park. This service is permitted to extend the Maximum Journey Time by 19 minutes.

2.8 Between 2050 and 2340 from London Victoria, four services shall be provided, at intervals of at least 30 minutes, but not exceeding 60 minutes, calling at the stations specified in Paragraph 1.1 and those specified in Paragraph 1.2. These services may omit to call at Purley, Coulsdon South, Merstham, Horley and Balcombe, except that the last service shall call at Balcombe. The last of these services is permitted to extend the Maximum Journey Time by 7 minutes.

#### Towards London Victoria

- 2.9 Four services shall be provided at hourly intervals from Three Bridges to London Victoria, calling at the stations specified in Paragraph 1.1, arriving at London Victoria between 0145 and 0515. The first of these services may start from Gatwick Airport.
- 2.10 One service shall be provided from Brighton to London Victoria, calling at the stations specified in Paragraph 1.1 and Redhill, Selhurst and Balham, arriving at London Victoria between 0545 and 0600. This service is permitted to extend the Maximum Journey Time by 34 minutes.
- 2.11 One service shall be provided from Brighton to London Victoria, calling at the stations specified in Paragraph 1.1, arriving half an hour later than the service specified in Paragraph 2.9. This service shall call additionally at Burgess Hill, but may omit calls at Horley and Purley.

#### 2.12 NOT USED

2.13 Three additional services shall be provided from Brighton to London Victoria, arriving at London Victoria between 1805 and 0040, calling at the stations specified in Paragraphs 1.1 and 1.2. One service may be provided from Three Bridges to London Victoria as a portion of a service requirement in Route D. All these services may omit to call at Balcombe, Merstham and Coulsdon South. Only the last service need call at Horley and Purley. Services arriving at London Victoria before 2100 may omit to call at Redhill.

#### 3 Service Pattern – Saturdays

#### Core Service Specification

3.1 There is no core service specification for Saturdays.

#### **Additional Services and Variations**

#### Towards Brighton

- 3.2 Between 0001 and 0030, one service shall be provided, calling at the stations specified in Paragraph 1.1, and at Coulsdon South, Merstham and Redhill. This service may terminate at Gatwick Airport.
- 3.3 Between 0030 and 0130, one service shall be provided, calling at the stations

- specified in Paragraph 1.1.
- 3.4 Between 0130 and 0330, two services shall be provided at hourly intervals, calling at the stations specified in Paragraph 1.1. These services may terminate at Three Bridges.
- 3.5 Between 0330 and 0430, one service shall be provided from London Victoria, calling at the stations specified in Paragraph 1.1.
- 3.6 Between 0445 and 0545, two services shall be provided from London Victoria, at half-hourly intervals, calling at the stations specified in Paragraph 1.1, and at all the stations specified in Paragraph 1.2, the first service may omit to call at Balcombe.
- 3.7 One additional service shall be provided from London Victoria departing no earlier than 2300 calling at the stations specified in Paragraph 1.1 and those specified in Paragraph 1.2. This service may omit to call at Horley, Purley, Coulsdon South and Merstham.

#### Towards London Victoria

- 3.8 Six services shall be provided at hourly intervals from Three Bridges to London Victoria, calling at the stations specified in Paragraph 1.1, arriving at London Victoria between 0145 and 0630. One of these services may start from Gatwick Airport. One service, to arrive at London Victoria before 0600, shall originate from Brighton. One service, to arrive at London Victoria before 0630, shall call at Redhill.
- 3.9 Four services shall be provided from Brighton to London Victoria, arriving at London Victoria between 0620 and 0735. These trains shall call at the stations specified in Paragraph 1.1, except that they may omit to call at Horley and Purley. Two of these services shall call at Preston Park; two shall call at Hassocks; two shall call at Burgess Hill; and two shall call at Wivelsfield.
- 3.10 Four services shall be provided, at hourly intervals, to arrive at London Victoria between 2110 and 0040, calling at the stations specified in Paragraphs 1.1 and 1.2. These services may omit to call at Balcombe, Merstham and Coulsdon South, and only the last train need call at Horley and Purley.

## 4 Service Pattern – Sundays

## Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Brighton, calling at Clapham Junction, East Croydon, Redhill, Gatwick Airport, Three Bridges, Balcombe, Haywards Heath, Wivelsfield, Burgess Hill, Hassocks and Preston Park.
- 4.2 Limited Stops shall be made at Purley, Coulsdon South, Merstham and

Horley.

#### **Additional Services and Variations**

## Towards Brighton

- 4.3 Between 0001 and 0030, one additional service shall be provided, calling at the stations specified in Paragraphs 4.1 and 4.2. This service shall terminate at Three Bridges.
- 4.4 Between 0030 and 0130, one additional service shall be provided, calling at the stations specified in Paragraphs 4.1 and 4.2. *This service may omit to call at Coulsdon South, Merstham, Redhill, Balcombe, Haywards Heath, Wivelsfield, Burgess Hill, Hassocks and Preston Park.*
- 4.5 Between 0130 and 0330, two additional services shall be provided at hourly intervals, calling at the stations specified in Paragraphs 4.1 and 4.2. *These services may terminate at Three Bridges and may omit to call at Coulsdon South, Merstham and Redhill.*
- 4.6 Between 0330 and 0530, two additional services shall be provided at hourly intervals from London Victoria, calling at the stations specified in Paragraph 4.1 and 4.2. The first of these services may omit to call at Coulsdon South, Merstham, Balcombe, Wivelsfield, Burgess Hill, Hassocks and Preston Park. The second service may omit to call at Coulsdon South, Merstham, Balcombe, Wivelsfield, Burgess Hill, Hassocks and Preston Park
- 4.7 Between 0540 and 0640, two additional services shall be provided from London Victoria, calling at the stations specified in Paragraphs 4.1 and 4.2. The first of these services may omit to call at Purley, Coulsdon South, Merstham and Balcombe.

#### Towards London Victoria

- 4.8 Six additional services shall be provided at hourly intervals from Three Bridges to London Victoria, arriving at London Victoria between 0145 and 0630, and calling at Gatwick Airport, Horley, Purley, East Croydon and Clapham Junction. One of these services may originate from Gatwick Airport. One service, to arrive at London Victoria before 0600, shall originate from Brighton, but need not make intermediate calls between Brighton and Three Bridges. The last of these six services shall call at Redhill.
- 4.9 One additional service shall be provided from Brighton to London Victoria, arriving at London Victoria between 0720 and 0735. This service shall call at the stations specified in Paragraphs 4.1 and 4.2, except that it may omit to call at Balcombe, Horley, Redhill, Merstham, Coulsdon South and Purley.
- 4.10 One additional service shall be provided from Brighton to London Victoria, departing after 2300 and arriving before 0045, calling at the stations specified in Paragraphs 4.1 and 4.2. This service may omit to call at Merstham and Coulsdon South.

# 5. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
London Victoria to Brighton	N/A	N/A	0900	
Brighton to London Victoria	N/A	N/A	0830	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Brighton	N/A	N/A	2330
Brighton to London Victoria	N/A	N/A	2200

# 6. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Brighton	1 hour 13 minutes	1 hour 20 minutes with two exceptions of 1 hour 25 minutes and one further exception of 1 hour 30 minutes permitted	1 hour 24 minutes with one exception of 1 hour 27 minutes permitted
Brighton to London Victoria	1 hour 25 minutes with one exception of 1 hour 35 minutes *	1 hour 23 minutes  with one exception  of 1 hour 36  minutes* and one further exception of 2 hours 6 minutes permitted	1 hour 19 minutes  with one exception of 1 hour 34  minutes* and a further exception of 2 hours and 4 minutes permitted

 $<sup>^{*}</sup>$  The services originating from Brighton specified in paragraphs 2.10, 3.8 and 4.8 may take up to 2 hours and 10 minutes.

#### **Route B4** London Bridge - Brighton

#### 1. Route Definition

1.1 Services shall be provided between London Bridge and Brighton, calling at East Croydon, Three Bridges, Haywards Heath, Burgess Hill, Hassocks and Preston Park.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 There is no core service specification between London Bridge and Brighton on Route B4 Mondays to Fridays.

#### Additional Services and Variations

## Towards Brighton

- 2.2 One additional service shall be provided departing London Bridge between 0700 and 0715, calling additionally at Gatwick Airport.
- 2.3 One additional service shall be provided departing London Bridge between 0930 and 0945, calling additionally at Gatwick Airport.
- 2.4 Between 1000 and 1645 fourteen services shall depart London Bridge at 30 minute intervals, calling additionally at Gatwick Airport.
- 2.5 Between 1830 and 1945 three services shall depart London Bridge at 30 minute intervals, calling additionally at Gatwick Airport.
- 2.6 One service shall be provided departing from London Bridge between **0730** and **0800**, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Gatwick Airport, but may omit to call at Hassocks and Preston Park.
- 2.7 One service shall be provided departing from London Bridge between 1745 and 1800, calling at the stations specified in Paragraph 1.1. This service shall have a Maximum Journey Time of 1 hour and 10 minutes.
- 2.8 Two services shall be provided departing from London Bridge between 2015 and 2100. These services may omit to call at Burgess Hill, Hassocks and Preston Park.

## Towards London Bridge

- 2.9 Not Used
- 2.10 One service shall be provided from Brighton to London Bridge departing between 0725 and 0735, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Wivelsfield and Balcombe, but may omit to call at Hassocks. This service shall have a Maximum Journey Time

of 1 hour and 13 minutes.

- 2.11 *One a*dditional service shall be provided departing Brighton between 0545 and 0600, calling additionally at Gatwick Airport.
- 2.12 Two additional services shall be provided departing Brighton between 0800 and 0900 at 30 minute intervals, calling additionally at Gatwick Airport.
- 2.13 Between 0915 and 1530 thirteen services shall be provided departing Brighton at 30 minute intervals, calling additionally at Gatwick Airport.
- 2.14 Three additional services shall be provided departing Brighton between 1715 and 1830 at 30 minute intervals, calling additionally at Gatwick Airport.

2.15

2.16 One service shall be provided from Brighton to London Bridge departing between 0630 and 0645, calling at the stations specified in Paragraph 1.1 with the exception of Hassocks, Burgess Hill and Three Bridges. This service may be provided as a portion of a service from another route (the portions joining at Haywards Heath).

One service shall be provided from Brighton to London Bridge departing between 21:30 and 21:45, calling at Haywards Heath, Three Bridges, Gatwick Airport and East Croydon.

## 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 There is no core service specification between London Bridge and Brighton on Saturdays.

#### Additional Services and Variations

## Towards Brighton

3.2 Between 0730 and 1900, services shall be provided at half hourly intervals from London Bridge to Brighton, calling additionally at Gatwick Airport.

#### Towards London Bridge

3.3 Between 0730 and 1800, services shall be provided at half hourly intervals from Brighton to London Bridge, calling additionally at Gatwick Airport.

## 4 Service Pattern – Sundays

## Core Service Specification

4.1 There is no core service specification between London Bridge and Brighton on Sundays.

# **5** Early and Late Services

There is no specified Early and Late Services on Route B4.

# 6 Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Bridge to Brighton	58 minutes, but 1 hour 14 minutes for the service specified in Paragraph 2.2 and 1 hour 9 minutes for the service specified in Paragraph 2.3	N/A	N/A
Brighton to London Bridge	1 hour 11 minutes, but 1 hour <i>12 minutes</i> for the service specified in Paragraph 2.5	N/A	N/A

## Section 2 – Sussex Coast Services

#### Route C1 LONDON VICTORIA – LITTLEHAMPTON via Hove

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Littlehampton, calling at Clapham Junction, East Croydon, Gatwick Airport, Haywards Heath, Hove, Shoreham-by-Sea, Lancing, Worthing, West Worthing, Durrington-on-Sea, Goring-by-Sea and Angmering.
- 1.2 Limited Stops shall be made at Horley, Three Bridges, Burgess Hill, Hassocks, Preston Park, Portslade and Southwick.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Littlehampton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Littlehampton

- 2.2 One interval after 1530 may be extended to 2 hours.
- 2.3 One additional service shall be provided from London Victoria to Littlehampton departing before 0800, calling at stations specified in Paragraph 1.1.
- 2.4 Three additional evening Peak services shall be provided from London Bridge to Littlehampton at intervals not less than 30 minutes, calling at stations specified in Paragraph 1.1. These services may omit to call at Gatwick Airport.
- 2.5 One additional service shall be provided from London Victoria to Worthing departing one hour later than the Late Service, calling at stations specified in Paragraph 1.1.
- 2.6 One additional service shall be provided from London Victoria to Worthing departing between 2359 and 0100, calling at stations specified in Paragraph 1.1. This service shall call additionally at Brighton to set down only.

#### Towards London Victoria

2.7 One additional service shall be provided from Havant to London Victoria to arrive no later than 0715, calling at stations specified in Paragraph 1.1. The service may omit to call at Littlehampton, Goring-by-Sea, Durrington-on-Sea and West Worthing. The service shall call additionally at Chichester and Barnham.

- 2.8 Two additional morning Peak services shall be provided from Littlehampton to London Bridge at an interval no less than 30 minutes, calling at stations specified in Paragraph 1.1. One of these services may omit to call at Gatwick Airport.
- 2.9 Three additional Morning Peak services shall be provided from Littlehampton to London Victoria departing between 0600 and 0800 at half-hourly intervals, calling at stations specified in Paragraph 1.1. Two of these services may omit to call at Gatwick Airport.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Littlehampton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Littlehampton

- 3.2 One additional service shall be provided from Haywards Heath to Littlehampton arriving an hour earlier than the Early Service, calling at stations specified in Paragraph 1.1.
- 3.3 One additional service shall be provided from London Victoria to Worthing departing one hour later than the Late Service, calling at stations specified in Paragraph 1.1.
- 3.4 One additional service shall be provided from London Victoria to Worthing departing between 2359 and 0100, calling at stations specified in Paragraph 1.1.

#### Towards London Victoria

3.5 One service shall be provided from Portsmouth & Southsea to London Victoria to arrive before 0700, calling at stations specified in Paragraph 1.1. The service may omit to call at Littlehampton, Angmering, Goring-by-Sea, Durrington-on-Sea, West Worthing and Lancing. The service shall call additionally at Fratton, Havant, Chichester and Barnham..

## 4. Service Pattern – Sundays

#### Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Littlehampton, calling at the stations specified in Paragraph 1.1.
- 4.2 All services, with the exception of the Early Service from Littlehampton, may omit to call at Lancing.

#### **Additional Services and Variations**

## Towards Littlehampton

4.3 Two additional services shall be provided from London Victoria to Worthing at an interval of an hour, the first service departing one hour later than the Late Service, calling at stations specified in Paragraph 1.1.

## Towards London Victoria

4.4 There are no variations to the core service from Littlehampton to London Victoria specified in Paragraph 4.1.

## 5. Limited Stops

- 5.1 **Horley** shall be served as follows:
  - (a) On Mondays to Fridays, one Evening Peak service shall call.
- 5.2 **Three Bridges** shall be served as follows:
  - (a) On Mondays to Fridays, two evening Peak service shall call, these shall originate from London Bridge.
  - (b) On Mondays to Fridays, the service specified in Paragraph 2.6 shall call.
  - (c) On Mondays to Fridays, the service specified in Paragraph 2.7 shall call.
  - (d) On Mondays to Fridays, one Morning Peak service shall call.
  - (e) On Saturdays, the service specified in Paragraph 3.4 shall call.
  - (f) On Saturdays, the Early Service from Littlehampton shall call.
  - (g) On Saturdays, the service specified in Paragraph 3.5 shall call.
  - (h) On Sundays, the last service specified in Paragraph 4.3 shall call.
- 5.3 **Wivelsfield** shall be served as follows:
  - (a) **NOT USED**
- 5.4 **Burgess Hill** shall be served as follows:
  - (a) On Mondays to Fridays, the services specified in Paragraph 2.4 shall call.
  - (b) On Mondays to Fridays, the services departing London Victoria between 1700 and 2100 shall call.
  - (c) On Mondays to Fridays, one service specified in Paragraph 2.8 shall call.
  - (d) On Mondays to Fridays, two services departing Littlehampton between 0700 and 0900 shall call.

- (e) On Mondays to Fridays, the Late Service from Littlehampton shall call.
- (f) On Sundays, the services specified in Paragraph 4.1 shall call.
- (g) On Sundays, one service specified in Paragraph 4.3 shall call.

#### 5.5 **Hassocks** shall be served as follows:

- (a) On Mondays to Fridays, the services specified in Paragraph 2.4 shall call.
- (b) On Mondays to Fridays, the services departing London Victoria between 1700 and 2100 shall call.
- (c) On Mondays to Fridays, the services specified in Paragraph 2.8 shall call.
- (d) On Mondays to Fridays, one service specified in Paragraph 2.9 shall call.

#### 5.6 **Preston Park** shall be served as follows:

- (a) On Mondays to Fridays, the services specified in Paragraph 2.4 shall call.
- (b) On Mondays to Fridays, three services departing London Victoria between 1700 and 2100 shall call.

## (c) NOT USED

(d) On Mondays to Fridays, one service specified in Paragraph 2.9 shall call.

#### 5.7 **Portslade** shall be served as follows:

- (a) On Mondays to Fridays, the services departing London Victoria between 1500 and up to and including the Late Service shall call.
- (b) On Mondays to Fridays, the services specified in Paragraphs 2.4, 2.5 and 2.6 shall call.
- (c) On Mondays to Fridays, the services specified in Paragraphs 2.7, 2.8 and 2.9 shall call.
- (d) On Mondays to Fridays, the Early Service from Littlehampton shall call.
- (e) On Saturdays, the services departing London Victoria between 1700 and up to and including the Late Service shall call.
- (f) On Saturdays, the services specified in Paragraphs 3.3 and 3.4 shall call.
- (g) On Saturdays, the Early Service from Littlehampton shall call.
- (h) On Sundays, the services specified in Paragraph 4.3 shall call.
- (i) On Sundays, the Early Service from Littlehampton shall call.

#### 5.8 **Southwick** shall be served as follows:

- (a) On Mondays to Fridays, the service specified in Paragraph 2.6 shall call.
- (b) On Saturdays, the services specified in Paragraphs 3.3 and 3.4 shall call.
- (c) On Sundays, the services specified in Paragraph 4.3 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Littlehampton	1030	0930	1015
Littlehampton to London Victoria	1030	0730	0915

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Littlehampton	2145	2145	2115
Littlehampton to London Victoria	1745	1745	2000

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Littlehampton	1 hour 42 minutes with two exceptions of 1 hour 43 minutes and four exceptions of 1 hour 49 minutes permitted	1 hour 41 minutes	1 hour 40 minutes with two exceptions of 1 hour 42 minutes and a further exception 1 hour and 49 minutes permitted
Littlehampton to London Victoria	1 hour 43 minutes with five exceptions of 1 hour 46 minutes and one further exception of 1 hour 49 minutes permitted	1 hour 43 minutes	1 hour 41 minutes with one exception of 1 hour and 50 minutes permitted

## Route C2 LONDON VICTORIA – LITTLEHAMPTON via Hove

(Stopping Servic

es)

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Littlehampton, calling at Clapham Junction, East Croydon, Gatwick Airport, Haywards Heath, Burgess Hill, Hove, Portslade, Shoreham-by-Sea, Worthing, West Worthing, Durrington-on-Sea, Goring-by-Sea and Angmering.
- 1.2 Limited Stops shall be made at Three Bridges, Balcombe, Hassocks, Preston Park, Lancing, and Southwick.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Littlehampton, calling at the stations specified in Paragraph 1.1. These services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route C1.

#### Additional Services and Variations

## Towards Littlehampton

- 2.2 Two additional services shall be provided from London Victoria to Bognor Regis departing one hour later than the Late Service at an hourly interval, calling at stations specified in Paragraph 1.1. These services shall omit to call at Littlehampton and shall call additionally at Ford, Barnham and Bognor Regis.
- 2.3 One additional service shall be provided from London Victoria to Worthing departing one hour later than the last service specified in Paragraph 2.2, calling at stations specified in Paragraph 1.1.

#### Towards London Victoria

- 2.4 Services departing from Littlehampton between 1800 and 2100 may omit to call at Burgess Hill.
- 2.5 Three additional services shall be provided at hourly intervals from Angmering to London Victoria departing no earlier than 1830, calling at stations specified in Paragraph 1.1. These services shall originate from Bognor Regis and shall call additionally at Ford and Barnham.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Littlehampton, calling at the stations specified in Paragraph 1.1. These services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route C1.

#### Additional Services and Variations

## Towards Littlehampton

- 3.2 Two additional services shall be provided from London Victoria to Bognor Regis departing one hour later than the Late Service at an hourly interval, calling at stations specified in Paragraph 1.1. These services shall omit to call at Littlehampton and shall call additionally at Ford, Barnham and Bognor Regis.
- 3.3 One additional service shall be provided from London Victoria to Chichester departing one hour later than the last service specified in Paragraph 3.1, calling at stations specified in Paragraph 1.1, this service shall omit to call at Littlehampton. The service shall call additionally at Ford, Barnham and Chichester.

#### Towards London Victoria

3.4 Three additional services shall be provided at hourly intervals from Angmering to London Victoria departing no earlier than 1830, calling at stations in Paragraph 1.1. These services shall originate from Bognor Regis and shall call additionally at Ford and Barnham. These services may omit to call at Burgess Hill.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 No Service.

#### 5. Limited Stops

- 5.1 **Three Bridges** shall be served as follows:
  - (a) On Mondays to Fridays, the services specified in Paragraph 2.2 shall call.
  - (b) On Mondays to Fridays, the service specified in Paragraph 2.3 shall call.
  - (c) On Mondays to Fridays, two services specified in Paragraph 2.5 shall call.
  - (d) On Saturdays, the services specified in Paragraph 3.2 shall call.

- (e) On Saturdays, the service specified in Paragraph 3.3 shall call.
- (f) On Saturdays, two services specified in Paragraph 3.4 shall call.

#### 5.2 **Balcombe** shall be served as follows:

(a) On Mondays to Fridays, one evening Peak service shall call.

#### 5.3 **Hassocks** shall be served as follows:

- (a) On Mondays to Fridays, two evening Peak services shall call.
- (b) On Mondays to Fridays, the Early Service from Littlehampton shall call.

#### 5.4 **Preston Park** shall be served as follows:

- (a) On Mondays to Fridays, the Early Service from Littlehampton shall call.
- (b) On Mondays to Fridays services specified in Paragraph 2.1 departing London Victoria between 0810 and 1525 and Littlehampton between 0910 and 1520 shall call
- (c) On Saturdays services specified in Paragraph 2.1 departing London Victoria between 0810 and 2025 and Littlehampton between 0610 and 1920 shall call

#### 5.5 **Southwick** shall be served as follows:

- (a) On Mondays to Fridays, one service specified in Paragraph 2.2 shall call.
- (b) On Mondays to Fridays, the service specified in Paragraph 2.3 shall call.
- (c) On Saturdays, one service specified in Paragraph 3.2 shall call.
- (d) On Saturdays, the service specified in Paragraph 3.3 shall call.

## 5.6 **Lancing** shall be served as follows:

- (a) On Mondays to Fridays, the services departing London Victoria between 1700 and up to and including the Late Service shall call.
- (b) On Mondays to Fridays, the services specified in Paragraph 2.2 shall call.
- (c) On Mondays to Fridays, the service specified in Paragraph 2.3 shall call.
- (d) On Mondays to Fridays, the Early Service from Littlehampton shall call.
- (e) On Mondays to Fridays, the services departing Littlehampton between 1800 and up to and including the Late Service shall call.
- (f) On Mondays to Fridays, the services specified in Paragraph 2.5 shall call.
- (g) On Saturdays, the services departing London Victoria between 1800 and

up to and including the Late Service shall call.

- (h) On Saturdays, the services specified in Paragraph 3.2 shall call.
- (i) On Saturdays, the service specified in Paragraph 3.3 shall call.
- (j) On Saturdays, the Early Service from Littlehampton shall call.
- (k) On Saturdays, the services departing Littlehampton between 1900 and up to and including the Late Service shall call.
- (l) On Saturdays, the services specified in Paragraph 3.4 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:				
Route	Monday - Friday Saturday Sunday				
London Victoria to Littlehampton	1000	1000	N/A		
Littlehampton to London Victoria	1000	0730	N/A		

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sun			
London Victoria to Littlehampton	2015	2015	N/A	
Littlehampton to London Victoria	2115	2115	N/A	

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Littlehampton	1 hour 41 minutes with two exceptions of 1 hour 43 minutes and a further exception of 1 hour 46 minutes permitted	1 hour 41 minutes	N/A
Littlehampton to London Victoria	1 hour 42 minutes with five exceptions of 1 hour 43 minutes, two exceptions of 1 hour 44 minutes and further two exceptions of 1 hour 45 minutes permitted	1 hour 43 minutes	N/A

## Route D1 LONDON VICTORIA - SOUTHAMPTON CENTRAL via Horsham

## 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Southampton Central, calling at Clapham Junction, East Croydon, Redhill, Gatwick Airport, Three Bridges, Crawley, Horsham, Barnham, Chichester, Southbourne, Emsworth, Havant, Cosham, Portchester, Fareham and Swanwick.
- 1.2 Limited Stops shall be made at Earlswood, Salfords, Horley, Ifield, Faygate, Littlehaven, Christ's Hospital, Billingshurst, Pulborough, Amberley, Arundel, Littlehampton, Ford, Fishbourne, Bosham, Nutbourne Warblington, Netley, Woolston and St. Denys.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Southampton Central, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Southampton Central

- 2.2 Two evening Peak services specified in Paragraph 2.1 may omit to call at Redhill and Gatwick Airport.
- 2.3 Two services departing from London Victoria after 1700 specified in Paragraph 2.1 may omit to call at Portchester.
- 2.4 One additional service shall be provided from Gatwick Airport to Southampton Central departing before 0615, calling at the stations specified in Paragraph 1.1. This service may omit to call at Southbourne and Emsworth.
- 2.5 One additional service shall be provided from Three Bridges to Chichester departing between 0700 and 0800, calling at the stations specified in Paragraph 1.1.
- 2.6 One additional service shall be provided from London Victoria to Southampton Central departing between 0645 and 0745, calling at the stations specified in Paragraph 1.1. This service may omit to call at Porchester. This service is permitted to extend the Maximum Journey Time by up to 17 minutes.

#### Towards London Victoria

2.7 One additional morning Peak service shall be provided from Southampton Central to London Victoria, calling at the stations specified in Paragraph 1.1. This service may omit to call at Redhill. This service is permitted to extend

- the Maximum Journey Time by up to 5 minutes.
- 2.8 Three additional services shall be provided from Southampton Central to London Victoria at hourly intervals, the first departing an hour later than the Late Service. These services are permitted to extend the Maximum Journey Time by up to 4 minutes. One of these services is permitted to extend the Maximum Journey Time by up to 8 minutes.
- 2.9 One additional service shall be provided from Southampton Central to Barnham at an hourly interval, departing no earlier than 2200. *This service shall be extended to terminate at Littlehampton.*
- 2.10 One additional service shall be provided from Chichester to Three Bridges departing after 2230, calling at the stations specified in Paragraph 1.1.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Southampton Central, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Southampton Central

- 3.2 The Late Service from London Victoria specified in Paragraph 3.1 may omit to call at Porchester.
- 3.3 One additional service shall be provided from Gatwick Airport to Southampton Central departing before 0615, calling at the stations specified in Paragraph 1.1.
- 3.4 One additional service shall be provided from London Victoria to Southampton Central departing between 0530 and 0630, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Purley, Coulsdon and Merstham.
- 3.5 One additional service shall be provided from London Victoria to Chichester departing after 2230, calling at the stations specified in Paragraph 1.1.

## Towards London Victoria

- 3.6 Three additional services shall be provided from Southampton Central to London Victoria at an hourly interval, the first departing an hour later than the Late Service.
- 3.7 One additional service shall be provided from Chichester to Three Bridges departing after 2230, calling at the stations specified in Paragraph 1.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 On Sundays there is no core service specified between London Victoria and Southampton Central.

## 5. Limited Stops

- 5.1 **Earlswood** and **Salfords** shall be served as follows:
  - (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
  - (b) On Saturdays, the service in Paragraph 3.4 shall call.
- 5.2 **Horley** shall be served as follows:
  - (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
  - (b) On Mondays to Fridays, one Evening Peak service from London Victoria shall call.
  - (c) On Saturdays, the service in Paragraph 3.4 shall call.
- 5.3 **Ifield** shall be served as follows:
  - (a) On Mondays to Fridays, the services in Paragraphs 2.4, 2.5 and 2.6 shall call.
  - (b) On Saturdays, the service in Paragraph 3.3 shall call.
  - (c) On Saturdays, the service in Paragraph 3.4 shall call.
  - (d) On Saturdays, the service in Paragraph 3.7 shall call.
- 5.4 **Faygate** shall be served as follows:
  - (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
- 5.5 **Littlehaven** shall be served as follows:
  - (a) On Mondays to Fridays, the services in Paragraphs 2.4, 2.5 and 2.6 shall call.
  - (b) On Mondays to Fridays, two Evening Peak services from London Victoria shall call.
  - (c) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
  - (d) On Saturdays, the service in Paragraph 3.3 shall call.
  - (e) On Saturdays, the service in Paragraph 3.4 shall call.

(f) On Saturdays, the service in Paragraph 3.7 shall call.

# 5.6 Christ's Hospital, Billingshurst, Pulborough, Amberley, Arundel and Ford shall be served as follows:

- (a) On Mondays to Fridays, the services in Paragraphs 2.4, 2.5 and 2.6 shall call.
- (b) On Mondays to Fridays, the services in Paragraph 2.8 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (d) On Saturdays, the services in Paragraphs 3.3 and 3.4 shall call.
- (e) On Saturdays, the services in Paragraph 3.5 and 3.6 shall call.

## 5.7 **Littlehampton** shall be served as follows:

- (a) On Mondays to Fridays, the service in Paragraph 2.5 shall call.
- (b) On Mondays to Fridays, the services in Paragraph 2.9 shall call.

#### 5.8 **Fishbourne** shall be served as follows:

- (a) On Mondays to Fridays, services from London Victoria departing after 1700 shall call.
- (b) On Mondays to Fridays, the Early Service from Southampton Central as specified in Paragraph 2.1 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.7 shall call.
- (d) On Saturdays, two services including the Late Service from London Victoria departing after 1800 shall call.

#### 5.9 **Bosham** shall be served as follows:

- (a) On Mondays to Fridays, services from London Victoria departing after 1600 shall call.
- (b) On Mondays to Fridays, the Early Service from Southampton Central as specified in Paragraph 2.1 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.7 shall call.
- (d) On Saturdays, services from London Victoria departing after 1800 shall call.

#### 5.10 **Nutbourne** shall be served as follows:

(a) On Mondays to Fridays, services from London Victoria departing after 1800 shall call.

- (b) On Mondays to Fridays, the service in Paragraph 2.7 shall call.
- (c) On Mondays to Fridays, the Early Service from Southampton Central as specified in Paragraph 2.1 shall call.
- (d) On Saturdays, two services from London Victoria departing after 1800 shall call.

## 5.11 **Warblington** shall be served as follows:

- (a) On Mondays to Fridays, services from London Victoria departing after 1800 shall call.
- (b) On Mondays to Fridays, the Early Service from Southampton Central as specified in Paragraph 2.1 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.7 shall call.
- (d) On Saturdays, services from London Victoria departing after 1800 shall call.

## 5.12 **Netley** shall be served as follows:

- (a) On Mondays to Fridays, one service departing London Victoria after 1500 shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.4 shall call.

#### 5.13 **Woolston** shall be served as follows:

- (a) On Mondays to Fridays, one service departing London Victoria after 1500 shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.4 shall call.
- (c) On Mondays to Fridays, one service from London Victoria departing after 1900 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.7 shall call.

## 5.14 **St. Denys** shall be served as follows:

- (a) On Mondays to Fridays, two services from London Victoria departing after 1500 shall call.
- (b) On Mondays to Fridays, the Early Service from Southampton Central as specified in Paragraph 2.1 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:				
Route	Monday - Friday Saturday Sunday				
London Victoria to Southampton Central	1115	1000	N/A		
Southampton Central to London Victoria	1100	0845	N/A		

	Late Service departs no earlier than:				
Route	Monday - Friday Saturday Sunday				
London Victoria to Southampton Central	2030	2030	N/A		
Southampton Central to London Victoria	1700	1700	N/A		

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Southampton Central	2 hours 27 minutes with four exceptions of 2 hours 29 minutes, three exceptions of 2 hours 31 minutes, and two further exceptions of 2 hours 35 minutes permitted	2 hours 27 minutes with three exceptions of 2 hours 33 minutes, and one further exception of 2 hours 59 minutes permitted	N/A
Southampton Central to London Victoria	2 hours 31 minutes with three exceptions of 2 hours 33 minutes, three exceptions of 2 hours 34 minutes and one further exceptions of 1 hour 35 minutes permitted	2 hours 31 minutes with one exception of 2 hours 32 minutes, and four further exceptions of 2 hours 39 minutes permitted	N/A

#### Route D2 LONDON VICTORIA – PORTSMOUTH HARBOUR via Horsham

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Portsmouth Harbour, calling at Clapham Junction, East Croydon, Redhill, Horley Gatwick Airport, Three Bridges, Crawley, Horsham, Barnham, Chichester, Havant Fratton and Portsmouth & Southsea.
- 1.2 Limited Stops shall be made at Christ's Hospital, Billingshurst, Pulborough, Amberley, Arundel, Ford, Fishbourne, Bosham, Nutbourne, Southbourne, Emsworth, Warblington, Bedhampton, and Hilsea.

### 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1. Services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route D1.

#### Additional Services and Variations

#### Towards Portsmouth Harbour

- 2.2 The Early Service as specified in Paragraph 2.1 may omit to call at Horley.
- 2.3 Two evening Peak services specified in Paragraph 2.1 may omit to call at Redhill, Horley and Gatwick Airport.
- 2.4 One peak service and the penultimate service from London Victoria specified in Paragraph 2.1 may terminate at Portsmouth & Southsea. These services shall be subject to the requirements of the Maximum Journey Times in Paragraph 6.
- 2.5 The Late Service and the penultimate service from London Victoria specified in Paragraph 2.1 may omit to call at Horley.
- 2.6 One additional service shall be provided from London Victoria to Portsmouth & Southsea departing after 2130, calling at the stations specified in Paragraph 1.1.
- 2.7 Three additional services shall be provided from London Victoria to Horsham at hourly intervals, the first departing no earlier than 2100, calling at the stations specified in Paragraph 1.1

## Towards London Victoria

- 2.8 Services from Portsmouth Harbour departing after 1500 specified in Paragraph 2.1 may omit to call at Horley.
- 2.9 One additional morning Peak service shall be provided from Havant to London Victoria arriving before 0735, calling at the stations specified in Paragraph 1.1. These services shall call additionally at Littlehaven and Ifield.
- 2.10 Two additional morning Peak services shall be provided from Portsmouth Harbour to London Victoria at an interval of one hour, calling at the stations specified in Paragraph 1.1. These services may omit to call at Gatwick Airport and Redhill. One service may be provided as a portion from Three Bridges as a service requirement of Route B (London Victoria to Brighton).
- 2.11 One additional service shall be provided from Portsmouth & Southsea to London Victoria departing between 1730 and 1830, calling at the stations specified in Paragraph 1.1. This service may omit to call at Horley.
- 2.12 One additional service shall be provided from Portsmouth Harbour to Three Bridges departing after 2130, calling at the stations specified in Paragraph 1.1.
- 2.13 Three additional services shall be provided from Horsham to London Victoria at hourly intervals departing between 1945 and 2245, calling at the stations specified in Paragraph 1.1. These services may omit to call at Horley

#### 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1. Services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route D1, for the purposes of this paragraph half-hourly shall mean between 25 minutes and 35 minutes.

#### Additional Services and Variations

## Towards Portsmouth Harbour

- 3.2 The Late Service from London Victoria specified in Paragraph 3.1 may omit to call at Horley.
- 3.3 Two additional services shall be provided from London Victoria to Portsmouth Harbour at an interval of one hour, the second shall arrive an hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1. These services shall call additionally at Purley, Coulsdon, Merstham, Earlswood, Salfords, Ifield and Littlehaven. One service may be provided as a portion, or connection, from Three Bridges as a service requirement of Route B (London Victoria to Brighton).
- 3.4 One additional service shall be provided from London Victoria to Portsmouth

& Southsea departing after 2130, calling at the stations specified in Paragraph 1.1. This service may omit to call at Horley.

#### Towards London Victoria

- 3.5 One additional service shall be provided from Havant to London Victoria arriving before 0800, calling at the stations specified in Paragraph 1.1. This service may omit to call at Horley.
- 3.6 One service shall be provided from Portsmouth & Southsea to London Victoria departing between 1730 and 1830, calling at the stations specified in Paragraph 1.1. This service may omit to call at Horley.
- 3.7 One additional service shall be provided from Portsmouth Harbour to Three Bridges departing after 2130, calling at the stations specified in Paragraph 1.1.

### 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, services one service per hour shall be provided at hourly intervals between Barnham and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1. Services shall be provided as a portion, from services originating at or terminating at London Victoria on Route C1 via Haywards Heath and Hove.

#### Additional Services and Variations

4.2 There are no variations to the core service between London Victoria and Portsmouth Harbour specified in Paragraph 4.1

## 5. Limited Stops

- 5.1 **Christ's Hospital** shall be served as follows:
  - (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
  - (b) On Mondays to Fridays, the service in Paragraph 2.8 shall call.
  - (c) On Mondays to Fridays, the service in Paragraph 2.11 shall call.
  - (d) On Saturdays, the services in Paragraph 3.3 shall call.
  - (e) On Saturdays, the service in Paragraph 3.4 shall call.
  - (f) On Saturdays, the service in Paragraph 3.5 shall call.
  - (g) On Saturdays, the service in Paragraph 3.7 shall call.

## 5.2 **Billingshurst and Pulborough** and **Arundel** shall be served as follows:

(a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.

- (b) On Mondays to Fridays, the service in Paragraph 2.8 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.11 shall call.
- (e) On Saturdays, the services in Paragraph 3.3 shall call.
- (f) On Saturdays, the service in Paragraph 3.4 shall call.
- (g) On Saturdays, the service in Paragraph 3.5 shall call.
- (h) On Saturdays, the service in Paragraph 3.6 shall call.
- (i) On Saturdays, the service in Paragraph 3.7 shall call.

## 5.3 **Amberley** shall be served as follows:

- (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
- (b) On Saturdays, the services in Paragraph 3.3 shall call.
- (c) On Saturdays, the service in Paragraph 3.4 shall call.
- (d) On Saturdays, the service in Paragraph 3.5 shall call.
- (e) On Saturdays, the service in Paragraph 3.6 shall call.

## 5.4 **Ford** shall be served as follows:

- (a) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.11 shall call.
- (c) On Saturdays, the services in Paragraph 3.3 shall call.
- (d) On Saturdays, the service in Paragraph 3.4 shall call.
- (e) On Saturdays, the service in Paragraph 3.5 shall call.
- (f) On Saturdays, the service in Paragraph 3.6 shall call.
- (g) On Saturdays, the service in Paragraph 3.7 shall call.

#### 5.5 **Fishbourne** and **Bosham** shall be served as follows:

- (a) On Mondays to Fridays, two evening Peak services from London Victoria shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.6 shall call.
- (c) On Mondays to Fridays, the services in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.10 shall call.

- (e) On Saturdays, the service in Paragraph 3.4 shall call.
- (f) On Saturdays, the service in Paragraph 3.6 shall call.

#### 5.6 **Nutbourne** shall be served as follows:

- (a) On Mondays to Fridays, one evening Peak services from London Victoria shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.6 shall call
- (c) On Mondays to Fridays, one of the services in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (e) On Saturdays, the service in Paragraph 3.4 shall call.
- (f) On Saturdays, the service in Paragraph 3.6 shall call.

#### 5.7 **Southbourne** shall be served as follows:

- (a) On Mondays to Fridays, services from London Victoria departing after 1600 shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.6 shall call
- (c) On Mondays to Fridays, the services in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (e) On Saturdays, the Late Service towards Portsmouth Harbour shall call.
- (f) On Saturdays, the service in Paragraph 3.4 shall call.
- (g) On Saturdays, the service in Paragraph 3.5 shall call.
- (h) On Saturdays, the service in Paragraph 3.6 shall call.
- (i) On Sundays, all services in Paragraph 4.1 shall call

## 5.8 **Emsworth** shall be served as follows:

- (a) On Mondays to Fridays, services from London Victoria departing after 1600 shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.6 shall call
- (c) On Mondays to Fridays, the services in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, the Early Service and one service no earlier than 1500 from Portsmouth Harbour as specified in Paragraph 2.1 shall call.
- (e) On Mondays to Fridays, the service in Paragraph 2.10 shall call.

- (f) On Saturdays, the Late Service towards Portsmouth Harbour shall call.
- (g) On Saturdays, the service in Paragraph 3.4 shall call.
- (h) On Saturdays, the service in Paragraph 3.5 shall call.
- (i) On Saturdays, the service in Paragraph 3.6 shall call.
- (j) On Sundays, all services in Paragraph 4.1 shall call

## 5.9 **Warblington** shall be served as follows:

- (a) On Mondays to Fridays, one evening Peak services from London Victoria shall call.
- (b) On Mondays to Fridays, the service in Paragraph 2.6 shall call
- (c) On Mondays to Fridays, one of the services in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (e) On Saturdays, the service in Paragraph 3.4 shall call.
- (f) On Saturdays, the service in Paragraph 3.6 shall call.

#### 5.10 **Bedhampton and Hilsea** shall be served as follows:

- (a) On Mondays to Fridays, two evening Peak services from London Victoria shall call.
- (b) On Mondays to Fridays, the services in Paragraph 2.9 shall call.
- (c) On Mondays to Fridays, the service in Paragraph 2.10 shall call.
- (d) On Mondays to Fridays, the service in Paragraph 2.11 shall call.
- (e) On Saturdays, the service in Paragraph 3.6 shall call.
- (f) On Saturdays, the service in Paragraph 3.7 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Victoria to Portsmouth Harbour	1015	1015	1045
Portsmouth Harbour to London Victoria	1030	0830	1000

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Portsmouth Harbour	2000	2000	2115
Portsmouth Harbour to London Victoria	1700	1700	1930

<sup>&</sup>lt;sup>1</sup> Sunday services are specified via Hove as portion of Route C1 service.

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Victoria to Portsmouth Harbour	2 hours with two exceptions of 2 hours 3 minutes and one further exception of 2 hours 13 minutes permitted	2 hours with one exception of 2 hours 4 minutes and two further exceptions of 2 hours 30 minutes permitted	2 hours 13 minutes with three exceptions of 2 hours 22 minutes permitted
Portsmouth Harbour to London Victoria	2 hours 4 minutes with four exceptions of 2 hours 6 minutes, two exceptions of 2 hours 8 minutes and one further exception of 1 hour 15 minutes permitted	2 hours 4 minutes	2 hours 12 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified via Hove as portion of Route C1 service.

#### Route D3 LONDON VICTORIA – BOGNOR REGIS via Horsham

## 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Bognor Regis, calling at Clapham Junction, East Croydon, Redhill, Gatwick Airport, Three Bridges, Crawley, Horsham, Christ's Hospital, Billingshurst, Pulborough, Amberley, Arundel, Ford, and Barnham.
- 1.2 Limited Stops shall be made at Horley, Ifield and Littlehaven.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

- 2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Bognor Regis, calling at the stations specified in Paragraph 1.1, for the purposes of this paragraph half-hourly shall mean between 25 minutes and 35 minutes.
- 2.2 Services may be provided as a portion from services originating at or terminating at London Victoria on Routes D1 or D2.

#### Additional Services and Variations

## Towards Bognor Regis

- 2.3 Four evening Peak services specified in Paragraph 2.1 may omit to call at Redhill and Gatwick Airport.
- 2.4 Services departing London Victoria between 0900 and 1330 specified in Paragraph 2.1 shall serve Christ's Hospital at an hourly interval.
- 2.5 Services departing London Victoria between 0900 and 1530 specified in Paragraph 2.1 shall serve Amberley at an hourly interval.
- 2.6 Three evening Peak services specified in Paragraph 2.1 may omit to call at Amberley.
- 2.7 One additional service shall be provided from London Victoria to Bognor Regis departing between 0530 and 0630, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Earlswood and Salfords. This service is permitted to extend the Maximum Journey Time by 10 minutes.
- 2.8 One additional evening Peak service shall be provided from London Bridge to Bognor Regis, calling at the stations specified in Paragraph 1.1. This service shall omit to call at Redhill and Gatwick Airport.

2.9 One additional service shall be provided from London Victoria to Barnham departing two hours later than the Late Service, calling at the stations specified in Paragraph 1.1. This service shall be extended to **Chichester**.

#### Towards London Victoria

- 2.10 Services departing Bognor Regis between 0900 and 1545 specified in Paragraph 2.1 shall serve Christ's Hospital at an hourly interval.
- 2.11 Services departing Bognor Regis between 0900 and 1745 specified in Paragraph 2.1 shall serve Amberley at an hourly interval.
- 2.12 Two additional morning Peak services shall be provided from Bognor Regis to London Victoria at an interval of half-hour, the last shall arrive before 0830, calling at the stations specified in Paragraph 1.1. These two services may omit to call at Gatwick Airport and Redhill and one service may omit Ford, and one service may omit Amberley. These services shall have Maximum Journey Times of 1 hour 43 minutes and 1 hour 38 minutes.
- 2.13 One additional morning Peak service shall be provided from Bognor Regis to London Bridge, calling at the stations specified in Paragraph 1.1. This service may omit to call at Ford, Amberley, Redhill and Gatwick Airport.
- 2.14 One additional morning Peak service shall be provided from Bognor Regis to London Victoria arriving between 0830 and 0930, calling at the stations specified in Paragraph 1.1. The service may omit to call at Ford, Gatwick Airport and Redhill. This service shall be provided as a portion up to and including Horsham, from a service terminating at London Victoria on Routes D1 or D2. This service shall have a Maximum Journey Time of 1 hour 44 minutes.
- 2.15 The Early Service from Bognor Regis to London Victoria may omit to call at Redhill

## 3. Service Pattern – Saturdays

#### Core Service Specification

- 3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Bognor Regis, calling at the stations specified in Paragraph 1.1, for the purposes of this paragraph half-hourly shall mean between 25 minutes and 35 minutes.
- 3.2 Services shall be provided as a portion up to and including Horsham, from services originating at or terminating at London Victoria on Routes D1 or D2.

#### Additional Services and Variations

#### Towards Bognor Regis

3.3 Services departing London Victoria between and including the Early and Late

- services specified in Paragraph 3.1 shall serve Christ's Hospital and Amberley at an hourly interval.
- 3.4 One additional service shall be provided from London Victoria to Barnham departing two hours later than the Late Service, calling at the stations specified in Paragraph 1.1. This service shall be extended to **Chichester**.

#### Towards London Victoria

3.5 Services departing Bognor Regis between the Early and Late Services specified in Paragraph 3.1 shall serve Christ's Hospital and Amberley at an hourly interval.

### 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Bognor Regis, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Bognor Regis

- 4.2 Services departing London Victoria between the Early and Late Services specified in Paragraph 3.1 shall call additionally at Purley.
- 4.3 The Early Service from Victoria may omit to call at Christ's Hospital and Amberley.

## Towards London Victoria

4.4 Services departing Bognor Regis between the Early and Late Services specified in Paragraph 3.1 shall call additionally at Purley.

## 5. Limited Stops

- 5.1 **Horley** shall be served as follows:
  - (a) On Mondays to Fridays, the service specified in Paragraph 2.7 shall call.
  - (b) On Mondays to Fridays, seven services departing London Victoria between 0900 and 1530 specified in Paragraph 2.1 shall call at an hourly interval.
  - (c) On Mondays to Fridays, two evening Peak services from London Victoria shall call.
  - (d) On Mondays to Fridays, one service specified in Paragraph 2.12 shall call.
  - (e) On Mondays to Fridays, the service specified in Paragraph 2.14 shall call.

- (f) On Mondays to Fridays, seven services departing Bognor Regis between 0800 and 1445 specified in Paragraph 2.1 shall call at an hourly interval.
- (g) On Saturdays, services departing London Victoria between 0800 and 1930 specified in Paragraph 3.1 shall call at hourly intervals.
- (h) On Saturdays, services departing Bognor Regis specified in Paragraph 3.1 shall call at hourly intervals.
- (i) On Sundays, services departing London Victoria specified in Paragraph 4.1 shall call.
- (j) On Sundays, services departing Bognor Regis specified in Paragraph 4.1 shall call.

#### 5.2 **Ifield** shall be served as follows:

- (a) On Mondays to Fridays, one evening Peak service from London Victoria shall call.
- (b) On Mondays to Fridays, the service specified in Paragraph 2.8 shall call.
- (c) On Mondays to Fridays, one service specified in Paragraph 2.12 shall call.
- (d) On Mondays to Fridays, the service specified in Paragraph 2.13 shall call.

#### 5.3 **Littlehaven** shall be served as follows:

- (a) On Mondays to Fridays, three evening Peak services from London Victoria shall call.
- (b) On Mondays to Fridays, the service specified in Paragraph 2.8 shall call.
- (c) On Mondays to Fridays, the services specified in Paragraph 2.12 shall call.
- (d) On Mondays to Fridays, the service specified in Paragraph 2.13 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Bognor Regis	1000	0930	0900
Bognor Regis to London Victoria	0945	0830	0900

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Bognor Regis	2030	2030	2200
Bognor Regis to London Victoria	1745	1745	2145

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
London Victoria to Bognor Regis	1 hour 46 minutes with three exceptions of 1 hour 47 minutes and one further exception of 1 hour 50 minutes permitted. Two services shall have a Maximum Journey Time of 1 hour 39 minutes and another two services shall have a Maximum Journey Time of 1 hour 43 minutes.	1 hour 46 minutes	1 hour 47 minutes with two exceptions of 1 hour and 51 minutes permitted
Bognor Regis to London Victoria	1 hour 48 minutes with nine exceptions of 1 hour 50 minutes and one further exceptions of 1 hour 51 minutes permitted	1 hour 48 minutes with one exception of 1 hour 54 minutes permitted and a further exception of 1 hour 49 minutes	1 hour 47 minutes with one exception of 1 hour and 51 minutes permitted

## Route E1 LONDON VICTORIA – HASTINGS via Lewes

## 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Hastings, calling at Clapham Junction, East Croydon, Gatwick Airport, Haywards Heath, Lewes, Polegate, Eastbourne, Hampden Park, Pevensey & Westham, Cooden Beach, Collington, Bexhill and St Leonards Warrior Square.
- 1.2 Limited Stops shall be made at Horley, Three Bridges, Wivelsfield, Plumpton, Cooksbridge, Glynde, Berwick, Pevensey Bay and Normans Bay.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification:

- 2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Hastings, calling at the stations specified in Paragraph 1.1.
- 2.2 Services may be extended to or from Ore.

#### Additional Services and Variations

#### **Towards Hastings**

- 2.3 One evening Peak interval may be extended to 1 hour 20 minutes.
- 2.4 One additional service shall be provided from London Victoria to Hastings departing before 0545, calling at stations specified in Paragraph 1.1. This service is permitted to extend the Maximum Journey Time by up to 17 minutes.
- 2.5 One additional evening Peak service shall be provided from London Victoria to Hastings departing no earlier than 1700, calling at stations specified in Paragraph 1.1.

## Towards London Victoria

- 2.6 One additional Morning Peak service shall be provided from Hastings to London Bridge calling at stations specified in Paragraph 1.1.
- 2.7 Two additional Morning Peak services shall be provided from Hastings to London Victoria departing between 0600 and 0800 at intervals between 15 minutes and 30 minutes, calling at stations specified in Paragraph 1.1. Two services may omit to call at Gatwick Airport. One of these services shall have a Maximum Journey Time of 2 hours 20 minutes and the other one shall have a Maximum Journey Time of 2 hours 12 minutes.

2.8 One additional service shall be provided from Hastings to London Victoria departing half an hour later than the Late Service as specified in Paragraph 2.1, calling at stations specified in Paragraph 1.1. This service is permitted to extend the Maximum Journey Time by 21 minutes.

## 3. Service Pattern – Saturdays

#### Core Service Specification:

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Hastings, calling at the stations specified in Paragraph 1.1. Services may be extended to or from Ore.

#### Additional Services and Variations

#### **Towards Hastings**

- 3.2 One additional service shall be provided from Gatwick Airport to Hastings departing before 0615, calling at stations specified in Paragraph 1.1.
- 3.3 One additional service shall be provided from Haywards Heath to Ore arriving one hour earlier than the Early Service as specified in Paragraph 3.1, calling at stations specified in Paragraph 1.1.

#### Towards London Victoria

3.4 One additional service shall be provided from Hastings to London Victoria departing no less than 45 minutes later than the Late Service as specified in Paragraph 3.1, calling at stations specified in Paragraph 1.1.

#### 4. Service Pattern – Sundays

#### Core Service Specification:

4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Hastings, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Hastings

4.2 One additional service shall be provided from London Victoria to Eastbourne an hour later than the Late Service, calling at stations specified in Paragraph 1.1 and Hampden Park.

All services except the last two as specified in Paragraph 4.1 shall be extended to terminate at Ore.

#### Towards London Victoria

4.3 Two additional services at an hourly interval shall be provided from Eastbourne to London Victoria, the second arriving an hour earlier than the Early Service as specified in Paragraph 4.1, calling at stations specified in Paragraph 1.1 and the first shall call at Hampden Park.

All services as specified in Paragraph 4.1 shall depart from Ore.

## 5. Limited Stops

- 5.1 **Horley** shall be served as follows:
  - (a) On Mondays to Fridays, one Morning Peak service shall call.
- 5.2 **Three Bridges** shall be served as follows:
  - (a) On Mondays to Fridays, services departing London Victoria before 0730 shall call.
  - (b) On Mondays to Fridays, one Evening Peak service shall call.
  - (c) On Mondays to Fridays, two Morning Peak services shall call.
  - (d) On Mondays to Fridays, the service specified in Paragraph 2.8 shall call.
  - (e) On Saturdays, the service specified in Paragraph 3.2, shall call.
- 5.3 **Wivelsfield** shall be served as follows:
  - (a) On Mondays to Fridays, services departing London Victoria before 0830 shall call.
  - (b) On Mondays to Fridays, services departing London Victoria after 1430 shall call.
  - (c) On Mondays to Fridays, the services as specified in paragraph 2.7, shall call.
  - (d) On Mondays to Fridays, the services departing Hastings between 1730 and up to and including the Late Service as specified in Paragraph 2.1, shall call.
  - (e) On Mondays to Fridays, the service specified in Paragraph 2.8 shall call.
  - (f) On Saturdays, services departing London Victoria between 1930 and up to and including the Late Service as specified in Paragraph 3.1, shall call.
  - (g) On Saturdays, three services departing Hastings between 1830 and up to and including the Late Service as specified in Paragraph 3.1, shall call.
  - (h) On Saturdays, the service specified in Paragraph 3.2, shall call.
  - (i) On Saturdays, the service specified in Paragraph 3.3, shall call.

- (j) On Saturdays, the service specified in Paragraph 3.4, shall call.
- (k) On Sundays, all services shall call.

## 5.4 **Plumpton** shall be served as follows:

- (a) On Mondays to Fridays, five services departing London Victoria between 0630 and 1230 shall call.
- (b) On Mondays to Fridays, services departing London Victoria after 1430 shall call.
- (c) On Mondays to Fridays, four Morning Peak services shall call.
- (d) On Mondays to Fridays, the Early Service as specified in paragraph 2.1 departing Hastings shall call.
- (e) On Mondays to Fridays, two services departing Hastings between 1730 and 2030 shall call.
- (f) On Mondays to Fridays, the service specified in Paragraph 2.8 shall call.
- (g) On Saturdays, four services departing London Victoria between 0730 and 1230 shall call.
- (h) On Saturdays, services departing London Victoria after 1530 shall call.
- (i) On Saturdays, the service specified in Paragraph 3.4 shall call.
- (j) On Sundays, all services departing London Victoria shall call, with the exception of three services between 1100 and 1600 that may omit to call.
- (k) On Sundays, all services departing Hastings shall call, with the exception of three services between 1100 and 1600 that may omit to call.

#### 5.5 **Cooksbridge** shall be served as follows:

- (a) On Mondays to Fridays, three services as specified in paragraph 2.1 departing London Victoria between and including the Early Service and 1000, shall call.
- (b) On Mondays to Fridays, five services as specified in paragraph 2.1 departing London Victoria between 1430 and 2100, shall call.
- (c) On Mondays to Fridays, three services as specified in paragraph 2.7, shall call.
- (d) On Mondays to Fridays, the Early Service as specified in paragraph 2.1 departing Hastings, shall call.

## 5.6 **Glynde** shall be served as follows:

- (a) On Mondays to Fridays, one service as specified in paragraph 2.1 departing London Victoria between 1700 and 1900, shall call.
- (b) On Mondays to Fridays, two services as specified in Paragraph 2.7 shall call.

#### 5.7 **Berwick** shall be served as follows:

- (a) On Mondays to Fridays, two services as specified in paragraph 2.1 departing London Victoria between 1500 and 1900, shall call.
- (b) On Mondays to Fridays, the Late Service departing London Victoria as specified in paragraph 2.1 shall call.
- (c) On Mondays to Fridays, the service as specified in Paragraph 2.6 shall call.
- (d) On Mondays to Fridays, three services as specified in Paragraph 2.7 shall call.
- (e) On Mondays to Fridays, the service as specified in Paragraph 2.8 shall call.
- (f) On Saturdays, the service specified in Paragraph 3.2, shall call.

## 5.8 **Pevensey Bay** and **Normans Bay** shall be served as follows:

- (a) On Mondays to Fridays, the Early Service departing London Victoria as specified in paragraph 2.1 shall call.
- (b) Two services as specified in Paragraph 2.1 departing London Victoria between 1230 and 1600 shall call.
- (c) The last service as specified in Paragraph 2.7 shall call.
- (d) One service as specified in Paragraph 2.1 departing Hastings between 1430 and 1700 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Hastings	0900	1000	1100	
Hastings to London Victoria	1100	0900	1030	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Hastings	2245	2245	2145
Hastings to London Victoria	2045	2045	2015

Route	Monday - Friday	Saturday	Sunday
London Victoria to Hastings	2 hours 2 minutes with five exceptions of 2 hours 5 minutes one exception of 2 hours 8 minutes and one further exception of 2 hours 10 minutes permitted	2 hours 3 minutes with one exception of 2 hours 4 minutes permitted	1 hour and 58 minutes with one exception of 2 hours and 1 minute and a further exception of 2 hours and 9 minutes permitted
Hastings to London Victoria	2 hours 1 minutes with four exceptions of 2 hours 2 minutes, three exceptions of 2 hours 3 minutes, two exceptions of 2 hours 4 minutes and two further exceptions of 2 hours 6 minutes permitted	2 hours 3 minutes with one exception of 2 hours 10 minutes permitted	2 hours 7 minutes

## Route E2 LONDON VICTORIA – EASTBOURNE via Lewes

## 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Eastbourne, calling at Clapham Junction, East Croydon, Gatwick Airport, Haywards Heath, Wivelsfield, Lewes, Polegate and Hampden Park.
- 1.2 Limited Stops shall be made at Horley, Three Bridges, Plumpton, Cooksbridge, Glynde and Berwick.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification:

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Eastbourne, calling at the stations specified in Paragraph 1.1. Services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route E1, for the purposes of this paragraph half-hourly shall mean between twenty-five minutes and thirty-five minutes.

#### **Additional Services and Variations**

#### Towards Eastbourne

- 2.2 Two evening Peak services as specified in Paragraph 2.1 shall originate from London Bridge. These services may omit to call at Gatwick Airport.
- 2.3 One additional service shall be provided from Haywards Heath to Eastbourne departing before 0715, calling at stations specified in Paragraph 1.1.
- 2.4 One additional service shall be provided from London Victoria to Eastbourne departing between 2359 and 0030, calling at stations specified in Paragraph 1.1. This service may omit to call at Wivelsfield.

#### Towards London Victoria

- 2.5 The Late Service as specified in Paragraph 2.1 from Hastings may omit to call Wivelsfield.
- One additional service shall be provided from Eastbourne to London Victoria arriving no later than 0645, calling at stations specified in Paragraph 1.1. This service may omit to call at Wivelsfield.
- 2.7 Two additional Morning Peak services shall be provided from Eastbourne to London Bridge, calling at stations specified in Paragraph 1.1. One service may omit to call at Wivelsfield.

## 3. Service Pattern – Saturdays

## Core Service Specification:

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Victoria and Eastbourne, calling at the stations specified in Paragraph 1.1. Services shall be so timed to provide a half-hourly interval in each hour with the services specified in Route E1, for the purposes of this paragraph half-hourly shall mean between twenty-five minutes and thirty-five minutes.

#### **Additional Services and Variations**

#### Towards Eastbourne

- 3.2 One service from London Victoria between 1800 and 2030 as specified in Paragraph 2.1 shall be extended to Hastings, calling additionally at Pevensey & Westham, Cooden Beach, Bexhill and St Leonards Warrior Square.
- 3.3 One additional service shall be provided from London Victoria to Eastbourne departing between 2359 and 0030, calling at stations specified in Paragraph 1.1. This service may omit to call at Wivelsfield and shall call additionally at Three Bridges.

#### Towards London Victoria

- 3.4 One additional service shall be provided from Eastbourne to Gatwick Airport arriving before 0615, calling at stations specified in Paragraph 1.1, the service may omit to call at Wivelsfield.
- 3.5 One additional service shall be provided from Eastbourne to London Victoria arriving between 0700 and 0800, calling at stations specified in Paragraph 1.1, the service may omit to call at Wivelsfield.

## 4. Service Pattern – Sundays

#### Core Service Specification:

4.1 There is no core service specification between London Victoria and Eastbourne on Sundays (services are provided in Route E1 London Victoria to Hastings).

## 5. Limited Stops

- 5.1 **Horley** shall be served as follows:
  - (a) On Mondays to Fridays, one Evening Peak service from London Bridge shall call.

## 5.2 **Three Bridges** shall be served as follows:

(a) On Mondays to Fridays, two Evening Peak services from London

Bridge shall call.

- (b) On Mondays to Fridays, one service as specified in Paragraph 2.7, shall call.
- (c) On Saturdays, the service as specified in Paragraph 3.4, shall call.

## 5.3 **Plumpton** shall be served as follows:

- (a) On Mondays to Fridays, one Evening Peak service from London Bridge shall call.
- (b) On Mondays to Fridays, all services departing Eastbourne shall call, with the exception of two services between 1130 and 1530 that may omit to call.
- (c) On Saturdays, all services departing Eastbourne shall call, with the exception of five services between 1000 and 1600 that may omit to call.

## 5.4 **Cooksbridge** shall be served as follows:

- (a) On Mondays to Fridays, one Evening Peak service from London Bridge shall call.
- (b) On Mondays to Fridays, the service as specified in Paragraph 2.3, shall call.
- (c) On Mondays to Fridays, the services as specified in Paragraph 2.7, shall call.
- (d) On Mondays to Fridays, the Early Service from Eastbourne as specified in Paragraph 2.1 shall call.
- (e) On Mondays to Fridays, services departing Eastbourne between 1430 and up to and including the Late Service as specified in Paragraph 2.1, shall call.

## 5.5 **Glynde** shall be served as follows:

- (a) On Mondays to Fridays, the service specified in Paragraph 2.3, shall call.
- (b) On Mondays to Fridays, two Evening Peak services from London Bridge shall call.
- (c) On Mondays to Fridays, services departing London Victoria between 1900 and up to and including the Late Service as specified in Paragraph 2.1, shall call.
- (d) On Mondays to Fridays, the services as specified in Paragraph 2.7, shall call.
- (e) On Saturdays, services departing London Victoria between 1900 and

up to and including the Late Service as specified in Paragraph 3.1, shall call.

#### 5.6 **Berwick** shall be served as follows:

- (a) On Mondays to Fridays, the service as specified in Paragraph 2.3, shall call.
- (b) On Mondays to Fridays, three Evening Peak services shall call; two of these shall depart from London Bridge.
- (c) On Mondays to Fridays, services departing London Victoria between 1900 and up to and including the Late Service as specified in Paragraph 2.1, shall call.
- (d) On Mondays to Fridays, the services as specified in Paragraph 2.7, shall call.
- (e) On Mondays to Fridays, one service departing Eastbourne between 0830 and 1100, shall call.
- (f) On Saturdays, services departing London Victoria between 1900 and up to and including the Late Service as specified in Paragraph 3.1, shall call.

# **6.** Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Eastbourne	1000	1000	N/A	
Eastbourne to London Victoria	1030	0830	N/A	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Eastbourne	2115	2115	N/A
Eastbourne to London Victoria	1845	1845	N/A

Route	Monday - Friday	Saturday	Sunday
London Victoria to Eastbourne	1 hour 27 minutes with six exceptions of 1 hour 33 minutes and one further exception of 1 hour 36 minutes permitted	1 hour 27 minutes with three exceptions of 1 hour 31 minutes and one further exception of 1 hour 36 minutes permitted	N/A
Eastbourne to London Victoria	1 hour 31 minutes with three exceptions of 1 hour 35 minutes and one further exception of 1 hour 41 minutes permitted	1 hour 30 minutes with one exception of 1 hour 34 minutes permitted	N/A

## Section 5 – Surrey & Sussex Borderers

## Route F1 LONDON BRIDGE – HORSHAM (Stopping Services)

#### 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Horsham, calling at Norwood Junction, East Croydon, Purley, Coulsdon South, Merstham, Redhill, Earlswood (Surrey), Salfords, Horley, Gatwick Airport, Three Bridges, Crawley, Ifield and Littlehaven.
- 1.2 Limited Stops shall be made at Faygate.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Horsham, calling at the stations specified in Paragraph 1.1.

### **Additional Services and Variations**

#### Towards Horsham

- 2.2 Services departing London Bridge between 0900 and 1430 specified in Paragraph 2.1 one service per hour may omit to call at Earlswood and Salfords provided each has a minimum hourly interval service.
- 2.3 One additional service shall be provided from Three Bridges to Horsham departing before 0730, calling at the stations specified in Paragraph 1.1.
- 2.4 NOT USED
- 2.5 NOT USED
- 2.6 Four additional evening Peak services shall be provided at half-hourly intervals from London Victoria to Three Bridges, calling at the stations specified in Paragraph 1.1. These services may omit to call at London Bridge, Norwood Junction and Purley, and shall call additionally at Clapham Junction. One interval may be extended to one hour five minutes.
- 2.7 Two additional evening Peak service shall be provided from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. This service shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction. This service shall be so timed with services as specified in Paragraph 2.5 as to provide a half-hourly evening Peak service interval from London Victoria.

- 2.8 Ten additional services shall be provided from London Victoria to Horsham at half-hourly intervals, the first departing after 1900, calling at the stations specified in Paragraph 1.1, one interval may be extended to 40 minutes. These services shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction. Four of these services may omit to call Salfords maintaining a minimum hourly interval. The last service in this pattern may also omit to call at Earlswood.
- 2.9 One additional service shall be provided from London Victoria to Gatwick Airport departing no earlier than 2359, calling at the stations specified in Paragraph 1.1. This service may omit to call at Earlswood and Salfords.

#### Towards London Bridge

- 2.10 Services departing Horsham between 0900 and 1600 specified in Paragraph 2.1 shall serve Salfords and Earlswood at an hourly interval.
- 2.11 Two additional services shall be provided from Horsham to London Victoria at a half-hourly interval, the second service arriving before the morning Peak, calling at the stations specified in Paragraph 1.1. These services shall omit to call at Norwood Junction, and shall call additionally at Clapham Junction. One service may omit to call at Salfords and Earlswood.
- 2.12 Three additional morning Peak services shall be provided from Three Bridges, at least one of these services shall be provided to London Victoria, calling at the stations specified in Paragraph 1.1. One service may start from Gatwick Airport and one may omit to call at Purley.
- 2.13 Three additional morning Peak services shall be provided from Horsham to London Victoria at hourly intervals, the last shall arrive before 0930, calling at the stations specified in Paragraph 1.1. These services shall omit to call at Norwood Junction, and shall call additionally at Clapham Junction. One service may omit to call at Purley. One service may be provided as a portion from Redhill.
- 2.14 Three additional morning Peak services shall be provided from Horsham to London Bridge, calling at the stations specified in Paragraph 1.1. These services may omit to call at Norwood Junction, two services may omit to call at Merstham and Purley, and one service may omit to call at Coulsdon South. One service may be provided as a portion from Redhill. One service may be provided as a portion from Redhill and may omit to call at Purley.
- 2.15 Two additional morning Peak services shall be provided from Horsham to London Victoria at an interval between 45 minutes and 60 minutes, the first shall arrive before 0900, calling at the stations specified in Paragraph 1.1. These services shall omit to call at Norwood Junction, and shall call additionally at Clapham Junction. One service may omit to call at Purley. One service may be provided as a portion from Redhill.

- 2.16 Three additional services shall be provided from Horsham to London Victoria, the first departing no earlier than 1715, calling at the stations specified in Paragraph 1.1. These services shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction. Seven additional services shall be provided from Horsham to London Bridge at half hourly intervals, the first departing no earlier than 1815, calling at stations specified in Paragraph 1.1. A minimum hourly service shall be provided at Salfords and Earlswood.
- 2.17 One additional service shall be provided from Horsham to London Victoria departing no earlier than 2300, calling at the stations specified in Paragraph 1.1. This service shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Horsham, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### Towards Horsham

- 3.2 Services departing London Bridge between and including the Early and Late Services specified in Paragraph 3.1, one service per hour may omit to call at Earlswood and Salfords provided each has a minimum hourly interval service.
- 3.3 Eight additional services shall be provided from London Victoria to Horsham at half-hourly intervals, the first departing after 2000, calling at the stations specified in Paragraph 1.1, one interval may be extended to 40 minutes. These services shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction. Four of these services may omit to call at Earlswood and Salfords maintaining a minimum hourly interval.
- 3.4 One additional service shall be provided from London Victoria to Gatwick Airport departing no earlier than 2359, calling at the stations specified in Paragraph 1.1. This service may omit to call at Earlswood and Salfords.

## Towards London Bridge

3.5 Services departing Horsham between and including the Early and Late Services specified in Paragraph 3.1 shall serve Earlswood and Salfords at an hourly interval.

- 3.6 Eight additional services shall be provided from Horsham to London Victoria at half-hourly intervals, the first departing after 1815, calling at the stations specified in Paragraph 1.1. These services shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction. Four of these services may omit to call Earlswood and Salfords maintaining a minimum hourly interval.
- 3.7 One additional service shall be provided from Horsham to London Victoria departing no earlier than 2300, calling at the stations specified in Paragraph 1.1. This service shall omit to call at London Bridge and Norwood Junction, and shall call additionally at Clapham Junction.

## 4. Service Pattern – Sundays

### Core Service Specification

4.1 Between and including the Early and Late Services, services shall be provided between London Bridge and Horsham at an hourly interval, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### Towards Horsham

- 4.2 Services departing London Bridge between and including the Early and Late Services specified in Paragraph 4.1 shall omit to call at Salfords.
- 4.3 One additional service shall be provided from Gatwick Airport to Horsham departing before 0700, calling at the stations specified in Paragraph 1.1.
- 4.4 One additional service shall be provided from London Victoria to Horsham departing after 2300, calling at the stations specified in Paragraph 1.1, this service shall omit to call London Bridge and Norwood Junction, and call additionally at Clapham Junction.

## Towards London Bridge

- 4.5 Services departing Horsham between and including the Early and Late Services specified in Paragraph 4.1 shall omit to call Earlswood and Salfords.
- 4.6 One additional service shall be provided from Horsham to Gatwick Airport departing before 0615, calling at the stations specified in Paragraph 1.1.
- 4.7 One additional service shall be provided from Horsham to London Victoria to arrive before 0800, calling at the stations specified in Paragraph 1.1, this service may omit to call Salfords, Earlswood and Norwood Junction.
- 4.8 *Two additional services* shall be provided from Horsham to London Victoria departing no earlier than *2200*, calling at the stations specified in Paragraph 1.1. *These services* shall omit to call at Salfords and Norwood Junction, and shall call additionally at Clapham Junction. The second one shall additionally omit to call at Earlswood.

## 5. Limited Stops

## 5.1 **Faygate** shall be served as follows:

- (a) On Mondays to Fridays, the service specified in Paragraph 2.3 shall call.
- (b) On Mondays to Fridays, one service specified in Paragraph 2.4 shall call.
- (c) On Mondays to Fridays, the Early Service from London Bridge as specified in Paragraph 2.1 shall call.
- (d) On Mondays to Fridays, one service departing from London Bridge between 1030 and 1130 shall call.
- (e) On Mondays to Fridays, two services departing from London Bridge between 1430 and 1600 shall call.
- (f) On Mondays to Fridays, four evening Peak services from London Bridge shall call.
- (g) On Mondays to Fridays, the services specified in Paragraph 2.13 shall call.
- (h) On Mondays to Fridays, two services from Horsham between 0900 and 1300 shall call.
- (i) On Mondays to Fridays, five services from Horsham between 1600 and 1900 shall call.
- (j) On Mondays to Fridays, the first service specified in Paragraph 2.7 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Bridge to Horsham	0900	0830	0900	
Horsham to London Bridge	1015	0645	0830	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Bridge to Horsham	1930	1930	2230
Horsham to London Bridge	1700	1800	2100

Route	Monday - Friday	Saturday	Sunday
London Bridge to Horsham	1 hour 15 minutes, with eleven exceptions of 1 hour 17 minutes and six further exceptions of 1 hour 18 minutes permitted	1 hour 14 minutes with one exception of 1 hour 15 minutes permitted	1 hour 11 minutes
Horsham to London Bridge	1 hour 13 minutes, with four exceptions of 1 hour 14 minutes, one exception of 1 hour 15 minutes and one further exception of 1 hour 19 minutes permitted	1 hour 13 minutes	1 hour 13 minutes with one exception of 1 hour 15 minutes permitted

#### **Route F2 LONDON BRIDGE - REIGATE**

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Reigate, calling at Norwood Junction, East Croydon, Purley, Coulsdon South, Merstham and Redhill,
- 1.2 There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

- 2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Reigate, calling at the stations specified in Paragraph 1.1.
- 2.2 Services on Route F2 (London Bridge Reigate) shall be so timed with services on Route F3 (London Bridge Tonbridge) as to provide a half-hourly service between London Bridge and Redhill. The combined off-peak core service patterns for Routes F1, F2 and F3 shall therefore provide four trains per hour between London Bridge and Redhill.

## Additional Services and Variations

#### **Towards Reigate**

- 2.3 Three additional services shall be provided from London Victoria to Reigate calling at the stations specified in Paragraph 1.1, arriving no later than 0800, at an interval of one hour. No more than two of these services may be formed by either a portion or connection from Redhill. These services may omit to call at Norwood Junction and one service may omit to call at Purley. These shall call additionally at Clapham Junction.
- 2.4 Two additional services shall be provided from Redhill to Reigate arriving no later than 0830, at an interval no less than one hour.
- 2.5 One additional service shall be provided from London Bridge to Reigate calling at the stations specified in Paragraph 1.1, arriving no later than 0915. This service may be formed by a portion from Redhill, and may omit to call at Norwood Junction, Purley and Merstham.
- 2.6 Two additional evening Peak services shall be provided from London Victoria to Reigate, calling at the stations specified in Paragraph 1.1, at an interval of one hour. These services may be formed by a portion from Redhill and may omit to call at Norwood Junction and Purley. These services shall call additionally at Clapham Junction.

- 2.7 One additional evening Peak services shall be provided from London Bridge to Reigate, calling at the stations specified in Paragraph 1.1. The service may omit to call at Purley. This service shall have a Maximum Journey Time of 39 minutes.
- 2.8 *One* additional service shall be provided from Redhill to Reigate, departing between 1845 and 2000.
- 2.9 *Five* additional services shall be provided from Redhill to Reigate, at hourly intervals, the last service shall depart no earlier than 2300.
- 2.10 One additional service shall be provided from London Bridge to Reigate, calling at the stations specified in Paragraph 1.1, departing after 1945.
- 2.11 Two additional services shall be provided from Redhill to Reigate, at hourly intervals, the second service shall depart no earlier than 2215.

## Towards London Bridge

- 2.12 One additional service shall be provided from Reigate to Redhill departing before 0700.
- 2.13 Two additional morning Peak services shall be provided from Reigate to London Bridge at an interval of 13 minutes, calling at the stations specified in Paragraph 1.1. These services may omit to call at Norwood Junction. One service may be formed by a portion from Redhill. Both services may be formed by a portion from Redhill until January 2017.
- 2.14 One additional morning Peak service shall be provided from Reigate to London Bridge between the services specified in Paragraph 2.13, calling at the stations specified in Paragraph 1.1. This service may omit to call at Merstham, Coulsdon South, Purley and Norwood Junction. This service need not operate until January 2017.
- 2.15 Two additional services shall be provided from Reigate to Redhill departing before 0815.
- 2.16 The Late Service as specified in Paragraph 2.1 may terminate at East Croydon.
- 2.17 One additional service shall be provided from Reigate to Redhill, departing between 1630 and 1730.
- 2.18 Two additional services shall be provided from Reigate to London Victoria, at an interval of no less than half an hour, calling at the stations specified in Paragraph 1.1, departing after 1730. These services may omit to call at Purley and Norwood Junction and one service may omit to call at Merstham, Coulsdon South,.
- 2.19 Eleven additional services shall be provided from Reigate to Redhill, at intervals between twenty and forty minutes, the last service shall depart no earlier than 2230.

## 3. Service Pattern – Saturdays

## Core Service Specification

- 3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Reigate, calling at the stations specified in Paragraph 1.1.
- 3.2 Services on Route F2 (London Bridge Reigate) shall be so timed with services on Route F3 (London Bridge Tonbridge) as to provide a half-hourly service between London Bridge and Redhill. The combined off-peak core service patterns for Routes F1, F2 and F3 shall therefore provide four trains per hour between London Bridge and Redhill.

#### **Additional Services and Variations**

## Towards Reigate

3.3 Two additional services shall be provided from Redhill to Reigate at an interval no less than one hour, the second service arriving one hour earlier than the Early Service.

## Towards London Bridge

3.4 One additional service shall be provided from Reigate to Redhill departing no earlier than 1900.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 On Sundays there is no core service specified between London Bridge and Reigate.

## 5. Limited Stops

There is no requirement for Limited Stops on this route.

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sund			
London Bridge to Reigate	1000	0900	N/A	
Reigate to London Bridge	1000	0800	N/A	

Route	Monday - Friday	Saturday	Sunday
London Bridge to Reigate	1600	1815	N/A
Reigate to London Bridge	1600	1800	N/A

Route	Monday - Friday	Saturday	Sunday
London Bridge to Reigate	42 minutes with two exceptions of 43 minutes permitted	41 minutes	N/A
Reigate to London Bridge	41 minutes with one further exception of 46 minutes permitted	41 minutes with one exception of 43 minutes	N/A

#### Route F3 LONDON BRIDGE - TONBRIDGE via Redhill

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Tonbridge, calling at Norwood Junction, East Croydon, Purley, Coulsdon South, Redhill, Nutfield, Godstone, Edenbridge, Penshurst and Leigh (Kent).
- 1.2 Limited Stops shall be made at Merstham.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

- 2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Tonbridge, calling at the stations specified in Paragraph 1.1.
- 2.2 Services on Route F3 (London Bridge Tonbridge) shall be so timed with services on Route F2 (London Bridge Reigate) as to provide a half-hourly service between London Bridge and Redhill. The combined off-peak core service patterns for Routes F1, F2 and F3 shall therefore provide four trains per hour between London Bridge and Redhill.

#### **Additional Services and Variations**

## Towards Tonbridge

- 2.3 Five additional services shall be provided from Redhill to Tonbridge, at intervals between twenty minutes and forty-five minutes, the first service shall arrive no later than 0615. Three of these services shall be extended to originate from London Victoria, calling at the stations specified in Paragraph 1.1. *Three* of these extended services may be provided as a portion from Redhill, and may omit to call at Norwood Junction and Purley; one of these extended services may also omit to call at Coulsdon South and Merstham..
- 2.4 Two additional services shall be provided from London Bridge to Tonbridge at an hourly interval, calling at the stations specified in Paragraph 1.1, arriving no later than 0930. No more than one of these services may be formed by a portion. These services may omit to call at Norwood Junction and Purley. These services are permitted to extend the Maximum Journey Time by 3 minutes.
- 2.5 One additional service shall be provided from Redhill to Tonbridge, calling at the stations specified in Paragraph 1.1, the service shall arrive no later than 0900.

## 2.6 NOT USED

- 2.7 Three additional Evening Peak service shall be provided from London Bridge to Tonbridge, calling at the stations specified in Paragraph 1.1. This service may be provided as a portion from Redhill and may omit to call at Norwood Junction. These services are permitted to extend the Maximum Journey Time by 10 minutes.
- 2.8 One additional services shall be provided from Redhill to Tonbridge at an interval no less than twenty minutes, calling at the stations specified in Paragraph 1.1, departing Redhill no earlier than 1800.
- 2.9 One additional evening Peak service shall be provided from London Victoria to Tonbridge, calling at the stations specified in Paragraph 1.1. This service may be provided as a portion from Redhill and may omit to call at Norwood Junction, and shall call additionally at Clapham Junction.
- 2.10 Four additional services shall be provided from Redhill to Tonbridge, at hourly intervals, calling at the stations specified in Paragraph 1.1. The fourth service shall depart no earlier than 2245.
- 2.11 One additional service shall be provided from Redhill to Tonbridge departing no earlier than 2345, this service shall call at Edenbridge only.

## Towards London Bridge

- 2.12 Two additional services shall be provided at an interval no less than twenty minutes from Tonbridge to Redhill, calling at the stations specified in Paragraph 1.1, arriving no later than 0600.
- 2.13 Two additional morning Peak services shall be provided at an interval no less than twenty-five minutes from Tonbridge to London Bridge, calling at the stations specified in Paragraph 1.1. These services may omit to call at Norwood Junction. These services are permitted to extend the Maximum Journey Time by 1 and 11 minutes respectively.
- 2.14 One additional service shall be provided from Tonbridge to Redhill, calling at the stations specified in Paragraph 1.1, arriving no later than 0730.
- 2.15 Three additional morning Peak services shall be provided from Tonbridge to London Victoria, calling at the stations specified in Paragraph 1.1. These services may omit to call at Purley and Norwood Junction and shall call additionally at Clapham Junction.
- 2.16 The penultimate service and the Late Service as specified in paragraph 2.1 may terminate at East Croydon.
- 2.17 Three additional services shall be provided from Tonbridge to Redhill departing *between 1630 and 1900 at hourly intervals*, calling at the stations specified in Paragraph 1.1. Two of these services shall be extended to London Victoria, and may omit to call at Purley and Norwood Junction and shall call additionally at Clapham Junction.

- One additional service hall be provided from Tonbridge to London Bridge, calling at the stations specified in Paragraph 1.1. This service shall depart no earlier than 1815. This train may omit to call Norwood Junction.
- 2.19 Five additional services shall be provided from Tonbridge to Redhill, calling at the stations specified in Paragraph 1.1, at intervals not less than 45 minutes, the last service shall depart no earlier than 2315.

## 3. Service Pattern – Saturdays

#### Core Service Specification

- 3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Tonbridge, calling at the stations specified in Paragraph 1.1.
- 3.2 Services on Route F3 (London Bridge Tonbridge) shall be so timed with services on Route F2 (London Bridge Reigate) as to provide a half-hourly service between London Bridge and Redhill. The combined off-peak core service patterns for Routes F1, F2 and F3 shall therefore provide four trains per hour between London Bridge and Redhill.

#### Additional Services and Variations

## Towards Tonbridge

- 3.3 Two additional services shall be provided from Redhill to Tonbridge at an hourly interval, the first arriving before 0700, calling at the stations specified in Paragraph 1.1.
- 3.4 Four additional services shall be provided from Redhill to Tonbridge at hourly intervals, calling at the stations specified in Paragraph 1.1, the last of these services departs no earlier than 2245.
- 3.5 One additional service shall be provided from Redhill to Tonbridge departing no earlier than 2345, this service shall call at Edenbridge only.

#### Towards London Bridge

- 3.6 One additional service shall be provided from Tonbridge to Redhill arriving before 0600, calling at the stations specified in Paragraph 1.1.
- 3.7 Five additional services shall be provided from Tonbridge to Redhill at hourly intervals, calling at the stations specified in Paragraph 1.1, the last of these services departs no earlier than 2300.

#### 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between Redhill and Tonbridge, calling at the

stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Tonbridge

4.2 There are no variations to the core service from Redhill to Tonbridge specified in Paragraph 4.1.

## Towards Redhill

4.3 There are no variations to the core service from Tonbridge to Redhill specified in Paragraph 4.1.

## 5. Limited Stops

#### 5.1 **Merstham** shall be served as follows:

- (a) On Mondays to Fridays, one service specified in Paragraph 2.4 shall call.
- (b) On Mondays to Fridays, the service specified in Paragraph 2.7 shall call.
- (c) On Mondays to Fridays, the service specified in Paragraph 2.9 shall call.
- (d) On Mondays to Fridays, these services specified in Paragraph 2.15 shall call.
- (e) On Mondays to Fridays, the services specified in Paragraph 2.17 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Tonbridge	1000	0900	0845
Tonbridge to London Bridge	1030	0730	0800

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Tonbridge	1545	1845	2300
Tonbridge to London Bridge	1615	1815	2215

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Redhill and Tonbridge

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Tonbridge	1 hour 3 minutes with one exception of 1 hour 4 minutes and one exception of 1 hour 4 minutes permitted	1 hour 3 minutes	30 minutes
Tonbridge to London Bridge	1 hour 6 minutes	1 hour 6 minutes	31 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Redhill and Tonbridge

## Route G1 LONDON VICTORIA – EAST GRINSTEAD

## 1. Route Definition

- 1.1 Services shall be provided between London Victoria and East Grinstead, calling at Clapham Junction, East Croydon, Sanderstead, Riddlesdown, Upper Warlingham, Woldingham, Oxted, Hurst Green, Lingfield and Dormans.
- 1.2 Limited Stops shall be made at Norwood Junction and South Croydon.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and East Grinstead, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

#### Towards East Grinstead

- 2.2 One additional service shall be provided from London Victoria to East Grinstead departing before 0530, calling at the stations specified in Paragraph 1.1, this service may omit to call at Sanderstead, Riddlesdown, Upper Warlingham and Woldingham.
- 2.3 Five additional services shall be provided from London Victoria to East Grinstead, at intervals of between 20 and 40 minutes, the first departing before 0600, calling at the stations specified in Paragraph 1.1. Four of these services may omit to call at Riddlesdown, Upper Warlingham and Woldingham. One of the services specified in paragraph 2.4 may count towards the interval required above.
- 2.4 *Three* additional services shall be provided from London Bridge to East Grinstead at a half-hourly interval, the first arriving before 0815, calling at the stations specified in Paragraph 1.1. *One of these services may omit to call at Riddlesdown, Upper Warlingham and Woldingham.*
- 2.5 Four additional evening Peak services shall be provided from London Bridge to East Grinstead at half-hourly intervals, calling at the stations specified in Paragraph 1.1.
- 2.6 One additional service shall be provided from London Victoria to Oxted departing after 2345, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from East Grinstead to Victoria departing between 2245 and 2300, calling at Oxted, East Croydon, Selhurst, and Clapham Junction. This services shall have a Maximum Journey Time of 45 minutes.

#### Towards London Victoria

- 2.7 One additional service shall be provided from Hurst Green to London Bridge arriving before 0645, calling at the stations specified in Paragraph 1.1.
- 2.8 One additional service shall be provided from East Grinstead to London Victoria arriving before 0659, calling at the stations specified in Paragraph 1.1. This service may omit to call at Woldingham, Upper Warlingham, Riddlesdown, and Sanderstead.
- 2.9 Six additional morning Peak services shall be provided from East Grinstead to London Bridge at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The last of these services shall arrive no earlier than 0900.
- 2.10 Four additional services shall be provided from East Grinstead to London Bridge at half-hourly intervals, departing no earlier than 1800, calling at the stations specified in Paragraph 1.1. Three of these services may terminate at East Croydon.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and East Grinstead, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards East Grinstead

- 3.2 One additional service shall be provided from London Victoria to East Grinstead departing before 0530, calling at the stations specified in Paragraph 1.1, this service may omit to call at Sanderstead, Riddlesdown, Upper Warlingham and Woldingham.
- 3.3 One additional service shall be provided from London Victoria to Oxted departing after 2345, calling at the stations specified in Paragraph 1.1.

## Towards London Victoria

3.4 There are no variations to the core service from East Grinstead to London Victoria specified in Paragraph 3.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, services shall be provided between London Victoria and East Grinstead at an hourly interval, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards East Grinstead

4.2

Between 0900 and 1930, additional services shall be provided at hourly intervals between London Victoria and East Grinstead, calling at the stations specified in Paragraph 1.1. These services shall be scheduled half an hour apart from services specified in Paragraph 4.1 to provide half hourly services between London Victoria and East Grinstead.

#### Towards London Victoria

4.3

Between 0930 and 2000, additional services shall be provided at hourly intervals between East Grinstead and London Victoria, calling at the stations specified in Paragraph 1.1. These services shall be scheduled half an hour apart from services specified in Paragraph 4.1 to provide half hourly services between East Grinstead and London Victoria.

## 5. Limited Stops

- 5.1 **Norwood Junction** shall be served as follows:
  - (a) On Mondays to Fridays, the service specified in Paragraph 2.2 shall call.
  - (b) On Mondays to Fridays, *two of the three* services specified in Paragraph 2.4 shall call.
  - (c) On Saturdays, the service specified in Paragraph 3.2 shall call.
- 5.2 **South Croydon** shall be served as follows: (The requirement to serve South Croydon may be transferred to Service Group G2 if Group G1 trains are formed of 12 cars)
  - (a) On Mondays to Fridays, four services towards East Grinstead specified in Paragraph 2.1 between 1430 and 1730 shall call.
  - (b) On Mondays to Fridays, the service specified in Paragraph 2.2 shall call.
  - (c) On Mondays to Fridays, the services specified in Paragraph2.5 shall call.
  - (d) On Mondays to Fridays, one morning Peak service specified in Paragraph 2.1 shall call.
  - (e) On Mondays to Fridays, the last service specified in Paragraph 2.9 shall call.
  - (f) On Saturdays, the service specified in Paragraph 3.2 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to East Grinstead	1000	0730	0845
East Grinstead to London Victoria	0745	0745	0915

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sund			
London Victoria to East Grinstead	2315	2315	2230	
East Grinstead to London Victoria	2230	2230	2300	

Route	Monday - Friday	Saturday	Sunday
London Victoria to East Grinstead	54 minutes with six exceptions of 55 minutes, two exceptions of 57 minutes and four further exceptions of 59 minutes permitted	54 minutes with one exception of 55 minutes and a further exception of 58 minutes permitted	56 minutes with one exception of 58 minutes and a further exception of 1 hour and 1 minute permitted
East Grinstead to London Victoria	55 minutes with two exceptions of 56 minutes, eighteen exceptions of 58 minutes and one exception of 59 minutes permitted	55 minutes with two exception of 58 minutes permitted	57 minutes

#### Route G2 LONDON BRIDGE – UCKFIELD

#### 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Uckfield, calling atEast Croydon, Oxted, Hurst Green, Edenbridge Town, Hever, Cowden, Ashurst, Eridge, Crowborough and Buxted
- 1.2 Limited Stops shall be made at Norwood Junction, Sanderstead, Riddlesdown, Upper Warlingham, and Woldingham. (South Croydon from December 2011 see Route G1)

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Uckfield, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### Towards Uckfield

- 2.2 Two Evening Peak services may omit to call at Hever, Cowden and Ashurst.
- 2.3 One additional service shall be provided from East Croydon to Uckfield departing before 0530, calling at the stations specified in Paragraph 1.1. This service may omit to call at Hever, Cowden and Ashurst.
- 2.4 Two additional services shall be provided from London Bridge to Uckfield at a half-hourly interval, the first departing before 0615, calling at the stations specified in Paragraph 1.1. These services are permitted to extend the Maximum Journey Time by 4 minutes.
- 2.5 One additional service shall be provided from Oxted to Uckfield arriving between 0745 and 0815, calling at the stations specified in Paragraph 1.1. The service may omit to call at Hever and Cowden.
- 2.6 Two additional services shall be provided from London Bridge to Uckfield at an hourly interval, the second departure shall be Evening Peak, calling at the stations specified in Paragraph 1.1.
- 2.7 Two additional services shall be provided from Oxted to Uckfield at an hourly interval, the first arriving an hour after second the service specified in Paragraph 2.6, calling at the stations specified in Paragraph 1.1.
- 2.8 One additional service shall be provided from London Bridge to Uckfield departing one hour later than the Late Service, calling at the stations specified in Paragraph 1.1, this service may omit to call at Hever, Cowden and Ashurst.

One additional service shall be provided from London Bridge to Oxted departing between 08:30 and 08:45, calling at East Croydon.

## Towards London Bridge

- 2.9 <sup>ii</sup>One additional service shall be provided from Uckfield to Oxted departing before 0615, calling at the stations specified in Paragraph 1.1. *This service may omit to call at Ashurst and Hever.*
- 2.10 Two additional morning Peak services shall be provided from Uckfield to London Bridge at an hourly interval, the first departing between 25 minutes and 35 minutes after the Early Service, calling at the stations as specified in Paragraph 1.1. One of these services may omit to call at Ashurst, Cowden and Hever. These services may terminate at East Croydon.
- 2.11 Four additional services shall be provided from Uckfield to London Bridge at hourly intervals, the first departing after 1700, calling at the stations specified in Paragraph 1.1. Two of these services may terminate at Oxted. Three of these services may omit to call at Ashurst, Cowden and Hever
- 2.12 One additional service shall be provided from Uckfield to East Croydon departing one hour later than the Late Service, calling at the stations specified in Paragraph 1.1, this service may omit to call at Hever, Cowden and Ashurst.

One additional service shall be provided between Oxted and East Croydon, departing between 09:00 and 09:14, without calling at intermediate stations.

## 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between London Bridge and Uckfield, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Uckfield

3.2 Two additional services shall be provided at an hourly interval from London Bridge to Uckfield. The first service shall depart one hour later than the Late Service, calling at the stations specified in Paragraph 1.1. These services shall call at Hever, Cowden and Ashurst to alight passengers only.

## Towards London Bridge

3.3 One additional service shall be provided from Uckfield to East Croydon departing after 2230, calling at the stations specified in Paragraph 1.1, this

service may omit to call at Hever, Cowden and Ashurst.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between Oxted and Uckfield, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Uckfield

4.2 The Early Service and the following service as specified in Paragraph 4.1 shall be extended to originate from East Croydon.

#### Towards Oxted

4.3 The Late Service as specified in Paragraph 4.1 shall be extended to terminate at East Croydon.

## 5. Limited Stops

- 5.1 **Norwood Junction** shall be served as follows:
  - (a) On Mondays to Fridays, all services departing from London Bridge after 2000, shall call.
  - (b) On Saturdays, all services departing from London Bridge after 2000 shall call.

### 5.2 **Sanderstead** shall be served as follows:

- (a) On Mondays to Fridays, the Early Servicefrom London Bridge as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, one service as specified in Paragraph 2.4 shall call.
- (c) On Mondays to Fridays, all services departing from London Bridge after 2000, shall call.

## 5.3 **Riddlesdown** shall be served as follows:

- (a) On Mondays to Fridays, the Early Service as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, one service as specified in Paragraph 2.4 shall call.

## 5.4 **Upper Warlingham** shall be served as follows:

- (a) On Mondays to Fridays, the Early Service from London Bridge as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, one service as specified in Paragraph 2.4 shall call.
- (c) On Mondays to Fridays, all services departing from London Bridge after 2000, shall call.

## 5.5 **Woldingham** shall be served as follows:

- (a) On Mondays to Fridays, the Early Service from London Bridge as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, one service as specified in Paragraph 2.4 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday Saturday Sunday		
London Bridge to Uckfield	0830	0730	1030
Uckfield to London Bridge	0800	0800	1130

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Uckfield	2200	2100	2230
Uckfield to London Bridge	2130	2130	2230

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Oxted and Uckfield.

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Uckfield	1 hour 13 minutes with five exceptions of 1 hour 15 minutes, three exceptions of 1 hour 17 minutes, one exception of 1 hour 19 minutes and one further exception of 1 hour 23 minutes permitted	1 hour 13 minutes	44 minutes
Uckfield to London Bridge	1 hour 15 minutes with three exceptions of 1 hour 17 minutes and two further exceptions of 1hour 18 minutes permitted	1 hour 15 minutes	44 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Oxted and Uckfield only.

## **Section 3 – East Coastway Services**

#### **Route H1 BRIGHTON – LEWES**

#### 1. Route Definition

- 1.1 Services shall be provided between Brighton and Lewes, calling at London Road Brighton, Moulsecoomb and Falmer.
- 1.2 There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between Brighton and Lewes, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Lewes

- 2.2 Two additional services at an hourly interval shall be provided from Brighton departing no later than 0700. One of these services is permitted to extend the Maximum Journey Time by 1 minute.
- 2.3 Two additional services shall be provided from Brighton departing between 0700 and 0730, at a 15 minute interval, calling at the stations as specified in Paragraph 1.1.
- One additional service shall be provided from Brighton departing between 0830 and 0915, calling at the stations as specified in Paragraph 1.1.
- 2.5 One additional service shall be provided from Brighton departing no earlier than 1915, calling at the stations as specified in Paragraph 1.1.
- 2.6 Two additional services shall be provided departing Brighton between 1700 and 1745, calling at the stations as specified in Paragraph 1.1. One of these services is permitted to extend the Maximum Journey Time by 1 minute.

#### Towards Brighton

- 2.7 Two additional services at an hourly interval shall be provided from Lewes departing no later than 0730, calling at the stations as specified in Paragraph 1.1.
- 2.8 Two additional services shall be provided from Lewes departing between 0735 and 0850, calling at the stations as specified in Paragraph 1.1.
- 2.9 The Early Service as specified in Paragraph 2.1 shall originate from **Newhaven Town**.

One additional service shall be provided from Lewes to Brighton departing between 1715 and 1730, calling at Falmer, Moulsecoomb and London Road.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Lewes, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Lewes

3.2 There are no variations to the core service from Brighton to Lewes specified in Paragraph 3.1.

## Towards Brighton

3.3 There are no variations to the core service from Lewes to Brighton specified in Paragraph 3.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 On Sundays there is no core service specified between Brighton and Lewes.

## Additional Services and Variations

## Towards Lewes

4.2 Two services shall be provided at a half-hourly interval from Brighton departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

## Towards Brighton

4.3 One service shall be provided arriving in Brighton before 0745, calling at the stations specified in Paragraph 1.1.

## 5. Limited Stops

There is no requirement for Limited Stops on this route

# **6.** Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday Saturday Sunday		
Brighton to Lewes	0845	0745	N/A
Lewes to Brighton	0815	0815	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Lewes	1615	1915	N/A
Lewes to Brighton	1745	1945	N/A

Route	Monday – Friday	Saturday	Sunday
Brighton to Lewes	16 minutes	15 minutes	15 minutes
Lewes to Brighton	16 minutes	16 minutes	16 minutes

#### **Route H2 BRIGHTON – SEAFORD**

## 1. Route Definition

- 1.1 Services shall be provided between Brighton and Seaford, calling at London Road Brighton, Moulsecoomb, Falmer, Lewes, Newhaven Town, Newhaven Harbour and Bishopstone.
- 1.2 Limited Stops shall be made at Southease.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between Brighton and Seaford, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Seaford

2.2 One service as specified in Paragraph 2.1 between 1730 and 1830 may omit to call at Newhaven Harbour.

#### 2.3 NOT USED

- 2.4 One additional service shall be provided from Brighton to Seaford arriving before 0630, calling at the stations specified in Paragraph 1.1.
- 2.5 One additional service shall be provided to Seaford arriving no later than 0730, calling at the stations specified in Paragraph 1.1. One these may start from Lewes.
- 2.6 One additional service shall be provided departing Brighton between 1745 and 1845, calling at the stations specified in Paragraph 1.1, this service may omit to call at Newhaven Harbour.
- 2.7 One additional service shall be provided from Lewes to Seaford, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from London Victoria as a portion of an evening Peak service requirement in Route E1 (London Victoria to Hastings), and may omit to call at Newhaven Harbour.
- 2.8 One additional service shall be provided departing Brighton one hour later than the Late Service, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Newhaven Harbour departing between 1915 and 1930, calling at all intermediate stations except Southease.

# Towards Brighton

- 2.9 Two service as specified in Paragraph 2.1 between 1815 and 1915 may omit to call at Newhaven Harbour
- 2.10 Six additional services shall be provided from Seaford to Brighton at intervals no less than 15 minutes, the first arriving before 0600 and the last after 0830, calling at the stations specified in Paragraph 1.1.
- 2.11 Two additional services shall be provided from Seaford to Lewes, calling at the stations specified in Paragraph 1.1. One shall depart between 0645 and 0700 and the other shall depart between 0815 and 0830 These service shall be extended to terminate at London Victoria as a portion of a morning Peak service requirement in Route E1 (London Victoria to Hastings).
- 2.12 Two additional services shall be provided from Seaford departing between *1845 and 2000*, calling at the stations specified in Paragraph 1.1.

#### 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided between Brighton and Seaford, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# Towards Seaford

- 3.2 One additional service shall be provided from Brighton to Seaford arriving before 0630, calling at the stations specified in Paragraph 1.1 and may omit to call at Bishopstone.
- 3.3 One additional service shall be provided departing Brighton one hour later than the Late Service, calling at the stations specified in Paragraph 1.1.

#### Towards Brighton

3.4 One additional service shall be provided from Seaford to Brighton arriving before 0600, calling at the stations specified in Paragraph 1.1.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided between Brighton and Seaford, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Seaford

4.2 There are no variations to the core service from Brighton to Seaford specified in Paragraph 4.1

# Towards Brighton

4.3 One additional service shall be provided from Seaford to Brighton arriving before 0845, calling at the stations specified in Paragraph 1.1.

# 5. Limited Stops

- 5.1 **Southease** shall be served as follows:
  - (a) On Mondays to Fridays, services specified in Paragraph 2.1 departing from Brighton between 0900 and 2000 shall call at hourly intervals.
  - (b) On Mondays to Fridays, one further service specified in Paragraph 2.1 departing from Brighton between 1500 and 1630 shall call to serve the needs of schools traffic.
  - (c) On Mondays to Fridays, two services specified in Paragraph 2.1 departing from Brighton between 2100 and the Late Service shall call at an hourly interval.
  - (d) On Mondays to Fridays, services specified in Paragraph 2.1 departing from Seaford between 0700 and 1830 shall call at hourly intervals.
  - (e) On Mondays to Fridays, one further service specified in Paragraph 2.1 departing from Seaford between 0700 and 0800 shall call.
  - (f) On Mondays to Fridays, two services specified in Paragraph 2.1 departing from Seaford between 1900 and 2100 shall call at an hourly interval.
  - (g) On Mondays to Fridays, the service specified in Paragraph 2.11 shall call.
  - (h) On Saturdays, services specified in Paragraph 3.1 departing from Brighton between 0900 and 1930 shall call at hourly intervals.
  - (i) On Saturdays, services in Paragraph 3.1 departing from Seaford between 0800 and 1930 shall call at hourly intervals.
  - (j) On Sundays, services specified in Paragraph 4.1 departing from Brighton between 0730 and 1930 shall call at hourly intervals.
  - (k) On Sundays, services specified in Paragraph 4.1 departing from Seaford between 0845 and 2030 shall call at hourly intervals.
  - (1) On Sundays, the service as specified in Paragraph 4.3 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
Brighton to Seaford	0715	0645	0800	
Seaford to Brighton	0930	0700	0915	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Seaford	2230	2230	2230
Seaford to Brighton	2315	2315	2245

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
Brighton to Seaford	36 minutes with four exceptions of 38 minutes and four further exceptions of 42 minutes permitted	36 minutes with four exceptio ns of 40 minutes permitted	35 minutes with one exception of 41 minutes permitted
Seaford to Brighton	35 minutes with four exceptions of 38 minutes and four further exceptions of 41 minutes permitted	35 minutes with four exceptio ns of 41 minutes permitted	33 minutes with fourteen exceptions of 41 minutes permitted

#### Route H3 BRIGHTON - HASTINGS via Eastbourne

# 1. Route Definition

- 1.1 Services shall be provided between Brighton and Seaford, calling at London Road Brighton, Moulsecoomb, Falmer, Lewes, Glynde, Berwick, Polegate, Hampden Park, Eastbourne, Hampden Park, Pevensey & Westham, Normans Bay, Cooden Beach, Collington, Bexhill and St Leonards Warrior Square.
- 1.2 Limited Stops shall be made at Pevensey Bay.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

- 2.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Hastings, calling at the stations specified in Paragraph 1.1.
- 2.2 Services as specified in Paragraph 2.1 may be extended to originate from or terminate at Ore.

#### Additional Services and Variations

#### **Towards Hastings**

2.3 One additional service shall be provided from Brighton to Eastbourne departing two hours later than the Late Service, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Eastbourne departing two hours earlier than the service specified in Paragraph 2.3, calling at the stations specified in Paragraph 1.1.

2.4 One additional service shall be provided from Brighton to Eastbourne departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

Two additional services shall be provided from Brighton to Eastbourne departing between 0615 and 0800 at an interval of 1h 30min, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Hastings to Ore departing between 0945 and 1000.

One additional service shall be provided from Brighton to Eastbourne departing between 1845 and 1900, calling at London Road, Moulsecoomb, Falmer, Lewes and Polegate.

Two additional services shall be provided from Eastbourne to Hastings departing between 0515 and 0645 at an interval of 1h 15min, calling at Pevensey, Bexhill and St Leonards. The second of these services shall call additionally at Normans Bay.

Two additional services shall be provided from Eastbourne to Hastings departing between 0600 and 0715 at an interval of 45 minutes, calling at the stations specified in Paragraph 1.1. These services may omit calling at Normans Bay.

One additional services shall be provided form Eastbourne to Ore departing between 0730 and 0745, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Pevenseay Bay.

# Towards Brighton

- 2.5 One additional service shall be provided from Eastbourne to Brighton arriving an hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1
- 2.6 The penultimate service to Brighton as specified in Paragraph 2.1 may originate from Eastbourne.
- 2.7 One additional service shall be provided from Hastings to Brighton departing no earlier than 2300, calling at the stations specified in Paragraph 1.1. This service may omit to call at Normans Bay, Berwick and Glynde.

One additional service shall be provided from Eastbourne to Brighton departing between 0645 and 0700, calling at Hampden Park, Polegate, Berwick, Glynde, Lewes, Falmer, Moulsecoomb, and London Road.

One additional service shall be provided from Hastings to Ore departing between 0530 and 0545, calling at St Leonards, Bexhill, Collington, Cooden Beach, Pevensey and Hampden Park.

# 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Ore, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# **Towards Hastings**

- 3.2 One additional service shall be provided from Brighton to Eastbourne departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1.
- 3.3 One additional service shall be provided from Brighton to Eastbourne departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

Four additional services shall be provided from Eastbourne to Ore at intervals of between 15 and 25 minutes, calling at Bexhill, St Leonards Warrior, Square and Hastings. The third of these services shall call additionally at Hampden Park, Pevensey & Westham, Normans Bay, Cooden Beach and Collington.

# Towards Brighton

3.4 One additional service shall be provided from Hastings to Brighton departing no earlier than 2300, calling at the stations specified in Paragraph 1.1. This service may omit to call at Normans Bay, Berwick and Glynde.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 On Sundays there is no core service specified between Brighton and Hastings.

#### Additional Services and Variations

# **Towards Hastings**

- 4.2 One additional service shall be provided from Eastbourne to Ore arriving no later than 0800, calling at the stations specified in Paragraph 1.1 and Hastings. This service may omit to call at Normans Bay.
- 4.3 Two additional services shall be provided from Brighton to Ore departing no later than 0900 at a one hour interval, calling at the stations specified in Paragraph 1.1 and Hastings. These services may omit to call at Glynde, Berwick, and Normans Bay.

# Towards Brighton

- 4.4 Two additional services shall be provided from Ore to Brighton departing no earlier than 2100 at a one hour interval, calling at the stations specified in Paragraph 1.1 and Hastings. These services may omit to call at Normans Bay, Berwick and Glynde.
- 4.5 One additional service shall be provided from Hastings to Eastbourne departing no earlier than 2300, calling at the stations specified in Paragraph 1.1. This service may omit to call at Normans Bay.

# 5. Limited Stops

- 5.1 **Pevensey Bay** shall be served as follows:
  - (a) On Mondays to Fridays, the penultimate service from Brighton as specified in Paragraph 2.1 shall call.
  - (b) On Mondays to Fridays, one train departing from Hastings after 1500 as specified in Paragraph 2.1 shall call

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
Brighton to Hastings	0915	0815	N/A	
Hastings to Brighton	0945	0745	N/A	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Hastings	1745	1845	N/A
Hastings to Brighton	2215	2215	N/A

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
Brighton to Hastings	1 hour 20 minutes with two exceptions of 1 hour 23 minutes permitted	1 hour 20 minutes	1 hour 13 minutes
Hastings to Brighton	1 hour 20 minutes with one exception of 1 hour 21 minutes	1 hour 20 minutes	1 hour 17 minutes

# Route H4 BRIGHTON – ASHFORD INTERNATIONAL via Eastbourne and Hastings

#### 1. Route Definition

- 1.1 Services shall be provided between Brighton and Ashford International, calling at, Lewes, Polegate, Eastbourne, Bexhill, St Leonards Warrior Square, Hastings, Rye, Appledore (Kent), and Ham Street.
- 1.2 Limited Stops shall be made at Pevensey & Westham, Ore, Three Oaks, Doleham and Winchelsea.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Ashford International, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Ashford International

- 2.2 One additional service shall be provided from Brighton to Ashford International departing no later than 0530, calling at the stations specified in Paragraph 1.1. This service may be provided by means of a connection at Eastbourne.
- 2.3 One additional service shall be provided from Hastings to Ashford International departing no later than 0530, calling at the stations specified in Paragraph 1.1.
- 2.4 One additional service shall be provided from Hastings to Ashford International departing no later than 0600, calling at the stations specified in Paragraph 1.1.
- 2.5 Two additional services shall be provided from Rye to Ashford International departing no later than 0830 at an interval no less than an hour, calling at the stations specified in Paragraph 1.1.
- 2.6 One additional service shall be provided from Hastings to Ashford International departing no earlier than 1700, calling at the stations specified in Paragraph 1.1.
- 2.7 Two additional services shall be provided from Rye to Ashford International departing no earlier than 1830 at an interval no less than an hour, calling at the stations specified in Paragraph 1.1.
- 2.8 One additional service shall be provided from Brighton to Hastings departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1.

2.9 Two additional services shall be provided from Brighton to Eastbourne at an hourly interval, departing an hour later than the service specified in Paragraph 2.6, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Ashford International, departing between 0615 and 0630, calling at stations specified in paragraph 1.1. This service shall call additionally at Hastings, Ore, Three Oaks and Winchelsea.

One additional service shall be provided from Brighton to Hastings departing between 2130 and 2145, calling at Lewes, Polegate, Eastbourne, Bexhill and St Leonards.

Two additional services shall be provided from Brighton to Eastbourne departing between 2228 and 2328 at an hourly interval, calling at Lewes and Polegate.

# Towards Brighton

- 2.10 One additional service shall be provided from Eastbourne to Brighton arriving before 0615, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Hampden Park, Falmer, Moulsecoomb and London Road (Brighton).
- 2.11 One additional service shall be provided from Hastings to Brighton arriving before 0730, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Hampden Park, Falmer, Moulsecoomb and London Road (Brighton).
- 2.12 Two additional services shall be provided from Ashford International to Rye departing no later than 0800 at an interval no less than an hour, calling at the stations specified in Paragraph 1.1.
- 2.13 One additional service shall be provided from Ashford International to Hastings arriving no later than 0945, calling at the stations specified in Paragraph 1.1.
- 2.14 Two additional services shall be provided from Ashford International to Rye departing no earlier than 1730 at an interval no less than an hour, calling at the stations specified in Paragraph 1.1.
- 2.15 One additional service shall be provided from Ashford International to Hastings departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1. This service shall call additionally atWinchelsea, Doleham and Three Oaks and Ore.

One additional service shall be provided from Ashford International to Hastings departing between 1945 and 2000, calling at stations specified in Paragraph 1.1.

# 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Ashford International, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# Towards Ashford International

- 3.2 One additional service shall be provided from Brighton to Eastbourne departing no less than one hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1.
- 3.3 Two additional services shall be provided at an hourly interval from Hastings, the first departing no later than 0545, calling at the stations as specified in Paragraph 1.1. The second service shall originate from Eastbourne.
- 3.4 Two additional services shall be provided from Brighton to Hastings at an hourly interval, the first departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1.
- 3.5 One additional service shall be provided from Brighton to Eastbourne departing an hour later than the service specified in Paragraph 3.3, calling at the stations specified in Paragraph 1.1

#### Towards Brighton

3.6 One additional service shall be provided from Ashford International to Hastings departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1.

Two additional services shall be provided from Eastbourne to Brighton at a 50 minutes interval, the second one arriving Brighton 65 minutes earlier than the Early Service, both of them calling at Hampden Park, Polegate, Lewes, Falmer, Moulsecoomb, and London Road.

# 4. Service Pattern – Sundays

# Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Ashford International, calling at the stations specified in Paragraph 1.1.
- 4.2 Between and including the Early and Late Services as specified in Paragraph 4.2, services shall call additionally at Glynde, Berwick and Hampden Park.

#### **Additional Services and Variations**

# Towards Ashford International

- 4.3 One additional service shall be provided from Brighton to Eastbourne departing no less than one hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1.
- 4.4 One additional service shall be provided from Hastings to Ashford International arriving no later than 0815, calling at the stations specified in Paragraph 1.1.
- 4.5 One additional service shall be provided from Eastbourne to Ashford International departing no less than one hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1.
- 4.6 One additional service shall be provided from Brighton to Hastings departing an hour later than the Late Service, calling at the stations specified in Paragraph 1.1.
- 4.7 Two additional services shall be provided from Brighton to Eastbourne at an hourly interval, the first departing two hours later than the Late Service, calling at the stations specified in Paragraph 1.1.

#### Towards Brighton

- 4.8 Two additional services shall be provided from Eastbourne to Brighton at an hourly interval, the first shall arrive no later than 0815, calling at the stations specified in Paragraph 1.1.
- 4.9 One additional service shall be provided from Ashford International to Hastings departing no earlier than 2215, calling at the stations specified in Paragraph 1.1.

# 5. Limited Stops

- 5.1 **Ore** shall be served as follows:
  - (a) On Mondays to Fridays, the Late Service from Brighton as specified in Paragraph 2.1 shall call.
  - (b) On Mondays to Fridays, the Early Service from Ashford International as specified in Paragraph 2.1 shall call.
  - (c) On Mondays to Fridays, the service specified in Paragraph 2.4 shall call.
  - (d) On Mondays to Fridays, the service specified in Paragraph 2.6 shall call.
  - (e) On Mondays to Fridays, the service specified in Paragraph 2.13 shall call.
  - (f) On Mondays to Fridays, the service specified in Paragraph 2.15 shall call.

- (g) On Saturdays, the Late Service from Brighton as specified in Paragraph 3.1 shall call.
- (h) On Saturdays, the Early Service from Ashford International as specified in Paragraph 3.1 shall call.
- (i) On Saturdays, the first service as specified in Paragraph 3.3 shall call.
- (j) On Saturdays, the service as specified in Paragraph 3.6 shall call.
- (k) On Sundays, the services specified in Paragraphs 4.4 and 4.9 shall call.

# 5.2 **Three Oaks**, **Doleham** and **Winchelsea** shall be served as follows:

- (a) On Mondays to Fridays, the Late Service from Brighton as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, the service specified in Paragraph 2.4 shall call.
- (c) On Mondays to Fridays, the service specified in Paragraph 2.6 shall call.
- (d) On Mondays to Fridays, the Early Service from Ashford International as specified in Paragraph 2.1 shall call.
- (e) On Mondays to Fridays, the service specified in Paragraph 2.13 shall call.
- (f) On Mondays to Fridays, the service specified in Paragraph 2.15 shall call.
- (g) On Saturdays, the first service as specified in Paragraph 3.3 shall call.
- (h) On Saturdays, the service as specified in Paragraph 3.6 shall call.
- (i) On Sundays, the services specified in Paragraphs 4.4 and 4.9 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Ashford International	0930	0830	1015
Ashford International to Brighton	0830	0830	1015

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Ashford International	2030	2030	2015
Ashford International to Brighton	2130	2130	2115

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
Brighton to Ashford International	1 hour 46 minutes with two exceptions of 1 hour 48 minutes permitted	1 hour <b>46</b> minutes	1 hour 48 minutes
Ashford International to Brighton	1 hour 47 minutes with two exceptions of 2 hours 7 minutes and one further exception of 1 hour and 48 minutes permitted	1 hour 47 minutes with one exceptions of 2 hours 7 minutes permitted	1 hour 50 minutes with one exception of 1 hour 56 minutes permitted

# Section 4 – West Coastway Services

#### **Route J1 BRIGHTON – HOVE**

#### 1. Route Definition

- 1.1 Services shall be provided between Brighton and Hove.
- 1.2 There is no requirement for Limited Stops on this route.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between Brighton and Hove, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Hove

- 2.2 Between and including the Early and Late Services specified in Paragraph 2.1, one interval may be extended to 40 minutes.
- 2.3 One additional service shall be provided from Brighton to Hove departing after 0005.

One additional service shall be provided from Brighton to Hove, departing between 0530 and 0545.

#### Towards Brighton

- 2.4 Between and including the Early and Late Services specified in Paragraph 2.1, two intervals may be extended to 40 minutes.
- 2.5 Two additional services shall be provided from Hove to Brighton at an interval not less than 30 minutes, the last service arriving before after 0645.
- 2.6 Two additional services shall be provided from Hove to Brighton at a half-hourly interval, the last arriving before 0800.
- 2.7 One additional service shall be provided to arrive between 0800 and 0830

#### 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between Brighton and Hove, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Hove

3.2 One additional service shall be provided from Brighton to Hove departing after 0005.

# Towards Brighton

3.3 There are no variations to the core service from Hove to Brighton specified in Paragraph 3.1.

# 4. Service Pattern – Sundays

# Core Service Specification

#### 4.1 NOT USED

Between and including the Early and Late services, one service per hour shall be provided from Brighton to Hove.

Between and including the Early and Late services, one service per hour shall be provided from Hove to Brighton.

#### Additional Services and Variations

#### Towards Hove

4.2 Three additional services shall be provided at an interval not less than one hour from Brighton to Hove, the first service shall depart no earlier than 2200.

One additional service shall be provided from Brighton to Hove departing between 0745 and 0800.

# Towards Brighton

4.3 One additional service shall be provided from Hove to Brighton arriving no later than 0815.

# 5. Limited Stops

There is no requirement for Limited Stops on this route

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
Brighton to Hove	0830	0600	1000	
Hove to Brighton	0900	0600	0915	

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday	
Brighton to Hove	2330	0000	1945	
Hove to Brighton	2345	2345	2100	

# 7. Maximum Journey Times

There is no specified Maximum Journey Time on this route.

#### Route J2 BRIGHTON – WEST WORTHING

#### 1. Route Definition

- 1.1 Services shall be provided between Brighton and West Worthing, calling at Hove, Aldrington, Portslade, Fishersgate, Southwick, Shoreham-by-Sea, Lancing, East Worthing, and Worthing.
- 1.2 There is no requirement for Limited Stops on this route.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between Brighton and West Worthing, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# **Towards West Worthing**

- 2.2 The Early Service as specified in Paragraph 2.1 may be provided by combining with a Route J4 (Brighton to Southampton Central) service.
- 2.3 One additional service shall be provided from Brighton to West Worthing arriving no later than 0545 calling at the stations specified in Paragraph 1.1, this service may be provided by combining with a Route J4 service.
- 2.4 Two additional services shall be provided from Brighton to Worthing at an hourly interval, the first departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Littlehampton departing between 0545 and 0600, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea and Angmering.

Two additional services shall be provided from Brighton to Littlehampton departing between 0720 and 0750 at a half hourly interval, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea and Angmering.

One additional service shall be provided from Brighton to West Worthing departing between 0600 and 0615, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Chichester departing between 0645 and 0700, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea, Angmering, Littlehampton, Ford and Barnham.

One additional service shall be provided from Brighton to Littlehampton departing not less than 20 minutes later than the Late Service, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea and Angmering.

# Towards Brighton

- 2.5 One additional service shall be provided from West Worthing to Brighton arriving no later than 0600, calling at the stations specified in Paragraph 1.1. This service shall depart from Barnham and call additionally at Ford, Littlehampton, Angmering, Goring By Sea and Durrington.
- 2.6 Two additional services shall be provided from West Worthing to Brighton between 0600 and 0900, calling at the stations specified in Paragraph 1.1. These services shall depart from Littlehampton and call additionally at Angmering, Goring By Sea and Durrington.
- 2.7 One additional service shall be provided each hour from West Worthing to Brighton between 1900 and 2200, calling at the stations specified in Paragraph 1.1. These services shall depart from Littlehampton, calling additionally at Angmering, Goring By Sea and Durrington.

Two additional services shall be provided from Bognor Regis to Brighton departing between 0530 and 0600 at an interval of 20 minutes, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea, Angmering, Ford and Barnham.

One additional service shall be provided from West Worthing to Brighton departing between 0700 and 0715, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Littlehampton to Brighton departing between 0715 and 0730, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea, Angmering, Ford and Barnham.

One additional service shall be provided from Bognor Regis to Brighton departing between 0800 and 0815, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea, Angmering, Ford and Barnham.

One additional service shall be provided from Brighton to Chichester, departing two hours after the Late service, calling at all intermediate stations.

# 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided between Brighton and West Worthing, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# **Towards West Worthing**

3.2 Two additional services shall be provided from Brighton to Worthing at an hourly interval, the first departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Littlehampton, departing between 0645 and 0700, calling at Hove, Portslade, Shoreham by Sea, Lancing, Worthing, West Worthing, Durrington, Goring By Sea and Angmering.

One additional service shall be provided from Brighton to Littlehampton, departing between 1815 and 1830, calling at all intermediate stations.

One additional service shall be provided from Brighton to Chichester, departing between 2230 and 2245, calling at all intermediate stations, including Littlehampton and excluding Bognor Regis.

#### Towards Brighton

- 3.3 One additional service shall be provided from Barnham to Brighton arriving no later than 0600, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Ford, Littlehampton, Angmering, Goring By Sea, Durrington and West Worthing.
- Four additional services shall be provided each hour from West Worthing to Brighton between 1845 and 2200, calling at all the intermediate stations.

#### 4. Service Pattern – Sundays

# Core Service Specification

4.1 There is no core service between Brighton and West Worthing on Sundays. Services are provided between Brighton and West Worthing within the requirements for the services as specified in Routes J3 and J4

#### Additional Services and Variations

# Towards West Worthing

4.2 One additional service shall be provided from Brighton to West Worthing arriving after 2300, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Brighton to Littlehampton departing between 2145 and 2200, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea and Angmering.

# Towards Brighton

#### 4.3 NOT USED.

#### 4.4 NOT USED

One additional service shall be provided from Littlehampton to Brighton departing between 0730 and 0745, calling at the stations specified in Paragraph 1.1 and West Worthing, Durrington, Goring By Sea and Angmering.

# 5. Limited Stops

There is no requirement for Limited Stops on this route

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to West Worthing	0900	0630	N/A
West Worthing to Brighton	0945	0645	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to West Worthing	1715	1745	N/A
West Worthing to Brighton	1700	1830	N/A

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
Brighton to West Worthing	28 minutes	28 minutes	N/A
West Worthing to Brighton	29 minutes	29 minutes	N/A

#### Route J3 BRIGHTON – PORTSMOUTH HARBOUR

# 1. Route Definition

- 1.1 Services shall be provided between Brighton and Portsmouth Harbour, calling at Hove, Portslade, Southwick, Shoreham-by-Sea, Lancing, Worthing, West Worthing, Durrington-on-Sea, Goring-by-Sea, Angmering, Ford, Barnham, Chichester, Southbourne, Emsworth, Havant, Fratton and Portsmouth & Southsea.
- 1.2 Limited Stops shall be made at Aldrington, Fishersgate, East Worthing, Fishbourne, Bosham, Nutbourne, Warblington, Bedhampton and Hilsea.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Portsmouth Harbour

- 2.2 The Early Service as specified in Paragraph 2.1 may omit to call at Portslade, Southwick, Lancing, Southbourne, and Emsworth.
- 2.3 Between the Early Service and 1530 services as specified in Paragraph 2.1 may omit to call at West Worthing, Durrington-on-Sea, Goring-by-Sea and Ford.
- 2.4 Between 1530 and 1930 two services specified in Paragraph 2.1 may terminate at Portsmouth & Southsea.
- 2.5 Between 1530 and up to and including the Late Service services as specified in Paragraph 2.1 may omit to call at Southbourne and Emsworth.
- 2.6 Four additional services shall be provided from Brighton to Portsmouth Harbour the first arriving no later than 0715 and last arriving no earlier than 0845, at intervals no less than twenty minutes, calling at the stations specified in Paragraph 1.1. Two of these services may omit to call at West Worthing, Durrington-on-Sea, Goring-by-Sea and Angmering, one of these services may also omit to call Ford, and one other service may omit to call at Southbourne and Emsworth.
- 2.7 One additional service shall be provided from Brighton to Portsmouth Harbour departing no earlier than 2115, calling at the stations specified in Paragraph 1.1.

Two additional services shall be provided from Barnham to Portsmouth Harbour, departing between 0500 and 0530 at an interval of 20 minutes,

calling at Chichester, Havant, Fratton and Portsmouth & Southsea. Two additional services shall be provided from Brighton to Bognor Regis departing between 0745 and 0815 at a 20-minute interval, calling at Hove, Portslade, Southwick, Shoreham By Sea, Lancing, Worthing, West Worthing, Durrington, Angmering, Ford and Barnham. The first of these service may omit to call at West Worthing and Ford.

# Towards Brighton

- 2.8 Between and including the Early Service and 2045 services as specified in Paragraph 2.1 may omit to call at Ford, Goring-by-Sea, Durrington-on-Sea and West Worthing.
- 2.9 Between 1830 and the Late Service three services specified in Paragraph 2.1 may originate from Portsmouth & Southsea.
- 2.10 Three additional services shall be provided from Portsmouth Harbour to Brighton the first arriving no later than 0715 and last arriving no earlier than 0830, at intervals no less than thirty minutes, calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Emsworth, Southbourne, Goring-by-Sea and West Worthing.
- 2.11 One additional service shall be provided from Portsmouth Harbour to Brighton departing between 1530 and 1730, calling at the stations specified in Paragraph 1.1. This service may omit to call at Ford, Angmering, Goringby-Sea, Durrington-on-Sea and West Worthing.
- 2.12 One additional service shall be provided from Portsmouth Harbour to Brighton departing no earlier than 2230, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Littlehampton.

# 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1. *The train required in the 2000-2100 hour shall be permitted to depart at 1956.* 

#### Additional Services and Variations

#### Towards Portsmouth Harbour

- 3.2 The Early Service as specified in Paragraph 3.1 may omit to call at Southbourne, and Emsworth.
- 3.3 Between the Early Service and 1830 services as specified in Paragraph 3.1

may omit to call at West Worthing, Durrington-on-Sea, Goring-by-Sea and Ford.

- 3.4 Between 1830 and up to and including the Late Service three services as specified in Paragraph 3.1 may omit to call at Southbourne and Emsworth.
- 3.5 One additional service shall be provided from Brighton to Portsmouth Harbour departing no earlier than 2115, calling at the stations specified in Paragraph 1.1.

Three additional services shall be provided from Barnham to Portsmouth Harbour at intervals of no more than 15 minutes, departing between 0515 and 0540, calling at Chichester, Havant, Fratton and Portsmouth & Southsea. The first of these services may terminate at Portsmouth & Southsea.

One additional service shall be provided from Littlehampton to Portsmouth Harbour departing between 0615 and 0630, calling at Barnham, Chichester, Havant, Fratton and Portsmouth & Southsea

# Towards Brighton

- 3.6 Between and including the Early Service and 2045 services as specified in Paragraph 3.1 may omit to call at Ford, Goring-by-Sea, Durrington-on-Sea and West Worthing.
- 3.7 Between 1800 and the Late Service as specified in Paragraph 3.1 one service may omit to call at Emsworth and Southbourne.
- 3.8 One additional service shall be provided from Portsmouth Harbour to Brighton departing no earlier than 2230, calling at the stations specified in Paragraph 1.1. This service may omit to call at Emsworth and Southbourne. This service shall call additionally at Littlehampton.

One additional service shall be provided from Chichester to Brighton between 0615 and 0630, calling at Barnham, Ford, Angmering, Goring By Sea, Durrington, West Worthing, Worthing, Lancing, Shoreham By Sea, Southwick, Portslade and Hove.

#### 4. Service Pattern – Sundays

#### Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Portsmouth Harbour, calling at the stations specified in Paragraph 1.1.
- 4.2 These services and the additional services below may be provided by combining between Brighton and West Worthing with the requirements for the services as specified in Route J2 Paragraph 4.

#### Additional Services and Variations

#### Towards Portsmouth Harbour

- 4.3 One additional service shall be provided from Ford to Portsmouth Harbour arriving no later than 0730, calling at the stations specified in Paragraph 1.1 and may omit to call at Southbourne and Emsworth. This service shall originate from Littlehampton.
- 4.4 Two additional services shall be provided at an interval of one hour from Brighton to Portsmouth Harbour arriving no later than 0930, calling at the stations specified in Paragraph 1.1. These services may omit to call at Portslade, Southwick, Lancing, West Worthing, Durrington-on-Sea, Goringby-Sea and Ford.

One additional service shall be provided from Brighton to Portsmouth Harbour, departing between 0715 and 0745, calling at Hove, Shoreham By Sea, Worthing, Angmering, Ford, Barnham, Chichester, Southbourne, Emsworth, Havant, Fratton and Portsmouth & Southsea.

4.5 One additional service shall be provided from Brighton to Chichester departing no earlier than 2230, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Littlehampton.

The service departing between 1800 and 1859 as defined in Paragraph 4.1 may terminate at Portsmouth & Southsea.

#### Towards Brighton

- 4.6 One additional service shall be provided from Barnham to Brighton arriving no later than 0700, calling at the stations specified in Paragraph 1.1 and may omit to call at Ford, Angmering, Goring-by-Sea, Durrington-on-Sea and West Worthing. This service shall call additionally at Littlehampton.
- 4.7 One additional service shall be provided from Portsmouth Harbour to Brighton departing no earlier than 2030, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Littlehampton.
- 4.8 Two additional services shall be provided from Portsmouth Harbour to Ford at a half hourly interval, departing no earlier than 2200, calling at the stations specified in Paragraph 1.1. These services shall call additionally and terminate at Littlehampton.

The service departing between 2000 and 2030 as defined in Paragraph 4.1 may originate at Portsmouth & Southsea.

One additional service shall be provided from Portsmouth Harbour to Brighton, calling at the stations specified in Paragraph 1.1 and Littlehampton, East Worthing, Fishergate and Aldrington.

# 5. Limited Stops

# 5.1 **Aldrington, Fishersgate** and **East Worthing** shall be served as follows:

- (a) On Mondays to Fridays, services between 1745 and up to and including the Late Service from Brighton as specified in Paragraph 2.1 shall call.
- (b) On Mondays to Fridays, services between 1715 and up to and including the Late Service to Brighton as specified in Paragraph 2.1 shall call.
- (c) On Mondays to Fridays, the first service to Brighton as specified in Paragraph 2.10 shall call.
- (d) On Mondays to Fridays, the service to Brighton as specified in Paragraph 2.11 shall call.
- (e) On Mondays to Fridays, the service to Brighton as specified in Paragraph 2.12 shall call.
- (f) On Saturdays, services between 1830 and up to and including the Late Service from Brighton as specified in Paragraph 3.1 shall call.
- (g) On Saturdays, services between 1845 and up to and including the Late Service to Brighton as specified in Paragraph 3.1 shall call.
- (h) On Saturdays, the service from Brighton as specified in Paragraph 3.5 shall call.
- (i) On Saturdays, the service specified in Paragraph 3.8 to Brighton shall

# 5.2 **Fishbourne, Bosham** and **Nutbourne** shall be served as follows:

- (a) On Mondays to Fridays, two services from Brighton as specified in Paragraph 2.6 shall call.
- (b) On Mondays to Fridays, the first service to Brighton as specified in Paragraph 2.10 shall call.
- (c) On Mondays to Fridays, the service to Brighton as specified in Paragraph 2.11 shall call.
- (d) On Mondays to Fridays, one service between 2030 and up to and including the Late Service to Brighton as specified in Paragraph 2.1 shall call.
- (e) On Saturdays, one service between 1930 and 2030 from Brighton as specified in Paragraph 3.1 shall call at Fishbourne and Nutbourne.
- (f) On Saturdays, the penultimate service before the Late Service to Brighton as specified in Paragraph 3.1 shall call.

- (g) On Sundays, the services specified in Paragraph 4.1 between Brighton and Portsmouth Harbour shall call.
- (h) On Sundays, the service specified in Paragraph 4.8 to Brighton shall call.

#### 5.3 **Warblington, Bedhampton and Hilsea** shall be served as follows:

- (a) On Mondays to Fridays, two services from Brighton as specified in Paragraph 2.6 shall call.
- (b) On Mondays to Fridays, the first service to Brighton as specified in Paragraph 2.10 shall call.
- (c) On Mondays to Fridays, the service to Brighton as specified in Paragraph 2.11 shall call at Warblington and Hilsea.
- (d) On Mondays to Fridays, one service between 2030 and the Late Service to Brighton as specified in Paragraph 2.1 shall call.
- (e) On Mondays to Fridays, the Late Service to Brighton as specified in Paragraph 2.1 shall call at Hilsea and Bedhampton.
- (f) On Mondays to Fridays, the service to Brighton as specified in Paragraph 2.12 shall call at Hilsea and Bedhampton.
- (g) On Saturdays, the penultimate service before the Late Service to Brighton as specified in Paragraph 3.1 shall call.
- (h) On Saturdays, the Late Service to Brighton as specified in Paragraph 3.1 shall call at Hilsea and Bedhampton.
- (i) On Saturdays, the service specified in Paragraph 3.8 to Brighton shall call at Hilsea and Bedhampton.
- (j) On Sundays, the services specified in Paragraph 4.1 between Brighton and Portsmouth Harbour shall call at Warblington and Bedhampton.
- (k) On Sundays, the service specified in Paragraph 4.8 to Brighton shall call at Bedhampton and Warblington.

#### 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
Brighton to Portsmouth Harbour	0930	0730	1015	
Portsmouth Harbour to Brighton	1000	0800	0900	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Portsmouth Harbour	2200	2200	2115
Portsmouth Harbour to Brighton	2200	2200	2100

# 7. Maximum Journey Times

Route	Monday – Friday	Saturday	Sunday
Brighton to Portsmouth Harbour	1 hour 25 minutes with three exceptions of 1 hour 26 minutes and one exception of 1 hour 28 minutes permitted	1 hour 23 minutes with one exception of 1 hour 26 minutes and one exceptions of 1 hour 30 minutes permitted	1 hour 37 minutes
Portsmouth Harbour to Brighton	1hour 22 minutes with one exception of 1 hour and 26 minutes and one further exceptions of 1 hour 33 minutes and two further exception of 1 hour 41 minutes permitted	1 hour 21 minutes with two exceptions of 1 hour 37 minutes and one further exception of 1 hour 41 minutes permitted	1 hour 3 7 minutes with two exceptions of 1 hour 48 minutes permitted

#### Route J4 BRIGHTON – SOUTHAMPTON CENTRAL

# 1. Route Definition

- 1.1 Services shall be provided between Brighton and Southampton Central, calling at Hove, Portslade, Southwick, Shoreham-by-Sea, Lancing, Worthing, West Worthing, Durrington-on-Sea, Goring-by-Sea, Angmering, Ford, Barnham, Chichester, Emsworth, Havant, Cosham, Fareham and; Swanwick or Eastleigh and Southampton Airport Parkway.
- 1.2 Limited Stops shall be made at Aldrington, Fishersgate, East Worthing, Bosham, Southbourne, Warblington, Porchester, Netley, Woolston and St.Denys.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 iiiBetween and including the Early and Late Services, one service per hour shall be provided at hourly intervals between Brighton and Southampton Central, calling at the stations specified in Paragraph 1.1

#### **Additional Services and Variations**

#### Towards Southampton Central

- 2.2 Between and including the Early and Late Services specified in Paragraph 2.1, services shall call at Eastleigh and Southampton Airport Parkway. Between 1300 and 1700, one service may omit to call at Southampton Airport Parkway, and the Late Service may omit to call at Eastleigh and Southampton Airport Parkway provided that the service shall call at Swanwick.
- 2.3 Between the Early Service and 1000 specified in Paragraph 2.1, one service may omit to call at Ford.
- 2.4 Between 1400 and the Late Service specified in Paragraph 2.1, one service may omit to call at West Worthing, Durrington-on-Sea, Goring-by-Sea and Angmering.
- 2.5 Between 1800 and up to and including the Late Service specified in Paragraph 2.1, two services may omit to call at Emsworth.
- 2.6 One additional service shall be provided from Barnham to Southampton Central arriving no later than 0600, calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick and may omit to call at Emsworth.
- 2.7 One additional service shall be provided from Brighton to Southampton Central arriving no later than 0715, calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick and may omit to call at Emsworth.

- 2.8 Two additional services shall be provided from Brighton to Southampton Central at an interval between forty-five minutes and one hour, arriving no later than 0815, calling at the stations specified in Paragraph 1.1. These services shall call at Swanwick. One service may omit to call at Portslade, Southwick, Lancing and Ford.
- 2.9 One additional service shall be provided from Brighton to Southampton Central arriving between 0830 and 0900, calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick, and may omit to call at Portslade, Southwick, Lancing Ford, and Emsworth.

# Towards Brighton

- 2.10 Between the and including the Early and Late Services specified in Paragraph 2.1, services shall call at Swanwick. Between 1300 and 1700, one service may omit to call at Swanwick provided that the service shall call at Eastleigh.
- 2.11 The Early Service specified in Paragraph 2.1 may omit to call at Emsworth, Goring-by-Sea and West Worthing.
- 2.12 Between 1300 and the Late Service specified in Paragraph 2.1, one service may omit to call at Swanwick, Cosham and Emsworth.
- 2.13 Between 1600 and the Late Service specified in Paragraph 2.1, one service may omit to call at Emsworth.
- 2.14 Between 1600 and the Late Service specified in Paragraph 2.1, three services may omit to call at Ford.
- 2.15 Between 1600 and the Late Service specified in Paragraph 2.1, two services may omit to call at Goring-by-Sea and West Worthing.
- 2.16 One additional service shall be provided from Southampton Central to Brighton departing no later than 0615 calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick.
- 2.17 One additional service shall be provided from Southampton Central to Brighton departing between 2100 and 2115, calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick and may omit to call at Emsworth.
- 2.18 One additional service shall be provided from Southampton Central to Barnham departing no earlier than 2130 calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick.
- 2.19 One additional service shall be provided from Southampton Central to Barnham departing no earlier than 2215, calling at the stations specified in Paragraph 1.1. This service shall omit to call at Southampton Airport Parkway and Eastleigh, and may omit to call at Swanwick and Emsworth. This service shall call additionally to terminate at Littlehampton.

One additional service shall be provided from Littlehampton to Brighton, departing between 1545 and 1600, calling at all intermediate stations.

# 3. Service Pattern – Saturdays

#### Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Southampton Central, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# Towards Southampton Central

- 3.2 Between and including the Early and Late Services specified in Paragraph 2.1, one interval may be extended to 1 hour ten minutes, and services shall call at Eastleigh and Southampton Airport Parkway. The Late Service may omit to call at Eastleigh and Southampton Airport Parkway provided that service shall call at Swanwick.
- 3.3 Two additional services shall be provided from Barnham to Southampton Central arriving no later than 0630 calling at the stations specified in Paragraph 1.1. These services shall call at Swanwick, and may omit to call at Emsworth.
- 3.4 One additional service shall be provided from Brighton to Southampton Central arriving no later than 0715 calling at the stations specified in Paragraph 1.1, this service may omit to call at Southwick. This service shall call at Swanwick.

# Towards Brighton

- 3.5 Between the and including the Early and Late Services specified in Paragraph 2.1, services shall call at Swanwick. Between 1300 and 1700, one service may omit to call at Swanwick and provided that the service shall call at Eastleigh.
- 3.6 Between 1300 and 1545 as specified in Paragraph 3.1, one service may omit to call at Cosham and Emsworth.
- 3.7 Between 1600 and the Late Service specified in Paragraph 3.1, two services may omit to call at Ford, Goring-by-Sea and West Worthing.
- 3.8 Two additional services shall be provided from Southampton Central to Brighton departing no earlier than 2100 at an interval of one hour, calling at the stations specified in Paragraph 1.1 and shall call at Swanwick. The first of these services may omit to call at Emsworth and shall call additionally at Littlehampton. The second service may be provided to Barnham and shall call additionally to terminate at Littlehampton.
- 3.9 One additional service shall be provided from Southampton Central to Barnham departing no earlier than 2115 calling at the stations specified in

- Paragraph 1.1 and shall call at Swanwick.
- 3.10 One additional service shall be provided from Southampton Central to Barnham departing no earlier than 2215, calling at the stations specified in Paragraph 1.1. This service shall call at Swanwick and may omit to call at Emsworth, and shall call additionally to terminate at Littlehampton.

#### 4. Service Pattern – Sundays

# Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided between Brighton and Southampton, calling at the stations specified in Paragraph 1.1. These services shall call at Swanwick.
- 4.2 These services and the additional services below may be provided by combining between Brighton and West Worthing with the requirements for the services as specified in Route J2 (Brighton to West Worthing) Paragraph 4...

#### Additional Services and Variations

# Towards Southampton Central

- 4.3 The Early Service from Brighton as specified in Paragraph 4.1 may omit to call at Swanwick provided this service shall call at Eastleigh and Southampton Airport Parkway.
- 4.4 One additional service shall be provided from Barnham to Southampton Central to arrive no later than 0845, calling at the stations as specified in Paragraph 1.1. This service shall call at Swanwick, and shall originate from Littlehampton.

#### Towards Brighton

- 4.5 Services from Southampton Central as specified in Paragraph 4.1 may omit to call at Ford, provided there is a connection at Barnham not exceeding 30 minutes.
- 4.6 Two additional services shall be provided at an interval of not less than 30 minutes from Southampton Central to Littlehampton, calling at the stations as specified in Paragraph 1.1, these services may omit to call at Ford. The first service shall depart no earlier than 45 minutes after the Late Service from Southampton Central.

One service departing Southampton Central between 0800 and 0859 may omit to call at Swanwick and call additionally at Southampton Airport Parkway and Eastleigh.

Two services departing Southamptpon Central between 1900 and 2059 may omit Swanwick and call additionally at Southampton Airport and Eastleigh.

# 5. Limited Stops

- 5.1 **Aldrington**, **Fishersgate** and **East Worthing** shall be served as follows:
  - (a) The service as specified in Paragraph 2.7 shall call.
  - (b) One service as specified in Paragraph 2.8 shall call.
  - (c) On Mondays to Fridays, services from Brighton between 1800 and up to and including the Late Service specified in Paragraph 2.1 shall call.
  - (d) On Mondays to Fridays, services from Southampton Central between 1700 and up to and including the Late Service specified in Paragraph 2.1 shall call.
  - (e) The service as specified in Paragraph 2.17 shall call.
  - (f) The Early Service from Brighton as specified in Paragraph 3.1 shall call.
  - (g) On Saturdays, the penultimate service and the Late Service from Brighton specified in Paragraph 3.1 shall call.
  - (h) On Saturdays, services from Southampton Central between 1800 and up to and including the Late Service specified in Paragraph 3.1 shall call.
  - (i) One service as specified in Paragraph 3.8 shall call.

#### 5.2 **Fishbourne** shall be served as follows:

- (a) The Early Service to Brighton as specified in Paragraph 2.1 shall call.
- (b) The service as specified in Paragraph 2.18 shall call.
- (c) The service as specified in Paragraph 3.9 shall call.

#### 5.3 **Bosham** shall be served as follows:

- (a) One service as specified in Paragraph 2.8 shall call.
- (b) The Early Service to Brighton as specified in Paragraph 2.1 shall call.
- (c) The service as specified in Paragraph 2.18 shall call.
- (d) The service as specified in Paragraph 3.4 shall call.
- (e) The service as specified in Paragraph 3.9 shall call.

#### 5.4 **Nutbourne** shall be served as follows:

- (a) The service as specified in Paragraph 2.18 shall call.
- (b) The service as specified in Paragraph 3.9 shall call.

#### 5.5 **Southbourne** shall be served as follows:

- (a) The services as specified in Paragraph 2.8 shall call.
- (b) The Early Service from Brighton as specified in Paragraph 2.1 shall call.
- (c) The service following the Early Service from Brighton as specified in Paragraph 2.1 shall call.
- (d) The service as specified in Paragraph 2.16 shall call.
- (e) The service as specified in Paragraph 2.18 shall call.
- (f) The Early Service from Brighton as specified in Paragraph 3.1 shall call.
- (g) The service as specified in Paragraph 3.4 shall call.
- (h) The second service as specified in Paragraph 3.8 shall call.
- (i) The service as specified in Paragraph 3.9 shall call.

# 5.6 **Warblington** shall be served as follows:

- (a) One service as specified in Paragraph 2.8 shall call.
- (b) The service as specified in Paragraph 2.18 shall call.
- (c) The service as specified in Paragraph 3.4 shall call.
- (d) The service as specified in Paragraph 3.9 shall call.

#### 5.7 **Porchester** shall be served as follows:

- (a) The service as specified in Paragraph 2.6 shall call.
- (b) One service as specified in Paragraph 2.8 shall call.
- (c) The service as specified in Paragraph 2.9 shall call.
- (d) The Early Service from Brighton as specified in Paragraph 2.1 shall call.
- (e) The penultimate service to the Late Service from Brighton as specified in Paragraph 2.1 shall call.
- (f) The service as specified in Paragraph 2.16 shall call.
- (g) The services as specified in Paragraph 2.17 shall call.
- (h) The service as specified in Paragraph 2.18 shall call.
- (i) The first service as specified in Paragraph 3.3 shall call.
- (i) The service as specified in Paragraph 3.4 shall call.

- (k) The services as specified in Paragraph 3.8 shall call.
- (1) The service as specified in Paragraph 3.9 shall call.

# 5.8 **Netley** shall be served as follows:

- (a) The service as specified in Paragraph 2.6 shall call.
- (b) One service as specified in Paragraph 2.8 shall call.
- (c) The service as specified in Paragraph 2.9 shall call.
- (d) The first service as specified in Paragraph 3.3 shall call.
- (e) The service as specified in Paragraph 3.9 shall call.

# 5.9 **Woolston** shall be served as follows:

- (a) The service as specified in Paragraph 2.6 shall call.
- (b) One service as specified in Paragraph 2.8 shall call.
- (c) The service as specified in Paragraph 2.9 shall call.
- (d) The service as specified in Paragraph 2.16 shall call.
- (e) The first service as specified in Paragraph 3.3 shall call.
- (f) The service as specified in Paragraph 3.9 shall call.

# 5.10 **St.Denys** shall be served as follows:

- (a) One service as specified in Paragraph 2.8 shall call.
- (b) The service as specified in Paragraph 2.9 shall call.
- (c) The service as specified in Paragraph 3.9 shall call.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Southampton Central	0930	0730	0945
Southampton Central to Brighton	0930	0830	0945

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Brighton to Southampton Central	2030	2015	2045
Southampton Central to Brighton	2030	2030	2115

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday
Brighton to Southampton Central	1 hour 50 minutes with two exceptions of 1 hour 55 minutes permitted	1 hour 54 minutes with one exceptions of 2 hours permitted	1 hour 48 minutes
Southampton Central to Brighton	1 hour 45 minutes with three exceptions of 1 hour 50 minutes and one further exception of 1 hour 55 minutes and one further exception of 2 hours 5 minutes permitted	1 hour 45 minutes with three exceptions of 1 hour 51 minutes, and one further exception of 2 hours 5 minutes permitted	1 hour 54 minutes with two exceptions of 1 hour 57 minutes and one further exception of 1 hour 59 minutes permitted

## Route J5 LITTLEHAMPTON – BOGNOR REGIS

#### 1. Route Definition

- 1.1 Services shall be provided between Littlehampton and Bognor Regis, calling at Ford and Barnham.
- 1.2 There is no requirement for Limited Stops on this route.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between Littlehampton and Bognor Regis, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Bognor Regis

- 2.2 Between the Early and Late Services one service as specified in Paragraph 2.1 may be omitted.
- 2.3 Two additional services shall be provided from Littlehampton to Bognor Regis arriving no later than 0700 calling at the stations specified in Paragraph 1.1.
- 2.4 One additional service shall be provided from Littlehampton to Bognor Regis arriving no later than 0830 calling at the stations specified in Paragraph 1.1.
- 2.5 One additional service shall be provided from Littlehampton to Bognor Regis arriving between 1530 and 1900 calling at the stations specified in Paragraph 1.1 This train shall operate as a through service from Brighton as specified in Route J2.
- 2.6 Two additional services shall be provided from Littlehampton to Bognor Regis at an interval of one hour, arriving no earlier than 2000 calling at the stations specified in Paragraph 1.1.
- 2.7 One additional service shall be provided from Littlehampton to Bognor Regis departing no earlier than 2230 calling at the stations specified in Paragraph 1.1.

## Towards Littlehampton

- 2.8 Between the Early and Late Services one service as specified in Paragraph 2.1 may be omitted.
- 2.9 One additional service shall be provided from Bognor Regis to Littlehampton arriving no later than 0545 calling at the stations specified in Paragraph 1.1.

2.10 Four additional services shall be provided from Bognor Regis to Littlehampton arriving between *0600* and 0800 calling at the stations specified in Paragraph 1.1.

#### 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Littlehampton and Bognor Regis, calling at the stations specified in Paragraph 1.1.

## **Additional Services and Variations**

# Towards Bognor Regis

3.2 There are no variations to the core service from Littlehampton to Bognor Regis specified in Paragraph 3.1.

#### Towards Littlehampton

3.3 One additional service shall be provided from Bognor Regis to Littlehampton arriving no later than 0545 calling at the stations specified in Paragraph 1.1.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 Between and including the Early and Late Services, one service per hour shall be provided between Littlehampton and Bognor Regis, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Bognor Regis

- 4.2 One additional service shall be provided from Littlehampton to Bognor Regis arriving no later than 0745 calling at the stations specified in Paragraph 1.1.
- 4.3 One additional service shall be provided from Littlehampton to Bognor Regis departing no earlier than 2300 calling at the stations specified in Paragraph 1.1.

## Towards Littlehampton

4.4 One additional service shall be provided from Barnham to Littlehampton departing no earlier than 2330 calling at the stations specified in Paragraph 1.1.

## 5. Limited Stops

There is no requirement for Limited Stops on this route

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunda			
Littlehampton to Bognor Regis	1030	0730	0845	
Bognor Regis to Littlehampton	0930	0630	0800	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Littlehampton to Bognor Regis	2100	2230	2145
Bognor Regis to Littlehampton	2230	2230	2200

Route	Monday - Friday	Saturday	Sunday
Littlehampton to Bognor Regis	21 minutes with four exceptions of 24 minutes and one further exception of 29 minutes permitted	18 minutes with two exceptions of 22 minutes permitted	16 minutes with one exception of 21 minutes permitted
Bognor Regis to Littlehampton	18 minutes with one exception of 19 minutes and one further exception of 25 minutes permitted	17 minutes with two exceptions of 29 minutes permitted	16 minutes

## Route J6 LITTLEHAMPTON – PORTSMOUTH & SOUTHSEA

#### 1. Route Definition

- 1.1 Services shall be provided between Littlehampton and Portsmouth & Southsea, calling at Ford, Barnham, Chichester, Fishbourne (Sussex), Bosham, Nutbourne, Southbourne, Emsworth, Warblington, Havant, Bedhampton, Hilsea and Fratton.
- 1.2 Limited Stops shall be made at Portsmouth Harbour.

## 2. Service Pattern – Mondays to Fridays

# Core Service Specification

- 2.1 (a)Between and including the Early and Late Services, one service per hour shall be provided between Littlehampton and Portsmouth & Southsea, calling at the stations specified in Paragraph 1.1.
- 2.1(b) Between 1700 and the Late Service one service may be provided by a service specified in the requirements of Route D2. This service may omit to call at Littlehampton.

#### Additional Services and Variations

#### Towards Portsmouth & Southsea

- 2.2 One additional service shall be provided from Littlehampton to Portsmouth & Southsea arriving no later than 0630 calling at the stations specified in Paragraph 1.1.
- 2.3 Two additional services shall be provided from Littlehampton to Portsmouth & Southsea arriving no later than 0830 calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Bosham and Warblington. One of these services shall be extended to Portsmouth Harbour.
- 2.4 Two additional services shall be provided from Littlehampton to Portsmouth Harbour departing no earlier than 1500 at an interval of one hour, calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Hilsea.

One additional service shall be provided from Littlehampton to Chichester, departing between 2200 and 2215, calling at all intermediate stations.

#### Towards Littlehampton

- 2.5 The Early Service and the following service as specified in Paragraph 2.1 may omit to call at Bosham. The Early Service may also omit to call Fishbourne (Sussex) and the following service may also omit to call at Warblington.
- 2.6 Two additional services shall be provided from Portsmouth & Southsea to Littlehampton at hourly intervals, calling at the stations specified in

Paragraph 1.1. These services may omit to call at Bedhampton.

2.7 One additional service shall be provided from Portsmouth Harbour to Littlehampton departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

The service after the Early service shall commence from Portsmouth Harbour.

One additional service shall be provided from Chichester to Littlehampton, departing between 0815 and 0830, calling at all intermediate stations.

One additional service shall be provided from Chichester to Bognor Regis, departing between 0900 and 0915, calling at all intermediate stations.

One additional service shall be provided from Chichester to Littlehampton, departing between 2345 and 0000, calling at all intermediate stations.

#### 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Littlehampton and Portsmouth & Southsea, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Portsmouth & Southsea

## 3.2 NOT USED

One additional service shall be provided from Littlehampton to Chichester between 2200 and 2215, calling at Ford and Barnham.

#### Towards Littlehampton

- 3.3 Between 1700 and the Late Services one service as specified in Paragraph 2.1 may be provided by a service specified in the requirements of Route D2 and may omit to call at Littlehampton & Southsea.
- 3.4 Two additional services shall be provided from Portsmouth Harbour to Littlehampton departing no earlier than 1730 at an interval of one hour, calling at the stations specified in Paragraph 1.1. These services may omit to call at Hilsea, Bedhampton, Warblington, Nutbourne, Bosham and Fishbourne (Sussex).
- 3.5 One additional service shall be provided from Portsmouth & Southsea to

Littlehampton, departing no earlier than 2030, calling at the stations specified in Paragraph 1.1, This service may omit to call at Bedhampton.

One additional service shall be provided from Portsmouth Harbour to Littlehampton between 2300 and 2315, calling at stations specified in Paragraph 1.1 and Portsmouth & Southsea.

One additional service shall be provided from Chichester to Littlehampton between 2345 and 0000, calling at Barnham and Ford.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 On Sundays there is no core service specified between Littlehampton and Portsmouth & Southsea.

## 5. Limited Stops

#### 5.1 **Portsmouth Harbour** shall be served as follows:

- (a) On Mondays to Fridays, the service specified in Paragraph 2.2 shall call.
- (b) On Mondays to Fridays, one service specified in Paragraph 2.3 shall call.
- (c) On Mondays to Fridays, the services specified in Paragraph 2.4 shall call.
- (d) On Saturdays, the services specified in Paragraph 3.4 shall call.
- (e) On Saturdays, the service specified in Paragraph 3.5 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Littlehampton to Portsmouth & Southsea	1000	0700	N/A
Portsmouth & Southsea to Littlehampton	0900	0700	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Littlehampton to Portsmouth & Southsea	1845	1845	N/A

to Littlehampton
------------------

Route	Monday - Friday	Saturday	Sunday
Littlehampton to Portsmouth & Southsea	56 minutes with one exception of 57 minutes and one further exception of 1 hour 2 minutes	56 minutes	N/A
Portsmouth & Southsea to Littlehampton	59 minutes	59 minutes	N/A

#### **Route J7** BARNHAM – BOGNOR REGIS

#### 1. Route Definition

- 1.1 Services shall be provided between Barnham and Bognor Regis.
- 1.2 There is no requirement for Limited Stops on this route.

# 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between Barnham and Bognor Regis.

#### Additional Services and Variations

# Towards Bognor Regis

- 2.2 One additional service shall be provided from Barnham to Bognor Regis arriving no later than 0600.
- 2.3 Four additional services shall be provided from Barnham to Bognor Regis arriving no later than 0930.
- 2.4 Five additional services shall be provided from Barnham to Bognor Regis departing between 1615 and 2230.
- 2.5 One additional service shall be provided from Barnham to Bognor Regis departing no earlier than 2330.

#### Towards Barnham

- 2.6 Four additional services shall be provided from Bognor Regis to Barnham arriving no later than 0930.
- 2.7 Four additional services shall be provided from Bognor Regis to Barnham departing between 1600 and 2215.
- 2.8 One additional service shall be provided from Bognor Regis to Barnham departing no earlier than 2300.

## 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between Barnham and Bognor Regis.

#### **Additional Services and Variations**

# Towards Bognor Regis

3.2 One additional service shall be provided from Barnham to Bognor Regis arriving no later than 0630.

#### 3.3 NOT USED

- 3.4 One additional service shall be provided from Barnham to Bognor Regis departing no earlier than 2200.
- 3.5 One additional service shall be provided from Barnham to Bognor Regis departing no earlier than 2330.

#### Towards Barnham

- 3.6 One additional service shall be provided from Bognor Regis to Barnham departing no earlier than 2145.
- 3.7 One additional service shall be provided from Bognor Regis to Barnham departing no earlier than 2300.

One additional service shall be provided from Bognor Regis to Barnham departing between 0600 and 0615.

# 4. Service Pattern – Sundays

## Core Service Specification

4.1 On Sundays there is no core service specified between Barnham and Bognor Regis.

#### Additional Services and Variations

# Towards Bognor Regis

- 4.2 One additional service shall be provided from Barnham to Bognor Regis arriving no later than 0700.
- 4.3 One additional service shall be provided from Barnham to Bognor Regis departing no earlier than 2245.

## Towards Barnham

4.4 One additional service shall be provided from Bognor Regis to Barnham departing no earlier than 2200.

# 5. Limited Stops

There is no requirement for Limited Stops on this route

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
Barnham to Bognor Regis	1000	0700	N\A
Bognor Regis to Barnham	1000	0700	N\A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
Barnham to Bognor Regis	1645	2145	N∖A
Bognor Regis to Barnham	1745	2130	N∖A

# 7. Maximum Journey Times

There are no specified Maximum Journey Times on this route.

## **Section 6 – South London Lines**

# Route K LONDON VICTORIA – HORSHAM via Hackbridge, Epsom and Dorking (semi-fast)

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Horsham, calling at Clapham Junction, Sutton (Surrey), Cheam, Ewell East, Epsom, Ashtead, Leatherhead, Boxhill & Westhumble, Dorking, Holmwood, Ockley and Warnham.
- 1.2 Limited Stops shall be made at Balham, Mitcham Junction, Hackbridge and Carshalton, and Limited Stops may be made at Mitcham Eastfields

# 2. Service Pattern – Mondays to Fridays

#### Core Service Specification

- 2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Dorking, calling at the stations specified in Paragraph 1.1.
- 2.2 One service per hour as specified in Paragraph 2.1 may omit to call at Ewell East and Boxhill & Westhumble.
- 2.3 One service per hour as specified in Paragraph 2.1 shall be extended to terminate at or originate from Horsham, calling at the stations specified in Paragraph 1.1,

#### Additional Services and Variations

## Towards Dorking and Horsham

- 2.4 Between 0545 and 0730, three additional services shall be provided at intervals between twenty-five minutes and fifty minutes from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Boxhill & Westhumble. These services may call at Mitcham Eastfields.
- 2.5 Between 0745 and 0800, one additional service shall be provided from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. This service may omit to call at Boxhill & Westhumble, Holmwood, Ockley and Warnham.
- 2.6 Between 0800 and 0830, one additional service shall be provided from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.

- 2.7 Between 0815 and 0840, one additional service shall be provided from Sutton to Dorking, calling at the stations specified in Paragraph 1.1. This service shall originate from London Victoria as a requirement of services specified in Route Q (London Victoria Sutton via Crystal Palace and West Croydon).
- 2.8 Between 0845 and 0914, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. This service may omit to call at Ewell East.
- 2.9 Between 1545 and 1614, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1.
- 2.10 Between 1615 and 1829, five additional services shall be provided at half-hourly intervals from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1.
- 2.11 Between 1830 and 1859, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.
- 2.12 Between 1900 and 1929, one additional service shall be provided from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.
- 2.13 Between 1930 and 2314, seven additional services shall be provided at half-hourly intervals from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. Two of these services may omit to call at Boxhill & Westhumble, provided that the interval between calls at this station does not exceed one hour. These services may call at Mitcham Eastfields.
- 2.14 Between 2315 and 2345, one additional service shall be provided from London Victoria to Horsham, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields. The calls at Holmwood, Ockley and Warnham may be to set down only.
- 2.15 On Fridays, between 2345 and 0030, one additional service shall be provided from London Victoria to *Epsom*, calling at the stations specified in Paragraph 1.1. This service shall also call at Balham, Mitcham Junction, Hackbridge and Carshalton, and may call at Mitcham Eastfields.

#### Towards London Victoria

- 2.16 Between 0545 and 0800, five additional services shall be provided from Horsham to London Victoria at half-hourly intervals, calling at the stations specified in Paragraph 1.1. One interval may be extended to 40 minutes. Two of these services may omit to call at Boxhill & Westhumble, and one may omit to call at Ewell East. These services may also call at Mitcham Eastfields.
- 2.17 Between 0900 and 0915, one additional service shall be provided from Horsham to London Victoria, calling at the stations specified in Paragraph 1.1. This service may also call at Mitcham Eastfields.

- 2.18 One additional service shall be provided in each of the time periods 0630-0700, 0845-0915 and 0945-1015 from Dorking to London Victoria, calling at the stations specified in Paragraph 1.1. The interval between the second and third of these services shall not exceed one hour. These services may also call at Mitcham Eastfields.
- 2.19 Between 1645 and 1715, one additional service shall be provided from Dorking to London Victoria, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.
- 2.20 Between 1645 and 1915, five additional services shall be provided at half-hourly intervals from Horsham to London Victoria, calling at the stations specified in Paragraph 1.1. These services may also call at Mitcham Eastfields.
- 2.21 Between 1945 and 2245, six additional services shall be provided at half-hourly intervals from Dorking to London Victoria, calling at the stations specified in Paragraph 1.1. These services may also call at Mitcham Eastfields. The second of these services shall be extended to originate from Horsham, calling at the stations specified in Paragraph 1.1 Two non-consecutive services may omit to call at Boxhill & Westhumble.
- 2.22 One additional service shall be provided from Dorking to West Croydon, departing no earlier than 2255, calling at the stations to Sutton (Surrey) specified in Paragraph 1.1. This service shall call additionally at Carshalton Beeches, Wallington and Waddon.
- 2.23 One additional service shall be provided from Dorking to Epsom, departing no earlier than 2325, calling at Leatherhead and Ashtead.

One additional service shall be provided from Dorking to West Croydon departing between 2300 and 2315, calling at the sations specified in Paragraph 1.1. This service shall call additionally at Box Hill & Westhumble.

# 3. Service Pattern – Saturdays

# Core Service Specification

- 3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Dorking, calling at the stations specified in Paragraph 1.1.
- 3.2 One service per hour as specified in Paragraph 3.1 may omit to call at Ewell East and Boxhill & Westhumble.
- 3.3 One service per hour as specified in Paragraph 3.1 shall be extended to terminate at or originate from Horsham, calling at the stations specified in Paragraph 1.1,

#### **Additional Services and Variations**

# Towards Dorking and Horsham

- 3.4 Between 0630 and 0700, one additional service shall be provided from West Croydon to Dorking, calling at the stations specified in Paragraph 1.1. This servicemay omit to call at Boxhill & Westhumble and shall call additionally at Waddon, Wallington and Carshalton Beeches.
- 3.5 Between 0630 and 0659, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.
- 3.6 Between 1745 and 1814, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1.
- 3.7 Between 1815 and 2305, ten additional services shall be provided at half-hourly intervals from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. These services may call at Mitcham Eastfields. Five of these services may omit to call at Ewell East and Boxhill & Westhumble, provided an hourly service is maintained at these stations.
- 3.8 Between 2315 and 2345, one additional service shall be provided from London Victoria to Dorking, calling at the stations specified in Paragraph 1.1. This service may call at Mitcham Eastfields.

#### Towards London Victoria

3.9 Between 0615 and 0815, four additional services shall be provided from Dorking to London Victoria at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The first and last of these services may omit to call at Boxhill & Westhumble. The second and last of these services may omit to call at Ewell East.

- 3.10 Between 1845 and 2245, eight additional services shall be provided from Dorking to London Victoria at half-hourly intervals, calling at the stations specified in Paragraph 1.1. These services may also call at Mitcham Eastfields. Services may omit to call at Boxhill & Westhumble and Ewell East, provided an hourly service is maintained at these stations by services departing from Dorking to London Victoria up to 2200.
- 3.11 One additional service shall be provided from Dorking to West Croydon departing no earlier than 2255, calling at the stations to Sutton (Surrey) specified in Paragraph 1.1 and additionally at Carshalton Beeches, Wallington and Waddon.
- 3.12 Between 2315 and 2345, one additional service shall be provided from Dorking to Epsom, calling at the stations specified in Paragraph 1.1. This service may omit to call at Boxhill& Westhumble.

# 4. Service Pattern – Sundays

# Core Service Specification

- 4.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and **Dorking**, calling at the stations specified in Paragraph 1.1. These services shall also call at Balham, Mitcham Junction, Hackbridge and Carshalton.
- 4.2 There is no requirement for services to call at Holmwood, Ockley, Warnham and Horsham on this route on Sundays.

#### Additional Services and Variations

#### **Towards Dorking**

# 4.3 NOT USED

One additional service shall be provided from London Victoria to Dorking departing between 0645 and 0700. This service may operate as part of the requirements for Route O between London Victoria and Sutton via West Croydon.

#### Towards London Victoria

- 4.4 One additional service shall be provided from Epsom to London Victoria arriving no later than 0800 calling at the stations specified in Paragraph 1.1. This service may count as the Early Service for the purposes of the arrival time specified in paragraph 6.
- 4.5 One additional service shall be provided from Dorking to London Victoria no less than 30 minutes later than the Late Service, calling at the stations specified in Paragraph 1.1.

# 5. Limited Stops

# 5.1 Balham, Mitcham Junction, Hackbridge and Carshalton

- (a) The services specified in Paragraph 2.4 shall call.
- (b) The service specified in Paragraph 2.9 shall call at Mitcham Junction.
- (c) The services specified in Paragraphs 2.10, 2.11, 2.12, 2.13, 2.14 and 2.15 shall call.
- (d) The services specified in Paragraph 2.16 shall call, one service may omit to call at Hackbridge, and one may omit to call at Balham.
- (e) The service specified in Paragraph 2.17 shall call.
- (f) One service specified in Paragraph 2.18 shall call.
- (g) The services specified in Paragraphs 2.19, 2.20 and 2.21 shall call.
- (h) On Saturdays, the services specified in Paragraphs 3.5 and 3.8 shall call.

## 5.2 **Mitcham Eastfields** may be served as follows by:.

- (a) The services as specified in Paragraph 2.1 at half-hourly intervals.
- (b) The services as specified in Paragraph 3.1 at half-hourly intervals.
- (c) The services as specified in Paragraph 4.1 at half-hourly intervals.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Victoria to Horsham	1045	0845	0845
Horsham to London Victoria	1130	0930	0800

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Victoria to Horsham	1520	1730	2300
Horsham to London Victoria	1600	1800	2245

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between London Victoria and Dorking.

Route	Monday - Friday	Saturday	Sunday
London Victoria to <b>Dorking</b>	53 minutes with sixteen exceptions of 58 minutes and three further exceptions of 1 hour permitted	53 minutes with two exception s of 1 hour permitted	59 minutes
<b>Dorking</b> to London Victoria	56 minutes with eleven exceptions of 1 hour and five further exceptions of 1 hour 5 minutes permitted	52 minutes with two exceptions of 54 minutes permitted	1 hour with one exception of 1 hour 3 minutes and one further exception of 1 hour 7 minutes permitted

# Route L LONDON VICTORIA – EPSOM via Hackbridge

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Epsom, calling at Clapham Junction, Balham, Mitcham Eastfields, Mitcham Junction, Hackbridge, Carshalton, Sutton (Surrey), Cheam and Ewell East.
- 1.2 Limited Stops may be made at Mitcham Eastfields

#### 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Epsom

- 2.2 Between 0700 and 0800, two additional services shall be provided from London Victoria to Sutton, calling at the stations specified in Paragraph 1.1. The second of these services shall depart London Victoria one hour before the Early Service. The first of these services shall call additionally at Wandsworth Common. One of these services may omit to call at Mitcham Eastfields.
- 2.3 Between 0830 and 0915, one additional service shall be provided from Sutton to Epsom, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from London Bridge via Streatham as a requirement of Route T (London Bridge to West Croydon via Peckham Rye, Tulse Hill and Selhurst).
  - Between 1845 and 1900, one additional service shall be provided from Sutton to Epsom, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from London Bridge via Streatham as a requirement of Route T (London Bridge to West Croydon via Peckham Rye, Tulse Hill and Selhurst). This service shall call additionally at Mitcham Eastfields, Hackbridge and Carshalton.
- 2.4 Between 1615 and 1744, two additional services shall be provided at an hourly interval from London Victoria to Epsom, calling at the stations specified in Paragraph 1.1.
- 2.5 Between 1745 and 1814, one additional service shall be provided from London Victoria to Epsom, calling at the stations specified in Paragraph 1.1. This service shall depart from London Victoria half an hour later than the last service specified in Paragraph 2.4.
- 2.6 Between 1930 and 2315 eight additional services shall be provided at half

hourly intervals from London Victoria to Epsom calling at the stations specified in Paragraph 1.1. These services may call at Mitcham Eastfields.

#### Towards London Victoria

- 2.7 Between 0515 and 0544, one additional service shall be provided from Sutton to Mitcham Junction, calling at the stations specified in Paragraph 1.1. This service shall be extended to terminate at London Bridge via Streatham as a requirement of Route T (London Bridge to West Croydon via Peckham Rye, Tulse Hill and Selhurst). This service may call at Mitcham Eastfields.
- 2.8 Between 0545 and 0614, one additional service shall be provided from Epsom to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.9 Between 0730 and 0759, one additional service shall be provided from Epsom to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.10 Between 0815 and 0845, one additional service shall be provided from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.11 Between 0840 and 0859, one additional service shall be provided from Epsom to London Victoria, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from **Guildford**.
- 2.12 Between 0900 and 0929, one additional service shall be provided from Epsom to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.13 Between 0930 and 0959, one additional service shall be provided from Epsom to London Victoria, calling at Ewell East, Cheam, Sutton (Surrey) and Clapham Junction.
- 2.14 Between 2015 and 2359 six additional services shall be provided at half-hourly intervals from Epsom to London Victoria calling at stations specified in Paragraph 1.1. These services may call additionally at Mitcham Eastfields. *One of these service may omit to call at Ewell East and Cheam.*

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### 3.2 NOT USED

One additional service shall be provided from London Victoria to Epsom between 0015 and 0030, calling at the stations specified in Paragraph 1.1.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

4.2 One additional service shall be provided from London Victoria to Epsom, departing between 2300 and 2330, calling at the stations specified in Paragraph 1.1..

# 5. Limited Stops

- 5.1 **Mitcham Eastfields** may be served as follows by:
  - (a) The services as specified in Paragraph 2.1 at half-hourly intervals.
  - (b) The services as specified in Paragraph 3.1 at half-hourly intervals.
  - (c) The services as specified in Paragraph 4.1 at half-hourly intervals.

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Epsom	0930	0800	1015	
Epsom to London Victoria	1100	0745	0930	

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Epsom	1545	2245	2315	
Epsom to London Victoria	1645	2245	2000	

Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom	42 minutes with six exceptions of 44 minutes and five further exceptions of 47 minutes permitted	42 minutes	41 minutes with one exception of 43 minutes and one further exception of 55 minutes permitted
Epsom to London Victoria	42 minutes with five exceptions of 44 minutes permitted	41 minutes with one exception of 42 minutes permitted	42 minutes with one exception of 46 minutes and one further exception of 50 minutes permitted

# Route M LONDON VICTORIA – EPSOM via Selhurst, West Croydon and Sutton

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Epsom, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Common, Norbury, Thornton Heath, Selhurst, West Croydon, Waddon, Wallington, Carshalton Beeches, Sutton (Surrey), Cheam and Ewell East.
- 1.2 There is no requirement for Limited Stops on this route.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### Towards Epsom

- 2.2 Services from London Victoria to Epsom as specified in Paragraph 2.1 may omit to call at Battersea Park.
- 2.3 Between 1845 and 1915, one additional service shall be provided between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.
- 2.4 Between 2245 and 2315, one additional service shall be provided between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### Towards London Victoria

- 2.5 Between 0530 and 0600, one additional service shall be provided from Epsom to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.6 Between 2315 and 2345, one additional service shall be provided from Epsom to West Croydon, calling at the stations specified in Paragraph 1.1.

Between 0445 and 0500, one additional service shall be provided from Balham to London Victoria, calling at Clapham Jn and Battersea Park.

# 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between London Victoria and Epsom, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

# Towards Epsom

3.2 Services from London Victoria to Epsom as specified in Paragraph 3.1 may omit to call at Battersea Park.

#### Towards London Victoria

3.3 Between 2245 and 2345, one additional service shall be provided from Epsom to West Croydon, calling at the stations specified in Paragraph 1.1.

# 4. Service Pattern – Sundays

4.1 There is no requirement for services to be provided on Route M on Sundays - services between London Victoria and Epsom on Sundays are provided by London Victoria-Dorking services, specified in Route K.

# 5. Limited Stops

5.1 There is no requirement for Limited Stops on this route.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom	1000	0800	N/A
Epsom to London Victoria	1100	0800	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom	1445	2245	N/A
Epsom to London Victoria	1600	2300	N/A

Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom	59 minutes	59 minutes	N/A
Epsom to London Victoria	54 minutes and one exception of 55 minutes permitted.	54 minutes with one exception of 55 minutes permitted	N/A

# Route N LONDON VICTORIA – EPSOM DOWNS via Selhurst, West Croydon and Sutton

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Epsom Downs, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Common, Norbury, Thornton Heath, Selhurst, West Croydon, Waddon, Wallington, Carshalton Beeches, Sutton (Surrey), Belmont and Banstead.
- 1.2 There is no requirement for Limited Stops on this route.

#### 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided between London Victoria and Epsom Downs, calling at the stations specified in Paragraph 1.1., except that services from London Victoria to Epsom Downs may omit calls at Battersea Park.

#### Additional Services and Variations

#### Towards Epsom Downs

- 2.2 Services from London Victoria to Epsom Downs as specified in Paragraph 2.1 may omit to call at Battersea Park.
- 2.3 Between 0545 and 0615, one additional service shall be provided from West Croydon to Epsom Downs, calling at the stations specified in Paragraph 1.1.This service shall commence from Norwood Junction.
- 2.4 Between 0545 and 0845, six additional services shall be provided at half-hourly intervals from London Victoria to Epsom Downs, calling at the stations specified in Paragraph 1.1.
- 2.5 Between 1545 and 1915, seven additional services shall be provided at half-hourly intervals from London Victoria to Epsom Downs, calling at the stations specified in Paragraph 1.1. The first of these services may omit to call at Battersea Park. One interval may be extended to 45 minutes. The last of these services may be provided between London Victoria and Sutton as a requirement of Route Q (London Victoria to Sutton via Crystal Palace).
- 2.6 Between 2015 and 2345, four additional services shall be provided at hourly intervals from Sutton to Epsom Downs, calling at the stations specified in Paragraph 1.1. These services shall be extended to originate from London Victoria as a requirement of Route Q.
  - Between 1615 and 1630, one additional service shall be provided from London Victoria to Selhurst, calling at the stations specified in Paragraph 1.1

#### Towards London Victoria

- 2.7 Between 0700 and 0915, three additional services shall be provided at hourly intervals from Epsom Downs to London Victoria, calling at the stations specified in Paragraph 1.1, the last of these services may omit to call at Battersea Park.
- 2.8 Between 1615 and 2030, eight additional services shall be provided at half-hourly intervals from Epsom Downs to London Victoria, calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Thornton Heath, Norbury and Wandsworth Common. Two services may omit to call at Battersea Park. Two intervals between services may be extended up to 40 minutes.
- 2.9 Between 2045 and 2300, three additional services shall be provided at hourly intervals from Epsom Downs to Sutton, calling at the stations specified in Paragraph 1.1. These services shall be extended to terminate at London Victoria as a requirement of Route Q.
- 2.10 Between 2330 and 2359, one additional service shall be provided from Epsom Downs to West Croydon, calling at the stations specified in Paragraph 1.1.

## 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between London Victoria and Epsom Downs, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Epsom Downs

- 3.2 Services from London Victoria to Epsom Downs as specified in Paragraph 3.1 may omit to call at Battersea Park.
- 3.3 Between 2315 and 2359, one additional service shall be provided from Sutton to Epsom Downs, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from London Victoria as a requirement of Route Q.

#### Towards London Victoria

- 3.4 One additional service shall be provided from Sutton to London Victoria, arriving one hour earlier than the Early Service, calling at the stations specified in Paragraph 1.1.
- 3.5 Between 2300 and 2359, two additional services shall be provided from Epsom Downs to West Croydon, calling at the stations specified in Paragraph 1.1.

# 4. Service Pattern – Sundays

4.1 There is no requirement for services to be provided on Route N between London Victoria and Epsom Downs on Sundays.

# 5. Limited Stops

5.1 There is no requirement for Limited Stops on this route.

# 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom Downs	1030	0730	N/A
Epsom Downs to London Victoria	0730	0830	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom Downs	1515	2215	N/A
Epsom Downs to London Victoria	1530	2230	N/A

Route	Monday - Friday	Saturday	Sunday
London Victoria to Epsom Downs	56 minutes with seven exceptions of 59 minutes and five further exceptions of 1 hour 3 minutes permitted	54 minutes with one exception of 55 minutes and a further exception of 1 hour and 7 minutes permitted	N/A
Epsom Downs to London Victoria	55 minutes with five exceptions of 1 hour permitted	53 minutes with one exception of 54 minutes permitted	N/A

# Route O LONDON VICTORIA – SUTTON via Selhurst and West Croydon

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Sutton, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Common, Norbury, Thornton Heath, Selhurst, West Croydon, Waddon, Wallington and Carshalton Beeches.
- 1.2 There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between London Victoria and Sutton at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### **Towards Sutton**

- 2.2 Between 0530 and 0600, one additional service shall be provided from West Croydon to Sutton, calling at the stations specified in Paragraph 1.1.
- 2.3 Between 0815 and 0845, one additional service shall be provided from London Victoria to Sutton, calling at the stations specified in Paragraph 1.1. This service may omit to call at Battersea Park.
- 2.4 Between 1915 and 2245, seven services shall be provided from London Victoria to Sutton at half-hourly intervals, calling at the stations specified in Paragraph 1.1.
- 2.5 One additional service shall be provided from London Victoria to Sutton departing between 2315 and 2340, calling at the stations specified in Paragraph 1.1.
- 2.6 One additional service shall be provided from London Victoria to Sutton departing no earlier than 2345, calling at the stations specified in Paragraph 1.1.
- 2.7 On Fridays, one additional service shall be provided from London Victoria to Sutton departing no earlier than 0015, calling at the stations specified in Paragraph 1.1.

#### Towards London Victoria

- 2.8 Services from Sutton to London Victoria as specified in Paragraph 2.1 may omit to call at Battersea Park.
- 2.9 Between 0600 and 0630, one additional service shall be provided from Sutton

- to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.10 Between 0700 and 0730, one additional morning Peak service shall be provided from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1.
- 2.11 Between 0745 and 0815, one additional morning Peak service shall be provided from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1. This service may omit to call at Wandsworth Common and Battersea Park.
- 2.12 Between 0900 and 0930, one additional service shall be provided from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1
- 2.13 Between 2045 and 2259, five additional services shall be provided at half-hourly intervals from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1. These services may omit to call at Battersea Park.
- 2.14 Between 2300 and 2329, one additional service shall be provided from Sutton to West Croydon, calling at the stations specified in Paragraph 1.1.

## 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided between London Victoria and Sutton at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### **Towards Sutton**

- 3.2 One additional service shall be provided from London Victoria to Sutton departing between 2315 and 2340, calling at the stations specified in Paragraph 1.1.
- 3.3 One additional service shall be provided from London Victoria to Sutton departing no earlier than 2345, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from London Victoria to Sutton departing no earlier than 0015, calling at the stations specified in Paragraph 1.1.

# Towards London Victoria

3.4 Services from Sutton to London Victoria as specified in Paragraph 2.1 may omit to call at Battersea Park.

One additional service shall be provided from Selhurst to London Victoria between 0600 and 0615, calling at all intermediate stations.

Two additional services shall be provided from Selhurst to London Victoria between 0730 and 0830, at a half hourly interval, calling at all intermediate stations except Battersea Park.

# 4. Service Pattern – Sundays

# Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Sutton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### **Towards Sutton**

- 4.2 Services may omit to call at Battersea Park and Wandsworth Common.
- 4.3 The Late Service shall be extended to Epsom, calling at the stations specified in Paragraph 1.1 and additionally at Cheam and Ewell East.

#### Towards London Victoria

- 4.4 Services departing Sutton may omit to call at Wandsworth Common and Battersea Park.
- 4.5 The Early Service shall be extended to originate from **Epsom**, calling additionally at Ewell East and Cheam and at the stations specified in Paragraph 1.1.
- 4.6 One additional service shall be provided from Sutton to London Victoria, calling at the stations specified in Paragraph 1.1. This service shall depart Sutton no less than 10 minutes later than the Late Service.

# 5. Limited Stops

5.1 There is no requirement for Limited Stops on this route.

# **6.** Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Sutton	1000	0830	0745	
Sutton to London Victoria	1045	0915	0745	

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to Sutton	1600	2300	2315
Sutton to London Victoria	1645	2315	2315

Route	Monday - Friday	Saturday	Sunday
London Victoria to Sutton	44 minutes	43 minutes with one exception of 44 minutes permitted	44 minutes
Sutton to London Victoria	43 minutes with seven exceptions of 45 minutes permitted	42 minutes with one exception of 43 minutes permitted	40 minutes with three exceptions of 44 minutes permitted

# Route P LONDON VICTORIA - CATERHAM via Selhurst and East Croydon

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Caterham, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Common, Norbury, Thornton Heath, Selhurst, East Croydon, South Croydon, Purley Oaks, Purley, Kenley, Whyteleafe and Whyteleafe South.
- 1.2 There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Caterham, calling at the stations specified in Paragraph 1.1.

#### **Additional Services and Variations**

#### Towards Caterham

- 2.2 Between 0600 and 0730 two additional services shall be provided from London Victoria to Purley calling at the stations specified in paragraph 1.1. These services may omit to call at South Croydon and Purley Oaks, and shall be extended to Tattenham Corner as a requirement of Route Y (London Bridge Tattenham Corner via Norwood Junction).
- 2.3 Between 0630 and 0700 one additional service shall be provided from London Victoria to East Croydon calling at the stations specified in paragraph 1.1. This service may omit to call at Battersea Park.
- 2.4 The Early Service may omit to call at Battersea Park.
- 2.5 Between 1630 and 1900, five additional services shall be provided at half-hourly intervals from London Victoria to Caterham, calling at the stations in Paragraph 1.1. These services may omit to call at Battersea Park, Wandsworth Common, Balham, Streatham Common, Norbury, Thornton Heath and Selhurst.
- 2.6 Between 1700 and 1830, three additional services shall be provided at half intervals from London Victoria to Selhurst calling at the stations specified in Paragraph 1.1.

.

2.7 Between 1830 and 1900, one additional service shall be provided from London Victoria to Caterham, calling at the stations specified in Paragraph 1.1.

- 2.8 Between 1915 and 2315, nine services shall be provided at half-hourly intervals from London Victoria to Purley, calling at the stations specified in Paragraph 1.1. These services shall be extended to Tattenham Corner as a requirement of services specified in Route Y.
- 2.9 On Mondays to Thursdays and between 0001 and 0030 one additional service shall be provided from London Victoria to East Croydon calling at the stations specified in Paragraph 1.1, this service may omit to call at Battersea Park and Wandsworth Common.
- 2.10 On Fridays, between 2315 and 0045, two additional services shall be provided at half-hourly intervals from London Victoria to Purley, calling at the stations specified in Paragraph 1.1. These services shall be extended to Tattenham Corner as a requirement of services specified in Route Y.
- 2.11 Between 0030 and 0100 one additional service shall be provided from London Victoria to East Croydon calling at stations in Paragraph 1.1.

Between 0745 and 0830, three additional services shall be provided from London Victoria to East Croydon, at intervals between 10 and 20 minutes, calling at Clapham Junction.

#### Towards London Victoria

- 2.12 Between 0700 and 0845, four additional Peak services shall be provided at half-hourly intervals from Caterham to London Victoria, calling at the stations specified in Paragraph 1.1. These services may omit to call at Selhurst, Thornton Heath, Norbury, Streatham Common, Balham, Wandsworth Common and Battersea Park.
- 2.13 Between 0545 and 0900 six additional services shall be provided at half-hourly intervals from East Croydon to London Victoria calling at stations specified in Paragraph 1.1. One of these services may originate from Selhurst.
- 2.14 The Late service may omit to call at Battersea Park.
- 2.15 Between 1830 and 2315, ten additional services shall be provided at half-hourly intervals from Purley to London Victoria, calling at the stations specified in Paragraph 1.1. These services shall originate from Tattenham Corner as a requirement of services specified in Route Y. The first of these services shall depart Purley no earlier than twenty-five minutes later than the Late Service, and may omit to call at Battersea Park.

Between 0045 and 0100, one additional service shall be provided from East Croydon to London Victoria, calling at Selhurst and Clapham Junction.

Between 0530 and 0600, one additional service shall be provided from Purley to London Victoria, calling at the stations specified in Paragraph 1.1. This train shall originate from Redhill and call additionally at Coulsdon South

Between 0630 and 0645, one additional service shall be provided from Caterham to London Victoria, calling at all intermediate stations up to and

including East Croydon and also calling at Clapham Junction.

Between 0630 and 0645, one service shall be provided from Selhurst to London Victoria, calling at West Norwood, Gipsy Hill, Crystal Palace, Sydenham, Forest Hill, Honor Oak Park, Brockley and New Cross Gate.

## 3. Service Pattern – Saturdays

# Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Caterham, calling at the stations specified in Paragraph 1.1.

#### Towards Caterham

- 3.2 Between 2300 and 0045, three services shall be provided at half-hourly intervals from London Victoria to Purley, calling at the stations specified in Paragraph 1.1. These services shall depart London Victoria no less than thirty minutes later than the Late Service, and shall be extended to Tattenham Corner as part of services specified in Route Y..
- 3.3 Between 0030 and 0100, one additional service shall be provided from London Victoria to East Croydon calling at the stations specified in Paragraph 1.1.

#### Towards London Victoria

3.4 Between 0630 and 0730, two services shall be provided at a half-hourly interval from Purley to London Victoria, calling at the stations specified in Paragraph 1.1. The first of these services shall depart Purley no later than one hour before the Early Service. These services shall start back from Tattenham Corner as part of services specified in Route Y.

One additional service shall be provided from East Croydon to London Victoria between 0045 and 0100, calling at Selhurst and Clapham Junction.

# 4 Service Pattern – Sundays

# Core Service Specification

4.1. Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Caterham, calling at the stations specified in Paragraph 1.1. These services may omit to call at Battersea Park, Wandsworth Common, Balham, South Croydon and Purley Oaks.

#### **Additional Services and Variations**

#### **Towards Caterham**

4.2. One additional service shall be provided from London Victoria to Caterham departing no earlier than 2345, calling at the stations specified in Paragraph 1.1. This service may omit to call at Battersea Park.

One additional service shall be provided from London Victoria to East Croydon departing between 0030 and 0045, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from London Victoria to East Croydon departing between 2345 and 0000, calling at Clapham Junction.

One additional service shall be provided from London Victoria to Selhurst departing between 2330 and 2345, calling at the stations specified in Paragraph 1.1.

#### Towards London Victoria

4.3. One additional service shall be provided from Caterham to East Croydon departing no earlier than 2300, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from East Croydon to London Victoria departing between 0045 and 0100, calling at Selhurst and Clapham Junction.

One additional service shall be provided from East Croydon to London Victoria departing between 0630 and 0645, calling at all intermediate stations.

# 5. Limited Stops

5.1 There is no requirement for Limited Stops on this route.

# 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday	Saturday	Sunday	
London Victoria to Caterham	0845	0745	0730	
Caterham to London Victoria	1000	0830	0830	

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday	
London Victoria to Caterham	1630	2230	2300	
Caterham to London Victoria	1745	2245	2230	

Route	Monday – Friday	Saturday	Sunday
London Victoria to Caterham	54 minutes with two exceptions of 57 minutes and one further exception of 1 hour permitted	53 minutes	52 minutes with one exception of 54 minutes and one further exception of 56 minutes permitted
Caterham to London Victoria	52 minutes with one exception of 53 minutes and one further exception of 56 minutes permitted	52 minutes	47 minutes

# Route Q LONDON VICTORIA – SUTTON via Crystal Palace and Norwood Junction

#### 1. Route Definition

- 1.1 Services shall be provided between London Victoria and Sutton, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Hill, West Norwood, Gipsy Hill, Crystal Palace, Norwood Junction, West Croydon, Waddon, Wallington and Carshalton Beeches.
- 1.2 Limited Stops shall be made at Cheam, Ewell East, Epsom, Ashtead and Leatherhead.

# 2. Service Pattern – Mondays to Fridays

# Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Sutton, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

#### Towards Sutton

- 2.2 One additional service shall be provided from London Victoria to West Croydon departing half an hour later that the Late Service.
- 2.3 On Fridays, three additional services at intervals not less than ten minutes shall be provided from London Victoria to West Croydon, the first of which shall depart no less than one hour later than the Late Service.

Between 2000 and 2259 one service per hour from London Victoria to Sutton shall be extended to Epsom, calling additionally at Cheam and Ewell East.

#### Towards London Victoria

- 2.4 Between 0530 and 0700, three services shall be provided at intervals between thirty and forty-five minutes from Norwood Junction and London Victoria calling at the stations specified in Paragraph 1.1. The first service shall originate from East Croydon and may omit to call at Wandsworth Common.
- 2.5 Between 0715 and 0900, four services shall be provided at half hourly intervals from West Croydon to London Victoria calling at the stations specified in Paragraph 1.1.

#### 2.6 NOT USED

One additional service shall be provided from Dorking to London Victoria departing between 1900 and 1915, calling at the stations specified in Paragraph 1.1 and Leatherhead, Ashtead, Epsom, Ewell East, Cheam and Sutton.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Sutton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## **Towards Sutton**

- 3.2 One additional service shall be provided arriving half an hour earlier than the Early Service and may originate from Streatham Hill.
- 3.3 One additional service shall be provided from London Victoria to West Croydon departing no less than half an hour later than the Late Service.
- 3.4 Three additional services shall be provided at intervals not less than ten minutes from London Victoria to West Croydon, the first of these shall depart an hour later that the Late Service.

### Towards London Victoria

3.5 Between 0545 and the Early Service, five additional services shall be provided at half hourly intervals from West Croydon to London Victoria calling at the stations specified in Paragraph 1.1. Three services may originate from Norwood Junction.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and Sutton, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

#### **Towards Sutton**

4.2 Between 2330 and 2359 one additional service shall be provided from London Victoria to Norwood Junction calling at the stations specified in Paragraph 1.1.

## Towards London Victoria

4.3 Between 0645 and 0845, three additional services shall be provided at half hourly intervals from West Croydon to London Victoria calling at the stations specified in Paragraph 1.1. The first service may originate from Norwood Junction.

One additional service shall be provided from London Victoria to west Croydon departing between 2315 and 2330, calling at the stations specified in Paragraph 1.1.

## 5. Limited Stops

- 5.1 **Cheam, Ewell East** and **Epsom** shall be served as follows:
  - (a) The Early Service and the service an hour later from London Victoria as specified in Paragraph 2.1 shall call.
  - (b) Three services between 2000 and the Late Service at hourly intervals from London Victoria as specified in Paragraph 2.1 shall call.
  - (c) Two services between 1700 and 2015 towards London Victoria as specified in Paragraph 2.1 shall call.
  - (d) Three services between 2015 and the Late Service at hourly intervals from Epsom as specified in Paragraph 2.1 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday			
London Victoria to Sutton	0715	0800	0815	
Sutton to London Victoria	1000	0900	0915	

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Su			
London Victoria to Sutton	2300	2300	2245	
Sutton to London Victoria	2300	2300	2300	

Route	Monday - Friday	Saturday	Sunday
London Victoria to Sutton	50 minutes with five exceptions of up to 52 minutes permitted	51 minutes	52 minutes with one exception of 53 minutes permitted
Sutton to London Victoria	51 minutes with three exceptions of 52 minutes and five further exceptions of 54 minutes permitted	51 minutes	47 minutes with one exception of 53 minutes permitted

## Route R LONDON VICTORIA – LONDON BRIDGE via Crystal Palace

## 1. Route Definition

- 1.1. Services shall be provided between London Victoria and London Bridge, calling at Battersea Park, Clapham Junction, Wandsworth Common, Balham, Streatham Hill, West Norwood, Gipsy Hill, Crystal Palace, Sydenham, Forest Hill, Honor Oak Park, Brockley and New Cross Gate.
- 1.2. There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1. Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and London Bridge, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards London Bridge

- 2.2. One morning peak service may omit to call at Battersea Park, and two morning peak services may omit to call at Wandsworth Common.
- 2.3. Between 0630 and 0930 six additional services shall be provided at half hourly intervals from Streatham Hill to London Bridge calling at the stations specified in paragraph 1.1. One interval of up to 40 minutes shall be permitted. Two of these services may start from Crystal Palace. Three of these services need not operate until January 2017 timetable change.

Between 0930 and 0945, one additional service shall be provided from Crystal Palace to London Bridge, calling at the stations specified in Paragraph 1.1.

## Towards London Victoria

- 2.4. One morning Peak service as specified in Paragraph 2.1 may omit to call at Wandsworth Common and Battersea Park.
- 2.5. The interval between the Early Service and the following service as specified in paragraph 2.1 may be extended by 40 minutes.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1. Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Victoria and London Bridge, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## 3.2. NOT USED

Two additional services shall be provided from Crystal Palace to London Victoria departing between 0730 and 0830 at a half hourly interval, calling at Gipsy Hill, West Norwood, Streatham Hill, Balham, Wandsworth Common, Clapham Junction and Battersea Park.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1. On Sundays there is no core service specified between London Victoria and London Bridge.

## 5. Limited Stops

5.1. There is no requirement for Limited Stops on this route.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to London Bridge	0715	0845	N/A
London Bridge to London Victoria	0715	0915	N/A

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Victoria to London Bridge	2345	2345	N/A
London Bridge to London Victoria	2345	2345	N/A

Route	Monday - Friday	Saturday	Sunday
London Victoria to London Bridge	52 minutes with three exceptions of 54 minutes permitted	52 minutes with one exception of 55 minutes and one further exception of 57 minutes permitted	N/A
London Bridge to London Victoria	49 minutes with seven exceptions of 50 minutes, and one further exceptions of 53 minutes permitted	47 minutes with 3 exceptions of 49 minutes permitted	N/A

## **Route S** Not Used

# Route T LONDON BRIDGE – WEST CROYDON via Peckham Rye, Tulse Hill and Selhurst

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and West Croydon, calling at South Bermondsey, Queens Road Peckham, Peckham Rye, East Dulwich, North Dulwich, Tulse Hill, Streatham, Streatham Common, Norbury, Thornton Heath and Selhurst.
- 1.2 Limited Stops shall be made at Streatham Hill, Tooting, Haydons Road, Wimbledon and East Croydon.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and West Croydon at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

**Additional Services and Variations** 

Towards West Croydon

- 2.2 Two services as specified in Paragraph 2.1 departing London Bridge before 1000 may omit to call at South Bermondsey and Queens Road Peckham
- 2.3 Six additional services shall be provided from London Bridge to Streatham departing between 0545 and 0830 at intervals between twenty and forty minutes, calling at the stations specified in Paragraph 1.1. One of these services shall be extended to Sutton and Epsom as a service requirement of Route L (London Victoria to Epsom via Hackbridge).
- 2.4 Between 1600 and 1700 two additional services shall be provided from London Bridge to Tulse Hill departing at a half hourly interval calling at stations specified in paragraph 1.1.
- 2.5 On Fridays two additional services shall be provided from London Bridge to West Croydon at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The first service shall depart half an hour later than the Late Service.

Between 0900 and 1000, three additional services shall be provided from London Bridger to Tulse Hill, at half hourly intervals, calling at all intermediate stations.

Between 1620 and 1650, two additional services shall be provided from London Bridge to Selhurst at a half hourly interval, calling at all intermediate stations.

## Towards London Bridge

- 2.6 One additional service shall be provided from Streatham to London Bridge departing between 0545 and 0600, calling at Tulse Hill and Peckham Rye.
- 2.7 One service arriving at London Bridge between 0900 and 0930 may omit to call at Queens Road Peckham and South Bermondsey.
- 2.8 One additional service shall be provided from Tulse Hill to London Bridge between 1545 and 1615, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from Streatham Hill.
- 2.9 Three additional services shall be provided from Streatham to London Bridge departing between 1800 and 1930 at intervals between thirty and forty minutes, calling at the stations specified in Paragraph 1.1.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and West Croydon, calling at the stations specified in Paragraph 1.1.

## **Additional Services and Variations**

3.2 There are no variations to the core service between London Bridge and West Croydon specified in Paragraph 3.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and **East Croydon** at half-hourly intervals, calling at the stations specified in Paragraph 1.1 these services shall omit to call at West Croydon.

## Additional Services and Variations

## Towards East Croydon

#### 4.2 NOT USED

Two additional services shall be provided from London Bridge to Norwood Junction departing between 0000 and 0030 at a half hourly interval, calling at South Bermondsey, Queens Road Peckham, Peckham Rye, East Dulwich, North Dulwich, Tulse Hill, West Norwood, Gipsy Hill and Crystal Palace.

One additional service shall be provided from London Bridge to West Croydon departing between 0015 and 0030, calling at the stations specified in Paragraph 1.1.

## Towards London Bridge

4.3 The Early Service as specified in Paragraph 4.1 may originate from Selhurst.

## 5. Limited Stops

- 5.1 **Streatham Hill** shall be served as follows:
  - (a) The service as specified in Paragraph 2.8 shall originate here.
- 5.2 **Tooting, Haydons Road** and **Wimbledon** shall be served as follows;
  - (a) Five services as specified in Paragraph 2.3 shall call.
  - (b) The three services as specified in Paragraph 2.9 shall call.

## 5.3 **East Croydon** shall be served as follows;

(a) The services as specified in Paragraph 4.1 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday1
London Bridge to West Croydon	0700	0730	0815
West Croydon to London Bridge	0715	0745	0730

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to West Croydon	2315	2359	2315
West Croydon to London Bridge	2245	2245	2230

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between London Bridge and East Croydon.

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to West Croydon	39 minutes with eight exceptions of 42 minutes permitted and two further exceptions of 46 minutes permitted	39 minutes	39 minutes with one exception of 41 minutes permitted
West Croydon to London Bridge	39 minutes with three exception of 42 minutes and two further exceptions of 44 minutes <i>permitted</i>	38 minutes	37 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between London Bridge and East Croydon.

## Route U LONDON BRIDGE – BECKENHAM JUNCTION via Tulse Hill and Crystal Palace

## 1. Route Definition

1.1 Services shall be provided between London Bridge and Beckenham Junction, calling at South Bermondsey, Queens Road Peckham, Peckham Rye, East Dulwich, North Dulwich, Tulse Hill, West Norwood, Gipsy Hill, Crystal Palace and Birkbeck.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Beckenham Junction, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Beckenham Junction

- 2.2 Two intervals between the Early and Late Services as specified in Paragraph 2.1 may be extended up to 45 minutes.
- 2.3 Not Used
- 2.4 The penultimate and Late Service from London Bridge as specified in Paragraph 2.1 may terminate at Crystal Palace.
- 2.5 One additional service shall be provided from London Bridge to Crystal Palace departing between 2345 and 0005, calling at the stations specified in Paragraph 1.1. This service shall be extended to terminate at Norwood Junction.
- 2.6 On Fridays, one additional service shall be provided from London Bridge to Crystal Palace, departing no earlier than 0015, calling at the stations specified in Paragraph 1.1.

## Towards London Bridge

- 2.7 Between 0545 and 0715 at an interval not less than forty-five minutes, two services shall be provided from Crystal Palace to London Bridge calling at the stations specified in Paragraph 1.1. These services shall be extended to originate from Norwood Junction. The first service may be further extended to originate from East Croydon.
- 2.8 Three intervals between the Early and Late Services as specified in Paragraph 2.1 may be extended up to forty minutes.

Between 1630 and 1730, two additional services shall be provided from Tulse Hill to London Bridge, at a half hourly interval, calling at all intermediate stations.

Between 1720 and 1750, two additional services shall be provided from Norwood Junction to London Bridge via Crystal Palace, at a half hourly interval, calling at all intermediate stations.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and Beckenham Junction, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Beckenham Junction

- 3.2 The penultimate and Late Service from London Bridge as specified in Paragraph 3.1 may terminate at Crystal Palace.
- 3.3 Two additional services shall be provided from London Bridge to Crystal Palace at a half-hourly interval departing no earlier than 2345, calling at the stations specified in Paragraph 1.1.

## Towards London Bridge

3.4 One additional service shall be provided from Crystal Palace to London Bridge departing no later than 0700, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from Norwood Junction.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and Crystal Palace, calling at the stations specified in Paragraph 1.1

## Additional Services and Variations

4.2 There are no variations to the core services between London Bridge and Crystal Palace as specified in Paragraph 4.1

## 5. Limited Stops

There is no requirement for Limited Stops on this route.

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunday <sup>1</sup>			
London Bridge to Beckenham Junction	0700	0815	0815	
Beckenham Junction to London Bridge	0800	0900	0745	

	Late Service departs no earlier than:		
Route	Monday - Friday Saturday Sunday <sup>1</sup>		
London Bridge to Beckenham Junction	2315	2315	2300
Beckenham Junction to London Bridge	2310	2310	2330

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Beckenham Junction	36 minutes with two exceptions of 39 minutes permitted	36 minutes	26 minutes with one exception of 28 minutes permitted
Beckenham Junction to London Bridge	35 minutes with five exceptions of 40 minutes	35 minutes with	26 minutes

<sup>&</sup>lt;sup>1</sup>Sunday services are specified between London Bridge and Crystal Palace

## Route V1 LONDON BRIDGE – NORWOOD JUNCTION Stopping Services

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Norwood Junction, calling at New Cross Gate, Brockley, Honor Oak Park, Forest Hill, Sydenham, Penge West and Anerley.
- 1.2 Limited Stops shall be made at West Croydon, Waddon, Wallington, Carshalton Beeches, Sutton, Cheam, Ewell East and Sutton.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and Norwood Junction at half-hourly intervals, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

### **Towards Norwood Junction**

2.2 One additional service shall be provided from London Bridge to Norwood Junction departing between 0530 and 0545, calling at the stations specified in Paragraph 1.1. This service shall be extended to terminate at Sutton.

## Towards London Bridge

2.3 One additional service shall be provided from Norwood Junction to London Bridge departing between 0530 and 0605, calling at the stations specified in Paragraph 1.1. This service shall be extended to originate from Epsom and call additionally at Ewell East, and Cheam.

Between, 1800 and 1815, one additional service shall be provided from Norwood Junction to London Bridge, calling at all intermediate stations.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Four services shall be provided from Norwood Junction to London Bridge departing between 0630 and 0800, calling at the stations specified in Paragraph 1.1. Two of these services shall be extended to originate from Sutton and one of the services shall be extended to Originate from East Croydon.

Four services shall be provided from London Bridge to Norwood Junction via Peckham Rye and West Norwood, departing between 2300 and 0033 at half hourly intervals, calling at all intermediate stations.

## 4. Service Pattern – Sundays

## Core Service Specification

- 4.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and West Croydon at half-hourly intervals, calling at the stations specified in Paragraph 1.1.
- 4.2 The services as specified in Paragraph 4.1 may omit to call at Penge West and Anerley.

#### Additional Services and Variations

4.3 There are no variations to the core service between London Bridge and West Croydon as specified in Paragraph 4.1.

## 5. Limited Stops

- 5.1 **West Croydon** shall be served as follows:
  - (a) On Mondays to Fridays, the services from London Bridge as specified in Paragraph 2.1 shall call.
  - (b) On Mondays to Fridays, the services as specified in Paragraphs 2.2 and 2.3 shall call.
  - (c) On Saturdays, the services as specified in Paragraph 3.1 shall call.
  - (d) On Sundays, the services as specified in Paragraph 4.1 shall call.
- 5.2 Waddon, Wallington, Carshalton Beeches, Sutton, Cheam, Ewell East and Sutton shall be served as follows:
  - (a) On Mondays to Fridays, the services as specified in Paragraphs 2.2 and 2.3 shall call..
  - (b) On Saturdays, the services as specified in Paragraph 3.1 shall call.

## 6. Early and Late Services

	Early Service arrives at destination no later than:				
Route	Monday - Friday Saturday Sunday <sup>1</sup>				
London Bridge to Norwood Junction	1715	N/A	1230		
Norwood Junction to London Bridge	0700	N/A	1245		

	Late Service departs no earlier than:				
Route	Monday - Friday Saturday Sunday <sup>1</sup>				
London Bridge to Norwood Junction	1830	N/A	2145		
Norwood Junction to London Bridge	0900	N/A	2230		

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between London Bridge and West Croydon.

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
London Bridge to Norwood Junction	23 minutes with one exception of 24 minutes permitted	N/A	30 minutes
Norwood Junction to London Bridge	27 minutes with two exceptions of 31 minutes permitted	33 minutes	27 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between London Bridge and West Croydon.

## Route V2 LONDON BRIDGE – EPSOM via Norwood Junction and West Croydon

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Epsom, calling at Norwood Junction, West Croydon, Waddon, Wallington, Carshalton Beeches, Sutton, Cheam and Ewell East.
- 1.2 Limited Stops shall be made at Ashtead, Leatherhead, Bookham, Effingham Junction, Horsley, Clandon, London Road (Guildford), Guildford.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided between London Bridge and Epsom at half-hourly intervals, calling at the stations specified in Paragraph 1.1. These services need not operate until January 2017.

## Additional Services and Variations

## Towards Epsom

- 2.2 Between 0500 and 0600 one additional service shall be provided from West Croydon to Sutton, calling at the stations specified in Paragraph 1.1.
- 2.3 One service as specified in Paragraph 2.1 departing London Bridge between the Early Service and 1845 shall be extended to terminate at Guildford, calling at Ashtead, Leatherhead, Bookham, Effingham Junction, Horsley, Clandon and London Road (Guildford).
- 2.4 Two services as specified in Paragraph 2.1 departing London Bridge between the Early Service and the Late Service shall be extended to Dorking, calling at Ashtead, Leatherhead and Boxhill & Westhumble.
- 2.5 One additional service from London Bridge to Epsom departing between 1920 and 1940 shall be provided, calling at the stations specified in Paragraph 1.1. This service shall be extended to terminate at Dorking calling at Leatherhead only.
- 2.6 One additional service from Sutton to Epsom departing between 0630 and 0645 shall be provided, calling at the stations specified in Paragraph 1.1. This service may omit to call at Cheam, Ewell East, and shall be extended to terminate at Guildford.
- 2.7 One additional service from Sutton to Epsom departing between 0720 and 0740 shall be provided, extended to terminate at Guildford. This service shall originate from London Victoria as a requirement of services specified in Route Q (London Victoria Sutton via Crystal Palace and West Croydon).

2.8 One additional service from Sutton to Epsom departing between 1720 and 1740 shall be provided, calling at the stations specified in Paragraph 1.1. This service shall be extended to terminate at Guildford. This service shall originate from London Victoria as part of services specified in Route Q (London Victoria – Sutton via Crystal Palace and West Croydon).

## Towards London Bridge

- 2.9 The Early Service as specified in Paragraph 2.1 may originate from Sutton.
- 2.10 Two additional services shall be provided from Sutton to London Bridge departing between 0530 and 0615 at a half-hourly interval, calling at the stations specified in Paragraph 1.1. The second service shall arrive half an hour earlier than the Early Service. The first service shall be extended to originate from Epsom and shall call at Ewell East and Cheam.
- 2.11 One service as specified in Paragraph 2.1 departing Epsom no later than 0800 may omit to call at Waddon. This service shall originate from Guildford.
- 2.12 One additional service from Guildford to Epsom departing between 0810 and 0825 shall be provided, this service shall continue to London Victoria as a requirement of services specified in Route L (London Victoria Epsom via Hackbridge).
- 2.13 Three additional services shall be provided from West Croydon to London Bridge at half-hourly intervals between 1700 and 1830, calling at the stations specified in Paragraph 1.1. These services may omit to call at Norwood Junction.
- 2.14 Two additional services shall be provided from Epsom to West Croydon departing between 1815 and 1945, calling at stations specified in Paragraph 1.1 The services shall originate from Guildford and one service may omit to call at Ewell East and Cheam. The first of these services shall be extended to terminate at London Victoria via Norwood Junction and Crystal Palace, as a requirement of Route Q. The second of these services may terminate at Sutton.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 There is no requirement to provide services between London Bridge and Epsom on Saturdays.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 There is no requirement to provide services between London Bridge and Epsom on Sundays.

## 5. Limited Stops

- 5.1 Ashtead, Leatherhead, Bookham, Effingham Junction, Horsley, Clandon, London Road (Guildford) and Guildford shall be served as follows:
  - (a) The service as specified in Paragraph 2.5 shall call.
  - (b) The service as specified in Paragraph 2.8 shall call at Leatherhead, Bookham, Effingham Junction and Guildford.
  - (c) The service as specified in Paragraph 2.9 shall call and may omit to call at Horsley and Clandon.
  - (d) The service as specified in Paragraph 2.10 shall.
  - (e) The service as specified in Paragraph 2.13 calling at London Road (Guildford), Effingham Junction, Bookham, Leatherhead and Ashtead
  - (f) The service as specified in Paragraph 2.14 shall call.
  - (g) The service specified in Paragraph 2.14 shall call. One service may omit to call at Ewell East and Cheam

## 6. Early and Late Services

	Early Service arrives at destination no later than:			
Route	Monday - Friday Saturday Sunda			
London Bridge to Epsom	1800	N/A	N/A	
Epsom to London Bridge	0715	N/A	N/A	

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sunday			
London Bridge to Epsom	1830	N/A	N/A	
Epsom to London Bridge	0815	N/A	N/A	

Route	Monday - Friday	Saturday	Sunday
London Bridge to Epsom	43 minutes with the one exception of 52 minutes permitted	N/A	N/A
Epsom to London Bridge	44 minutes with one exception of 51 minutes permitted	N/A	N/A

## Route W LONDON BRIDGE – SUTTON and WIMBLEDON via Tulse Hill and Hackbridg

e

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Wimbledon, calling at South Bermondsey, Queens Road Peckham, Peckham Rye, East Dulwich, North Dulwich, Tulse Hill, Streatham, Mitcham Junction, Hackbridge, Carshalton, Sutton, West Sutton, Sutton Common, St Helier, Morden South, South Merton and Wimbledon Chase.
- 1.2 Limited Stops may be made at Mitcham Eastfields.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Services shall be provided between London Bridge and Wimbledon, calling at the stations specified in Paragraph 1.1, as follows:

## Towards Wimbledon (via Sutton)

2.2 Four evening Peak services shall be provided from London Bridge to Wimbledon at not less than half-hourly intervals, calling at the stations specified in Paragraph 1.1. One service may be provided to Sutton, and extended from Sutton to **Epsom**, calling at Cheam, Ewell East and Epsom.

## Towards London Bridge (via Sutton)

2.3 Six morning Peak services shall be provided from Wimbledon to London Bridge, at intervals between 30 and 40 minutes calling at the stations specified in Paragraph 1.1. The first service may arrive London Bridge before 0700

## 3. Service Pattern – Saturdays

3.1 There is no requirement to provide services between London Bridge and Wimbledon on Route W on Saturdays.

## 4. Service Pattern – Sundays

4.1 There is no requirement to provide services between London Bridge and Wimbledon on Route W on Sundays.

## 5. Limited Stops

5.1 Mitcham Eastfields may be served by the services specified in Paragraph 2.

## 6. Early and Late Services

The detailed specification of the Route W service defines Early Service and Late Service without tables.

Route	Monday - Friday	Saturday	Sunday
London Bridge to Wimbledon	1 hour	N/A	N/A
Wimbledon to London Bridge	1 hour 1 minute	N/A	N/A

## Route X LONDON BRIDGE – CATERHAM via Norwood Junction and East Croydon

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Caterham, calling at New Cross Gate, Brockley, Honor Oak Park, Forest Hill, Sydenham, Penge West, Anerley, Norwood Junction, East Croydon, South Croydon, Purley Oaks, Purley, Kenley, Whyteleafe and Whyteleafe South.
- 1.2 There is no requirement for Limited Stops on this route.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Caterham, calling at the stations specified in Paragraph 1.1.

## **Additional Services and Variations**

## Towards Caterham

- 2.2 Between 0645 and 0700, one additional service shall be provided from London Bridge to Caterham, calling at the stations specified in Paragraph 1.1.
- 2.3 Between the Late Service and 1850, five additional services shall be provided from London Bridge to Caterham at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The first of these services shall depart London Bridge no later than fifty minutes later than the Late Service. These services shall omit to call at New Cross Gate, Brockley, Honor Oak Park, Forest Hill, Sydenham, Penge West and Anerley.
- 2.4 Between 1900 and 0045, twelve additional services shall be provided from London Bridge to Caterham at half-hourly intervals, calling at the stations specified in Paragraph 1.1. The last of these services shall depart no earlier than 0015, and may terminate at East Croydon.

## Towards London Bridge

- 2.5 Between 0500 and 0515, one additional service shall be provided from Purley to London Bridge, calling at the stations specified in Paragraph 1.1.
- 2.6 Between 0545 and 0600, one additional service shall be provided from Caterham to London Bridge, calling at the stations specified in Paragraph 1.1.

- 2.7 Between 0615 and 0845, six additional services at half-hourly intervals shall be provided from Caterham to London Bridge, calling at the stations specified in Paragraph 1.1. These services may omit to call at Anerley, Penge West, Sydenham, Forest Hill, Honor Oak Park, Brockley and New Cross Gate. These services shall be extended from Purley to London Bridge as a portion of a service requirement in Route Y (London Bridge to Tattenham Corner via Norwood Junction).
- 2.8 Between 1615 and 1715 services departing Caterham as specified in paragraph 2.1 may terminate at New Cross Gate.
- 2.9 Between 1700 and 1815 of services as specified in Paragraph 2.1 there may be one interval of fifty minutes and one service from Caterham in this period may omit to call at Anerley, Penge West, Sydenham, Forest Hill, Honor Oak Park, Brockley and New Cross Gate.
- 2.10 Between 1715 and 1830 one additional service shall be provided from Norwood Junction to London Bridge, calling at the stations specified in Paragraph 1.1.
- 2.11 Between 2300 and 2330, one additional service shall be provided from Caterham to Norwood Junction, calling at the stations specified in Paragraph1.1. This service shall depart Caterham no earlier than thirty minutes later than the Late Service specified in Paragraph 2.1.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Caterham, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Caterham

#### 3.2 NOT USED

One additional service shall be provided from London Bridge to East Croydon departing no earlier than 0015, calling at the stations specified in Paragraph 1.1.

## Towards London Bridge

3.3 Between 2300 and 2330, one additional service shall be provided from Caterham to Norwood Junction, calling at the stations specified in Paragraph 1.1. This service shall depart Caterham no earlier than forty minutes later than the Late Service as specified in Paragraph 3.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 On Sundays there is no core service specified between London Bridge and Caterham. Sunday services to/from Caterham shall be provided as a requirement in Route P (London Victoria to Caterham).

One additional services shall be provided from London Bridge to Caterham departing between 2330 and 2359 at a 25-minute interval, calling at the stations specified in Paragraph 1.1.

One additional service shall be provided from Caterham to London Bridge departing between 0715 and 0730, calling at the stations specified in Paragraph 1.1.

## 5. Limited Stops

There is no requirement for Limited Stops on this route.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday – Friday	Saturday	Sunday
London Bridge to Caterham	0700	0730	N/A
Caterham to London Bridge	1015	0700	N/A

	Late Service departs no earlier than:			
Route	Monday - Friday Saturday Sunda			
London Bridge to Caterham	1600	0000	N/A	
Caterham to London Bridge	2230	2230	N/A	

Route	Monday - Friday	Saturday	Sunday
London Bridge to Caterham	53 minutes with five exceptions of 56 minutes permitted	50 minutes with one exception of 52 minutes and a further exception of 56 minutes permitted	N/A
Caterham to	52 minutes with one	51 minutes with one	N/A

London Bridge	exception of 55 minutes	exception of 52 minutes	
	permitted	permitted	

## Route Y LONDON BRIDGE – TATTENHAM CORNER via Norwood Junction and East Croydon (semi-fast)

## 1. Route Definition

- 1.1 Services shall be provided between London Bridge and Tattenham Corner, calling at Norwood Junction, East Croydon, Purley, Reedham, Smitham, Woodmansterne, Chipstead, Kingswood and Tadworth.
- 1.2 Limited stops shall be made at South Croydon and Purley Oaks.

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Tattenham Corner, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Tattenham Corner

- 2.2 The Late Service as specified in Paragraph 2.1 may be formed from a portion detached at Purley from a service as a requirement of Route X (London Bridge to Caterham via Norwood Junction and East Croydon).
- 2.3 Between 0645 and 0815, two additional services shall be provided at an hourly interval from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services shall originate from London Victoria between 0600 and 0730 as a requirement of Route P (London Victoria to Caterham via Selhurst).
- 2.4 Between 0800 and 0900, two additional services shall be provided at a half-hourly interval from London Bridge to Tattenham Corner, calling at the stations specified in Paragraph 1.1. One of these services may omit to call at Norwood Junction.
- 2.5 Between 0730 and 0830, two additional services shall be provided at an interval between thirty minutes and forty-five minutes from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services may be formed of a portion detached from a service from London Bridge provided in Route X.
- 2.6 Between 1000 and 1615, seven additional services shall be provided at hourly intervals from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1.
- 2.7 Between 1730 and 1930, four additional services shall be provided from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services shall be formed by a portion detached from services originating at London Victoria as a requirement in Route P.

- 2.8 Between 1945 and 2359, nine additional services shall be provided at half-hourly intervals from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services shall originate from London Victoria as a requirement of services specified in Route P.
- 2.9 On Fridays, between 2359 and 0100, two additional services shall be provided at half-hourly intervals from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services shall originate from London Victoria as a requirement of services specified in Route P.

Between 0715 and 0745, two additional services shall be provided from London Bridge to East Croydon, calling at the stations specified in Paragraph 1.1. One of these services shall call additionally at New Cross Gate.

Between 2030 and 2045, one additional service shall be provided from London Bridge to East Croydon, calling at the stations specified in Paragraph 1.1.

Between 2100 and 2300, three additional services shall be provided from London Bridge to East Croydon, at an hourly interval, calling at Norwood Junction.

Between 2330 and 0100, three additional services shall be provided from London Bridge to East Croydon, at half hourly intervals. These services may omit to call at Norwood Junction.

## Towards London Bridge

- 2.10 Between 0530 and 0615, one additional service shall be provided from Tattenham Corner to London Bridge, calling at the stations specified in Paragraph 1.1. This service shall call additionally at Purley Oaks and South Croydon.
- 2.11 Between 0630 and 0830, five additional services shall be provided at half-hourly intervals from Tattenham Corner to London Bridge. These services may be attached at Purley to services to London Bridge as a requirement of Route X.
- 2.12 Between 0640 and 0825, four additional services shall be provided at half-hourly intervals from Tattenham Corner to Purley, calling at the stations specified in Paragraph 1.1. These services shall be extended to terminate at London Victoria, this may be by attachment at Purley, as a requirement of Route P.
- 2.13 Between 0930 and 1635, eight additional services shall be provided at hourly intervals from Tattenham Corner to Purley, calling at the stations specified in Paragraph 1.1. The last of these services shall be extended to London Victoria as a requirement of Route P.
- 2.14 Between 1800 and 2250, ten additional services shall be provided at half-hourly intervals from Tattenham Corner to Purley, calling at the stations specified in Paragraph 1.1. These services shall be extended to London Victoria as a requirement of services specified in Route P.

2.15 Between 2300 and 0100, four additional services shall be provided between East Croydon and London Bridge, at half hourly intervals. These services may omit to call at Norwood Junction.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, two services per hour shall be provided at half-hourly intervals between London Bridge and Tattenham Corner, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Tattenham Corner

- 3.2 One additional service shall be provided from Norwood Junction to Tattenham Corner, calling at the stations specified in Paragraph 1.1. This service shall depart East Croydon no less than thirty minutes earlier than the Early Service.
- 3.3 Between 0800 and 2215, fifteen additional services shall be provided at hourly intervals from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1.
- 3.4 Between the Late Service and 2359, one additional services shall be provided from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. This service shall originate from London Victoria as a requirement of services specified in Route P.
- 3.5 Between 2359 and 0100, two additional services shall be provided at half-hourly intervals from Purley to Tattenham Corner, calling at the stations specified in Paragraph 1.1. These services shall originate from London Victoria as a requirement of services specified in Route P.

## Towards London Bridge

- 3.6 Between 0600 and 0700, two services shall be provided at half-hourly intervals from Tattenham Corner to Purley, calling at the stations specified in paragraph 1.1. These services shall be extended to London Victoria as a requirement of Route P.
- 3.7 Between 0830 and 2245, fifteen additional services shall be provided at hourly intervals from Tattenham Corner to Purley, calling at the stations specified in Paragraph 1.1.

## 4. Service Pattern – Sundays

## Core Service Specification

4.1 Between and including the Early and Late Services, a half hourly service shall be provided between London Bridge and Tattenham Corner, calling at the stations specified in Paragraph 1.1.

#### Additional Services and Variations

## Towards Tattenham Corner

4.2 Two additional services shall be provided at a 90-minute interval from London Bridge to East Croydon the first departing no earlier than 2300, calling at the stations as specified in Paragraph 1.1.

## Towards London Bridge

- 4.3 Between 0630 and the Early Service, two additional services shall be provided at half hourly intervals from Purley to London Bridge, calling at the stations specified in Paragraph 1.1. The last of these shall depart Purley no later than 30 minutes before the Early Service.
- 4.4 Between 2230 and 2300, one additional service shall be provided from Tattenham Corner to Norwood Junction, calling at the stations specified in Paragraph 1.1. This service shall depart Tattenham Corner no earlier than 30 minutes after the Late Service.

## 5. Limited Stops

## 5.1 New Cross Gate

- (a) On Mondays- Fridays, the Early Service as specified in Paragraphs 2.1 shall call.
- (b) On Sundays, the services specified in Paragraph 4.1, 4.2 and 4.3 shall call.

## 5.2 Brockley, Honor Oak Park Forest Hill and Sydenham

(a) On Sundays, the services specified in Paragraph 4.1, 4.2 and 4.3 shall call.

## 5.3 **Penge West** and **Anerley**

(a) On Sundays, the services in paragraphs 4.1, 4.2 and 4.3 shall call.

## 5.4 South Croydon and Purley Oaks

(a) On Mondays – Fridays the services specified in Paragraphs 2.12 and 2.13 shall call.

(b) On Sundays, the services specified in Paragraph 4.1, 4.3 and 4.4 shall call.

## **6.** Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday
London Bridge to Tattenham Corner	1015	0745	0845
Tattenham Corner to London Bridge	0945	0815	0900

	Late Service departs no earlier than:		
Route	Monday - Friday	Saturday	Sunday
London Bridge to Tattenham Corner	1845	2250	2245
Tattenham Corner to London Bridge	1745	2250	2215

Route	Monday - Friday	Saturday	Sunday
London Bridge to Tattenham Corner	49 minutes with three exceptions of 52 minutes and two further exceptions of 1h 7 minutes permitted	49 minutes	1 hour 3 minutes with three exceptions of 1 hour 8 minutes permitted
Tattenham Corner to London Bridge	50 minutes, with five exceptions of 56 minutes and one further exception of 58 minutes permitted	48 minutes	1 hour 1 minute with two exceptions of 1 hour 3 minutes permitted

## Route Z EAST CROYDON – MILTON KEYNES CENTRAL via Selhurst, Kensington Olympia and Watford Junction

## 1. Route Definition

1.1 Services shall be provided between East Croydon and Milton Keynes Central, calling at Selhurst, Thornton Heath, Norbury, Streatham Common, Balham, Wandsworth Common, Clapham Junction, Imperial Wharf, West Brompton, Kensington Olympia, Shepherds Bush, Wembley Central, Harrow & Wealdstone, Watford Junction, Hemel Hempstead, Berkhamsted, Tring, Leighton Buzzard and Bletchley.

## 1.2 NOT USED

## 2. Service Pattern – Mondays to Fridays

## Core Service Specification

2.1 Between and including the Early and Late Services, one service per hour shall be provided at hourly intervals between East Croydon and Milton Keynes Central, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Milton Keynes Central

- 2.2 Services as specified in Paragraph 2.1 including the Early Service and up to 0915 arriving at Milton Keynes Central may originate from Clapham Junction, with a Maximum Journey Time of 1 hour 26 minutes.
- 2.3 Two services as specified in Paragraph 2.1 arriving between 1715 and 1915 at Milton Keynes Central may omit to call at Tring and Bletchley.
- 2.4 Two additional services shall be provided at an interval not less than 50 minutes from Clapham Junction to Watford Junction, calling at the stations specified in Paragraph 1.1, departing before 0600. One service may omit to call at Wembley Central.
  - Between 0615 and 0630, one additional service shall be provided from Clapham Junction to Watford Junction, calling at the stations specified in Paragraph 1.1. This service may omit calling at Harrow and Wealdstone.
- 2.5 One additional service shall be provided to Milton Keynes Central arriving before 0700, calling at the stations specified in Paragraph 1.1. This service may originate from Balham, and may omit to call at Wandsworth Common.
- 2.6 One additional service shall be provided from East Croydon to Watford Junction, calling at the stations specified in Paragraph 1.1, arriving no later than 0915.. This train shall commence from Coulsdon Town valling at Purley, Purley Oaks and South Croydon.

Between 0800 and 0815 one additional service shall be provided between Clapham Junction and Shepherds Bush calling at the stations specified in Paragraph 1.1.

- 2.7 One additional service shall be provided from East Croydon to Shepherds Bush arriving no later than 0930, calling at the stations specified in Paragraph 1.1.
- 2.8 Two additional services shall be provided at an interval not less than 30 minutes from Clapham Junction to Shepherd's Bush, calling at the stations specified in Paragraph 1.1, arriving no later than 0959.
- 2.9 One additional service shall be provided from Clapham Junction to Shepherd's Bush, calling at the stations specified in Paragraph 1.1, departing between 1700 and 1730.
- 2.10 One additional service shall be provided from Clapham Junction to Milton Keynes Central, calling at the stations specified in Paragraph 1.1, departing one hour later than the Late Service.
- 2.11 Two additional services shall be provided at an hourly interval from Clapham Junction to Watford Junction, calling at the stations specified in Paragraph 1.1, departing no earlier than 2130. These services may omit to call at Wembley Central.

Between 1645 and 1700, one additional service shall be provided from Clapham Junction to Kensington Olympia, calling at all intermediate stations.

## Towards East Croydon

- 2.12 One service as specified in Paragraph 2.1 departing Milton Keynes Central no earlier than 1800 may terminate at Selhurst.
- 2.13 Services as specified in Paragraph 2.1 departing Milton Keynes Central between 1900 and the Late Service may terminate at Clapham Junction, with a Maximum Journey Time of 1 hour 21 minutes. The penultimate and the Late Service may omit to call at Wembley Central.
- 2.14 Two additional services shall be provided at an hourly interval from Watford Junction to Clapham Junction, calling at the stations specified in Paragraph 1.1, arriving before 0740.

Between 0700 and 0730, one additional service shall be provided from Watford Junction to Clapham Junction, calling at all intermediate stations.

Between 0900 and 0930, one additional service shall be provided from Watford Junction to Clapham Junction, calling at all intermediate stations.

Between 1630 and 1705, two additional services shall be provided from Kensington Olympia to Clapham Junction, at a half hourly interval, calling at all intermediate stations.

- 2.15 One additional service shall be provided from Milton Keynes Central to East Croydon, calling at the stations specified in Paragraph 1.1, departing before 0715.
- 2.16 Three additional services shall be provided from Shepherd's Bush to Clapham Junction, calling at the stations specified in Paragraph 1.1, departing between 0815 and 0945 at intervals between 25 minutes and 40 minutes.
- 2.17 One additional service shall be provided from Shepherd's Bush to East Croydon, calling at the stations specified in Paragraph 1.1, departing between 1600 and 1859.
- 2.18 Two additional services shall be provided at an hourly interval from Watford Junction to Clapham Junction, calling at the stations specified in Paragraph 1.1, departing no earlier than 2215. These services may omit to call at Wembley Central.
- 2.19 The Late Service from Milton Keynes Central as specified in Paragraph 2.1 may terminate at Selhurst and may omit to call at Wembley Central and Wandsworth Common.

Between 0730 and 0745, one additional service shall be provided from Coulsdon Town to Watford Junction, calling at all intermediate stations.

## 3. Service Pattern – Saturdays

## Core Service Specification

3.1 Between and including the Early and Late Services, one service per hour shall be provided between East Croydon and Milton Keynes Central, calling at the stations specified in Paragraph 1.1.

## Additional Services and Variations

## Towards Milton Keynes Central

- 3.2 The Late Service as specified in Paragraph 3.1 may omit to call at Wembley Central.
- 3.3 The penultimate service and the Late Service as specified in Paragraph 3.1 may terminate at Watford Junction.
- 3.4 Two additional services shall be provided at an hourly interval from Clapham Junction to Watford Junction, calling at the stations specified in Paragraph 1.1, arriving before 0700. One service may omit to call at Wembley Central.
- 3.5 One additional service shall be provided to Milton Keynes Central an hour earlier than the Early Service, this service may originate from Balham, calling at the stations specified in Paragraph 1.1. This service may omit to call at Wandsworth Common.

3.6 Three additional services shall be provided at intervals not less than an hour from Clapham Junction to Watford Junction, calling at the stations specified in Paragraph 1.1, the third service departing no earlier than 2230. These services may omit to call at Wembley Central

All services except the Early Service as specified in Paragraph 3.1 shall be extended to depart from South Croydon.

## Towards East Croydon

- 3.7 Two additional services shall be provided at an hourly interval from Watford Junction to East Croydon, calling at the stations specified in Paragraph 1.1, arriving before 0800. One service may omit to call at Wembley Central.
- 3.8 One additional service shall be provided between 1900 and 1945 from Watford Junction to Clapham Junction, calling at the stations specified in Paragraph 1.1.
- 3.9 The Late Service as specified in Paragraph 3.1 may omit to call at Wembley Central.
- 3.10 Four additional services shall be provided at intervals not less than 35 minutes from Watford Junction to Clapham Junction, calling at the stations specified in Paragraph 1.1. The fourth service departing no earlier than 2315. These services may omit to call at Wembley Central.

All services except the last two as specified in Paragraph 3.1 shall be extended to terminate at South Croydon.

## 4. Service Pattern – Sundays

## Core Service Specification

- 4.1 Between and including the Early and Late Services, one service per hour shall be provided between Clapham Junction and Watford Junction, calling at the stations specified in Paragraph 1.1.
- 4.2 The services as specified in Paragraph 4.1 shall omit to call at Wembley Central.

## Additional Services and Variations

## **Towards Watford Junction**

4.3 One additional service shall be provided from Clapham Junction to Shepherd's Bush not less than 50 minutes earlier than the Early Service, calling at the stations specified in Paragraph 1.1.

## Towards Clapham Junction

4.4 Two additional services shall be provided at an hourly interval from Shepherd's Bush to Clapham Junction, arriving no later than 0915.

One additional service shall e provided from Watford Junction to East Croydon departing between 2315 and 2330, calling at Harrow & Wealdstone, Shepherds Bush, Kensington Olympia, West Brompton, Imperial Wharf, Clapham Junction and Selhurst.

## 5. Limited Stops

- 5.1 **South Croydon** shall be served as follows;
  - (a) Services specified in Paragraph 2.1 towards Milton Keynes Central departing East Croydon between 0930 and 1730 shall originate from here.
  - (b) Services specified in Paragraph 2.1 towards East Croydon departing Milton Keynes between 0800 and 1530 shall terminate here.
  - (c) NOT USED.
  - (d) Services specified in Paragraph 3.1 towards Milton Keynes Central departing East Croydon between 0700 and 1830 shall originate from here.
  - (e) Services specified in Paragraph 3.1 towards East Croydon departing Milton Keynes between 0700 and 1630 shall terminate here.
  - (f) Service specified in Paragraph 3.7 shall terminate here.
- 5.2 **Imperial Wharf** may be served on Mondays to Fridays, Saturdays and Sundays.

## 6. Early and Late Services

	Early Service arrives at destination no later than:		
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
East Croydon to Milton Keynes Central	0815	0815	0900
Milton Keynes Central to East Croydon	1015	0915	1000

	Late Service departs no earlier than:			
Route	Monday - Friday	Saturday	Sunday <sup>1</sup>	
East Croydon to Milton Keynes Central	1900	1900	2200	
Milton Keynes Central to East Croydon	2200	1900	2215	

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Clapham Junction and Watford Junction.

# 7. Maximum Journey Times

Route	Monday - Friday	Saturday	Sunday <sup>1</sup>
East Croydon to Milton Keynes Central	1 hour 51 minutes with two exceptions of 1 hour 54 minutes permitted	1 hour 50 minutes	41 minutes with one exception of 43 minutes permitted
Milton Keynes Central to East Croydon	1 hour 44 minutes with four exceptions of 1 hour 44 minutes and one further exception of 2 hours 3 minutes permitted	1 hour 43 minutes with two exception of 1 hour 45 minutes	41 minutes

<sup>&</sup>lt;sup>1</sup> Sunday services are specified between Clapham Junction and Watford Junction.

8.

ECML 1: East Coast Mainline Minimum Service Specification and First and Last Train Specification

	TABLE 1: NORTHBOUND								Monday	to Friday										Satu	rdav				1				Sunday			
	Departure time from Londor	n	Journe	ey Time	0200-0659	0700-1559	0700-1559	1600-1859			2000-2059	2100-21:59	2200-0159	Servi	ice leaves L	ondon	Journe	y Time	0200-0659			2200-0159	Service leav	ves Londor	n Journe	ey Time	0200-0859	0900-2059	9 0900-2059	2100-0159	9 Service leav	ves London
															Last (no	Last (Fri								Last (no								Last (no
	Direct services from London to each station	n Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	First (no later than)	earlier than)	nights only)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	er earlier than)
_	Old Street	Moorgate			4	6	58	32	12	7	6	6	10	06:00	00:00	J,	. m en a ge		5	4	60	10	06:00	23:45	- manage		12 (note 1)		48	14	06:00 (note 1	
<u>\$</u>	Essex Road	Moorgate			4	6	58	32	12	7	6	6	10	06:00	00:00				5	4	60	10	06:00	23:45			12 (note 1)	4 (note 1)	48	14	06:00 (note 1	
×	Highbury & Islington Drayton Park	Moorgate Moorgate			4	6	58 58	32 32	12 12	7 7	6	6	10 10	06:00 06:00	00:00				5	4	60 60	10 10	06:00 06:00	23:45 23:45			12 (note 1) 12 (note 1)	4 (note 1) 4 (note 1)	48 48	14 14	06:00 (note 1 06:00 (note 1	
and	Finsbury Park	London			12	10	95	50	18	12	10	10	21	05:30	01:30				13	8	120	24	05:30	01:00			18	7 (note 7)	87	26	06:00	01:30
elq	Harringay	London			6	6	55	18	6	6	6	6	10	05:30	00:00	01:00			6	4	60	13	05:30	01:00			12	4	48	15	06:00	00:00
latfi	Hornsey Alexandra Palace	London London			6	6	55 58	18 27	6 10	6	6	6	10 12	05:30 05:30	00:00 01:30	01:00			6	4	60 60	13 13	04:30 05:30	01:00			12 12	4	48 48	15 15	06:00	00:00 01:30
, F	New Southgate	London	1		2	3	27	16	6	3	3	3	6	06:15	01:30	1			2	2	30	7	06:15	01:00	1		5	2	24	8	06:45	01:30
B	Oakleigh Park	London			2	3	27	16	6	3	3	3	6	06:15	01:30				2	2	30	7	06:15	01:00			5	2	24	8	06:45	01:30
otter	New Barnet Hadley Wood	London London			2 2	3	27 27	15 14	5	3	3	3	6	06:15 06:15	01:30 01:30				2 2	2 2	30 30	7	04:30 06:15	01:00 01:00			5 5	2 2	24 24	8	06:45 06:45	01:30 01:30
9	Potters Bar	London			5	5	45	18	6	4	5	5	11	05:30	01:30				5	4	60	13	05:30	01:00			7	3 (note 7)	39	13	06:45	01:30
Park	Brookmans Park	London			2	3	27	11	4	2	3	3	5	06:15	00:00	01:00			2	2	30	7	06:15	01:00			5	2	24	8	06:45	00:00
LIY E	Welham Green Hatfield	London London			2 5	3 5	27 45	11 23	4 8	2 5	5	5	11	06:15 04:15	00:00 01:30	01:00			5	2	30 60	7 13	06:15 04:30	01:00 01:00			5 7	2 3 (note 7)	24 39	8 13	06:45 06:45	00:00 01:30
spn	Welwyn Garden City	London			5	5	45	23	8	5	5	5	11	04:15	01:30				5	4	60	13	04:30	01:00			7	3 (note 7)	39	13	06:45	01:30
E G		London King's Cross	28	40	3	2	19	12	4	3	2	2	7				26	28	3	2	30	4			25	25	2	1	15	6		
cept		of which Moorgate	26 48	28 50	3 2	2	18 26	7 11	2 4	1 2	2 3	3	4				N/A 49	N/A 49	2	2	30	9			N/A 50	N/A 52	5	2	24	7		
ě	Bowes Park	London			4	3	28	12	4	5	3	3	5	05:30	00:00		, i		4	2	30	6	05:30	00:45	1 -		7	2	24	7	06:00	00:00
oul	Palmers Green Winchmore Hill	London London	l		4	3	31 31	21 21	8	5	3	3	6	05:30 05:30	01:00 01:00				4	2 2	30 30	6	05:30 05:30	00:45 00:45			7 7	2	24 24	7	06:00 06:00	01:00 01:00
ses	Grange Park	London	l		4	3	28	12	4	5	3	3	5	05:30	00:00				4	2	30	6	05:30	00:45		1	7	2	24	7	06:00	00:00
Ğ.	Enfield Chase	London			4	3	31	21	8	5	3	3	6	05:30	01:00				4	2	30	6	05:30	00:45			7	2	24	7	06:00	01:00
ές O	Gordon Hill Crews Hill	London London			4	3	31 27	21 6	8	5	3	3	6	05:30 05:30	01:00 00:00				4	2 2	30 30	6	05:30 05:30	00:45 00:45			7 7	2	24 24	7	06:00 06:00	01:00 00:00
<u>u</u>	Cuffley	London			4	3	28	15	6	5	3	3	6	05:30	01:00				4	2	30	6	05:30	00:45			7	2	24	7	06:00	01:00
¥	Bayford	London			4	3	27	6	2	3	3	3	5	05:30	00:00				4	2	30	6	05:30	00:45			7	2	24	7	06:00	00:00
	Hertford North Watton-at-Stone	London London	48	50	4 3 (note 2)	3	28 13	15	6	5	3	3	6	05:30	01:00		50	52	4	2	30	6	05:30 06:30	00:45 00:45	50	52	7	2	24 12	7	06:00 06:30	01:00 00:00
	Welwyn North	London King's Cross	1		3 (note 2)	2	18	9	2	2	2	2	6	04:15	01:30	1			3	2	30	6	05:30	00:45	1		2	1 (note 7)	15	5	07:15	01:30
	Knebworth	London King's Cross			3	2	18	10	3	2	2	2	6	04:15	01:30				3	2	30	6	05:30	00:45			2	1 (note 7)	15	5	07:15	01:30
	Stevenage Hitchin	London King's Cross London King's Cross			6	4	36 36	17 17	6	4 5	4	4	10 10	04:15 04:15	01:30 01:30				6	4	60 60	14 14	05:30 05:30	00:45 00:45			6	3 (note 7) 3 (note 7)	39 39	12 12	06:45 06:45	01:30 01:30
	Letchworth	London King's Cross	45	67	3	2	18	12	4	2	3	3	7	04:15	01:00		41	58	3	2	30	7	05:45	00:45	41	54	5	2 (note 7)	25	7	06:45	01:00
		of which	43	55	3	2	18	12	4	1	3	3	6				34	36	2	1	15	2			34	34	2	1	12	2		
≥	Baldock	of which London King's Cross	34	37	2	1 2	9 18	5 10	2	0	1 2	1 2	1	05:45	01:00		N/A	N/A	3	2	30	5	05:45	00:15	N/A	N/A	5	2	24	7	06:45	01:00
88 01	Ashwell & Morden	London King's Cross			3 (note 3)	1	12 (note 3)	10	3	1	2	2	5	05:45	01:00				3	1	15	4	05:45	00:00			2	1	12	5	07:15	01:00
Ş.	Royston	London King's Cross			3	2	18	12	4	1	3	3	5	04:15	01:00		53	69	3	2	30	5	05:45	00:15	54	67	5	2 (note 7)	25	7	06:45	01:00
Ser	Meldreth Shepreth	London King's Cross London King's Cross			3 (note 3) 3 (note 3)	1	11 (note 3) 11 (note 3)	7 (note 4) 7 (note 4)	1 1	1 1	2	2 2	4	05:45 05:45	00:00 00:00				3	1 1	15 15	3	05:45 05:45	00:00			2 2	1 1	12 12	5	07:15 07:15	00:00 00:00
nter	Foxton	London King's Cross			3 (note 3)	1	11 (note 3)	7 (note 4)	1	1	2	2	4	05:45	00:00				3	1	15	3	05:45	00:00			2	1	12	5	07:15	00:00
2	Cambridge	London King's Cross	78	96 72	3	2	18 9	7	2	1	2	2	4	05:45	00:00	00:30	72 62	83 62	3	2	30 15	4	05:45	00:15	74 63	92 63	5	2	24	7	06:45	00:00
Ö	Arlesey	,,,of which London King's Cross	00	12	3	2	18	6	2	2	2	2	3	05:30	00:30		02	02	3	2	15 30	6	05:30	00:45	0.3	03	1 (note 6)	1 1 (note 7)	12	5	07:15	00:30
ш ш	Biggleswade	London King's Cross	l		3	2	18	8	2	2	2	2	6	05:30	01:30				3	2 (note 5)	33	6	05:30	00:45			1 (note 6)	1 (note 7)	15	5	07:15	01:30
	Sandy St. Neots	London King's Cross London King's Cross	l		3	2	18 18	6 11	2	2	2 2	2	3	05:30 05:30	00:30 01:30				3	2 2 (note 5)	30 33	6 6	05:30 05:30	00:45 00:45		1	1 (note 6) 1 (note 6)	1 (note 7) 1 (note 7)	14 15	5	07:15 07:15	00:30 01:30
	Huntingdon	London King's Cross			3	2	18	11	4	3	2	2	6	05:30	01:30				3	2 (note 5) 2 (note 5)	33	6	05:30	00:45		1	1 (note 6)	1 (note 7)	15	5	07:15	01:30
	Peterborough	London King's Cross	80	100	3	2	18	10	4	3	2	2	6	05:30	01:30		81	100	3	2 (note 5)	33	6	05:30	00:45	79	100	1 (note 6)	1 (note 7)	15	5	07:15	01:30
		of which	73 66	80 70	1 0	1	9	8	4 2	2	2	2	4 2				73 63	80 70	1	1	18 3	2			76 62	80 62	1 0	1	12	3		
× -	Letchworth	London King's Cross	- 00	,,,	0	0	0	1	0	2	2	2	3	-	-		- 00	70	0	0	4	2	- 1	-	02	02	0	0	0	1	-	-
CEX	Royston	London King's Cross	49	54	0	0	0	5	2	2	2	2	3	- 06:45	23:00		48	F.1	0	0	4	2	- 06:30	23:00	47	52	0	0	0	1	- 09:15	-
Prior to	Cambridge the Passenger Change Date in May 2017	London King's Cross	49	54	1 1	2	18	ь		2	- 2	Z		Ub:45	23:00		48	51	1 1	2	30	Z	06:30	23:00	4/	52	U	1	12	3	09:15	23:00
ML	Waterbeach	London King's Cross			1 (note 8)	1	11	3	1	1	2	2	3	06:00	23:00				1 (note 9)	1	15	2	07:00	23:00			1	1	12	3	08:00	23:00
Z E	Ely Littleport	London King's Cross London King's Cross			1 (note 8) 1 (note 8)	1	11 10	5 3	2	2	2 2	2	3	06:00 06:00	23:00 23:00				1 (note 9) 1 (note 9)	1 1	16 15	2	07:00 07:00	23:00 23:00		1	1 1	1 1	12 12	3	08:00 08:00	23:00 23:00
and Se	Downham Market	London King's Cross London King's Cross			1 (note 8) 1 (note 8)	1	10	4	1	1 1	2	1 1	2	06:00	23:00				1 (note 9) 1 (note 9)	1	16	2	07:00	23:00		1	1	1 1	12	3	08:00	23:00
CEx a	Watlington	London King's Cross	l		1 (note 8)	1	10	3	1	1	2	1	2	06:00	23:00				1 (note 9)	1	15	2	07:00	23:00		1	1	1	12	3	08:00	23:00
	King's Lynn e Passenger Change Date in May 2017	London King's Cross	98	104	1 (note 8)	1	10	4	1	1	2	1	2	06:00	23:00		97	103	1 (note 9)	1	16	2	07:00	23:00	99	101	1	1	12	3	08:00	23:00
⊢ o	Waterbeach	London King's Cross			2 (note 8)	1	11	3	1	1	2	2	3	06:00	23:00	1			2 (note 9)	1	15	2	06:00	23:00	1	I	1	1	12	3	08:00	23:00
Vice V	Ely	London King's Cross	l		2 (note 8)	2	18	6	2	2	2	2	3	06:00	23:00				2 (note 9)	2	30	2	06:00	23:00			1	1	12	3	08:00	23:00
Sen	Littleport	London King's Cross	l		2 (note 8)	1 2 (note 42)	10	3	1	1	2	1	2	06:00	23:00				2 (note 9)	1 2 (note 10)	15	2	06:00	23:00			1	1	12	3	08:00	23:00
x ar	Downham Market Watlington	London King's Cross London King's Cross	l		2 (note 8) 2 (note 8)	2 (note 10) 1	16 10	6 3	2 1	2	2 2	1	2	06:00 06:00	23:00 23:00				2 (note 9) 2 (note 9)	2 (note 10) 1	29 15	2	06:00 06:00	23:00 23:00			1 1	1 1	12 12	3	08:00 08:00	23:00 23:00
80	King's Lynn	London King's Cross	98	104	2 (note 8)	2 (note 10)	16	6	2	2	2	2	3	06:00	23:00		97	103	2 (note 9)	2 (note 10)	29	2	06:00	23:00	99	101	1	1	12	3	08:00	23:00
	•	_																														

Note 1: The minimum requirement at Old Street, Essex Road, Highbury & Islington and Drayton Park before 10:00 on a Sunday morning may be reduced, but only to the extent that the Franchisee (having used all reasonable endeavours) is unable to secure the necessary rights to operate services on the Moorgate branch.

Note 2: One additional service shall be provided from Hertford North to Stevenage, calling at Watton-at-Stone, arriving at Stevenage before 06:15.

Note 3: The specified totals must include a total of six services from London King's Cross to Ashwell & Morden, Meldreth, Shepreth and Foxton, and continuing to Cambridge, so as to provide a total of six services from each of those stations arriving at Cambridge between 07:15 and 10:15, with a maximum interval of 40 minutes between consecutive arrivals at Cambridge.

Note 4: Intervals between the arrival times of consecutive services at Meldreth, Shepreth and Foxton between 16:00 and 18:59 shall be no more than 40 minutes.

Note 5: The requirement increases to three trains per hour between 16:00 and 18:59 on Saturdays.

Note 6: One additional service shall be provided from Stevenage to Peterborough, calling at all stations, arriving at Peterborough before 08:45.

Note 7: The requirement at Finsbury Park, Potters Bar, Hatfield, Welwyn Garden City, Welwyn North, Knebworth, Stevenage, Hitchin, Biggleswade, St. Neots, Huntingdon and Peterborough increases by one train per hour between 16:00 and 18:59 on Sundays.

The requirement at Affects and Sandy increases by one train per hour between 16:00 and 18:59 on Sundays.

The requirement at Letchworth and Royston increases by one train per hour between 18:00 and 18:59 on Sundays.

The requirement at Letchworth and Royston increases by one train per hour between 18:00 and 18:59 on Sundays.

Note 9: One additional service shall be provided from Cambridge to King's Lynn, calling at all stations, arriving at King's Lynn before 07:30.

Note 10: The frequency of service may be reduced to one train per hour to hours of the

	TABLE 2: SOUTHBOUND							Mondoy	to Friday									Ç <sub>0</sub> +	urday				_			e	ınday			
	Arrival time in Londor	1	Journ	ney Time	0200-0659	0700-0959	0800-085			2000-2059	2100-2159	2200-015	9 Service ar	rives London	Journe	ey Time	0200-0659		0700-2159	2200-0159	Service arr	rives London	Journe	ey Time	0200-0859	9 0900-2059		32100-0159	9 Service arrive	es London
														Last (no		ĺ						Last (no		Í						Last (no
	Direct services to London from each station	Destination	Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	First (no later than	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)
Prior to t	e Passenger Change Date in May 2017	Destination	Average	IVIdXIIIIUIII	TUIdi	TUIdi	TUlai	IFF	TULAI	TUIdi	TOTAL	TOTAL	iatei tiiaii	) lilali)	Average	IVIAXIIIIUIII	TUIdi	IFN	TUIdi	TUIdi	iatei tiiaii)	liidii)	Avelage	IVIdXIIIIUIII	Tutai	IFH	TULAI	TUIdi	triari)	liidii)
es ME	King's Lynn	London King's Cross	98	103	1	5	2	1	10	1	1	3	06:45	00:30	97	102	0	1	16	2 (note 1)	07:45	23:30	101	103	0	1	12	4	10:00	00:30
집음	Watlington	London King's Cross			1	4	2	1 1	10	1	1	3	06:45	00:30			0	1 1	15	2 (note 1)	07:45	23:30			0	1 1	12	4	10:00	00:30
Ser	Downham Market	London King's Cross London King's Cross			1	5	2	1 1	11 10	1 1	1 1	3	06:45 06:45	00:30 00:30			0	1 1	16 15	2 (note 1) 2 (note 1)	07:45 07:45	23:30 23:30			0	1 1 1	12 12	4	10:00 10:00	00:30 00:30
xa	Littleport Elv	London King's Cross			1	5	2	1 1	14	Ιi	l i	3	06:45	00:30			1	1 1	16	2 (note 1)	06:45	23:30			0	1 1	12	4	10:00	00:30
9 0	Waterbeach	London King's Cross			1	4	2	1	12	1	1	3	06:45	00:30			1	1	15	2 (note 1)	06:45	23:30			0	1	12	4	10:00	00:30
From the	Passenger Change Date in May 2017	1																												
Ces	King's Lynn Watlington	London King's Cross London King's Cross	98	103	1 1	6	2 2	2 (note 2)	18 10	2	2	5 3	06:45 06:45	00:30 00:30	97	102	0	2 (note 2)	29 15	5 (note 1) 3 (note 1)	07:45 07:45	00:30 00:30	101	103	0	1 1	12 12	4	10:00 10:00	00:30 00:30
er E	Downham Market	London King's Cross			l i	6	2	2 (note 2)	18	2	2	5	06:45	00:30			0	2 (note 2)		5 (note 1)	07:45	00:30			0	l i l	12	4	10:00	00:30
and rr Se	Littleport	London King's Cross			1	6	2	1	10	1	1	3	06:45	00:30			0	1	15	3 (note 1)	07:45	00:30			0	1	12	4	10:00	00:30
CEX	Ely	London King's Cross			1	6	2	2	20	2	2	5	06:45	00:30			1	2	30	5 (note 1)	06:45	00:30			0	1	12	4	10:00	00:30
00	Waterbeach Cambridge	London King's Cross	50	EA	1	5	2	1	10	1	1 2	3	06:45 06:45	00:30 23:30	E0.	54	1	1	15 30	3 (note 1)	06:45	00:30	40	E1	0	1 1	12	4	10:00 10:00	00:30 23:15
CEX	Royston	London King's Cross London King's Cross	50	54	1 1	5	2	0	20 0	0	0	0	00.45	23.30	50	34	Ιί	0	0	0	06:45	23:30	49	51	0	0	12 0	3	10.00	23:15
0 0	Letchworth	London King's Cross			1	3	0	0	0	0	0	0	-	-			1	0	0	0	-	-			0	0	0	0	-	-
	Peterborough	London King's Cross	82	100	4	10	6	2 (note 3)	21	2	2	4	05:00	23:45	82	100	3	2 (note 4)	32	5	05:00	00:00	75	100	2	1 (note 5)	15	4	07:30	00:15
		of which	75 68	80 70	2	7	4	1 0	11 1	1 0	1 0	0			72 64	80 70	0	1 0	15 3	0			73 60	80 70	0	0	15 3	3 0		
	Huntingdon	London King's Cross	00	10	4	10	6	2 (note 3)	22	2	2	4	05:00	23:45	04	10	3	2 (note 4)	33	5	05:00	00:00	- 00	10	2	1 (note 5)	15	4	07:30	00:15
	St. Neots	London King's Cross	I		4	10	6	2 (note 3)	22	2	2	4	05:00	23:45			3	2 (note 4)	33	5	05:00	00:00	I		2	1 (note 5)	15	4	07:30	00:15
	Sandy	London King's Cross			2	5	2	2	20	2	2	4	06:00	23:45			2	2	30	5	06:00	00:00	1		2	1 (1.1.5)	12	4	07:30	00:15
l luc	Biggleswade Arlesev	London King's Cross London King's Cross			3	5	3	2 (note 3)	21	2	2 2	4	05:00 06:00	23:45 23:45			3	2 (note 4)	33	5	05:00 06:00	00:00 00:00			2	1 (note 5)	15 12	4	07:30 07:30	00:15 00:15
es es	Cambridge	London King's Cross	78	91	0	7	2	2	20	2	2	5	07:15	00:30	75	86	0	2	30	5	07:30	00:30	73	85	3	2	24	7	07:45	00:30
Si.		of which	63	67	0	2	0	1	10	1	1	0			62	62	0	1	15	0			61	64	1	1	12	3	7	
Se	Foxton	London King's Cross			0	6	2	1 (note 6)	13	2	1 1	3	07:15	00:30			0	1	15	3	07:30	00:30			1	1 1	12	4	08:30	00:30
ute	Shepreth Meldreth	London King's Cross London King's Cross			0	6	2 2	1 (note 6) 1 (note 6)	13 13	2	1 1	3	07:15 07:15	00:30 00:30			0	1 1	15 15	3	07:30 07:30	00:30 00:30			1 1	1 1 1	12 12	4	08:30 08:30	00:30 00:30
Ŏ	Royston	London King's Cross			2	12	5	2	20	2	2	5	06:30	00:30	58	66	1	2	30	5	06:30	00:30	56	65	3	2	24	7	07:45	00:30
S	Ashwell & Morden	London King's Cross			2	9	3	1 (note 6)	13	2	1	3	06:30	00:30			1	1	15	3	06:30	00:30			1	1	12	4	08:30	00:30
ш	Baldock	London King's Cross	46	57	2	9 13	3	2	20 20	2	2	3 5	06:30	00:30	45	52	1	2	30 30	5	06:30	00:30	44		3	2	24	7	07:45	00:30
	Letchworth	London King's Cross of which	37	38	2	13	2	1	10	1	2	0	06:30	00:30	45 36	36	0	1	15	5	06:30	00:30	36	52 39	3	2 (note 7)	26 12	3	07:45	00:30
	Hitchin	London King's Cross	- 51	- 50	6	18	7	4	40	4	4	9	05:00	00:30	- 50	- 50	4	4	60	10	05:00	00:30	- 50	- 55	5	3 (note 7)	38	11	07:30	00:30
	Stevenage	London King's Cross			6	20	9	4 (note 8)	41	4	4	9	05:00	00:30			4	4	60	10	05:00	00:30			5	3 (note 7)	38	11	07:30	00:30
	Knebworth Welwyn North	London King's Cross			3	9	3	2	20 20	2	2	5	06:00 06:00	00:30 00:30			3	2	30 30	5	06:00 06:00	00:30 00:30	1		2	1 (note 7) 1 (note 7)	14 14	4	07:30 07:30	00:30 00:30
_	Watton-at-Stone	London King's Cross London			2	7	2	1	16	1	1	4	06:00	00:30			2	1	15	3	06:00	00:30			1	1 (note 7)	12	4	08:30	00:30
- W	Hertford North	London	47	50	3	15	6	3	33	4	2	6	05:00	01:00	47	47	3	2	30	6	05:00	00:30	47	48	5	2	24	8	07:00	01:00
S ∈	Bayford	London			2	8	3	3	30	2	2	6	06:00	01:00			3	2	30	6	06:00	00:30			5	2	24	8	07:00	01:00
Je J	Cuffley	London			3 2	15 8	6	3	33 30	4	2	6	05:00 06:00	01:00 01:00			3	2	30 30	6	05:00 06:00	00:30 00:30			5 5	2 2	24 24	8	07:00 07:00	01:00 01:00
a G	Crews Hill Gordon Hill	London London			3	22	3 8	3	37	4	2	6	05:00	01:00			3	2	30	6	05:00	00:30	1		5	2	24	8	07:00	01:00
#iel	Enfield Chase	London			3	22	8	3	37	4	2	6	05:00	01:00			3	2	30	6	05:00	00:30	1		5	2	24	8	07:00	01:00
五	Grange Park	London	I		2	15	6	3	31	2	2	6	06:00	01:00			3	2	30	6	06:00	00:30	I		5	2	24	8	07:00	01:00
Bar,	Winchmore Hill Palmers Green	London London	1		3	19 19	7	3	37 37	4	2 2	6	05:00 05:00	01:00 01:00		1	3	2 2	30 30	6	05:00 05:00	00:30 00:30	1		5 5	2 2	24 24	8	07:00 07:00	01:00 01:00
ars 6	Bowes Park	London	1		2	18	7	3	31	2	2	6	06:00	01:00		1	3	2	30	6	06:00	00:30	1		5	2	24	8	07:00	01:00
otte	Welwyn Garden City	London			6	22	8	5	52	4	4	9	05:00	00:30			5	4	60	10	05:00	00:30			6	3 (note 7)	38	10	07:15	00:30
<u>ب</u> د		London King's Cross	30	41	4	13	4	2	28	2	2	4 4			29	41	4	2	30	6			29	31	2	1	14	4		
Park, City		of which Moorgate	28 48	31 50	2 2	8 9	2 4	2 3	28 24	2 2	2 2	5			29 47	31 47	2	2 2	30 30	6 4			N/A 47	N/A 47	4	2	24	6		
ury len o	Hatfield	London	40	- 50	6	19	6	5	52	4	4	9	05:00	00:30	4,		5	4	60	10	05:00	00:30		7,	6	3 (note 7)	38	10	07:15	00:30
တ္က မ	Welham Green	London	1		2	10	4	3	30	2	2	4	06:15	23:45		1	1	2	30	4	06:45	23:45	1		4	2	24	6	07:15	23:45
g. Fin	Brookmans Park	London London	1		3 6	10 18	4 6	3	30 52	2	2	4 9	05:00 05:00	23:45 00:30		1	2	2	30 60	4 10	05:00 05:00	23:45 00:30	1		4 6	2 (noto 7)	24	6	07:15 07:15	23:45 00:30
ceb	Potters Bar Hadley Wood	London	I		3	18	4	3	30	2	2	4	05:00	23:45			3	2	30	4	05:00	23:45	I		4	3 (note 7)	38 24	10 6	07:15	23:45
×Θ	New Barnet	London	1		4	15	6	3	30	2	2	4	05:00	23:45		1	3	2	30	4	05:00	23:45	1		4	2	24	6	07:15	23:45
yluc	Oakleigh Park	London	1		4	15	6	3	30	2	2	4	05:00	23:45		1	3	2	30	4	05:00	23:45	1		4	2	24	6	07:15	23:45
es c	New Southgate Alexandra Palace	London Lonodon	+	+	7	15	10	6	30 60	2	2	10	05:00 05:00	23:45 01:00		<del>                                     </del>	3	2	30 60	10	05:00 05:00	23:45 00:30	<del>                                     </del>	1	4	4	24 48	6 14	07:15 07:15	23:45 01:00
<u>Ş</u> .	Hornsey	London	1		5	28 23	8	6	60	4	4	10	05:00	01:00		1	5	4	60	10	05:00	00:30	1		9	4 4	48 48	14	07:15 07:15	01:00
S	Harringay	London	1		5	23	8	6	60	4	4	10	05:00	01:00			5	4	60	10	05:00	00:30	I		9	4	48	14	07:15	01:00
ner	Finsbury Park	London	I		11	52	19	10	108	8	8	17	05:00	01:00			8	8	120	17	05:00	00:30	I		12	7 (note 7)	86	24	07:15	01:00
드	Drayton Park Highbury & Islington	Moorgate Moorgate	1		5	32 32	12 12	6	67 67	6	4	10 10	06:00 06:00	00:30 00:30		1	4	4	60 60	10 10	06:00 06:00	00:30 00:30	1		7 (note 8) 7 (note 8)	4 (note 8) 4 (note 8)	48 48	10 10	07:30 (note 7) 07:30 (note 7)	00:30 00:30
Ö	Essex Road	Moorgate	1		5	32	12	6	67	6	4	10	06:00	00:30			4	4	60	10	06:00	00:30	I		7 (note 8)		48	10	07:30 (note 7)	00:30
ш	Old Street	Moorgate			5	32	12	6	67	6	4	10	06:00	00:30			4	4	60	10	06:00	00:30			7 (note 8)	4 (note 8)	48	10	07:30 (note 7)	00:30

Note 1: One additional service shall be provided from King's Lynn to Cambridge, calling at all stations, departing King's Lynn no earlier than 23:00.

Note 2: The frequency of service may be reduced to one train per hour in up to two hours of the day between 10:00 and 19:59 Monday to Friday and one hour between 07:00 and 21:59 on a Saturday to allow for pre-existing freight services to operate.

Note 3: The requirement increases to three trains per hour between 09:00 and 11:59 on Suturdays.

Note 5: The requirement increases to two trains per hour between 10:00 and 12:59 on Sundays.

Note 6: The requirement increases to two trains per hour arriving at London King's Cross between 17:00 and 19:59 on weekdays.

Note 7: The requirement increases by one train per hour arriving at London King's Cross between 10:00 and 11:59 on Sundays.

Note 8: The minimum requirement at Drayton Park, Highbury & Islington, Essex Road and Old Street before 10:00 on a Sunday morning may be reduced, but only to the extent that the Franchisee (having used all reasonable endeavours) is unable to secure the necessary rights to operate services on the Moorgate branch.

## ECML 2: East Coast Mainline Minimum Service Specification and First and Last Train Specification

	TABLE 1: NORTHBOUND								Monday	to Friday							1			Satu	rday				I				Sunday			$\overline{}$
	Departure time from Londo	n	Journe	ey Times	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Servi	ice leaves L		Journey	y Times	0200-0659	0700-2159	0700-2159	2200-0159	Service lea		Journe	ey Times	0200-0859	9 0900-2059	0900-2059	2100-0159	Service leave	
															Last (no									Last (no								Last (no
	District for the last of the last				T 1	TPH	T	T 1	T 1	T 1	T	T 1	T 1	First (no	earlier	nights			T	TDU	T 1	T 1	First (no	earlier			T	TDU	T. 1.1	T	First (no later	
	Direct services from London to each statio		Average	Maximum	Total	IPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	only)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum		TPH	Total	Total 14	than)	than)
€	Old Street Essex Road	Moorgate		1	4	0	73 73	38 38	14 14	8	8		10	06:00 06:00	00:00 00:00		l		5	8	120 120	10 10	06:00 06:00	23:45 23:45			12 (note 1) 12 (note 1)	) 6 (note 1) ) 6 (note 1)	72 72	14	06:00 (note 1) 06:00 (note 1)	00:00
- €	Highbury & Islington	Moorgate Moorgate		1	4	8	73	38	14	8	8	8	10	06:00	00:00		l		5	8	120	10	06:00	23:45			12 (note 1)	) 6 (note 1)	72	14	06:00 (note 1)	) 00:00
×	Drayton Park	Moorgate		1	4	8	73	38	14	8	8	8	10	06:00	00:00		l		5	8	120	10	06:00	23:45			12 (note 1	) 6 (note 1)	72	14	06:00 (note 1)	) 00:00
2	Finsbury Park	London		1	12	14	136	68	24	16	14	14	20	05:30	01:30		l		12	14	210	24	05:30	01:00			18	10	120	26	06:00	01:30
a	Harringay	London		1	6	6	54	18	6	6	6	6	10	05:30	00:00	01:00	l		6	8	120	11	05:30	01:00			12	6	72	15	06:00	00:00
<u>=</u>	Hornsey	London		1	6	6	54	18	6	6	6	6	10	05:30	00:00	01:00	l		6	8	120	11	05:30	01:00			12	6	72	15	06:00	00:00
Ŧ	Alexandra Palace	London			6	8	73	40	13	9	8	8	11	05:30	01:30				6	8	120	13	05:30	01:00			12	6	72	15	06:00	01:30
	New Southgate	London			2	4	39	24	8	6	4	4	6	06:00	01:30				2	4	60	7	06:15	01:00			5	2	24	8	06:45	01:30
Bar	Oakleigh Park	London		1	2	4	39	24	8	6	4	4	6	06:00	01:30		l		2	4	60	7	06:15	01:00			5	2	24	8	06:45	01:30
ers	New Barnet	London		1	2	4	39	24	8	6	4	4	6	06:00	01:30		l		3	4	60	7	06:15	01:00			5	2	24	8	06:45	01:30
Pott	Hadley Wood	London		1	2	2	20	14	5	2	2	2	6	06:00	01:30		l		2	2	30	7	06:15	01:00			5	2	24	8	06:45	01:30
	Potters Bar	London		1	5	6	59 20	24 12	8 4	8	6 2	6	11	05:30 06:00	01:30	01:00	l		5	6	90	13 6	05:30 06:15	01:00 01:00			_ ′	4	48 24	14 8	06:45 06:45	01:30 00:00
Park	Brookmans Park Welham Green	London London		1	2	2	20	12	4	2 2	2	2	5	06:00	00:00 00:00	01:00	l		2	2 2	30 30	6	06:15	01:00			5	2 2	24	8	06:45	00:00
Şi d	Hatfield	London		1	5	6	59	24	8	8	6	6	11	05:30	01:30	01.00	l		5	6	90	13	05:30	01:00			7	4	48	14	06:45	01:30
	Welwyn Garden City	London	I	1	5	6	59	30	10	8	6	6	11	05:30	01:30	l			5	6	90	13	05:30	01:00	I	1	7	4	48	13	06:45	01:30
-insb		London Kings Cross	32	35	1	0	3	6	2	0	0	0	2	00.00	01.00		26	33	2	0	0	4	55.55	000	N/A	N/A	0	0	0	0	555	000
t t		Moorgate	46	53	2	4	36	12	4	4	4	4	5				46	48	2	4	60	4			50	52	5	2	24	8		
8		London St. Pancras	28	33	2	2	20	12	4	4	2	2	4				26	26	1	2	30	5			26	26	2	2	24	5		
ě	Bowes Park	London			4	4	36	13	4	5	4	4	5	05:30	00:00				4	4	60	5	05:30	00:45			7	4	48	7	06:00	00:00
l É	Palmers Green	London		1	4	4	37	26	10	5	4	4	5	05:30	01:00		l		4	4	60	6	05:30	00:45			7	4	48	7	06:00	01:00
88	Winchmore Hill	London		1	4	4	37	26	10	5	4	4	5	05:30	01:00		l		4	4	60	6	05:30	00:45			7	4	48	7	06:00	01:00
.ĕ	Grange Park	London		1	4	4	37	22	6	5	4	4	5	05:30	00:00		l		4	4	60	5	05:30	00:45			7	4	48	7	06:00	00:00
Je J	Enfield Chase	London		1	4	4	37	26	10	5 5	4 4	4 4	5	05:30	01:00		l		4	4	60	6	05:30	00:45			7	4	48	7	06:00	01:00
<u>Б</u>	Gordon Hill	London London		1	4	4	37 18	26 6	10 2	3	2	4	5	05:30 05:30	01:00 00:00		l		4	4	60 30	0	05:30 05:30	00:45 00:45			7	2	48 24	7	06:00 06:00	01:00 00:00
Ĕ	Crews Hill Cuffley	London		1	4	Z	36	15	6	5	4	Z /	5	05:30	01:00		l		4	1 4	60	5	05:30	00:45			7	4	48	7	06:00	01:00
	Bayford	London		1	4	2	18	6	2	3	2	2	5	05:30	00:00		l		4	2	30	5	05:30	00:45			7	2	24	7	06:00	00:00
S	Hertford North	London	46	52	4	4	36	15	6	5	4	4	5	05:30	01:00		46	49	4	4	60	6	05:30	00:45	48	49	7	4	48	7	06:00	01:00
ш	Watton-at-Stone	London			3 (note 2)	2	18	6	2	2	2	2	3	06:00	23:45				3	2	30	4	06:00	00:45		"	3	1	12	4	06:30	23:45
	Welwyn North	London		1	3	2	18	6	2	2	2	2	6	05:30	01:30				3	4	60	6	05:30	00:45			2	2	24	7	07:15	01:30
	Knebworth	London		1	3	2	18	10	3	2	2	2	6	05:30	01:30		l		3	4	60	6	05:30	00:45			2	2	24	7	07:15	01:30
	Stevenage	London		1	6	6	54	18	6	6	6	6	10	05:30	01:30		l		6	6	90	15	05:30	00:45			6	4	48	12	06:45	01:30
	Hitchin	London			6	6	54	18	6	6	6	6	10	05:30	01:30				6	6	90	15	05:30	00:45			6	4	48	12	06:45	01:30
	Letchworth	London	NI/A	N//A	4	4	36	12	4	4	4	4	6	05:45	01:00		N//A	NI/A	3	4	60	10	05:45	00:45	N1/A	NI/A	5	2	24	7	06:45	01:00
		London Kings Cross	N/A	N/A	2	0	0	0	0	0	0	0	1				N/A	N/A	1	0	0	0			N/A	N/A 48	2	0	0	0 7		A I
		London St. Pancras (total) of which	41 35	48 36	2	4	36 18	12 6	4 2	2	2	2	5				41 35	46 35	2	4 2	60 30	10			47 N/A	N/A	0	2 0	24 0	0		A l
_	Baldock	London	33	36	1 3	2	19	10	3	2	2	2	5	05:45	01:00		33	33	3	2	30	5	05:45	00:15	IN/A	IN/A	5	2	24	7	06:45	01:00
<u> </u>	Ashwell & Morden	London		1	3 (note 3)	1	10 (note 3)	10	3	2	1	1	5	05:45	01:00		l		3	1 1	15	5	05:45	00:00			2	1	12	4	07:15	01:00
Se	Royston	London		1	4	4	36	12	4	4	4	4	5	05:45	01:00		l		3	4	60	10	05:45	00:00			5	2	24	7	06:45	01:00
. <u>ö</u>	Meldreth	London		1	3 (note 3)	1	10 (note 3)	6	2	2	1	1	4	05:45	00:00		l		3	1	15	3	05:45	00:00			2	1 1	12	4	07:15	00:00
Ser	Shepreth	London		1	3 (note 3)	1	10 (note 3)	6	2	2	1	1	4	05:45	00:00		l		3	1	15	3	05:45	00:00			2	1	12	4	07:15	00:00
9	Foxton	London		1	3 (note 3)	1	10 (note 3)	6	2	2	1	1	4	05:45	00:00		l		3	1	15	3	05:45	00:00			2	1	12	4	07:15	00:00
Out	Cambridge	London			4	4	36	12	4	4	4	4	4	05:45	00:00	00:30			3	4	60	10	05:45	00:15			5	2	24	7	06:45	00:00
¥		London Kings Cross	N/A	N/A	2	0	0	0	0	0	0	0	1				N/A	N/A	1	0	0	0			N/A	N/A	2	0	0	0		
5		London St. Pancras (total)	69	81	2	4	36 18	12	4	4	4	4	3				67	75	2	4	60	10			75	75	3	2	24	7		
	Arlanay	of which	60	61	1	2		6	2	2	2	2	0	05:20	00-20		59	59	1	2	30	5	05:00	00:45	N/A	N/A	2 (n=+= 5)	0	0	0	07:45	00:20
	Arlesey Biggleswade	London London	I	1	3	2	18 18	6	2 2	2 2	2 2	2	6	05:30 05:30	00:30 01:30	l			3	2 2 (note 4)	30 33	6	05:30 05:30	00:45 00:45	I	1	2 (note 5) 2 (note 5)	2 2	24 24	5	07:15 07:15	00:30 01:30
	Sandy	London			3	2	18	6	2	2	2	2	5	05:30	00:30				3	2 (note 4)	30	6	05:30	00:45	l	1	2 (note 5) 2 (note 5)	2	24	5	07:15	00:30
	St. Neots	London			3	2	18	11	4	3	2	2	6	05:30	01:30				3	2 (note 4)	33	6	05:30	00:45	l	1	2 (note 5)	2	24	5	07:15	01:30
	Huntingdon	London	I	1	3	2	18	11	4	3	2	2	6	05:30	01:30	l			3	2 (note 4)	33	6	05:30	00:45	I	1	2 (note 5)	2	24	5	07:15	01:30
	Peterborough	London			3	2	18	10	4	3	2	2	6	05:30	01:30				3	2 (note 4)	33	6	05:30	00:45	l	1	2 (note 5)	2	24	5	07:15	01:30
	_	London Kings Cross	62	62	1	0	0	4	2	0	0	0	1				91	100	2	0	3	2			N/A	N/A	1	0	0	0		
		London St. Pancras	78	84	2	2	18	6	2	2	2	2	5				76	76	1	2	30	4			77	77	1	2	24	5		4
× >	Letchworth	London Kings Cross			0	0	0	1	0	2		4	3						0	0	5	3		-		1	0	0	0	1	-	- 1
CEX	Royston	London Kings Cross	l	l	0	0	0	6	2	2	1	4	3	l		l	l !		0	0	5	3	l		l	l	0	0	0	1	1	
<u> </u>	Cambridge	London Kings Cross	49	54	1	2	18	6	2	2	<u> </u>	4	3	06:45	23:00		49	51	1 1	2	30	3	06:45	23:00	47	52	1 1	2	24	5	08:45	23:00
M ≥ ≤	Waterbeach	London Kings Cross			2 (note 6)	1	10	3	1	1	1	1	2	05:45	23:00				2 (note 7)	1 1	15	3	06:00	23:00	l	1	1	1 1	12	3	08:00	23:00
집절	Littleport	London Kings Cross			2 (note 6)	2	18	6	2	2	2	2	3	05:45	23:00				2 (note 7)	2	30 15	3	06:00	23:00	l	1	1 1	1 1	12	3	08:00	23:00
and E	Littleport Downham Market	London Kings Cross London Kings Cross	I	1	2 (note 6) 2 (note 6)	2 (note 0)	10 16	6	1 2	1 2	1 2	1 2	3	05:45 05:45	23:00 23:00	l			2 (note 7) 2 (note 7)	2 (noto 0)	15 29	3	06:00 06:00	23:00 23:00	I	1	1 1	1 1	12 12	3	08:00 08:00	23:00 23:00
x a	Watlington	London Kings Cross London Kings Cross	I	1	2 (note 6) 2 (note 6)	2 (note 8)	16	3	1	1	1	1	3	05:45	23:00	l			2 (note 7) 2 (note 7)	2 (note 8)	29 15	3	06:00	23:00	I	1	1	1 1	12	3	08:00	23:00
O G	King's Lynn	London Kings Cross	98	104		2 (note 8)	16	6	2	2	2	2	3	05:45	23:00		97	103	2 (note 7)	2 (note 8)	29	3	06:00	23:00	99	101	1	1	12	3	08:00	23:00
تت	· ···· g = -y · · · ·	zonaon rango oroso	30	107	~ (Hote 0)	~ (HOLO 0)	10	U						00.40	20.00		J1	100	- (HOIC 1)	- (HOLO 0)	23		00.00	20.00	55	101		<u> </u>	14	U	00.00	20.00

Note 1: The minimum requirement at Old Street, Essex Road, Highbury & Islington and Drayton Park before 10:00 on a Sunday morning may be reduced, but only to the extent that the Franchisee (having used all reasonable endeavours) is unable to secure the necessary rights to operate services on the Moorgate branch. Note 2: One additional service shall be provided from Hertford North to Stevenage, calling at Watton-at-Stone, arriving at Stevenage before 06:15.

Note 3: The specified totals must include four services from London to each of Ashwell & Morden, Meldreth, Shepreth and Foxton, arriving at Cambridge between 07:15 and 09:15, with a maximum interval of 35 minutes between consecutive arrivals at Cambridge.

Note 4: The requirement increases to three trains per hour between 16:00 and 18:59 on Saturdays.

Note 5: One additional service shall be provided from Stevenage to Peterborough, calling at all stations, arriving at King's Lynn before 07:15.

Note 6: One additional service shall be provided from Cambridge to King's Lynn, calling at all stations, arriving at King's Lynn before 07:30.

Note 7: One additional service shall be provided from Cambridge to King's Lynn, calling at all stations, arriving at King's Lynn before 07:30.

Note 8: The frequency of service may be reduced to one train per hour in up to two hours of the day between 07:00 and 15:59 Monday to Friday and one hour between 07:00 and 21:59 on a Saturday to allow for pre-existing freight services to operate.

	TABLE 2: SOUTHBOUND							Monday	to Eriday									Çatı	ırday							c	unday			
	Arrival time in Londor	1	Journe	ey Times	0200-0659	0700-0959	0800-0859			2000-2059	2100-2159	2200-0159	Service an	ives London	Journe	y Times	0200-0659			2200-0159	Service arri	ves London	Journe	y Times	0200-0859			2100-0159	Service arrive	es London
				ĺ										Last (no								Last (no								Last (no
													First (no	earlier							First (no	earlier							First (no later	earlier
	Direct services to London from each station		Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)		Maximum	Total	TPH	Total	Total	than)	than)
₩ Ses	King's Lynn	London Kings Cross	98	103	1	6	2	2 (note 2)	18	2	2	5	06:45	00:30	97	103	0	2 (note 2)	29	5 (note 1)	07:45	00:30	98	98	0	1	12	4	10:00	00:30
8 ≥ 2	Watlington	London Kings Cross			1	6	2	1 (	10 18	1 2	1 2	3	06:45	00:30			0	1 (	15	3 (note 1)	07:45	00:30			0	1	12	4	10:00 10:00	00:30
and l	Downham Market Littleport	London Kings Cross London Kings Cross			1	6	2	2 (note 2)	10	1	1	3	06:45 06:45	00:30 00:30			0	2 (note 2)	29 15	5 (note 1) 3 (note 1)	07:45 07:45	00:30 00:30			0	1	12 12	4	10:00	00:30 00:30
× ž	Elv	London Kings Cross			1	6	2	2	20	2	2	5	06:45	00:30			1	2		5 (note 1)	06:45	00:30			ő	i	12	4	10:00	00:30
Bo	Waterbeach	London Kings Cross			1	5	2	1	10	1	1	3	06:45	00:30			1	1	20	3 (note 1)	06:45	00:30			0	1	12	4	10:00	00:30
× >	Cambridge	London Kings Cross	49	53	1	6	2	2	20	2	2	4	06:45	23:30	50	54	1	2	30	5	06:45	23:30	48	50	0	2	24	4	09:45	23:30
CEX	Royston	London Kings Cross			1	6	2	0	0	0	0	2	-	- 1			1	0	0	2	-	-			0	0	0	0	-	- 1
	Letchworth Peterborough	London Kings Cross London			4	10	0	2	21	2	2	6	05:00	00:45			2	2 (note 3)	33	2	05:00	00:00			0	2	24	4	07:15	00:15
	reterbolough	London Kings Cross	69	74	2	4	4	0	1	0	0	2	05.00	00.43	91	100	0	0	3	1	03.00	00.00	N/A	N/A	0	0	0	2	07.15	00.15
		London St. Pancras	73		2	6	2	2	20	2	2	4			76	76	3	2	30	4			74	74	2	2	24	2		
	Huntingdon	London			4	10	6	2	20	2	2	6	05:00	00:45			3	2 (note 3)	33	5	05:00	00:00			2	2	24	4	07:15	00:15
	St. Neots	London			4	10	6	2	20	2	2	6	05:00	00:45			3	2 (note 3)	33	5	05:00	00:00			2	2	24	4	07:15	00:15
	Sandy	London	1		2	6	2	2	20	2	2	6	05:45	00:45			2	2	30	5	06:00	00:00			2	2	24	4	07:15	00:15
	Biggleswade Arlesey	London London			3	7	3	2	20	2	2	6	05:00 05:45	00:45 00:45			3	2 (note 3)	33 30	5	05:00 06:00	00:00 00:00			2	2	24 24	4	07:15 07:15	00:15 00:15
>	Cambridge	London			0	13	4	4	40	4	4	5	07:00	00:45			1	4	60	10	06:00	00:00			3	2	24	7	07:15	00:15
l f		London Kings Cross	N/A	N/A	0	0	0	0	0	0	0	2	07.00	00.00	N/A	N/A	0	0	0	1	30.40	00.00	N/A	N/A	0	0	0	1	07.10	00.00
ses		London St. Pancras (total)	67	77	0	13	4	4	40	4	4	3			67	76	1	4	60	9			72	73	3	2	24	6		
3		of which	58	60	0	7	2	2	20	2	2	0			58	58	0	2	30	4					0	0	0	0		
Se	Foxton	London			0	6	2	1 (note 4)	13	1	1	3	07:15	00:30			0	1	15	4	07:15	00:30			1 1	1	12	3	07:15	00:30
E E	Shepreth Meldreth	London London			0	6	2	1 (note 4) 1 (note 4)	13 13	1	1	3 3	07:15 07:15	00:30 00:30			0	1 1	15 15	4	07:15 07:15	00:30 00:30			1 1	1	12 12	3	07:15 07:15	00:30 00:30
0	Royston	London			2	13	4	4	40	4	4	5	06:15	00:30			2	4	60	10	06:30	00:30			3	2	24	7	07:15	00:30
l ≅	Ashwell & Morden	London			2	8	3	1 (note 4)	13	1	1	3	06:15	00:30			2	1	15	3	06:30	00:30			1 1	1	12	3	07:15	00:30
Щ	Baldock	London			2	12	4	2	20	2	2	4	06:15	00:30			2	2	30	6	06:30	00:30			3	2	24	7	07:15	00:30
	Letchworth	London			2	14	6	4	40	4	4	5	06:15	00:30			2	4	60	10	06:30	00:30			3	2	24	7	07:15	00:30
		London Kings Cross	N/A	N/A	0	0	0	0	0	0	0	2			N/A	N/A	0	0	0	1			N/A	N/A	0	0	0	1		
		London St. Pancras (total) of which	42 35	51	2	14	6	4	40 20	4	4	3			41	46	2	2	30 30	9			45 N/A	46 N/A	3	2	24	6		
	Hitchin	London	33	30	6	20	8	6	60	6	6	10	05:00	00:45	33	33	4	8	120	15	05:00	00:30	IN/A	IN/A	5	4	48	8	07:15	00:30
	Stevenage	London			6	22	9	6	60	6	6	10	05:00	00:45			4	8	120	15	05:00	00:30			5	4	48	8	07:15	00:30
	Knebworth	London			3	9	3	2	20	2	2	4	05:45	00:30			3	4	60	6	06:00	00:30			2	2	24	5	07:15	00:30
	Welwyn North	London			3	6	2	2	20	2	2	4	05:45	00:30			3	4	60	6	06:00	00:30			2	2	24	5	07:15	00:30
ء	Watton-at-Stone	London	40	50	3	6 15	2	2 4	20 40	2 4	2	4	05:45	00:15	46	49	3	2	30	5	06:00	00:15	47	40	1 1	1	12	4	08:30	00:30
<u>\$</u>	Hertford North Bayford	London London	46	53	3	12	4	2	20	2	2	6	05:00 05:45	00:15 00:15	46	49	4 3	4 2	60 30	5	05:00 06:00	00:15 00:15	47	48	5 5	4 2	48 24	8	07:00 07:00	00:30 00:30
×	Cuffley	London			4	15	6	4	40	4	4	6	05:00	00:15			4	4	60	6	05:00	00:15			5	4	48	8	07:00	00:30
D E	Crews Hill	London			3	8	3	2	20	2	2	6	05:45	00:15			3	2	30	5	06:00	00:15			5	2	24	8	07:00	00:30
20	Gordon Hill	London			4	25	10	4	40	4	4	6	05:00	00:15			4	4	60	6	05:00	00:15			5	4	48	8	07:00	00:30
fje	Enfield Chase	London			4	25	10	4	40	4	4	6	05:00	00:15			4	4	60	6	05:00	00:15			5	4	48	8	07:00	00:30
Ξ̈́	Grange Park	London			3	23	8 10	4	40 40	4	4	6	05:45 05:00	00:15			3	4	60	6	06:00	00:15			5	4	48	8	07:00 07:00	00:30
3ar,	Winchmore Hill Palmers Green	London London	1		4	25 25	10	4 4	40 40	4	4	6	05:00	00:15 00:15			4	4	60 60	6	05:00 05:00	00:15 00:15			5 5	4	48 48	8 8	07:00	00:30 00:30
2	Bowes Park	London	<u> </u>	<u> </u>	3	18	7	4	40	4	4	6	05:45	00:15		l	3	4	60	6	06:00	00:15		<u></u>	5	4	48	8	07:00	00:30
otte	Welwyn Garden City	London			6	30	10	6	66	6	6	8	05:00	00:30			5	6	90	10	05:00	00:30			6	4	48	9	07:15	00:30
ď.		London Kings Cross	37		1	6	2	0	3	0	0	1			26	33	0	0	0	2			N/A	N/A	0	0	0	0		
Park, City		Moorgate London St. Pancras	47		2	12	4	4	40	4	4	4			46	48	3	4	60	5			47	47	4	2	24	6		
en P	Hatfield	London St. Pancras London	28	33	3	12 29	4 10	2 6	23 63	2 6	2 6	3 8	05:00	00:30	26	26	2 6	2 6	30 90	3 10	05:00	00:30	26	27	6	2	24 48	3 11	07:15	00:30
I de	Welham Green	London			2	12	4	2	20	2	2	4	06:15	23:30			1	2	30	4	06:45	23:30			4	2	24	6	07:15	23:45
ြဋ္ဌိ	Brookmans Park	London			3	12	4	2	20	2	2	4	05:00	23:30			2	2	30	4	05:00	23:30			4	2	24	6	07:15	23:45
Į,	Potters Bar	London	1		6	29	10	6	63	6	6	8	05:00	00:30			6	6	90	10	05:00	00:30			6	4	48	11	07:15	00:30
xce	Hadley Wood	London			3	12	4	2	20	2	2	4	05:00	23:30			2	4	60	4	05:00	23:30			4	2	24	6	07:15	23:45
>	New Barnet	London			3	24 24	8 8	4 4	40 40	4	4	4	05:00 05:00	23:30			3	4 4	60 60	5 5	05:00	23:30 23:30			4	2 2	24 24	6	07:15 07:15	23:45 23:45
lo u	Oakleigh Park New Southgate	London London			3	24	8	4	40	4	4	4	05:00	23:30 23:30			3	4	60	5	05:00 05:00	23:30			4	2	24	6	07:15	23:45
seo	Alexandra Palace	London			6	35	12	8	80	6	6	10	05:00	00:15			7	8	120	10	05:00	00:15			9	6	72	14	07:00	00:30
Z.	Hornsey	London	1		5	23	8	6	60	6	6	10	05:00	00:15			5	8	120	10	05:00	00:15			9	6	72	14	07:00	00:30
Š	Harringay	London			5	23	8	6	60	6	6	10	05:00	00:15			5	8	120	10	05:00	00:15			9	6	72	14	07:00	00:30
neu	Finsbury Park	London	1		16	68	24	14	153	14	14	19	05:00	00:45			10	14	210	24	05:00	00:15			16	10	120	22	07:00	00:30
느	Drayton Park	Moorgate Moorgate	1		4	37 37	14	8	84	8	8	10	06:00	23:45			4	8	120	10	06:00	00:15 00:15			7 (note 5)	6 (note 5)	72	14	07:30 (note 5)	00:30
S	Highbury & Islington Essex Road	Moorgate Moorgate			4	37	14 14	8	84 84	8 8	8	10 10	06:00 06:00	23:45 23:45			4	8 8	120 120	10 10	06:00 06:00	00:15			7 (note 5) 7 (note 5)	6 (note 5) 6 (note 5)	72 72	14 14	07:30 (note 5) 07:30 (note 5)	00:30 00:30
ш	Old Street	Moorgate			4	37	14	8	84	8	8	10	06:00	23:45			4	8	120	10	06:00	00:15			7 (note 5)	6 (note 5)	72	14	07:30 (note 5)	00:30
																									/	\/				

Note 1: One additional service shall be provided from King's Lynn to Cambridge, calling at all stations, departing King's Lynn no earlier than 23:00.

Note 2: The frequency of service may be reduced to one train per hour in up to two hours of the day between 10:00 and 19:59 Monday to Friday and one hour between 07:00 and 21:59 on a Saturday to allow for pre-existing freight services to operate.

Note 3: The requirement increases to three trains per hour between 09:00 and 11:59 on Saturdays.

Note 4: The requirement increases to two trains per hour arriving in London between 17:00 and 19:59 on weekdays.

Note 5: The minimum requirement at Drayton Park, Highbury & Islington, Essex Road and Old Street before 10:00 on a Sunday morning may be reduced, but only to the extent that the Franchisee (having used all reasonable endeavours) is unable to secure the necessary rights to operate services on the Moorgate branch.

#### MML 1: Midland Mainline Minimum Service Specification and First and Last Train Specification

	TABLE 1: NORTHBOUND							M	londay to Fr	iday								Sat	turday							Sun	day			
	Departure time from London St. Pancras		Journe	ey Time	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service leaves Lone	don Jou	mey time	0200-0659	0700-2159	9 0700-2159	2200-0159	Service lea	ves London	Journ	ney time	0200-0859	0900-2059	0900-2059	2100-0159	Service lea	ves Londor
														Last (r	10							Last (no								Last (no
														First (no earlie	r			1			First (no	earlier							First (no	earlier
	Direct services from London (note 1) to each station	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than) than	Average	e Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	n Total	TPH	Total	Total	later than)	than)
	Kentish Town	London St Pancras			9	6 (note 4)	51	15	4	5	6	5	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
	West Hampstead Thameslink	London St Pancras	1		12	4	43	17	6	6	4	4	13		1		12	4	60	13		01:30	l		5	6	75	13	06:15	, I
	Cricklewood	London St Pancras			8	4	36	12	4	4	4	4	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
	Hendon	London St Pancras			8	4	36	12	4	4	4	4	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
	Mill Hill Broadway	London St Pancras			8	4	37	15	6	4	4	4	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
	Elstree & Borehamwood	London St Pancras			8	4	39	16	6	5	4	4	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
	Radlett	London St Pancras			8	4	39	16	6	4	4	4	7				7	4	60	8		01:30	l		5	2	24	9	06:15	, 1
တ္	St. Albans	London St Pancras	27	37	13	8	75	35	13	10	8	8	13		27	33	12	8	120	13		01:30	28	33	5	6	75	13	06:15	, 1
<u>.</u> . <u></u>		of which	21	25	5	4 (note 5)	36	19	7	6	4	4	6		21	25	4	4	52	2			24	25	0	4	38	2	/	, 1
€	Harpenden	London St Pancras			12	6	54	26	9	8	6	6	13	(note 2)			9	6	90	13	(note 2)	01:30			5	6	75	13	06:15	(note 2)
8	Luton Airport Parkway (note 3)	London St Pancras			12	6	57	23	8	6	6	6	13				12	6	90	13		01:30	l		5	6	75	13	06:15	, 1
₹	Luton	London St Pancras	40	53	12	6	56	28	10	8	6	6	13		42	50	9	6	90	13		01:30	43	48	5	6	75	13	06:15	, 1
		of which	36	40	5	4	38	19	7	6	4	4	6		36	40	4	4	56	2			39	39	0	4	38	2	/	, 1
	Leagrave	London St Pancras			8	4	36	15	5	5	4	4	12				7	4	62	13		01:30			5	4	51	13	06:15	,
	Harlington	London St Pancras			8	4	36	15	5	4	4	4	12				7	4	62	13		01:30	l		5	4	51	13	06:15	, 1
	Flitwick	London St Pancras			8	4	37	22	8	5	4	4	12				7	4	62	13		01:30	l		5	4	51	13	06:15	, 1
	Bedford	London St Pancras	63	75	8	4	38	22	8	6	4	4	12	1	64	75	7	4	62	13	1	01:30	69	75	5	4	51	13	06:15	, 1
		of which	61	70	2	4	37	21	7	6	4	4	5		62	65	2	4	56	2			66	70	0	4	38	2		, 1
		of which	56	60	0	0	6	11	5	3	0	0	0		N/A	N/A							N/A	N/A				1 1		,

Note 1: All services must serve all Thameslink core stations between London St. Pancras and Blackfriars throughout Monday to Sunday, except City Thameslink, which must be served by all services arriving at City Thameslink between 05:00 and 23:00 Monday to Friday, and between 09:00 and 21:00 on Saturdays. Note 2: There must be a minimum of one train per hour, and a maximum interval of 80 minutes between consecutive departures from London St. Pancras to Bedford, throughout the period 22:00-06:59, with the exception of Saturday noming. Note 3: There must be a minimum of two trains per hour arriving at Luton Airport Parkway between 03:00 and 06:59 from Blackfriars on Monday to Saturday morning, with a maximum interval of 40 minutes between consecutive arrivals. Note 4: May be reduced to two trains per hour between 07:00-08:59.

TABLE 2: SOUTHBOUND					·	·		to Friday	·	· ·	·				·			ırday							Sund				
Arrival time at London St. F	Pancras	Journ	ney Time	0200-0659	0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-015	9 Service an	ives London	Journe	ey time	0200-0659	0700-2159	0700-2159	2200-0159	Service arriv	ves London	Journ	ey time	0200-0859	0900-2059	0900-2059	2100-015	9Service ar	rrives Lor
												First (no	Last (no earlier							First (no	Last (no earlier							First (no	Last earli
Direct services to London (note 1) from each		Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than	
Bedford	London St Pancras	59	72	10	19	8	4	46	2	2	7			60	71	9	4	60	6	1 1	00:45	65	75	6	4	48	10	07:00	
	of which	57	60	4	19	7	4	44	2	2	2			58	60	2	4	58	1	1 1		63	65	0	4	44	0		4
	Sevenoaks			1	2	2		2	0	0	0					0		0	0	1 1				0		0	0		
	Gatwick Airport			9	10	3		38	2	2	7					9		60	6	1 1				5		48	9		4
	Brighton			7	6	2		21	2	2	2					6		30	4					4		37	9		
Flitwick	London St Pancras			10	17	6	4	42	2	2	7					9	4	60	6		00:45	l		6	4	48	10	07:00	
Harlington	London St Pancras			9	12	4	4	40	2	2	7	1				9	4	60	6	1 1	00:45	l		6	4	48	10	07:00	
Leagrave	London St Pancras			9	15	6	4	40	2	2	7	1				9	4	60	6	1 1	00:45	l		6	4	48	10	07:00	
Luton	London St Pancras	39	49	16	27	11	6	64	5	4	11			40	48	11	6	90	10	1 1	00:45	43	51	6	6	72	10	07:00	
	of which	34	35	8	19	7	4 (note 3)	44	2	2	2			34	35	2	4	58	5	1 1		39	40	0	4	44	0		4
	Sutton (via Wimbledon)			1	5	2		19	2	2	4					0		28	4	1 1				0		24	1		4
	Sutton (not via Wimbledon)			2	3	1		2	0	0	0					2		4	0	1 1				0		0	0		4
	Sevenoaks			1	4	3		3	1	0	0					0		0	0	1 1				0		0	0		4
	Gatwick Airport			9	10	3		38	2	2	7					9		60	6	1 1				5		48	9		4
	Brighton			7	6	2		21	2	2	2					6		30	4	1 1				4		37	9		4
Luton Airport Parkway	London St Pancras			15	20	7	6	67	5	4	11	(no	te 2)			11	6	90	10	(note 2)	00:45	l		6	6	72	10	07:00	(note
Harpenden	London St Pancras			16	27	11	6	65	5	4	11					11	6	90	10		00:45	l		6	6	72	10	07:00	
St. Albans	London St Pancras	26	36	16	35	15	8	88	7	4	11			27	34	11	8	120	10	1 1	00:45	28	36	6	6	72	10	07:00	
	of which	21	25	9	21	8	4	48	5	2	3			20	25	2	4	58	5	1 1		24	25	0	4	44	0		4
	Sutton (via Wimbledon)			1	6	3		20	2	2	4					0		28	4	1 1				0		24	1		4
	Sutton (not via Wimbledon)			2	6	2		20	2	0	0					2		28	0	1 1				0		0	0		4
	Sevenoaks			1	7	4		5	1	0	0					0		0	0	1 1				0		0	0		4
	Gatwick Airport			9	10	3		38	2	2	7					9		60	6	1 1				5		48	9		4
	Brighton			7	6	2		21	2	2	2					6		30	4	1 1				4		37	9		4
Radlett	London St Pancras			7	17	8	4	43	5	2	8	1				7	4	60	7	1 1	00:45	l		5	2	24	10	07:00	
Elstree & Borehamwood	London St Pancras			7	17	8	4	43	5	2	8	1				7	4	60	7	1 1	00:45	l		5	2	24	10	07:00	
Mill Hill Broadway	London St Pancras			7	15	7	4	40	2	2	8	1				7	4	60	7	1 1	00:45	l		5	2	24	10	07:00	
Hendon	London St Pancras			7	10	4	4	40	2	2	8	1		1	l	7	4	60	7	1	00:45	I	1	5	2	24	10	07:00	
Cricklewood	London St Pancras			7	10	4	4	40	2	2	8	1		1	l	7	4	60	7	1	00:45	I	1	5	2	24	10	07:00	
West Hampstead	London St Pancras			16	18	8	4	49	7	4	11	1		1	l	11	4	62	10	1	00:45	I	1	6	6	72	10	07:00	
Kentish Town	London St Pancras	1	1	9	11	4	6 (note 4)	53	3	2	8	1			l	9	4	60	7	1 1	00:45	I	1	5	2	24	10	07:00	

Note 1: All services must serve all Thameslink core stations between London St. Pancras and Blackfriars throughout Monday to Sunday, except City Thameslink, which must be served by all services arriving at City Thameslink between 05:00 and 23:00 Monday to Friday, and between 09:00 and 21:00 on Saturdays. Note 2: There must be a minimum of one train per hour, and a maximum interval of 80 minutes between consecutive arrivals at London St. Pancras from Bedford, throughout the period 22:00-06:59, with the exception of Saturday night / Sunday morning. Note 3: May be reduced to three trains per hour between 17:00-18:59.

Note 4: May be reduced to four trains per hour between 18:00-18:59 and one train per hour between 19:00-19:59.

## MML 2: Midland Mainline Minimum Service Specification and First and Last Train Specification

	TABLE 1: NORTHBOUND							M	onday to Fri	day								Satu	ırday							Sun	iday		•	
	Departure time from London St. Pancras		Journe	ey Times	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service leaves Lond	on Jour	ney Times	0200-0659	0700-2159	0700-2159	2200-0159	Service lea	ives London	Journe	ey Times	0200-0859	0900-2059	0900-2059	2100-0159	Service leav	es Londor
														Last (n								Last (no						i T		Last (no
														First (no earlier	1		1				First (no	earlier						. '	First (no	earlier
	Direct services from London (note 1) to each station	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than) than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	Kentish Town	London St. Pancras			9	6 (note 3)	57	12	4	6	6	6	12		1		7	6	90	15		01:30			6	4	48	9	06:15	
	West Hampstead Thameslink	London St. Pancras			9 (note 4)	4	57	22	8	6	6	6	12		1		9 (note 4)	4	60	13		01:30			6	8	96	12	06:15	
	Cricklewood	London St. Pancras			8	4	36	22	8	6	4	4	7		1		7	4	60	10		01:30			6	4	48	9	06:15	
	Hendon	London St. Pancras			8	4	36	22	8	6	4	4	7		1		7	4	60	10		01:30			6	4	48	9	06:15	
	Mill Hill Broadway	London St. Pancras			8	4	36	22	8	6	4	4	7		1		7	4	60	10		01:30			6	4	48	9	06:15	
	Elstree & Borehamwood	London St. Pancras			8	4	36	22	8	6	4	4	7		1		7	4	60	10		01:30			6	4	48	9	06:15	
	Radlett	London St. Pancras			8	4	36	22	8	6	4	4	7		1		7	4	60	10		01:30			6	4	48	9	06:15	
98	St. Albans	London St. Pancras	26	35	11 (note 4)	8	84	44	16	12	8	8	12		26	34	9 (note 4)	8	120	19		01:30	29	33	6	8	96	12	06:15	
.≅		of which	19	23	3	4	42	22	8	6	4	4	5	(note 2)	18	18	2 (note 4)	4	60	9	(note 2)		24	24	0	4	48	3	1	(note 2)
S	Harpenden	London St. Pancras			10	6	60	32	12	9	6	6	12	(Hote 2)	1		9	6	90	15	(Hote 2)	01:30			6	6	72	12	06:15	(Hote 2)
₹	Luton Airport Parkway	London St. Pancras	N/A	30 (note 5)	) 10 (note 4)	6	63	32	12	9	6	6	12		N/A	30 (note 5	9 (note 4)	6	90	15		01:30	N/A	N/A	6	6	72	12	06:15	
	Luton	London St. Pancras	37	48	10	6	63	32	12	9	6	6	12		38	48	9	6	90	15		01:30	43	48	6	6	72	12	06:15	
		of which	33	36	3	4	42	22	8	6	4	4	5		32	32	2	4	60	9			39	39	0	4	48	3	1	
	Leagrave	London St. Pancras			8	4	38	22	8	6	4	4	12				7	4	60	13		01:30			5	4	48	11	06:15	
	Harlington	London St. Pancras			8	4	36	18	6	6	4	4	12		1		7	4	60	13		01:30			5	4	48	11	06:15	
	Flitwick	London St. Pancras			8	4	38	22	8	6	4	4	12		1		7	4	60	13		01:30			5	4	48	11	06:15	
	Bedford	London St. Pancras	57	74	8	4	41	22	8	6	4	4	12		57	57	7	4	60	13	1	01:30	68	75	5	4	48	12	06:15	
		of which	57	60	3	4	41	22	8	6	4	4	7		N/A	N/A							66	66	0	4	48	3	(	

Note 1: All services must serve all Thameslink core stations between London St. Pancras and Blackfriars throughout Monday to Sunday, except City Thameslink, which must be served by all services arriving at City Thameslink between 05:00 and 23:00 Monday to Friday, and between 09:00 and 21:00 on Saturdays. Note 2: There must be a minimum of one train per hour, and a maximum interval of 80 minutes between consecutive departures from London St. Pancras to Bedford, throughout the period 22:00-06:59, with the exception of Saturday morning.

Note 3: May be reduced to four trains per hour between 07:00-08:59.

Note 4: There must be a minimum of two trains per hour arriving at Luton Airport Parkway between 03:00 and 06:59 from Blackfriars on Monday to Saturday morning, with a maximum interval of 40 minutes between consecutive arrivals, until the Passenger Change Date in December 2018.

Note 5: Four trains per hour shall have a maximum journey time of 30 minutes.

	TABLE 2: SOUTHBOUND							Monday	to Friday									Satu	ırday				I			Sur	nday			
	Arrival time at London St. Pancras		Journe	ey Times	0200-0659	0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-0159	9 Service ar	rives London	Journe	y Times	0200-0659	0700-2159	0700-2159	2200-0159	Service arri	ves London	Journe	y Times	0200-0859	0900-2059	0900-2059	2100-0159	Service arriv	ves Londor
														Last (no								Last (no								Last (no
													First (no	earlier							First (no	earlier								
	Direct services to London (note 1) from each station			Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)		Maximum	Total	TPH	Total	Total	later than)	than)
	Bedford	London St. Pancras	58	72	10	24	7	4	40	4	4	12			57	57	9	4	60	9		00:45	65	72	5	4	48	9	07:00	ı
		of which	57	60	7	24	7	4	40	4	4	10			N/A	N/A							62	62	1	4	48	0		ı I
	Flitwick	London St. Pancras			10	18	6	4	40	4	4	12					9	4	60	9		00:45			5	4	48	9	07:00	
	Harlington	London St. Pancras			10	12	4	4	40	4	4	12					9	4	60	9		00:45			5	4	48	9	07:00	
	Leagrave	London St. Pancras			10	18	6	4	40	4	4	12					9	4	60	9		00:45			5	4	48	9	07:00	
	Luton	London St. Pancras	38	48	13	36	12	6	64	6	6	16			38	44	11	6	90	12		00:45	43	48	5	6	72	10	07:00	ı I
		of which	34	38	7	24	8	4	40	4	4	10			34	34	4	4	60	6			38	38	1	4	48	0	4 1	ı I
Se Se	Luton Airport Parkway	London St. Pancras	N/A	30 (note 4)	13	36	12	6	64	6	6	16			N/A	30 (note 4)	) 11	6	90	12		00:45	N/A	N/A	5	6	72	10	07:00	
ĕ	Harpenden	London St. Pancras			13	36	12	6	64	6	6	16	(2)	ote 2)			11	6	90	12	(note 2)	00:45			5	6	72	10	07:00	(note 2)
S	St. Albans	London St. Pancras	26	33	18	48	16	8	88	4	8	20	(11)	ne 2)	26	30	11	8	120	15	(Hote 2)	00:45	28	33	5	8	96	10	07:00	(Hote 2)
		of which	21	23	7	24	8	4	40	4	4	10			21	21	4	4	60	6			23	23	1	4	48	0	4 1	ı I
_ `	Radlett	London St. Pancras			12	24	8	4	48	5	4	10					7	4	60	9		00:45			4	4	48	10	07:00	
	Elstree & Borehamwood	London St. Pancras			12	24	8	4	48	5	4	10					7	4	60	9		00:45			4	4	48	10	07:00	
	Mill Hill Broadway	London St. Pancras			12	24	8	4	48	5	4	10					7	4	60	9		00:45			4	4	48	10	07:00	
	Hendon	London St. Pancras			9	24	8	4	48	5	4	10					7	4	60	9		00:45			4	4	48	10	07:00	
	Cricklewood	London St. Pancras			9	24	8	4	48	5	4	10					7	4	60	9		00:45			4	4	48	10	07:00	
1	West Hampstead	London St. Pancras	I		14	24	8	4	68	5	4	10	1		1	l	12	4	60	10		00:45			5	8	96	10	07:00	
	Kentish Town	London St. Pancras	I		13	12	4	6 (note 3)	65	5	6	14					9	6	90	12		00:45			4	4	48	10	07:00	

Note 1: All services must serve all Thameslink core stations between London St. Pancras and Blackfriars throughout Monday to Sunday, except City Thameslink, which must be served by all services arriving at City Thameslink between 05:00 and 23:00 Monday to Friday, and between 09:00 and 21:00 on Saturdays. Note 2: There must be a minimum of one train per hour, and a maximum interval of 50 minutes between consecutive arrivals at London St. Pancras from Bedford, throughout the period 22:00-06:59, with the exception of Saturday night / Sunday morning. Note 3: May be reduced to five trains per hour between 19:00-19:59.

Note 4: Four trains per hour shall have a maximum journey time of 30 minutes.

# South East London and Kent 1: Minimum Service Specification and First and Last Train Specification

	TABLE 1: SOUTHBOUND							M	londay to Fr										Satu	ırday								nday			
	Departure time from London Blackfriars		Journ	ey Time	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service lea	ves London	Journe	y Time	0200-0659	0700-2159	0700-2159	2200-0159	Service lea	ves Londor	n Journ	ney Time	0200-0859	0900-2059	0900-2059	2100-0159	Service lea	es London
															Last (no								Last (no								Last (no
				1	l 1									First (no	earlier	l				l I		First (no	earlier						1 '	First (no	earlier
	Direct services from London (note 1) to each station	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	Elephant & Castle	Blackfriars (note 1)			4	2	22	10	3	2	2	2	4	05:30	23:30			2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Herne Hill	Blackfriars (note 1)			0	0	0	2	1	0	0	0	0	17:00	17:45			0	0	0	0	-	-			0	0	0	0	-	-
	West Dulwich	Blackfriars (note 1)	l	1	0	0	0	2	1	0	0	0	0	17:00	17:45	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Sydenham Hill	Blackfriars (note 1)		1	0	0	0	2	1	0	0	0	0	17:00	17:45	l		0	0	0	0	-	-			0	0	0	0	-	-
	Penge East	Blackfriars (note 1)		1	0	0	0	2	1	0	0	0	0	17:00	17:45	l		0	0	0	0	-	-			0	0	0	0	-	-
	Kent House	Blackfriars (note 1)	l	1	0	0	1	2	1	0	0	0	0	08:30	17:45	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Beckenham Junction	Blackfriars (note 1)			0	0	1	1	0	0	0	0	0	07:45	16:45			0	0	0	0	-	-			0	0	0	0	-	-
		Blackfriars (note 1)			4	2	21	8	2	2	2	2	4	05:30	23:30			2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Peckham Rye	Blackfriars (note 1)		1	4	2	21	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
<u>~</u>	Nunhead	Blackfriars (note 1)	l	1	4	2	21	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
5	Crofton Park	Blackfriars (note 1)		1	4	2	20	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
8	Catford	Blackfriars (note 1)	l	1	4	2	21	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
.≘	Bellingham	Blackfriars (note 1)		1	4	2	21	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
Se	Beckenham Hill	Blackfriars (note 1)	l	1	3	2	18	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
Je Je	Ravensbourne	Blackfriars (note 1)	l	1	4	2	20	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
=	Shortlands	Blackfriars (note 1)		1	4	2	20	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
t t	Bromley South	Blackfriars (note 1)	l	1	3	2	18	8	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
3	Bickley	Blackfriars (note 1)		1	4	2	20	7	1	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Petts Wood	Blackfriars (note 1)			0	0	1	1	0	0	0	0	0	08:00	16:15			0	0	0	0	-	-			0	0	0	0	-	-
	Orpington	Blackfriars (note 1)			0	0	1	1	0	0	0	0	0	08:00	16:15			0	0	0	0	-	-			0	0	0	0	-	-
	St. Mary Cray	Blackfriars (note 1)			4	2	19	6	2	2	2	2	4	05:30	23:30			2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Swanley	Blackfriars (note 1)	l	1	4	2	19	7	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
	Eynsford	Blackfriars (note 1)		1	4	2	18	7	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Shoreham	Blackfriars (note 1)		1	4	2	18	7	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30			3	2	24	6	07:45	23:30
	Otford	Blackfriars (note 1)	l	1	4	2	19	7	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
	Bat & Ball	Blackfriars (note 1)	l	1	4	2	18	7	2	2	2	2	4	05:30	23:30	l		2	2	30	4	06:15	23:30	1		3	2	24	6	07:45	23:30
	Sevenoaks	Blackfriars (note 1)	63	74	4	2	19	7	2	2	2	2	4	05:30	23:30	61	61	2	2	30	4	06:15	23:30	61	61	3	2	24	6	07:45	23:30
Ş		Blackfriars (note 1)			0	0	6 (note 3)	6	2	2	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	-
ō	Bromley South	Blackfriars (note 1)	l	1	0	0	6 (note 3)	6	2	2	0	0	0	-	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
ĕ	Bickley	Blackfriars (note 1)		1	0	0	6 (note 3)	6	2	2	0	0	0	-	-	l		0	0	0	0	-	-			0	0	0	0	-	-
Ξ		Blackfriars (note 1)	l	I	0	0	6 (note 3)	6	2	2	0	0	0	-	-	I		0	0	0	0	-	-	1		0	0	0	0	-	-
Se	Swanley	Blackfriars (note 1)	l	1	0	0	6 (note 3)	6	2	2	0	0	0	-	-			0	0	0	0	-	-	1		0	0	0	0	-	-
ter	Otford	Blackfriars (note 1)	l	I	0	0	6 (note 3)	6	2	2	0	0	0	-	-	I		0	0	0	0	-	-	1		0	0	0	0	-	-
l õ	Borough Green & Wrotham	Blackfriars (note 1)	l	1	0	0	6 (note 3)	6	2	2	0	0	0	-	-			0	0	0	0	-	-	1		0	0	0	0	-	-
Ę		Blackfriars (note 1)	l	I	0	0	6 (note 3)	6	2	2	0	0	0	-	-	I		0	0	0	0	-	-	1		0	0	0	0	-	-
χ	Maidstone East (note 2)	Blackfriars (note 1)	70	70	0	0	6 (note 3)	6	2	2	0	0	0	-	-	N/A	N/A	0	0	0	0	-	-	N/A	N/A	0	0	0	0	-	-

Note 1: All Kent Inner services must serve all Thameslink core stations between London St. Pancras and London Blackfriars throughout Monday to Friday (with a maximum journey time of 12 minutes), but may originate at London Blackfriars at weekends. Note 2: Outer services serving stations between Elephant & Castle and Maidstone East are required to operate from January 2018 only, and are permitted to originate at Blackfriars.

Note 3: Two of these services must operate from London St. Pancras, departing London Blackfriars between 15:00 and 15:59.

		TABLE 2: NORTHBOUND							Monday	to Friday									Sati	ırdav							Sui	ndav			
		Arrival time at London Blackfriars		lourn	ev Time	0200-0659	0700-0959	0800-0850	1000-1959		2000-2059	2100-2159	2200-0159	Service arr	ives London	lourne	ev Time	0200-0659			2200-0159	Service arr	ives London	lourne	ev Time	0200-0859			2100-0150	Service arr	ives London
		Annvar time at London blackmans		Journ	I	0200-0003	0700-0333	0000-0008	1000-1909	1000-1303	2000-2003	2100-2133	2200-0133	Service arri	Last (no	Journe	I	0200-0033	0700-2133	0700-2133	2200-0133	Service arr	Last (no	Journe	I	0200-0033	0300-2033	0300-203	2100-0133	Service arr	Last (no
							1		1		1			First (no	earlier							First (no	earlier							First (no	earlier
		Direct services to London (note 1) from each station	Destination	Avorago	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
_		Maidstone East (note 2)				10tai	TOTAL	10(a)	IFII		10tai	0	0	,		N/A	N/A	10tai		10tai	O	iatei tiiaii)		N/A	N/A		IFII	10tai	10tai	iatei tiiaii)	lilali)
	S		Blackfriars (note 1)	72	72	0	0	2	0	8 (note 3)	0	0	0	-	-	IN/A	IN/A	0	0	0	0	-	-	IN/A	IN/A	0	0	0	0	· -	· ·
	8	West Malling Borough Green & Wrotham	Blackfriars (note 1)			0	6	2 2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	· ·
	2		Blackfriars (note 1)			0	0	_	0	8 (note 3)	0	ľ	ı ~		- 1			0	1 -	0	0	-	-			0	0	0	0	· -	· ·
	ຼ ≥	Otford	Blackfriars (note 1)			0	ь	2	ı ~	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	· ·
	only only	Swanley	Blackfriars (note 1)			0	6	2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	- 1
	ō .	St. Mary Cray	Blackfriars (note 1)			0	6	2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	- 1
	i.	Bickley	Blackfriars (note 1)			0	6	2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	- 1
	ž	Bromley South	Blackfriars (note 1)			0	6	2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	- 1
		Elephant & Castle	Blackfriars (note 1)			0	6	2	0	8 (note 3)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	-
		Sevenoaks	Blackfriars (note 1)	63	66	1	7	3	2	21	2	2	4	06:45	23:30	62	62	1	2	30	4	06:45	23:30	62	62	2	2	24	5	08:30	23:15
			St Albans			0	4	0	0	7	0	0	0					0	0	0	0					0	0	0	0		
			Luton			0	2	0	0	6	0	0	0					0	0	0	0					0	0	0	0		
			Bedford			0	1	0	0	6	0	0	0					0	0	0	0					0	0	0	0		
		Bat & Ball	Blackfriars (note 1)			1	7	3	2	21	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		Otford	Blackfriars (note 1)			1	7	3	2	21	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		Shoreham	Blackfriars (note 1)			1	7	3	2	20	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		Eynsford	Blackfriars (note 1)			1	7	3	2	20	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		Swanley	Blackfriars (note 1)			1	7	3	2	21	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		St. Mary Cray	Blackfriars (note 1)			1	7	3	2	21	2	2	4	06:45	23:30			1	2	30	4	06:45	23:30			2	2	24	5	08:30	23:15
		Orpington	Blackfriars (note 1)	41	44	3	4	2	0	1	0	0	0	05:30	18:00	N/A	N/A	0	0	0	0	-	-	N/A	N/A	2	0	0	0	07:30	07:45
			St Albans			0	3	0	0	1	0	0	0					0	0	0	0					0	0	0	0		
			Luton			0	2	0	0	1	0	0	0					0	0	0	0					0	0	0	0		
	>		Bedford			0	0	0	0	1	0	0	0					0	0	0	0					0	0	0	0		
	Ę.	Petts Wood	Blackfriars (note 1)			3	4	2	0	1	0	0	0	05:30	18:00			0	0	0	0	-	-			2	0	0	0	07:30	07:45
	S	Bickley	Blackfriars (note 1)	1	1	4	11	5	2	21	2	2	4	05:30	23:30		1	1	2	30	4	06:45	23:30	1		4	2	24	5	07:30	23:15
	<u>.</u>	Bromley South	Blackfriars (note 1)			4	13	6	2	22	2	2	4	05:30	23:30			1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	2	Shortlands	Blackfriars (note 1)			4	13	6	2	20	2	2	4	05:30	23:30			1 1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	Š	Ravensbourne	Blackfriars (note 1)			3	8	3	2	20	2	2	4	05:30	23:30			1 1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	in e	Beckenham Hill	Blackfriars (note 1)			3	8	3	2	20	2	2	4	05:30	23:30			1 1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	=	Bellingham	Blackfriars (note 1)			3	11	5	2	21	2	2	4	05:30	23:30			1 1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	e.	Catford	Blackfriars (note 1)			3	11	5	2	21	2	2	4	05:30	23:30			I i I	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
	소	Crofton Park	Blackfriars (note 1)			3	8	3	2	20	2	2	4	05:30	23:30			1 1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
		Nunhead	Blackfriars (note 1)			3	8	3	2	21	2	2	4	05:30	23:30			1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
		Peckham Rye	Blackfriars (note 1)	1		3	10	4	2	21	2	2	4	05:30	23:30		l	I i I	2	30	4	06:45	23:30	l		4	2	24	5	07:30	23:15
		Denmark Hill	Blackfriars (note 1)			3	11	5	2	21	2	2	4	05:30	23:30			1	2	30	4	06:45	23:30			4	2	24	5	07:30	23:15
		Beckenham Junction	Blackfriars (note 1)	24	28	1	3	2	0	2	0	0	0	05:45	18:00	N/A	N/A	0	0	0	0	-	-	N/A	N/A	0	0	0	0	-	
			St Albans			0	3	0	0	0	0	0	0					0	0	0	0					0	0	0	0		
			Luton			o o	2	2	0	0	0	0	0					o o	0	0	0					0	0	0	0		
			Bedford			0	1	1	0	2	0	0	0					0	0	0	0					0	0	0	0		
		Kent House	Blackfriars (note 1)			1	4	2	0	1	0	0	0	05:45	17:45			0	0	0	0					0	0	0	0		
		Penge East	Blackfriars (note 1)			1 1	4	2	I 0	Ιi	l ő	0	0	05:45	17:45			I ŏ l	0	0	0		l -			I 0	0	0	0	_	. <b> </b>
		Sydenham Hill	Blackfriars (note 1)				4	2	I ŏ	l i	I ŏ	l ŏ	0	05:45	17:45			I ŏ l	l ő	l ő	l ő	l .	l .			I ŏ	1 0	I 0	l ő	١.	. <b> </b>
		West Dulwich	Blackfriars (note 1)				1 7	2	1 0		1 0	١	0	05:45	17:45			0	0	١٠	١	1 [	l I			1 0	0	١٠	0		[
		Herne Hill	Blackfriars (note 1)				1 4	2	1 0		1 0	0	0	05:45	17:45			0	0	1 0	0	1 [	1 [			0	0	1 0	0		[
		Elephant & Castle		1	-	1	15	7	1 2	22	1 2	2	4				-	1	2	20	4	06:45	22:20	-	-		2	24		07:20	23:15
		Elephani α Castie	Blackfriars (note 1)	1	1	4	15	7	2	23	2	2	4	05:30	23:30		1	1	2	30	4	06:45	23:30	ı	1	4	2	24	5	07:30	23:15

Note 1: All Kent Inner services must serve all Thameslink core stations between London Blackfriars and London St. Pancras throughout Monday - Friday (with a maximum journey time of 10 minutes), but may terminate at London Blackfriars at weekends. Note 2: Outer services serving stations between Maidstone East and Elephant & Castle shall operate from January 2018 only, and are permitted to terminate at Blackfriars.

Note 3: Two trains per hour must operate to London St. Pancras arriving at London Blackfriars between 10:00 and 11:59.

# South East London and Kent 2: Minimum Service Specification and First and Last Train Specification

		TABLE 1: SOUTHBOUND	Г						M	onday to Fri	day						I			Satu	urday				l			Su	nday			
		Departure time from London Blackfriars	Ī	Journe	ey Times	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service leav	ves London	Journe	/ Times	0200-0659	0700-2159	0700-2159	2200-0159	Service lea	ves London	Journe	ey Times	0200-0859	0900-2059	9 0900-2059	2100-0159	3 Service les	aves London
			ì													Last (no								Last (no							1	Last (no
															First (no	earlier				l .			First (no	earlier							First (no	earlier
		Direct services from London (note 1) to each station	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
			Blackfriars (note 1)			4	2	20	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
			Blackfriars (note 1)			4	2	20	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Peckham Rye	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
			Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Crofton Park	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
	⋛	Catford	Blackfriars (note 1)			4	2	20	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
1 3	ō	Bellingham	Blackfriars (note 1)			4	2	20	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
	ĕ	Beckenham Hill	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
1	ž	Ravensbourne	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
	se	Shortlands	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
3	ē	Bromley South	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
3	≧	Bickley	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
1 3	Ĕ	St. Mary Cray	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
2	ž	Swanley	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Eynsford	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Shoreham	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Otford	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Bat & Ball	Blackfriars (note 1)			4	2	18	6	2	2	2	2	4	05:30	23:30			3	2	30	4	06:00	23:30			3	2	24	6	07:45	23:30
		Sevenoaks	Blackfriars (note 1)	61	63	4	2	18	6	2	2	2	2	4	05:30	23:30	60	60	3	2	30	4	06:00	23:30	61	61	3	2	24	6	07:45	23:30
			Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	T -	-
es		Bromley South	Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	- 1
ĕ		Bickley	Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	1 - 1	-			0	0	0	0	-	-			0	0	0	0	-	- 1
Ser	_	St. Mary Cray	Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	1 - 1	-			0	0	0	0	-	-			0	0	0	0	-	-
9	Ę	Swanley	Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	l - I	-			0	0	0	0	-	-			0	0	0	0	-	- 1
Ę	0	Otford	Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	l - I	-			0	0	0	0	-	-			0	0	0	0	-	- 1
=			Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	1 - 1	-	1		0	0	0	0	-	-	l	1	0	0	0	0	-	1 - 1
è			Blackfriars (note 1)			0	0	6 (note 2)	6	2	2	0	0	0	-	-			0	0	0	0	-	-	l		0	0	0	0	-	-
1 -			Blackfriars (note 1)	70	70	0	0	6 (note 2)	6	2	2	0	0	0	1 - 1	-	N/A	N/A	0	0	0	0	-	-	N/A	N/A	0	0	0	0	-	- 1

Note 1: All services must serve all Thameslink core stations between London Blackfriars and London St. Pancras throughout Monday to Saturday, but may originate at London Blackfriars on Sundays. Note 2: Two of these services must operate from London St. Pancras, departing London Blackfriars between 15:00-15:59.

	TARLE & MORTUROUND							Manda	to Eddan									0-1	les							0	a dans			
	TABLE 2: NORTHBOUND		<u> </u>						to Friday							_			urday				<u> </u>	_			nday			
	Arrival time at London Blac	friars	Journ	ey Times	0200-0659	0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-0159	Service arr		Journe	y Times	0200-0659	0700-2159	0700-2159	2200-0159	Service arr		Journe	y Times	0200-0859	0900-2059	0900-2059	2100-0159	Service arri	
			1.			l							First (no	Last (no earlier	<b>.</b>						First (no	Last (no earlier							First (no	Last (no earlier
	Direct services to London (note 1) from each s				Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)		Maximum	Total	TPH	Total	Total	later than)	than)
(0	Maidstone East	Blackfriars (note 1)	72	72	0	6	2	0	8 (note 2)	0	0	0	-	-	N/A	N/A	0	0	0	0	-	-	N/A	N/A	0	0	0	0	-	· - I
ë	West Malling	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	· - I
- ≥	Borough Green & Wrotham	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
s >	Otford	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
e e	Swanley	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
0 0	St. Mary Cray	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
ŧ	Bickley	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
\$	Bromley South	Blackfriars (note 1)	1		0	6	2	0	8 (note 2)	0	0	0	-	-	l	1	0	0	0	0	-	-			0	0	0	0	-	- 1
	Elephant & Castle	Blackfriars (note 1)			0	6	2	0	8 (note 2)	0	0	0	-	-			0	0	0	0	-	-			0	0	0	0	-	-
	Sevenoaks	Blackfriars (note 1)	62	62	1	6	2	2	20	2	2	4	06:45	23:30	61	61	1	2	30	6	06:45	00:30	62	62	4	2	24	5	07:30	23:15
	Bat & Ball	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
	Otford	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
	Shoreham	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
	Eynsford	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
≥	Swanley	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
9	St. Mary Cray	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
es	Bickley	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
ĕ	Bromley South	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
Se	Shortlands	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
ē	Ravensbourne	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
_	Beckenham Hill	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
Ħ	Bellingham	Blackfriars (note 1)	1		1	6	2	2	22	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
\$	Catford	Blackfriars (note 1)	1		1	6	2	2	22	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30			4	2	24	5	07:30	23:15
	Crofton Park	Blackfriars (note 1)	1		I 1	6	2	2	20	2	2	4	06:45	23:30	l		1 1	2	30	6	06:45	00:30	I		4	2	24	5	07:30	23:15
	Nunhead	Blackfriars (note 1)	1		1	6	2	2	20	2	2	4	06:45	23:30	l		1	2	30	6	06:45	00:30	I		4	2	24	5	07:30	23:15
	Peckham Rve	Blackfriars (note 1)	1		I 1	6	2	2	20	2	2	4	06:45	23:30	l	1	1	2	30	6	06:45	00:30	I		4	2	24	5	07:30	23:15
	Denmark Hill	Blackfriars (note 1)	1		1 1	6	2	2	22	2	2	4	06:45	23:30	l		1	2	30	6	06:45	00:30	I		4	2	24	5	07:30	23:15
	Elephant & Castle	Blackfriars (note 1)	1		I i	6	2	2	22	2	2	4	06:45	23:30	l		Ιi	2	30	6	06:45	00:30	I		4	2	24	5	07:30	23:15
	1	Diagramate (Hote 1)			<u> </u>						_	<u> </u>	00.10	20.00			<u> </u>		- 00		00.10	00.00			<del></del>			Ü	01.00	200

Note 1: All services must serve all Thameslink core stations between London Blackfriars and London St. Pancras throughout Monday to Saturday, but may terminate at London Blackfriars on Sundays. Note 2: Two trains per hour must operate to London St. Pancras arriving at London Blackfriars between 10:00-11:59.

BML Inner 1: Brighton Main Line Minimum Service Specification (Inner Services only) and First and Last Train Specification

Direct services from specified origin to each station ersea Park sham Junction laws of the man of the man distriction adsworth Common	0					1	i .																Last (no							
ersea Park bham Junction		Averson	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	First (no later than)	Last (no earlier than)	Last (Fri nights only)	Averson	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)	Averson	Maximum	Total	TPH	Total	Total	First (no later than)
	Origin London Victoria	Average	Maximum	7	8	72	25	8	7	8	8	19	06:00	00:30	only)	Average	Maximum	1	8	120	23	06:45	00:30	Average	Maximum	8	4	48	19	07:00
	London Victoria London Victoria			25 8	24 10	211 90	73 26	24 8	20 7	20 8	20 8	43 19	02:00 06:00	01:00 00:30				14 3	24 10	360 150	53 25	02:00 06:30	01:00 00:30			31 8	18 4	216 48	47 20	02:00 07:00
lam I Phant & Castle I	ondon Victoria ondon Blackfriars			10 4	12 4	108 36	34 12	11 4	10 4	12 4	12 4	25 5	06:00 06:00	00:30 23:45				4	12 4	180 60	29 4	06:30 06:00	00:30 23:45			17	10	120 48	29 1	06:45 09:00
	London Blackfriars London Blackfriars			3	4	36 36	12 12	4	4	4	4	3	06:15 06:00	23:15 23:45				1	4	60 60	3	07:00 06:00	23:15 23:45			1	4	48 48	1	09:00
h Bermondsey I	ondon Bridge			7	4	42	18	6	4	4	4	8	06:00	00:00	00:30			1	4	60	10	06:45	00:30			7	4	48	13	07:30
	ondon Bridge ondon Bridge			7	4	42 42	18 18	6	4	4 4	4	8	06:00 06:00	00:00 00:00	00:30 00:30			1	4	60 60	10 10	06:45 06:45	00:30 00:30			7	4 4	48 48	13 13	07:30 07:30
Dulwich I n Dulwich I	London Bridge London Bridge			7	4	42 42	18 18	6	4	4 4	4	8 8	06:00 06:00	00:00 00:00	00:30 00:30			1 1	4 4	60 60	10 10	06:45 06:45	00:30 00:30			7 7	4	48 48	13 13	07:30 07:30
Hill (note 1)	ondon			11	8	78	30	10	10	8	6	13	06:00	00:00	00:30			5	8	120	14	06:00	00:30			8	8	96 48	14	07:30
kley	London Bridge London Bridge			6	4	55 37	12	4	4	4	4	10	05:45 05:45	00:30 00:30				2	4	90 60	12 10	06:30 06:30	00:30 00:30			7	4	48	11	07:30 07:30
	London Bridge London Bridge			6	4	37 37	12 12	4 4	4	4 4	4	10 10	05:45 05:45	00:30 00:30				2 2	4 4	60 60	10 10	06:30 06:30	00:30 00:30			7 7	4 4	48 48	11 11	07:30 07:30
enham stal Palace (via Forest Hill) [	ondon Bridge ondon Bridge			6	4	37 18	12	4	4	4	4	10	05:45 06:30	00:30 23:45				2	4	60 30	10	06:30 07:30	00:30 23:45			7	4	48	11	07:30
atham Hill (note 1)	.ondon			5	4	36	12	4	4	4	4	8	05:30	23:45	00:30			1	4	60	11	05:30	00:30			4	2	24	9	07:30
	_ondon _ondon			6	6	54 54	18 18	6	8	6	6	12 12	06:15 06:15	00:00 00:00	00:30 00:30			0	6	90 90	16 16	07:15 07:15	00:30 00:30			7	4	48 48	16 16	07:30 07:30
	London London Victoria	28	31	6	6 4	54 36	18 12	6	8 4	6	6 4	12 8	05:45	00:00	00:30	27	28	0	6	90 60	16 11	07:15	00:30	28	30	7	4 2	48 24	16 9	07:30
Į.	ondon Bridge ondon Bridge	29	34	2	2	18	6	2	4	2	2	4	06:30	22:15		28	29	0	2	30	5	07:30	22:15	27	28	3	2	24	7	
kenham Junction	ondon Bridge	36	39	2	2	18	6 6	2 2	2	2 2	2 2	1	06:30	22:15		36	36	0	2	30 30	1	07:30	22:15			0	0	0	0	- 1
	_ondon			9	6 8	57 72	24 22	8 6	6 5	5 6	5 6	9 15	06:00 06:00	23:45 00:30	00:15			5 4	6 8	90 120	9 19	06:00 06:30	00:15 00:30			5 13	6	72 72	7 21	07:30 06:45
	London			6	8	72	22	6	5	6	6	15	06:00	00:30 00:30				4	8	120	19	06:30	00:30 00:30			13	6	72	21	06:45
urst	ondon ondon			19	8	72 72	22 22	6	5	6	6	15 19	06:00 02:00	01:00				14	8	120 120	19 23	06:30 02:00	01:00			13 18	6	72 72	21 24	06:45 02:00
ge West I rley I	London London			4 4	2 2	19 19	6	2 2	2 2	2 2	2 2	6	05:45 05:45	00:30 00:30				2 2	2 2	30 30	6 6	06:30 06:30	00:30 00:30			4 4	2 2	24 24	8	07:30 07:30
	ondon London			7 (note 3) 36	10 26	90 231		) 8 (note 4)	9 23	6 19	6 19	17 44	05:30 02:00	00:30				5	10 24	150 361	23 44	05:30 02:00	00:30			14 34	8 20	96 240	26 57	07:30
th Croydon I	London			4	4	37	20	6	4	4	4	9	05:30	00:00	00:15			4	4	60	10	05:30	00:15			4	2	24	9	07:30
ey l	London London			3 12	4 10	37 90	14 23	6	4 9	4 6	6	9 15	06:15 02:00	00:00 01:00	00:15			3 12	4 10	60 150	10 18	06:30 02:00	00:15 01:00			4 18	7	24 84	9 22	07:30 02:00
ley	London London			3	4	37 37	14 14	4	2 2	2 2	2 2	5 5	06:15 06:15	00:00 00:00				3	4	60 60	7 7	06:30 06:30	23:45 23:45			5	2 2	24 24	8	06:45 06:45
teleafe South	London			3	4	37	14	4	2	2	2	5	06:15	00:00				3	4	60	7	06:30	23:45			5	2	24	8	06:45
	London London Victoria	53	59	0	2	37 18	14 8	2	2	0	0	0	06:15	00:00		51	52	1	2	60 30	3	06:30	23:45	49	56	5	2	24 24	8 7	06:45
	of which London Bridge	46 53	47 56	0	0 2	1 19	5 6	2 2	0 2	0 2	0 2	0 4				N/A 51	N/A 52	0 2	0 2	0 30	0			N/A 52	N/A 53	0	0	0	0	
	of which	47	48	0	0	0	4	2	0	0	0	0	06-45	23-4E	00:45	N/A	N/A	0	0	0	0	07:00	00:15	N/A	N/A	0	0	0 24	0	07:30
	London London			2	2	18 18	11	4	2	2	2	3	06:15 06:15	23:45 23:45	00:15 00:15			1	2	30 30	5 -	07:00 07:00	00:15			4	2	24	5	07:30
	_ondon _ondon			2 2	2 2	18 18	11 11	4 4	2 2	2 2	2 2	3	06:15 06:15	23:45 23:45	00:15 00:15			1 1	2 2	30 30	5 5	07:00 07:00	00:15 00:15			4	2 2	24 24	5 5	07:30 07:30
swood	London London			2	2 2	18 18	11	4 4	2 2	2 2	2 2	3	06:15 06:15	23:45 23:45	00:15 00:15			1 1	2	30 30	5	07:00 07:00	00:15 00:15			4	2	24 24	5	07:30 07:30
enham Corner	ondon	60		2	2	18	11	4	2	2	2	3	06:15	23:45	00:15		N-/-0	1 (note 6)	2	30	5	07:00	00:15	****	****	4	2	24	5	07:30
Į.	ondon Victoria ondon Bridge	62 53	69 67	1	0 2	1 17	5 6	2 2	2 0	2 0	2 0	3 0				N/A 48	N/A 48	0	0 2	0 30	0 5			N/A 64	N/A 68	0 4	0 2	0 24	0 5	
	ondon ondon			4 4	4	36 37	19 20	7 7	5 5	6 6	5 5	6 6	06:00 06:00	23:15 23:15	00:15 00:15			3	4	60 60	4 4	06:00 06:00	00:15 00:15			4	6	72 72	5 5	07:45 07:30
kbridge	ondon			4	4	37	19	7 7	5	6	5	6	06:00	23:15	00:15 00:15			3	4	60	4	06:00 06:00	00:15 00:15			4	6	72	5	07:45 07:45
	_ondon			7 (note 3)	8	72	24 (note 4)	) 10 (note 4)	8	6	6	13	05:45	23:15	00:15			3 (note 5)	8	120	20	06:30	00:15			12	6	72	13	07:45
	London Victoria (via Norbury) London Victoria (via Gipsy Hill)	31 38	34 39	2 2	4 2	36 18	6	2 2	2 2	2 2	2 2	5 4				31 39	31 38	0	4 2	60 30	8 7			30 38	31 40	5 4	2 2	24 24	5 5	
ı	ondon Bridge (via Norwood Junction) ondon Bridge (via Tulse Hill)	30 40	46 46	1 2	0 2	0 18	6	4 2	1 2	0 2	0 2	0 4				N/A 38	N/A 38	0	0 2	0 30	0 5			N/A 39	N/A 39	0	0 2	0 24	0	
ddon I	London London			5 (note 3) 5 (note 3)	6	54 54	14 (note 4)		4	4	4	8	05:45 05:45	23:45 23:45	00:30 00:30	30	~	2 (note 5) 2 (note 5)	6	90	11 11	06:30 06:30	00:30 00:30	33	35	9	4	48	10 10	07:00 07:00
shalton Beeches I	ondon			5 (note 3) 5 (note 3)	6 6	54		) 4 (note 4) ) 4 (note 4)	4	4	4	8	05:45	23:45	00:30 00:30			2 (note 5) 2 (note 5)	6	90	11	06:30	00:30			9	4	48 48	10 10	07:00
dons Road (not via Sutton)	_ondon _ondon			5 5	2 2	20 20	6	2 2	2 2	2 2	2 2	4 4	06:00 06:00	23:45 23:45				2 2	2 2	30 30	4	06:30 06:30	23:45 23:45			0	2 2	24 24	1	09:15 09:15
nbledon (not via Sutton)	ondon ondon			5 2	2 2	20 18	6 (note 4)	2 (note 4) 2	2 2	2 2	2 2	4 4	06:00 06:00	23:45 23:45				2 2	2 2	30 30	4	06:30 06:30	23:45 23:45			0	2 2	24 24	1	09:15 09:15
th Merton (via Wimbledon)	London			2	2	18	6	2	2	2	2	4	06:00	23:45				2	2	30	4	06:30	23:45			0	2	24	1	09:15
Helier (via Wimbledon)	London London			2	2	18 18	6	2 2	2	2 2	2 2	4	06:00 06:00	23:45 23:45				2	2 2	30 30	4	06:30 06:30	23:45 23:45			0	2 2	24 24	1	09:15 09:15
on Common (via Wimbledon)	London London			2 2	2 2	18 18	6	2 2	2 2	2 2	2 2	4 4	06:00 06:00	23:45 23:45				2 2	2 2	30 30	4	06:30 06:30	23:45 23:45			0	2 2	24 24	1	09:15 09:15
ton (note 2)	ondon	20		11 (note 3)	14	126	42 (note 4)	) 14 (note 4)	11	12	12	19	05:45	23:45	00:30			7	14	210	22	06:00	00:30			13	12	144	16	07:00
L.	London Victoria (Hackbridge fast) London Victoria (Hackbridge stopping)	32 33	32 37	0 2	2	18 18	9	3	0 3	0 4	0 4	6				31 32	31 32	1	2	30 30	3			31 42	32 33	0 4	0 4	0 48	0 5	
ľ	London Victoria (via Norbury) London Victoria (via Gipsy Hill)	44 51	47 52	2 2	4 2	36 18	6	2 2	1 2	2 2	2 2	5 3				43 51	44 51	1 0	4 2	60 30	8			43 52	44 53	5 4	2 2	24 24	5 4	
ı	London Bridge (via Tulse Hill)  London Bridge (via West Croydon)	42	54 54	2	0	0	4	2	0	0	0	0				N/A N/A	N/A N/A	0	0	0	0			N/A N/A	N/A N/A	0	0	0	0	
Į.	ondon Blackfriars (via Wimbledon)	49	49	2	2	18	6	2	2	2	2	4				N/A 49	49	2	2	30	4			N/A 49	49	0	2	24	2	
	London Blackfriars (via Hackbridge) London Victoria	37	37	3 2 (note 3)	1	18 12	5	2	2	1	1	1	06:00	22:30		37	37	1	1	30 15	2	06:30	22:30	37	37	0	0	24 0	0	-
tead l	London Victoria London Victoria	56	62	2 (note 3) 2 (note 3)	1	12	5	2	2	1 1	1 1	1 1	06:00 06:00	22:30 22:30		53	53	1 1	1	15	2	06:30 06:30	22:30 22:30			0	0	0	0	-
m l	ondon	50	υz	4	5	45	17	5	4	5	5	8	06:00	23:15	00:15	55	- 33	2 (note 5)	5	75	8	07:00	00:15			4	4	48	6	07:00
om l	_ondon _ondon			4	4 5	36 45	17 17	5 5	4	5 5	5 5	8 8	06:00 06:00	23:15 23:15	00:15 00:15			2 (note 5) 2 (note 5)	4 5	60 75	6 8	07:00 07:00	00:15 00:15			4	4 4	48 48	6 6	07:00 07:00
	London Victoria (Hackbridge fast) London Victoria (Hackbridge stopping)	41 43	41 47	0 2	2 2	18 18	0 11	0 3	0	0 4	0 4	0				41 42	41 42	0	2 2	30 30	3 4			N/A 43	N/A 44	0 4	0 4	0 48	0	
ı	ondon Victoria (via Norbury)	59	59	0	1	9	0	0	0	0	0	1				55	55	1	1	15	1			59	62	0	Ö	0	1	
ı	London Victoria (via Gipsy Hill) London Bridge (via West Croydon)	62 48	62 64	0	0	0	1 5	2	0 1	1 0	1 0	1 0				N/A N/A	N/A N/A	0	0	0	0			N/A N/A	N/A N/A	0	0	0	0	
ead l	London London			3	2 2	18 18	11 11	4 4	3	2 2	2 2	3	06:00 06:00	23:15 23:15				1 (note 5) 1 (note 5)	2 2	30 30	4	07:00 07:00	23:15 23:15			4	2 2	24 24	5 5	07:00 07:00
kham [	London London			1	0	0	2	1 1	0	0	0	0	06:45 06:45	17:15 17:15				0	0	0	0	-	-			0	0	0	0	-
ley I	_ondon			0	0	0	2 2	1 1	0	0	0	0	16:45	17:15				0	0	0	0	-	-			0	0	0	0	-
	London London			0	0	0	2 2	1 1	0	0	0	0	16:45 06:45	17:15 17:15				0	0	0	0	-	:			0	0	0	0	-
dford	London London Victoria			1	0	0	2	1 0	ō	0	0	0	06:45	17:15				0	0	0	0	-	-			0	0	0	0	-
	ondon Bridge	81	85	0	0	0	1	1	0	0	0	0						0	0	0	0					0	0	0	0	
Hill and Westhumble ting	London London		Ţ	1 2	1 2	11 18	9	1	3	2 2	1 2	3	06:00 06:00	23:15 23:15	]		Ţ	1 1	1 2	15 30	3 4	07:00 07:00	23:15 23:15			4	2 2	24 24	5 5	07:00 07:00
nwood	London Victoria			2 2	1	9	5	2 2	1	0	0	1 1	06:00 06:00	23:15 23:15				0	1	15 15	0	07:45 07:45	21:00 21:00			0	0	0	0	-
nham [	ondon Victoria	70		2	1	9	5	2	1	0	0	1 1	06:00	23:15			74	0	1	15	0	07:45	21:00			0	0	0	0	-
sham (via Dorking)	ondon Victoria	78	83	2	1	10	5	2	1	0	0	1	06:00	23:15		74	74	0	1	15	0	07:45	21:00		1	0	1 0	0	0	-
	Note 1: 1	The parliame	entary service	between Stre Pancras to S	eatham Hill	and Tulse H	Hill must con	tinue to opera	ate.	as via Wimble	adon mueto	ontinue to ~	nerate se tho	Current "M's	mhledon I c~	n" service :~i	ned through	Sutton												
		Similarly, ser-	rvices from Lo	ondon St. Par	ncras to Sur	tton via Wim	nbledon and	from Sutton	to London S	St. Pancras v	ia Hackbrid	ge must conf	tinue to be jo	ned through	Sutton.			JUNUTI.												
	7	The second of	of these servi	hall operate fri ices shall star	t at Norwoo	od Junction	and extend t	to Epsom Do	wns, calling	at Belmont a	and Banstea	ad.																		
	Note 4: I	rom Januar	ry 2017, the r	equirement be hall operate from	etween 16:0	00-18:59 at	Norwood Ju	inction and W	est Croydo	on increases b	by four, whil	le the require							n) and Wim	bledon incre	ases by thre	e. The requir	rement betwe	een 17:00-1	7:59 at the st	tations desc	cribed increas	es by two.		
	7	The first of th	hese services	hall operate fr s shall operate ices shall exte	from Stream	atham Hill, c	alling at We	st Norwood,	Gipsy Hill, (	Crystal Palaci	e and Norw			nun peècne	o, arriving at	Juliuri Detwe	on or:uu and	u ur.30.												

TABLE 2: NORTHBOUND		lavar	Time	0000 0000	0700-0959	0000 0050	Monday to	Friday	2000 2000	2400 2450	1 2200 0450	C				0200-0659	Satu	irday	2000 0450	C		laura a	T	0000 0050	Sun	nday	2400 0450	C	1 4
Arrival time in London		Journ	ey Time	0200-0658	0700-0959	0000-0009	1000-1959	1000-1958	2000-2059	2100-2159	2200-0158		Last (no	Journe	y Time	0200-0659	0700-2159	0700-2159	2200-0159		Last (no	Journe	ey Time	0200-0659	0900-2059	0900-2059	2100-0159		Last (no
Direct services from each station to specified destination	Destination	Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)
Horsham (via Dorking) Warnham	London Victoria London Victoria	80	87	0	5 5	2	1	12 12	1	1	0	07:15 07:15	21:15 21:15	74	74	0	1 1	15 15	0	08:00 08:00	21:00 21:00			0	0	0	0	-	- 1
Ockley Holmwood	London Victoria London Victoria			0	5	2	1	12 12	1	1	0	07:15 07:15	21:15 21:15			0	1 1	15 15	0	08:00 08:00	21:00 21:00			0	0	0	0	-	:
Dorking	London			0	7	3	2	20	3	2	3 (note 5)	07:15 07:15	23:15			0	2	30 16	3	07:15 07:45	23:15			2	2	24 24	7	08:30 08:30	00:15
Box Hill and Westhumble Guildford	London London			0	2	1	0	1	0	0	1 (note 5)	08:45	19:45			0	0	0	0	- 07:45	22:45			0	0	0	0		00:15
	London Victoria London Bridge	69	69	0	1 1	0	0	1	0	0	0					0	0	0	0					0	0	0	0		
London Road (Guildford)	London London			0	2	1 0	0	1	0	0	0	08:45 09:45	19:45 19:45			0	0	0	0	-	-			0	0	0	0	-	-
Clandon Horsley	London			0	i	0	0	1	0	0	0	09:45	19:45			0	0	0	0	:	- :			0	0	0	0	- :	- :
Effingham Junction Bookham	London London			0	2 2	1	0	1	0	0	0	08:45 08:45	19:45 19:45			0	0	0	0	1	-			0	0	0	0	-	
Leatherhead Ashtead	London London			0	9	4	2 2	21 21	3	2	3 (note 5) 3 (note 5)	07:15 07:15	23:15 23:15			0	2	30 30	3	07:15 07:15	23:15 23:15			2 2	2	24 24	7	08:30 08:30	00:15 00:15
Epsom	London			3	13	6	5	50	4	5	9 (note 5)	06:15	23:30			0	5	75	9	07:15	23:45			4	4	48	7	07:45	00:15
	London Victoria (Hackbridge fast) London Victoria (Hackbridge stopping)	41 44	41 47	0	9	0 4	2 2	20 20	2	4	7			41 42	41 42	0	2 2	30 30	3 4			N/A 44	N/A 51	3	4	48	7		
	London Victoria (via Norbury) London Victoria (via Gipsy Hill)	55 62	55 64	1 0	0	0	1 0	10 0	0 2	0	0 2			55 N/A	55 N/A	0	1 0	15 0	2			50 N/A	50 N/A	1 0	0	0	0		
Evell East	London Bridge (via West Croydon)	45	51	1 2	4 12	2	0	0 41	0	0	0 9 (note 5)	06:15	23:30	N/A	N/A	0	0	0 60	0	07:15	23:45	N/A	N/A	0	0	0 48	0 7	07:45	00:15
Ewell East Cheam	London London			3	13	6	5	50	4	5	9 (note 5)	06:15	23:30			0	5	75	9	07:15	23:45			4	4	48	7	07:45	00:15
Epsom Downs Banstead	London Victoria London Victoria	56	62	0	5 5	2	1	13 13	2 2	1	2 (note 5) 2 (note 5)	07:30 07:30	23:45 23:45	54	54	0 0	1 1	15 15	2 2	08:00 08:00	23:15 23:15			0	0	0	0 0		- :
Belmont Sutton (note 1)	London Victoria London			0 7	5 41	2 16	1 14	13 140	2 10	10	2 (note 5) 19 (note 5)	07:30	23:45 23:45	-		0	1 14	15 210	2 24	08:00 07:15	23:15 00:00			0 7	0 12	0 144	0 27	07:45	00:15
Cutton (note 1)	London Victoria (Hackbridge fast)	31	31	0	0	0	2	20	0	0	0	00.13	25.45	31	31	0	2	30	3	07.13	00.00	N/A	N/A				_	07.40	00.13
	London Victoria (Hackbridge stopping) London Victoria (via Norbury)	33 44	36 47	1	10 9	4	2 4	20 40	0	0	4			32 43	32 44	0	4	30 60	9			33 40	40 43	3 2	4 2	48 24	8		
	London Victoria (via Gipsy Hill) London Bridge (via Tulse Hill)	51 41	54 43	0 2	1 5	0	2	20 0	2	2	4 0			51 N/A	51 N/A	0	2 0	30 0	4			48 N/A	53 N/A	1 0	2	24 0	6		
	London Bridge (via West Croydon)	34	41	2	5	2	0	0	0	0	0			N/A 50	N/A	0	0	0	0			N/A	N/A	0	0	0	0		
	London Blackfriars (via Wimbledon) London Blackfriars (via Hackbridge)	50 34	50 35	0	6 5	1 2	2	20 20	2	2	0			50 34	50 35	0	2	30 30	0			50 34	50 35	0	2	24 24	2		
West Sutton (via Wimbledon) Sutton Common (via Wimbledon)	London London			2	6	2	2 2	20 20	2 2	2	4	07:00 07:00	23:30 23:30	1		0	2 2	30 30	4 4	07:30 07:30	23:30 23:30			0	2	24 24	2 2	09:30 09:30	21:45 21:45
St. Helier (via Wimbledon) Morden South (via Wimbledon)	London London			2 2	6	2 2	2 2	20 20	2 2	2 2	4 4	07:00 07:00	23:30 23:30	1		0	2 2	30 30	4	07:30 07:30	23:30 23:30			0	2 2	24 24	2 2	09:30 09:30	21:45 21:45
South Merton (via Wimbledon) Wimbledon Chase (via Wimbledon)	London London			2	6	2	2 2	20 20	2 2	2	4 4	07:00 07:00	23:30 23:30	1		0	2	30 30	4	07:30 07:30	23:30 23:30			0	2 2	24 24	2 2	09:30 09:30	21:45 21:45
Wimbledon (not via Sutton)	London			2	10	3	2	23	2	2	4	07:00	23:30			0	2	30	4	07:30	23:30			0	2	24	2	09:30	21:45
Haydons Road (not via Sutton) Tooting (not via Sutton)	London London			1	6	2	2	23 23	2 2	2	4	07:00 07:00	23:30 23:30			0	2 2	30 30	4	07:30 07:30	23:30 23:30			0	2 2	24 24	2 2	09:30 09:30	21:45 21:45
Carshalton Beeches Wallington	London London			3	15 15	6	6 6	60 60	4	4	8 (note 5) 8 (note 5)	06:15 06:15	23:45 23:45			0	6	90 90	13 13	07:30 07:30	00:00 00:00			4	4	48 48	14 14	07:45 07:45	00:15 00:15
Waddon	London			3	14	5 10	6	60 80	4	4	8 (note 5)	06:15	23:45			0	6	90	13 17	07:30	00:00			4	4	48	14 19	07:45	00:15
West Croydon	London London Victoria (via Norbury)	31	34	1	25 8	4	4	40	2	2	12 (note 5	06:15	23:45	31	31	0	4	120 60	9	07:00	00:00	28	31	3	2	72 24	8	07:45	00:15
	London Victoria (via Gipsy Hill) London Bridge (via Norwood Junction)	39 22	42 29	0 2	6 5	2	2	20 0	2	2	4 0			38 N/A	39 N/A	0	2 0	30 0	4 0			36 N/A	51 N/A	3 0	2	24 0	6		
Carshalton	London Bridge (via Tulse Hill) London	39	44	3	6 19	2	2	20 41	2	2	4 7	06:15	23:30	38	38	1	2	30 60	4	07:30	23:30	38	38	2	2	24 72	5 11	08:00	00:15
Hackbridge	London			3	18	7	4	41	4	5	7	06:15	23:30			ő	4	60	4	07:30	23:30			3	6	72	11	08:00	00:15
Mitcham Junction Mitcham Eastfields	London London			3	19 20	8	4	41 41	4	5	7	06:15 06:15	23:30 23:30			0	4	60 60	4	07:30 07:30	23:30 23:30			3	6 6	72 72	11 11	08:00 08:00	00:15 00:15
Tattenham Corner	London London Victoria	64	75	1 0	10 4	4 2	2	21 0	2	2	4 0	07:00	23:45	64	64	0	2 0	30 2	4 0	07:15	23:30	N/A	N/A	1 0	2	24 0	5 (note 7) 0	09:00	23:15
Tadworth	London Bridge London	49	58	1	6 10	2	2 2	20 21	2	2	4 4	07:00	23:45	48	48	2	2	30 30	4 4	07:15	23:30	62	63	1 1	2 2	24 24	5 5 (note 7)	09:00	23:15
Kingswood	London			1	10	4	2	21	2	2	4	07:00	23:45			0	2	30	4	07:15	23:30			1	2	24	5 (note 7)	09:00	23:15
Chipstead Woodmansterne	London London			1	10 10	4	2 2	21 21	2 2	2	4	07:00 07:00	23:45 23:45			0	2 2	30 30	4	07:15 07:15	23:30 23:30			1 1	2	24 24	5 (note 7) 5 (note 7)	09:00 09:00	23:15 23:15
Coulsdon Town Reedham	London London			1	10 10	4	2 2	21 21	2 2	2	4 4	07:00 07:00	23:45 23:45			0	2 2	30 30	4	07:15 07:15	23:30 23:30			1 1	2 2	24 24	5 (note 7) 5 (note 7)	09:00 09:00	23:15 23:15
Caterham	London London Victoria	44	44	2	12 6	4	4 2	40 20	2	2	3 (note 6) 0	06:45	23:30	51	52	1 0	4 2	60 30	7	07:00	23:45	AG.	47	3 2	2	24 24	5 (note 7) 5	08:15	23:15
	of which	51	56	0	5	2	0	0	0	0	0			N/A	N/A	0	0	0	0			N/A	N/A	0	0	0	0		
	London Bridge of which	49 43	55 45	2	6 5	2	2 0	20 1	0	0	0			51 N/A	52 N/A	0	0	30 0	0			52 N/A	52 N/A	0	0	0	0		
Whyteleafe South Whyteleafe	London London			2	12 12	4	4	40 40	2	2	3 (note 6) 3 (note 6)	06:45 06:45	23:30 23:30			1	4	60 60	7	07:00 07:00	23:45 23:45			3	2	24 24	5 (note 7) 5 (note 7)	08:15 08:15	23:15 23:15
Kenley Purley	London London			2	12 23	4 8	4	40 100	2	2	3 (note 6) 13 (note 6)	06:45	23:30 01:45	1		1 1	4	60 150	7 17	07:00 02:15	23:45			3 15	2 7	24 84	5 (note 7) 20 (note 7)	08:15 02:15	23:15 01:45
Purley Oaks	London	l		4	12	4	4	41	4	4	7 (note 6)	06:30	23:45	1		1	4	60	7	07:00	23:45			4	2	24	5 (note 7)	07:30	23:15
South Croydon  East Croydon	London London	-		4 28	19 78 (note 3)	6 28 (note 3)	4 24	41 244	4 20	20	7 (note 6) 36 (note 6)		23:45 01:45	<b> </b>		1 18	4 24	60 364	7 37	07:00 02:15	23:45 01:45			4 29	2 19	24 239	5 (note 7) 54 (note 7)	07:30 02:15	23:15 01:45
Norwood Junction Anerley	London London			10 4	25 6	8 2	10 2	100 21	4 2	4 2	7 3	05:45 05:45	23:45 23:30	1		4	10 2	150 30	15 3	06:30 07:00	00:15 23:30			13 4	8 2	96 24	20	06:00 07:30	23:45 23:15
Penge West	London	-		4	6	2	2	21	2	2	3	05:45	23:30	ļ		1 15	2	30 120	3	07:00	23:30			4	2	24	5 27	07:30 02:15	23:15
Selhurst Thorton Heath	London London			7	22 22	8	8	80	6	6	17	05:00	23:45	1		15 4	8	120	19	06:00	00:00			11	6	72	18	06:00	00:15
Norbury Streatham Common	London London			7	22 22	8	8	80 80	6	6 6	12 12	05:00 05:00	23:45 23:45	1		4 4	8 8	120 120	17 17	06:00 06:00	00:00 00:00			11 11	6	72 72	18 18	06:00 06:00	00:15 00:15
Streatham Beckenham Junction	London London Bridge	36	40	5 0	22 5	8	6	63 20	6 2	6 2	8 4	06:15 08:00	23:30 23:45	35	35	2	6 2	90 30	8 4	06:15 07:30	23:30 23:45			4 0	6	72 0	11 0	07:30	23:15
Birkbeck	London Bridge London	L		0	5	2	2	20	2	2	4	08:00	23:45	<u> </u>		0	2	30 90	4	07:30	23:45			0 7	0 4	0	0	07:30	00:00
Crystal Palace (via Gipsy Hill)	London Victoria	27	32	2	12	4	4	40	4	4	10	00:15	00:30	27	28	2	4	60	10	06:30	00:30	27	32	3	2	24	6	07.30	00.00
Gipsy Hill	London Bridge London	28	32	1	6 20 (note 4)	2 6 (note 4)	2 6	20 60	6	2 6	4 14	06:15	00:30	28	29	0 2	2 6	30 90	4 14	06:30	00:30	26	26	4 7	2 4	24 48	7 13	07:30	00:00
West Norwood Streatham Hill (note 2)	London London			3 2	20 (note 4) 14 (note 4)		6 4	60 40	6	6 4	14 12	06:15 06:15	00:30 00:30	1		2 2	6 4	90 60	14 10	06:30 06:30	00:30 00:30			7 4	4 2	48 24	13 6	07:30 07:30	00:00 23:45
Crystal Palace (via Forest Hill)	London Bridge London Bridge	1		0 4	10 (note 4)	2 (note 4)	2	20 41	2 4	2 4	6	07:15 05:45	00:30	1		0	2 4	30 60	6	07:30 07:00	00:30			0	0 4	0 48	0	07:30	23:15
Sydenham Forest Hill	London Bridge			4	16 (note 4) 16 (note 4)	4 (note 4)	4	41	4	4	9	05:45	00:30 00:30	1		1 1	4	60	9	07:00	00:30 00:30			6	4	48	10	07:30	23:15
Honor Oak Park Brockley	London Bridge London Bridge			4	16 (note 4) 16 (note 4)		4	41 41	4 4	4	9	05:45 05:45	00:30 00:30	1		1 1	4 4	60 60	9	07:00 07:00	00:30 00:30			6 6	4	48 48	10 10	07:30 07:30	23:15 23:15
New Cross Gate Tulse Hill (note 2)	London Bridge London	-		4	16 (note 4)	4 (note 4)	6	60 88	4 8	4 8	9	05:45 06:15	00:30 23:45	ļ		2	6	90	13 12	06:45 06:15	00:30 23:45			6 7	4 8	48 96	10	07:30 07:30	23:15
North Dulwich	London Bridge			3	17	6	4	48	4	4	8	06:15	23:45	1		0	4	60	8	07:15	23:45			7	4	48	12	07:30	00:00
East Dulwich Peckham Rye	London Bridge London Bridge	l		3	17 17	6	4	48 48	4	4	8	06:15 06:15	23:45 23:45	1		0	4	60 60	8	07:15 07:15	23:45 23:45			7	4	48 48	12 12	07:30 07:30	00:00 00:00
Queens Road Peckham South Bermondsey	London Bridge London Bridge			3	17 17	6	4	48 48	4	4	8	06:15 06:15	23:45 23:45	1		0	4	60 60	8 8	07:15 07:15	23:45 23:45			7 7	4	48 48	12 12	07:30 07:30	00:00 00:00
Herne Hill	London Blackfriars	1		3	11	4	4	40	4	4	4	06:15	23:30	1		2	4 4	60	4	06:15	23:30			0	4 4	50	6	10:30	22:30
Loughborough Junction Elephant & Castle	London Blackfriars London Blackfriars			2	11 11	4	4	40 40	4	4 4	3 4	06:30 06:15	23:00 23:30			2	4	60 60	3 4	06:45 06:15	23:00 23:30			ō	4	50 50	6 6	10:30 10:30	22:30 22:30
Balham Wandsworth Common	London Victoria London Victoria			9	36 26	13 9	12 10	120 100	11 8	11 8	25 18	05:00 06:30	00:30 00:30	I		4 3	12 10	180 150	27 23	06:00 06:30	00:30 00:30			13 7	10 4	120 48	26 14	06:00 07:15	00:15 00:15
Clapham Junction Battersea Park	London Victoria London Victoria	l		17	71 25	26	24	230 81	19	21	44 16	03:00 05:15	01:45 00:30	1		12	24	360 120	49 18	02:15 05:15	01:00 00:30			25 7	18	216 48	46 14	02:15 06:00	01:45 00:15
Dattersea Fair	CONGOT VICTORIA			- /	20	3	٥	01	/	- /	10	UJ:15	00:30			ا ا	٥	120	10	UJ: 15	00:30			/	4	46	14	UU.UU	UU. 13

Note 1: Services from Sutton to London St. Pancras via Hackbridge and from London St. Pancras via Wimbledon must continue to operate as the current "Wimbledon Loop' service, joined through Sutton. Similarly, services from Sutton to London St. Pancras via Wimbledon and from London St. Pancras via Wimbledon and Via Hackbridge must continue to be joined through Sutton. Note 3: From January 2017, one additional services shall be provided from Crystal Palace to London Bridge, arriving between 08:00 and 08:29.

Note 4: From January 2017, one additional services shall be provided from Crystal Palace to London Bridge at maximum intervals of 40 minutes, calling at all stations via Forest Hill, arriving between 07:30 and 08:59.

Two of these services shall devent to Strestham Hill, calling at Gipsy Hill and West Norwood. In Minutes, calling at Carshalton Beeches, Wallington and Waddon, departing between 23:15 and 23:59.

One of these services shall start at Epsom, calling at Box Hill & Westhumble, Leatherhead, Ashtead, Epsom, Ewell East and Cheam.

One of these services shall start at Epsom Downs, calling at and Belmont.

A further additional service shall operate from Dorking to Epsom, calling at all stations, departing after 23:15.

Note 6: An additional service shall operate from Purley to East Croydon, with a maximum interval of 10 minutes, calling at all stations, departing between 23:00 and 23:29.

One of these services shall operate from Tattenham Corner, calling at all stations, and extend to Norwood Junction.

One of these services shall operate from Careham to Normood Junction.

One of these services shall operate from Careham to Normood Junction.

BML Inner 2: Brighton Main Line Minimum Service Specification (Inner Services only) and First and Last Train Specification

Column	TABLE 1: SOUTHBOUND		lacco	ou Tim-	0200 0050	0700 455	0700 455	1600 4050	Monday to	Friday	2050 2400	) 24EAl AA	200 0450	C '	o lon /	andon	la	w Time	0200 0050	Satur		2200 0452	Conésa I	woo I		ou Tim -	0200 0050	Suno		2100 0450	Cond I	nunc I and
Septiment of the septim	Departure time from London		Journe	ey ime	0200-0659	0700-1558	9 0700-1558	1600-1859	1/00-1/59 1	900-1959 2000	J-2059 2100	F2159 22	200-0159		Last (no	Last (Fri	Journe	ey rime	0200-0659	0700-2159	0700-2159 2	2200-0159		Last (no	n Journe	ey ime	0200-0859	1900-2059	0900-2059	2100-0159		Last (no
Series Control of the			Average	Maximum	Total	TPH		Total	Total	Total T	otal To	otal		later than)	than)	nights only)	Average	Maximum	Total	TPH			later than)	than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	
Part	Clapham Junction	London Victoria					211	26 75					45	02:00	01:00						360	53	02:00	00:45							07:00 02:00	00:30 01:00
Column												8 12	18 24						3 4			25 29									07:00 06:45	00:30 00:30
March   Marc		London Blackfriars			7 7																						1	4		1	09:00 09:00	21:15 21:15
General Confession - Confession	Herne Hill	London Blackfriars			7	4	36	12	4 8	4	4	4	8	05:30	23:45	00-30			4	4	60	6		23:45	ļ		1 7	4	48	1	09:00 07:30	21:15
March   Marc	Queen's Road Peckham	London Bridge			7	4	41	21	8	6	5	4	8	06:00	00:00	00:30			1	4	60	10	06:45	00:30			7	4	48	13	07:30	00:00
March   Marc	East Dulwich	London Bridge			7	4	41	21	8	6	5	4	8	06:00	00:00	00:30			1	4	60	10	06:45	00:30			7	4	48	13	07:30 07:30	00:00 00:00
Property of the content of the con	Tulse Hill (note 1)	London				4	77	33			9	8	16	05:30	00:00				1 5	8	120	14	06:00	00:30			8	8	96	13 14	07:30 07:30	00:00 00:00
Company   Comp									5											6 4	90 60						7 7	4			07:30 07:30	00:30 00:30
Control Property   Control Pro																															07:30 07:30	00:30 00:30
Second	Sydenham	London Bridge			6									05:45	00:30				2				06:30	00:30							07:30	00:30
Company   Comp	Streatham Hill (note 1)	London					36					4		05:30	23:45					4			05:30	00:30				2		9	07:30	23:45
Company of the Comp	Gipsy Hill	London			6		54	18	- 1		6	6	12	06:15	00:00	00:30			0 (note 4)	6	90	16	07:15	00:30				4	48	16	07:30 07:30	23:45 23:45
Section   Control of the control o	Crystal Palace (via Gipsy Hill)		27	32	6	6 4			6		6	6 4		05:45	00:00	00:30	27	27		6			07:15	00:30	28	30	7	2			07:30	23:45
Part	Rirkheck	London Bridge			2	2	18	6 7	2			2		06:30	22:15					2	30		07:30	22:15				2	24	7		<u> </u>
Part	Beckenham Junction	London Bridge	36	36		2	18	7	2	2	3	2	1	06:30	22:15	00-45	36	36	0	2	30	1	07:30	22:15			0	0	0	0	07:20	- 22-45
Proper   P	Streatham Common	London	1		6	8	72	23	6		6	6	14	06:00	00:30	υU:15				8	120	19	06:30	00:30				6	72		07:30 06:45	23:15 00:30
Part					6	8 8		23			6	6	14	06:00	00:30					8	120	19	06:30	00:30	1			6		21	06:45 06:45	00:30 00:30
March			1			8 2	72	23	6 2	5 2	6 2	6 2	15	02:00					8 2	8 2	120	19	02:00	00:30	<del>                                     </del>		20	6 2	72	24	02:00 07:30	01:00 00:30
Part	Anerley	London	1		4	2	18	6	2	2	2	2	6	05:45	00:30				2	2	30	6	06:30	00:30			4	2	24	8	07:30 07:30	00:30 00:30
Marco   Marc	East Croydon	London			48	26	240	100		24	22 2	22	48	02:00	01:00	0.7.1			24 (note 5)		360	46	02:00	00:45			35		240	52	02:00	01:00
Markey  Michael Control 19	Purley Oaks	London				4	41	16	4	4	4	4	8	05:30	23:45				3	4	60	10	06:30	00:15	1		4	2	24	9	07:30 07:30	23:45 23:45
Month of the control	Purley		1									8 4	18 7	02:00	01:00							19 7	02:00		1					22 8	02:00 06:45	01:00 23:45
Contact   Cont	Whyteleafe	London			3	4	43	15	4	4	4	4	7	06:00	23:45				2	4	60		06:30	23:45	1		5	2	24	8	06:45 06:45	23:45 23:45
Margin   M		London			3	4	43	15	4	4	4	4	7						2	4	60	7					5	2	24	8	06:45	23:45
Part								8 5												2 0								0		7 0		
Product   Prod		London Bridge		61	3		25 6	6	2		2	2	4				53	53	1 0	2								0	0	1		
Non-conference   Conference		London			3	2				2	2	2	3							2		5						2		5	07:30 07:30	23:15 23:15
Programmer	Woodmansterne	London			3	2	18	11	4	2			3	06:15	23:15	00:15			1 (note 5)	2	30	5	07:00	00:15			4	- 1	24	5	07:30	23:15
Transfer Gener   Control Contr		London					18							06:15	23:15	00:15			1 (note 5)	2	30		07:00	00:15					24	5	07:30 07:30	23:15 23:15
Marcian Estimate  Marcian Esti					3															2 2										5	07:30 07:30	23:15 23:15
Mexican Fundamental   Section   Se			63		1 2	0		5	2	0	0	0	0							0	0	0						0		0		
London-Color   G   4   77   19   7   0   6   6   10   100   200   201   20   0   10   0   0   0   0   0   0   0		London	77	30	6		36		7	5	6	6							3	4		6			- 03	- 05	4	6		5	07:45 07:45	23:00 23:00
Mest Civysfor   Cortical Victoria (wa Machinary)   Sample   Samp	Hackbridge	London			_		37	19			6	6	10	05:30	23:45	00:15					60		06:00	00:15			4	6	72	5	07:45	23:00
London Victoria (Parl Minister)   27   30   2   2   18   6   2   2   2   2   4		London				8	72			8	6	6	13							8	120						12	6	72	13	07:45 07:00	23:00 23:15
Company   Comp					2 2			6											0	2								2 2		5 5		
Washer Marked Control		London Bridge (via Norwood Junction)						9												- 1										0		
Considerable   Cons		London		1.0	5 (note 3)	6	54	17	6	4	4	4	8				00		2 (note 4)	6	90	11			0.	0.	9	4	48	10	07:00 07:00	23:15 23:15
Hyprone Road (not we Sultano)  Winnelson (not with a Sultano)  Winnelson (not with Sultano)  London  B 2 2 20 8 8 4 2 2 2 2 4 66-56 23-56  Control winnelson (not with a Sultano)  London  Lon	Carshalton Beeches	London				6	54		6	4	4	4	8	05:45	23:45					6	90	11	06:30	00:30			9	4	48		07:00	23:15
Windedon Chase (vis Windedon)   London	Haydons Road (not via Sutton)				6	2 2		8 8		2 2	2	2		05:45					2 2	2 2				23:45				2		1	09:15 09:15	21:15 21:15
South Martin (via Wimbledon)   London     3   2   18   8   4   2   2   2   2   4   65.45   23.45     2   2   2   30   4   66.30   23.45     0   0   2   24   1					_				- 1				4							2 2								- 1		1	09:15 09:15	21:15 21:15
Self-leffer (will will will will will will will wil	South Merton (via Wimbledon)	London			3	2	18		4	2			4	05:45	23:45						30	4	06:30	23:45			0		24	1	09:15 09:15	21:15 21:15
West Sutton (via Winnelbedon)   London   Londo	St. Helier (via Wimbledon)	London				2	18	8	4	2			4	05:45	23:45					- 1	30	4	06:30	23:45			ō	2	24	1	09:15	21:15
Sutton (note 2)  London (Victoria (Hackbridge fast)) 32 32 02 02 18 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0											2	2																		1	09:15 09:15	21:15 21:15
London Victoria (lab Noturo) 44 48 2 4 36 10 2 1 2 2 5 5 44 46 1 4 60 8 8 43 44 5 2 24 5 5 1 6 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Sutton (note 2)		32	32	15 (note 3) 0	14 2		45 0	17 0	11	12 1	0	22 0	05:30	23:45	00:30	32	32	7	14 2	210	22 3		00:30	31	32	13 0	12 0	144 0	16 0	07:00	23:15
London Victoria (Na Gispay Hill)   51   52   2   2   18   6   2   2   2   2   2   3   8   8   1   1   0   2   30   4   52   53   4   2   24   4   4   4   4   4   4		London Victoria (Hackbridge stopping)	32	37	2		18		3	3	4	4	6				33	33	1	2	30				32	33		4		5		
London Bridge (vis Tubes Hill) London Ridge (vis West Croyoton) London Bridge (vis West Croyoton) London Ridge (vis West Croyoton) London		London Victoria (via Gipsy Hill)	51	52	2	2	18	6	2	2	2	2	3				51	51	0	2	30	4			52	53	4	2	24			
London Blackfrians (via Hockorina)   52   52   2   2   18   6   2   2   2   2   4       52   52   2   2   30   4     52   52   52   0   2   24   2   2   2   4     2   38   38   2   2   30   0   4   52   52   52   0   2   24   2   2   2   4   2   38   38   3   2   2   30   0   38   38   0   2   2   2   4   2   30   30   3   38   3   2   2   30   0   0   0   0   0   0   0   0				58				4																						0		
Belmont   London (National Alexandrians) (via) Hackbridge)   38   38   3   2   18   6   2   2   2   2   4				41 52													N/A 52	N/A 52	0 2							N/A 52				0		
Banstead   London Victoria   Sporm Downs   London Downs   London Downs   London Downs   London Victoria   Sporm Downs   London L	Belmont	London Blackfriars (via Hackbridge)	38	38	3	2	18	6	2	2	2	2		06:00	22:3N				2	2	30 15		06:30	22:30	38			2		0		
Cheam	Banstead	London Victoria			2 (note 3)	1	12	5	2	2	1	1	1	06:00	22:30		F.	_	1	1	15	2	06:30	22:30			0	0	0	0	-	-
Evel East   London	Cheam	London	57	63	4		45				5	5		06:00	23:15		53		1 2 (note 4)	5	75		07:00	00:15	1		4		48	6	07:00	23:15
London Victoria (Hackbridge fast) London							36 45				5	5			23:15				2 (note 4)	4 5	60 75				1		4		48	6	07:00 07:00	23:15 23:15
London Victoria (via Motbury) 59 59 0 1 9 0 0 0 0 0 1 1 1 1 1 5 59 59 5 1 1 1 1 15 1 59 62 0 0 0 1 1 0 0 0 1 1 1 1 1 1 1 1 1 1 1		London Victoria (Hackbridge fast)			0	2	18	0	0	0			0						0		30	3					0		0	0		
London Bridge (via West Croydon)   42   44   0   0   0   5   2   1   0   0   0   0   0   0   0   0   0		London Victoria (via Norbury)	59	59	0	1	9		ō	0	0	0	1				59	59	1	1	15	1			59	62	0	0	0	1		
Ashtead London 3 2 18 11 3 3 2 2 3 06:00 23:15 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 Leatherhead London 3 2 18 11 3 3 3 2 2 3 06:00 23:15 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 5 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 5 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 5 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 5 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 5 1 (note 4) 2 30 4 07:00 23:15 4 2 24 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		London Bridge (via West Croydon)		44		0	0			1	0	0	o						0	0	0	0				N/A N/A	0		0	0		
Bookham	Leatherhead	London	<u> </u>		3	2 2			3	3	2	2	3	06:00	23:15					2	30 30	4	07:00 07:00	23:15 23:15				2	24 24	5 5	07:00 07:00	23:00 23:00
Horsley London     0   0   0   2   1   0   0   0   16:45   17:15     0   0   0   0   -   -   0   0   0		London	1		1	0		2 2	1	0	0	0		06:45 06:45						0		0	-	-			0	0	0	0	-	:
	Horsley	London				ō	0	2	1	0	0		0	16:45	17:15				0		0	0	-	1	1		0	0	0		-	-
London Road (Guildford) London     1   0   0   2   1   0   0   0   06:45   17:15       0   0   0   0   -   -     0   0	London Road (Guildford)	London	1		1	ō	0	2	1	ō	ŏ	ŏ	0	06:45	17:15				0	0	0	0	-	-	1		0	0	0	0	-	-
Guildford         London         1         0         0         2         1         0	Guildford	London Victoria						1	1 0		0	0		06:45	17:15					0			-	-				0		0	-	-
London Bridge         77         77         0	Box Hill and Westhumble	London Bridge	77	77			0	1 9	1 2				0	06;00	23:15					0	0	0	07:00	23:15				0 2	0		07:00	23:00
Dorking   London     2   2   18   9   2   3   2   2   3   06:00   23:15       1 (note 4)   2   30   4   07:00   23:15     4   2   24   5	Dorking	London	1		2		18	9				2		06:00	23:15				1 (note 4)	2	30	4	07:00	23:15	1			2	24	5	07:00	23:00
Holmwood   London Victoria   2   1   9   5   2   1   0   0   1   06:00   23:15   0   1   15   0   07:45   17:30   0   0   0   0   0   0   0   0   0	Ockley	London Victoria			2		9	5		1		0	1	06:00	23:15					1	15	0	07:45	17:30	1		0	0	0	0	-	-
Warnham         London Victoria         2         1         9         5         2         1         0         1         06:00         23:15         0         0         1         15         0         07:45         17:30         0         0         0         0           Horsham (via Dorking)         London Victoria         78         83         2         1         11         6         2         1         0         0         1         0         0         1         15         0         07:45         17:30         0         0         0         0			78	83	2 2	1 1	9 11	5 6	2 2	1	0	0	1				74	74	0	1	15	0		17:30 17:30	1		0	0	0	0	-	-

Note 1: The parliamentary service between Streatham Hill and Tulse Hill must continue to operate.

Note 2: Services from London St. Pancras to Sutton via Hackbridge and from Sutton to London St. Pancras via Wimbledon must continue to operate as the current "Wimbledon Loop" service, joined through Sutton.

Similarly, services from London St. Pancras to Sutton via Wimbledon and from Sutton to London St. Pancras via Hackbridge must continue to be joined through Sutton.

Note 3: Two additional services shall operate from West Croydon to Sutton, with a maximum interval of 15 minutes, calling at Waddon, Wallington and Carshalton Beeches, arriving before 06:15.

The second of these services shall operate from West Croydon to Sutton at a maximum interval of 30 minutes, liging at Waddon, Wallington, and Carshalton Beeches, arriving before 07:30.

The first of these two services shall operate from West Croydon to Sutton at a maximum interval of 30 minutes.

The second of these two services shall operate from Streatham Hill. calling at Waldon and Leatherhead.

The second of these two services shall operate from Streatham Hill. calling at Waldon and I Leatherhead.

The second of these two services shall operate from Streatham Hill. calling at Waldon and I Leatherhead.

The second of these two services shall operate from Streatham Hill. calling at West Morwood, Glosy Hill. Crostal Palace and Norwood Junction.

Note 5: An additional service shall operate from Norwood Junction to Tattenham Corner, calling at East Croydon and all stations from Purley, arriving before 07:15.

Secretary of the property of t	TABLE 2: NORTHBOUND  Arrival time in London	1	Journe	ey Time	0200-0659	0700-0959	0800-0859	Monday to		2000-2059	2100-2159	2200-0159	9 Service arr	ves London	Journe	y Time	0200-0659	Saturda 0700-2159 07	ay '00-2159 220	0-0159 Servic	arrives Lo	ndon	Journey T	ime lo	0200-0859	Suno 0900-2059	day 0900-2059	2100-0159	Service arr	rives Londo
Column														Last (no							Last	(no								Last (no
March   Marc								TPH					later than)	than)				TPH		otal later t	an) tha	n) Ave	erage M	aximum						
March   Marc	Warnham		79	85	0	5	2 2	1		1	1		07:15		/4	/4	0	1								- 1			-	:
Part					0	5	- 1	1		1	1	0					0	1							0	- 1		0	-	-
Column	Dorking	London			ő	7		2	20		3		07:15	23:15			0	2	30 3 (r	ote 3) 07:	5 23:	15			2		24	7		00:15
Part					0	5	1	1 0	15	0	0						0	0			5 22:	45			0	0		7	08:30	00:15
March and Marc	Guidiora	London Victoria			0	1																						_		
Mary Content	London Road (Guildford)		70	70	0	_	1 1	0	1				08:45	19:45			0	0	0	0 -					0	0	0	0	-	-
Mary No.					0	1		- T	1			1 1					0				-								-	-
March   Marc		London			0	2	1	- 1	1			1 -	08:45	19:45			0												-	
Service Servic					0	2	1	0	1 21	0	0	0 2 (noto 3)					0	0	0 2 /	0 -	- 22:				0	0	0	0		- 00:15
Septimal Control and Control a	Ashtead							2			3	3 (note 3)	07:15	23:15			0	2	30 3 (r	ote 3) 07:	5 23:	15				2	24	7	08:30	00:15
Series - Ser	Epsom		//1	//1	3		6	5			5	9 (note 3)	06:15	23:30	/1	A1	0	5			5 23:		U/Δ	N/A	3	4		7	07:45	00:15
Septiminal properties of the p		London Victoria (Hackbridge stopping)	43	45	1	9	4	2	20	2	4	7			42	42	ő	2	30	4			44	51	2	4	48	7		
Mary Control of Mary Control o					0						1						0	0							0					
Control   Cont		London Bridge (via West Croydon)			1	4	2		0	0	Ö	0					0		0	0	_	1				0	0			
Series - Ser					3	13	6	5	41 50	4	5						0	5							3	4		7		00:15
Part		London Victoria	55	60	0	5		1			2	2 (note 3)			54	54	0	1	15 2 (r	ote 3) 08:	0 23:	15							-	-
Septiminal property of the control o	Belmont				0	5	2	1		2	2						0	1							0	0	0	0	-	
See - Control of Manufall Annel Manu	Sutton (note 1)		31	31	7								06:15	01:00	31	32	2				0 23:		31	31	8				07:45	01:00
		London Victoria (Hackbridge stopping)	32	35	1		3	2	21	2	2	7			32	32	ō	2	30	4			33	40	3	4	48	7		
					1 0	9											0								2	2		8		
Separate supplies sup		London Bridge (via Wimbledon)	N/A	N/A	0	2	2		0	0	0	0			N/A	N/A	0	0	0	0		1	N/A	N/A	0	0	0	-		
Second control of the control of t		London Bridge (via Tulse Hill)			2	5		0				0					0								0	0				
March Control   March Contro		London Blackfriars (via Wimbledon)	50	50	1	6	2	2	20	2	2	4			50	50	1	2	30	4			50	50	1	2	24	2		
Mary Control Professor   Mary Control Profes	West Sutton (via Wimbledon)		34	35	0	6		2			2		07:00	23:30	34	35	1	2			0 23-		34	35	1	2		4 2	09:00	21:45
March of the Property of the Control of the Contr	Sutton Common (via Wimbledon)	London	1		2	8			20	2	2	4	07:00	23:30			1	2	30	4 07:0	0 23:	30			- 1	2	24	2	09:00	21:45
Marchane			1		2 2	8	4 4	2 2			2 2	4 4					1 1	2 2							0	2		2 2		21:45 21:45
March   Marc	South Merton (via Wimbledon)	London			2	8		2			2	4	07:00				1	2		4 07:0	0 23:	30			0	2		2	09;00	21:45
Property of Surfey   Propert					2 2							1 .					1 1											2 2		21:45
Company control   Company   Compan	Haydons Road (not via Sutton)	London			2	8					2	4	07:00	23:30			1	2		4 07:0	0 23:	30						2	09:00	21:45
Marcong   Marc					3	15	6	6	60	4	4	8 (note 3)					0	6	90 13 (						4	4	24 48	14		21:45 00:15
Martine description of the property of the pro		London			3	15	6	6			4	8 (note 3)	06:15	23:45			0	6	90 13 (	note 3) 07:					4	4	48			00:15
Manufacture					3		10	8		6	6						1	8							8	4 4				00:15
Companies   Comp		London Victoria (via Norbury)			1							4					0		60	9							24			
Control   Cont					2												0								3	2	24	6		
Marcine Cargonia	O and after	London Bridge (via Tulse Hill)		44	2		4	2		2	2	4	00.45	22-20	38	38	1	2			E 22.		38	38	2	2		5	00-00	00-45
Marie Confessor   Marie Conf	Hackbridge	London			3	19	7	4	41			7	06:15	23:30			1		60	7 06:4	5 23:	30				6	72	11	08:00	00:15
The proper work of the property of the propert					3	20	8 8	4	41 41	5	5	7 7					1 1	4							3	6		11		00:15
Part		London			1		1 : 1	2									2	2	30	4 06:		45			1	2	24	6		23:45
Table 1					0		- 1	0 2			0 2	1 -					0 2	0 2		-					0	0 2		6		1
Controllers   1 1 0 0 4 2 2 3 0 2 2 2 4 0 000 2 2 1 5 0 4 0 0 0 2 2 4 0 0 0 1 1 2 2 2 4 0 0 0 4 0 0 0 0 0 0 0 0 0 0 0 0		London	''		1	10	4		20	2		4							30	4 06:		45	-	.	1		24	6		
Windowshare					1 1																				1 1			6		23:45 23:45
Property Company   Property Co	Woodmansterne	London			1	10			20	2		4	07:00	23:45			2	2	30	4 06:	0 23:	45			1	2	24	6	07:45	23:45
Cardon London London London London London London Discoler 1					1 1		4 4	2 2		2 2	2 2	4 4					2 2	2 2							1 1	2 2		6		23:45 23:45
Mythyladed South  Mythyladed S		London			2			4					06:45			50	1			7 07:		45	40	47					08:15	23:15
Mysteled South					0												0	2	30	4					2	2	24	5		
Weighted Sputh Weight																52	1	2	30	3					1	0	0	0		
Mythelete			35	35	2							8	06:45	23:45	N/A	N/A	1	4	60	7 07:	0 23:		WA	IN/A	3	2	24	5 (note 4)	08:15	23:15
Purkey   Lunden	Whyteleafe	London	1		2			4		-	4	8		23:45			1	4		7 07:0	0 23:	45			3	- 1		5 (note 4)	08:15	23:15
Pulsy Cologs	Purley	London	1		11	30	11		100	6		14	03:00	01:45			9		150	17 03:0	0 01:	00				7	84	21 (note 4)	02:15	01:45
East Croyden	Purley Oaks		1		4							7 7					1 1		60							2			07:30	23:45
Anterly	East Croydon	London	t		31	99	35	26	264	22	22	38	02:15	01:45			16	24	364	45 02:	5 01:	45			29		235	47 (note 4)	02:15	01:45
Perge West			1		15												6								13 6					23:45 23:45
Throno-Health	Penge West	London	ļ		3	7	2	2	22	2	2	3	05:45	23:30			1	2	30	3 07:	0 23:	30			6	2	24	6	07:30	23:45
Norbury Lundon   7   22   8   8   8   80   6   6   12   65.00   22.45   8   8   80   0   6   6   12   65.00   22.45   8   8   80   0   6   6   12   65.00   22.45   8   8   80   0   6   6   12   65.00   22.45   8   8   8   8   8   8   8   8   8			1				8 8	8 8			6						6 3	8 8								7 6				01:45 00:15
Streatharn   London Bridge   35   40   0   5   25   10   6   63   6   6   8   06.15   23.30   2   2   6   00   10   06.15   23.30   4   6   72   11   0   73.00   23.15	Norbury	London	1		7	22	8	- 1	80	6	6	12	05:00	23:45			3	8	120	17 06:	0 00:	00			11	- 1	72	18	06:00	00:15
Bekenham Junction  Endom Bridge  Bekenham Junction  Endom Bridge  Bekenham Junction  Endom Bridge  Bri			1		7 5		10	6		6	6						3 2	6								6				00:15 23:15
Christal Palace (vis Giosy Hill)  Condon Victoria  28 31 2 12 4 4 40 0 4 4 9 9  Condon Victoria  28 31 12 12 4 4 40 0 4 4 9 9  Condon Victoria  28 31 12 12 4 4 4 00 4 4 4 9 9  Condon Victoria  28 31 12 12 4 4 4 0 0 4 4 4 9 9  Condon Victoria  28 31 12 12 4 4 4 0 0 4 4 4 9 9  Condon Victoria  28 31 12 12 2 2 2 2 2 2 2 6 6 0 14 6630  Condon Victoria  Condon Bridge	Beckenham Junction	London Bridge	35	40	0			2	20	2	2	4	08:00	23:45	35	35	0	2	30	4 07:	0 23:	45					0	0		-
London Rédige   28   31   2   12   4   4   40   4   4   9   9   27   28   0   4   60   10   27   32   4   2   24   6   6   6   6   6   6   6   6   6			<del>                                     </del>		3	18	6	6	60	6	6						2	6	90	14 06:		30	_		7	4	48	13	07:30	00:00
Cigsy Hill		London Victoria										9				28			60	10							24			
West Norwoord   London   3   22   7   6   60   6   6   15   06:15   00:30   2   6   90   14   06:30   00:30   7   4   48   13   07:30   07:5	Gipsy Hill	London	28	34		22			60			15			28	29	2	6	90	14 06:		30	20	20	7		48	13		00:00
Cystal Palace (via Forest Hill)	West Norwood		1		3	22	7	6		6	6	15		00:30			2	6	90	14 06:	0 00:	30				4	48		07:30	00:00
Sydenham	Crystal Palace (via Forest Hill)	London Bridge	1			13	4	2	20		2		07:15	00:30			0		30	6 07:	0 00:	30	_		0	0			-	-
Honor Calk Park	Sydenham	London Bridge	1						42				05:45	00:30			1		60	9 07:0	0 00:	30								23:45
Brockley   London Bridge   3 20 6 4 42 4 4 9 05.45 00.30   1 4 60 9 07.00 0.30   8 4 48 11 07.30 23.45	Honor Oak Park	London Bridge	1			20		4	42	4	4	9	05:45	00:30			1	4	60	9 07:0	0 00:	30				4	48	11	07:30	23:45
Tulse Hill (note 2)  London Bridge London Br	Brockley	London Bridge	1		3		6 7	4		4	4		05:45				1 2	4	60						8		48		07:30	23:45 23:45
East Dulwich   London Bridge   3   19   8   4   48   4   4   8   06:15   23:45   0   4   60   8   07:15   23:45   7   4   48   12   07:30   00:00	Tulse Hill (note 2)	London	1			31			88			12	06:15	23:45			2		120	14 06:	5 23:	45			7	8	96	18	07:30	00:00
Peckham Rye			1																						7					00:00 00:00
Queens Road Peckham   London Bridge   3   19   8   4   48   4   4   8   0615   23.45   0   4   60   8   07.15   23.45   7   4   48   12   07.30   00.05	Peckham Rye	London Bridge	1		3	19	8	4	48	4	4	8	06:15	23:45			0		60	8 07:	5 23:	45			7	4	48	12	07:30	00:00
Heme Hill			1		3		8 8	4		4	4	8 8					0	4								4				00:00
Loughborough Junction   London Blackfriars   3   12   4   4   40   4   4   3   06.30   23.30   2   4   60   6   06.45   23.15   0   4   48   6   09.15   22.35	Herne Hill	London Blackfriars	l e			12			40				06:15	23:30			2		60	6 06:	5 23:	30			0		48	6	09:15	22:30
Balham         London Victoria         10         36         13         12         120         11         11         25         05.00         00:30         4         12         180         27         06:00         00:30	Loughborough Junction		1		3	12		4						23:30			2		60						0			6	09:15	22:30 22:30
Wandsworth Common   London Victoria   4 26 9 10 100 8 8 18 06:30 00:30   3 10 150 23 06:30 00:30   7 4 49 14 07:15 00:15   17 72 26 24 230 21 21 46 06:30 01:45   11 24 360 30 01:00   25 18 216 46 02:15 01:45   11 24 360 30 01:00   25 18 216 46 02:15 01:45   30 30 30 30 30 30 30 30 30 30 30 30 30	Balham	London Victoria	1			36	13		120	11	11	25	05:00	00:30			4	12	180	27 06:0	0 00:	30				10	121	26	06:00	00:15
		London Victoria	I		4		9		100	8		18	06:30	00:30					150	23 06:	0 00:	30			7	4	49	14	07:15	00:15
		London Victoria																												

Note 1: Services from Sutton to London St. Pancras via Hackbridge and from London St. Pancras to Sutton via Wimbledon must continue to operate as the current "Wimbledon Loop" service, joined through Sutton. Similarly, services from Sutton to London St. Pancras via Wimbledon and from London St. Pancras to Sutton via Hackbridge must continue to be joined through Sutton. Note 2: The parliamentary service between Streatham Hill and Tulse Hill must continue to operate.

Note 3: Four additional services shall operate from Sutton to West Croydon, with a maximum interval of 15 minutes, calling at Carshalton Beeches, Wallington and Waddon, departing between 23:15 and 23:59. One of these services shall start at Eposom, calling at Ewell East and Cheam.

One of these services shall start at Eposom, calling at Ewell East and Cheam.

One of these services shall start at Eposom Calling at Banstead and Belmont.

A further additional service shall operate from Dorking to Eposom, calling at Isanstead and Belmont.

A further additional service shall operate from Dorking to Eposom, calling at Isanstead and Belmont.

Note 4: An additional service shall operate from Caterham to East Croydon, calling at all stations, departing after 23:15.

BML Outer 1: Brighton Main Line Minimum Service Specification (Outer Services only) and First and Last Train Specification

Departure time from London  Direct services from specified origin to each station Sanderstead Riddlesdown Uoper Warlingham							. M	londay to Frida	y								Saturda								Sat	aturday			
Sanderstead Riddlesdown	`	Journ	ney Time	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759 1	1900-1959	2000-2059	2100-2159			Last (no	Journey Ti	ne 0200-065	0700-2159	0700-2159	2200-0159		Last (no	Journe	y Time	0200-0859	9 0900-205	59 0900-205	9 2100-015	9 Service lea	Last
Riddlesdown	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total		First (no ater than)	earlier than) Av	verage Max	imum Total	TPH	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	n) than
	London London			2	2	22 20	10	4 4	2	2	2	4	06:00 06:00	23:45 23:45		2 2	2 2	30 30	4	06:15 06:15	23:45 23:45			2 2	2	24 24	2	07:45 07:45	
Woldingham	London London			2 (note 3) 2	2 2	20 20	10 10	4 4	2 2	3 2	3 2	6	06:00 06:00	23:45 23:45		2 2	2 2	30 30	5 4	06:15 06:15	23:45 23:45			2 2	2 2	24 24	2 2	07:45 07:45	22:3
Oxted Hurst Green	London London			6 (note 3) 6 (note 3)	3	31 30	14 14	5	3	3	3	6	05:30 05:30	23:45 23:15		4 4	3	45 45	6	05:15 05:15	23:45 23:30			2 2	2 2	24 24	2 2	07:45 07:45	22:3
Lingfield	London			4	2	20	10	4	2	2	2	3	05:30	23:15		3	2	30	3	05:15	23:30			2	2	24	2	07:45	22:3
Dormans East Grinstead	London London			4	2	20 20	10 10	4 4	2	2 2	2 2	3	05:30 05:30	23:15 23:15		3	2 2	30 30	3	05:15 05:15	23:30 23:30			2 2	2	24 24	2 2	07:45 07:45	22:3
	London Bridge London Victoria	52 55	58 68	0 4	0 2	3 17	4	2 2	0 2	0 2	0 2	0 3				N/A 0 68 3	0 2	0 30	0 3			N/A 55	N/A 68	0 2	0 2	0 24	0 2		
Edenbridge Town Hever	London Bridge (note 2) London Bridge (note 2)			2 (note 3) 2	1	10 10	6 (note 4) 4	2 (note 4)	1	1	1	2	06:15 06:15	23:00 22:00		1 1	1	15 15	2	06:15 06:15	23:00 22:00			0	1	12 12	2	09:45 09:45	
Cowden	London Bridge (note 2)			2	1	10	5 (note 4)	1 (note 4)	1	1	1	1	06:15	22:00		1 1	1	15	1	06:15	22:00			0	1	12	2	09:45	22:3
Ashurst Eridge	London Bridge (note 2) London Bridge (note 2)			2 2 (note 3)	1	10 10	4 6 (note 4)	2 (note 4)	1	1	1	2	06:15 06:15	22:00 23:00		1 1	1	15 15	2	06:15 06:15	22:00 23:00			0	1	12 12	2 2	09:45 09:45	22:3
Crowborough Buxted	London Bridge (note 2) London Bridge (note 2)			2 (note 3) 2 (note 3)	1 1	10 10	6 (note 4) 6 (note 4)		1	1 1	1 1	2 2	06:15 06:15	23:00 23:00		1 1	1 1	15 15	2 2	06:15 06:15	23:00 23:00			0	1 1	12 12	2 2	09:45 09:45	22:3
Uckfield Coulsdon South	London Bridge (note 2) London	76	82	2 (note 3)	1 4	10	6 (note 4)	2 (note 4)	1	1 2	1 2	2	06:15 05:00	23:00 23:45	76	82 1	1 4	15 60	2	06:15 05:00	23:00 23:45	76	82	0	1 3	12 36	2	09:45 06:30	22:3
Merstham	London			6	3	33	13	4	5	2	2	5	05:00	23:45		4	3	57	5	05:00	23:45			3	2	24	6	06:30	23:4
Redhill	London London Blackfriars			0	6	54 14	17	1	0	0	0	10	04:00	01:00		0	2	90 30	10	04:00	01:00			10	0	48	12	04:00	01:0
	London Bridge London Victoria			0 7	2	22 18	6	2 2	2	2	2	3 7				1 5	2	30 30	3 7					2 8	2	24	4 8		
Reigate (note 1)	London London Bridge	41	42	1 0	1 0	9	5 2	2	1	0	0	0	07:00	19:45	N/A	1 N/A 0	1 0	15	0	06:00	18:15	N/A	N/A	0	0	0	0	-	-
	London Victoria	41 37	42 48	1	1	8	3	1	ó	0	0	ō				48 1	1	15	0			N/A N/A	N/A N/A	0	0	0	0		
Nutfield Godstone	London London			3	1 1	11 11	5	1 1	0	0	0	0	06:00	18:15 18:15		1 1	1 1	15 15	0	06:30 06:30	18:45 18:45			0	1	12 12	2 2	09:00	
Edenbridge Penshurst	London London			3	1	11	5	1 1	0	0	0	0	06:00	18:15 18:15		1	1	15 15	0	06:30 06:30	18:45 18:45			0	1	12	2	09:00	22:0
Leigh	London			3	1	11	5	1	0	0	0	0	06:00	18:15		1	1	15	0	06:30	18:45			0	1	12	2	09:00	22:0
Tonbridge (note 1)	London London Bridge	67	67	3	1 0	11	5	1 1	0	0	0	0	06:00	18:15	N/A	N/A 0	0	15 0	0	06:30	18:45	67	67	0	1	12 12	2	09:00	22:0
Earlswood	London Victoria London	67	93	3	1 2	7	2	0	0	0	0	0	05:30	23:00		93 1	1 2	15	0	05:30	23:15	N/A	N/A	0	0	0	0	08:30	22:3
Salfords	London			4	2	19	11	4	3	1	1	2	05:30	23:00		4	2	30	3	05:30	23:15			1	1	12	2	08:30	22:3
forley Satwick Airport (including Gatwick Express)	London London	1		9 32	3 18 (note 7)	27 156	17 47	6 15	5 17	2 14	2 12	6 31	02:00 02:00	01:00 01:30		7 28	3 18	45 270	6 34	02:00 02:00	01:00 01:00			10 37	2 14	24 165	7 40	02:00 02:00	
	London Blackfriars	48	56 52	11	4	34	10	3	4	2	2	4			48	56 10	4	56	7					5	4	44	8		1
	London Bridge of which	41 29	53 30	0	2	35 18	9	0	2	2	0	4 0			29	53 1 30 0	4 2	61 30	0			N/A	N/A	2	1	12	2		
Gatwick Airport (Express services only)	London Victoria London Victoria (note 5)	34 N/A	55 31 (note 6)	21 10	10 4	87 36	28 12	10 4	10 4	8	8	23 12	03:30	01:30	34	55 17 note 6) 10	10 4	149 60	23 10	03:30	00:30	N/A	31 (note 6)	30	9	109 48	30 16	03:30	01:3
Three Bridges	London	l		22 (note 8)	8	75	39	13	11	8	6	19	02:00	01:00	31	17 (note 8	8	120	19	02:00	01:00	.40	J. (. Die 0)	17 (note 12		3) 80	21	02:00	
	London Blackfriars London Bridge	52 41	57 58	11 0	2 4	19 35	6 15	5	4 3	4	2 2	7				9	4	30 60	7					5 2	1	44 12	6 2		
Crawley	London Victoria London	42	60	11 3 (note 8)	2	20 36	18 15	5	4	2	2	9	06:00	23:45		7 3 (notes 8 &	0) 4	30 60	8	06:00	23:45			10 4 (note 12	2 2	24 24	7 5	07:00	23:0
field	London			2 (note 8)	2	19	9	3	4	2	3	6	06:30	23:45		3 (notes 8 &		30	4	06:00	23:45			3 (note 12	2) 1	12	3	07:00	
Faygate Littlehaven	London London	1		0 2 (note 8)	0 2	7 19	4 11	1 4	4	0 2	3	6	07:15 06:30	19:00 23:45		0 3 (notes 8 &	0 2	30	4	06:00	23:45			0 3 (note 12	2) 1	12	3	07:00	
Horsham	London London Bridge	71	75	3 (note 8)	4	36 18	15	5	6	4 2	4 2	7	06:00	23:45	71	3 (notes 8 & 75 1		60	5	06:00	23:45	71	75	4 (note 12		24	5	07:00	
	London Victoria	71 55	75 83	3	2	18	8	2	4	2	2	4				83 2	2	30	1			71 63	75 83	2 2	1	12	3		
Christ's Hospital Billingshurst	London London			2 (note 8) 2 (note 8)	1 2	12 18	7 7	3 3	2 2	2 2	2 2	1	06:00 06:00	22:30 22:30		2 (notes 8 & 2 (notes 8 &		15 30	1	06:00 06:00	22:30 22:30			1 2	1	12 12	2 2	08:00 07:00	22:0
Pulborough	London	1		2 (note 8) 2 (note 8)	2	18 11	7	3 2	2 2	2 2	2 2	1	06:00	22:30 22:30		2 (notes 8 & 2 (notes 8 &	0) 2	30 15	1	06:00 06:00	22:30 22:30			2	1	12	2	07:00	22:0
Amberley Arundel	London London			2 (note 8) 2 (note 8)	2	11	7	3	2	2	2	1	06:00	22:30		2 (notes 8 & 2 (notes 8 &		30	1	06:00	22:30			2	1	12	2	08:00 07:00	22:0
Ford Barnham	London London			2 (note 8) 2 (note 8)	2	18 18	5 7	2 3	2	2	2	2	06:00	22:30 22:30		2 (notes 8 & 2 (notes 8 &		30 30	3	06:00 06:00	23:15 23:15			2	1 2	12	2	07:00 07:00	
Bognor Regis	London	98	98	1 (note 8)	2	18	7	3	2	2	2	1 0	06:00	22:15	N/A	0 N/A 0	2	30	1	07:30	23:15	21/2	N/A	0	1	12	2	07:00	22:0
	London Bridge London Victoria	98 104	98	0	2	18	6	2	0 2	2	0 2	1			104	99 0	2	30	1			N/A 104	N/A 99	2	1	0 12	2		
Chichester Fishbourne	London Victoria London Victoria			1 (note 8) 0	0	18 0	6	2 2	1	1	1	0	06:30 17:15	22:30 21:30		2 (notes 8 & 0	0 2	30 1	0	06:00	23:15			0	0	13	0	08:15	21:1
Bosham Nuthoume	London Victoria London Victoria			0	0	0	5 2	2	1	1	1	0	16:45 17:15	21:30 21:30		0	0	1 1	0	-	-			0	0	0	1	21:30 21:30	
Southbourne	London Victoria			0	1	9	6	2	2	2	2	0	07:15	21:30		1 (note 8)	1	15	0	06:00	21:30			1	1	13	1	08:15	21:1
Emsworth Warblington	London Victoria London Victoria			0	1 0	9	6 2	2	2	2	2	0	07:15 17:15	21:30 21:30		1 (note 8)	1 0	15 1	0	06:00	21:30			1 0	1 0	13	1 0	08:15	21:1
Havant	London Victoria			0 (note 8)	2	18	6	2	2	2	2	0	07:15 16:15	21:30 17:00		2 (notes 8 &	0) 2	30	0	06:00	21:30			1	1	13	1	08:15	21:1
Bedhampton Hilsea	London Victoria London Victoria			0	0	0	2	1	0	0	0	0	16:15	17:00		0	0	0	0	-	-			0	0	0	0	-	+ :
Fratton Portsmouth & Southsea	London Victoria London Victoria	118	142	0	1	9	3	1 1	1 1	1	1 1	0	08:00 08:00	21:30 21:30	118	1 (note 10 142 1 (note 10		15 15	0	06:30 06:30	21:30 21:30	118	142	1 1	1 1	13 13	1 1	08:15 08:15	
Portsmouth Harbour Cosham	London Victoria London Victoria			0 0 (note 8)	1	9	2	0	0	1 1	1	0	08:00 07:15	20:00		1 (note 10 1 (note 8)	1 1	15 15	0	06:30 06:00	20:00 20:30			1 0	1 0	13	1 0	08:15	21:1
Portchester	London Victoria			0 (note 8)	1	9	2	1	1	ó	ó	ő	08:30	19:30		1 (note 8)	1	15	0	06:00	20:30			0	0	0	0	-	-
Fareham Swanwick	London Victoria London Victoria	1		0 (note 8) 0 (note 8)	1 1	9	3	1	1 1	1 1	1 1	0	07:15 07:15	20:30 20:30		1 (note 8) 1 (note 8)	1 1	15 15	0	06:00 06:00	20:30 20:30			0	0	0	0	1 :	:
Netley Woolston	London Victoria London Victoria	1		0 (note 8) 0 (note 8)	0	1 1	0	0	0	0	0	0	15:45 15:45	15:30 19:30		0	0	0	0	:	:			0	0	0	0	-	:
St. Denys	London Victoria	l .		0	0	1	0	0	1	0	0	0	15:45									i							1 -
Southampton Central Balcombe	London Victoria London	149	164	0 (note 8) 4	1 1	9	3	1 1	1					19:30		0	0	0	0					0	0	0	0	-	1 -
	London London Blackfriars	1					6	1	1	1	1	3	07:15 05:00	20:30 23:30	149	0 0 164 1 (note 8)		0 15 15	0 0 4	06:00 04:30	20:30 23:30	149	164	0 0 3	0 0	0 0 12	0 0 3	06:30	
Haywards Heath				15	6	56 19	32	1 13	1 8	1 8	1 6	12	07:15	20:30	149	0 0 164 1 (note 8) 5 9	0 1 1 6	90	0 0 4 16			149	164	0 0 3 13	0 0 1 5	0 0 12 60	0 0 3 15	-	
Haywards Heath	London Bridge			15 7 0	6 2 2	19 17	32 7 9	2 3	1 8 2 2	1 8 2 2	1 6 2 0	12 3 0	07:15 05:00	20:30 23:30	149	0 0 164 1 (note 8) 5 9 6 0	0 1 1 6 2 2	90 29 28	0 0 4 16 7	04:30	23:30	149	164	0 0 3 13 5 0	0 0 1 5	0 0 12 60 24 0	5	06:30	
	London Bridge London Victoria			15 7	6 2 2 2 2 3	19	32		1 8 2 2 4 4	1 8 2 2 4 4	1 1 6 2 0 4	12	07:15 05:00	20:30 23:30	149	0 0 1 (note 8) 5 9 6 0 3 7 (note 9)	0 1 1 6	90 29	0 0 4 16 7 1 8 8	04:30	23:30	149	164	0 0 3 13	0 0 1 5 2 0 3 2	0 0 12 60		06:30	01:0
Wivelsfield Plumpton	London Bridge London Victoria London London			15 7 0 8 9 (note 9) 1 (note 9)	6 2 2 2 2 3	19 17 20 28 9	32 7 9 17 12 5	2 3 7 5	2	2 4 4 1	0	12 3 0 9 5	07:15 05:00 04:00 04:00 06:45	20:30 23:30 01:00 23:30 22:45	149	5 9 6 0 3	0 1 1 6 2 2 2 2 3	90 29 28 30 45	7 1 8 8	04:30 04:00	23:30 01:00	149	164	0 0 3 13 5 0 8 5	0 3 2	0 0 12 60 24 0 36 24 12	5 0 10 5	06:30 04:00	23:3
Wivelsfield Plumpton Coksbridge -ewes	London Bridge London Victoria London London London London London	65	67	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9)	6 2 2 2 3 1 0 2	19 17 20 28 9 3 18	32 7 9 17 12 5 5	2 3 7 5 2 2 2 3	2 4 4 1 1 2	2 4 4 1 0 2	0 4 4 1 1 2	12 3 0 9 5 1 0 2	07:15 05:00 04:00	20:30 23:30 01:00		5 9 6 0 3 7 (note 9) 0 0 67 0 (note 9)	0 1 1 6 2 2 2 2 3 1 0 2	90 29 28 30 45 15 0	7 1 8 8 1 0 2	04:30 04:00	23:30 01:00 23:45	149	164	0 0 3 13 5 0 8 5	0 3 2 1 0	0 0 12 60 24 0 36 24 12 0	5 0 10 5 2 0 3	06:30 04:00	23:3
Whelsfield Plumpton Cookabridge Cookabridge Southwase Newhasen Town	London Bridge London Victoria London London London London London London Victoria London Victoria	65	67	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 0 (note 9)	6 2 2 2 3 1 0 2	19 17 20 28 9 3	32 7 9 17 12 5	2 3 7 5	2 4 4 1 1 2 0 0	2 4 4 1 0 2 0	0 4 4 1 1 2 0 0	12 3 0 9 5	07:15 05:00 04:00 04:00 06:45 06:45	20:30 23:30 01:00 23:30 22:45 21:45		5 9 6 0 3 7 (note 9) 0	0 1 1 6 2 2 2 2 2 3 1 0 0	90 29 28 30 45 15 0	7 1 8 8 1 0 2	04:30 04:00 04:30 08:30	23:30 01:00 23:45 22:45		104	0 0 3 13 5 0 8 5 1 0	0 3 2	0 0 12 60 24 0 36 24 12 0 12	5 0 10 5 2 0 3	06:30 04:00 05:45 08:45	23:3 22:4
Wivelsfield Plumpton Cooksbridge Leves Southease Venhauen Town Venhauen Harbour	London Bridge London Victoria London London London London London London Victoria London Victoria London Victoria	65	67	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9)	0	19 17 20 28 9 3 18	32 7 9 17 12 5 5	2 3 7 5 2 2 2 3	2 4 4 1 1 2 0 0	2 4 4 1 0 2 0 0	0 4 4 1 1 2 0 0	12 3 0 9 5 1 0 2	07:15 05:00 04:00 04:00 06:45 06:45 05:30 -	20:30 23:30 01:00 23:30 22:45 21:45 23:45 		5 9 6 0 3 7 (note 9) 0 0 67 0 (note 9)	0 1 1 6 2 2 2 2 2 3 1 0 0 0 0	90 29 28 30 45 15 0 30	7 1 8 8 8 1 0 2 0 0	04:30 04:00 04:30 08:30	23:30 01:00 23:45 22:45		104	0 0 3 13 5 0 8 5 1 0	0 3 2 1 0 1	0 0 12 60 24 0 36 24 12 0 12 0	5 0 10 5 2 0 3 0 0	06:30 04:00 05:45 08:45 08:45	23:3 22:4
Wivelsfield Plumpton Cooksbridge Leves Southease Newhaven Town Newhaven Harbour Blahopstone Beatord	London Bridge London Victoria London London London London London London London Victoria	65	67	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9)	0	19 17 20 28 9 3 18 0	32 7 9 17 12 5 5 7 0	2 3 7 5 2 2 3 0	2 4 4 1 1 2 0 0	2 4 4 1 0 2 0	0 4 4 1 1 2 0 0	12 3 0 9 5 1 0 2	07:15 05:00 04:00 04:00 06:45 06:45 05:30 	20:30 23:30 01:00 23:30 22:45 21:45 23:45 17:45 17:45 17:45	65	5 9 6 0 3 7 (note 9) 0 0 67 0 (note 9)	0 1 1 6 2 2 2 2 2 3 1 0 0	90 29 28 30 45 15 0 30 0	7 1 8 8 1 0 2	04:30 04:00 04:30 08:30	23:30 01:00 23:45 22:45 23:45		104	0 0 3 13 5 0 8 5 1 0 1	0 3 2 1 0 1	0 0 12 60 24 0 36 24 12 0 12	5 0 10 5 2 0 3	06:30 04:00 05:45 08:45	23:3 22:4 
Whelsfield Plumpton Contarions Contarions Contarions Contarions Control Contro	London Bridge London Victoria London London London London London London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria			15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9)	0	19 17 20 28 9 3 18 0	32 7 9 17 12 5 5 7 0	2 3 7 5 2 2 3 0	2 4 4 1 1 2 0 0	2 4 4 1 0 2 0 0	0 4 4 1 1 2 0 0	12 3 0 9 5 1 0 2	07:15 05:00 04:00 04:00 06:45 06:45 05:30 18:00 18:00 18:00 05:30	20:30 23:30 01:00 23:30 22:45 21:45 23:45 - 17:45	65	5 9 6 0 3 7 (note 9) 0 0 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 2 3 1 0 0 0 0	90 29 28 30 45 15 0 30 0	7 1 8 8 8 1 0 2 0 0	04:30 04:00 04:30 08:30 07:45	23:30 01:00 23:45 22:45 23:45	65	67	0 0 3 13 5 0 8 5 1 0	0 3 2 1 0 1	0 0 12 60 24 0 36 24 12 0 12 0	5 0 10 5 2 0 3 0 0	06:30 04:00 05:45 08:45 08:45	23:3 22:4 
Whelsfield Plumpton Cocksbridge Leves Leves Southease Southease Southease Seaford Slabpoptone Seaford Servick Phopate Denvick	London Bridge London Victoria London London London London London London London London Victoria			15 7 0 8 9 (note 9) 1 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2	0 0 0 0 0	19 17 20 28 9 3 18 0 0 0 0 0 1	32 7 9 17 12 5 5 7 0	2 3 7 5 2 2 2 3 0 1 0 1 1 2 2 3	2 4 4 1 1 2 0 0 0 0 0 0 0	2 4 1 0 2 0 0 0 0 0 0	0 4 4 1 1 2 0 0 0 0 0 0	12 3 9 5 1 0 2 0 0 0 0 0	07:15 05:00 04:00 04:00 06:45 06:45 05:30 18:00 18:00 05:30 05:30	20:30 23:30 01:00 23:30 22:45 21:45 23:45 17:45 17:45 17:45 21:15 22:45 23:45	65	5 9 6 0 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 3 3 1 0 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 3 3 3 3 3 0	7 1 8 8 8 1 0 2 0 0	04:30 04:00 04:30 08:30 07:45	23:45 22:45 22:45 - - - - - - - - -	65	67	0 0 3 13 5 0 8 5 1 0 1 0 0 0	0 3 2 1 0 1 0 0 0 0	0 0 12 60 24 0 36 24 12 0 0 0 0 0 0 0 0 0 12	5 0 10 5 2 0 3 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45	23:3 22:4 
Wivehsfield Wivehsfield Windows Windows Leves Southease Southease Seatord Selexpetrore	London Bridge London Victoria London Victoria	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2	0 0 0 0 0 2 2 2	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18	32 7 9 17 12 5 5 7 0 1 1 0 1 1 3 4 7 7	2 3 7 5 2 2 3 0 1 1 1 1 2 2 3 3	2 4 1 1 2 0 0 0 0 0 1 1 1 2 2	2 4 1 0 2 0 0 0 0 0 0 1 1 1 2 2	0 4 4 1 1 2 0 0 0 0 0 0 1 1 1 2 2	12 3 0 9 5 1 0 0 0 0 0 0 0 1 2 2 2	07:15 05:00 04:00 04:00 06:45 06:45 06:45 05:30 - 18:00 18:00 18:00 05:30	23:30 01:00 23:30 01:00 22:45 21:45 23:45 -17:45 17:45 17:45 17:45 21:15 22:45	65	5 9 6 0 0 3 3 7 (note 9) 0 (note 9)	0 1 1 6 2 2 2 2 3 1 0 0 0 0 0 0 0 0 0 2 2 2 2 2 2 2 2 2 2	90 29 28 30 45 15 0 30 0 0 0 0 0 3 3 3 3 3 3 3 3 3 3 3 3	7 1 8 8 8 1 0 2 0 0 0 0 0 0 0 1 2 2 2 3	04:30 04:00 04:30 08:30 07:45	23:45 22:45 23:45 23:45	65	67	0 0 3 13 5 0 8 5 1 0 0 0 0 0 0	0 3 2 1 0 1 0 0 0 0 0 0 0	0 0 0 12 60 24 0 36 24 12 0 0 0 0 0 0 0 0	5 0 10 5 2 0 3 0 0 0 0 0 0 0 0 0 0 3 3 3 3 3 3 3	06:30 04:00 05:45 08:45	23:3 22:4 23:4 - - - - - - - - - - - - - - - - - - -
Wivehsfield Wivehsfield Windows Windows Leves Southease Southease Seatord Selexpetrore	London Bridge London Victoria London London London London London London London London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Lo	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2	0 0 0 0 0 2 2 2	19 17 20 28 9 3 18 0 0 0 0 0 1 18 18 18	32 7 9 17 12 5 7 0 1 0 1 1 1 3 4 7 7 7	2 3 7 5 2 2 3 0 1 0 1 1 2 2 3 3	2 4 4 1 1 2 0 0 0 0 0 0 0 1 1 2 2	2 4 4 1 0 2 0 0 0 0 0 0 1 1 1 2 2	0 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 1 2 2	12 3 0 9 5 1 0 2 0 0 0 0 0 0 1 2 2	07:15 05:00 04:00 04:00 06:45 06:45 06:30 18:00 18:00 05:30 05:30 05:30 05:45	20:30 23:30 01:00 23:30 22:45 21:45 23:45 17:45 17:45 17:45 21:15 22:45 23:45 23:45 23:45	65 92	5 9 9 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 3 1 1 0 2 0 0 0 0 0 0 0 2 2 2 2 2 2 3	90 29 28 30 45 15 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 8 1 0 2 0 0 0 0 0 0 0 0 1 2 2 3 3 0 0	04:30 04:00 04:30 08:30 07:45	23:45 22:45 23:45 22:45 	65 92 N/A	67 92	0 0 3 13 5 0 8 5 1 0 0 0 0 0 0	0 3 2 1 0 1 0 0 0 0	0 0 12 60 24 0 36 62 12 0 0 0 0 0 0 0 0 12 12 12 12 0 0 0 0	5 0 10 5 2 0 3 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45	23:3 22:4 23:4 - - - - - - - - - - - - - - - - - - -
Whelsfield Plumpton Cockstridge Lowes Lowes Lowes Blance B	London Bridge London Victoria London London London London London London London London London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2	0 0 0 0 0 2 2 2 2	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18	32 7 9 17 12 5 5 7 0 1 0 1 1 1 3 4 7 7 7	2 3 7 5 5 2 2 2 3 3 0 1 1 2 2 3 3 3 3 1 2 2 2	2 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 0 0 0	2 4 4 1 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 0 0 0 0	0 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 1 2 2 2 0 0 0 0	12 3 0 9 5 1 0 0 0 0 0 0 0 0 1 2 2 2 0 0 0 0 0 0 0	07:15 05:00 04:00 04:00 06:45 06:45 06:30 18:00 05:30 05:30 05:30 05:30	20:30 23:30 01:00 23:30 01:00 22:45 21:45 23:45 23:45 21:15 22:45 23:45 23:45 23:45 23:45 23:45	65 92	5 9 6 0 0 3 3 7 (note 9) 0 (note 9)	0 1 1 6 6 2 2 2 2 3 3 1 0 0 0 0 0 0 0 0 0 0 2 2 2 2 0 0 2 1 1	90 29 28 30 45 15 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 2 2 3 3 0 0 1	04:30 04:00 04:30 08:30 07:45	23:45 22:45 23:45 22:45 	65	67	0 0 3 13 13 5 0 8 5 5 1 1 0 0 0 0 0 0 0 1 1 1 1 0 0 1 1 1 1	0 3 2 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1	0 0 12 60 24 0 36 24 12 0 0 0 0 0 0 0 0 12 12 12 0 0 0 0 0 0	5 0 10 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45	23:3 22:4 - - 23:4 - - - - - - - - - - - - - - - - - - -
Plumpton Cookebridge Lewes Southease Southease Newharen Torn Newharen Farbour Saaford Ginnde Berwick Polegate Hampden Park Eastbourne Pewensey & Wesham Pewensey Bay Normans Bay	London Bridge London Victoria London Victoria London Victoria	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2 2 0 2 2 1 1	0 0 0 0 0 2 2 2	19 17 20 28 9 3 18 0 0 0 0 0 1 18 18 18	32 7 9 17 12 5 5 7 0 1 1 1 3 4 7 7 7 7 2 5 5 4 0 0	2 2 3 7 7 5 5 2 2 2 3 3 0 1 1 2 2 2 3 3 3 1 1 2 2 2 0 0 0	2 4 4 1 1 2 0 0 0 0 0 0 0 1 1 2 2	2 4 4 1 0 2 0 0 0 0 0 0 1 1 1 2 2	0 4 4 1 1 1 2 0 0 0 0 0 0 0 0 0 1 1 1 2 2 2 2	12 3 0 9 5 1 0 2 0 0 0 0 0 0 1 2 2	07:15 05:00 04:00 04:00 06:45 06:45 06:45 05:30 18:00 18:00 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30	20:30 23:30 01:00 01:00 23:30 22:45 23:45 23:45 17	65 92	5 9 9 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 3 1 1 0 2 0 0 0 0 0 0 0 2 2 2 2 2 2 3	90 29 28 30 45 15 0 0 0 0 0 0 0 0 3 3 3 3 3 0 0 0 0 0 0	7 1 8 8 8 1 0 2 0 0 0 0 0 0 0 0 1 2 2 3 3 0 0	04:30 04:00 04:30 08:30 07:45 - - 07:45 07:45 07:45	23:45 22:45 22:45 	65 92 N/A	67 92	0 0 3 13 5 0 8 5 1 0 0 0 0 0 0	0 3 2 1 0 1 0 0 0 0 0 0 0 0 0	0 0 12 60 24 0 36 24 12 0 0 0 0 0 0 0 0 12 12 12 0 0 0 0 0 0	5 0 10 5 2 0 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45 08:45 08:45	23:3 22:4 
WiveIsfield Plumptor	London Bridge London Victoria London Victoria London London London London London London London London London Victoria London Londo	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2	0 0 0 0 0 2 2 2 2 0 0	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18 18 0 18 2	32 7 9 17 12 5 5 7 0 1 1 1 1 3 4 7 7 7 2 5 5	2 2 3 7 5 5 2 2 2 3 3 0 1 1 1 1 2 2 2 3 3 3 3 1 1 2 2 0 0 0 2 2 2	2 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 0 0 0 0	2 4 4 1 1 0 0 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2 2	0 4 4 1 1 1 2 2 0 0 0 0 0 1 1 1 2 2 2 2 1 0 0 0 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 06:45 06:45 06:45 05:30 18:00 18:00 05:30 05:30 05:30 05:30 05:30	20:30 23:30 01:00 10:00 22:45 21:45 21:45 21:45 17:45 17:45 21:15 22:45 23:45 23:45 23:45 23:45 23:45 23:45	65 92	5 9 6 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 3 3 1 0 0 2 2 0 0 0 0 0 0 2 2 2 2 1 1 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 0 3 3 3 3 3 3 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 2 2 3 3 0 0 1	04:30 04:00 04:30 08:30 07:45 07:45 07:45	23:45 22:45 22:45 23:45 23:45 23:45 23:45 23:45	65 92 N/A	67 92	0 0 3 13 13 5 0 8 8 5 1 0 0 0 1 1 1 1 0 0 1 1 1 0 0 1 1 1 0 0 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0	0 3 2 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1 0 1	0 0 12 60 24 0 4 12 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0 10 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45 08:45 08:45	01:0 23:3 22:4 - - - - - - - - - - - - - - - - - - -
WiveIsfield Plumotion Cockebridge Cockebridge Cockebridge Southease Newhaven Town Newhaven Harbour Bishcopstone Seaford Glynde Berwick Berwick Berwick Eastbourne Pevensey & Westham Pevensey & Westham Pevensey Bay Normans Bay Collination	London Bridge London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria	92	92	15 7 8 8 9 (note 9) 1 (note 9) 1 (note 9) 1 0 (note 9) 1 1 2 2 2 2 0 0 2 2 1 1 1 2 2 2 2 2 2 2	0 0 0 0 0 2 2 2 2 0 0	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18 18 0 18 2	32 7 9 17 12 5 5 5 7 0 1 0 1 1 1 3 4 4 7 7 7 2 5 5	2 2 3 7 5 5 2 2 2 3 3 0 0 1 1 0 1 1 1 2 2 3 3 3 3 1 1 2 2 0 0 0 2 2 2 2 2 2	2 4 4 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 2 0 0 1 1 1 2 1 0 0 1 1 1 0 0 1 1 1 0 0 1 1 1 0 0 1	2 4 4 1 1 0 2 0 0 0 0 0 0 1 1 1 2 2 2 2 0 0 2 1 1 0 0 0 1 1 1 1	0 4 4 1 1 1 2 0 0 0 0 0 0 1 1 1 2 2 2 2 1 0 0 0 1 1 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 04:00 06:45 06:45 06:45 06:30 - 18:00 05:30 05:30 05:30 07:00 07:00 07:00 07:30 05:30 05:30	20:30 23:30 01:00 01:00 23:30 22:45 22:45 23:45 21:45 23:45 21:15 22:45 23:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24:45 24	65 92	5 9 9 6 0 3 3 7 (note 9) 0 0 0 1 (note 9) 0	0 1 1 6 2 2 2 2 3 3 1 0 0 2 2 0 0 0 0 0 0 2 2 2 2 1 1 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 0 3 3 3 3 3 3 3 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 2 2 3 3 0 0 1	04:30 04:30 08:30 07:45 07:45 07:45 07:45 07:45 07:45	23:30 01:00 23:45 22:45 23:45 23:45 23:45 23:45 22:45 22:45 22:45 22:45 22:45	65 92 N/A	67 92	0 0 0 3 13 5 0 8 5 1 0 0 0 0 0 0 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 1 0 0 1 1 1 1 1 0 0 1	0 3 2 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1 0 1	0 0 0 12 24 0 0 0 0 0 0 0 0 0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:4 23:4 
Whelsfield Plumpton Coackaindige Coackaindige Coackaindige Southease Newharean Town Newharean Town Newharean Town Newharean Town Seartord Glynde Berwick Planden Berwick Berwich Berwi	London Bridge London Victoria London Victoria London Victoria	92	92	15 7 0 8 9 (note 9) 1 (note 9) 1 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2 2 0 2 2 1 1	0 0 0 0 0 2 2 2 2 0 0	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18 18 0 18 2	32 7 9 17 12 5 5 7 0 1 1 0 1 1 3 4 7 7 7 7 2 5 5	2 2 3 7 5 5 2 2 2 3 3 0 1 1 1 1 2 2 2 3 3 3 3 1 1 2 2 0 0 0 2 2 2	2 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 0 0 0 0	2 4 4 1 0 0 2 0 0 0 0 0 0 0 1 1 1 2 2 2 2 2 0 0 0 0	0 4 4 1 1 2 0 0 0 0 0 0 1 1 2 2 2 2 0 0 2 1 1 0 0 0 1 1 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 04:00 06:45 06:45 06:45 05:30 18:00 18:00 05:30 05:30 05:30 07:00 07:00 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30	20:30 01:00 01:00 23:30 01:00 23:30 22:45 21:45 23:45 21:45 22:45 23:45 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 24:55 25 26 26 26 26 26 26 26 26 26 26 26 26 26	92 N/A 88	5 9 9 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 3 3 1 0 0 2 2 0 0 0 0 0 0 2 2 2 2 1 1 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 2 2 3 0 0 0 0 1 1 1 2 3 3 0 0 1 1 1 2 3 3 1 1 1 1 2 3 3 1 1 1 1 1 1 1	04:30 04:00 04:30 08:30 07:45 - - - 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:30 01:00 23:45 22:45 23:45 23:45 23:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45	65 92 N/A	67 92	0 0 3 13 13 5 0 8 8 5 1 0 0 0 1 1 1 1 0 0 1 1 1 0 0 1 1 1 0 0 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 1 1 1 1 0	0 3 2 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1 0 1	0 0 0 0 244 0 365 244 0 0 12 2 0 0 0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:4 - 23:4
Winshield Winshield Winshield Donkship Onkship	London Bridge London Victoria London Victoria London Victoria	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 1 2 (note 9) 0 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2 2 2 2 1 1 1 2 2 2 2 2 2 2 2	0 0 0 0 0 2 2 2 2 0 0	19 17 20 28 9 3 18 0 0 0 0 0 1 18 18 0 18 2 2 2 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	32 7 9 17 12 5 5 7 0 1 1 0 1 1 1 3 4 7 7 7 7 2 5 5 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2 2 3 7 5 5 2 2 2 3 3 1 1 2 2 2 3 3 3 3 1 1 2 2 2 0 0 0 2 2 2 2 2 2 2 2 2 2 2 2	2 4 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 2 1 0 0 0 1 1 1 1	2 4 4 1 1 0 2 2 0 0 0 0 0 1 1 1 2 2 2 0 0 0 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 1 1 1 2 2 2 0 0 1 1 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 0 0 1 1 0 1 1 0 1 1 0 1 1 1 0 1 1 1 0 1	07:15 05:00 04:00 04:00 04:00 06:45 06:45 06:45 05:30 - 18:00 - 18:00 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30	20:30 23:30 01:00 23:30 01:00 23:30 22:45 22:45 23:45 24:55 24:55 24:55 24:55 24:55 24:55 24	92 N/A 88	5 9 9 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 6 2 2 2 3 3 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	04:30 04:30 08:30 08:30 07:45 - - - 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:45 22:45 23:45 23:45 23:45 23:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 3 5 0 0 8 8 5 1 0 0 1 1 0 0 0 0 0 0 0 1 1 1 1 0 0 1	0 3 2 1 1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 04:00 05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:4 
Wivelsfield Wivelsfield Windows Windows Western Southease Southease Southease Seatord Starter Seatord Starter Seatord	London Bridge London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 0 (note 9) 1 1 2 2 2 2 2 1 1 1 2 2 2 2 2 2 2 2 2	0 0 0 0 0 2 2 2 2 0 0	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18 18 0 18 2	32 7 9 17 12 5 5 5 7 0 1 0 1 1 1 3 4 4 7 7 7 2 5 5	2 2 3 7 5 2 2 2 3 0 1 1 0 1 1 1 1 2 2 3 3 3 3 1 1 2 2 0 0 0 2 2 2 2 2 2 2	2 4 4 1 1 2 0 0 0 0 0 0 0 1 1 2 2 2 2 0 0 1 1 1 2 1 0 0 1 1 1 0 0 1 1 1 0 0 1 1 1 0 0 1	2 4 4 1 1 0 2 0 0 0 0 0 0 1 1 1 2 2 2 2 0 0 2 1 1 0 0 0 1 1 1 1	0 4 4 1 1 1 2 0 0 0 0 0 0 1 1 1 2 2 2 2 1 0 0 0 1 1 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 04:00 06:45 06:45 06:45 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30 06:30	20:30 23:30 01:00 23:30 22:45 21:45 23:45 21:45 23:45 21:45 23:45	92 N/A 88	5 9 9 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 3 3 1 0 0 2 2 0 0 0 0 0 0 2 2 2 2 1 1 0 0 0 0	90 29 28 30 45 15 15 0 0 30 30 30 30 30 31 5 15 15 15 15 15 15 15 15 15 15 15 15	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 2 2 3 0 0 0 0 1 1 1 2 3 3 0 0 1 1 1 2 3 3 1 1 1 1 2 3 3 1 1 1 1 1 1 1	04:30 04:30 08:30 07:45 	23:45 22:45 23:45 23:45 23:45 23:45 23:45 22:45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 13 5 0 8 5 1 0 0 0 0 0 0 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 0 0 0 1 1 1 1 0 0 1 1 1 1 1 0 0 1	0 3 2 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1 0 1	0 0 0 0 244 0 365 244 0 0 12 2 0 0 0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:32 22:43 23:40 00:00 00:00 21:41 21:41 21:41 21:41 19:40 01:51
Whelsfield Whelsfield Whenthor Doubthidge Outhhease Wesharen Town Wesharen Town Wesharen Town Wesharen Harbour Balkoptstone Selford Singhold Windel Wesharen Town Wesharen Town Wesharen Town Wesharen Harbour Balkoptstone Selford Wesharen	London Bridge London Victoria London Victoria London London London London London London London London London Victoria	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 1 2 (note 9) 0 0 (note 9) 0 (note 9) 0 (note 9) 1 1 2 2 2 2 2 1 1 1 2 2 2 2 2 2 2 2 2	0 0 0 0 0 2 2 2 2 1 0 0 0 1 1 1 1 1 1	19 17 20 28 9 3 18 0 0 0 0 0 0 1 18 18 18 18 9 2 2 2 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	32 7 9 17 12 5 5 7 0 1 1 0 1 1 1 3 4 7 7 7 7 2 5 5 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2 3 7 5 2 2 3 0 1 1 0 1 1 1 2 2 3 3 3 3 1 1 2 2 2 0 0 0 2 2 2 2 2 2 2 2 8 8	2 4 4 1 1 1 2 2 0 0 0 0 0 1 1 2 2 2 2 0 0 1 1 1 1	2 4 4 1 1 0 2 2 0 0 0 0 0 1 1 2 2 2 0 0 0 1 1 1 1	0 4 4 1 1 2 2 0 0 0 0 0 0 1 1 2 2 2 2 0 0 1 1 1 1	12 3 0 9 5 1 0 0 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 04:00 04:00 06:45 06:45 06:45 06:30 18:00 18:00 05:30 05:30 05:30 07:00 05:30	20:30 23:30 01:00 22:35 22:45 21:45 23:45 17:45 17:45 17:45 17:45 17:45 22:45 23:45 23:45 23:45 23:45 23:45 22:33 22:33 22:33 22:33 22:33 22:33 22:33 22:33 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:33 23:33	92 N/A 88	5 9 9 6 0 3 3 7 (note 9) 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 1 6 2 2 2 2 2 3 3 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 99 28 30 45 15 0 0 0 0 0 0 3 3 3 30 30 30 30 30 15 15 15 15 15 15 15 15 15 15 15 15 15	7 1 8 8 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	04:30 04:30 08:30 07:45	23:30 01:00 23:45 22:45 	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 0 0 8 5 5 1 1 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1	0 3 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 0 0 0 0 0	0 0 0 0 244 0 364 242 0 0 122 0 0 0 0 0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 1 1 1 1 1 1 1 0 0 15	05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:4 23:4 23:4 21:4 21:4 21:4 21:4 21:4 21:4 21:4 21
Ministribud Ministribud Ministribud Ministribud Conkstribud Conkst	London Bridge London Victoria London London London London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 1 0 (note 9) 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 10 10 8	0 0 0 0 0 0 2 2 2 2 1 1 0 0 1 1 1 1 1 1	19 17 20 28 9 3 18 0 0 0 0 0 0 0 1 18 18 0 0 0 0 0 1 18 18 0 0 0 0	32 7 9 17 12 5 5 7 0 1 1 0 1 1 1 3 4 7 7 7 2 5 5 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2 2 3 7 5 2 2 3 3 0 1 1 2 2 3 3 3 3 1 2 2 2 2 2 2 8 8 8 8 3 3 3 3	2 4 4 1 1 1 2 0 0 0 0 0 0 1 1 1 2 2 2 0 0 0 1 1 1 1	2 4 4 1 1 0 2 2 0 0 0 0 0 1 1 2 2 2 0 0 0 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 1 1 1 2 2 2 2 1 1 0 0 0 1 1 1 1	12 3 0 9 5 1 0 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 2 1 1 1 1	07:15 05:00 04:00 04:00 04:00 04:00 04:00 06:45 06:45 06:45 06:45 06:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 05:30 07:00	20:30 23:30 01:00 23:30 01:00 22:45 21:45 21:45 21:45 21:15 22:45 23	92 N/A 88	5 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 6 2 2 2 3 3 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 29 28 30 45 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	7 1 8 8 1 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 3 0 0 3 3 1 1 0 0 1 1 1 1 1 1 1 1 0 0 1 1 1 1	04:30 04:30 08:30 07:45 	23.45 22.45 22.45 23.45 23.45 23.45 23.45 22.45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 0 0 8 5 5 1 1 0 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1	0 3 2 1 0 0 0 0 0 0 0 0 0 0 1 1 1 0 0 0 0 0	0 0 0 12 60 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:2 23:4 23:4 21:4 21:4 21
Whatefield  Whatefield  Munichon  Leves  Southease  Southease  Weenhaven Town  Weenhaven Town  Weenhaven Harbour  Bilarboptione  Bearbord  Bearbor	London Bridge London Victoria London London London London London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 2 2 2 2 2 2 2 2 2 2 2 1 10 10 8 0 0	0 0 0 0 0 0 2 2 2 2 0 0 1 1 1 1 1 1 1 1	19 17 20 28 9 3 18 0 0 0 0 0 1 18 18 0 18 9 2 2 9 9 9 9 9 18 18 9 9 9 9 9 9 9 9 9 9 9 9	32 9 9 177 122 5 5 7 7 0 1 1 1 1 3 3 4 7 7 7 7 2 2 5 4 0 0 0 4 4 4 4 4 4 3 2 23 21 16 8 8 8 8 8 8 8 8 8 8	2 2 3 7 5 2 2 3 3 0 1 1 1 1 2 2 3 3 3 3 1 1 2 2 2 2 2 2 2 2	2 4 4 1 1 2 0 0 0 0 0 0 1 1 1 2 2 2 2 0 0 2 1 1 1 1	2 4 4 4 1 1 0 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 1 1 0 0 0 1 1 1 1	0 4 4 1 1 1 2 0 0 0 0 0 0 0 0 0 0 1 1 1 2 2 2 2	12 3 0 9 5 1 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 2 0 0 0 0	07:15 05:00 04:00 04:00 04:00 04:00 06:45 06:45 06:45 06:30 18:00 18:00 05:30 05:30 05:30 07:00 05:30	20:30 23:30 01:00 22:35 22:45 21:45 23:45 17:45 17:45 17:45 17:45 17:45 22:45 23:45 23:45 23:45 23:45 23:45 22:33 22:33 22:33 22:33 22:33 22:33 22:33 22:33 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:33 23:33	92 N/A 88	5 9 9 6 0 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 6 2 2 2 3 3 1 1 0 2 2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90   90   28   30   30   45   50   64   51   50   64   64   64   64   64   64   64   6	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 1 2 2 3 3 0 0 0 1 1 1 1 1 1 1 1 1 0 0 8 4	04:30 04:30 08:30 07:45	23:30 01:00 23:45 22:45 	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 0 0 8 5 5 1 1 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1	0 3 2 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 0 0 12 24 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3 22:4 23:4 23:4 23:4 21:4 21:4 21:4 21:4 23:3 23:3 23:3 23:3 23:3 23:3 23:3
Wiselstind Plantind Plantind Plantind Plantind Plantind Cookstrice  Everse Southease S	London Bridge London Victoria London Victoria London Victoria London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 2 2 2 2 2 2 2 2 2 2 2 1 10 10 8 0 0	0 0 0 0 0 0 2 2 2 2 1 1 0 0 1 1 1 1 1 1	19 17 20 28 9 3 18 0 0 0 0 0 0 0 1 18 18 0 0 0 0 0 1 18 18 0 0 0 0	32 9 17 12 5 5 7 0 1 1 1 3 4 4 7 7 7 7 2 5 5 4 0 0 4 4 4 4 4 4 4 4 4 4 4 1 8 8 8 8 8 8 8 8	2 3 7 5 2 2 3 0 1 1 0 1 1 1 2 2 3 3 3 3 1 2 2 0 0 0 2 2 2 2 2 2 8 8 8 8 3 3 3 3 3 0 0	2 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 0 0 1 1 1 1	2 4 4 4 1 1 0 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 0 0 0 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 0 0 1 1 1 1	12 3 0 9 5 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1 2 2 2 2	07-15 05:50 04:00 04:00 05:50 04:00 06:50 06:50 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:50	22:30 01:00 22:30 01:00 22:45 21:45 22:45 21:45 21:15 21:15 21:15 22:45 23:45	65 92 N/A 88	5 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 1 0 2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 29 28 30 45 15 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 1 2 2 3 3 0 0 0 1 1 1 1 1 1 1 1 1 0 0 8 4	04:30 04:30 08:30 08:30 07:45 	23:45 22:45 22:45 23:45 23:45 23:45 23:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 22:45 23:45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 5 0 8 8 5 5 1 0 0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 1	0 3 2 1 1 1 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	08:30 04:00 04:00 05:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45 08:45	23:3:22/2 23:4:25/2 23:4:25/2 21:4:25/2 21:4:21:4:21:4:21:4:21:4:21:4:21:4:21:
Wisefafiald Pilentation Cocklatride Leves Southease Sout	London Bridge London Victoria London Victoria London London London London London Victoria London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 2 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 0 (note 9) 1 2 2 2 2 2 2 2 2 2 2 2 1 10 10 8 0 0	0 0 0 0 0 0 2 2 2 2 1 0 0 0 1 1 1 1 1 1	199 177 200 28 9 9 3 18 18 18 18 9 9 9 9 9 9 9 9 9 9 9 9 9 9	32 9 9 177 122 5 5 7 7 0 1 1 1 1 3 3 4 7 7 7 7 2 2 5 4 0 0 0 4 4 4 4 4 4 3 2 23 21 16 8 8 8 8 8 8 8 8 8 8	2 3 7 5 2 2 3 0 0 1 1 0 1 1 1 2 2 3 3 3 3 1 2 2 0 0 2 2 2 2 2 2 8 8 8 3 3 3 3 3 3 0 3 3 3 3 0 3 3 3	2 4 4 4 1 1 1 2 2 0 0 0 0 0 0 0 1 1 1 1 1 1 1 6 6 4 3 2 2 2 2 2 0 0 2 2 2	2 4 4 4 1 1 0 2 2 0 0 0 0 0 1 1 2 2 2 2 0 0 2 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	12 3 0 9 5 5 1 1 0 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 1 1 1 1	07-15 05:00 04:00 04:00 04:00 05:00 04:00 06:00	20:30 22:30 01:00  22:30 01:00  22:46 22:46 22:46 23:46 17:45 17:45 17:45 12:46 22:46 23:46 23:46 23:46 23:46 22:45 23:46 22:46 23:30 23:30 23:30 23:30 23:30 23:36 23:46 23:46 23:46 23:46 23:46 23:46 23:46	65 92 N/A 88	5 9 9 6 6 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 1 0 2 2 2 0 0 0 0 0 0 0 0 1 1 1 1 1 1	90 29 28 30 45 5 15 0 0 0 0 30 30 15 15 15 15 15 15 15 15 15 15 15 15 15	7 1 8 8 1 0 0 0 0 0 0 0 0 0 1 1 2 2 3 3 0 0 0 1 1 1 1 1 1 1 1 1 0 0 8 4	04:30 04:00 08:30 07:45	23:45 22:45 22:45 23:45 23:45 23:45 23:45 23:45 23:45 22:45 23:45 22:45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 5 0 8 8 5 5 1 0 0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 1	0 3 2 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 0 0 12 24 0 0 0 0 0 0 0 0 0 0 0 0 12 12 12 12 12 12 12 13 13 13 12 13 13 13 13 13 15 16 16 16 16 16 16 16 16 16 16 16 16 16	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	06:30 04:00 05:45 06:45	23:3:22/2 23:4:22/2 23:4:22/2 23:4:22/2 21:4:22/2 21:4:21/2 21:4:21/2 21:4:21/2 23:3:23:3:23:3:23:3:23:3:23:3:23:3:23
Minishified Minishified Minishified Conditioning Leves Southease S	London Bridge London Victoria London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 0 (note 9) 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 10 10 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 2 2 2 2 1 0 0 0 1 1 1 1 1 1	199	32 7 9 177 12 5 5 5 7 7 0 0 1 1 1 1 3 3 4 7 7 7 7 7 7 2 2 5 5 4 4 4 4 4 4 4 3 3 3 3 21 16 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	2 3 7 7 5 2 2 3 3 0 1 1 1 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	2 4 4 4 1 1 1 1 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	2 4 4 4 1 1 0 2 2 0 0 0 0 0 1 1 1 2 2 2 2 0 0 2 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	12 3 0 9 5 5 1 1 0 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 1 1 1 1	07-15 (05:50) 04:00 04:00 04:00 04:00 04:00 06:45 05:30 05:3	20:30 22:30 01:00	65 92 N/A 88	5 9 9 6 6 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9	0 1 1 6 2 2 2 3 3 1 1 0 2 2 2 2 1 1 0 0 2 2 2 2 2 2 2 2 2	90 29 28 28 30 445 50 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 1 0 2 2 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1	04:30 04:00 08:30 07:45 	23.45 22.45 22.45 2.245 2.	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 5 0 8 8 5 5 1 0 0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 1	0 3 2 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 0 0 122 24 24 20 0 0 0 122 12 12 12 12 12 12 12 12 12 12 12 12	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 05:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45	23:3-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2
WiveIsfield Phirmpton Cookebridge Lewes Southease Southease Newharen Town Newharen Harbour	London Bridge London Victoria London London London London London Victoria London	92 90 88	92 93 104	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 0 (note 9) 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 10 10 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 2 2 2 2 1 1 0 0 1 1 1 1 1 1 1	199 177 200 288 9 3 188 188 18 18 18 18 18 18 18 18 18 18 1	32 7 9 17 12 5 5 7 7 0 1 1 0 1 1 1 1 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 2 3 7 5 2 2 2 3 3 0 1 1 1 1 2 2 3 3 3 3 1 2 2 2 2 2 2 2 2 2	2 4 4 4 1 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2	2 4 4 4 1 1 0 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	0 4 4 1 1 1 2 0 0 0 0 0 0 0 0 0 0 1 1 1 2 2 2 2	12 3 0 9 5 5 1 1 0 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 1 1 1 1	07:15 05:50 04:50 04:50 04:50 05:50 04:50 05:50 04:50 05:50 04:50 05:50	2030 01:00 0	65 92 N/A 88	5 9 9 6 6 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 29 28 28 30 445 50 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 8 1 1 0 2 2 0 0 0 0 0 0 0 0 1 1 2 2 3 3 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	04:30 04:00 08:30 07:45 	23:45 22:45 22:45 23:45	65 92 N/A 88	67 92 N/A 104	0 0 0 3 3 13 5 5 0 8 8 5 5 1 0 0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 1	0 3 2 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 0 0 12 24 25 25 25 25 25 25 25 25 25 25 25 25 25	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 06:45	23:3:22-4 23:4 23:4 23:4 23:4 23:4 21:4 21:4:4 21:4:4 21:4:4 22:3:3 23:4 23:3:2 23:4 23:4 23:4:2 23:4 23:4
Wiselsfield Plampton Cocksbridge Lewes Southease Southease Newhateen Town Newhateen Newhateen Search Ghride Berwick Polessian Polessian Polessian Polessian Polessian Polessian Newhateen	London Bridge London Wictoria London Victoria London London London London London London London London London Victoria London	92 90 88 123	92 93 104 139	15 7 0 8 9 (note 9) 1 (note 9) 1 (note 9) 0 (note 9) 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 10 10 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 2 2 2 2 1 0 0 0 1 1 1 1 1 1	199 199 288 188 18 18 18 18 18 18 18 18 18 18 18	32 7 9 177 12 5 5 5 7 7 0 0 1 1 1 1 3 3 4 7 7 7 7 7 7 2 2 5 5 4 4 4 4 4 4 4 3 3 3 3 21 16 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	2 2 3 7 7 7 7 5 2 2 2 3 3 0 1 1 1 2 2 2 3 3 3 3 1 2 2 2 2 2 2 8 8 8 3 3 3 3 3 3 3 3 3 3 3	2 4 4 4 1 1 1 2 0 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	2 4 4 4 1 1 0 2 2 0 0 0 0 0 1 1 1 2 2 2 2 0 0 2 1 1 1 1	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	12 3 0 9 5 5 1 1 0 0 0 0 0 0 0 0 0 0 0 1 1 2 2 2 0 0 1 1 1 1	07-15 (05:50) 04:00 04:00 04:00 04:00 04:00 06:45 05:30 05:3	20:30 22:30 01:00	92 N/A 88 123 80 N/A	5 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 28 8 65 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	7 1 8 8 1 1 0 2 2 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1	04:30 04:00 08:30 07:45 	23.45 22.45 22.45 2.245 2.	65 92 N/A 88 123 80 N/A	67 92 N/A 104 139	0 0 0 3 3 13 5 5 0 8 8 5 5 1 0 0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 1	0 3 2 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 0 0 12 24 24 26 21 22 22 22 2	5 0 10 5 2 0 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 05:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45 06:45	23:3:22-4 23:4 23:4 23:4 23:4 23:4 21:4 21:4:4 21:4:4 21:4:4 22:3:3 23:4 23:3:2 23:4 23:4 23:4:2 23:4 23:4
Wivelsfield Primotion Cewes Southease Newhaven Town Newhaven From Newhaven From Newhaven From Seaford Seaford Southease Newhaven From Seaford Southease Newhaven From Newhaven From Newhaven From Newhaven Newhave	London Bridge London Victoria London Victoria London London London London London London London London Victoria London Lond	92 90 88 123	92 93 104 139 82	15 7 0 8 8 9) (note 9) 7 0 0 8 9 9) 7 0 1 1 (note 9) 2 (note 9) 0 (note 9) 0 (note 9) 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	199 177 200 187 188 18 18 18 18 18 18 18 18 18 18 18 18	32 7 9 17 12 5 5 5 7 0 1 1 0 0 1 1 1 3 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 3 7 7 5 2 2 3 3 0 1 1 1 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3	2 4 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	2 4 4 4 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	12 3 0 9 9 1 1 0 2 0 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1	07-15 05:50 04:50 05:50 04:50 05:50	2030 01:00 0	65 92 N/A 88 123	5 9 9 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 1 0 2 2 2 2 2 2 2 2 2 2 2 2 8 (note 13)	90 29 28 30 46 5 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	04:30 04:00 08:30 07:45 	23:45 22:45 22:45 23:45	65 92 NA 88 123 80 NA 102	67 92 N/A 104 139	0 0 3 3 5 8 8 5 1 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1	0 3 3 2 1 1 0 0 1 1 1 0 0 0 0 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0 0 100 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	05:45 06:45	01:0 23:3 22:4 23:4 21:4:4 21:4:4 21:4:4 22:4:4 23:3 23:3 23:3 23:4 23:4:4 21:4 21
WiveIsfield Plumption Plumption Southease Southease Newhaven Town Newhaven Harbour Behopstone Seaford Southease Newhaven Harbour Behopstone Seaford Southease Newhaven Harbour Behopstone Seaford Southease Polecate Hampden Park Eastbourne  Penensor & Westham Penensor & Westham Penensor & Westham Penensor Bon Normane Bay Cooden Beach Collination St. Leonards Warrior Square Hassings Burpess Hil Hassocks Preston Park How Portslade Snoreham-by-Sea Snoreham-by-Sea East Worthing West Worthing West Worthing West Worthing West Worthing Morthing Morthing Anometing Littlehampton	London Bridge London Wictoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London	92 90 88 123 80 99 102 81	92 93 104 139 82 102 109 88	15	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	199 177 200 187 188 18 18 18 18 18 18 18 18 18 18 18 18	32 7 9 7 12 2 5 5 7 7 0 1 1 1 3 3 4 4 7 7 7 7 2 5 5 4 4 0 0 0 4 4 4 4 4 4 4 4 3 3 23 1 16 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	2 3 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	2 4 4 4 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	2 4 4 4 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 4 4 1 1 1 1 2 2 2 0 0 0 0 1 1 1 1 1 1 1 1 1	12 3 0 9 5 5 1 0 0 0 0 0 0 0 0 0 0 1 2 2 2 2 2 2 2 2	07:15 05:50 04:50 04:50 04:50 05:50 04:50 05:50 04:50 05:50 04:50 05:50	2030 01:00 0	65 92 N/A 88 123 80 N/A 102 81	5 9 9 6 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 1 0 2 2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0	90 90 90 90 90 90 90 90 90 90 90 90 90 9	7 1 8 8 1 1 0 2 2 0 0 0 0 0 1 1 2 2 2 3 3 3 1 1 1 1 1 1 1 1 1 1 1 1	04:30 04:00 06:30 07:45 	23:45 22:45	65 92 N/A 88 123	67 92 N/A 104 139	0 0 3 3 5 8 8 5 1 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1	0 3 3 2 1 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0 0 100 15 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1	05:45 05:45 06:45	01:0 23:3 22:4 23:4 21:4:4 21:4:4 21:4:4 22:4:4 23:3 23:3 23:3 23:4 23:4:4 21:4 21
Wivelsfield Phumpton Cockebridge Lewes Southease Southease Southease Newharen Town Newharen Harbour Newharen Harbour Newharen Harbour Seaford Glynde Berwick Polesais Hamber Seaford S	London Bridge London Victoria London Victoria London London London London London London London London Victoria London Lond	92 90 88 123	92 93 104 139 82	15 7 0 9 (note 9) 1 (note 9) 0 (note 9) 1 (note 9) 0 (note 9) 1 (note 9) 0 (note 9) 1 (n	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	199 17 200 17 200 19 19 19 19 19 19 19 19 19 19 19 19 19	32 7 9 17 12 5 5 5 7 0 1 1 0 0 1 1 1 3 3 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	2 3 7 7 5 2 2 3 3 0 1 1 1 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3 3	2 4 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	2 4 4 4 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 4 4 1 1 1 2 2 0 0 0 0 0 0 1 1 1 2 2 2 2 2 2	12 3 0 9 9 1 1 0 2 0 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1	07-15 05:50 04:50 05:50 04:50 05:50	2030 01:00 0	65 92 N/A 88 123 80 N/A 102 81 62 62	5 9 9 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 1 1 1 6 2 2 2 3 3 1 1 0 2 2 2 2 2 2 2 2 2 2 2 2 8 (note 13)	90 29 28 30 46 5 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 1 8 8 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	04:30 04:00 06:30 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:45 22:45	65 92 NA 88 123 80 NA 102	67 92 N/A 104 139	0 0 3 3 5 8 8 5 1 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1 1 1 1	0 3 3 2 1 1 0 0 1 1 1 0 0 0 0 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0 0 100 15 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1	05:45 05:45 06:45	01:0 23:3 22:4 23:4 21:4:4 21:4:4 21:4:4 22:4:4 23:3 23:3 23:3 23:4 23:4:4 21:4 21

<sup>55 55 0 2 30 0</sup>Note 15: States pervises possible to thereon Robelli all Q Robelli and Torbidose, or not included in Table 1. The auarthum of services between fines stations is specified in Table 3.

Note 2: All services on a Suickey to these stations are required to coverage from Osed orbit. The specification shown is based on stemping and Coverage from Osed orbit. The specification shown is based on stemping and Coverage from Osed orbit. The specification shown is based on stemping and Coverage from Osed orbits of the Services departing to these stations are stemping and particular state of the Services of the Services departing London between 16:00 and 16:59 are permitted to specification shown is all coverage from Osed orbits of the Services of the Ser

	TABLE 2: NORTHBOUND	4		rney Time	Longo ger	1 0700 005	0 0000 0050	Monday	to Friday	2000 2050	2400 245	ol 2200 0450l	C		la-sa-sa-	Time	0000 0000	Saturd: 0700-2159	3y	2200 0450	Carrier and				2222 2222	Satur	rday	12400 0450	Service arriv	
	Arrival time in Lor		Jour	rney i ime	0200-0658	0/00-095	9 0800-0859	1000-1959	1000-1959	2000-2059	2100-215	9 2200-0159	First (no	Last (no earlier	Journey	rime	0200-0659	0/00-2159	0/00-2159	2200-0159		Last (no earlier	Journe	y i me	0200-0859	J900-2059	0900-2059	2100-0159		Last (no earlier
	Direct services from each station to specified destinated Brighton	London	Average		Total 4	Total 25	Total 7	TPH 8 (note 7)	Total 75	Total 6	Total 5	Total 13	later than) 06:00	than) 01:00		Maximum	Total 4	TPH 8 (note 8)	Total 115	Total 11	later than) 06:00	than) 00:45	Average	Maximum	Total 8	TPH 5	Total 60	Total 15	later than) 06:00	
		London Blackfriars Luton	79	89	2	9	2	2	20 20	2	2	6			79	89	0	2 2	29 29	0			79	89	5	2	24 23	9		
		Bedford Midland London Bridge	61	74	0	8 4 12	1	2	20 20	0	0	1			61	74	0	2 2	29 28 58	0 1 10			61	74	0	0	23 0	0		
	Littlehampton	London Victoria of which	61 56	93 56	0	0 7	0	2	35 20	1	0	2	07:45	22:45	61 56	93 56	0	2	30	0	07:30	22:45	61 N/A	93 N/A	0	0	36 0	0	09:00	21:45
	Littlenampton	London London Bridge London Victoria	103 100	115 105	0	3 4	1 2	0 2	20 0 20	0	0	0	07.45	22.45	N/A 100	N/A 105	0	0	30 0 30	0	07.30	22.45	N/A 100	N/A 105	0	0	12 0 12	0	09.00	21.45
	Angmering Goring-by-Sea	London London	100	100	0	7	3	2	20 20 20	2 2	2	2	07:00 07:45	22:45 22:45	100	103	0	2 2	30 30	2 2	07:30 07:30	22:45 22:45	100	103	1	1	13	1	09:00 09:00	21:45 21:45
	Ourrington-on-Sea West Worthing	London London			0	6	3	2 2	20 20	2 2	2 2	2 2	07:45 07:45	22:45 22:45			0	2	30 30	2 2	07:30 07:30	22:45 22:45			1 1	1	13	1	09:00	21:45 21:45
	Worthing East Worthing	London London	80	84	0	7	3	2	20	2	2	2	07:00	22:45	80	84	1 0	2	30	2	07:00	22:45	80	84	1 0	1 0	13	1 0	09:00	21:45
	Lancing Shoreham-by-Sea	London London			0	7	3	1 2	10 20	2 2	2 2	2 2	07:00 07:00	22:45 22:45			0	1 2	16 30	2 2	07:30 07:00	22:45 22:45			1 1	0	1 13	0	09:00	21:45
	Portslade Hove	London London			0	7	3	1 2	10 20	2 2	2 2	2 2	07:00 07:00	22:45 22:45			0	1 2	16 30	2 2	07:30 07:00	22:45 22:45			1	0	1 13	0	09:00	21:45
	Preston Park Hassocks	London London			2 2	17 18	6	3	30 30	3	2 2	5 7	06:30 06:30	01:00 01:00			1	3	47 47	8	06:30 06:30	00:45 00:45			7	1 4	12 51	4 20	07:30 07:00	00:30 01:00
	Burgess Hill Ore	London Londond Victoria			0	21 0	7	1	40 10	1	1	9 2	06:30 11:00	01:00 23:45			0	1	60 15	10	06:30 09:00	00:45 23:45			0	5 1	64 12	21	07:00 10:30	01:00 22:15
	Hastings St. Leonards Warrior Square	Londond Victoria Londond Victoria	125	142	0	5 5	1 1	1	10 10	1	1	2 2	07:15 07:15	23:45 23:45	125	142	0	1 1	15 15	2 2	09:00 09:00	23:45 23:45	125	142	0	1	12 12	2	10:30 10:30	22:15 22:15
	Bexhill Collington	Londond Victoria Londond Victoria			0	5	1 1	1 1	10 10	1 1	1 1	2 2	07:15 07:15	23:45 23:45			0	1 1	15 15	2	09:00 09:00	23:45 23:45			0	1	12 12	2 2	10:30 10:30	22:15 22:15
	Cooden Beach Normans Bay	Londond Victoria Londond Victoria Londond Victoria			0	5	0	0	10	0	0	0	07:15 10:00 10:00	23:45 17:45 17:45			0	0	15 0 0	2 0 0	09:00	23:45			0	0	12 0 0	0	10:30	22:15
	Pevensey Bay Pevensey & Westham Eastbourne	Londond Victoria Londond Victoria London			0 1 (note 2)	5	1 2	1 2	10 20	1 2	1 2	2	07:15 06:45	23:45 23:45			0	1 2	15 30	2 2	09:00	23:45 23:45			0	1	12 12	2	10:30 08:30	22:15 22:15
	Lastbourne	London Bridge London Victoria	93 89	94 103	1	3 4	2	0 2	0 20	0 2	0 2	0 2	00.40	20.40	N/A 89	N/A 103	0	0	0	0 2	00.00	20.40	N/A 89	N/A 103	0	0	0	0 2	00.00	ZZ.10
1	Hampden Park Polegate	London London	-		1 (note 2) 1 (note 2)	7 7	3	2 2	20 20 20	2 2	2 2	2 2	06:45 06:45	23:45 23:45			0	2 2	30 30	2 2	08:00 08:00	23:45 23:45			1	1	12 12	2 2	08:30 08:30	22:15 22:15
L	Berwick Glynde	London London	1		0	6	2 2	0	1 0	0	0	1 0	07:15 08:15	23:45 09:45		_	0	0	1 1	0	_:	_		L I	0	0	0	0	L: 1	_:`
	Seaford Bishopstone	Londond Victoria Londond Victoria	98	100	0	2 2	1	0	0	0	0	0	08:45 08:45	09:45 09:45	98	100	0	0	0	0	-	-	98	100	0	0	0	0	-	-
	Newhaven Harbour Newhaven Town	Londond Victoria Londond Victoria			0	2 2	1 1	0	0	0	0	0	08:45 08:45	09:45 09:45			0	0	0	0	-	:			0	0	0	0	:	. :
	Southease Lewes	Londond Victoria London	65	74	0 1 (note 2)	7	3	2	0 20 7	2	2	2	08:45 06:45	09:45 23:45	65	74	0	2	30	2	08:00	23:45	65	74	0	1	12	2	08:30	22:15
	Cooksbridge Plumpton	London London			1	5 6	3	1	11	1 1	1 1	1	08:00 06:45	20:15 23:45			0	0	0 15	1	08:30	23:45			0	0	0 12	2	08:30	22:15
1	Wivelsfield Haywards Heath	London London			5 (note 2)	18 35	13	3 6	30 61	5	3 5	7 11	06:30 06:00	01:00 01:00			1 4	3 6	45 90	10 12	06:30 06:00	00:45 00:45			9	5	24 63	6 16	07:30 06:00	00:30 01:00
1		London Blackfriars London Bridge London Victoria			2 0 3	9 8 18	2 4 7	2 2 2	2 20 21	2 0 3	0 3	6 1 4					3 0 1	2 2	29 30 31	6 1 5					4 0 5	3 0 2	40 0 23	7 0 9		
	Balcombe Southampton Central	London Victoria  Londond Victoria	150	159	3	7	2	1	11	1	1	4	06:30 08:45	01:00 22:45	150	159	1	1	15	5	06:30 08:45	00:45 22:45	150	159	1	1 0	12	4	08:30	00:45
	St. Denys Woolston	Londond Victoria Londond Victoria			0	0 2	0	0	1 0	0	0	0	10:45 09:45	10:30 09:30			0	0	0	0	1 :	-			0	0	0	0	-	. :
	Netley Swanwick	Londond Victoria Londond Victoria			0	0 2	0	0	0	0	0	0	08:45	22:45			0	0	0 15	0	08:45	22:45			0	0	0	0	-	-
	Fareham Portchester	Londond Victoria Londond Victoria			0	2 2	1	1	10 10	1	1	1 1	08:45 08:45	22:45 22:45			0	1 1	15 15	1 1	08:45 08:45	22:45 22:45			0	0	0	0	-	:
	Cosham Portsmouth Harbour	Londond Victoria Londond Victoria			0	2	1 0	1 1	10	1 0	1 0	1 0 (note 4)	08:45 08:00	22:45 19:15			0	1 1	15 15	1 0 (note 4)	08:45 08:15	22:45 19:15			0	1	0 12	0	10:00	21:45
	Portsmouth & Southsea Fratton	Londond Victoria Londond Victoria	120	141	0	2 2	0	1	10 10	1	0	0 (note 4) 0 (note 4)	08:00 08:00	20:15 20:15	120	141	1	1 1	15 15	0 (note 4) 0 (note 4)	07:00 07:00	20:15 20:15	120	141	0	1	12 12	1 1	10:00 10:00	21:45 21:45
	Hilsea Bedhampton	Londond Victoria Londond Victoria			0	3	1	0	0	1	0	0 (note 4) 0 (note 4)	08:00	20:15 20:15			0	0	1	0 (note 4) 0 (note 4)	<del>  </del>	-			0	0	0	0	-	
	Havant Warblington	Londond Victoria Londond Victoria			0	3	1 1	0	20	1	0	1 (note 4) 0	07:00 08:45	22:45 20:15			0	0	30 1	1 (note 4)	07:00	22:45			0	0	12 0	0	10:00	21:45
	Emsworth Southbourne	Londond Victoria Londond Victoria			0	4 4 3	1	1 1 0	12 10 1	2	1 1 0	1 1 0	08:00 08:00 08:45	22:45 22:45 20:15			0	1	16 16 1	1 1 0	07:45 07:45	22:45 22:45			0	1 1 0	12 12 0	1 1 0	10:00 10:00	21:45 21:45
	Nutbourne Bosham Fishbourne	Londond Victoria Londond Victoria Londond Victoria			0	4	1	0	2	1	0	0	08:00 08:00	20:15 20:15 20:15			0	0	1	0	1 - 1	- :			0	0	0	0	:	
	Chichester  Boanor Reais	Londond Victoria London			0	6	1 2	2	20	2	2	1 (note 4)	07:00	22:45 22:15			1 0	2	30 30	1 (note 4)	07:00 08:15	22:45 22:30			0	1 1	12	1 3	10:00 08:45	21:45 23:45
		London Bridge London Victoria	104 105	104 116	0	1 4	1	0 2	0 20	0	0	0			104 105	104 116	0	0 2	0 30	0			N/A 105	N/A 116	0	0	0 12	0		
	Barnham Ford	London London			0	2	1	2	20 20	2	2 2	2 (note 4) 2 (note 4)	07:00 08:30	22:45 22:45			0	2 2	30 30	2 (note 4) 2 (note 4)	07:15 07:45	22:45 22:45			1 1	1	24 12	3	08:45 08:45	23:45 23:45
	Arundel Amberley	London London			0	3	0	1	20 10	1	1	1 (note 4) 1 (note 4)	07:45 08:00	22:45 22:45			0	1	30 16	1 (note 4) 1 (note 4)	07:45 07:45	22:45 22:45			1	1	12 12 12	3	08:45 08:45	23:45 23:45 23:45
	Pulborough Billingshurst Christ's Hospital	London London London			0	7 7 7	2 2	2 2	20 20 13	2 2 1	1	1 (note 4) 1 (note 4) 1 (note 4)	07:45 07:45 07:45	22:45 22:45 22:45			0	2 2	30 30 15	1 (note 4) 1 (note 4) 1 (note 4)	07:45 07:45 07:45	22:45 22:45 22:45			1	1	12 12 12	3 3	08:45 08:45 08:45	23:45 23:45 23:45
	Horsham	London London Bridge	72	76	2	14 (note 1: 3 (note 11	1 4 (note 11) 1 1 (note 11)	4 2	40 18	4 2	4 2	7 (note 4) 2	06:30	00:15			1	4	60	5 (note 4)	06:45	00:15			3 (note 6)	2	24	7 2	07:45	00:15
	Littlehaven	London Victoria London	56	78	2	11 11 (note 1:	3 1 4 (note 11)	2	22 20	2 2	2 2	5 4 (note 4)	06:30	00:15			0	2 2	30 30	1 4 (note 4)	06:45	00:15			2 2 (note 6)	1	12	5 4	07:45	00:15
	Faygate Ifield	London London			0 2		) 0 (note 11)	0 2	7 20	0 2	0 2	0 4 (note 4)	08:00 06:30	19:30 00:15			0	0 2	0	0 4 (note 4)	06:45	00:15			0 2 (note 6)	0	0	0 4	07:45	00:15
	Crawley Three Bridges	London London			2 14 (note 2	14 (note 1: ) 34 (note 1:	1 4 (note 11) 1 11 (note 11	4 8	40 81	4 8	4 9	5 (note 4) 17	06:30 02:15	00:15 01:00			1 15	4 8	60 120	5 (note 4) 16	06:45 02:15	00:15 01:00			3 (note 6) 17 (note 6)	2 7 (note 9)	24 79	7 20	07:45 02:00	00:15 01:00
1		London Blackfriars London Bridge	52 41	57 58	7	8 9 (note 11	2 ) 4 (note 11)	2 4	20 38	4 2	4 2	6 3			52 41	57 58	8	2 4	30 60	6 5			52 41	57 58	6	4	43 12	9 2		
	Gatwick Airport (including Gatwick Express)	London Victoria London	42	60	7 22	17 44 (note 11	5 1) 12 (note 11)	2 18	23 180	2 15	3 15	8 33	02:15	01:45	42	60	6 21	2 18	30 270	5 32	02:15	01:45	42	60	10 32	2 14	24 168	9 38	02:00	01:30
1		London Blackfriars Luton	47	59	8	9	2 2	2	39 39	3	4	6			47	59	7 7	4	59 59	6			47	59	6	4	46 46	9		
		Bedford Midland London Bridge of which	41 29	59 32	8 0 0	8 8 (note 11 5	2 3 (note 11)	2 4 2	39 28 20	3 2 0	4 2 0	6 3 1			41 29	59 32	7 1 0	4	59 58 30	6 5 0			41 N/A	59 N/A	6 1 0	1	46 12 0	9 2 0		
1	Gatwick Airport (Express services only)	London Victoria London Victoria	34 N/A	57 31 (note 3)	16	27 13	7 4	10 4	20 97 40	10	9	1 24 14	02:15	01:15	34	57 31 (note 3)	13	10	149 60	21 14	02:15	01:15	34	57 31 (note 3)	25 14	9	109 48	27 18	02:15	01:15
1	Horley Salfords	London London	1	2.,1000 0)	6	15 10 (note 1	6	3 2	30 20	2	2	6	02:45 06:45	01:45 00:15		. ,	6	3 2	45 30	6 4	02:45 06:45	01:45 00:15		,	9	2	24 12	8	02:00 08:15	00:45 23:00
	Earlswood Tonbridge (note 1)	London London			1 0	10 (note 1:	1 5 (note 11) 2	1	20 12	1 0	1 0	3	06:45 07:30	00:15 19:45			1 0	2	30 15	4 0	06:45 07:30	00:15 19:45			1 0	1	12 12	3 2	08:15 09:45	23:00 22:45
1		London Bridge London Victoria	69 65	77 74	0	3	1	0	1 11	0	0	0			69 65	77 74	0	0	0 15	0			69 N/A	77 N/A	0	1 0	12 0	2 0		
	Leigh Penshurst	London London			0	6	2 2	1 1	12 12	0	0	0	07:30 07:30	19:45 19:45			0	1 1	15 15	0	07:30 07:30	19:45 19:45			0	1	12 12	2 2	09:45 09:45	22:45 22:45
1	Edenbridge Godstone	London London			0	6	2 2	1 1	12 12	0	0	0	07:30 07:30	19:45 19:45			0	1 1	15 15	0	07:30 07:30	19:45 19:45			0	1	12 12	2 2	09:45 09:45	22:45 22:45
	Nutfield Reigate (note 1)	London London London Bridge	20	48	0	6 3 3	2 1 (note 11) 1 (note 11)	1 0	12 11 0	0	0	0	07:30 08:30	19:45 18:45	N/A	N/A	0	1	15 15 0	0	07:30 07:30	19:45 18:45	N/A	N/A	0	0 0	12 0 0	0	09:45	22:45
	Redhill	London Victoria	38 46	48 46	0	0	1 (note 11) 0 1 6 (note 11)	1	0 11 60	0	0	0	05:00	01:00	N/A 46	N/A 46	0	1	0 15 90	0	05:00	00:45	N/A N/A	N/A N/A	0	0	0	0	06:00	01:00
1	recuiii	London London Bridge London Blackfriars			12 6 1	17 (note 11 6 (note 11 3	1 6 (note 11) 1 (note 11)	6 2 2	18 16	2 0	2 0	9 2 0	U0:UU	01:00			5 1 1	2 2	30 30	9 4 1	US:0U	UU.45			6 1 0	2 0	48 24 0	14 4 1	06:00	01:00
	Merstham	London Victoria London			5	8	4 5	2	26	2 2	2 2	7	06:30	00:15			3 2	2	30	1 4 4	06:45	00:15			5	2	24	9	07:45	00:30
	Coulsdon South Uckfield	London London Bridge (note 10)	75	81	2	11 12 5	5 2	3 4 1	37 40 10	2	2	4 4 2 (note 5)	06:30 06:30	00:15 23:45	75	81	0	4	60 60 15	4 4 1 (note 5)	06:45 08:00	00:15 22:45	75	81	0	3	24 36 10	9 3 (note 5)	07:45 11:30	00:30
	Buxted Crowborough	London Bridge (note 10) London Bridge (note 10)			2 2	5	2 2	1 1	10 10	2 2	2 2	2 (note 5) 2 (note 5)	06:30 06:30	23:45 23:45			0	1 1	15 15	1 (note 5) 1 (note 5)	08:00 08:00	22:45 22:45			0	1	10 10	3 (note 5) 3 (note 5)	11:30 11:30	23:15 23:15 23:15
	Eridge Ashurst	London Bridge (note 10) London Bridge (note 10)			0	5 4	2	1 1	10 10	2 2	2	2 (note 5)	06:30 07:45	23:45 22:45			0	1 1	15 15	1 (note 5)	08:00 08:00	22:45 22:45			0	1	10 10	3 (note 5) 3	11:30 11:30	23:15 23:15
	Cowden	London Bridge (note 10) London Bridge (note 10)			0	4	1	1	10 10	2 2	1 1	1 1	06:30 07:45	22:45 22:45			0	1 1	15 15	1	08:00 08:00	22:45 22:45			0	1 1	10 10	3	11:30 11:30	23:15 23:15
	Hever	London Bridge (note 10)		1	2	5	2	1 2	10 21	2	2	2 (note 5) 5	06:30 06:45	23:45 23:30			0	1 2	15 30	1 (note 5)	08:00 07:45	22:45 23:30			0	1	10 12	3 (note 5)	11:30 09:15	23:15 00:00
	Hever Edenbridge Town East Grinstead	London			2	10					-					ALC:							NAME OF TAXABLE PARTY.	MICO				4		
	Edenbridge Town East Grinstead	London London Bridge London Victoria	50 56	53 59	1 1 2	5 5	2 2	0 2 2	1 20 20	2	0 2	0 5	06-45	23-30	N/A 56	N/A 59	0	0 2	0	0 4		23-20	N/A 56	N/A 59	0	0 1	0 12	0 4	00-15	00:00
	Edenbridge Town  East Grinstead  Dormans Lingfield	London London Bridge London Victoria London London	50 56	53 59	1 2 2	5 5 10 10	2 2 4 4	2 2 2	20 20 20	2 2 2	2 2 2	0 5 4 4	06:45 06:45 06:45	23:30 23:30 23:45	N/A 56	N/A 59	0		0 30 30 30	0 4 4 4	07:45 07:45	23:30 23:30 23:30		N/A 59	0		0 12 12 12	0 4 4 4	09:15 09:15	00:00 00:00 00:00
	Edenbridge Town East Grinstead  Dormans	London London Bridge London Victoria London	50 56	53 59	1	5 5 10 10 15 15	2 2 4 4 6 6	2 2 2 3 3	20 20 20 30 30	2 2 2 4 4	2 2 2 4 4	0 5 4 4 5 (note 4) 6 (note 5)	06:45 06:45 06:30	23:30 23:45 23:45	N/A 56	N/A 59	0 0 0		0 30 30 30 45 45	0 4 4 4 5 (note 5) 5 (note 5)	07:45	23:30 23:30 23:30 23:30 23:30		N/A 59	0 0 0		0 12 12 12 12 12	0 4 4 4 4 (note 5) 4 (note 5)	09:15 09:15	00:00 00:00 00:00
	Edenbridge Town East Grinstead  Dommans Lingfield Hurst Green Oxted	London Bridge London Victoria London London London London London London	50 56	53 59	1 2 2 3 4	5 5 10 10	2 2 4 4 6	2 2 2 3	20 20 20 30	2 2 2 4	2 2 2 4	0 5 4 4 5 (note 4)	06:45 06:45	23:30 23:45	N/A 56	N/A 59	0 0 0		0 30 30 30 30 45	0 4 4 4 5 (note 5)	07:45 07:45 07:45 07:45	23:30 23:30 23:30		N/A 59	0		0 12 12 12 12	0 4 4 4 4 (note 5)	09:15 09:15 09:15	00:00

Note 1: Stutile services operating between Reignes and Redvill, and Torbodge and Redvill, are not included in Table 2. The quantum of services between these stations is specified in Table 4.

Note 3: The torbodge and Redvill and Torbodge and Redvill, are not included in Table 2. The quantum of services between these stations is specified in Table 4.

Note 3: The torbodge and the torbodge and Redvill and Torbodge and Redvill, are not included in Table 2. The quantum of services between these stations is specified in Table 4.

Note 5: An additional service shall operate from Chichester for Three Bridges, at a maximum internal of 70 minutes, arriving between 23:00 and 25:90. One service shall extend to Portsmouth Harbour. One service need not call at Amberiev, Littlehaven or field.

Note 5: An additional service shall operate from Chichester for Three Bridges, and the torbodge of these temperature and the torbodge of the

	TABLE 3: REDHILL OUTBOUND			Monday	to Friday			Satu	rday			Sur	nday	
	Departure time from Redhill		0200-0859	1800-0159	Service le	aves RDH	0200-0659	1900-0159	Service le	aves RDH	0700-0859	2200-2359	Service le	aves RDH
	Direct services from Redhill to each station	Origin	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)
	Reigate	Redhill	4	8	06:45	23:00	0	0	-	-	0	0		-
88	Nutfield	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	
.io ≡	Godstone	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	- 1
흔들	Edenbridge	Redhill	2	6	05:45	23:45	1	5	06:30	23:45	2	1	-	- 1
Se es	Penshurst	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	-
# E	Leigh	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	- 1
<i>듀</i> 2	Tonbridge	Radbill	2	-	05-45	22-45	4	-	06-30	22-45	2	4		

	TABLE 4: REDHILL INBOUND				to Friday			Satu					nday	
	Arrival time in Redhill		0200-0859	1800-0159	Service ar	rives RDH	0200-0659	1900-0159	Service ar	rives RDH	0700-0859	2200-2359	Service a	rrives RDH
	Direct services to Redhill from each station	Destination	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)
b	Tonbridge		3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
88	Leigh	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	1 - 1
-8	Penshurst	Redhill	3	6	05:45	23:45	- 1	5	06:00	23:45	2	1	-	1 - 1
Je.	Edenbridge	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
2 1	Godstone	Redhill	3	6	05:45	23:45	- 1	5	06:00	23:45	2	1	-	- 1
국 등	Nutfield	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
S S	Reigate	Redhill	3	11	07:00	23:15	0	1	-	-	0	0	-	-



BML Outer 2: Brighton Main Line Minimum Service Specification (Outer Services only) and First and Last Train Specification

TABLE 1: SOUTHBOUND  Departure time from London	, [	Journe	ey Time	0200-0659	9 0700-155	9 0700-15	559 1600-18	fonday to Frid	day 1900-195	9 2000-2059	2100-2159	2200-015	9 Service lear	ves London	Journe	/Time	0200-0659	Saturda 0700-2159	y 0700-2159 2	2200-0159	Service leave	es London	Journe	y Time	0200-0856	Su 9 0900-2059	inday 9 0900-205	2100-0159	e leaves Lo	ondon
													First (no	Last (no								Last (no earlier							First (no	Last (no earlier
Direct services from specified origin to each station	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
Sanderstead Riddlesdown	London London			5	2	22	12 12	4	2 2	3	3 2	6	06:00 06:00	23:45 23:45			2	2 2	30	5 4	06:15 06:15	23:45 23:45			2	2 2	24 24	2	07:45 07:45	22:30 22:30
Upper Warlingham	London			6 (note 3)	2	22	12	4	2	3	3	6	06:00	23:45			2	2	30 30	5	06:15	23:45			2	2	24	2	07:45	22:30
Woldingham Oxted	London London			4 7 (note 3)	2 3	21 31	12 16	4 5	2 3	2 3	2 3	4 6	06:00 05:30	23:45 23:45			2 4	2 3	30 45	4 6	06:15 05:15	23:45 23:45	l		2 2	2 2	24 24	2 2	07:45 07:45	22:30 22:30
Hurst Green	London			7 (note 3)	3	31	16	5	3	3	3	5	05:30	23:15			4	3	45	5	05:15	23:30			2	2	24	2	07:45	22:30
Lingfield Dormans	London London			5 5	2 2	21 21	12 12	4 4	2 2	2 2	2 2	3	05:30 05:30	23:15 23:15			3	2 2	30 30	3	05:15 05:15	23:30 23:30	l		2 2	2 2	24 24	2 2	07:45 07:45	22:30 22:30
East Grinstead	London London Bridge	52	58	5 2	2	21	12	4 2	2	2	2	3	05:30	23:15	N/A	N/A	3	2	30	3	05:15	23:30	N/A	N/A	2	2	24	2	07:45	22:30
	London Victoria	52 55	68	3	2	18	6	2	2	2	2	3			55	68	3	2	30	3			55	68	2	2	24	2		
Edenbridge Town Hever	London Bridge (note 2) London Bridge (note 2)			2 (note 3) 2	1	10	6 (note 4	2 (note 4)	1	1 1	1	2	06:15 06:15	23:00 22:00			1	1	15 15	2	06:15 06:15	23:00 22:00			0	1	12 12	2	09:45 09:45	22:30 22:30
Cowden	London Bridge (note 2)			2	i	10	5 (note 4	) 1 (note 4)	1	1	i	1	06:15	22:00			i	i	15	1	06:15	22:00			0	1	12	2	09:45	22:30
Ashurst Eridge	London Bridge (note 2) London Bridge (note 2)			2 2 (note 3)	1	10 10	4 6 (note 4	1 2 (note 4)	1 1	1 1	1 1	1 2	06:15 06:15	22:00 23:00			1 1	1 1	15 15	1 2	06:15 06:15	22:00 23:00			0	1	12 12	2 2	09:45 09:45	22:30 22:30
Crowborough	London Bridge (note 2)			2 (note 3)	1	10	6 (note 4	) 2 (note 4)	1	1	1	2	06:15	23:00			1	1	15 15	2	06:15	23:00	l		0	1	12	2	09:45	22:30
Buxted Uckfield	London Bridge (note 2) London Bridge (note 2)	76	82	2 (note 3) 2 (note 3)	1 1	10	6 (note 4	2 (note 4) 2 (note 4)	1 1	1 1	1 1	2 2	06:15 06:15	23:00	76	82	1	1	15 15	2 2	06:15 06:15	23:00			0	1 1	12 12	2 2	09:45 09:45	22:30
Coulsdon South	London			8	4	36	16	6	5	4	4	9	05:00	00:15			5	4	60	9	05:00	00:15			4	3	36	8	06:30	23:45
Merstham Redhill	London London			8 11	6	36 56	15 19	6	5 7	6	4 6	9 13	05:00 02:30	00:15 01:00			5	6	60 90	9 13	05:00 04:00	00:15 01:00			4 10	2 5	24 60	6 12	06:30 04:00	23:45 01:00
	London Bridge			1 10	4	36 20	4 15	4	4	4	4	7					0	4	60 30	8					2	3	36 24	6		
Reigate (note 1)	London Victoria London			0	1	9	5	2	1	0	0	0	07:00	19:45			1	1	15	0	06:00	21:00			0	0	0	0		
	London Bridge London Victoria	41 37	42 48	0	0	0	0	0	0	0	0	0			N/A	N/A	0	0	0	0			N/A N/A	N/A N/A	0	0	0	0		
Nutfield	London	31	40	3	1	10	4	1	0	0	0	0	06:00	18:45	37	40	1	1	15	0	06:30	21:00	INA	INA	0	1	12 12	2	09:00 09:00	22:00 22:00
Godstone Edenbridge	London London			3	1	10	4	1	0	0	0	0	06:00 06:00	18:45 18:45			1	1	15 15	0	06:30 06:30	21:00 21:00			0	1	12 12	2	09:00 09:00	22:00 22:00
Penshurst	London			3	1	10	4	1	0	0	0	0	06:00	18:45			i	1	15	0	06:30	21:00	l		0	1	12	2	09:00	22:00
Leigh	London London			3	1	10 10	4	1 1	0	0	0	0	06:00 06:00	18:45 18:45			1	1	15 15	0	06:30 06:30	21:00 21:00			0	1	12 12	2	09:00 09:00	22:00 22:00
Tonbridge (note 1)	London Bridge	67	67	0	0	2	0	0	0	0	0	0	00.00	10.40	N/A	N/A	0	0	0	0	00.30	21.00	67	67	0	1	12	2	05.00	22:00
Enthumed	London Victoria London	67	93	3	1	8	4	1 4	0	0	0	0	05:30	23:45	67	93	1 4	1	12	0	05:30	23:45	N/A	N/A	0	0	0	0	07:45	22.22
Earlswood Salfords	London			6	2	19	12 12	4	2	2	2 2	4	05:30	23:45			4	2	30 30	4	05:30	23:45	l		1	1	24 12	2	08:30	22:30 22:30
Horley Gatwick Airport (including Gatwick Express)	London			9 32	3 18 (note 7	27 160	18 59	6	4 20	2 18	2 18	6 36	02:00 02:00	01:00 01:00			7 27	3 18	45 270	6 36	02:00 02:00	01:00	l		10 36	3 15	36 180	7 40	02:00 02:00	01:00 01:00
Gatwick Airport (including Gatwick Express)	London London Bridge			3	8	72	29	20 10	20 10	8	8	36 16	02:00	01:00			10	8	120	17	02.00	01:00			36 6	6	72	15	02:00	U1:U0
Gatwick Airport (Express society and	London Victoria	N/A	31 (note 6)	21	10	88	30	10	10	10	10	20	03:30	00:30	N/A	21 (nets 6)	17	10	150	16	03:30	00:30	N/A	31 (note 6)	30	9	108 48	25	03:30	00:30
Gatwick Airport (Express services only) Three Bridges	London Victoria (note 5) London			22 (note 8)	8	36 76	12 46	4 15	10	8	9	20	03:30	00:30 01:00		31 (note 6)	17 (note 8)	8	60 121	10 19	03:30	00:30 01:00	N/A		16 (note 12		92	14 21	03:30	00:30
	London Blackfriars (not London Bridge) London Bridge	52 41	57 58	15 3	0	0 56	0 29	0	0 8	0	0	1 14			N/A 41	N/A 58	0	0	0 90	0 13			N/A 41	N/A 58	2	0	0 68	1 12		
	London Victoria	41 42	58 60	11	2	20	19	5	2	2	3	8			41 42	58 60	7	2	31	6			41 42	58 60	10	2	24	7		
Crawley Ifield	London London			6 (note 8) 5 (note 8)	4 2	36 19	16 10	5	5	4 2	4 2	5	05:30 05:30	23:45 23:45			3 (notes 8 & 10) 3 (notes 8 & 10)	4 2	60 30	5	06:00 06:00	23:45 23:45			4 (note 12 3 (note 12		36 24	7	07:00 07:00	23:00 23:00
Faygate	London			3	ő	6	4	1	1	0	0	0	05:30	19:00			0	ō	0	ō	-	-	l		0	0	0	ő	-	-
Littlehaven Horsham	London London			5 (note 8) 6 (note 8)	2	19 36	12 15	4 5	3 5	2	2	4 5	05:30 05:30	23:45 23:45			3 (notes 8 & 10) 3 (notes 8 & 10)	2	30 60	4 5	06:00 06:00	23:45 23:45	l		3 (note 12 4 (note 12	2) 2	24 36	5 7	07:00 07:00	23:00 23:00
Horsilan	London Bridge	71	75	3	2	19	7	3	2	2	2	4	00.30	23.40			1	2	30	4	00.00	23.45			0	2	24	4	07:45	22:30
Christ's Hospital	London Victoria London	55	83	4 2 (note 8)	2	27 13	12	2	2	2	2	2	06:00	22:30			2 2 (notes 8 & 10)	2	30 15	1 1	06:00	22:30			2	1 1	12 12	2	08:00	22:00
Billingshurst	London			2 (note 8)	2	18	7	3	2	2	2	1	06:00	22:30			2 (notes 8 & 10)	2	30	1	06:00	22:30			2	1	12	2	07:00	22:00
Pulborough Amberley	London London			2 (note 8) 2 (note 8)	2	18 11	7	3 2	2	2	2	1 1	06:00 06:00	22:30 22:30			2 (notes 8 & 10) 2 (notes 8 & 10)	2	30 15	1	06:00 06:00	22:30 22:30	l		2	1 4	12 12	2 2	07:00 08:00	22:00 22:00
Arundel	London			2 (note 8)	2	18	7	3	2	2	2	1	06:00	22:30			2 (notes 8 & 10)	2	30	1	06:00	22:30	l		2	1	12	2	07:00	22:00
Ford	London			2 (note 8) 2 (note 8)	2	18	5	2	2	2 2	2 2	2	06:00 06:00	22:30 22:30			2 (notes 8 & 10) 2 (notes 8 & 10)	2	30 30	3	06:00	23:15	l		2 2	1 2	12 24	2 3	07:00 07:00	22:00 22:00
Barnham Bognor Regis	London London			1 (note 8)	2	18	7	3	2	2	2	1	06:00	22:15			0	2	30 0	1	07:30	23:15			2	1	12	2	07:00	22:00
	London Bridge London Victoria	98 104	98 99	0	0	16	2	1 2	0 2	0 2	0	0			N/A 104	N/A 99	0	0 2	0 25	0			N/A 104	N/A 99	0 2	0	0 12	0 2		
Chichester	London Victoria	1.04		1 (note 8)	2	18	6	2	2	2	2	1	06:30	22:30		, s	2 (notes 8 & 10)	2	30	2	06:00	23:15		35	1	1	12	1	08:15	21:15
Fishbourne Bosham	London Victoria London Victoria			0	0	0	4 5	2 2	1 1	1 1	1 1	0	16:15 16:15	21:30 21:30			0	0	1	0			l		0	0	0	0	:	:
Nutbourne	London Victoria			0	0	0	2	1	1	1	1	0	17:15	21:30			ő	ō	1	ő			l		ő	0	0	1		-
Southbourne Emsworth	London Victoria London Victoria			0	1 1	9	6	2 2	2 2	2 2	2 2	0	07:15 07:15	21:30 21:30			1 (note 8) 1 (note 8)	1 1	15 15	0	06:00 06:00	21:30 21:30	l		1 1	1 1	12 12	1 1	08:15 08:15	21:15 21:15
Warblington	London Victoria			0	0	0	2	1	1	1	1	0	17:15	21:30			0	o l	1	0	- 1		l		0	0	0	0	l - I	-
Havant Bedhampton	London Victoria London Victoria			0 (note 8)	2	18	6 2	2	2	2	2	0	07:15 16:15	21:30 17:00			2 (notes 8 & 10)	2	30	0	06:00	21:30	l		1 0	1 0	12 0	1 0	08:15	21:15
Hilsea	London Victoria			0	0	0	2	1	0	0	0	0	16:15	17:00			0	0	0	0		-	1		0	0	0	0		Ė
Fratton Portsmouth & Southsea	London Victoria London Victoria	118	142	0	1 1	9	3	1 1	1 1	1 1	1 1	0	08:00 08:00	21:30 21:30	118	142	1 (note 10) 1 (note 10)	1	15 15	0	06:30 06:30	21:30 21:30	l		1 1	1 1	12 12	1 1	08:15 08:15	21:15 21:15
Portsmouth Harbour	London Victoria			0	1	9	2	0	o	1	1	0	08:00	21:00		-	1 (note 10)	1	15	0	06:30	21:00			1	1	12	1	08:15	21:15
Cosham Portchester	London Victoria London Victoria			0 (note 8) 0 (note 8)	1 1	9	3 2	1 1	1 1	1 0	1 0	0	07:15 08:30	21:00 19:30			1 (note 8) 1 (note 8)	1	15 15	0	06:00 06:00	21:00 21:00	l		0	0	0	0	:	:
Fareham	London Victoria			0 (note 8)	i	9	3	i	1	1	1	0	07:15	21:00			1 (note 8)	1	15	0	06:00	21:00	l		ő	0	ŏ	0	:	-
Swanwick Netley	London Victoria London Victoria			0 (note 8) 0 (note 8)	1 0	9	3 0	1 0	1 0	1 0	1 0	0	07:15	21:00			1 (note 8) 0	1 0	15 0	0	06:00	21:00	l		0	0	0	0	:	:
Woolston	London Victoria			0 (note 8)	0	1	0	0	1	0	0	0	15:45	19:30			0	0	0	0	-	-	l		0	0	0	0	-	-
St. Denys Southampton Central	London Victoria London Victoria	149	164	0 0 (note 8)	0	9	0	0	1 1	0	0	0	15:45 07:15	19:30 21:00	149	164	0 1 (note 8)	0	0 15	0	06:00	21:00	l		0	0	0	0	:	:
Balcombe	London	3		4	1	9	7	2	2	2	2	3	05:00	23:30			5	1	15	4	04:30	23:30	1		3	1	12	3	06:30	23:30
Haywards Heath	London London Bridge			15	6	56 36	39 22	15	8	8	8	14	04:00	01:00			9	6	97 60	17	04:00	01:00			12	5 2	60 24	15 5	04:00	01:00
	London Victoria			8	2	20	17	7	4	4	4	9					3	2	30	9					9	3	36	10	05.15	
Wivelsfield Plumpton	London London			9 (note 9) 1 (note 9)	3	9	15	6 2	1	1	5	8	04:00 06:45	23:30 22:45			7 (note 9) 0	3	45 15	8	04:30 08:00	23:45 22:45	-		5	1	24 12	5 2	05:45 08:45	23:30 22:45
Cooksbridge	London			1	0	3	5	2	1	0	1 1	o	06:45	21:45			0	0	0	o o	-	-	l		0	0	0	0	-	-
Lewes Southease	London London Victoria	65	67	2 (note 9)	0	18	7 0	3	0	0	0	0	05:30	23:45	65	67	0 (note 9) 0	0	30 0	2	07:45	23:45	-	-	1 0	1 0	12	3	08:45	23:45
Newhaven Town	London Victoria			0 (note 9)	0	0	1	1	0	0	0	0	-	-			0	0	0	0	-	-	l		0	0	0	0	-	-
Newhaven Harbour Bishopstone	London Victoria London Victoria			0 (note 9) 0 (note 9)	0	0	0	0	0	0	0	0	:	- 1			0	0	0	0	:	-	l		0	0	0	0	:	:
Seaford	London Victoria	92	92	0 (note 9)	0	0	1	1 1	ō	0	0	0			92	92	0	0	0	0	-	-			0	0	0	0	-	-
Glynde Berwick	London London			1 1	0	1	3 4	2 2	1 1	1 1	1 1	0	05:30 05:30	21:15 22:45			0 0 (note 9)	0	3	0	: 1	- 1	l		0	0	0	0	:	:
Polegate	London			2	0 2	18	7	2	2	2	2	2	05:30	23:45			0 (note 9)	2	30	2	07:45	00:00	l		1	0	12	0	08:45	00:00
Hampden Park Eastbourne	London London			2 2	2 2	18 18	7 7	3	2 2	2 2	2 2	2	05:45 05:30	23:45 23:45			0 (note 9) 0 (note 9)	2 2	30 30 0 27 15	2 2	07:45 07:45	23:45 23:45	l		1 1	1 1	12 12	3	08:45 08:45	00:00 00:00
Lancoullie	London Bridge	90	93	0	0 2	0	7 5		0 2	0	0	0	33.30	20.40	N/A	N/A 104	0	0	0	0	51.745	20.40	N/A	N/A	0	0	0	0	33.43	30.00
Pevensey & Westham	London Victoria London Victoria	88	104	0 2	2	17	5 4	1 2 2	2	2	1 1	0	05:30	22:45	88	104	0 0 (note 9)	2	27 15	5	07:45	22:45	88	104	0	1 1	0 12 12	1	08:45	21:45
	London Victoria London Victoria			1	0	2	0	0	Ö	o	o	0	07:00	15:45			0 (note 9)	o l	0	0		- 45	l		o	0	0	0	08:45	- 21:45
Pevensey Bay			1	1	0	2	0 4	0 2	0	0	0	0	07:00 05:30	15:45 22:45			0 0 (note 9)	0	0 15	0	07:45	22:45	l		0	0	0 12	0	08:45	21:45
Pevensey Bay Normans Bay	London Victoria		1		1 1	1 ~	4	2	1	1	1	1	05:30	22:45			0 (note 9)	1	15	i	07:45	22:45	l		i	1	12	1	08:45	21:45
Pevensey Bay Normans Bay Cooden Beach Collington	London Victoria London Victoria			2	1	9		2	1 1	1 1	1	1	05:30	22:45			0 (note 9)	1	15	1 1	07:45	22:45		i .	1 4	1 1	12	1 1	08:45	21:45
Pevensey Bay Normans Bay Cooden Beach Collington Beshill	London Victoria London Victoria London Victoria			2 2	1	9	4	2					05:30	22:45 22:45	123	139	0 (note 9) 0 (note 9)		10				l		1 1	1 2	40			24.45
Pevensey Bay Normans Bay Cooden Beach Collington Beshill St. Leonards Warrior Square	London Victoria London Victoria	123	139	2	1 1 1	9 9	4 4 4	2 2	1	1 1	1	1	05:30					1 I	15 15	1 1	07:45 07:45	22:45 22:45			1	1 1	12 12	1	08:45 08:45	21:45 21:45
Pewnisey Bay Normans Bay Cooden Beach Coolington Beshill St. Leonards Warrior Square Hastings Ore	London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria	123	139	2 2	1 1 1 1	9	4 4 4 3	2 2 2	1 1 1	1 1 0	1 1 0	1 0	05:30	19:45			0 (note 9)	1 1	15	1 0	07:45	22:45 21:45			1 1 1	1 1	12 12 12	1 1 0	08:45	21:45 21:45 19:45
Pewensey Bay Normans Bay Cooden Beach Collington Beshill St. Leonards Warrior Square Hastings Ore Burgess Hill Hassocks	London Victoria London Victoria London Victoria London Victoria London Victoria	123	139	2 2	1 1 1 1 1 5		4 4 4 3 25	2 2 2 2 9 8	1 1 1 6 4	1 1 0 5 4	1 1 0 5 4	1 0 10 9	05:30 04:00 04:00	23:30 23:30				1 1 5 3	15 75 46	1 0 10 9	07:45 04:30 04:30	22:45 21:45 23:45 23:45			1 1 1 1 9 8	1 1 1 4 3	12 12	1 1 0 11 8	08:45 05:45 05:45	21:45 21:45 19:45 01:00 23:30
Pewensey Bay Normans Bay Cooden Basch Coolination Beshill St. Leonards Warrior Square St. Leonards Warrior Square Green Burgess Hill Hassocks Preston Park	London Victoria London London London London	123	139	2 2 2 2 2 2	1 1 1 1 5 3	9 9 45 27 28	4 4 4 3 25 21 16	2 2 2 9	1 1 1 6 4 4	1 1 0 5 4 4	1 0 5	1 0 10	05:30 04:00 04:00 04:00	23:30 23:30 23:30			0 (note 9) 7 7 7	1 1 5 3 3	15 75 46 47	1 0 10	07:45 04:30 04:30 04:30	22:45 21:45 23:45 23:45 23:45			1 1 1 9 8 4	1 1 1 4 3 1	12 12 12 48 36 12	1 1 0 11 8 3	08:45 05:45 05:45 05:45	21:45 21:45 19:45 01:00 23:30 23:30
Peensey Bay Normans Bay Cooden Beach Coolination Beshill St. Leonards Warrior Square Hastings Generals Hill Hassocks Preston Park Hove Portslade	London Victoria London London London London London	123	139	2 2 2 2 2 2 10 10 8 0	3 2 1	9 9 45 27 28 18 9	4 4 4 3 25 21 16	2 2 2 9 8 7 5	2 2	2 2	1 0 5 4 4 2 2	1 0 10 9 8 4 4	05:30 04:00 04:00 04:00 07:45 08:00	23:30 23:30 23:30 23:45 23:45			0 (note 9) 7 7 7 7 0 (note 11) 0 (note 11)	1 1 5 3 3 2 1	15 75 46 47 30 15	1 0 10 9 8 4 4	07:45 04:30 04:30 04:30 07:45 08:00	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45			1 1 1 1 9 8 4	1 1 1 1 4 3 1	12 12 12 48 36 12 12 0	3 4 4	08:45 05:45 05:45 05:45 08:15 21:15	21:45 21:45 19:45 01:00 23:30 23:30 23:45 23:45
Pewniere Bay Normans Bay Cooden Bach Cothration St. Leonards Warrior Square Hastings Ore Burgess Hill Hassooks Physion Park Profision Park Portslade Shoreham-by-Sea	London Victoria London London London London London London London	123	139	2 2 2 2 2 2 10 10 8 0 0	1 1 1 1 5 3 3 2 1 2	9 9 45 27 28 18 9 18	4 4 4 3 25 21 16 13 12 13	2 2 2 9 8 7 5 5	2 2 2	2 2 2	1 0 5 4 4 2 2 2	1 0 10 9 8 4 4	05:30 04:00 04:00 04:00 07:45 08:00 07:45	23:30 23:30 23:30 23:45 23:45 23:45 23:45			0 (note 9)  7  7  7  0 (note 11)  0 (note 11)  0 (note 11)	1 1 5 3 2 1 2	15 75 46 47 30 15	1 0 10 9 8 4 4 4	07:45 04:30 04:30 04:30 07:45 08:00 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45			1 0 1	1 1 0 1	12 12 12 48 36 12 12 0 12	3 4 4 4	08:45 05:45 05:45 05:45 05:45 08:15 21:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45
Peensey Bay Normans Bay Cooden Beach Coolination Beshill St. Leonards Warrior Square Hastings Generals Hill Hassocks Preston Park Hove Portslade	London Victoria London London London London London			2 2 2 2 2 2 10 10 8 0	3 2 1 2 1 0	9 9 45 27 28 18 9 18 9	4 4 4 3 25 21 16 13 12 13 12 0	2 2 2 9 8 7 5	2 2 2 2 2 0	2 2	1 0 5 4 4 2 2	1 0 10 9 8 4 4	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45	23:30 23:30 23:30 23:45 23:45			0 (note 9) 7 7 7 7 0 (note 11) 0 (note 11)	1 1 5 3 3 2 1 2 1 0	15 75 46 47 30 15 30 15 0	1 0 10 9 8 4 4	07:45 04:30 04:30 04:30 07:45 08:00 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45			1 1 1 1 9 8 4 1 0 1 0 0	1 1	12 12 12 48 36 12 12 0 12 0	3 4 4	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45
Pewensey Bay Normans Bay Cooden Beach Coolination Berbill St. Leonards Warrior Square Hastings O're Burgess Hill Hassocks Prestor Park Portslade Shoreham-by-Sea Lancing East Worthing Worthing Worthing	London Victoria London	123	139	2 2 2 2 2 2 2 10 10 8 0 0 0	3 2 1 2 1 0	9 9 45 27 28 18 9 18 9	4 4 4 3 25 21 16 13 12 13 12 0	2 2 2 9 8 7 5 5 5 5	2 2 2 2 2 0	4 2 2 2 2 2 0 2	1 0 5 4 4 2 2 2 2 2 0	1 0 10 9 8 4 4 4	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45	23:30 23:30 23:30 23:45 23:45 23:45 23:45 -	80	82	0 (note 9) 7 7 7 0 (note 11) 0 0 (note 11)	1 1 5 3 3 2 1 2 1 0 2	15 75 46 47 30 15 30 15 0	1 0 10 9 8 4 4 4 4 4 0	07:45 04:30 04:30 04:30 07:45 08:00 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45			4 1 0 1 0	1 1 0 1	12 12 12 48 36 12 12 0 12 0	3 4 4 4 3	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45 23:45
Pevensey Bay Normans Bay Cooden Basch Coolination Bestill Set Statement One Burges Hill Hassocks Preston Park Hove East Worthing Worthing Worthing West Worthing	London Victoria London			2 2 2 2 2 2 2 10 10 8 0 0 0 0	3 2 1 2 1 0 2 2	9 9 45 27 28 18 9 18 9 1 1 18	4 4 4 3 25 21 16 13 12 13 12 0 13 13	2 2 9 8 7 5 5 5 5 5 5	4 2 2 2 2 2 0 2 2	4 2 2 2 2 2 0 2	1 0 5 4 4 2 2 2 2 2 2 2 2 2 2 2	1 0 10 9 8 4 4 4 4 4 1	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45	23:30 23:30 23:30 23:45 23:45 23:45 23:45 23:45 23:45 22:45	80	82	0 (note 9) 7 7 7 7 0 (note 11)	1 1 5 3 3 2 1 2 1 0 2 2	15 75 46 47 30 15 30 15 0 30	1 0 10 9 8 4 4 4 4 4 0 4 2	07:45 04:30 04:30 04:30 07:45 08:00 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45			4 1 0 1 0	1 1 0 1	12 12 12 48 36 12 12 0 12 0 12 0 12	3 4 4 4 3 0	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:35 23:45 23:45 23:45 23:45 23:45 21:15
Pevensey Bay Normans Bay Cooden Basch Collination Beshill St. Leonards Warrior Square More Coolen Basch Designer Burgess Hill Hassocks Preston Park Hove Hove Lancing Union Worthing Worthing Worthing West Worthing Durrington-on-Sea Goring-by-Sea	London Victoria London			2 2 2 2 2 2 2 10 10 8 0 0 0 0 0 0	3 2 1 2 1 0	9 9 45 27 28 18 9 18 9 1 18 18 18	4 4 4 3 25 21 16 13 12 13 12 0 13 13 13 13 13	2 2 9 8 7 5 5 5 5 5 5 5 5	4 2 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2	4 2 2 2 2 2 0 2	1 0 5 4 4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 10 9 8 4 4 4 4 4	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:45 23:45 23:45 23:45 23:45 23:45 22:15 22:15 22:15	80	82	0 (note 9) 7 7 7 0 (note 11)	1 1 5 3 2 1 2 1 0 2 2 2 2 2	15 75 46 47 30 15 30 15 0 30 30 30 30 30	1 0 10 9 8 4 4 4 4 0 4 2 2 2	07:45 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15			4 1 0 1 0	1 1 0 1	12 12 12 48 36 12 12 0 12 0 0 12 12 12 12	3 4 4 4 3 0	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 - 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45 23:45 21:15 21:15
Pewares Bay Normans Bay Cooden Bach Collingtion St. Leonards Warrior Square Hastings Ore Burgess Hill Hassouss Pression Park Pression Park Portainde Shoreham-by-Sea Lancing East Worthing Worthing Worthing Worthing Worthing Worthing Gonne-by-Sea Gonne-by-Sea Gonne-by-Sea Gonne-by-Sea Angmering	London Victoria London			2 2 2 2 2 2 10 10 0 0 0 0 0 0	3 2 1 2 1 0 2 2 2 2	9 9 9 45 27 28 18 9 18 18 18 18 18	4 4 4 3 25 21 16 13 12 13 12 0 13 13 13 13 13	2 2 9 8 7 5 5 5 5 5 5 5 5 5	4 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 2 0 2 2	1 0 5 4 4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 10 9 8 4 4 4 0 4 1 1 1	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:45 23:45 23:45 23:45 23:45 22:15 22:15 22:15 22:15	80	82	0 (note 9) 7 7 7 0 (note 11)	1 1 5 3 3 2 1 2 1 0 2 2 2 2 2 2	15 75 46 47 30 15 30 15 0 30 30 30 30 30	1 0 10 9 8 4 4 4 4 4 0 4 2	07:45 04:30 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15			4 1 0 1 0	1 1 0 1	12 12 12 48 36 12 0 12 0 12 12 12 12 12 12	3 4 4 4 3 0	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45 23:45 21:15 21:15 21:15 21:15
Pevensey Bay Normans Bay Cooden Basch Collination Beshill St. Leonards Warrior Square More Coolen Basch Designer Burgess Hill Hassocks Preston Park Hove Hove Lancing Union Worthing Worthing Worthing West Worthing Durrington-on-Sea Goring-by-Sea	London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London	80	82	2 2 2 2 2 2 2 10 10 8 0 0 0 0 0 0	3 2 1 2 1 0 2 2 2 2	9 9 45 27 28 18 9 1 18 18 18 18 18 18	4 4 4 3 25 21 16 13 12 13 12 0 13 13 13 13 13 13 13	2 2 9 8 7 5 5 5 5 5 5 5 5	4 2 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 2 0 2 2	1 0 5 4 4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 10 9 8 4 4 4 4 4 1	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:45 23:45 23:45 23:45 23:45 23:45 22:15 22:15 22:15	N/A	N/A	0 (note 9) 7 7 7 0 (note 11)	1 1 5 3 3 2 1 1 0 2 2 2 2 2 2 2 2	15 75 46 47 30 15 30 15 0 30 30 30 30 30 30 30 30	1 0 10 9 8 4 4 4 4 0 4 2 2 2	07:45 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15	N/A	N/A	4 1 0 1 0	1 1 0 1	12 12 12 48 36 12 0 12 0 0 12 12 12 12 12 12 12 12 12 12	3 4 4 4 3 0	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 - 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45 23:45 21:15 21:15
Pewniere Bay Normans Bay Cooden Bach Coolination St. Leonards Warrior Square Hastings Ore Burgess Hill Burgess Hill Burgess Hill Hose Pestatos Pestatos Pestatos Portalade Shoreham-by-Sea Lancing West Worthing West Worthing Unrindton-on-Sea Gorins-by-Sea Angmering Littlehampton	London Victoria London	80	82	2 2 2 2 2 2 10 110 8 8 0 0 0 0 0 0 0 0 0 0 1 1 0 0 0 0	3 2 1 2 1 0 2 2 2 2 2 2 2 2 2 2 2	9 9 9 45 27 28 18 9 1 1 18 18 18 18 18 18 18 18	4 4 4 4 3 25 21 16 13 12 12 13 12 2 0 13 13 13 13 13 13 13 13 15 5 5 5 5 7	2 2 2 9 8 7 5 5 5 5 5 5 5 5 5 5 5 5 5 1 1 4 4 4 4 4	4 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	4 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 5 4 4 4 2 2 2 2 2 2 2 2 2 2 2 2 0 1 1	1 0 10 9 8 4 4 4 4 4 1 1 1 1 1 0 0	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:35 23:45 23:45 23:45 23:45 22:15 22:15 22:15 22:15 22:15 21:45			0 (note 9) 7 7 7 0 (note 11)	3 2 1 2 1 0 2 2 2 2 2 2 2 2	15 75 46 47 30 15 30 15 0 30 30 30 30 30 30 30 30 27	1 0 10 9 8 4 4 4 4 4 0 4 2 2 2 2 1 1 0 5 5	07:45 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15 23:15 23:15	N/A 102	N/A 109	4 1 0 1 0 0 1 1 1 1 1 1 1 0 0	1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	12 12 12 48 36 12 12 0 0 12 12 12 12 12 12 12 12 12 12 12 13	3 4 4 4 3 0 4 1 1 1 1 1	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:30 23:45 23:45 23:45 21:15 21:15 21:15 21:15
Pewares Bay Normans Bay Cooden Bach Collingtion St. Leonards Warrior Square Hastings Ore Burgess Hill Hassouss Pression Park Pression Park Portainde Shoreham-by-Sea Lancing East Worthing Worthing Worthing Worthing Worthing Worthing Gonne-by-Sea Gonne-by-Sea Gonne-by-Sea Gonne-by-Sea Angmering	London Victoria London Victoria London Victoria London Victoria London Victoria London Victoria London	80	82 102 109	2 2 2 2 2 2 2 2 2 2 0 100 100 8 8 0 0 0 0 0 0 0 0 0 0 0 0 0	3 2 1 2 1 0 2 2 2 2 2 2 2 2	9 9 45 27 28 18 9 18 18 18 18 18 18 18 18 18 18 18 18 18	4 4 4 4 3 25 21 16 16 13 12 12 13 13 13 13 13 13 15 5 5 26 0	2 2 2 9 8 7 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	4 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 5 4 4 2 2 2 2 2 2 2 2 2 2 2 2 9 0 1 9 0 0	1 0 10 9 8 4 4 4 4 4 0 0 4 1 1 1 1 0 0 0 0 1 1 3 0 0 0 1 1 0 0 0 1 1 1 0 0 0 0	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:45 23:45 23:45 23:45 23:45 22:15 22:15 22:15 22:15	N/A	N/A	0 (note 9) 7 7 7 0 (note 11)	3 2 1 2 1 0 2 2 2 2 2 2 2 2	15 75 46 47 30 15 30 30 30 30 30 30 30 30 27 120	1 0 10 9 8 4 4 4 4 4 0 4 2 2 2 2 2	07:45 04:30 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15	N/A 102		4 1 0 1 0 0 1 1 1 1 1	1 0 1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1	12 12 12 48 36 12 0 12 0 0 12 12 12 12 12 12 12 12 12 12 12 13 60 0	3 4 4 3 0 4 1 1 1	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:45 23:45 23:45 23:45 23:45 21:15 21:15 21:15 21:15
Pewniere Bay Normans Bay Cooden Bach Coolination St. Leonards Warrior Square Hastings Ore Burgess Hill Burgess Hill Burgess Hill Hose Pestatos Pestatos Pestatos Portalade Shoreham-by-Sea Lancing West Worthing West Worthing Unrindton-on-Sea Gorins-by-Sea Angmering Littlehampton	London Victoria London	80 99 102	82	2 2 2 2 2 2 100 100 8 8 0 0 0 0 0 0 0 0 0 0 1 1 0 0 0 155	3 2 1 2 1 0 2 2 2 2 2 2 2 2 2 2 8 (note 7)	9 9 45 27 28 18 9 18 18 18 18 18 18 18 18 18 18 18 18 18	4 4 4 4 3 25 21 16 16 13 12 12 12 12 13 13 13 13 13 13 15 5 5 26 0	2 2 2 9 8 7 5 5 5 5 5 5 5 5 5 5 5 5 5 1 4 4 8 8 8 8 8 7 8 8 8 7 8 8 8 8 8 8 8 8	4 2 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2	4 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1 0 5 4 4 4 2 2 2 2 2 2 2 2 2 2 2 2 1 9 9	1 0 10 9 8 4 4 4 4 4 0 4 1 1 1 1 1 0 0 0 0 0 13	05:30 04:00 04:00 04:00 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	23:30 23:30 23:35 23:45 23:45 23:45 23:45 22:15 22:15 22:15 22:15 22:15 21:45	N/A	N/A	0 (note 9) 7 7 7 0 (note 11)	3 2 1 2 1 0 2 2 2 2 2 2 2 2 2 2 8	15 75 46 47 30 15 30 15 0 30 30 30 30 30 30 30 30 27	1 0 10 9 8 4 4 4 4 4 4 4 2 2 2 2 1 1 0 5 5 15	07:45 04:30 04:30 07:45 08:00 07:45 07:45 07:45 07:45 07:45 07:45 07:45 07:45	22:45 21:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:45 23:15 23:15 23:15 23:15 23:15	N/A 102		4 1 0 0 1 1 1 1 1 1 1 0 0 0	1 0 0 1 1 1 1 1 1 1 1 1 5	12 12 12 48 36 12 12 0 0 12 12 12 12 12 12 12 12 12 12 12 12 12	3 4 4 3 0 4 1 1 1 1 1 0 1	08:45 05:45 05:45 05:45 08:15 21:15 08:15 22:15 08:15 08:15 08:15 08:15 08:15	21:45 21:45 19:45 01:00 23:30 23:30 23:45 23:45 23:45 21:15 21:15 21:15 21:15

BLE 2: NORTHBOUND  Arrival time in Londo	n .	Journe	y Time	0200-0659	0700-0959	9 0800-0859	Monday 1000-1959	1000-1959	2000-2059	2100-2159	2200-015	9 Service an	Last (no	Journe	y Time	0200-0659	Saturd 0700-2159	0700-2159	2200-0159	Service arri	ves London Last (no	Journe	y Time	0200-0859	Sun 0900-2059	0900-2059	2100-0159	Service an
Direct services from each station to specified destination	n Destination	Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no later than)	earlier than)	Average	Maximum	Total	TPH	Total	Total	First (no
ghton	London London Blackfriars (not London Bridge)	83	85	4	23	8	8	80	8	9	15	06:00	01:00	N/A	N/A	4	8	120	21	06:00	00:45	N/A	N/A	8	5	60	14	06:00
	London Bridge London Victoria	61 62	66 119	2 2	12 11	4	4 4	40 40	4 4	4 5	4 9			62	119	2 2	4	60 60	9 12					4	2	24 36	8 6	
lehampton	of which London	56	56	0	0 8	0 4	2	20 20	1 1	0	1	07:30	22:45	56	56	0	2	30 30	0	07:30	22:45	N/A	N/A	0	0	0 12	0	09:00
	London Bridge London Victoria	103 100	115 105	0	4	2 2	0 2	0 20	0	0	0			N/A 100	N/A 105	0	0	0 30	0			N/A 100	N/A 105	0	0	0 12	0	
gmering ring-by-Sea	London London	100	100	0	9	4 4	2 2	20 20	2	2 2	2 2	07:00 07:30	22:45 22:45	100	100	0	2	31 31	2 2	07:30 07:30	22:45 22:45	100	100	1	1	12	1 1	09:00 09:00
rrington-on-Sea est Worthing	London London			0	8	4	2	20 20	2	2 2	2 2	07:30 07:30	22:45 22:45 22:45			0	2	31 31	2 2	07:30 07:30	22:45 22:45 22:45			1	1 1	12 12		09:00 09:00
rthing	London	80	84	0	9	4	2	20	2	2	2	07:00	22:45	80	84	1	2	32 0	2	07:00	22:45			1	1	12	1	09:00
st Worthing ncing	London London			0	9	0 4	1	10	2	2	2	07:00	22:45			0	1	16	0 2	07:30	22:45			1	0	0 1	0	-
oreham-by-Sea rtslade	London London			0	9	4	2	20 10	2 2	2 2	2 2	07:00 07:00	22:45 22:45			1 0	2	32 16	2 2	07:00 07:30	22:45 22:45			1	1 0	12 1	1 0	09:00
ve iston Park	London London			0 2	9	6	3	20 32	3	4	8	07:00 06:15	22:45 01:00			1 2	3	32 48	2 8	07:00 06:30	22:45 00:45			1 2	1 1	12	1 4	09:00 07:30
ssocks gess Hill	London London			2	19 22	7 9	3 5	32 50	3	4 5	10	06:15 06:15	01:00 01:00			2 2	3	48 75	9	06:30 06:30	00:45 00:45			7	3 4	12 36 48	12 13	07:00 07:00
stings	London Victoria London Victoria	125	142	0	0	0	1	10 10	1	1	2 2	11:00 07:15	23:45 23:45	125	142	0	1	15 15	1 2	08:00 09:00	23:45 23:45			0	1	12 12	2 2	10:00
Leonards Warrior Square	London Victoria	125	142	0	5	1	1	10	1 1	1	2	07:15	23:45	125	142	0	1	15	2	09:00	23:45			0	1 1	12	2	10:00
chill lington	London Victoria London Victoria			0	5 5	1	1	10 10	1	1	2 2	07:15 07:15	23:45 23:45			0	1	15 15	2 2	09:00 09:00	23:45 23:45			0	1	12 12	2 2	10:00 10:00
oden Beach rmans Bay	London Victoria London Victoria			0	5	0	0	10 1	0	0	0	07:15 10:00	23:45 17:45			0	1	15 0	2 0	09:00	23:45			0	0	12 0	0	10:00
vensey Bay vensey & Westham	London Victoria London Victoria			0	1 5	0	0	1 10	0	0	0 2	10:00 07:15	17:45 23:45			0	0	0 15	0 2	09:00	23:45			0	0	0 12	0 2	10:00
stbourne	London London Bridge	93	94	1 (note 2) 0	7 2	3 2	2	20	2	2	2	06:45	23:45	N/A	N/A	0	2	30 0	2	08:00	23:45	N/A	N/A	1	1 0	12 0	2	08:30
mpden Park	London Victoria London	89	103	1 1 (note 2)	5	1	2 2	20	2	2 2	2	06:45	23:45	1671	1671	0	2	30	2	08:00	23:45	1671	1671	1	1	12	2	08:30
egate wick	London London	1		1 (note 2) 1 (note 2)	7	3 3 2	2 0	20 20 1	2	2 0	2 2	06:45 07:15	23:45 23:45 23:45			0	2	30 30	2 2 0	08:00	23:45			1 0	1 0	12 12 0	2 2 0	08:30
nde	London			0	6 4	2	0	Ö	0	0	0	08:00	09:45		***	0	0	1	0	- :	- :			0	0	0	0	-
aford hopstone	London Victoria London Victoria	98	100	0	2 2	1	0	0	0	0	0	08:45 08:45	09:45 09:45	98	100	0	0	0	0	:	- :			0	0	0	0	-
whaven Harbour whaven Town	London Victoria London Victoria			0	2	1 1	0	0	0	0	0	08:45 08:45	09:45 09:45			0	0	0	0	:	- :			0	0	0	0	-
uthease wes	London Victoria London	65	74	0 1 (note 2)	7	3	2	0 20	2	2	2	08:45 06:45	09:45 23:45	65	74	0	2	30	2	08:00	23:45			0	1	12	0 2	08:30
oksbridge mpton	London London	1		0	5 6	3	0	7	1 1	0	0	08:00 06:45	20:15 23:45			0	0	0 15	0	08:00	23:45			0	0	0 12	0 2	08:30
velsfield ywards Heath	London London			2 5 (note 2)	21 38	8 15	3 6	30 62	4	4 7	9	06:15 06:00	01:00 01:00			1 4	3	45 91	10 13	06:30 06:00	00:45 00:45			3 10	2 5	24 60	6 15	07:30 06:00
	London Bridge London Victoria			2 3	18 20	8 7	4 2	40 22	4 2	4 3	4 9	55.00	51.00			2 2	4	60 31	9	30.00	20.40			4	2 3	24 36	8 7	50.00
combe uthampton Central	London	450	159	3	8	2	1	10	1	1	4	06:15	01:00	150	450	1	1	15	4	06:30	00:45			1	1	12	4	08:30
Denys	London Victoria London Victoria	150	159	0	0	0	0	10	0	0	0	08:45	22:45	150	159	0	0	15 0	0	08:00	22:45			0	0	0	0	-
olston dey	London Victoria London Victoria			0	0	0	0	0	0	0	0	08:45	09:30			0	0	0	0	:	- :			0	0	0	0	-
anwick eham	London Victoria London Victoria	1		0	2 2	1 1	1	10 10	1 1	1 1	1 1	08:45 08:45	22:45 22:45			0	1	15 15	1 1	08:00 08:00	22:45 22:45			0	0	0	0	-
tchester sham	London Victoria London Victoria	1		0	2 2	1 1	1 1	10	1 1	1 1	1 1	08:45 08:45	22:45 22:45			0	1	15	1 1	08:00	22:45 22:45			0	0	0	0	-
tsmouth Harbour	London Victoria London Victoria London Victoria	120	141	0	2 2	0	1	10	0	0	0 (note 4)	08:00	19:15 20:15	120	141	0	1		0 (note 4) 0 (note 4)		21:00 21:00			0	1 1	12 12	1 1	10:00
tton	London Victoria	120	147	0	2	0	1	10	1	0	0 (note 4) 0 (note 4)	08:00	20:15	120	141	1	1		0 (note 4)	07:00 07:00	21:00 21:00			0	1	12	1	10:00
sea Hampton	London Victoria London Victoria	1		0	3	1	0	0	1	0	0 (note 4) 0 (note 4)	08:00	20:15 20:15			0	0	1	0 (note 4) 0 (note 4)					0	0	0	0	
vant ırblington	London Victoria London Victoria	1		0	6 3	1 1	0	20 1	1	1 0	1 (note 4) 0	08:45	22:45 20:15			1 0	2	30 1	1 (note 4) 0	07:00	22:45			0	1 0	12 0	0	10:00
sworth uthbourne	London Victoria London Victoria	1		0	4	1 1	1	11 10	2 2	1 1	1 1	08:00 08:00	22:45 22:45			0	1	16 16	1	07:45 07:45	22:45 22:45			0	1 1	12 12	1 1	10:00 10:00
thoume sham	London Victoria London Victoria	1		0	3 4	1 1	0	1 2	1 1	0	0	08:45	20:15			0	0	1	0	-	-			0	0	0	0	-
hbourne ichester	London Victoria London Victoria	1		0	4	1	0	1 20	1 2	0	0 1 (note 4)	08:00	20:15 20:15 22:45			0	0	1 30	0 1 (note 4)	07:00	22:45			0	0	0	0	10:00
gnor Regis	London London London Bridge	104	104	0	5	2	2	20	1	1 0	1 (note 4)	08:00	22:15	N/A	N/A	0	2	30	1 (note 4)	07:30	22:30			1 0	1 0	12	3	08:45
	London Victoria	104 105	104 116	0	4	1	2	20	1	1	1	07.00	20.15	N/A 105	N/A 116	0	2	30	1	07.00	22.15			1	1	12	3	00.15
nham d	London London	1		0	8 2	1	2 2	20 20	2	2	2 (note 4) 2 (note 4)	08:30	22:45 22:45			0	2	30 30	2 (note 4) 2 (note 4)	07:45	22:45 22:45			1 1	1	24 12 12	3	08:45 08:45
indel berley	London London			0	6 3	0	1	20 11	1	1	1 (note 4) 1 (note 4)	08:00	22:45 22:45			0	1	30 16	1 (note 4) 1 (note 4)	07:45	22:45 22:45			1	1 1	12	3	08:45 08:45
borough ingshurst	London London			0	6	2 2	2 2	20 20	2 2	1	1 (note 4) 1 (note 4)	07:30	22:45 22:45			0	2	30 30	1 (note 4) 1 (note 4)	07:45 07:45	22:45 22:45			1	1 1	12 12	3 3	08:45 08:45
rist's Hospital rsham	London London			0 2	6 18	2 5	1 4	14 40	1 4	1 4	1 (note 4) 6 (note 4)		22:45 00:15			0	1 4	15 60	1 (note 4) 6 (note 4)	07:45 06:45	22:45 00:15			1 3 (note 6)	1 3	12 36	3 7	08:45 07:45
	London Bridge London Victoria	72 56	76 78	2 0	8	3	2	20	2	2 2	5	00.10	00.10			1 0	2	30	5	00.40	00.10			2	2	24	3 4	08:15 08:45
lehaven	London	56	/0	2	15	5	2	20 20	2	2	5 (note 4)		00:15			1	2	30 30	5 (note 4)	06:45	00:15			2 (note 6)	2	12 24	4	07:45
xgate Id	London London			0 2	4 14	3 4	2	7 20	0 2	2	0 5 (note 4)	08:00 06:15	19:30 00:15			0	2	30	0 5 (note 4)	06:45	00:15			0 2 (note 6)	2	0 24	0 4	07:45
wley ree Bridges	London London	1		2 14 (note 2)	18 42	5 14	8	40 80	8	9	6 (note 4)	06:15	00:15 01:00			1 15	8	60 121	6 (note 4) 18	06:45 02:15	00:15 01:00	-		3 (note 6) 17 (note 6)	8	36 92 (note 7	7	07:45 02:00
	London Blackfriars (not London Bridge) London Bridge	52	61	2 7	0 26	10	0	0 60	0	0	2 9			N/A	N/A	0 10	0	0 90	0 14			N/A	N/A	0	0	0 68	0 10	
twick Airport (including Gatwick Express)	London Victoria London			5 22	16 57	4	2 18	20 180	2	2 18	6 34	02:15	01:45			5 21	2	31 270	4 35	02:15	01:45			9	2 15	24 180	9 41	02:00
	London Bridge			7	30 14	10	8	80 62	8	8	9	GE.13	51.45			5 12	8	121 91	18	06:30 03:00	00:00 00:30			10 23	6	68 61	11 27	JE.00
twick Airport (Express services only)	London Victoria London Victoria	N/A	31 (note 3)	6	13	4	4	40	4	4	14	02:15	01:15			6	4	60	14	02:15	01:15			14	4	48	18	02:15
rley fords	London London	1		1	23 13	6	2	41 20	2	2	6 5	02:45 06:45	01:45 00:15			6 1	4	60 30	9 5	02:45 06:45	01:45 00:15			9	3 1	36 12	8	02:00 08:15
rlswood nbridge (note 1)	London London			1 0	13 6	6 2	1	20 10	0	0	5	06:45 07:15	00:15 19:45			1 0	1	30 15	5	06:45 07:30	00:15 21:00			0	1	24 12	3 2	08:15 09:45
	London Bridge London Victoria	69 65	77 74	0	3 5	1 2	0	1 9	0	0	0			N/A 65	N/A 74	0	0	0 15	0			69 N/A	77 N/A	0	1 0	12 0	2 0	
gh nshurst	London London			0	6	2 2	1	10 10	0	0	0	07:15 07:15	19:45 19:45			0	1	15 15	0	07:30 07:30	21:00 21:00			0	1 1	12 12	2 2	09:45 09:45
enbridge dstone	London London	1		0	6	2 2	1	10 10	0	0	0	07:15 07:15	19:45 19:45			0	1	15 15	0	07:30 07:30	21:00 21:00 21:00			0	1 1	12 12	2 2	09:45 09:45
field	London London London			0	6	2	1	10	0	0	0	07:15 07:15 08:30	19:45	$\sqcup$		0	1	15 15	0	07:30 07:30 07:30	21:00		$\sqcup$	0	1	12	2	09:45
igate (note 1)	London Victoria	38	48	0	3	2	1	11	0	0	0	U8:30	18:45	38	48	0	1	15	0	07:30	21:00	N/A	N/A	0	0	0	0	-
dhill	London Bridge London	46	46	8	24	8	6	64	6	6	9	02:30	01:00	N/A	N/A	5	6	90	12	05:00	00:45	N/A	N/A	6	0 4	0 48	14	06:00
	London Bridge London Victoria			5	24 14 10	4	4 2	40 24	4 2	4 2	5 4					3 2	4 2	60 30	9					2 4	3 2	24 24	4 10	
rstham ulsdon South	London London			4	17	8 8	4	42 42	4	4	5	06:15 06:15	00:15 00:15			3	4	60	9	06:30 06:30	00:15 00:15			2	2	24 36	6 9	07:45 07:45
kfield ded	London Bridge (note 8) London Bridge (note 8)	75	81	2 2	5	2 2	1	10	2 2	2 2	1 (note 5)	06:30	22:45 22:45			0	1	15 15	1 (note 5) 1 (note 5)	08:00	22:45 22:45			0	1	10	2 (note 5) 2 (note 5)	11:30 11:30
wborough	London Bridge (note 8)	1		2	5	2	1	10	2	2	1 (note 5)	06:30	22:45			0	1	15	1 (note 5)	08:00	22:45			0	1	10	2 (note 5)	
dge nurst	London Bridge (note 8) London Bridge (note 8)	1		0	5	1	1	10 10 10	2 2	1	1 (note 5)	07:45	22:45 22:45			0	1	15 15	1 (note 5)	08:00	22:45 22:45			0	1	10	2 (note 5) 2	11:30
wden ver	London Bridge (note 8) London Bridge (note 8)	1		2 0	4	1 1	1 1	10 10 10	2 2	1 1	1 1	06:30 07:45	22:45 22:45			0	1	15 15	1	08:00 08:00	22:45 22:45			0	1 1	10 10	2 2	11:30 11:30
enbridge Town st Grinstead	London Bridge (note 8) London	1	-	2 2	5	4	1 2	10	2 2	2 2	1 (note 5)	06:30	22:45 23:30	$\vdash$		0	2	15	1 (note 5)	08:00 07:45	22:45 23:30	-	$\vdash$	0	1 2	10 24	2 (note 5) 4	11:30 09:15
	London Bridge London Victoria	50 56	53 59	1 1	6	2 2	0 2	4 20	0 2	0 2	0 7	1		N/A 56	N/A 59	0	0	0	0 4		20.00	N/A 56	N/A 59	0	0 2	0 24	0 4	22.10
mans	London	36	39	2	11	4	2	21	2	2	4	06:45	23:30	30	28	0	2	30	4	07:45	23:30	36	Ja	0	2	24	4	09:15
gfield rst Green	London London	1		3	12	6	3	31	4	4	7 (note 5)		23:30 23:30			0	3	30 45	4 5 (note 5)	07:45 07:45	23:30 23:30		$\vdash$	0	2	24 24	4 4 (note 5)	09:15 09:15
ed Ildingham	London London	1		4 2	17 11	6 4	3 2	31 21	4 2	4 2	7 (note 5) 4	06:30	23:30 23:30			0	3 2	45 30	5 (note 5) 4	07:45 07:45	23:30 23:30			0	2 2	24 24	4 (note 5) 4	09:15 09:15
per Warlingham Idlesdown	London London	1		2 2	12 11	4 4	2 2	21	2 2	2 2	4 4	06:30 06:30	23:30 23:30			0	2 2	30 30	4 4	07:45 07:45	23:30 23:30			0	2 2	24 24	4 4	09:15 09:15
desdown	London			2	12	4	2	21	2	2	4	06:30	23:30			ő	2	30	4	07:45	23:30	l		0	2	24	4	09:15
	Note	1: Shuttle servi	ices operatin	g between Re	eigate and	Redhill, and	Tonbridge a	nd Redhill, a	re not includ	led in Table	2. The qua	entum of serv	ices betwee	n these statio	ns is specifi	ied in Table 4	l.											
		<ol><li>An additional</li></ol>																										
	Note	3: Two trains p											104															
	Note Note Note		nal services : al service sha	shall operate all operate fro	from Chich m Uckfield	ester to Thre to East Croy	ee Bridges, a don, calling	at Buxted, C	n interval of i	70 minutes, Eridge, Ed	, arriving bet enbridge To	tween 23:00 a	and 23:59. C reen and Ox	ne service sh led, departing	nall extend to after 22:30	o Portsmouth	Harbour. One se	ervice need n	ot call at Am	nberley, Little	haven or Ifie	eld.						

		TABLE 3: REDHILL OUTBOUND			Monday t	o Friday			Satu	rday			Sur	nday	
		Departure time from Redhill		0200-0859	1800-0159	Service le	aves RDH	0200-0659	1900-0159	Service le	aves RDH	0700-0859	2200-2359	Service le	eaves RDH
		Direct services from Redhill to each station	Origin	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)
Г		Reigate	Redhill	4	8	06:45	23:00	0	0	-	-	0	0	-	-
	88	Nutfield		2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	-
	.S ==	Godstone	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	-
	호등	Edenbridge	Redhill	2	6	05:45	23:45	1	5	06:30	23:45	2	1	-	-
	9 8	Penshurst	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	-
	5 €	Leigh	Redhill	2	5	05:45	22:45	1	4	06:30	22:45	2	1	-	-
	for Sh	Tonbridge	Redhill	2	6	05:45	23:45	1	5	06:30	23:45	2	1	-	-

	TABLE 4: REDHILL INBOUND			Monday t				Satu					nday	
	Arrival time in Redhill		0200-0859	1800-0159	Service at	rives RDH	0200-0659	1900-0159	Service at	rrives RDH	0700-0859	2200-2359	Service a	rrives RDH
	Direct services to Redhill from each station	Destination	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)	Total	Total	First (no later than)	Last (no earlier than)
2	Tonbridge		3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
8	Leigh	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
Š	Penshurst	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
2	Edenbridge	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
9 11	Godstone	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
Juttle	Nutfield	Redhill	3	6	05:45	23:45	1	5	06:00	23:45	2	1	-	-
윤 등	Reigate	Redhill	3	11	07:00	23:15	0	1	-	-	0	0	-	-

#### West Coastway: Minimum Service Specification and First and Last Train Specification

	TABLE 1: WESTBOUND							Mo	nday to Frid	av						1			Sati	urdav				т —			Sur	nday			
	Departure time from origin station		Journe	ev Time	0200-0659	0700-1559	0700-1559	1600-1859			2000-2059	2100-2159	2200-0159	Service lea	ves Brighton	Journe	ev Time	0200-0659			2200-015	9 Service lea	ves Brighto	n Journ	nev Time	0200-085			9 2100-015	9 Service leav	ves Brighto
	Dopartare anno nom engin etation		Court	1	0200 0000	0700 1000	0700 1000	1000 1000	1100 1100	1000 1000	2000 2000	2100 2100	2200 0100	0011100100	Last (no		I	0200 0000	0.00 2.00	0.00 2.00	2200 010	C COI VICO IOU	Last (no		1	0200 000	0 0000 200	0 0000 200	12100 010	550111001001	Last (no
	Direct services from specified origin to each station			1										First (no	earlier							First (no	earlier			1				First (no	earlier
	(excluding services from London)	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	Hove	Brighton	Avelage	Widalindin	Q	6 (note 4)	56	16	6	1000	1000	10001	Q .	05:15	00:00	riverage	IVIGXIIIIGIII	11	6	90	9	05:15	00:00	Avelage	IVIGAIIIIGIII	6	3	36	8	07:15	00:00
		Brighton		1	5	2	18	6	2	2	2	2	1 4	05:15	00:00			1 7	0	30	4	05:30	23:45			2	3	24	6	07:13	22:45
		Brighton		1	6	4 (note 4)	37	10	4	2	2	2	1 4	05:15	00:00			8	1 4	60	4	05:15	23:45			3	2	24	5	07:30	22:45
		Brighton		1	5	2 (note 4)	19	6	2	2	2	2	7	05:15	00:00			1 %	2	30	4	05:30	23:45			3	2	24	5	07:30	22:45
		Brighton		1	5	2 (Hote 4)	36	10	4	2	2	2	4	05:15	00:00			4	1 4	60	4	05:30	23:45			3	2	24	5	07:30	22:45
		Brighton		1	0	4 (note 4)	38	10	4	2	2	2	7	05:15	00:00			"	1 7	60	4	05:30	23:45			5	2	24	5	07:15	22:45
		Brighton			0	4 (note 4)	37	10	4	2	2	2	4	05:15	00:00				1 4	60	4	05:15	23:45			2	2	24	5	07:13	22:45
				1	5	4 (11016 4)	18	6	2	2	2	2	4	05:15	00:00			l °	2	30	4	05:30	23:45			3	2	24	5	07:30	22:45
		Brighton Brighton		1	3	4 (note 4)	38	10	4	2	2	2	7	05:15	00:00			4	1 4	60	4	05:15	23:45			5	2	24	5	07:15	22:45
	· ·				7	3 (note 4)	31	0	4	2	2	2	4	05:15	00:00			0	3	45	4	05:15	23:45			3	2	24	5	07:13	22:45
		Brighton Brighton	I	1	,	1 (note 4)	16	7	4	2	2	2	2	05:15	22:30	l		5	1 1	15	2	05:15	22:30	1		3	2	24	1 1	07:30	22:45
1		Brighton	I	1	6	1 (note 4)	16	7	4	2	2	2	2	05:15	22:30			5		15	2	05:15	22:30	1	1	3	2	24	4	07:30	22:45
	Angmering	Brighton			6	2 (note 4)	22	8	4	2	2	2	2	05:15	22:30			5	2	30	2	05:15	22:30			5	2	24	4	07:30	22:45
		Brighton	<b>†</b>	<b>-</b>	2	0	3	4	2	0	0	0	1	06:00	22:30			1	0	1	1	07:00	22:30	+	1	0	0	0	2	- 07.13	
		Brighton	<b>†</b>	<b>-</b>	4	1 (note 4)	12	7	2	2	2	2	2	05:15	22:30			4	1	15	2	05:15	22:30	+	1	5	2	24	3	07:15	22:45
		Brighton			6	2 (note 4)	20	7	2	2	2	2	2	05:15	22:30			4	2	30	2	05:15	22:30			5	2	24	3	07:15	22:45
3.		Brighton	<b>†</b>	<b>-</b>	0	0	20	1	0	0	0	0	0	00.10				0	0	0	0	- 00.10		+	1	0	0	0	0	- 07.13	-
9	Chichester	Brighton	-	<b>-</b>	6	2 (note 4)	20	6	2	2	2	2	2	05:15	22:30			4	2	30	2	05:15	22:30	+	+	5	2	24	3	07:15	22:45
to		Brighton			1 1	2 (11016 4)	1	0	0		0	Ĺ _	_ ^	03.13	22.50			1 7	1 6	1 1	1 6	03.13	22.50			2	1	12	1	07:13	21:15
ig.		Brighton			2	0	1	0	0	0	0	lő	l o	[	1 .			1 1	1 0	0	0	1 :	1			2	1 1	12	1 1	07:30	21:15
Ā		Brighton			1	0	1 1	0	0	١٠	0	l ő	ľ	Ι.	_			l .	0	1 1	0	_				2	1 1	12	1	07:30	21:15
E		Brighton		1	3	1 (note 4)	11	2	1	0	0	l ĭ	ľ	05:30	21:30			0	1 1	15	0	05:15	21:30			1	1 ;	12	1 1	07:15	21:15
Ŧ		Brighton			3	2 (note 4)	19	5	2	0	0		n	05:30	21:30			3	2	30	0	05:15	21:30			5	2	24	2	07:15	21:15
vay		Brighton			2	0	1	1	1	0	0	l i	n	00.00	21.00			1 1	0	0	0	00.10	21.00			2	1	12	1	07:30	21:15
stv		Brighton		1	5	2 (note 4)	20	6	2	2	2	2	1	05:15	22:00			1 ;	2	30	1	05:15	22:00			5	2	24	2	07:15	21:15
og		Brighton			1	0	1	0	0	1 6	l 6	<u> </u>	, ,	03.13	22.00			1 7	1 6	0	l i	05.15	22.00			2	1	12	1	07:30	21:15
5		Brighton		<b> </b>	1	0	1	0	0	0	0	0	n	-				0	0	0	0	<del>-</del>	<del></del>	_		0	0	0	<u> </u>	-	-
Ves		Brighton			2	1 (note 4)	10	3	1	l ĭ	l ĭ	2	1	06:00	22:00			1 1	l ĭ	15	1 1	06:15	22:00			4	1 1	12	1	07:15	21:15
>		Brighton	78	82	2	1 (note 4)	10	3	1		l i	2		06:00	22:00	78	86	1	1 1	15	1 1	06:15	22:00	8	a as	3 4	1	12	1 1	07:15	21:15
		Brighton	I '°	32	2	1 (note 4)	10	2	1	i .	Ιi	2	1	06:00	22:00	l "	1 ~	T ;	Ιi	15	1 1	06:15	22:00	ı ~		1 4	Ιi	12	1 1	07:15	21:15
		Brighton	<b>1</b>	<b>i</b>	3	1 (note 4)	10	3	1	1	1	1	0	05:15	21:00			3	1 1	15	0	05:15	21:00	1	1	1	1	12	<del>   </del>	08:00	21:00
		Brighton	I	1	I 1	0 (note 4)	2	0	0	1	0	0	0	05:30	19:30	l		I 1	0	0	0	1 -	-	1		0	0	0	0	-	
		Brighton		1	3	1 (note 4)	10	3	1	l i	1	1	0	05:15	21:00			3	1 1	15	0	05:15	21:00			1	1	12	1	08:00	21:00
		Brighton		1	3	1 (note 3/4)	9 (note 3)	3	1	1	1	1	0	05:15	21:00			1 1	1 (note 3)	10 (note 3)	) 0	05:15	21:00			1	1	12	1 1	08:00	21:00
		Brighton			0	0 (note 4)	1	0	0	0	0	0	0		-			0	0	0	'l ŏ	-				0	0	0	0	-	
		Brighton			0	0 (note 4)	l i	0	0	lő	0	Ö	0					0	0	0	l ŏ	_				0	0	0	0	_	1 - '
		Brighton			1	0 (note 4)	1	0	0	0	0	٥	l ő					0	0	0	0	_				0	0	0	0	_	1 - '
		Brighton			0	0 (note 4)	l i	0	0	ő	ő	Ö	0	l -				0	0	0	0	_				0	0	0	0	- '	1 - '
		Brighton		1	1	0 (note 4)	1	0	0	0	0	0	0	l -				0	0	0	0					0	0	0	0	_	1 - '
		Brighton			0	0 (note 4)	1	0	0	0	0	٥	0					0	0	0	0	_				0	0	0	0	1 - '	1 - '
		Brighton		i	Ö	0 (11010 4)	1	ő	0	o o	0	0	0	-	-			2	ő	5	0	-	-	1	1	0	0	0	1 0	-	
		Brighton	I	1	0	0	1	o l	0	l ŏ	0	0	0					2	0	5	0	1 -		1	1	0	0	0	0	1 - '	1 - '
	, ,	Brighton	<b>†</b>	l	l ĭ	0 (note 4)	1	ő	0	ő	0	0	0	<u> </u>	-			0	Ö	0	0	1 -	-	1	1	0	0	0	0	-	
		Brighton	114	120	3	1 (note 4)	10	3	1	l ĭ	l ĭ	l ĭ	0	05:15	21:00	111	120	nl š	I i	15	0	05:15	21:00	10	7 109	1	I 1	12	I 1	08:00	21:00
1	outriampton central	Dirigition	114	120	٥	1 (11016 4)	10	3					U	03.13	21.00	- 111	120	ر ا		10	U	03.13	21.00	10	100	ווי		12	1 '	00.00	21.00

	TABLE 2: WESTBOUND	i			Monday t	o Friday			ı		Saturday			1		Sunday		
	Departure time from origin station		0200-0959	1000-1459	,	1900-0159	Services I	eaves LIT	0200-0659	0700-1859	,	Services	eaves LIT	0200-0859	0800-2159	,	Services	leaves LIT
	Direct services from specified origin to each station		Total	TPH	Total	Total	First (no later than)	Last (no earlier	Total	TPH	Total	First (no later than)	Last (no earlier	Total	TPH	Total	First (no later than)	Last (no earlier
	(excluding services from London)						,	than)			_	,	than)				,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	than)
	Ford	Littlehampton	8	2	9	_	05:45	22:30	2	2	5	06:00	22:30	4	1	1 1	06:30	23:00
		Littlehampton	8	2	9	/	05:45	22:30	3	2	5	06:00	22:30	5	1	1	06:30	23:00
	Bognor Regis	Littlehampton	3	1	3	9	06:15	22:30	3	1	4	07:15	22:30	3	0	1	07:30	23:00
		Barnham Littlehampton	5 (note 5)	1	5	3	05:45	22:00	3 (note 6)	1	3	06:00	22:00	2	0	0	-	
9		Littlehampton	5 (HOLE 5)	¦	6	1	05:45	18:45	3 (Hote 6)	1	0	06:00	18:45	2	0	0	-	-
D (p		Littlehampton	3	¦	6	0	05:45	18:45	2	¦ ¦	0	06:00	18:45	0	0	0	-	-
ğ.		Littlehampton	-	¦	6	0	05:45	18:45	2	¦ ¦	0	06:00	18:45	0	0	0	-	-
har		Littlehampton	5	¦	6	0	05:45	18:45	2	1	0	06:00	18:45	0	0	0	-	
Littleh		Littlehampton	5	;	6	0	05:45	18:45	2		0	06:00	18:45	1 1	0	0	_	-
5		Littlehampton	3	¦	6	0	05:45	18:45	2	1	0	06:00	18:45	, ,	0	0	-	
E C		Littlehampton	5 (note 5)	;	6	0	05:45	18:45	3 (note 6)	1	0	06:00	18:45	2	0	١٠		
ŧ		Littlehampton	5 (11016-3)	;	6	0	05:45	18:45	3 (11016-0)	1	0	06:00	18:45	0	0	١ ،		
va)	Hilsea	Littlehampton	5	1	5	0	05:45	18:45	2	1	0	06:00	18:45	0	0	0	-	-
astr		Littlehampton	5 (note 5)		6	0	05:45	18:45	3 (note 6)	1	0	06:00	18:45	Ĭ	0	0	-	
Coa		Littlehampton	5 (note 5)	l i	6	0	05:45	18:45	3 (note 6)	1	0	06:00	18:45	l i	0	ő	-	
st		Littlehampton	2 (note 5)	Ö	2	0	05:45	16:30	1 (note 6)	ó	0	-	-	1	0	0	-	
We		Littlehampton	0 (note 5)	0	0	0	-	-	0 (note 6)	0	0	-	-	1	0	0	-	-
		Littlehampton	0 (note 5)	0	0	0	-	-	0 (note 6)	0	0	-	-	0	0	0	-	-
	Fareham	Littlehampton	0 (note 5)	0	0	0	-	-	0 (note 6)	0	0	-	-	1	0	0	-	- 1
	Swanwick	Littlehampton	0 (note 5)	0	0	0	-	-	0 (note 6)	0	0	-	-	1	0	0	-	-
	Southampton Central	Littlehampton	0 (note 5)	0	0	0	-	-	0 (note 6)	0	0	-	-	1	0	0	-	-

Note 1: Table 1 includes services to West Coastway stations from Brighton only. All Littlehampton services serving West Coastway stations are specified in Table 2.

Note 2: Shuttle services operating between Littlehampton, Barnham and Bognor Regis are not included in the Table 1. The quantum of services between these stations is specified in Table 2.

Note 3: At times when trains are specified to call at Eastleigh and Southampton Airport Parkway, there shall be no requirement to serve Swanwick.

Note 4: An additional service shall be provided to Southampton Central, calling at Hove, Portamouth Parbour, calling at Fratton and Portsmouth As Southsea, arriving before 09:15.

Note 5: Three additional services shall be provided from Barnham with a maximum interval of 20 minutes, calling at Chichester and Havant, departing between 04:45 and 05:45.

Two of these services shall extend to Portsmouth Harbour, calling at Fratton and Portsmouth & Southsea. One shall extend to Southampton Central, calling at Coisham, Portchester and Havant, departing between 04:45 and 05:45.

Three of these services shall extend to Portsmouth Harbour, calling at Fratton and Portsmouth & Southsea. One of these may terminate at Portsmouth & Southsea.

Two shall extend to Southampton Central, calling at Coisham, Portchester, Fareham, Swanwick, Netley and Woolston.

Note 6: Three of these services shall extend to Portsmouth Harbour, calling at Fratton and Portsmouth & Southsea. One of these may terminate at Portsmouth & Southsea.

Two shall extend to Southampton Central, calling at Coisham, Portchester, Fareham, Swanwick, Netley and Woolston.

Note 6: Three dolitional services shall extend to Portsmouth Harbour, calling at Fratton and Portsmouth & Southsea.

The services shall extend to Southampton Central, calling at Coisham, Portchester, Fareham, Swanwick, Netley and Woolston.

Note 6: Three addit

	TABLE 3: EASTBOUND							Monday t	o Friday						ı .			Satu	ırdav				T			Sun	dav			
	Arrival time at destination station	1	Journe	ey Time	0200-0659	0700-0959	0800-0859			2000-2059	2100-2159	2200-015	Service arriv	es Brighton	.lourn	ey Time	0200-0659			2200-0159	Service arriv	ves Brighto	n Journ	ev Time	0200-0859	0900-2059		2100-0159	Service arr	ives Brighto
	/ invariance at acountation station		oodiii	I	0200 0000	0700 0303	0000 0000	1000 1303	1000 1000	2000 2000	2100 2100	2200 010	ocivice aim	Last (no	Journ	I I	0200 0000	0700 2100	0700 £100	2200 0100	OCIVICE and	Last (no	Joann	I	0200 0000	0300 2003	0300 2003	2100 0100	DCI VICE AIT	Last (no
	Direct services from each station to specified destination	.l		1								1	First (no	earlier	l						First (no	earlier	1						First (no	earlier
	(excluding services to London)	Destination	Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	Southampton Central		108	125	0	10tai	1 Utai			1 Utai	1 Otal	10(a)			Average 108			1511		10(a)	, , ,		Average 112		0				, ,	23:15
		Brighton	106	125	0	2		1	10			2	08:00	23:15	100	125	0		15	2	08:00	23:15	112	119	0	1 0	12	3	09:30	23.15
	St. Denys	Brighton		<u> </u>	0	0	0	0	0	0	0	0					0	0	0	0		-			0	U	U	2	<u> </u>	<u> </u>
	Southampton Airport Parkway	Brighton		1	0	0	0	0		0	0	0	-	-	l		0	0		0	-	-	1		0	0	1	2	-	-
	Eastleigh	Brighton		<u> </u>	v	0	0	0	1	0	0	0		-			0	0	1	0	-	-			0	·	1			-
	Bitterne	Brighton		1	0	0	0	0	0	0	0	0	-	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Woolston	Brighton		1	0	1	1	0	0	0	0	0	1 - 1	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Sholing	Brighton		1	0	0	0	0	0	0	0	0	-	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Netley	Brighton		1	0	0	0	0	0	0	0	0	-	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Hamble	Brighton		1	0	0	0	0	0	0	0	0	-	-	l		0	0	0	0	-	-	1		0	0	0	0	-	-
	Bursledon	Brighton		1	0	0	0	0	0	0	0	0	-	-	l		0	0	0	0	-		1		0	0	0	0	-	-
	Swanwick	Brighton	I	I	0	2	1 1	1 (note 3)	9 (note 3)	1 1	1 1	2	08:00	23:15	I	1	0	1 (note 3)	14 (note 3)	2	08:00	23:15	I		0	1 (note 3)	11 (note 3)	1	09:30	23:15
	Fareham	Brighton	l	1	0	2	1 !	1	10	1	1	2	08:00	23:15			0	1	15	2	08:00	23:15	1		0	1	12	3	09:30	23:15
	Portchester	Brighton		1	0	1	1	0	0	0	0	1	-	-	l		0	0	0	1	-	-	1		0	0	0	0	-	-
	Cosham	Brighton			0	2	1	1	10	1	1	2	08:00	23:15			0	1	15	2	08:00	23:15			0	1	12	3	09:30	23:15
	Portsmouth Harbour	Brighton		1	0	4	1	1 (note 4)	11	0	0	2	07:15	00:15	l		0	1	15	3	08:00	00:15	1		1	1	12	3	09:00	23:30
	Portsmouth & Southsea	Brighton	79	97	0	4	1	1 (note 4)	11	1	1	3	07:15	00:15	78	97	0	1	15	3	08:00	00:15	95	104	1	1	12	4	09:00	23:30
Ė	Fratton	Brighton		1	0	4	1	1 (note 4)	11	1	1	3	07:15	00:15	l		0	1	15	3	08:00	00:15	1		1	1	12	4	09:00	23:30
₽	Hilsea	Brighton			0	1	0	0 (note 4)	1	0	0	3	-	-			0	0	0	3	-	-	ļ		0	0	0	0	-	-
6	Bedhampton	Brighton		1	0	1	0	0	0	0	0	3	-	-	l		0	0	0	3	-	-	1		1	1	12	2	09:00	22:45
Ę	Havant	Brighton		1	0	6	2	2 (note 4)	21	2	2	5	07:15	00:15	l		0	2	30	5	08:00	00:15	1		1	2	24	7	09:00	23:30
. <u>E</u>	Warblington	Brighton		1	0	1	0	0 (note 4)	1	0	0	1	- 1	-	l		0	0	0	1	-	-	1		1	1	12	2	09:00	22:45
2	Emsworth	Brighton		1	0	4	1	2 (note 4)	21	1	2	3	07:15	23:45	l		0	2	30	3	08:00	23:45	1		1	2	24	7	09:00	23:30
· 🚖	Southbourne	Brighton		1	0	4	1	1 (note 4)	11	1	1	2	07:15	23:45	l		0	1	15	2	08:00	23:45	1		1	1	12	4	09:00	23:30
ž	Nutbourne	Brighton		1	0	1	0	0 (note 4)	1	0	0	1	-	-	l		0	0	0	1	-	-	1		1	1	12	2	09:00	22:45
as	Bosham	Brighton		1	0	2	0	0 (note 4)	1	0	0	1	- 1	-	l		0	0	0	1	-	-	1		1	1	12	2	09:00	22:45
റ്	Fishbourne	Brighton		1	0	2	0	0 (note 4)	1	0	0	1	-	-	l		0	0	0	1	-	-	1		1	1	12	2	09:00	22:45
est	Chichester	Brighton			0	6	2	2 (note 4)	21	2	2	5	07:15	00:15			0	2	30	5	07:30	00:15			1	2	24	7	09:00	23:30
š	Bognor Regis	Brighton			2	1	0	0	0	0	0	0	-	-			1	0	0	0	-	-			0	0	0	0	-	-
	Barnham	Brighton	I	I	3	7	2	2 (note 4)	21	2	2	5	06:00	00:15	I	1	2	2	30	5	06:00	00:15	I		2	2	24	7	07:00	23:30
	Ford	Brighton	<u> </u>		3	6	2	1	10	0	0	5	06:00	00:15			2	1	15	5	06:00	00:15			1	2	24	7	09:00	23:30
	Littlehampton	Brighton	<u> </u>		1	3	3	0	2	1	1	3	06:00	00:15			1	0	3	3	06:00	00:15	<u> </u>		2	0	1	3	07:00	23:30
	Angmering	Brighton	I	I	3	10	5	2	22	3	3	6	06:00	00:15	I		2	2	30	6	06:00	00:15	I		2	2	24	7	08:30	23:30
	Goring-by-Sea	Brighton	I	1	3	7	4	1	11	2	2	6	06:00	00:15	I		2	1	15	6	06:00	00:15	1		2	2	24	7	08:30	23:30
	Durrington-on-Sea	Brighton	I	I	3	9	5	1	12	2	2	6	06:00	00:15	I		2	1	15	6	06:00	00:15	I		2	2	24	7	08:30	23:30
	West Worthing	Brighton	I	I	3	9	4	3	30	2	2	6	06:00	00:15	I		2	3	45	6	06:00	00:15	I		2	2	24	7	08:30	23:30
	Worthing	Brighton	l	1	3	12	5	4 (note 4)	41	3	3	6	06:00	00:15			2	4	60	6	06:00	00:15	1		3	2	24	7	07:00	23:30
	East Worthing	Brighton	I	I	3	7	3	2 (note 4)	21	3	3	6	06:00	00:15	I	1	2	2	30	6	06:00	00:15	I		2	2	24	7	08:30	23:30
	Lancing	Brighton	l	1	3	12	5	4 (note 4)	41	3	3	6	06:00	00:15			2	4	60	6	06:00	00:15	1		3	2	24	7	07:00	23:30
	Shoreham-by-Sea	Brighton	I	I	3	12	5	4 (note 4)	41	3	3	6	06:00	00:15	I		2	4	60	6	06:00	00:15	I		3	2	24	7	07:00	23:30
	Southwick	Brighton	I	I	3	12	5	4 (note 4)	41	3	3	6	06:00	00:15	I		2	4	60	6	06:00	00:15	I		3	2	24	7	07:00	23:30
	Fishersgate	Brighton	l	1	3	7	3	2 (note 4)	21	3	3	6	06:00	00:15			2	2	30	6	06:00	00:15	1		2	2	24	7	08:30	23:30
	Portslade	Brighton	l	1	3	12	5	4 (note 4)	41	3	3	6	06:00	00:15			2	4	60	6	06:00	00:15	1		3	2	24	7	07:00	23:30
	Aldrington	Brighton	I	I	3	7	3	2 (note 4)	21	3	3	6	06:00	00:15	I		2	2	30	6	06:00	00:15	I		2	2	24	7	08:30	23:30
	Hove	Brighton	I	I	5	18	7	6 (note 4)	61	5	5	10	06:00	00:15	I	1	5	6	90	10	06:00	00:15	I	1	4	3	36	8	07:00	23:30

	TABLE 4: EASTBOUND			•	Monday 1	o Friday					Saturday				•	Sunday		
	Arrival time at destination station		0200-0959	1000-1559	1600-1959	2000-0159	Services a	arrives LIT	0200-0659	0700-1759	1800-0159	Services a	arrives LIT	0200-0659	0700-2159	2200-0159	Services	arrives LIT
	Direct services from each station to specified destination (excluding services to London)		Total	TPH	Total	Total	First (no later than)	Last (no earlier than)	Total	TPH	Total	First (no later than)	Last (no earlier than)	Total	TPH	Total	First (no later than)	Last (no earlier than)
	Southampton Central	Littlehampton	0	0	0	2 (note 5)	-	-	0	0	2	-	-	0	0	2	-	-
	Swanwick	Littlehampton	0	0	0	2	-	-	0	0	2	-	-	0	0	2	-	-
	Fareham	Littlehampton	0	0	0	2 (note 5)	-	-	0	0	2	-	-	0	0	2	-	-
	Portchester	Littlehampton	0	0	0	1	-	-	0	0	1	-	-	0	0	0	-	-
نے :	Cosham	Littlehampton	0	0	0	2 (note 5)	-	-	0	0	2	-	-	0	0	2	-	-
from.	Portsmouth Harbour	Littlehampton	1	0	1	3	09:45	00:15	0	0	4	-	-	0	0	2	-	-
	Portsmouth & Southsea	Littlehampton	2	1	5	4	08:45	00:15	1	1	6	07:00	23:59	0	0	2	-	-
btc	Fratton	Littlehampton	2	1	5	4	08:45	00:15	1	1	6	07:00	23:59	0	0	2	-	-
Littlehampton	Hilsea	Littlehampton	2	1	5	2	08:45	00:15	1	1	4	07:00	23:59	0	0	0	-	-
<del>P</del>	Bedhampton	Littlehampton	2	1	3	2	08:45	00:15	1	1	3	07:00	23:59	0	0	1	-	-
耳	Havant	Littlehampton	2	1	5	6 (note 5)	08:45	00:15	1	1	8	07:00	23:59	0	0	4	-	-
9	Warblington	Littlehampton	2	1	5	4	08:45	00:15	1	1	4	07:00	23:59	0	0	1	-	-
<u>\$</u>	Emsworth	Littlehampton	2	1	5	5	08:45	00:15	1	1	7	07:00	23:59	0	0	4	-	-
ž	Southbourne	Littlehampton	2	1	5	5	08:45	00:15	1	1	7	07:00	23:59	0	0	2	-	-
Coas	Nutbourne	Littlehampton	2	1	5	4	08:45	00:15	1	1	4	07:00	23:59	0	0	1	-	-
ŏ	Bosham	Littlehampton	1	1	5	4	09:45	00:15	1	1	4	07:00	23:59	0	0	1	-	-
est	Fishbourne	Littlehampton	1	1	5	4	09:45	00:15	1	1	4	07:00	23:59	0	0	1	-	-
We	Chichester	Littlehampton	3	1	5	7 (note 5)	08:30	00:15	1	1	9	07:00	23:59	0	0	4	-	┷
	Bognor Regis	Littlehampton	6	1	3	3	05:45	22:45	2	1	5	05:45	22:45	0	1	1	08:00	22:15
	Bognor Regis (additional shuttle to Barnham)	Barnham	5	1	4	3			2	1	6	-		0	0	1		
	Barnham	Littlehampton	9	2	8	10	05:45	00:15	3	2	14	05:45	23:59	0	1 1	6	08:00	23:59
	Ford	Littlehampton	9	2	8	8	05:45	00:15	3	2	14	05:45	23:59	0	1	5	08:00	23:45

Note 1: Table 3 includes services from West Coastway stations to Brighton only. All Littlehampton services serving West Coastway stations are specified in Table 4.

Note 2: Shuttle services operating between Bognor Regis, Barnham and Littlehampton are not included in Table 3. The quantum of services from these stations are specified in Table 4.

Note 3: At times when trains are specified to call at Eastleigh and Southampton Airport Parkway, there shall be no requirement to serve Swanwick.

Note 4: An additional service shall be provided from Portsmouth Harbour, departing after 16:30.

Note 5: One additional service shall be provided from Southampton Central to Barnham, calling at St Denys, Fareham, Cosham, Havant and Chichester, departing after 23:00.

#### East Coastway: Minimum Service Specification and First and Last Train Specification

	TABLE 1: EASTBOUND							Mo	onday to Frida	ay										Satu	ırday			I			Sund	day			$\overline{}$
	Departure time from origin station		Journ	ey Time	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service lea	ves Brighton	Journe	y Time	0200-0659	0700-2159	0700-2159	2200-0159	Service lear	ves Brighton	Journe	ey Time	0200-0859	0900-2059	0900-2059	2100-0159	Service leav	ves Brighton
	Direct services from specified origin to each station													First (no	Last (no earlier							First (no	Last (no earlier							First (no	Last (no earlier
	(excluding services from London)	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	London Road (Brighton)	Brighton			5	4	39	14	6	4	2	2	4	05:45	23:30			4	4	60	4	06:00	23:30			7	2	24	6	07:15	23:30
	Moulsecoomb	Brighton			5	4	39	14	6	4	2	2	4	05:45	23:30			4	4	60	4	06:00	23:30	l		7	2	24	6	07:15	23:30
	Falmer	Brighton		1	5	4	39	14	6	4	2	2	4	05:45	23:30			4	4	60	4	06:00	23:30	l		7	2	24	6	07:15	23:30
	Lewes	Brighton		1	7	5	49	17	7	5	3	3	6	05:15	23:30			6	5	75	6	05:15	23:30	l		8	3	36	9	07:15	23:30
	Southease	Brighton			0	1	9	4	2	2	1	1	1	08:00	23:00			0	1	15	0	08:00	21:00			2	1	12	0	08:00	20:00
	Newhaven Town	Brighton		1	3	2	18	7	2	3	2	2	3	05:45	23:30			3	2	30	3	06:00	23:30	l		4	2	24	4	07:15	22:30
	Newhaven Harbour (note 1)	Brighton			3	2	18	5	2	3	2	2	3	05:45	23:30			3	2	30	3	06:00	23:30	l		4	2	24	4	07:15	22:30
	Bishopstone	Brighton		1	3	2	18	7	2	2	2	2	3	05:45	23:30			2	2	30	3	06:15	23:30	l		4	2	24	4	07:15	22:30
	Seaford	Brighton	36	42	3	2	18	7	2	2	2	2	3	05:45	23:30	36	40	3	2	30	3	06:00	23:30	35	41	4	2	24	4	07:15	22:30
	Glynde	Brighton			1	1	10	2	1	1	0	0	1	06:30	23:00			1	1	15	1	07:00	23:00			2	1	12	3	07:15	23:15
	Berwick	Brighton			1	1	10	3	2	1	0	0	1	06:30	23:00			1	1	15	1	07:00	23:00	l		2	1	12	3	07:15	23:15
	Polegate	Brighton		1	3	2	19	7	3	2	1	1	3	05:15	23:15			3	2	30	3	05:15	23:15	l		4	1	12	3	07:15	23:15
욛	Hampden Park	Brighton			1	1	10	3	2	1	0	0	1	06:30	23:00			1	1	15	1	07:00	23:00	l		4	1	12	3	07:15	23:15
. <u>D</u>	Eastbourne	Brighton	37	42	3 (note 2)	2 (note 4)	19	7	3	2	1	1	3	05:15	23:15	37	42	3 (note 7)	2	30	3	05:15	23:15	38	41	4 (note 9)	1	12	3	07:15	23:15
ω.		of which	32	34	2	1	9	3	1	1	1	1	2			33	35	1	1	15	2			N/A	N/A				A 7	i l	i
ě	Pevensey & Westham	Brighton			0 (note 2)	1 (note 4)	10	3	1	0	0	0	0	07:45	18:45			1 (note 7)	1	15	0	07:00	21:00			2 (note 9)	0	0	0	- 1	-
_ <del>_</del>	Pevensey Bay	Brighton			0	0 (note 4)	0	1	0	0	0	0	0	-	- 1			0 (note 7)	0	0	0	-	-	l		0	0	0	0	i - I	i - I
, ka	Normans Bay	Brighton		1	0 (note 2)	1 (note 4)	9	2	1	0	0	0	0	08:00	17:45			1 (note 7)	1	15	0	07:00	21:00	l		0	0	0	0	i - I	i - I
astr	Cooden Beach	Brighton		1	0 (note 2)	1 (note 4)	9	2	1	0	0	0	0	08:00	17:45			1 (note 7)	1	15	0	07:00	21:00	l		2 (note 9)	0	0	0	i - I	i - I
ő	Collington	Brighton			0 (note 2)	1 (note 4)	9	2	1	0	0	0	0	08:00	17:45			1 (note 7)	1	15	0	07:00	21:00	l		2 (note 9)	0	0	0	i - I	i - I
, to	Bexhill	Brighton		1	1 (note 2)	2 (note 4)	18	5	2	1	1	1	0	06:30	21:30			2 (note 7)	2	30	1	06:45	22:15	l		3 (note 9)	1	12	1	07:45	21:15
ä	St. Leonards Warrior Square	Brighton		1	1 (note 2)	2 (note 4)	18	5	2	1	1	1	0	06:30	21:30			2 (note 7)	2	30	1	06:45	22:15	l		3 (note 9)	1	12	1	07:45	21:15
-	Hastings	Brighton	71	82	1 (note 2)	2 (note 4)	18	5	2 (note 6)	1	1	1	0	06:30	21:30	71	80	2 (note 7)	2	30	1	06:45	22:15	67	73	3 (note 9)	1	12	1	07:45	21:15
	_	of which	64	64	1	1	9	3	1	1	1	1	0			63	66	1	1	15	1			N/A	N/A				1	1	ı
	Ore	Brighton			0 (note 3)	1 (note 4)	9	2	1 (note 6)	0	1	1	0	08:00	21:00			1 (note 8)	1	15	0	07:00	21:00			2 (note 9)	0	0	0	- 1	-
1	Three Oaks	Brighton		I	1 (note 3)	0	5	1	1 (note 6)	1	0	1	0	06:30	21:00			0 (note 8)	0	8	0	07:45	21:00	I	1	0 (note 9)	0	0	0	ı - I	1 - I
	Doleham	Brighton	I	1	0 (note 3)	0	0	0	0 (note 6)	0	0	1	0	-	-			0 (note 8)	0	1	0	-	-	ı	1	0 (note 9)	0	0	0	1	1 - I
	Winchelsea	Brighton	I	1	1 (note 3)	0	4	2	0 (note 6)	0	0	1	0	06:30	21:00			1 (note 8)	0	7	0	06:45	21:00	ı	1	0 (note 9)	0	0	0	1	1 - I
	Rye	Brighton	1	1	1 (note 3)	1 (note 5)	9	3	1 (note 6)	1 (note 6)	1	1	0	06:30	21:00			1 (note 8)	1	15	0	06:45	21:00	I	1	1 (note 9)	1	12	0	08:30	20:15
1	Appledore	Brighton	I	1	1 (note 3)	1 (note 5)	9	3	1 (note 6)	1 (note 6)	1	1	0	06:30	21:00			1 (note 8)	1	15	0	06:45	21:00	ı	1	1 (note 9)	1	12	0	08:30	20:15
1	Ham Street	Brighton	1	1	1 (note 3)	1 (note 5)	9	3	1 (note 6)	1 (note 6)	1	1	0	06:30	21:00			1 (note 8)	1	15	0	06:45	21:00	I	1	1 (note 9)	1	12	0	08:30	20:15
	Ashford International	Brighton	107	108	1 (note 3)	1 (note 5)	9	3	1 (note 6)	1 (note 6)	1	1	0	06:30	21:00	106	106	1 (note 8)	1	15	0	06:45	21:00	108	108	1 (note 9)	1	12	0	08:30	20:15

Note 1: The parliamentary service to Newhaven Marine must continue to operate unless and until that station is formally closed.

Note 2: Three additional services shall be provided between Eastbourne and Hastings, calling at Pevensey & Westham, Beshill and St Leonards Warrior Square, departing Eastbourne between 05:00 and 06:45. One service shall call additionally at Normans Bay, Cooden Beach and Collington.

Note 3: Three additional services shall be provided at half hourly intervals between Hastings and Ashford International, calling at Rey, Appledore and Ham Street, departing Eastbourne between 07:00 and 06:30. One of these trains shall call at limitermediate stations. Two services shall be provided between Eastbourne and Hastings, departing Eastbourne between 07:00 and 07:40. One service shall be extended to Ore and call at all all intermediate stations. Services shall be provided between Rye and Ashford International calling at all intermediate stations.

Note 6: Two additional services shall be provided between Rye and Ashford International calling at all intermediate stations.

Note 7: Four additional services shall be provided between Eastbourne and Ore calling at Bexhill, St Leonards Warrior Square and Hastings, departing Eastbourne between 05:00 and 07:00. One service shall call at all intermediate stations.

Note 7: Four additional services shall be provided between Eastbourne and Ore calling at Bexhill, St Leonards Warrior Square and Hastings, departing Eastbourne no later than 06:00.

Note 9: One additional service shall be provided between Hastings and Ashford International, calling at all intermediate stations, departing Eastbourne no later than 06:00.

Note 9: One additional service shall be provided between Hastings and Ashford International, calling at all intermediate stations, departing Eastbourne no later than 06:00.

Note 9: One additional service shall be provided between Eastbourne and Ashford International calling at all intermediate stations, departing Eastbourne no later than 06:00.

	TABLE 2: WESTBOUND							Monday	to Friday									Sat	urday							Sund	day			
	Arrival time at destination station		Journ	ney Time	0200-0659	9 0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-0159	Service arr	ives Brighton	Journe	ey Time	0200-0659	0700-2159	0700-2159	2200-0159	Service arri	ives Brighton	) Journe	ey Time	0200-0859	0900-2059	0900-205	9 2100-0159	Service arri	ves Brightor
														Last (no								Last (no								Last (no
	Direct services from each station to specified destination		1	1					1			1	First (no	earlier					l	1	First (no	earlier	1						First (no	earlier
	(excluding services to London)	Destination	Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
	Ashford International	Brighton	108	108	0	2 (note 3)	1	1	10 (note 4)	1	1	2 (note 5)	08:30	23:15	107	107	0	1	15	2 (note 5)	08:00	23:15	111	116	0	1	12	3 (note 5)	10:00	23:00
	Ham Street	Brighton	1		0	2 (note 3)	1	1 1	10 (note 4)	1	1 1	2 (note 5)	08:30	23:15			0	1 1	15	2 (note 5)	08:00	23:15	1	l	0	1	12	3 (note 5)	10:00	23:00
	Appledore	Brighton	1		0	2 (note 3)	1	1	10 (note 4)	1	1	2 (note 5)	08:30	23:15			0	1	15	2 (note 5)	08:00	23:15	1	l	0	1	12	3 (note 5)	10:00	23:00
	Rye	Brighton	1		0	2 (note 3)	1	1	10 (note 4)	1	1	2 (note 5)	08:30	23:15			0	1	15	2 (note 5)	08:00	23:15	1	l	0	1	12	3 (note 5)	10:00	23:00
	Winchelsea	Brighton	1		0	2 (note 3)	1	0	5	0	1	1 (note 5)	08:30	23:15			0	0	8	1 (note 5)	08:00	23:15	1	l	0	0	1	0 (note 5)	1 - '	1 - '
	Doleham	Brighton	1		0	2 (note 3)	1	0	0	0	0	0 (note 5)	-	-			0	0	1	0 (note 5)	-	-	1	1	0	0	1	0 (note 5)	- 1	1 - '
	Three Oaks	Brighton	I	1	0	2 (note 3)	1	0	5	0	1 1	1 (note 5)	08:30	22:15		1	0	0	7	1 (note 5)	08:00	22:15	ı		0	0	1 1	0 (note 5)		
	Ore	Brighton	1		0	3 (note 3)	1	1 1	10	0	1 1	1 (note 5)	08:30	23:45		l	0	1	16	2 (note 5)	07:45	23:45	l	l	0	0	1	2 (note 5)	10:00	23:30
	Hastings	Brighton	73	81	1 (note 2)	4	1	2	20	2	2	4 (note 5)	07:30	00:30	73	80	0	2	30	5 (note 5)	07:45	00:30	70	77	0	1	12	5 (note 6)	10:00	23:30
		of which	67	72	0	2	1	1	10	1	1	3			66	66	0	1	15	2			N/A	N/A					1 '	1 '
Ė	St. Leonards Warrior Square	Brighton	1		1 (note 2)	4	1	2	20	2	2	4	07:30	00:30			0	2	30	5	07:45	00:30	1	l	0	1	12	5 (note 6)	10:00	23:30
£	Bexhill	Brighton	1		1 (note 2)	4	1	2	20	2	2	4	07:30	00:30			0	2	30	5	07:45	00:30	1	1	0	1	12	5 (note 6)	10:00	23:30
5	Collington	Brighton	1		1 (note 2)	1	0	1	10	1	1	2	09:45	00:30			0	1	15	3	07:45	00:30	1	l	0	0	0	2 (note 6)	1 - '	1 - '
Ĕ	Cooden Beach	Brighton	1		1 (note 2)	1 1	0	1 1	10	1	1	2	09:45	00:30			0	1 1	15	3	07:45	00:30	1	l	0	0	0	2 (note 6)	1 - '	1 - '
, je	Normans Bay	Brighton	1		0	1 1	0	1	10	0	0	1	09:45	23:45			0	1	15	1	07:45	23:45	1	l	0	0	0	0	1 - '	1 - '
2	Pevensey Bay	Brighton	1		0	0	0	0	1 1	0	0	0	l <sup>-</sup>	l			0	0	0	0		l	1	l	0	0	0	0	1 - '	1 - '
· -	Pevensey & Westham	Brighton	l		1 (note 2)	1 1	0	1	10	1	1	2	09:45	00:30		l	0	1	15	3	07:45	00:30			0	0	0	2 (note 6)		
ž	Eastbourne	Brighton	38	43	1	6	2	2	20	2	2	5	06:15	00:30	38	41	1	2	30	5	06:30	00:30	39	43	1	1	12	5 (note 6)	08:15	23:30
as		of which	. 34	35	0	2	1	1	10	1	1	2			34	35	0	1	15	2			N/A	N/A					1	1 '
8	Hampden Park	Brighton	1		1	4	1	1	11	1	2	3	06:15	00:30			1	1	15	3	06:30	00:30	1	l	1	1	12	5	08:15	23:30
ast	Polegate	Brighton	1		1	6	2	2	20	2	2	5	06:15	00:30			1	2	30	5	06:30	00:30	1	l	1 1	1 1	12	5	08:15	23:30
шĭ	Berwick	Brighton	1		0	3	1	1 !	10	1	1 !	2	07:30 07:30	23:45			0	1 !	15	2	07:45	23:45	1	l	1 !	1 !	12	3	08:15	23:00
	Glynde	Brighton	0.5		0	3	1	1	10	1	1	2		23:45			0	1	15	2	07:45	23:45			1	1	12	3	08:15	23:00
	Seaford	Brighton	35	41	2	5	2	2	22	2	2	5	06:00	23:45	34	41	1	1 2	30	5	05:45	23:45	37	41	1 1	2	24	6	08:45	23:15
	Bishopstone	Brighton	I	1	1 2	5	2	2	22	2	2	5	06:00	23:45		I	1	1 2	30	5	05:45	23:45	I	1	1 1	2	24	6	08:45	23:15
	Newhaven Harbour (note 1)	Brighton	I	1	2	5	2	2	20	2	2	5	06:00	23:45		1	1	1 2	30	5	05:45	23:45	ı		1 !	2	24	6	08:45	23:15
	Newhaven Town	Brighton	I	1	2	6	3	2	22 10	2	2	5	06:00	23:45 21:00		1	1 1	1 2	30 15	5	05:45 08:00	23:45 21:00	ı		1 1	2	24 12	6	08:45	23:15 20:30
	Southease	Brighton		+	0	3	2	1		1	1	- 0	08:00				0	1	15	0				-	1 1	1	12	0	08:45	
	Lewes	Brighton	I	1	1 4	16	7	5	52	4	4	10	06:00	00:30		1	2	5	75	10	05:45	00:30	ı		3	3	36	11	07:45	23:30
	Falmer	Brighton	I	1	1 4	14	6	4	42 42	3	3	1 8	06:00	00:30		1	2	1 4	60	8	05:45	00:30	ı		3	2	24	8	07:45	23:30
	Moulsecoomb	Brighton	I	1	4	14	6	4	42	3	3	8	06:00	00:30		I	2	4	60	8	05:45	00:30	I	1	3	2	24	8	07:45	23:30
	London Road (Brighton)	Brighton			4	14	6	4	42	3	3	8	06:00	00:30		l	2	4	60	8	05:45	00:30			3	2	24	8	07:45	23:30

Note 1: The parliamentary service to Newhaven Marine must continue to operate unless and until that station is formally closed.

Note 2: One additional service shall be provided departing Hastings no later than 05:45. This service may terminate at Eastbourne.

Note 3: Three additional services shall be provided between Ashford International and Rye calling at all intermediate stations, departing Ashford International between 06:30 and 09:30. One service shall be extended to Hastings calling at all intermediate stations.

Note 4: Three additional services shall be provided between Ashford International and Rye calling at all intermediate stations, departing Ashford International between 17:30 and 20:30. One service shall be extended to Hastings.

Note 5: One additional service shall be provided between Ashford International and Hastings calling at all intermediate stations, departing Ashford International no earlier than 22:30

Note 6: One additional service shall be provided between Hastings and Eastbourne calling at all intermediate stations (except Normans Bay and Pevensey Bay), departing Hastings no earlier than 23:15

# Milton Keynes-Croydon: Minimum Service Specification and First and Last Train Specification

	TABLE 1: NORTHBOUND							Monday t	to Friday									Satu	ırday							Sun	nday			
	Arrival time at Kensington (Olympia)	)	Journe	ey Time	0200-0659	0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-0159	Service a	rrives KPA	Journe	y Time	0200-0659	0700-2159	0700-2159	2200-0159	Service a	rrives KPA	Journe	ey Time	0200-0859	0900-2059	0900-2059	2100-0159	Service a	
														Last (no								Last (no								Last (no
	Direct services from each station to specified destination									l			First (no	earlier	_						First (no	earlier	١.	l					First (no	earlier
	(excluding services to London terminals)		Average	Maximum	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
		Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45			0	0	0	0	-	-
		Shepherd's Bush	44	46	0	3	2	1	10	0	0	0	-	-	43	43	1	1	13	0	-	-	N/A	N/A	0	0	0	0	-	-
:		Watford Junction	72	74	0	3	2	1	10	0	0	0	-	-	71	71	1	1	13	0	-	-	N/A	N/A	0	0	0	0	-	-
E		Milton Keynes	112	114	0	2	1	1	10	0	0	0	-	-	110	110	1	1	11	0	-	-	N/A	N/A	0	0	0	0	-	-
- €	Selhurst	Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45	l		0	0	0	0	-	-
<u>[a]</u>	Thornton Heath	Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45	l		0	0	0	0	-	-
Ĕ	Norbury	Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45	l		0	0	0	0	-	-
र्त	Streatham Common	Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45	l		0	0	0	0	-	-
2		Kensington Olympia			1	4	2	1	10	0	0	0	05:45	19:45			2	1	15	0	06:00	19:45	l		0	0	0	0	-	-
25	Wandsworth Common	Kensington Olympia			0	4	2	1	10	0	0	0	08:30	19:45			1	1	15	0	07:00	19:45	l		0	0	0	0	-	-
ı <u>ë</u>	Clapham Junction	Kensington Olympia			5	8	3	1	12	1	1	1	05:15	22:45			4	1	15	1	05:30	22:45	l		2	1	12	2	07:45	22:15
E S		Shepherd's Bush	14	18	5	8	3	1	12	1	1	1	-	-	14	14	4	1	15	1	-	-	13	16	2	1	12	2	-	-
×		Watford Junction	43	45	5	4	2	1	12	1	1	1	-	-	42	45	4	1	15	1	-	-	40	43	1	1	12	2	-	-
P		Milton Keynes	83	86	2	3	1	1	12	1	1	1	-	-	82	84	2	1	11	1	-	-	N/A	N/A	0	0	0	0	-	-
	Imperial Wharf	Kensington Olympia			5	8	3	1	12	1	1	1	05:15	22:45			4	1	15	1	05:30	22:45			2	1	12	2	07:45	22:15
		Kensington Olympia			5	8	3	1	12	1	1	1	05:15	22:45			4	1	15	1	05:30	22:45			2	1	12	2	07:45	22:15

	TABLE 2: NORTHBOUND						Mo	onday to Frid	lay							Satu	ırday					Sun	day		
	Departure time from Kensington (Olympia)		0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service le	aves KPA	0200-0659	0700-2159	0700-2159	2200-0159	Service le	aves KPA	0200-0859	0900-2059	0900-2059	2100-0159	9 Service le	aves KPA
													Last (no						Last (no						Last (no
	Direct services from specified origin to each station						l					First (no	earlier	l				First (no	earlier					First (no	earlier
	(excluding services from London terminals)	Origin	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Total	TPH	Total	Total	later than)	than)	Total	TPH	Total	Total	later than)	than)
	Shepherd's Bush	Kensington Olympia	5	1	14	4	2	1	1	1	1	05:15	22:45	4	1.00	15	1	05:30	22:45	2	1	12	2	07:45	22:15
	Wembley Central	Kensington Olympia	4	1	10	3	1	1	1	1	0	05:45	20:45	3	1.00	15	0	06:00	18:45	0	0	0	0	- 1	1 - I
Ę :	Harrow & Wealdstone	Kensington Olympia	4	1	10	3	1	1	1	1	1	05:15	22:45	4	1.00	15	1	05:30	22:45	1	1	12	2	08:30	22:15
p o	Watford Junction	Kensington Olympia	5	1	10	3	1	1	1	1	1	05:15	22:45	4	1.00	15	1	05:30	22:45	1	1	12	2	08:30	22:15
nsi ia)	Hemel Hempstead	Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0	- /	1 - I
중 문	Berkhamsted	Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0	- 1	1 - I
₽ 중	Tring	Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0	- 1	1 - I
60	Leighton Buzzard	Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0	- 1	1 - I
		Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0	1 - 1	1 - I
	Milton Keynes	Kensington Olympia	2	1	9	3	1	1	1	1	0	05:45	20:45	2	1.00	15	0	06:00	17:45	0	0	0	0		-

	TABLE 3: SOUTHBOUND						Monday	to Friday							Satu	urday			1		Sun	day		$\overline{}$
	Arrival time at Kensington (Olympia)		0200-0659	0700-0959	0800-0859	1000-1959	1000-1959	2000-2059	2100-2159	2200-0159	Service ar	rives KPA	0200-0659	0700-2159	0700-2159	2200-0159	Service ar	rrives KPA	0200-0859	0900-2059	0900-2059	2100-0159	Service at	rrives KPA
	Direct services from each station to specified destination										First (no	Last (no earlier					First (no	Last (no earlier					First (no	Last (no earlier
	(excluding services to London terminals)	Destination	Total	Total	Total	TPH	Total	Total	Total	Total	later than)	than)	Total	TPH	Total	Total	later than)	than)	Total	TPH	Total	Total	later than)	than)
-	Milton Keynes		0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	-	-
. <u>e</u>	Bletchley	Kensington Olympia	0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	- 1	1 - 1
Ę	Leighton Buzzard	Kensington Olympia	0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	- 1	1 - 1
Ō.	Tring	Kensington Olympia	0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	- 1	1 - 1
E -	Berkhamsted	Kensington Olympia	0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	-	1 - 1
of gt	Hemel Hempstead	Kensington Olympia	0	2	1	1	10	1	1	2	08:15	23:15	0	1	15	0	08:30	20:15	0	0	0	0	-	1 - 1
Sir.	Watford Junction	Kensington Olympia	1	5	1	1	10	1	1	4	06:30	23:45	1	1	16	3	06:30	23:45	0	1	12	3	10:00	23:45
je je	Harrow & Wealdstone	Kensington Olympia	1	5	1	1	10	1	1	4	06:30	23:45	1	1	16	3	06:30	23:45	0	1	12	3	10:00	23:45
.0	Wembley Central	Kensington Olympia	1	5	1	1	10	1	1	0	06:30	21:15	0	1	15	0	07:30	20:00	0	0	0	0	-	1 - 1
1	Shepherd's Bush	Kensington Olympia	1	8	2	1	11	1	1	4	06:30	23:45	1	1	16	3	06:30	23:45	2	1	12	3	08:00	23:45

TABLE 4: S	SOUTHBOUND								nday to Frid										Satu									urday			
	Departure time from Kensington (Olympia)		Journe	y Time	0200-0659	0700-1559	0700-1559	1600-1859	1700-1759	1900-1959	2000-2059	2100-2159	2200-0159	Service le	aves KPA	Journe	y Time	0200-0659	0700-2159	0700-2159	2200-0159	Service le	eaves KPA	Journ	ey Time	0200-0859	0900-2059	0900-2059	2100-015	9 Service le	
					1										Last (no		l .						Last (no								Last (no
Dir	irect services from specified origin to each station				1									First (no	earlier		l .					First (no	earlier							First (no	earlier
	(excluding services from London terminals)	Origin	Average	Maximum	Total	TPH	Total	Total	Total	Total	Total	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)	Average	Maximum	Total	TPH	Total	Total	later than)	than)
West Bromp	pton	Kensington Olympia			1	1	14	6	3	1	1	1	4	06:30	23:45		l .	1	1	16	3	06:30	23:45			2	1	12	3	08:00	23:45
Imperial Wh	harf	Kensington Olympia			1	1	14	6	3	1	1	1	4	06:30	23:45		l .	1	1	16	3	06:30	23:45			2	1	12	3	08:00	23:45
: Clapham Ju	lunction	Kensington Olympia			1	1	14	6	3	1	1	1	4	06:30	23:45		l .	1	1	16	3	06:30	23:45			2	1	12	3	08:00	23:45
9		Shepherd's Bush	13	14	1	1	14	4	2	1	1	1	4	-	-	13	14	1	1	16	3	-	-	13	13	2	1	12	3	/ - /	- /
<u>(a)</u>		Watford Junction	41	44	1	1	11	3	1	1	1	1	4	-	-	41	42	1	1	16	3	-	-	39	41	0	1	12	3	4 - 7	- /
Ë		Milton Keynes	80	84	0	1	8	3	1	0	1	1	2	-	-	80	80	1	1	13	0	-	-	N/A	N/A	0	0	0	0	4 - 7	- /
	th Common	Kensington Olympia			0	1	9	4	2	1	0	0	0	08:15	19:15			1	1	15	0	06:30	20:15			0	0	0	0	- '	-
Balham		Kensington Olympia			0	1	9	4	2	1	0	0	1	08:15	23:15			1	1	15	1	06:30	23:15			0	0	0	0	- '	-
용 Streatham C	Common	Kensington Olympia			0	1	9	4	2	1	0	0	1	08:15	23:15		l .	1	1	15	1	06:30	23:15			0	0	0	0	- '	-
.⊊ Norbury		Kensington Olympia			0	1	9	4	2	1	0	0	1	08:15	23:15		l .	1	1	15	1	06:30	23:15			0	0	0	0	- '	-
Thornton He	leath	Kensington Olympia			0	1	9	4	2	1	0	0	1	08:15	23:15		l .	1	1	15	1	06:30	23:15			0	0	0	0	- '	-
Selhurst		Kensington Olympia			0	1	9	4	2	1	0	0	1	08:15	23:15		l .	1	1	15	1	06:30	23:15			0	0	0	1	- '	-
East Croyd		Kensington Olympia			0	1	9	4	2	0	0	0	1	08:15	23:15		l .	1	1	15	0	06:30	20:15			0	0	0	1	- '	-
Ē.		Shepherd's Bush	68	69	0	1	9	4	2	0	0	0	1	-	-	65	68	1	1	14	0	-	-	N/A	N/A	0	0	0	1		- /
		Watford Junction	40	45	0	1	9	3	1	0	0	0	1	-	-	37	39	1	1	14	0	-	-	N/A	N/A	0	0	0	1	/	4 - /
		Milton Keynes	107	108	0	1 1	9	3	1	0	0	0	1	_	-	104	105	1	1	13	0			N/A	N/A	0	0	0	0	/ - /	/ - /

#### Thameslink Core: Minimum Service Specification and First and Last Train Specification

T.D		THBOUND											0										$\overline{}$
IABLE	1: NOR I	IHBOUND											Service arrives de		1								
_							Monday to Fr							Saturday						Sunday			
Route Number	Opera	ration	Origin	Destination	Principal stations served (all trains serve London Blackfriars, City Thameslink, Farringdon, St Pancras International)	First (arriving at destination no later than)	Frequency between first and last trains	Last (arriving at destination no later than)	Average Journey Time	Maximum Journey Time	Exceptions	First (no later than)		Last (no earlier than)	Number of trains Timetable	Average Journey Tim	Maximum Journey Time Except	ons First (no later than)	Frequency between first and last trains		Average Journey Time	Maximum Journey Time	Exceptions
1		Mon-Sat	Brighton	Cambridge	Haywards Heath, Three Bridges, Gatwick Airport, East Croydon, London Bridge, Finsbury Park, Stevenage, Letchworth Garden City, Royston	08:00	Two trains per hour at half hourly intervals	22:30	135	142		08:00	Two trains per hour at half hourly intervals	01:00	36	133	133			NO SERVICE			
2		Daily	Tattenham Corner	Cambridge	Purley, East Croydon, London Bridge, Finsbury Park, Welwyn Garden City, Stevenage, Letchworth Garden City, Royston	08:30	Two trains per hour at half hourly intervals	01:00	137	140	Note 2	08:30	Two trains per hour at half hourly intervals	01:15	36	134	134	10:30	Two trains per hour at half hourly intervals	00:30	151	151	
3		Daily	Horsham	Peterborough	Three Bridges, Gatwick Airport, Redhill, Purley, East Croydon, London Bridge, Finsbury Park, Stevenage	08:00	Two trains per hour at half hourly intervals	01:30	166	172		08:30	Two trains per hour at half hourly intervals	01:30	36	164	164	10:00	Two trains per hour at half hourly intervals	00:30	164	164	Note 11
4	Mo	on-Fri AM Peak	Caterham	Finsbury Park	Purley, East Croydon, London Bridge	07:45	Six trains provided at half hourly intervals	10:00	56	56	Note 3			NO SERVICE						NO SERVICE			
4	Mor	on-Fri PM Peak	Caterham	Welwyn Garden City	Purley, East Croydon, London Bridge, Finsbury Park, Potters Bar, Hatfield	16:45	Two trains per hour at half hourly intervals	19:30	82	83				NO SERVICE						NO SERVICE			
5		Daily	Brighton	Bedford	Haywards Heath, Three Bridges, Gatwick Airport, East Croydon, London Bridge, St Albans, Luton Airport Parkway, Luton	07:30	Two trains per hour at half hourly intervals	00:45	134	168	Note 4	08:00	Two trains per hour at half hourly intervals	00:45	35	133	133	08:45	Two trains per hour at half hourly intervals	02:00	147	157	
6		Daily	Gatwick Airport	Bedford	Redhill, Purley, East Croydon, London Bridge, St Albans, Luton Airport Parkway, Luton	Note 1	Two trains per hour at half hourly intervals	Note 1	106	131		Note 1	Two trains per hour at half hourly intervals	Note 1		105	109	10:30	Two trains per hour at half hourly intervals	21:45	116	125	Notes 1 & 12
7	Mo	on-Fri AM Peak	East Grinstead	West Hampstead TL	Oxted, East Croydon, London Bridge	07:15	Seven trains provided at half hourly intervals	09:45	78	79	Note 5			NO SERVICE			· ·			NO SERVICE			
7A	Mor	on-Fri PM Peak	East Grinstead	Bedford	Oxted, East Croydon, London Bridge, St Albans, Luton Airport Parkway, Luton	18:30	Two trains per hour at half hourly intervals	20:45	122	122	Note 6			NO SERVICE						NO SERVICE			
8	Mo	on-Fri AM Peak	Littlehampton	West Hampstead TL	Worthing, Hove, Haywards Heath, Gatwick Airport, East Croydon, London Bridge	07:45	Six trains provided at half hourly intervals	09:00	126	130	Note 7			NO SERVICE						NO SERVICE			
9		Daily	Sutton	St Albans	Wimbledon, Tulse Hill, Herne Hill, Elephant & Castle, Kentish Town, West Hampstead Thameslink	07:45	Two trains per hour at half hourly intervals	23:00	96	99	Г	07:45	Two trains per hour at half hourly intervals	00:00	34	95	95	09:45	Two trains per hour at half hourly intervals	22:30	91	91	Note 13
10		Daily	Sutton	Luton	Mitcham Junction, Tulse Hill, Herne Hill, Elephant & Castle, Kentish Town, West Hampstead Thameslink, St Albans, Luton Airport Parkway	08:15	Two trains per hour at half hourly intervals	22:30	96	96	Note 8	07:45	Two trains per hour at half hourly intervals	00:00	34	96	96	09:45	Two trains per hour at half hourly intervals	21:30	N/A	N/A	Note 14
11		Daily	Sevenoaks	Kentish Town	Swanley, Bromley South, Peckham Rye, Denmark Hill, Elephant & Castle	09:30	Two trains per hour at half hourly intervals	23:15	77	77	Note 9	07:00	Two trains per hour at half hourly intervals	00:45	37	76	76	07:30	Two trains per hour at half hourly intervals	23:15	N/A	N/A	Note 15
11A	Mo	on-Fri AM Peak	Sevenoaks	St Albans	Swanley, Bromley South, Peckham Rye, Denmark Hill, Elephant & Castle, West Hampstead Thameslink	07:15	Two trains per hour at half hourly intervals	09:00	99	101			İ	NO SERVICE						NO SERVICE	•		
11B	Mor	on-Fri PM Peak	Sevenoaks	Luton	Swanley, Bromley South, Peckham Rye, Denmark Hill, Elephant & Castle, West Hampstead Thameslink, St Albans, Luton Airport Parkway	17:15	Two trains per hour at half hourly intervals	20:30	118	118				NO SERVICE						NO SERVICE			
12	Mor	on-Fri Mornings	Maidstone East	West Hampstead TL	Swanley, Bromley South, Elephant & Castle	07:45	Two trains per hour at half hourly intervals	12:00	91	92				NO SERVICE						NO SERVICE			
12	Mor	on-Fri PM Peak	Maidstone East	London St Pancras	Swanley, Bromley South, Elephant & Castle		Four trains provided to arrive between 1800 and 2000		81	81	Note 10			NO SERVICE						NO SERVICE			

- Note 1: Services on Route 6 shall operate 24 hours a day and as such there must be a minimum of one train per hour, and a maximum interval of 80 minutes between consecutive departures to Bedford from London St Pancras, throughout the period 00:45 and 06:15, with the exception of Saturday right / Sunday morning when no service is required. These services need not call at Redfill or Purley.

  Note 3: Two trains shall be extended to Wileyin Cardian City.

  Note 5: Two trains from the exception of Saturday right / Sunday morning when no service is required. These services need not call at Redfill or Purley.

  Note 5: Two trains shall be extended to Netfort or call at Redfill or Purley.

  Note 5: Two trains or call be extended to Netfort or call at Redfill or Purley.

  Note 5: Two trains shall be extended to Netfort or call at Redfill or Purley.

  Note 5: Two trains shall be extended to Netfort or call at Redfill or Purley.

  Note 6: Trains exervices shall be extended to Redfort or call at Redfill or Purley.

  Note 7: Two services shall be extended to Redfort or call at Redfill or Purley.

  Note 7: Two services shall be extended to Redfort or call at Redfill or Purley.

  Note 7: Two services shall be extended to Redfort or call at Redfill or Purley.

  Note 9: A time services from the standard to Redfort or call at Redfill or Purley.

  Note 9: A time services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Two services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Purley.

  Note 10: Services shall be extended to Redfort or call at Redfill or Pur

TABLE 2: SOUTHBOUND				Service arrives destination station																			
					Monday to Fi							Saturday					Sunday						
Route Numb	Oper	eration	Origin	Destination	Principal stations served (all trains serve St Pancras International, Farringdon, City Thameslink, London Blackfriars)	First (arriving at destination no later than)	Frequency between first and last trains	Last (arriving at destination no later than)	Average Journey Time	Maximum Journey Time	Exceptions	First (no later than)	Frequency between first and last trains	Last (no earlier than)		Average Journey Time	Maximum Journey Time Exception	First (no later than)	Frequency between first and last trains		erage ey Time Jo	Maximum lourney Time Exce	otions
1		Mon-Sat	Cambridge	Brighton	Royston, Letchworth Garden City, Stevenage, Finsbury Park, London Bridge, East Croydon, Gatwick Airport, Three Bridges, Haywards Heath	08:15	Two trains per hour at half hourly intervals	23:00	131	136		08:15	Two trains per hour at half hourly intervals	00:30	34	128	128		NO	SERVICE			
2		Daily	Cambridge	Tattenham Corner	Royston, Letchworth Garden City, Stevenage, Welwyn Garden City, Finsbury Park, London Bridge, East Croydon, Purley	08:15	Two trains per hour at half hourly intervals	00:00	134	135		07:45	Two trains per hour at half hourly intervals	01:00	36	131	134	09:00	Two trains per hour at half hourly intervals	23:45	152	152	
3	:	Daily	Peterborough	Horsham	Stevenage, Finsbury Park, London Bridge, East Croydon, Purley, Redhill, Gatwick Airport, Three Bridges	07:45	Two trains per hour at half hourly intervals	01:00	160	163		08:15	Two trains per hour at half hourly intervals	01:00	35	159	159	09:00	Two trains per hour at half hourly intervals		165	165 No	a 10
	Mo	Ion-Fri AM Peak	Welwyn Garden City	Caterham	Hatfield, Potters Bar, Finsbury Park, London Bridge, East Croydon, Purley	07:00	Nine trains provided at half hourly intervals	10:30	83	85	Note 2		1	NO SERVICE						SERVICE			
'	Mo	Ion-Fri PM Peak	Welwyn Garden City	Caterham	Hatfield, Potters Bar, Finsbury Park, London Bridge, East Croydon, Purley	17:00	Six trains provided at half hourly intervals	19:00	83	85	Note 3		1	NO SERVICE					NO	SERVICE			$\overline{}$
Ę		Daily	Bedford	Brighton	Luton, Luton Airport Parkway, St Albans, London Bridge, East Croydon, Gatwick Airport, Three Bridges, Haywards Heath	08:15	Two trains per hour at half hourly intervals	00:30	140	169		07:45	Two trains per hour at half hourly intervals	00:30		136	136	09:00	Two trains per hour at half hourly intervals	00:15	148	161	
6	i	Daily	Bedford	Gatwick Airport	Luton, Luton Airport Parkway, St Albans, London Bridge, East Croydon, Redhill, Purley	Note 1	Two trains per hour at half hourly intervals	Note 1	110	133		Note 1	Two trains per hour at half hourly intervals	Note 1		107	114	10:15	Two trains per hour at half hourly intervals	21:15	113	133 Nr	te 11
7	Mo	Ion-Fri PM Peak	West Hampstead TL	East Grinstead	London Bridge, East Croydon, Oxted	17:00	Two trains per hour at half hourly intervals	19:15	74	75	Note 4			NO SERVICE						SERVICE			$\overline{}$
7.	A Mo	Ion-Fri AM Peak	Bedford	East Grinstead	Luton Airport Parkway, St Albans, London Bridge, East Croydon, Oxted	07:00	Nine trains may be provided at half hourly intervals	08:15	106	106	Note 5			NO SERVICE						SERVICE			
8	Mo	Ion-Fri PM Peak	West Hampstead TL	Littlehampton	London Bridge, East Croydon, Gatwick Airport, Haywards Heath, Hove, Worthing	18:15	Five trains provided at half hourly intervals	20:00	120	122			1	NO SERVICE					NO	SERVICE			
9		Daily	St Albans	Sutton	West Hampstead Thameslink, Kentish Town, Elephant & Castle, Herne Hill, Tulse Hill, Wimbledon	07:15	Two trains per hour at half hourly intervals	00:30	94	95	Notes 6 & 7	07:45	Two trains per hour at half hourly intervals	00:00		93	93	10:15	Two trains per hour at half hourly intervals	22:00	92	92	
1	0	Daily	Luton	Sutton	Luton Airport Parkway, St Albans, West Hampstead Thameslink, Kentish Town, Elephant & Castle, Herne Hill, Tulse Hill, Mitcham Junction	06:45	Two trains per hour at half hourly intervals	00:30	94	96	Note 8	07:45	Two trains per hour at half hourly intervals	00:00		93	93	10:15	Two trains per hour at half hourly intervals	21:30	N/A	N/A No	e 12
1	1	Daily	Kentish Town	Sevenoaks	Elephant & Castle, Denmark Hill, Peckham Rye, Bromley South, Swanley	06:30	Two trains per hour at half hourly intervals	00:30	74	74	Note 9	07:00	Two trains per hour at half hourly intervals	00:45		74	74	08:45	Two trains per hour at half hourly intervals	00:30	N/A	N/A No	e 13
- 11	, Mo	Ion-Fri AM Peak	St Albans	Sevenoaks	West Hampstead Thameslink, Elephant & Castle, Denmark Hill, Peckham Rye, Bromley South, Swanley	08:30	Two trains per hour at half hourly intervals	10:45	102	102		-		NO SERVICE			•			SERVICE			
- 1 "	^ Mo	Ion-Fri PM Peak	St Albans	Sevenoaks	West Hampstead Thameslink, Elephant & Castle, Denmark Hill, Peckham Rye, Bromley South, Swanley	19:00	Two trains per hour at half hourly intervals	21:15	111	111				NO SERVICE						SERVICE			
1	2 Mc	Ion-Fri Afternoon	West Hampstead TL	Maidstone East	Elephant & Castle, Bromley South, Swanley	16:30	Two trains per hour at half hourly intervals	20:45	89	89			1	NO SERVICE					NO	SERVICE			

- Note 1: Services on Route 6 shall apparate 24 hours a day and as such there must be a minimum of ner train per householders of the between the period 00:30 and 04:00, with the exception of Saturday night / Sunday morning when no service is required. These services need not call at Purley or Redhill.
  Note 2: Services arriving at London SP pancas between 15:00 and 16:59 may start from Einstury Park.
  Note 3: Services arriving at London SP pancas between 15:00 and 16:59 may start from Einstury Park.
  Note 4: One train may start from Einstury Park and omit call at West Hampstead Thamestink.
  Note 5: There is the services may start from Einstury Park and omit call at West Hampstead Thamestink.
  Note 6: There is the services may start from Einstury Park and omit call at West Hampstead Thamestink.
  Note 6: An interval of up to 45 minutes is permitted between the penultimate service and the last service.
  Note 6: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton, calling at Luton Airport Parkway.
  Note 7: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton Airport Parkway.
  Note 8: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton Airport Parkway.
  Note 9: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton Airport Parkway.
  Note 9: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton Airport Parkway.
  Note 10: Services arriving at London SP pancas between 07:00 and 09:29 and of 19:59, shall start from Luton Route 11:
  Note 11: Sunday services may can be considered to ported to exceed to excee



# Thameslink, Southern and Great Northern Franchise

Train Service Requirements: Capacity Tables

May 2014

# **CR1 Morning Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras	108	40
London Bridge	Trains arriving from the south	28	0
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	280	128
Kings Cross	Any train	280	140
Moorgate	Any train	186	72

# **CR1 Evening Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from St Pancras and continuing south towards Elephant & Castle	128	44
London Bridge	Trains departing to the south	44	12
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	268	108
Kings Cross	Any train	274	114
Moorgate	Any train	189	72

# **CR2 Morning Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras	236	96
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	280	128
Kings Cross	Any train	280	140
Moorgate	Any train	186	72

## **CR2 Evening Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from St Pancras and continuing south towards Elephant & Castle	228	72
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	268	108
Kings Cross	Any train	274	114
Moorgate	Any train	189	72

**CR3 Morning Peak** 

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras	236	96
London Bridge*	Trains arriving from the south	550*	214*
Victoria	Any train	719	310
Kensington Olympia	Trains arriving from the south	52	21
Kensington Olympia	Trains arriving from the north	52	13
Brighton	Trains arriving from the London Road direction or the Hove direction, but not from the Preston Park direction, and not counting trains that run only between Brighton and Hove	82	32
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	280	128
Kings Cross	Any train	280	140
Moorgate	Any train	186	72

<sup>\*</sup> The stated requirements include a minimum of three ten-car trains arriving in London Bridge from Uckfield between 0700 and 0959, with two of those trains arriving between 0800 and 0859, and a minimum of 40 vehicles arriving in London Bridge from Uckfield must be provided between 0700 and 0959.

## **CR3 Evening Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from St Pancras and continuing south towards Elephant & Castle	228	72
London Bridge**	Trains departing to the south	494**	194**
Victoria	Any train	750	280
Kensington Olympia	Trains departing to the south	30	10
Kensington Olympia	Trains departing to the north	25	10
Brighton	Trains departing in the London Road direction or the Hove direction, but not the Preston Park direction, and not counting trains that run only between Brighton and Hove	75	26
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	268	108
Kings Cross	Any train	274	114
Moorgate	Any train	189	72

<sup>\*\*</sup> The stated requirements include a minimum of two ten-car trains departing London Bridge for Uckfield between 1600 and 1859, with one of those trains departing between 1730 and 1829, and a minimum of 34 vehicles must depart from London Bridge to Uckfield between 1600 and 1859.

**CR4 Morning Peak** 

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras	276	96
London Bridge*	Trains arriving from the south	605* \$ +	236* \$ +
Victoria	Any train	740	310
Kensington Olympia	Trains arriving from the south	52	21
Kensington Olympia	Trains arriving from the north	52	13
Brighton	Trains arriving from the London Road direction or the Hove direction, but not from the Preston Park direction, and not counting trains that run only between Brighton and Hove	85	34
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	292	128
Kings Cross	Any train	280	140
Moorgate	Any train	186	72

<sup>\*</sup> The stated requirements include a minimum of three ten-car trains arriving in London Bridge from Uckfield between 0700 and 0959, with two of those trains arriving between 0800 and 0859, and a minimum of 40 vehicles arriving in London Bridge from Uckfield must be provided between 0700 and 0959.

<sup>\$ -</sup> The requirements specified for London Bridge in columns (3) and (4) shall be increased by 32 vehicles and 24 vehicles respectively, applicable from the Passenger Change Date in January 2017, subject to completion by Network Rail of the necessary infrastructure improvements at Bermondsey.

<sup>+</sup> The requirements for London Bridge may be reduced in the event that the Franchisee, having used all reasonable endeavours, is unable to secure the necessary rights to provide the specified level of capacity into London Bridge as a consequence of Network Rail's infrastructure works.

#### **CR4** Evening Peak

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from St Pancras and continuing south towards Elephant & Castle	260	80
London Bridge**	Trains departing to the south	560** \$ +	216** \$ +
Victoria	Any train	752	280
Kensington Olympia	Trains departing to the south	33	11
Kensington Olympia	Trains departing to the north	27	10
Brighton	Trains departing in the London Road direction or the Hove direction, but not the Preston Park direction, and not counting trains that run only between Brighton and Hove	86	33
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	272	108
Kings Cross	Any train	274	118
Moorgate	Any train	189	72

<sup>\*\*</sup> The stated requirements include a minimum of two ten-car trains departing London Bridge for Uckfield between 1600 and 1859, with one of those trains departing between 1730 and 1829, and a minimum of 34 vehicles must depart from London Bridge to Uckfield between 1600 and 1859.

<sup>\$ -</sup> The requirements specified for London Bridge in columns (3) and (4) shall be increased by 24 vehicles and 16 vehicles respectively, applicable from the Passenger Change Date in January 2017, subject to completion by Network Rail of the necessary infrastructure improvements at Bermondsey.

<sup>+</sup> The requirements for London Bridge may be reduced in the event that the Franchisee, having used all reasonable endeavours, is unable to secure the necessary rights to provide the specified level of capacity out of London Bridge as a consequence of Network Rail's infrastructure works.

#### **CR5 Morning Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars %	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras @	200 %	80 %
London Bridge ^ %	Trains arriving from the south	713 ^ %	276 ^ %
Victoria	Any train	740	310
Kensington Olympia	Trains arriving from the south	52	21
Kensington Olympia	Trains arriving from the north	52	13
Brighton	Trains arriving from the London Road direction or the Hove direction, but not from the Preston Park direction, and not counting trains that run only between Brighton and Hove	85	34
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	292	128
Kings Cross	Any train	280	140
Moorgate	Any train	186	72

<sup>^</sup> The stated requirements include a minimum of three ten-car trains arriving in London Bridge from Uckfield between 0700 and 0959, with two of those trains arriving between 0800 and 0859, and a minimum of 40 vehicles arriving in London Bridge from Uckfield must be provided between 0700 and 0959. % The requirements specified for London Bridge in columns (3) and (4) may be reduced by up to 84 vehicles and 36 vehicles respectively, so long as the reduction is offset by an equivalent increase in the requirements specified in those columns for Blackfriars.

<sup>@</sup> Trains arriving from Maidstone East and terminating at Blackfriars may also be counted towards the requirements specified in this row.

## **CR5** Evening Peak

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars %%	Trains arriving from St Pancras and continuing south towards Elephant & Castle @	192 %%	64 %%
London Bridge ^^ %%	Trains departing to the south	664 ^^ %%	260 ^^ %%
Victoria	Any train	752	280
Kensington Olympia	Trains departing to the south	33	11
Kensington Olympia	Trains departing to the north	27	10
Brighton	Trains departing in the London Road direction or the Hove direction, but not the Preston Park direction, and not counting trains that run only between Brighton and Hove	86	33
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	272	108
Kings Cross	Any train	274	118
Moorgate	Any train	189	72

<sup>^^</sup> The stated requirements include a minimum of two ten-car trains departing London Bridge for Uckfield between 1600 and 1859, with one of those trains departing between 1730 and 1829, and a minimum of 34 vehicles must depart from London Bridge to Uckfield between 1600 and 1859.

<sup>%%</sup> The requirements specified for London Bridge in columns (3) and (4) may be reduced by up to 92 vehicles and 36 vehicles respectively, so long as the reduction is offset by an equivalent increase in the requirements specified in those columns for Blackfriars.

<sup>@</sup> Trains departing for Maidstone East, originating at Blackfriars, may also be counted towards the requirements specified in this row.

## **CR6 Morning Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0700 and 0959 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of vehicles scheduled to arrive at the station specified in column (1) between 0800 and 0859 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from the Elephant & Castle direction and continuing at least as far as St Pancras	192	64
London Bridge \$	Trains arriving from the south	921 \$	374 \$
Victoria	Any train	825	322
Kensington Olympia	Trains arriving from the south	52	21
Kensington Olympia	Trains arriving from the north	52	13
Brighton	Trains arriving from the London Road direction or the Hove direction, but not from the Preston Park direction, and not counting trains that run only between Brighton and Hove	85	34
St Pancras	Trains arriving from the Midland Main Line and continuing south at least as far as Blackfriars	468	168
Kings Cross or St Pancras	Any train arriving from the East Coast Main Line	400	164
Moorgate	Any train	222*	84*

<sup>\$</sup> The stated requirements include a minimum of three ten-car trains arriving in London Bridge from Uckfield between 0700 and 0959, with two of those trains arriving between 0800 and 0859, and a minimum of 40 vehicles arriving in London Bridge from Uckfield must be provided between 0700 and 0959.

\* The stated requirements for Moorgate in columns (3) and (4) apply from the December 2018 timetable change date and may be reduced by up to 36 vehicles and 12 vehicles respectively between May and December 2018.

## **CR6 Evening Peak**

Column (1)	Column (2)	Column (3)	Column (4)
Specified station	Specified description	Minimum number of vehicles scheduled to depart from the station specified in column (1) between 1600 and 1859 on weekdays (other than bank holidays), being vehicles forming part of trains that meet the description specified in column (2)	Minimum number of trains scheduled to depart from the station specified in column (1) between 1730 and 1829 on weekdays (other than bank holidays), being vehicles forming part of trains meeting the description specified in column (2)
Blackfriars	Trains arriving from St Pancras and continuing south towards Elephant & Castle	192	64
London Bridge \$\$	Trains departing to the south	880 \$\$	316 \$\$
Victoria	Any train	809	302
Kensington Olympia	Trains departing to the south	33	11
Kensington Olympia	Trains departing to the north	27	10
Brighton	Trains departing in the London Road direction or the Hove direction, but not the Preston Park direction, and not counting trains that run only between Brighton and Hove	86	33
St Pancras	Trains arriving from Blackfriars and continuing north onto the Midland Main Line	456	160
Kings Cross or St Pancras	Any train departing towards the East Coast Main Line	396	144
Moorgate	Any train	222*	84*

<sup>\$\$</sup> The stated requirements include a minimum of two ten-car trains departing London Bridge for Uckfield between 1600 and 1859, with one of those trains departing between 1730 and 1829, and a minimum of 34 vehicles must depart from London Bridge to Uckfield between 1600 and 1859.

<sup>\*</sup> The stated requirements for Moorgate in columns (3) and (4) apply from the December 2018 timetable change date and may be reduced by up to 33 vehicles and 12 vehicles respectively between May and December 2018.